Synergies of Innovation: Proceedings of NCSTEM 2023

Editors: Meenakshi M. Pawar, Swati P. Pawar, Dipti A. Tamboli, Jyoti S. Shinde

SVERI's College of Engineering Pandharpur



Editors

Meenakshi M. Pawar SVERI's College of Engineering, Pandharpur Swati P. Pawar SVERI's College of Engineering, Pandharpur

Dipti A. Tamboli SVERI's College of Engineering, Pandharpur Jyoti S. Shinde SVERI's College of Engineering, Pandharpur

ISBN 978-81-962882-7-3

ISBN 978-81-962882-8-0 (eBook)

© The Editor(s) (if applicable) and The Author(s), under exclusive license to SVERI's College of Engineering, Pandharpur. This work is subject to copyright. All rights are solely and exclusively licensed by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed. The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use. The publisher, the authors, and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, expressed or implied, with respect to the material contained herein or for any errors or omissions that may have been made. The publisher remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Contents-Computer Science and Engineering			
Sr. No	r. No Title of the Chapter Name of Authors		Page Number
1.	IOT Based Women Safety Application With Location Tracking Using Gps Module	Vimala Kulkarni, Sheetal Phulari	1
2.	GNDECB Campus Keeper	Ajit Patil Balaji Patil , Abhishek Puranik, Gururaj Surampalli	5
3.	Machine-Learning-Based Prediction of Stock Market Price	Reshma Mahadeo Patil, Swati P Pawar	12
4.	Implementing KNN For Movie Recommendation: An Exploration of Recommender System	Pramod D. Mane, Kanchan Iccharam Chouhan, Nimisha Dinesh Deval, Aishwarya Pawar	20
5.	Heart Disease Prediction Using Machine Learning	.P.T.Ingale, S.V. Saraf, R.K.Malgonde, G.S Misal	25
6.	Sonovisual: Transforming Image Description Into Dynamic Narratives	Vaishali Rajmane, Trupti Dhumal, Maina Awatade	30
7.	Fake Currency Detector	A.M.Dyade ,Atharv Davale, Rushikesh Waghule, Digvijay Shinde	34
8.	Revolutionizing Hematology: Deep Learning Approaches For Blood Cell Detection In Microscopic Images	Shradha S. Jadhav, Vanita D. Jadhav ,Lalit V. Patil	40
9.	Reversible Data Hiding In Encrypted Images Using Deep Neural Networks And Gan Model	Nagma Shaikh, Mohini Jagtap,Amruta Survase ³	
10.	Image Splicing	Unnati Dhaygude	54
11.	Revolutionizing Motorcycle Safety: Smart Airbag Systems For Rider Protection With Integrated Alarm Technology	Priyanka Lalasaheb Naikal ,. Priyanka Parmeshwar Deshmukh , Parmeshwar Suresh DeshmukhSarika Ashok Hajare	
12.	Automated News Crawling And Summarization: A Python-Based Approach	Rajeshwari Ghodake, Nupur Pawar, Samrudhi Masalekar, Sarika Hajare	66
13.	Location Based Medical Emergency Vehicles Booking Using Android	Mansab Shaikh, Vaishali Rajmane	70
14.	Optical Character Reader(OCR) Using Yolo And Tessreact For Text Extraction	Nimisha Deval, Sanika Patil, Gayatri Sudake	78
15.	Automatic Timetable Generator	P.A.Satarkar 'Nikita Tad, Nikita Pore, Dhanashree Lokhande,	81

Sr. No	Title of the Chapter Name of Authors		Page Number
16.	Virtual Classroom	S.S.Pawar, A.V.Patil, V.N.Jagdale, S.H.Bagal	85
17.	A Study On Security Risks In Mobile Inter-App Communication	S.W.Koli, S.M.Shinde S.P.Pawar	88
18.	Leaf Based Plant Disease Detection Using Convolution Neural Network	Rajshekar Gaithond Arpita, Amisha	94
19.	Interactive Context For A Review On Medical Store Website	D.P.Bhojane, V.S.Boragi, P.A.Pawar, G.S.Kusumade	104
20.	A Decision Support System Using Sensor-Cloud Based Smart Agricultural System For Smart Irrigation Scheduling	Malage Varsharani , Malage Rajshri . Pawar Minakshi , More Archana ⁴ . ,Jagtap Mohini,	109
21.	An Overview of BlockChain Technology	Sanjivani G Atakare ,Antosh M Dyade,Swati P Pawar	113
22.	Online Social Media Content Management System	Mihir Apte, Ajeet Lokhande, Rohan Mujumdar, Tejal Khadke	118
23.	Fire Eye: A Real-Time Forest Fire And Smoke Detection By Using Machine Learning	Sonali Wadekar, Mrs. V. D. Jadhav, Swati Pawar	124
24.	Data-Powered Voicematic Chatbot Conversation For Enhanced Learning	Mohini Jagtap, Archana More, Nagma Shaikh	132
25.	Inclusive Public Transportation: A Holistic Approach To Accessibility And User Experience For Visually Impaired Travelers	Atharva Hande, Shruti Deshpande, Vaibhavi Gawade, Harsh Ghodgaonkar, Jitendra Chavan	135
26.	Discover Your Strengths With Achieveai's NLP- Powered Platform	Malhar Gadade, Vinayak Jadhav, Om Dixit, Mr.Pramod Mane	139
27.	5G And Beyond Communication	Ujwala VijayKumar Bhagat, Vinayak M. Sale	143
28.	A Data-Driven Approach using Placement Assistance Systems	Shraddha B. Gavhane, Avinash K. Parkhe, Dr. S. P. Pawar	146

Contents-Electrical Engineering

Sr. No	Title of the Chapter	Name of Authors	Page Number	
29.	Adaptive Load Frequency Controller using Fuzzy Logic Technique for Parallel Resonant SIC Inverter in Surface Hardening Application	Vijay Patil, Anwar Mulla	153	
30.	A Fuzzy based Approach for Battery Controller for Microgrid	Mitkari Mohit M.,Bawage Ankita S., Mantri Lata R. Pate Pawan V.	159	
31.	Analysis of Electricity Consumption Using LSTM Model	Durga Prasad Ananth, T. Vinay Kumar Neelashetty K, Rupesh M	166	
32.	Mathematical Optimizing Solar Energy Utilization: Novel PCM-Integrated Solar Water Heating Systems	Nitin Morkane, Sarika Sakhare	172	
33.	Soldier's Uniform used for Temperature Control and Health Monitoring System based on IOT	Suresh Patil	177	
34.	Design and Development of Solar Hybrid Bicycle	Neelashetty K	183	
35.	Semi-automated Spacer Fitting Mechanism forHotline Maintenance	Vijay A Sawant, Dhanraj D Daphale, Sanjivani G Atakare	190	
36.	Analytical Approaches for Analysis of Sub- Synchronous Resonance (SSR) In PMSG based Wind Farm under Weak Grid: A Review	Prakash D. Kadam, Sampath Kumar Bodapatla, Sagar M. Ghodake, R. T. Bansode	196	
37.	Implementation of Arduino based Smart Management System for Household and Corporate Application	Dhanraj D Daphale, Vijay A Sawant , Shreya Moholkar	205	
38.	Quasi-Z-Source Solar Inverter fed BLDC Drive System	Shreya K. Sindol, Shruti S. Ankalgi, Prajakta S. Vedpathak, Priyanka S. Devmare' S.V. Moholkar	211	
39.	Smart Doormat with IOT-Enabled Visitor Detection	Mr. Shreays S. Kulkarni, Ms. S.V. Moholkar , Mr. D. D. Daphale, Mr. V. A.Sawant		
40.	Simulation and Analysis of Thyristor- Based Industrial Battery Charge Controller using MATLAB	Priyanka Ankush Pawar, Pallavi Suresh Bachute, Pradnya Sanjay Mane, D. A. Tamboli	223	
41.	Fault Investigation in High Voltage Transmission Line	Akshay Kale, Amit Jadhav, Ravikant Khandekar, Prof. H. M. Mallad	231	
42.	A Review of Innovations in EV Battery Charging Electrification	V. S. Biradar , A. A. Kotmale, M. T. Shinde ,S S Kulkarni	240	
43.	Automatic Load Sharing of Distribution Transformer using PLC	Pramod Korake, Harshwardhan Murade, Rushikesh Doke, Vikas Narale, Suhas B. Khadake, Aniket S Chavan	253	

Sr. No	Title of the Chapter	Name of Authors	Page Number
44.	Solar Outdoor Air Purifier with Air Quality Monitoring System	Seema S Landage, Sonali R Chavan, Pooja A Kokate, Sonal P Lohar, M. K. Pawar, Suhas B Khadake	260
45.	DTMF Based Irrigation Water Pump Control System	Prachi S Bhosale, Pallavi D Kokare, Dipali S Potdar, Shrutika D Waghmode, V A Sawant, Suhas B Khadake	267
46.	Development of a Test-Bench for Battery-Motor-Propeller Assemblies Designed for Multirotor Drone	R. N. Surwase, S. S. Lokhande, S. U. Patil, A. A. Shaikh, M. P. Thakre ⁵ , S. S. Kawade	274
47.	IOT-enabled Solar-based EV Charging Station	Dhanashri Dalave, Rutuja Sarvgod, Aishwarya Gaikwad3	288
48.	Pick and Place Robotic Vehicle with Soft Catching Gripper	G.D.Pawar, T.P.Suryawanshi, R.S.Nikam, D.D.Daphale	295
49.	Solar Powered Wireless Charging System	Mahek Patil, Monika Salunkhe, Karthika Nagane, ,Ranjana. N. Khandebharad, Dr. Dipti. A. Tamboli	302

Contents-Electronics and Telecommunication Engineering			
Sr. No	Title of the Chapter	Name of Authors Parallel Number Number	
50.	Baby Care Monitoring System	Rahul Pawar, Priti Tawashe, Anjali Manakoji,	310
51.	Smart Farm Hub	Sohel Sayyad, Disha Bhattad, Ashitosh Mane, Kenny Santani,Nikhil Gaikwad, Kajal Ingale, Dr.Meenakshi M Pawar, Amol Chounde	317
52.	Automatic Seed Sowing Robot	Adsul Abhishek, Atar Sahil, Divane Saurabha, Ms. S S Pawar	326
53.	Crop Monitoring System Using Wireless Sensor	Shivani Atkale, Aishwarya Bhosale, Dr.Nikhilesh k Mishra	329
54.	MIMO Technology - A Review	Mr.K.S.Deshpande, Dr.M.S.Mathpati, Dr.N.P.Kulkarni, Mr. R.B.Pawar	335
55.	Smart Seed Sowing Tractor Machine Using Arduino	Mr. P. S. Deshmukh1, Mrs. Priyanka P.Naikal-Deshmukh, Mr.R.B. Pawar	
56.	Conservation Of Biodiversity In River Ecosystem By Using Embedded System	Mr. S.A.Inamdar, Avishkar Ubale, Ritesh 344 n Chavan	
57.	IOT BASED SMART DUSTBIN	Ms.Suranjali B. Jagtap, Mrs.M.M.Pawar	350
58.	Handwritten Notes Recognition Using Machine Learning	Sakshi Rananavare, Swarali Joshi, Aishwarya Shinde , Seema Atole	356
59.	Adiabatic Technique Based Low Power 12t Static Ram For Data Communication Systems	Shilpa D B, Dr. Kishan Singh	361

Sr. No	Title of the Chapter	Name of Authors	Page Number	
60.	Flex grip: Enhancing Robotic Manipulation With Sensor-Enabled Gloves	Rahul Pawar, Jeevan Musale, Aditya Ghavate, Sandip Khatake, Shubham More	365	
61.	Anti-Theft Vehicle Tracking System	Shreyash Kamble , Amol Kore,Kunal Korade, Mr.A.A.Garad	370	
62.	Indian Currency Detection And Recognition From Image By Using MATLAB	Ms.Megha Arun Sontakke	376	
63.	Secure Attendance Tracking With Facial Recognition And Liveness Detection	Bhavana Bidarkar, Meer Hyder Ali, SK Tazim, MD Afnan Khan	382	
64.	Eco charge: Harnessing Solar And Traditional Power For Hybrid E-Vehicle Charging	Snehal Kadam, Swapnali Pawar, Kanchan Vyavahare and Nirmala Pujari	389	
65.	IoT Based Smart Irrigation System Using Arduino	Tushar Kore1, Rohit Sonar, Vishal B H, Premala Bhande	395	
66.	Online Voting System	Pratiksha Patil, Aakanksha Bhajibhakare, Heena Mulani and Manoj Deshmukh	402	
67.	Efficient Crowd Density Estimation Using Deep Learning	Ms. Shweta Jadhav, Ms.J.A. Kendule	407	
68.	Model For Retinal Disease Diabetic Retinopathy	Isharani, Vimala Kulkarni, Prerana, Amul	416	
69.	Design And Analysis Of Microstrip Array Antenna For IoT Applications	Dr.Gurudev Zalki, Dr.Mohammed Bakhar	420	
70.	Leaf Based Plant Disease Detection Using Convolution Neural Network	Rajshekar Gaithond, Arpita, Amisha	426	
71.	Suspicious Activity Tracking & Detecting Ai Camera	Aishwarya H, Namrata S, Md. Bakhar	436	
72.	Visualizing And Forecasting Stocks	Smita Gawade, Sunil Kadam, Yash Bagal, Adesh Kulkarni	442	
73.	Directivity Enhancement Of Multilayer Direct Feed Hexagonal Micro Strip Patch Antenna Array For Wireless Applications	Anil K Rathod, Dr.Md.Bakhar, Dr. S.R. Chougule, Amol C Bhosale		
74.	The Design Of Multiband Microstrip Patch Antenna Use Of Truncation And Slot Techniques For Wireless Application	Dr. Basavalinga Swami, Dr. Naganath Biradar	457	
75.	Unmanned Aerial Vehicles (UAVs) And IoT Integration In Agriculture: A Comprehensive Review Of Smart Farming Solutions For Precision And Efficiency Enhancement	466 Ms.Pooja Ausekar, Dr. Mrs.M.M.Pawar		

Sr. No	Title of the Chapter	Name of Authors	Page Number
76.	The Impact Of Text Preprocessing Techniques On Word Sense Disambiguation (WSD) In Natural Language Processing (NLP)	Somehekhar Aland, Rajivkumar Mente	475
77.	Sentiment Analysis Of COVID-19 Tweets: A Supervised Machine Learning Approach With NLP And Feature Extraction For Polarity And Subjectivity Classification	Parth Aland, Shivkumar R Raut, Ashok Shinde	483
78.	Gas Detection And Safety Management For Underground Mining Using Ardiuno	Ashish Biradar, Shubha S	492
79.	Soldier's Uniform Used For Temperature Control And Health Monitoring System Based On IoT	Suresh Patil	
80.	Digital Scarecrow	Rahul Pawar, Shashikant Kamble, Abhijit Khule, Shubham Kamble, Vishal Kulkarni	503
81.	A Review On Automatic Counting Of RBC And WBC In Blood Sample By Using Machine Learning	Shivani S. Jadhav, Jyoti A. Kendule, Vikram R. Chavan	510

	Business Administration			
Sr. No Title of the Chapter		Name of Authors	Page Number	
82.	Performance Analysis Of IPO's Issued During Third Quarter Of FY 2023-24	Dr. K.P.Galani	520	
83.	A Study On Identifying factors Affecting Pre- purchase Customer Dissatisfaction With Respect Tomoze Properties	Mr. Varshaket A. Wagh, M. M Bhore	526	
84.	Financial awareness survey among the students at Pandharpur's rural region	Pravin S. More, Komal P. Kondubhairi, Sagar D. Sarik	532	
85.	A Study On The Technique Of Shivamurut Milk And Milk Products By Using Inventory Management	Komal P. Kondubhairi, Pravin S. More, Sagar D. Sarik, Amad A. Ahmed	535	
86.	Cashless Impact on Banking Customers: Trends, Challenges and Opportunities	Mahesh Kumar M, Gayatri Patil, Swati Kote, Urmila Kore	543	
87.	Unveiling Secrets: A Holistic Analysis of Threats Posed by Spam Messages, Junk Calls, Social Media Marketing, And Dark Web Exploitation Via Telegram App	Md. Jeelani, Kalyanrao K, Deepak M [,] Shadrak	549	

Sr. No	Title of the Chapter	Name of Authors	Page Number	
88.	Comparative Analysis of India's Digital Rupee with Asia Pacific countries	Jyoti Ainapur, Priyanka Karmalkar [,] Sagar Karmalkar	555	
89.	Digital Dynamics: Investigating the Impact of Social Media on Youth Well- being, Relationships, and Behavior	Mr.Sagar Dnyaneshwar Sarik Mrs. Komal Padmakr kondubhari, Mr. Amad Abdul Ahmed,Mr. Pravin Sudam More	569	
90.	An empirical view of skill gap analysis in retail industry in Pune city	Miss. Siddhi Budhuk, Dr. Nagesh S. Magar	575	
91.	The Neuroscience Of Public Relations With Respect To Higher Educational Institutions	Dr. Sangeeta Swamy, Dr. Vageesha Salanki Mathada	585	

COMPUTER SCIENCE AND ENGINEERING

IOT BASED WOMEN SAFETY APPLICATION WITH LOCATION TRACKING USING GPS MODULE

Vimala Kulkarni¹, Sheetal Phulari² Guru Nanak Dev Engineering College Bidar, Karnataka.India sheetalphulari07@gmail.com

Abstract— The project presents a wearable safety device for women using the Arduino. The purpose of this device is to safeguard women in the event they might face any danger. The device uses wireless sensor network to communicate and to send alerts to them. The GPS and GSM are used to share the used to share the user's location directly to the relevant authorities and saved contacts. The switch in the device work for sending manual alerts in case of emergency and as panic switch to get the shock, then the Buzzer will also activate along laser diode. Keywords: Women Safety, Safety using Ardunio, IOT Based Women Safety. **Keywords:** Arduino. GPS. GSM.

I. INTRODUCTION

In today's world, women safety has become a major issue as they can't step out of their house at any given time due to physical/ sexual abuse and a fear of violence. Even in the 21st century where the technology is rapidly growing and new gadgets were developed but still women's and girls are facing problems. Women are adept at mobilizing diverse groups for a common reason. They often work across ethnic, religious, political, and cultural divides to promote liberty. We are all aware of importance of women safety, but we must analyze that they should be properly protected. Women are not as physically fit as men; in an emergency situation a helping hand would be assistance for them.

The best way to curtail your probability of becoming a dupe of violent crime (robbery, sexual assault, rape, domestic violence) is to recognize, defense and look up resources to help you out of hazardous situation. Today's world, women safety has become a major issue as they can't step out of their house at any given time due to physical abuse and a fear of violence. Women are adept at mobilizing diverse groups for a common reason. They often work across ethnic, religious, political, and cultural divides to promote liberty.

II. SYSTEM ANALYSIS

Existing System: This method has many drawbacks as she need to unlock the mobile phone which may take more time during panic situations, and also these applications are installed only on smartphones with access to GPS and Web. **Proposed System:** The proposed system consists of Push button, ESP32 microcontroller and Wi-Fi module, GPS and GSM module, ESP32 Cam module.

III. PROBLEM IDENTIFICATION

India is one of the fastest economic development countries and it also has super powers, still there are many crimes against women. So we are designing a system to reduce these kinds of atrocities on women who are facing many problems such as women abuse, Harassment, etc.

IV. OBJECTIVES

Women safety devices using GPS and GSM module are designed to alert family members or authorities in case of danger. These devices typically consist of a microcontroller, GPS module, GSM module, and a switch or sensor. The switch or sensor is used to trigger the device when the woman senses danger.

A. GPS (global positioning system)

GPS stands for global positioning system, which provides unequalled accuracy and flexibility of positioning for navigation. The GPS provides continuous three-dimensional positioning 24 hrs. a day throughout the world. GPS provides accurate location and time information for an unlimited number of people in all weather, day and night, anywhere in the world.

V. BLOCK DIAGRAM

1



COMPONENTS USED

The Global Positioning System (GPS) is a satellite-based navigation system made up of a network of 24 satellites placed into orbit

- B. NEO-6M GPS Module with EPROM
- 5Hz position update rate.
- The cold start time of 38 s and Hot start time of 1 s.
- Configurable from 4800 Baud to 115200 Baud rates. (Default 9600).
- Super Sense ® Indoor GPS: -162 dBm tracking sensitivity.
- Support SBAS (WAAS, EGNOS, MSAS, GAGAN).
- Separated 18 x 18mm GPS antenna.
- C. GSM
- D. ULTRA SONIC SENSOR

Ultrasonic sensors work on a principle similar to radar or sonar which evaluates attributes of a target by interpreting the echoes from radio or sound waves respectively. Ultrasonic sensors generate high frequency sound waves and evaluate the echo which is received back by the sensor. Sensors calculate the time interval between sending the signal and receiving the echo to determine the distance to an object. This technology can be used for measuring: speed and direction

VI. HARDWARE REQUIREMENTS

A. Arduino UnoThe Arduino Uno is an open-source microcontroller board based on the Microchip ATmega328P micro controller (MCU) and developed by Arduino.cc.The microcontroller board is equipped with sets of digital and analog input/output (I/O) pins that may be interfaced to various expansion boards (shields) and other circuits



Fig 1: Arduino Uno

B. Regulated power supply

Every embedded system requires dc voltage and that will be 5v DC supply

So, the battery voltage needs to converted to 5V dc

Digital electronic devices need digital supply and we can get supply from regulated power supply block

Rectifier is used to protect the circuit from battery polarity reversal.

C. LCD Display

D. Buzzer

When a voltage is applied across the two electrodes, the piezoelectric material mechanically deforms due to the applied voltage. This movement of the piezo disk within the buzzer creates sound in a similar manner as the movement of the ferromagnetic disk in a magnetic buzzer or the speaker cone mentioned above

E. Oscillator

An electronic oscillator is an electronic circuit that produces a repetitive electronic signal, often a sine wave or a square wave.

PIC micro controller internally having 4mhz clock frequency.

We are giving the 20Mhz clock frequency as an external source for increasing the system performance

VII. SOFTWARE REQUIREMENTS

- Arduino IDE for compiling and dumping code into Microcontroller
- Express SCH for Circuit design.
- Proteus for hardware simulation
- .

VIII. ADVANTAGES & APPLICATIONS

ADVANTAGES:

- Compact in size
- Maintenance is very easy
- Effortless to use
- It can be used for security applications
- Pre registered mobile numbers can be easily changed

APPLICATIONS:

- Can be utilized for the security of ladies, kids, impaired and matured individuals.
- Can be utilized as a legitimate proof of wrongdoing with correct.
- Location data for indictment

IX. CONCLUSION

In conclusion, the IoT-based women safety application with GPS-based location tracking emerges as a potent solution for addressing safety concerns. The integration of GPS modules enhances real-time tracking, providing a reliable and efficient mechanism to monitor and respond to potential threats. This application empowers women with a tool that not only offers precise location data but also ensures swift assistance in emergencies. The seamless connectivity within the IoT framework enables quick communication and intervention, fostering a safer environment. Moreover, the application's user-friendly interface enhances accessibility, encouraging widespread adoption. As society grapples with safety challenges, embracing technology-driven solutions like this offers a promising path forward. By leveraging the power of IoT and GPS technology, this application stands as a tangible step towards fostering a secure and supportive ecosystem for women, ultimately contributing to the broader goal of ensuring their well-being and peace of mind.

X. REFERENCES

1. Dr.C K Gomathy, Article: A Semantic Quality of Web Service Information Retrieval Techniques Using Bin Rank, International Journal of Scientific Research in Computer Science Engineering and Information Technology (IJSRCSEIT) Volume 3 | Issue 1 | ISSN : 2456-3307, P.No:1563-1578, February-2018

- 2. Dr.C K Gomathy, Article: A Web Based Platform Comparison by an Exploratory Experiment Searching For Emergent Platform Properties, IAETSD Journal For Advanced Research In Applied Sciences,
- 3. Volume 5, Issue 3, P.No-213-220, ISSN NO: 2394-8442, Mar/2018
- 4. Dr.C K Gomathy, Article: A Study on the Effect of Digital Literacy and information Management, IAETSD Journal For Advanced Research In Applied Sciences, Volume 7 Issue 3, P.No-51-57, ISSN NO: 2279-543X,Mar/2018
- Dr.C K Gomathy, Article: A Semantic Quality of Web Service Information Retrieval Techniques Using Bin Rank A Cloud Monitoring Framework Perform in Web Services, International Journal of Scientific Research in Computer Science Engineering and Information Technology IJSRCSEIT | Volume 3 | Issue 5 | ISSN : 2456-3307,May-2018
- Dr.C K Gomathy, Article: Supply chain-Impact of importance and Technology in Software Release Management, International Journal of Scientific Research in Computer Science Engineering and Information Technology (IJSRCSEIT) Volume 3 | Issue 6 | ISSN : 2456-3307, P.No:1-4, July-2018
- 7. A Smart friendly IoT device for women safety with GSM and GPS location tracking Publisher: IEEE Authors N Saranya; R Aakash; K Aakash; K Marimuthu 20 January 2022 5th International Conference .
- 8. WOMEN SAFETY DEVICE USING IOT Dr.C K Gomathy,Ms.S.Geetha Sri Chandrasekharendra SaraswathiViswa Mahavidyalaya , Kanchipuram , International Journal of Scientific Research in Engineering and Management (IJSREM) Volume: 05 Issue: 10 | Oct 2021 ISSN: 2582-3930
- 9. Sriranjini R, "GPS and GSM based self defene system for women safety", Journel of Electrical and Electronic system.
- 10. Vikash Singh, Vinush Singh and Vinush kumavat, "women safety device based on GPS and GSM", International Journal of Engineering applied science and technology.
- 11. Shaista khanum and Tripti shah, "Self defence device with GSM alert and GPS tracking with fingerprint", International conference on Electronics Communication and Aerospace Technology.
- 12. Tripti Paul, Tanusri Dey, Sanjana Mukharje and Upama Bhattacharjee, "WeRsafe an android app for society", International Journal of emerging trends and technology on computer science.

GNDECB CAMPUS KEEPER

Ajit Patil¹, Balaji Patil², Abhishek Puranik³, Gururaj Surampalli⁴ ¹²³⁴ Department of Computer Science and Engineering Guru Nanak Dev Engineering College, Bidar, Karnataka, India

Abstract - Campus Keeper is a cutting-edge mobile application designed to revolutionize and streamline the maintenance processes on college campuses. With its user- friendly interface and advanced features, Campus Keeper empowers students, faculty, and staff to report maintenance issues seamlessly, ensuring a swift and efficient resolution to campus-related concerns. This innovative app goes beyond traditional issue reporting by providing real-time updates on the status of reported problems. Users can track the progress of their submissions, fostering transparent communication between the campus community and maintenance teams. The incorporation of personalized reminders ensures that both users and maintenance personnel stay on top of scheduled tasks, contributing to a more proactive and organized approach to campus upkeep.

Keywords-Maintenance, Remainder, Task, Administration

I. INTRODUCTION

In the fast-paced and dynamic landscape of higher education, maintaining a well-functioning and aesthetically pleasing campus environment is paramount. Enter "Campus Keeper," a revolutionary mobile application meticulously designed to streamline and transform college campus maintenance. This innovative solution redefines the way campuses manage and address maintenance issues, offering a comprehensive platform that ensures efficient communication, real-time updates, and proactive task management. Empowering the Campus Community: Campus Keeper serves as a user-friendly and empowering tool for students, faculty, and staff. With its intuitive interface, individuals can effortlessly report maintenance issues, creating a seamless connection between the campus community and maintenance teams [1]. This user-centric approach not only expedites issue resolution but also fosters a sense of shared responsibility for the well-being of the campus environment.

Real-Time Updates for Transparent Communication: What sets Campus Keeper apart is its ability to provide real-time updates on the status of reported issues. Users can track the progress of their submissions, ensuring transparency in communication [2]. This feature facilitates a dynamic feedback loop, allowing maintenance teams to relay information back to the campus communication in campus maintenance [4], Campus Keeper establishes a new standard for efficient and open communication in campus maintenance.

Personalized Reminders for Proactive Maintenance: Campus Keeper introduces a proactive approach to maintenance with personalized reminders. Users receive timely notifications, keeping them informed about scheduled tasks and upcoming maintenance activities. This proactive feature not only aids in preventing issues from escalating but also cultivates a culture of responsibility and mindfulness among the campus community. Faster, Efficient, and Centralized Communication: Centralization is at the core of Campus Keeper's success.

By consolidating issue reporting, updates, and reminders within a single platform, the app significantly accelerates communication between users and maintenance teams [8]. This streamlined communication process reduces response times, ensuring that maintenance issues are addressed swiftly and effectively.

Data-Driven Insights for Strategic Maintenance: Campus Keeper goes beyond immediate issue resolution by providing comprehensive analytics and reporting features. Administrators can harness these insights to identify recurring problems, allocate resources strategically [10], and implement preventive maintenance measures. The app empowers maintenance teams with the information they need to make data-driven decisions, resulting in a more resilient and well- maintained campus infrastructure.

RELATED WORK

As educational institutions embrace the digital era, there is a growing emphasis on leveraging technology to enhance campus management and maintenance. One notable related work is "Campus Connect Maintenance," a mobile application designed to facilitate streamlined communication between students, faculty, and maintenance teams. Similar to Campus Keeper, Campus Connect Maintenance offers a user- friendly platform for reporting issues, complete with real- time updates. However

Campus Keeper differentiates itself through its personalized reminders and proactive maintenance approach, providing a comprehensive solution that goes beyond issue resolution.

Another relevant solution is "Facility Sync University," a cloud-based campus management system that includes maintenance functionalities. This platform centralizes various campus-related tasks, from facility scheduling to maintenance ticketing. While Facility Sync University excels in providing an overarching management system, Campus Keeper stands out by focusing specifically on the intricacies of maintenance communication. The personalized reminders and data-driven insights within Campus Keeper create a unique value proposition, ensuring a holistic and user-centric approach to campus upkeep In the broader landscape, "Smart Campus Solutions" is an integrated suite of tools designed for efficient campus management. While it covers diverse aspects such as security, energy management, and maintenance, Campus Keeper stands out for its singular focus on revolutionizing the maintenance aspect. By concentrating on creating a seamless communication channel for maintenance tasks, Campus Keeper addresses a critical need in campus management that sets it apart from more generalized solutions like Smart Campus. This highlights the importance of tailored solutions in meeting the unique challenges of campus maintenance in educational institutions.

II. OBJECTIVES

Objective 1: - Develop a user-friendly interface: Create an intuitive and visually appealing user interface for the Maintenance and Reminder Application, ensuring accessibility and ease of navigation to enhance user experience and encourage widespread adoption.

Objective 2: - Transparent status tracking system. Implement a robust status tracking system within the application, providing users with real-time updates on the progress of their maintenance tasks. This transparency enhances accountability, allowing homeowners to monitor service provider activities seamlessly.

Objective 3: - Timely Maintenance Reminder. Incorporate a reliable reminder system that sends timely notifications to users, ensuring they are promptly alerted to upcoming maintenance tasks. This feature aims to prevent oversights and promote proactive home management, contributing to the overall efficiency of the maintenance process.

Maintenance Management.

Maintenance management is a critical aspect of ensuring the longevity and functionality of various systems, including those in residential properties. Research by Pantaloon and Parodi (2018) emphasizes the importance of implementing proactive maintenance strategies to reduce operational costs and increase the lifespan of equipment. Maintenance management software, such as Computerized Maintenance Management Systems (CMMS), has been extensively used in industrial settings to streamline maintenance processes.

Reminder Systems

Reminder systems have seen widespread application in various domains, from healthcare to personal task management. Research by Munshietal. (2020) discusses the effectiveness of reminder apps in enhancing compliance with medication regimens. In the context of home maintenance, reminder systems can be valuable for scheduling routine inspections, filter replacements, and other preventive tasks.

III. METHODOLOGY

Architectural Foundation:

The Maintenance and Reminder Application's system architecture is meticulously designed to establish a robust and scalable foundation. Embracing a microservices architecture lays the groundwork for modular and independent service components. This architectural choice enhances flexibility, allowing for easier maintenance, updates, and the seamless addition of new features. The decision to adopt Next.js as the frontend framework ensures a responsive and dynamic user interface, facilitating a smooth user experience. This architectural foundation sets the stage for a system capable of accommodating the evolving needs of users while maintaining a high level of performance.

Technological Stack:

The choice of technology stack plays a pivotal role in shaping the application's capabilities. The frontend, powered by Next.js, offers a modern and efficient user interface, while the backend is built on Node.js, known for its scalability and event-driven architecture. MongoDB serves as the primary database, providing a flexible and scalable solution for data storage. Prisma acts as the ORM (Object-Relational Mapping) tool, streamlining interactions with the database and simplifying data management.

This carefully selected technological stack ensures a cohesive and efficient synergy between different components, contributing to the overall robustness and performance of the application.

Database Management and Integration:

MongoDB, chosen as the primary database, allows for agile and schema-less data storage, accommodating the diverse and dynamic nature of maintenance and reminder data. Prisma, as the ORM tool, optimizes database interactions, simplifying queries and ensuring efficient data retrieval. This combination of MongoDB and Prisma establishes a responsive and scalable database management system integral to the application's functionality. Furthermore, the integration of blockchain technology enhances the security and transparency of transaction recording, adding an extra layer of data integrity to the system. This strategic integration ensures that the application not only meets the functional requirements but also upholds a high standard of data security and transparency.

Scalability and Future-Proofing:

The architecture's scalability is a key consideration in ensuring the application's longevity and adaptability. By adopting a microservices approach and leveraging technologies known for scalability, such as Node.js and MongoDB, the system is wellequipped to handle increased user loads and evolving requirements. The modular nature of microservices allows for the independent scaling of different components, providing a foundation that can grow seamlessly with the application's user base. This forward-looking approach to scalability and future-proofing ensures that the Maintenance and Reminder Application remains agile and responsive to the dynamic landscape of user needs and technological advancements.

IV. DEVELOPMENT TOOLS AND TECHNOLOGIES

The development process leverages industry-standard tools and technologies. Visual Studio Code is chosen as the primary integrated development environment (IDE) for its versatility and extensive plugin support. Git version control ensures collaborative development, and GitHub serves as the repository for source code management. Continuous integration and deployment (CI/CD) pipelines are implemented using platforms like Jenkins or GitHub Actions to automate the testing and deployment processes.

V.IMPLEMENTATION PROCESS

The implementation process follows an iterative and agile methodology. User stories are defined to capture functional requirements, and sprints are conducted to incrementally develop and test features. Next.js facilitates server-side rendering for enhanced performance, while Prisma ensures efficient interaction with the MongoDB database. Smart contracts on a blockchain network handle secure transaction, ensuring the integrity of user data

- Caur	er		
Gre	nate table		
(De)	ele lacha 💦 🌫		
	Psin and remind ev	ery 3 days or custom 🌫	
		Add task details	>
•	Gond task details (reminder) -		
은 User	TasksRemind er	C CronJobsApi	TaskDeta
	(the second sec	A. 1999	<u></u>

Fig 1: Sequence Diagram of reminders.



Fig 2: Sequence Diagram if Maintenance

VI. APPLICATION VISUALS

3	GNDECE					(‡ A
98	Dashboard	Dashboard				
AL	Statts	Durrium of Remaindure, Maintunance Statts				
Ą	Reminders	Reminders 💭	Maintenance	20	Staffs	恖
20	Maintanance	4 Veral Hammoorn	13 Total Matemanice		5 Total biolity	
ĉ	Profile					
(1)	Settingi	Reminders		Mainten	ance	
		Pending	1	Pending		3
		inprogress	t	Inprogress		3
		Completed	0	Completed		(#)
		Expried	1	Rejected		2
		and the second s				

Fig 3: Application Dashboard

The Application dashboard which given the stats of the Reminders and maintenance like pending, in progress, completed and expired tasks.

Generati				
Title	Status	Start Date	Due Date	Description
Deadline for Report Submission	@ Pending	Tue Oct 03 2023	Sun Nov 05 2023	
Tax Planning Meeting	() Inprogress	Sat Nov 11 2023	Set Nov 18 2023	Schedule a meeting with a tax
Stock Market Analysis	O Completed	Wed Nov 08 2025	Sat Sep 30 2023	Analyze the current state of th
Monthly Expense Report	Expired	Sun Oct 29 2023	Wed Sep 20 2023	Compile and submit the month

Fig 4: Reminders Records.

Structured representation of reminder tasks with their status which helps to visualize and maintain the reminders.

Aaintenances 😁				(- 4	aka Nangaan
Search					
Title	Turture	Requestori du	Taper	Calegory	
ELECTRICAL FAN	() Encyland	Nga	Mandanata	Institut	
Lala Bapalpromoto	() ingringrate	Admin	them :	Biactroat	
MBA Lais Installation	G Completer	Bala) Parti	No	Discussion	1
D2 Maintenance	© Eargebraid		11410 ·	Cint	
Name Province Rev. et al.	C. Constants	A.A	No.	and a	

Fig 5: Maintenances Records.



Fig 6: Maintenance Requests

The Maintenance requests which are requested by the staff. Requests are examined and made to accomplish their tasks.



Fig 7: Maintenance Categories

The maintenance is structured with multiple categories like Civil, electrical, system and software.

Heardh						15 View
tame	femail	Phone number	Gender	Rate	Dato	
Nerrin	admin@gmail.com	1234567890	MALE	SUPERADMIN	September 25th, 2023	
s) (t	0@gmail.com	8971860300	MALE	ADMIN	October 4th, 2023	<u>i</u> (11)
(yiz	xyz@gmail.com	8971860200	MALE	STATE	October Sth. 2023	
Aphronick	124@gmail.com	9019536879	MALE	ADMIN	October 9th: 2023	
latayi Patil	balajuati/1220@gmail.com	9353704589	MALE	ADMIN	October 11th, 2023	

Fig 8: Staff Management

The staff management module enables the addition of staff members with defined privileges and permissions, ensuring a streamlined and secure administrative framework within the Campus Keeper application. This feature facilitates efficient task delegation and enhances overall organizational control.

VII. CONCLUSION

In conclusion, the Maintenance and Reminder Application stands as a testament to the successful fusion of cutting-edge technology and thoughtful design principles. The adoption of a microservices architecture, with Next.js driving the frontend and Node.js powering the backend, establishes a flexible and scalable foundation. This architectural choice, combined with MongoDB as the primary database and Prisma as the ORM tool, ensures not only optimal performance but also a responsive and adaptable user interface.

VIII. REFERENCES

- 1. Spezie, M., & Bragantini, D. (2023). The Development of Task Management Software (TMS). PM World Journal, 2023.
- 2. Nagel, S., & Schwade, F. (2020). Analysis of Task Management in Virtual Academic Teams. Hawaii International Conference on System Sciences, 2020.
- 3. Hedaoo, G., et al. (2018). Online Task Management System (OTMS). IRE Journal, 2018.
- 4. Jyothi, N., & Parkavi, A. (2016). A Study on Task Management System. In International Conference on Research Advances in Integrated Navigation Systems (RAINS), 2016, (pp. pppp).
- 5. Haseena, V., & Shaheer, K. (2015). Project Management Systems: Challenges and IT Solutions. International Journal of Science and Research (IJSR), Volume 6, Issue 5, 2015.
- 6. Johnson, L., & Williams, M. (2019). Enhancing Collaborative Task Management through Cloud-Based Solutions. Journal of Information Technology Management, 2019.
- 7. Chen, W., & Liu, Y. (2017). Agile Task Management: A Comprehensive Review. IEEE Transactions on Software Engineering, 2017.
- 8. Patel, R., & Gupta, S. (2022). Task Management Systems: A Comparative Analysis of Features and User Satisfaction. International Journal of Computer Applications, Volume 11, Issue 3, 2022.
- 9. Li, Q., & Wang, J. (2014). Towards an Integrated Framework for Task Management and Collaboration. In International Conference on Software Engineering and Knowledge Engineering (SEKE), 2014.
- 10. Santos, A., & Oliveira, T. (2017). User Adoption of Task Management Apps: An Empirical Study. Computers in Human Behavior, Volume 76, 2017.

MACHINE-LEARNING-BASED PREDIC TION OF STOCK MARKET PRICES

Reshma Mahadeo Patil¹, Swati P Pawar² ¹SVERI's College of Engineering Pandharpur. ² SVERI's College of Engineering Pandharpur

Abstract - Investment corporations, hedge funds, and even individual investors must study financial models to understand market behavior and build profitable investments and trades. Investors usually produce educated guesses by analyzing information about old stock prices, the company's performance behavior, etc. The initial phase of revealing theories in the guesswork indicates that stock unit prices are entirely random and unpredictable. In the betterment of the guesswork, quantitative analysts get deployed to make prophetic models. The paper focuses on using machine-learning techniques to develop better models for enabling appropriate recommendations for financial investments.

Keywords: - Stock Price Prediction, Machine Learning, Random Forest Regression.

I. INTRODUCTION

The world's stock markets comprehend enormous wealth. As with the extended market, investors hunted for ways to amass data regarding the companies listed in the market. In the past, investors relied upon their expertise to spot market patterns, but this is not possible nowadays. Easily applied math analysis of financial information provides some insights. However, in recent years, investment firms have used numerous artificial intelligence (AI) systems to look for patterns in vast amounts of real-time equity and financial information. These systems support investment decision-making, and they have currently been used for a sufficiently long amount that their features and performance will be reviewed and analyzed to Identify those systems and improve prophetic performance compared with alternative techniques. When the prediction goes correct, the vendor and stock broker make enormous profits. Frequently when the prediction goes in unexpected ways it is expected by analyzing the history of several securities markets. Machine learning is economical, thanks to representing such processes. It predicts a market price value near the physical weight with increasing accuracy—the introduction of machine letups of research attributable to its economic and correct values measurements. Dataset is the important part of machine learning used in education. The information set ought to be as concrete as potential, resulting from which amendment within the data will uphold massive changes within the outcome. This project uses supervised machine learning on a dataset obtained from Yahoo Finance. This dataset has five variables: open, close, low, high, and volume. With nearly direct names, airy, compact, soft, and increased area units indicate different bid costs at other times. Throughout the fundamental measure, shares are passed from one owner to another. The test information is then used to develop a model. A regression model and an LSTM model are used to test this conjecture, one by one. During working hours, regression minimizes errors, and LSTM contributes to the cognitive process of information and result. Finally, yet importantly, graphs for the fluctuation of cost with dates (for the regression-based model) and between actual and expected prices (for the LSTM-based model) are planned.

Stock Market Prediction aims to predict the longer-term price of a corporation's money stocks. Market prediction technologies use machine learning to make predictions based on current exchange indices and coaching on previous values. By employing different models, machine learning creates more accurate and detailed forecasts. Our primary focus is on utilizing regression and LSTM machine learning techniques and developing a deep understanding of stock values. Several factors are considered, including the area of the unit, the low, the high, and the volume of stock values. This Paper introduces several techniques for calculating the prices like the R factor, Quantitative Analysis factor - The chance/praise ratio, often called the "R/R ratio," compares the capability income of a change to its capability loss. It is a calculation that uses the distinction between the access factor to locate praise. Quantitative Analysis - Quantitative evaluation (QA) in finance is a technique that emphasizes mathematical and statistical evaluation to assist decides the price of a monetary asset, along with an inventory or option. Quantitative buying and

selling analysts (additionally recognized as "quanta") use several data—including historical funding and inventory marketplace data—to increase buying and selling algorithms and pc models.

II. RELATED WORK

Stock price prediction can be predicted using AI and machine learning models in machine learning fields. It uses the SVM model for stock price prediction. Support vector machine which works on classification algorithms. It is used to get a new text as an output. Applying Multiple Linear Regression with Interactions to predict the trend in stock [1] Using data from stock markets around the globe, Beginner's checks whether the markets are efficient and whether there are any anomalies. Whenever a market anomaly is found, scholars first confirm the anomaly and then search for existing models to explain the anomaly. Suppose scholars are unable to estimate, evaluate, and forecast any model to explain the anomaly. In that case, scholars will use quantitative analysis, modeling, or even a new theory of information to explain the anomaly that led to Behavioral Finance. In the event of an unexplained anomaly, one may be able to exploit the monster in order to profit. Investors can get valuable investment advice this way, on the one hand [2] The real Gross Domestic Product reflects the relationship between the stock market and the economic activity of the five European countries: Germany, France, Italy, the Netherlands, and the UK. This analysis includes variables such as stock market returns, actual economic activity, and interest rates in addition to the variables commonly used in such analyses. In the empirical VAR model, the authors have included the composite leading indicator [3]. The weak-form potency and stochastic process behavior of the CIVETS stock markets throughout the amount 2002–2012. We tend to apply unit root tests, serial autocorrelation, and variance quantitative relation tests. Our unit root results imply that CIVETS follow a stochastic process [4]. To predict the stock value of NSE and securities markets, two leading stock markets worldwide, the authors use four-deciliter architectures. We tend to train four networks, MLP, RNN, LSTM, and CNN, with the stock value of TATA MOTORS from NSE. From the NSE stock exchange, the models were used to predict the stock values of MARUTI, HCL, and AXIS BANK, and from the securities market, BANK OF AMERICA (BAC) and CHESAPEAKE ENERGY (CHK). Based on the results obtained, it is clear that the models can describe the patterns found in each stock market [5]. The importance of predicting the securities exchange price is well known among financial specialists since they need to know what kind of return they will receive for their investments. Generally, specialized experts and intermediaries use chronicled prices, volumes, value designs, and basic patterns to predict stock costs. Stock value expectations today are even more baffling than before since the organization's money-related status, as well as the socio-practical state of the nation, political environment, and cataclysmic events, influence stock costs. [6].

III. PROPOSED SYSTEM

This paper introduced LSTM (Long Short-Term Memory) model in stream-lit, which will predict the values based on the old dataset. The Prediction values are High, Low, Open, and Close. It is a reliable application for students and beginners who want to trade. They can quickly identify the trends in the market, whether the market is going upward or downward, or else it will remain sideways. The model generates the confusion matrix for the classification report. This paper introduced the two regression and classification methods for stock market prediction. In the regression method, the closing price of company stock is predicted. The classification method will predict company stocks' closing price that will increase or decrease in upcoming days. Figure1 shows the proposed system design—this paper takes datasets from yahoo finance data. The first step is to train the data, and in the second step, data was tested, and with the LSTM model, forecast the values to get the prediction value.

The objective of the System

- 1. Explore Stock Prices.
- 2. Implement Basic Model Using Linear Regression
- 3. Implement LSTM Model with confusion matrix and classification report.

Proposed Architecture:



Fig.1. Proposed Architecture

Method of Implementation

1. R Factor

There are two types of equity market risks: systematic and non-systematic. Rising oil prices, currency movements, and changing government policies are familiar sources of frequent hazards. Unsystematic risks, however, are caused by factors unique to a company or industry. In addition, management and labor relations, increased competition, the entry of competing players, and customers' preference for a company's products all contribute to unsystematic risk.

2. Stock Analysis Candle Stick Chart

Candlestick charts show price movements of securities, derivatives, and currencies. As with a graph, each candle represents all four significant pieces of information for that day: open and close in the thick body; high and low in the wick. Two ways can be used to visualize buying and selling pressure using candlesticks.

3. LSTM Model

Long Short-Term Memory fashions are extraordinarily effective time-collection fashions. They can expect an arbitrary wide variety of steps into destiny. An LSTM module (or molecular) has five essential additives which permit it to version each long-time period and quick-time period data. Cell nation (ct) - This represents the inner reminiscence of the molecular, which shops each quick period of reminiscence and long-time period recollections Hidden nation (ht) - This is output nation records calculated w.r.t. modern enter, preceding remote country, and current molecular enter that you use to expect the destiny inventory marketplace prices. Additionally, the hidden nation can determine to handiest retrieve the short or long-time period or each variety of reminiscence saved withinside the molecular country to make the following prediction. Input gate (it) - Decides how many records from current enter flow to the molecular nation.

Forget gate (ft) - Decides how many records from the modern enter and the preceding molecular nation flows into the contemporary molecular country. Output gate (to) - Decides how many records from the modern molecular nation flow into the hidden government, so that if wanted, LSTM can handily select out the long-time period recollections or quick-time period recollections.

IV. MATHEMATICAL FORMULATION

Confusion Matrix is the visual illustration of the particular VS foretold values. It measures the performance of our Machine Learning classification model and appears sort of a table-like structure. This is. However, a Confusion Matrix of a binary classification downside sounds like

Precision: It may be outlined because of the range of correct outputs provided by the model or, out of all positive categories appropriately foretold by the model, what number of them were valid. It may be calculated as mistreatment by the below formula.

 $Precision = \frac{TP}{TP + FP}$

Recall: - It is outlined because the out of total positive categories, however our model foretold properly. The recall should be as high as doable

 $Recall = \frac{TP}{TP + FN}$

R factor

Mean: - In other words, it is by far the most common of the datasets within the diverse fields of arithmetic. As a result, if we take five numbers in statistics set, say 12, 13, 6, 7, 19, 21, the suggestion system would be

$x1 + x2 + x3 + \dots + xn$

n

Mode: - As a concept, mode refers to the number in a data set that is repetitive and occurs most frequently. The mode is also known as a modal value, which represents the highest number of occurrences in the group. A mode is also a value that represents the whole data collection, like mean and median. There may be more than one mode in a given data set in some cases, so it is imperative to keep this in mind. Bimodal data sets have two modes. As shown in the excel sheet, the mode can be calculated as follows:

Mode. SNGL (B1: B5)

Dataset Use: (<u>https://www.kaggle.com/achintyatripathi/eda-autoviz-class-one-line-code-yahoo-stock-price?scriptVersionId=42446951</u>)

Software requirement specification

Python

Stream lit web framework

Hardware requirement specification

> Laptop

RESULTS



Figure2. Actual & Predicted Result

Figure2 shows the actual & predicted result.

2. R Factor



Figure3. R factor

The above Fig 3 graph shows the close Price of NKE (Nike) company with actual value denoted in green color and compared with MA (Moving Average), which has MA53 days characterized with blue color and MA30 days is marked with red paint. The graph shows the total daily returns of NKE (Nike) company. On the y-axis, the value is denoted, and on the x-axis, the year is denoted.

3. Stock Analysis Candle Stick Chart



Figure4. Stock Analysis candle stick chart

In above Fig 4 we see the candle stick chart, which is compared with Boll (Bollinger Band) denoted with red color, RSI (Relative Strength Index) denoted by blue color, and volume by gray color. On the y-axis price of the stock is denoted, and on the x-axis, month and year are denoted.

1. Quantitative Analysis

A quantitative analyst is the only one who designs a complicated framework for monetary establishments that aids them in charging and exchanging securities within the financial market.

Quants may be of types:

Front workplace quant: These are those who immediately offer the dealer the charge of the monetary securities or the buying and selling tools.

Back workplace quant's - These quants are there to validate the framework and create new techniques after research. Following are the contents of the Quantitative Analysis of NKE (Nike) company.

- 1. Mean
- 2. Median
- 3. Mode
- 4. Maximum
- 5. Minimum
- 6. Variance



Figure 5. Quantative Analysis



Figure 6. Quantative Analysis pie chart

In above Figure 6 we see Variance value denoted in blue color, Maximum value in red color, Mean value in green color, Median value in violet color, Mode value in orange color and Minimum value in light blue color.

1. Forecasting of Values

Forecasting is used for the time series to predict the value of the stock with the terms of the day like - short term (5-10 days), medium term (20-100 days), and long term (200 days). NKE stock forecasts on the testing set, testing error 17.41%. The Below Graph Show the predicted value for the previous data.

The Graph Shows value up to 24 April 2022.



Figure7 forecasting of values

In above Figure 7 Predicted closing price is denoted with a blue color line, CL+ (closing positive) fee, which is above the predicted price and denoted with orange color and also, we see CL- (closing negative) price, which is below the expected price and denoted with orange paint.

1. Confusion Matrix

Classification	report : precision	recall	f1-score	support	
1 0	0.67 0.71	0.50 0.83	0.57 0.77	4 6	
accuracy macro avg weighted avg	0.69 0.70	0.67 0.70	0.70 0.67 0.69	10 10 10	

Figure8 Confusion Matrix

III.CONCLUSION

By training with a broader range of knowledge sets, we may be able to improve prediction. It is also possible to analyze specific business aspects within the forecast of varied shares. In this paper, we examine the various patterns of share prices in multiple sectors. The algorithm might analyze a graph with various periods to improve its accuracy. A framework such as this can assist in marketing research and predicting growth for other corporations over several years. Prediction accuracy may be enhanced by incorporating alternative parameters (e.g., capitalist sentiment, election outcome, and government stability).

IV.REFERENCES

1. C Osman Hegazy, Omar S. Soliman, Mustafa Abdul Salam, "A Machine Learning Model for Stock Market Prediction" International Journal of Computer Science and Telecommunications [Volume 4, Issue 12, December 2013].

- 2. Kai-Yin Woo, Chilin Mai, Michael McAleer, Wing-Keung Wong "Review on Efficiency and Anomalies in Stock Markets" 22 December 2019; Accepted: 4 March 2020; Published: 12 March 2020.
- 3. Boriss Seliverstov's, Manh Ha Duong "On the role of stock market for real economic activity" JUNE 9 2006.
- 4. Fahad Almudhaf, Yaser A. AlKulaib OCTOMBER 2012
- 5. "ARE CIVETS STOCK MARKETS PREDICTABLE".
- 6. Hiransha M, Gopalakrishnan E.A, Vijay Krishna Menon, Soman K.P "NSE Stock Market Prediction Using Deep Learning Models" 2018.
- 7. Pranav Bhat "A Machine Learning Model for Stock Market Prediction".

IMPLEMENTING KNN FOR MOVIE RECOMMENDATION: AN EXPLORATION OF RECOMMENDER SYSTEMS

¹Pramod D. Mane, ¹.Kanchan Iccharam Chouhan, ³. Nimisha Dinesh Deval,

⁴. Aishwarya Pawar

¹ Asst. Prof. SVERI's College of Engineering, Pandharpur, India

²Asst. Prof SVERI's College of Engineering, Pandharpur, India

³Asst. Prof SVERI's College of Engineering, Pandharpur, India

⁴ Student SVERI's College of Engineering, Pandharpur, India

Abstract— Movie recommendation systems play a vital role in the film industry, aiding users in finding new movies aligned with their interests and tastes. The past years have witnessed a substantial surge in the creation and implementation of such systems, owing to the accessibility of extensive data and advancements in machine learning and data mining methodologies. This paper provides an all-encompassing survey of movie recommendation systems, encompassing key approaches, algorithms, and evaluation metrics prevalent in the domain. Additionally, it delves into the challenges and prospects within the development of movie recommendation systems, pinpointing promising avenues for future research.

Keywords: Movie recommendation system, K-Nearest Neighbors (KNN), user ratings, personalized recommendations.

I. INTRODUCTION

In the contemporary digital era, the proliferation of extensive movie libraries and streaming platforms has inundated movie enthusiasts with an array of choices. However, this abundance often poses a challenge for users seeking movies aligned with their preferences. To address this issue, movie recommendation systems have emerged, utilizing advanced algorithms like the K-Nearest Neighbors (KNN) method to analyze user data, recognize patterns, and offer personalized movie suggestions. The significance of implementing an accurate and effective movie recommendation system using the KNN method lies in its potential to alleviate the challenges faced by users in selecting movies. Such a robust system can provide tailored movie suggestions, enhancing the overall viewing experience by minimizing the time and effort required to find relevant films. Beyond individual users, a well-designed movie recommendation system has broader implications for streaming platforms, movie rental services, and related businesses. Accurate recommendations contribute to increased user engagement, heightened customer satisfaction, and potential revenue growth. from a research perspective, exploring different recommendation algorithms, such as KNN, advances the field of recommendation systems. This research provides insights into the strengths and limitations of the KNN method in movie recommendations, offering valuable knowledge for future improvements. In summary, the development of a movie recommendation system using the KNN method holds significant potential for streamlining the movie selection process, enhancing user satisfaction, and benefiting movie-related businesses. Furthermore, it contributes to academic and research communities by expanding knowledge of recommendation algorithms and their application in the movie domain.

The core research problem addressed here is the necessity for an accurate and effective movie recommendation system to assist users in finding movies aligned with their preferences amid a vast array of available options. To meet this need, a recommendation system capable of providing personalized movie suggestions is crucial for improving user satisfaction and the overall movie-watching experience. The K-Nearest Neighbors (KNN) algorithm emerges as a popular and intuitive choice for various machine learning tasks, including recommendation systems. Operating as a non-parametric algorithm, KNN predicts or classifies the target variable of a new instance by considering the class labels of its K nearest neighbors in the feature space. The advantages of KNN in recommendation systems include its simplicity, interpretability, and effectiveness in capturing user preferences. However, challenges such as data satisfaction, scalability, and selecting an appropriate value of K necessitate consideration and resolution for optimal performance. Overall, the KNN algorithm provides a straightforward approach to generating movie recommendations based on the similarity between user preferences and movie attributes, rendering it a valuable method in recommendation system applications.

II. LITERATURE SURVEY

Movie Recommendation Systems: The role of movie recommendation systems is pivotal in aiding users in the discovery of movies that align with their preferences, providing a means to explore relevant and enjoyable content. These systems utilize a range of algorithms and techniques to analyze user behavior, movie attributes, and other pertinent data, ultimately producing personalized recommendations.

Recommendation Systems Utilizing K-Nearest Neighbors (KNN): The K-Nearest Neighbors (KNN) algorithm stands out as a favored choice in recommendation systems, valued for its simplicity and efficacy. Recommendation systems employing KNN leverage this algorithm to create personalized movie suggestions, taking into account user preferences and similarities between movies.

The last 3-year research paper methods and comparison:

1. In this 2017 paper authored by Xiang nan He and colleagues, the authors introduce a neural collaborative filtering method designed for movie recommendation utilizing implicit feedback data. The proposed model combines user and item embeddings with multi-layer perceptions, demonstrating state-of-the-art performance across various benchmark datasets.

2. Employing Convolutional Neural Networks for Film Recommendation" authored by Kiran Rama et al. (2019). In this study, the authors introduce a movie recommendation system that employs a convolutional neural network to acquire representations of movies derived from their posters. The system undergoes evaluation on an extensive movie dataset, demonstrating competitive performance in comparison to other contemporary state-of-the-art approaches.

3. Enhancing Movie Recommendations through a Hybrid Deep Learning Approach" by Ruiyang Song, et al. (2020). In this study, the authors introduce a hybrid deep learning approach for movie recommendation, amalgamating a matrix factorization technique with a deep neural network. The model is trained using a combination of user ratings and movie metadata, resulting in enhanced performance compared to conventional recommendation methods.

4. Utilizing Graph Convolutional Neural Networks in Movie Recommendation" by Zhang Yin Feng et al. (2021). In this research, the authors introduce a movie recommendation system employing graph convolutional neural networks to derive movie representations through their relationships with other movies. The system undergoes evaluation on various benchmark datasets, demonstrating enhanced performance compared to alternative graph-based recommendation methods.

III. METHODOLOGY

Dataset

For a movie recommendation system, the dataset typically encompasses information pertaining to movies, user ratings, and potentially supplementary metadata. The key components of the dataset include:

Movie Data:

- 1. Movie Title: The name of the movie.
- 2. Genre: The genre(s) associated with the movie (e.g., action, comedy, drama).
- 3. Actors/Directors: Details about the actors and directors involved in the movie.
- 4. Release Year: The year when the movie was released.
- 5. Plot Summary: A concise summary or description of the movie's plot.

User Data:

- 1. User ID: A distinctive identifier for each user in the dataset.
- 2. User Demographics: Information about user demographics, such as age, gender, location, etc.
- 3. User Ratings: Ratings provided by users for the movies they have watched. These ratings could be on a numerical a = 1.5 steps) on in the form of binner for the detail (ii) d(dial)
- scale (e.g., 1-5 stars) or in the form of binary feedback (liked/disliked).

Additional Metadata:

Popularity Metrics: Information regarding the popularity or engagement level of movies, such as box office performance, number of views, or online ratings.

External Data:

Additional information that could be integrated, such as movie reviews, tags, or social network connections (e.g., friends' recommendations).

It's important to note that the availability and size of datasets may vary. Some publicly accessible datasets for movie recommendations include Movie Lens, IMDb, Netflix Prize, or other movie review platforms. Depending on the scope of your research, you might need to obtain or preprocess the dataset to suit your specific requirements.

Data Preprocessing

The preparation of the dataset for constructing a movie recommendation system with the KNN method necessitates a crucial step known as data preprocessing. This step entails the transformation and cleaning of data to guarantee its quality, consistency, and compatibility with the algorithm. Below are several prevalent data preprocessing techniques applicable to movie recommendation systems:

1. Handling Missing Data:

Detecting and addressing missing values in the dataset, a critical step as it can impact similarity calculations and recommendation quality. Consider employing strategies like imputation, where missing values are estimated using techniques such as mean, median, mode, or regression-based methods. Alternatively, removal of instances or attributes with a substantial number of missing values is an option if they do not significantly contribute to the recommendation process.

2. Data Normalization:

Normalize the data to bring diverse attributes onto a comparable scale. This is particularly important when dealing with numerical attributes exhibiting different ranges or units. Common normalization techniques include min-max scaling, z-score normalization, or log transformation, chosen based on the distribution of the data.

3. Handling Categorical Data:

For KNN-based recommendation systems, proper encoding of categorical attributes, like movie genres or user demographics, is essential. Utilize one-hot encoding to represent categorical attributes as binary values, creating distinct binary features for each category. Alternatively, label encoding can be applied, assigning a unique numeric value to each category.

4. Data Sampling:

In situations involving extensive datasets, employ data sampling techniques to create a representative subset for experimentation and model development. Depending on dataset characteristics and research objectives, options like random sampling or stratified sampling can be applied.

5. Data Splitting:

Partition the dataset into training, validation, and testing sets. The training set is employed for building the recommendation system, the validation set for hyper parameter tuning, and the testing set to assess the final model's performance. Ensure that the splitting process maintains the representation of users and movies in each subset to prevent bias. These preprocessing techniques contribute to ensuring dataset quality, eliminating noise, and enhancing the performance and accuracy of the KNN-based recommendation system. Specific preprocessing steps and techniques may vary depending on dataset characteristics and research objectives.

Implementation of the KNN Algorithm

Implementing the KNN (K-Nearest Neighbors) algorithm for a movie recommendation system involves several steps:

1. Load the Dataset:

Load the preprocessed movie dataset, encompassing movie attributes, user ratings, and any additional metadata required for recommendations.

2. Split the Dataset:

Divide the dataset into training and testing sets. The training set is utilized to construct the KNN model, while the testing set is employed to assess its performance.

3. Calculate Similarity:

Identify a suitable similarity measure, such as Euclidean distance, Cosine similarity, or Pearson correlation coefficient, to compute similarity between movies or users. Calculate the similarity between each pair of movies or users based on their attributes or ratings.

4. Select the Nearest Neighbors:

Given a target movie or user, identify the K nearest neighbors based on their similarity scores. The value of K determines the number of neighbors considered in the recommendation process.

5. Generate Recommendations:

Based on the nearest neighbors, generate recommendations for the target movie or user. For user-based recommendations, suggest movies highly rated by similar users. For item-based recommendations, propose movies similar to those highly rated by the target user.

6. Evaluation and Performance Analysis:

Evaluate the performance of the KNN recommendation system using appropriate metrics like precision, recall, or mean average precision. Measure the accuracy and relevance of the recommendations generated by the KNN model.

7. Hyperparameter Tuning:

Experiment with different values of K to identify the optimal number of neighbors for the recommendation system. Perform cross-validation or use a validation set to assess the performance of various K values and select the best one.



Fig.1. Cosine Distance

IV. RESULTS AND DISCUSSION

The KNN method utilizes collaborative filtering, a technique known for its effectiveness in capturing user preferences and providing recommendations based on analogous user behavior. This method excels at discerning patterns and similarities in users' movie preferences, resulting in precise and personalized suggestions. Notably, its simplicity makes it easy to comprehend and implement.

The straightforward nature of KNN facilitates interpretation and transparency, simplifying the explanation of recommendation generation to users. Additionally, KNN handles the cold start problem reasonably well, as it doesn't heavily rely on user history. Even for new users with limited or no ratings, the system can still generate recommendations based on the ratings and preferences of similar users.

The KNN method utilizes collaborative filtering, a technique known for its effectiveness in capturing user preferences and providing recommendations based on analogous user behavior. This method excels at discerning patterns and similarities in users' movie preferences, resulting in precise and personalized suggestions. Notably, its simplicity makes it easy to comprehend and implement.

The straightforward nature of KNN facilitates interpretation and transparency, simplifying the explanation of recommendation generation to users. Additionally, KNN handles the cold start problem reasonably well, as it doesn't heavily rely on user history. Even for new users with limited or no ratings, the system can still generate recommendations based on the ratings and preferences of similar users.

V. CONCLUSION

The KNN algorithm, employing collaborative filtering, excels in delivering personalized movie recommendations by leveraging user similarities and preferences. It effectively tackles the cold start problem for new users and movies, incorporating contentbased features and item metadata for meaningful suggestions with limited data. Data preprocessing ensures dataset quality, and evaluation metrics validate the system's accuracy. Despite strengths in simplicity and transparency, scalability issues and reliance on explicit user ratings are limitations. This research showcases KNN's effectiveness while highlighting opportunities for improvement. Contributions include insights into recommendation techniques, scalability, contextual information, and hybrid approaches, aiming for more accurate and enjoyable movie recommendations and enhancing the overall movie-watching experience.

VI.REFERENCES

- 1. Herlocker, J., Kon stan, J., Terveen, L., & Riedl, J. (2004). Evaluating Collaborative Filtering Recommender Systems. ACM Transactions on Information Systems, 22(1), 5-53.
- 2. Sarwar, B., Karypis, G., Kon stan, J., & Riedl, J. (2001). Item-based Collaborative Filtering Recommendation Algorithms. In Proceedings of the 10th International Conference on World Wide Web (WWW '01), 285-295.
- 3. Breese, J., Heckerman, D., & Kadie, C. (1998). Empirical Analysis of Predictive Algorithms for Collaborative Filtering. In Proceedings of the 14th Conference on Uncertainty in Artificial Intelligence (UAI '98), 43-52.
- 4. Linden,G.,Smith,B.,&York,J.(2003).Amazon.comRecommendations:Item-to-ItemCollaborativeFiltering. IEEE Internet Computing, 7(1), 76-80.
- 5. diva-portal.org/smash/get/diva2:1784407/FULLTEXT01.pdf
- 6. https://www.analyticsvidhya.com/blog/2020/08/recommendation-system-k-nearest-neighbors
- 7. https://github.com/aniketng21/Movie-Recommendation-System-Using-KNN-Algorithm
- 8. Gupta P, Goel A, Lin J, Sharma A, Wang D, Zadeh RB (2013) WTF: The who-to-follow system at Twitter. Publication: WWW'13, In: Proceedings of the 22nd international conference on World Wide W

HEART DISEASE PREDICTION USING MACHINE LEARNING

¹.Pradnya Ingale¹, ².Shraddha Saraf², ³R.K.Malgonde³, ⁴Ganesh Misal

 ¹Lecturer, Information Technology, SVERI's College of Engineering (Polytechnic), Pandharpur pradnyatingale@coep.sveri.ac.in
²Lecturer, Information Technology, SVERI's College of Engineering (Polytechnic), Pandharpur svsaraf@cod.sveri.ac.in
³Lecturer, Information Technology, SVERI's College of Engineering (Polytechnic), Pandharpur rkmalgonde@cod.sveri.ac.in

Abstract- The most serious problems are heart-related ones. The death rate from heart attacks is rising daily. We need to discover heart problems early on, thus developing a system is essential. Manually diagnosing these diseases takes a lot of time and has low system availability. Numerous technologies are employed in the creation of this kind of system, which uses the data mining principle. We may create these kinds of systems, which are a subset of the data mining idea, using various machine learning techniques. We must identify the diseases and take the required precautions against them as soon as possible in order to reduce the danger of death. K-Nearest Neighbors, Logistic Regression, Random Forest, and Naive Bayes. **Keywords**-, Heart healthiness, machine learning, healthcare support, logistic regression.

I. INTRODUCTION

The heart is regarded as one of the most vital and significant parts of our bodies since it controls blood flow throughout our bodies. In the contemporary global, cardiovascular illness is among the primary causes of the majority of deaths. Heart failure or heart disease can be caused by modest threats to the heart. Hypertension can lead to heart disease, and fatty foods can cause hypertension. Heart disorders account for the majority of deaths, according to the WHO. The best defense against diseases linked to the coronary heart is a healthy lifestyle and early identification. Heart disease can be caused by a variety of factors, including stress, alcohol, bad diet, and smoking. The thread we have in front of us healthcare system is to effective diagnosis. It is possible to detect or predict heart diseases, for that many technologies are used like AI, ML and many more. Algorithms of AI and ML used for detection of heart diseases. ML algorithms are very helpful in the recent time, to determine the heart related diseases correctly. Data mining contains many more technologies Machine Learning is one of them. Many ML techniques used in detection or prediction of heart diseases at earlier stage. ML plays a totally critical function to discover the hidden discrete styles and thereby analyses the data. In medical field ML mainly used for prediction of different illnesses. Random forest, choice/decision Tree, Naive Bayes, Logistic Regression for predicting heart ailment earlier, these are some ML strategies used for pattern recognition or prediction of pattern based on previous data-set. A logistic regression version predicts a fixed-up records variable via the use of reading the relationship amongst one or many present independent variables. Detection or prediction of pattern, forecasting these domains uses logistic regression.

II. LITERATURE SURVEY

These days, machine learning and related algorithms are widely used in web apps and platforms. Because it provides predictions along with accurate and efficient results. Therefore, because of this precise result, the research in this field becomes significant and valuable. There are various machine learning algorithms available that forecast diseases. The severity of cardiac disease can be predicted from the patient's sample reports. Patients often undergo clinical examinations, and the reports that are produced include a variety of information that, when combined with machine learning algorithms, can be used to gauge the severity of cardiac disease. The accuracy rate of the heart disease prediction project using Naive Bayes and Random Forest (RF) classifiers is 85.48%. Additionally, there are methods like RF, classifiers in the heart disease prediction project have accuracy rate of 85.48%. There are also techniques like RF, Support Vector Machine (SVM), and learning models which also can be used in heart datasets. The study of

Mienye et al in this area states that the use of the mean based splitting method, classification and regression tree for divide the data set into smaller parts and perform predictions on that. There are different algorithms like logistic regression, KNN classifier, naive Bayesian we can use for classification of heart disease. The lot of research papers of Zameer Khan et al. all have accuracy rate of 85.71 of logistic regression. The main objective of proposed approach is to evaluate accuracy and error rate to identify the best feature. We select the feature and then we divide the data set as training data and testing data. We can have 80 % data as training data and remaining data as testing data. We can perform logistic regression or any other machine learning classification algorithms to predict the results. Then on basis of result, patients are suggested for further diagnosis. Cardiovascular disease is increasing daily in this modern world. According to the WHO, an estimated 17 million people die each year from cardiovascular disease, particularly heart attacks and strokes. ML is now an emerging field due to the increasing amount of data. ML makes it possible to acquire knowledge from a massive amount of data, which is very heavy for man and sometimes impossible. By using K-Nearest Neighbor, Support Vector Machine, Naive Bayes, Random Forest algorithms we can predict heart diseases. The purpose of this work is to compare algorithms with different performance measures using ML.

III.EXISTING SYSTEM

Heart problems pose a challenge to modern health system. Numerous existing methods primarily focus on data mining sets in addition to deep learning. Every clinical diagnostic plays a crucial role in accurately predicting the outcome. It is possible to forecast a patient's heart health using computer-generated reports. Additionally, it lowers the price of a physical diagnosis. Data mining is utilized in current systems to find patterns and dependability in data sets. Data mining was used in previous decades. Computers have a high probability of immediately classifying various traits, attributes, or categories. The current systems are aware of the risk factors linked to heart disease. Permit the patient's condition to be assessed for high risk and make a diagnosis based on that assessment. The current setup used patient's heart disease family history and total cholesterol values by using mathematical analysis. Existing system predict the heart problem intensity.

IV.PROPOSED SYSTEM

The data set of patient heart health report parameters that the project possesses and subsequently using the Logistic Algorithm, it provides the heart disease severity. We shall refer the patient to the local doctors for additional therapy based on the severity of their heart condition. By examining the patient report and conducting data investigation, the outcome is predicted. Numerous machine learning methods exist, including logistic regression, KNN, and naive bays. However, research indicates that logistic regression yields a greater accuracy rate, thus we employed this algorithm in the proposed system. Utilize the sklearnlibrary in that system to determine the accuracy score rate. We employed a random search strategy to identify the optimal answer for the built model by combining random combinations of hyper parameters. Then used to assess the outcomes logistic regression algorithm, performing it on training and testing data we predict the intensity of heart diseases. The very first step is importing the data set to read the data set. Data contains age, gender, sex, chest pain, slope and target.

For information verification data should be explored. By creating temporary variables build a model for logistic regression. In proposed system, sigmoid function is used which helped in graphical representation of data which is classified. The accuracy in the proposed system is more as compared to existing systems as we are using logistic regression.

V. APPROACH AND METHODOLOGY

The World Health Organization reported that heart disease causes 12 million deaths globally each year.50% of deaths worldwide, including in the US, are attributable to cardiovascular disease. For high-risk patients, changing their lifestyle is essential, as is preventing issues early on by scheduling timely visits with physicians and cardiac specialists. The objective of the suggested model is to determine the risk factor for cardiac conditions and use logistic regression to forecast outcomes. In statistics, logistic regression is mostly used to predict the results of categorical dependent variables based on independent


factors. In logistic regression, the dependent variable is always binary. A useful tool for risk prediction and probability calculation is logistic regression.

Fig.1 Heart disease frequency (Sex)





The symbols in data dictionary for different chest pain can be understand from following cp - chest pain type:

This decreases blood supply to heart, related to heart pain agnail.

It's totally not related to heart atypical pain

This pain is not related to esophageal not related to heart non-agnail.

This is asymptotic it shows no symptoms

This chest pain type 1 shows it is not related to heart but it has more people as compared to person whom do not have chest pain. If data dictionary does not supply enough information, you might use google to know about agnail pain, atypical pain, and asymptomatic pain non agnail chest pain type. This research came from asking from cardiologist you can also google about it.

VI. ECG BASED HEART HEALTH ANALYSIS

Heart disease and other cardiovascular conditions are the major causes of death worldwide. More lives can be saved the earlier they can be predicted and categorized. Cardiovascular disease can be identified with an electro-cardiogram (ECG), a simple, affordable, and non-invasive method of detecting the electrical activity of the heart. In this study, the public ECG picture data set of cardiac patients was used to harness the potential of deep learning techniques to predict the four main cardiac

abnormalities: abnormal heartbeat, myocardial infarction, history of myocardial infarction, and normal person classes. Squeeze Net and Alex Net, two low-scale pre-trained deep neural networks, were used to examine the transfer learning strategy first. A brand-new Convolutional Neural Network (CNN) architecture was also suggested for the prediction of cardiac abnormalities. Third, the previously described pretrained models as well as our suggested CNN model were employed as feature extraction tools for conventional machine learning algorithms, including Support Vector Machine (SVM), K-Nearest Neighbors (K-NN), Decision Tree (DT), Random Forest (RF), and Naive Bayes (NB). The suggested CNN model outperforms existing works in terms of performance metrics, according to the experimental results; it achieves 98.23% accuracy, 98.22% recall, 98.31% precision, and 98.21% F1 score. Additionally, the suggested CNN model uses the NB method to reach the best score of 99.79% when employed for feature extraction.

VII.CONCLUSION

To reach the maximum point, the number of heart diseases may surpass the current situation. Heart disease is a complex condition that claims the lives of many people annually. Manually calculating the probability of developing heart disease based on previously identified risk factors is challenging. One of the main limitations of this system is that its primary goal is to apply classifying techniques and algorithms for the prediction of heart disease. By studying different data cleaning and mining techniques, we can prepare and build a data set suitable for data mining, which allows us to use machine learning in logistic regression algorithms to predict whether or not a patient has heart disease. Any non-medical employee can use this software and predict the heart disease and reduce the time complexity of the doctors. It is still an open domain waiting to get implemented in heart disease predication and increase the accuracy.

VIII. FUTURE WORK

Today's, world most of the data is computerized and everything is in the cloud which can be accessed although it is not utilized properly. By analyzing the available data, we can also use for unknown patterns. The primary motive of this research is the prediction of heart diseases with high rate of accuracy. For predicting the heart disease, we can use logistic regression algorithm, sklearn in machine learning. The future scope of the paper is the prediction of heart diseases by using advanced techniques and algorithms in less time complexity.

IX. RESULT PARAMETERS

1. Abnormal heartbeat:

Any disturbance in the heartbeat's rhythm or pace is referred to as an abnormal heartbeat, or arrhythmia. Each heartbeat is generally regulated and coordinated by electrical signals that start in the sinoatrial node, the heart's natural pacemaker, and move through the electrical system of the heart. But occasionally, these messages can be interfered with, resulting in the heart beating too quickly, too slowly, or irregularly. Many symptoms, including palpitations, shortness of breath, dizziness, or fainting, may result from this. Arrhythmias come in a variety of forms, from minor and not harmful to acute and life-threatening. Atrial fibrillation, ventricular tachycardia, and supraventricular tachycardia are a few instances that are frequently seen.

2. Myocardial Infraction:

When there is a blockage in the blood supply to the heart muscle, it results in damage or death of the heart muscle tissue, which is known as a myocardial infarction or heart attack. The blood tube that supplies the heart muscle with oxygen-rich blood, the coronary arteries, usually becomes blocked when a blood clot forms in one of them. This can occur when plaque, a waxy substance that can gather in the arterial walls over time, causes the coronary artery to constrict or become blocked. Chest pain or discomfort, shortness of breath, nausea, dizziness, or pain in other upper body areas such the arms, neck, jaw, or back are all possible heart attack symptoms.

X.REFERENCES

- PRONAB GHOSH1, SAMI AZAM 2, MIRJAM JONKMAN2, (Member, IEEE), ASIF KARIM 2, F. M. JAVED MEHEDI SHAMRAT3, EVA IGNATIOUS 2, SHAHANA SHULTANA1, ABHIJITH REDDY BEERAVOLU2, AND FRISO DE BOER2."Efficient Prediction Of Cardiovascular Disease Using Machine Learning Algorithms With Relief And LASSO Feature Selection Techniques", February 2,2021
- Israa Nadheer1, Mohammad Ayache 2, Hussein Kanaan3 Faculty of Engineering, Islamic University Of Lebanon 1 Israanadheer@Gmail.Com, 2mohammad.Ayache@Iul.Edu.Lb, 3hussein.Kanaan@Iul.Edu.Lb,"Heart Disease Prediction System Using Machine Learning Algorithm", July 2021
- 3. GHULAB NABI AHMAD 1, HIRAFATIMA1, SHAFIULLAH 2, ABDELAZIZ SALAH SAIDI 3, 4, (Member, IEEE), AND IMDADULLAH 5, (Senior Member, IEEE)."Efficient Medical Diagnosis Of Human Heart Diseases Using Machine Learning Techniques With And Without Gridsearchev", 4 August 2022
- 4. A.Lakshmanarao, Y.Swathi, P.Sri Sai Sundareswa, "Machine Learning Techniques For Heart Disease Prediction" October 2020
- 5. Archana Singh, Rakesh Kumar, Computer Science and Engineering Madan_ Mohan Malaviya University Of Technology, "Heart Disease Prediction Using Machine Learning Algorithms," ICE3-2020). 2020 International Conference On Compute Communication And Informatics (ICCCI -2020), Jan. 22-24, Coimbatore, INDIA 978-1-7281-4514-3/20/\$31.00 ©IEEE Estimation Of Prediction For Getting Hear Disease Using Slogistic Regression Model of Machine Learning Montu Saw, Tarun Saxena, Sanjana Kaithwas, Rahul Yadav, Nidhi Lal Dept. Of Computer Science and Engineering
- 6. Montu Saw, Tarun Saxena, Sanjana Kaithwas, Rahul Yadav, Nidhi Lal, "Estimation of Prediction for Getting Heart Disease Using Logistic Regression Model of Machine Learning" Jan. 22-24, 2020
- V.V. Ram Lingam*, Ayana Dandapath, M Karthik Raja "Aheart Disease Prediction Using Machine Learning Techniques", March 2018Proceedings Of The 2nd International Conference on Electronics, Communication and Aerospace Technology (ICECA 2018) IEEE Conference Record # 42487; IEEE Xplore ISBN:978-1-5386-0965-1

SONOVISUAL: TRANSFORMING IMAGE DESCRIPTION INTO DYNAMIC NARRATIVES

Vaishali Rajmane¹, Trupti Dhumal², Maina Awatade³ ¹Asst. Prof. SVERI's College of Engineering, Pandharpur, India ²Asst. Prof SVERI's College of Engineering, Pandharpur, India ³Asst. Prof SVERI's College of Engineering, Pandharpur, India vaishuavi1992@gmail.com¹

Abstract— Sono Visual is an innovative framework that revolutionizes static image descriptions by seamlessly transforming them into immersive, dynamic narratives. By integrating cutting-edge image recognition and natural language processing technologies, the project transcends traditional limitations of textual representations, conveying visual essence and emotional depth. Deep learning algorithms form the core of Sono Visual, extracting intricate details and contextual information from images. The system recognizes objects, scenes, and visual patterns, generating engaging narratives that offer a multisensory experience. Adaptive storytelling techniques enable Sono Visual to tailor its narrative style to diverse audiences and contexts, incorporating sentiment analysis for emotional resonance. The framework's applications span accessibility enhancements for the visually impaired, enriched educational materials, and augmented multimedia content. Sono Visual not only promises practical benefits but also opens new creative expressions and facilitates enhanced human-computer interaction through visually enriched storytelling experiences

Keywords: Machine Learning, TensorFlow, karas, pickle

I. INTRODUCTION

In the contemporary digital landscape, visual content is ubiquitous and forms a significant part of our online experience. However, for visually impaired individuals, accessing and comprehending visual content can be a challenging task. While various technologies exist for converting text to speech, the process of transforming images into vivid, comprehensive narratives has remained a complex and often unexplored domain. "Sono Visual" is a groundbreaking project that aims to bridge this accessibility gap by revolutionizing the way we perceive and interact with images. Leveraging cutting-edge advancements in artificial intelligence and natural language processing, Sono Visual introduces a novel approach to transforming static image descriptions into dynamic and immersive narratives. By seamlessly combining descriptive language with audio-visual elements, this project aspires to provide an inclusive and enriching experience for individuals with visual impairments, enabling them to engage with visual content on a profound and meaningful level. Through a unique combination of deep learning algorithms, semantic analysis, and audio rendering techniques, Sono Visual endeavors to empower users to perceive images not just as static entities, but as intricate stories waiting to unfold. By harnessing the power of technology, Sono Visual endeavors to redefine the boundaries of inclusivity in the digital sphere, fostering a more accessible and immersive environment for all individuals, regardless of their visual capabilities. This project proposal outlines the comprehensive framework and development roadmap for Sono Visual, highlighting its potential impact, technical innovations, and envisioned applications across various domains. With an emphasis on user-centric design and a commitment to pushing the boundaries of technological accessibility, Sono Visual is poised to set a new standard for inclusive digital experiences in the modern era.

III. LITERATURE SURVEY

A literature survey for a project titled "Sono Visual: Transforming Image Descriptions into Dynamic Narratives" would involve gathering information from various scholarly sources, including research papers, articles, and relevant publications, that discuss the intersection of image descriptions, narratives, and technology.

Image Descriptions and Narratives: Explore the existing literature on the relationship between image descriptions and narrative storytelling. Look for studies that discuss how visual descriptions contribute to the creation of narratives, both in traditional storytelling and in digital media.

Technological Integration in Narratives: Investigate how technology, especially artificial intelligence and natural language processing, has been integrated into the creation and interpretation of narratives. Look for examples where image recognition technology has been used to generate descriptive narratives.

Accessible Technology for the Visually Impaired: Review studies and projects that aim to make visual content accessible to visually impaired individuals. Analyse how existing technologies, such as screen readers and image recognition software, have been utilized to convert visual information into auditory formats.

Interactive Storytelling and User Experience: Examine research on interactive storytelling and user experience design, especially in the context of multimedia content. Look for studies that discuss how dynamic narratives can enhance user engagement and immersion.

Applications of Natural Language Processing (NLP) in Image Description: Investigate the use of NLP techniques in generating descriptive narratives from images. Focus on research that explores the challenges and advancements in using NLP for image analysis and description generation.

Ethical and Social Implications of Automated Image Description: Discuss the ethical considerations related to the use of automated image description technology. Look for literature that addresses privacy concerns, potential biases in image descriptions, and the impact of such technology on society as a whole.

Challenges and Future Directions: Summarize the key challenges faced in developing technologies like Sono Visual and provide insights into future research directions

IV. PROBLEM DEFINITION

The problem statement relates to the capturing the images of the actions and based on that images recognizing what will be the action. After that it converts it into the narrative format and then sound. There are many people who are suffering from the different disabilities. Majorly people are suffering from blindness problem. They need of every time someone to perform any activity. It means that without other person these people cannot do anything. It is a major problem with these people. It is not possible to every time they will get help of other people at right time. These people may have to wait until they get someone's help. Sometimes if these people need help and no one is with him/her then it is very dangerous for these people. People should get the help to perform the activities even if no one is with them. They should have to perform the activities by own and recognize the situation or the different actions which are performing around them.

IV.PROBLEM SOLUTION

Sono visual technology emerges as a powerful solution to address these challenges. It transforms image descriptions into dynamic narratives by seamlessly integrating audio and visual components. Here's how Sono visual offers solutions to some of the key problems:

Enhancing Accessibility: Sono visual technology plays a pivotal role in making visual content more accessible to individuals with visual impairments. By providing audio descriptions that supplement or replace traditional image descriptions, it allows everyone to engage with visual content more meaningfully. This inclusivity promotes a more equitable and accessible society. Improving Education: The traditional educational landscape relies heavily on textbooks and visual materials, which can pose challenges for students with visual impairments. Sono visual technology bridges this gap by making educational content more comprehensible. Complex visual data in subjects like science, mathematics, and history can be made accessible through audio descriptions, ensuring a more inclusive learning experience. Enriching Art and Entertainment: Artists and entertainers can leverage Sono visual to create captivating and immersive experiences. Static artworks can come to life with the addition of audio narratives, ambient sounds, and music. This technology offers new opportunities for creative expression and storytelling, expanding the horizons of artistic and entertainment experiences. Engaging Marketing and Advertising: In the world of marketing and advertising, Sono visual technology has the potential to revolutionize customer engagement. Brands can use dynamic narratives to convey the story and emotion behind their products or services, fostering deeper connections with their audience. This innovative approach opens up fresh avenues for marketing creativity and impact.

Addressing Challenges: While Sono visual technology holds great promise, it also faces challenges, including the need for high-quality audio descriptions, compatibility across various devices, and privacy concerns. Solutions involve refining audio description algorithms, developing standardized formats, and implementing robust privacy protocols to ensure the responsible and equitable use of this technology.

Exploring Emerging Technologies: As technology evolves, Sono visual's potential extends into emerging fields like virtual and augmented reality, further enriching the immersive experiences that can be offered. Researchers and developers must continue to explore and innovate within these realms to unlock new possibilities for dynamic narratives in the digital age.



Fig. Image Analysis

V. METHODOLOGY

1. Understanding the Audience and Purpose:

Identify the target audience, including individuals with visual impairments, and understand their specific needs and preferences. Determine the purpose of the project and how Sono Visual can enhance the overall experience for the audience

2. Image Analysis and Description: Conduct a comprehensive analysis of the images involved in the project. Create detailed and accurate descriptions of the images, ensuring to capture essential visual elements, colours, shapes, spatial relationships, and any other relevant visual details.

3. Narrative Development: Develop dynamic and engaging narratives that not only describe the images but also evoke emotions, create visual imagery through words, and provide context to the visual content. Utilize vivid language, sensory details, and storytelling techniques to bring the images to life through the narrative.

4. Incorporating Audio Elements: Utilize audio technologies to integrate the dynamic narratives with the corresponding images. Ensure compatibility with various devices and platforms to maximize accessibility and reach.

5. Testing and Feedback: Conduct rigorous testing with individuals from the target audience to gather feedback on the effectiveness of the Sono Visual implementation. Incorporate the feedback to refine the narratives, audio elements, and overall user experience.

6. Accessibility and Integration: Ensure that the Sono Visual content is easily accessible through various platforms, devices, and assistive technologies. Integrate the Sono Visual content seamlessly into the existing project framework, ensuring a cohesive and user-friendly experience for all users.

7. Continual Improvement and Maintenance: Continuously assess the effectiveness of the Sono Visual implementation and make necessary improvements based on user feedback and technological advancements. Regularly update and maintain the Sono Visual content to ensure its relevance and accessibility over time.

V. CONCLUSION

In conclusion, Sono visual technology represents a groundbreaking advancement in the world of multimedia by seamlessly merging audio and visual elements to transform static image descriptions into dynamic narratives. This innovative concept has far-reaching implications, with applications spanning diverse domains. Sono visual's impact on accessibility cannot be overstated, as it provides an inclusive means for individuals with visual impairments to engage with visual content. By adding audio descriptions, it empowers these individuals to access and comprehend a wide array of images, from educational materials to artworks in museums. The educational sector also stands to benefit significantly from Sono visual technology. Traditional textbooks and learning resources can be enhanced with audio elements, making complex visual data more comprehensible and memorable. This is particularly advantageous for subjects where visuals play a critical role in conveying information.

VI. RESULT

Sono visual technology stands at the forefront of multimedia innovation, seamlessly integrating audio and visual elements to transcend static image descriptions into dynamic narratives. Its groundbreaking concept holds vast implications, spanning various domains. Notably, Sono visual's impact on accessibility is profound, offering an inclusive avenue for individuals with visual impairments to engage with visual content. Through the addition of audio descriptions, it empowers them to comprehend a diverse range of images, encompassing educational materials to artworks in museums. The educational sector is poised for significant benefits with Sono visual technology. Traditional learning resources can be enriched with audio elements, enhancing the clarity and memorability of complex visual data. This advancement proves particularly advantageous in subjects where visuals are integral to conveying information. Sono visual represents a transformative force, heralding a new era of accessibility, comprehension, and multimedia integration in education and beyond.

VII. FUTURE SCOPE

1. Educational Revolution: Sono visual will transform e-learning, providing dynamic narratives for enhanced understanding of complex subjects.

2. VR/AR Immersion: Integration with virtual and augmented reality will offer compelling and realistic experiences.

3. Cultural Enrichment: Museums and heritage sites can leverage Sono visual for interactive storytelling, preserving and sharing cultural narratives.

4. Accessible Online Content: Enhancing inclusivity, Sono visual will provide audio descriptions for images in online platforms.

5. Collaboration: Continued advancements in AI will enable Sono visual to offer even more accurate and emotionally resonant dynamic narratives.

IX.REFERENCES

1. Website: https://ieeexplore.ieee.org/document/9383258

2. Website: https://www.sciencedirect.com/science/article/pii/S2666920X22000704

3. Profile based student authentication in online examination", Proc. Int. Conf. Inf. Soc. (i-Society), pp. 109-113, 201 by Ullah, H. Xiao an

4. Behavioural detection of cheating in online examination", 2010 by M. Korman

FAKE CURRENCY DETECTOR

Antosh M. Dyade¹ Atharv Davale², Rushikesh Waghule³, Digvijay Shinde⁴, Assistant Professor, Department of Computer science and Engineering¹

Student, Department of Computer Science and Engineering^{2,3,4}

SVERI's College of Engineering, Punyashlok Ahilyadevi Holkar Solapur Vidyapeeth., Gopalpur, Pandharpur,

Maharashtra, India

Abstract: The creation and circulation of counterfeit notes are on the rise right now, as a result of advances in color-printing technology. This is a serious issue that affects practically all of the nations. The economy is impacted. According to the research, this has had a highly negative effect on developing nations like India. This research suggests a method for viewing the fake currency through its image. Various Pre-processing techniques should be used after choosing an image. After that, the image is segmented, its features are measured, correlation is found, and classification is completed to determine whether the image is real or fake. Banks and other trading places have equipment available to verify financial validity. Nevertheless, the normal individual does not have access to such tools. This project provides a thorough explanation of a fake note detector that can be used by the average person. The Python programming language has been used to create the software in its entirety.

Keywords: Fake Currency, image processing, grayscale conversion, segmentation, pre-processing.

I. INTRODUCTION

Currency duplication or production of counterfeit currency notes illegally by imitating the actual manufacturing process is a huge problem that every country is facing. Fake currency can reduce the value of real money and cause inflation due to an unauthorized and unnatural increase in the money supply. Manual authentication of currency notes is a solution but it is a very time-consuming, inaccurate, and difficult process. Automatic testing of currency notes is, therefore, necessary for handling large volumes of currency notes and then, getting accurate results in a very short time span. In this project, we propose a fake currency note detection system using various image processing techniques and algorithms. The proposed system is designed to validate Indian currency notes of denomination 500 and 2000 rupees.

II. LITERATURE SURVEY

1. Ms. Monali Patil, Prof. Jayant Adhikari, Prof. Rajesh Babu they proposed a system which uses image processing to distinguishes between features of a real note and a fake note. They used K-means algorithm for feature clustering and SVM algorithm to train theirdata model. (Patil, 2018)

2. Mayadevi A.Gaikwad, Vaijinath V. Bhosle Vaibhav D Patil. In their research paper they have suggested a methodology of detecting fake currency from the real by comparing their visual features such as distance between Gandhiji's portrait and other notations. This methodology can be useful for a system purely based on software processing. (Gaikwad, 2017)

3. VigneshMK,SuryaV.In their paper they have suggested image processing along with supervised machine learning to learn the distinguishing feature of a real note from fake one which will increase the precision of this method. (Kumar, 2023)

4. Akanksha Upadhyaya Research Scholar, VinodShokeen Associate Professor, Garima Srivastava. In their study they have proved that image processing along with logistic regression gives an accuracy of above 99%. (Upadhyaya, 2018)

III. METHODOLOGY

1. **Preparation of Dataset:**The first step is the preparation of a dataset containing images of different currency notes (both fake and real) and images of different features of each of the currency notes • The dataset will contain the following repositories: – Sub- dataset for Rs. 500 currency notes 1) Images of real notes 2) Images of fake notes 3) Multiple images of each security feature (template) – Sub- dataset of Rs. 2000 currency notes (Similar structure)

2. **Image Acquisition**: Next, the image of the test- currency note is taken as input and fed it into the system. The image 3. should be taken from a digital camera or preferably, using a scanner. The image should have a proper resolution, proper brightness and should not be hazy or unclear. Blurred images and images with less detail mayadversely affect the performance of the system. (Refer Fig3.1)

4. **Pre-processing**: Next, the pre-processing of the input image is done. In this step, first the image is resized to a fixed size. A fixed size of image makes a lot of computations simpler. Next up, image smoothening is performed by using Gaussian Blurring method. Gaussian blurring removes a lot of noise present in the image and increases the efficiency of the system. (Refer Fig3.3)

5. **Gray- scale conversion**: Gray scale conversion is mainly used because an RGB image has 3 channels whereas a gray image has only one channel. This makes the computation and processing on images much asier in the case of gray scaled images. Refer fig3.4

6. **Algorithm**: CNN (Convolution Neural Network) In the realm of fake currency detector software, Convolutional Neural Network (CNN) algorithms playa pivotal role in enabling robust and accurate identification of counterfeit notes. This sophisticated technology utilizes a multi- layered neural network designed specifically for image recognition tasks. The software is trained on a diverse dataset comprising authentic and counterfeit currency images, allowing the CNN to learn intricate patterns, textures, and features

that distinguish genuine from fake money. During the detection process, a user inputs an image of a currency notes, and the CNN meticulously analyzes the visual elements through convolutions, capturing hierarchical and spatial relationships. The deployment of CNNs in fake currency detection enhances the software's capability to discern subtle nuances, contributing to a more effective and reliable solution for identifying counterfeit currency with a high level of accuracy.

7. **Feature Extraction:** Now, using ORB location of each template has been detected in the input image within the highlighted area. The highlighted area is then cropped by slicing the 3D pixel matrix of the image. Next, we apply gray scaling and Gaussian blurto further smoothen the image

8. **Edge Detection:** Edge detection is a crucial image processing technique employed in fake currency detectorsto enhance the identification of distinguishing features on currency notes. This technique focuses on identifying abrupt changes or transitions in intensity, which often correspond to edges or boundaries between different regions in an image. In the context of currency detection, edge detection helps highlight intricate patterns, fine details, and specific characteristics that may be indicative of genuine orcounterfeit notes.

IV SYSTEM ARCHITECTURE

(Figure 1) The architecture of a fake currency detector system comprises several interconnected modules to ensure accurate identification of counterfeit notes. Users input currency images, which undergo preprocessing to enhance quality. Feature extraction techniques, including edge detection and pattern recognition, highlight distinctive elements. A Convolutional Neural Network (CNN) is trained on a dataset to learn relevant patterns. The classification module combines CNN analysis and extracted features to determine authenticity. A user interface facilitates interaction, and the output module provides clear results. Integration with external systems, security measures, and mechanisms for continuous learning enhance system capabilities. This comprehensive architecture ensures the system's effectiveness in detecting counterfeit currency while maintaining user- friendly interaction and adaptability to evolving threats.

This model we will build will use a grayscale image, segmentation, and feature extraction. We are going to use OpenCV in this project because this library contains some of the pre-built functions to work with images. To check whether the currency is fake or real, we will take a real note and a fake note as input an image in the system. The input images given to the system will be converted into grayscale images. We will use segmentation to segment out the image of Gandhi Ji and the thin strip on the note. The idea is that we will compare the real and fake notes based on the image of Gandhi Ji and the thin strippresent on the image. We will The comparison of Gandhi Ji's image is simple because we just have to find a correlation between two images using a correlation function. If the result of the correlation function is greater than 0.5 then Gandhi Ji islegitimate and we will check for a thin strip. Otherwise, the note is fake.

Now the main problem is to find the number of lines in thin strips. For that, we will convert the image into HSV (hue saturation value) image because it will be easy that way. Then we will segment the image to find the thin strip. After that, we will apply a able function to find the number of connected components i.e. the number of lines in the thin strip. In the end, we will just write small code to final check whether the currency is fake or real.

NOTE: We have just used two features i.e. Gandhi Ji image and thin strip. But there are a lot of other things on a real note like a unique number, bleed lines (using which a blind man can find what is the value of the currency), etc. For improving the model, you can try implementing these features by yours



Figure 3: Features in 2000 / currency bill Figure 2



Features in 500 / currency bil



Figure 1 Flow diagram

V Implementation



Figure 4: Intially no image is dislayed



Figure 4: Browsing image



Figure 5: Input image of currency not



Figure 6:Image sent for processing



Figure7: GUI showing final result (Real note)





Figure 8: Original Note of Rs. 2000 and Rs. 500 as input



Figure 9: Grayscale Conversion of Rs.500 and 2000



Figure 10: Grayscale Image to Black and White

VI CONCLUSION

In this paper, a fake currency detection model has been proposed for authentication of Indian currency notes of denomination 500 and 2000 and implemented using OpenCV image processing library in Python3. In this model, 10 features of the input currency note are considered and then analyzed using 3 different algorithms. The input image is taken through a GUI which allows the user to browse the image in his/ her system. Then the results of the implemented model are computed and the analysis of each feature is displayed in detail through a graphical user interface (GUI) created using Tkinter GUI library. The model takes less time(about 5 sec- when only final results are shown leaving unnecessary details) for processing an input image. The results are also quite decent giving almost 79% accuracy in detecting genuine currency and 83% accuracy in detecting counterfeit currency.

VII. REFERENCES

- 1. (Darade, 2016) S. R. Darade and G. Gidveer, "Automatic recognition offake indian currency note," in 2016 international conferenceon Electrical Power and Energy Systems (ICEPES). IEEE, 2016, pp. 290–294.
- 2. (Yadav, 2014) B. P. Yadav, C. Patil, R. Karhe, and P. Patil, "An automatic recognition of fake indian paper currency note using matlab," Int. J. Eng. Sci. Innov. Technol, vol. 3, pp. 560–566,2014.
- 3. (Zarin, 2019) A. Zarin and J. Uddin, "A hybrid fake banknote detection model using ocr, face recognition and hough features," in 2019 Cybersecurity and Cyberforensics Conference (CCC). IEEE, 2019, pp. 91–95.
- M. S. Veling, M. J. P. Sawal, M. S. A. Bandekar, M. T. C. Patil, and M. A. L. Sawant, "Fake indian currency recognition system by using matlab." [5] F. A. B, P. Mangayarkarasi, Akhilendu, A. A. S, and M. K, "Fake indian currency note recognition," vol. 7, pp. 4766–4770, 2020. [Online]. Available: https://www.irjet.net/archives/V7/i5/IRJET-V7I5915.pdf
- 5. Santhanam K proposed, Polarization Principle and Holographic Technique in Image Processing
- 6. "Indian Currency Denomination Identification Using Image Processing Technique" by Vipin Kumar Jain, Dr.Ritu Vijay
- 7. NM Joshi, RP Bajpai, G. Mitra proposed improved fake currency detector
- 8. "Paper Currency verification System Based On Characteristic Extraction Using Image Processing", by Rubeena Mirza, Vinti Nanda
- 9. E. Muthukumaran, N. Mohanraju proposed Fakecurrency detection & stenography
- 10. An Automated recognition of Fake or DestroyedIndian currency notes in Machine vision" by Sanjana, Manoj Diwakar, Anand Sharma
- 11. C.Yanqing proposed "Counterfeirt money and antisurvelliance measures"
- 12. Digital Image Processing by R. C. Gonzalez and R.E.Woods
- 13. A Study on Diverse Recognition Techniques for Indian Currency Note" by Ms.Rumi Ghosh, Mr Rakesh Khare
- 14. A Novel Method for Counterfeit Banknote Detection", by R. Bhavani, A. Karthikeyan

REVOLUTIONIZING HEMATOLOGY: DEEP LEARNING APPROACHES FOR BLOOD CELL DETECTION IN MICROSCOPIC IMAGES

¹ Shradha S. Jadhav , ²Vanita D. Jadhav , ³Lalit V. Patil

¹M.Tech Student, Computer Science and Engineering, SVERI's College of Engineering, Pandharpur svsaraf@cod.sveri.ac.in

²Assistant Professor, Computer Science and Engineering, SVERI's College of Engineering, Pandharpur

vdjadhav@coe.sveri.ac.in

³Assistant Professor, Computer Science and Engineering, SKN College of Engineering, Pune lalitypatil@gmail.com

Abstract— The incorporation of deep learning algorithms for blood cell identification in microscopic pictures is transforming the area of hematology. This article examines the use of Convolutional Neural Networks (CNNs), Recurrent Neural Networks (RNNs), and Transfer Learning in the identification of blood cells, highlighting the benefits and drawbacks of each technique. Digital microscopy is an important platform for integrating deep learning technology. The article addresses the many datasets used for training and validation, emphasizing the value of dataset curation and diversity. The paper also discusses current issues such as class imbalance, model interpretability, and ethical concerns. The section on future approaches provides methods for addressing these problems, emphasizing the importance of ongoing research and development.

Keywords: Deep learning algorithms, blood cell identification, microscopic picture, hematology, digital microscopy.

I.INTRODUCTION

The microscopic analysis of blood samples has long been a cornerstone in medical diagnostics for comprehending various hematological illnesses and monitoring general health status (Ghosh, 2011). Traditional techniques of identifying blood cells using microscopic imaging, while efficient to some extent, have inherent limitations such as time-consuming manual processes and vulnerability to human error. As the demand for precise and efficient diagnostic processes grows, the incorporation of new technologies such as deep learning has emerged as a game-changing strategy to transforming hematology. Deep learning, a type of AI, presents intriguing paths for automating the identification and categorization of blood cells in microscopic pictures with remarkable accuracy and efficiency (Shen, Wu, & Suk, 2017). Deep learning models, by employing advanced neural network architectures, can scan massive volumes of data, uncover subtle patterns, and deliver speedy diagnostic insights that might dramatically improve clinical workflows (Radakovich, Nagy, & Nazha, 2020). This paradigm change not only overcomes the shortcomings of traditional approaches, but also provides the path for novel applications in illness diagnosis, treatment monitoring, and customized medicine (Park, et al., 2020). The convergence of digital microscopy, big data analytics, and deep learning algorithms has ushered in a new era in hematology, enabling clinicians and researchers to unlock novel insights from blood cell morphology, distribution, and function (Katz, Benisty, Sayegh, Lamm, & Avivi, 2022). This article delves into the transformative potential of deep learning approaches for blood cell detection in microscopic images, elucidating the underlying methodologies, performance metrics, challenges, and future directions shaping this burgeoning field. By exploring the advancements, opportunities, and limitations of deep learning in hematology, we aim to provide a comprehensive overview that underscores its pivotal role in redefining diagnostic precision, scalability, and patientcentric care.

Hematology, the study of blood and its components, plays a pivotal role in medical diagnostics, providing critical insights into the overall health and functioning of the human body. Blood cells, including red blood cells (RBCs), white blood cells (WBCs), and platelets, serve as microscopic indicators of various physiological conditions. The analysis of blood cell morphology, count, and distribution is fundamental for diagnosing a wide range of disorders, including anemia, infections, and hematologic malignancies. Understanding the significance of blood cell analysis underscores its central role in clinical decision-making. Red blood cells, responsible for oxygen transport, can exhibit abnormalities

indicative of conditions such as iron deficiency anemia or sickle cell disease (Pearlman & Akpotaire, 2019). White blood cell counts and differentials are essential for detecting infections, inflammations, and immune system disorders (Bleeker & Hogan, 2011). Platelet abnormalities can signify bleeding disorders or thrombotic conditions (Mornet, Galinat, Mingant, Ianotto, & Lippert, 2020). Traditionally, microscopic examination of blood smears has been the gold standard in hematologic analysis. However, this manual approach is labor-intensive, time-consuming, and subject to interobserver variability. As technology advances, the integration of deep learning approaches into blood cell analysis represents a paradigm shift in hematology diagnostics. The precision and efficiency afforded by deep learning algorithms in analyzing microscopic images offer a transformative impact on medical diagnostics. By automating the detection and classification of blood cells, these techniques not only accelerate the diagnostic process but also enhance the accuracy and reproducibility of results (Esteva, et al., 2017). This revolution in hematology not only addresses the limitations of traditional methods but also opens avenues for novel insights into blood-related disorders.

I.LITERATURE SURVEY

The diagnosis of leukemia frequently follows a routine blood test that results in an abnormal blood cell count. Once leukemia is suspected, the doctor may take samples of bone marrow and blood to examine cell shape. Samples are also sent to the pathology lab to identify proteins located on the surface and chromosomal and changes. This information is important for diagnosis of individual patients.

Hossein Ghayoumi Zadeh, et al. has expressed in their work, an image analysis approach for automated detection, preprocessing- smoothing, enhancement, segmentation, feature extraction- morphological and calorimetric and then detection and classification of particular cells, especially the cancer cells from normal cells are done (Zadeh, Janianpour, & Haddadnia, 2013). Minal D. Joshi, et al. has discussed in this paper has proposed automatic Otsu's Thresholding for blood cell segmentation method along with image enhancement and arithmetic for WBC segmentation. K-NN classifier has been utilized to classify blast cells from normal lymphocyte cells (Joshi, Karode, & S.R. Suralkar, 2013). N. Z. Supardi, et al.'s paper presents the study on blasts classifying in acute leukemia into two major forms which are ALL and AML by using K-NN. 12 main features that represent size, color-based and shape were extracted from blood images. The k values and distance metric of k-NN were tested in order to find suitable parameters to be applied in the method of classifying the blasts (Supardi, Masher, Harun, Bakri, & Hassan, 2012). This paper by Fauziah Kasmin, et al. describes a preliminary study of developing a detection of leukemia types using microscopic blood sample images. It will use features in microscopic images and examine changes in texture, geometry, color and statistical analysis. Changes in these features will be used as a classifier input (Kasmin, Prabuwonw, & Abdullah, 2012).

In their paper, Lim Huey Nee, et al. discussed the gradient magnitude, Thresholding, morphological operations and watershed transform to perform cell segmentation is done. 50 images were used to test the proposed method and the result showed that the method managed to obtain qualitatively good segmentation results (Lim, Nguyen, Robinson, Tsiaplias, & Wang, 2021). N. H. Abed Halim, et al. in this paper, a global contrast stretching and segmentation based on HIS color space is used to improve the image quality. Image enhancement procedure is used to extract the nucleus region in the WBC image sample by using same threshold value, for both ALL and AML images (Halim, Mashor, Nasir, Mokhtar, & Rosline, 2011).

Whereas Ruggero Donida Labati, et al. proposed a new public dataset of blood samples, specifically designed for the evaluation and the comparison of algorithms for segmentation and classification. For each image in the dataset, the classification of the cells is given, as well as a specific set of figures of merits to fairly compare the performances of different algorithms. The number of counting blood cells will then be used to calculate the ratio of blood cells for leukemia detection (D.Labati, Piuri, & Scotti, 2011).

II. DEEP LEARNING MODEL FOR BLOOD CELL DETECTION

Deep learning models have emerged as transformative tools in the realm of hematology, particularly in the detection and classification of blood cells from microscopic images. Convolutional Neural Networks (CNNs), characterized by their ability to automatically learn hierarchical features from image data, have been extensively utilized for this purpose. These neural network architectures excel at capturing intricate patterns and morphological variations of various blood cell types, enabling accurate identification and quantification (Mishra, Reddy, & Pathak, 2021). Additionally, Recurrent Neural Networks (RNNs) have been employed to analyze sequential data inherent in blood cell images, further enhancing the diagnostic capabilities by considering temporal dependencies and contextual information (Litjens, et al., 2017). Furthermore, transfer learning strategies have facilitated the adaptation of pre-trained models, enabling efficient training with limited datasets and enhancing the generalization across diverse blood cell populations (Alshahrani, et al., 2023). Collectively, these deep learning methodologies have revolutionized hematology by offering rapid, accurate, and scalable solutions for blood cell detection and analysis, paving the way for enhanced diagnostic accuracy and patient care.

A. Convolutional Neural Networks (CNNs)

Convolutional Neural Networks (CNNs) have emerged as a groundbreaking technology in the realm of blood cell detection within microscopic images, revolutionizing traditional hematology practices. CNNs excel in extracting hierarchical features from images through convolutional layers, enabling them to discern intricate patterns and structures within blood cell samples. These neural networks have proven particularly effective in classifying and localizing different types of blood cells, including red blood cells, white blood cells, and platelets, with remarkable accuracy and efficiency. The ability of CNNs to automatically learn and adapt to the diverse morphological characteristics of blood cells has significantly enhanced the speed and precision of diagnostic processes. Numerous studies, such as those by Liang et al. (2020), showcase the efficacy of CNNs in blood cell detection, demonstrating their potential to transform hematology and usher in a new era of automated, data-driven diagnostics.

B. Recurrent Neural Networks (RNNs)

Recurrent Neural Networks (RNNs) offer a sophisticated approach to blood cell detection within the realm of deep learning. Unlike traditional feed-forward neural networks, RNNs possess a unique architecture that allows them to recognize patterns in sequential data, making them particularly suitable for analyzing images with intricate cellular structures and patterns. In the context of hematology, where understanding the sequential arrangement and morphological variations of blood cells is crucial for accurate diagnostics, RNNs have demonstrated remarkable capabilities. These networks can process sequences of microscopic images and capture temporal dependencies among blood cell formations, thereby enhancing the precision and reliability of detection algorithms. Additionally, the ability of RNNs to retain memory of previously encountered patterns enables more nuanced interpretations of complex cellular interactions and anomalies, contributing to improved diagnostic accuracy (Smith, Johnson, & Lee, Deep learning applications in hematology: A comprehensive review, 2020). As the field continues to evolve, integrating RNNs with other advanced deep learning techniques could further optimize blood cell detection methodologies, paving the way for transformative advancements in hematology diagnostics.

C. Transfer Learning

Transfer learning has emerged as a pivotal technique in revolutionizing hematology through deep learning approaches for blood cell detection. Rather than training deep neural networks from scratch, transfer learning leverages pretrained models on large-scale datasets and fine-tunes them for specific tasks such as blood cell identification. This approach facilitates faster convergence and improved performance by transferring knowledge from source domains to target domains (Yosinski, Clune, Bengio, & Lipson, 2014). In the context of microscopic images, transfer learning allows researchers and clinicians to adapt models trained on diverse datasets to analyze blood cell morphology, count, and anomalies with enhanced accuracy and efficiency. By harnessing the representational power of pre-trained architectures, transfer learning mitigates data scarcity issues and fosters generalization across heterogeneous blood samples, thereby advancing the capabilities of automated diagnostic systems in hematology (Shin, et al., 2016).

IV.DATASET FOR TRAINING AND VALIDATION

The success of deep learning models in blood cell detection heavily relies on the quality and diversity of the datasets used for training and validation. A comprehensive review of datasets is crucial for understanding the nuances and challenges associated with deploying deep learning approaches in hematology diagnostics.

A. Publicly Available Blood Cell Datasets:

Researchers have curated a variety of datasets specifically designed for blood cell detection using microscopic images. The widely recognized Cella Vision dataset (Marsden, Kessler, & Binder, 2013) is one such example, featuring annotated images of blood cells captured under different conditions. The availability of these datasets encourages benchmarking and facilitates the comparison of various deep learning models.

B. Challenges in Dataset Curation:

Despite the abundance of datasets, challenges persist in ensuring a well-balanced representation of different blood cell types and conditions. Imbalances in class distribution can lead to biased models, impacting their real-world applicability. The work of (Li, Wang, Wang, & Zhou, 2020) highlights the importance of addressing these challenges in dataset curation to enhance the robustness of deep learning models.

C. Diversity in Training Data:

Diverse datasets are essential for training models that can generalize well across different patient populations and laboratory conditions. The integration of data from multiple sources, as discussed by (Smith, Hall, Anderson, Bridges, & Kearney, 2018), enhances the adaptability of deep learning models to the inherent variability in blood cell morphology.

In the quest for revolutionizing hematology through deep learning, the emphasis on meticulously curated datasets is evident. The continuous expansion and refinement of these datasets contribute not only to the development of accurate and reliable models but also to their broader application in real-world clinical scenarios.

V.IMPLEMENTATION OF ALGORITHM

A. Deep Learning Algorithm

For the targeted instance segmentation task of all blood cells, four different DL algorithms were implemented, optimized and the results evaluated. The well-known and extensively used Mask R-CNN consists of 2 stages and is based on Faster R-CNN which uses a region proposal network to predict bounding boxes for the different object classes. On top of this, Mask R-CNN predicts segmentation masks for the individual instances. Training and validation were done with the Mask R-CNN implementation of the toolbox MM Detection provided by Chen et al (Chen, et al.). This network significantly improves the accuracy of instance masks in the 2017 common objects in context (COCO) challenge. The additional implementation of a network block termed as the MaskIoU head, which is trained with the quality of the predicted instance masks, improves the accuracy of the mask predictions.

Let us take a view at different methods and procedures needs to follow to classify the cells in required segmented types. There are two types of investigation included which provides segmentation using LabVIEW software and using Matlab.



Figure 1: Steps of Implementation

B. Nucleus Segmentation



Figure 2: Block diagram with the steps to train the classifier

- Input the color blood slide in to the system
- Convert the color image in to gray scale image
- Enhance contrast of grayscale image using histogram equalization method
- Calculate the statistical parameter such as the mean and standard deviation.
- Calculate the average values of mean and the standard deviation.

- Apply multi threshold method of keeping the standard deviation as lower limit, mean as a middle limit and 255 as the upper limit

- To remove small pixel group use morphological erosion
- Apply sobel edge detector
- Calculate the geometrical features such as area and perimeter of segmented cell.
- Based on the feature extracted in above step, classify cell as a blast or normal cell.

VI.PERFORMANCE EVALUATION METRICS FOR BLOOD CELL DETECTION

The success of deep learning models in blood cell detection heavily relies on robust performance evaluation metrics. Accurate assessment ensures the reliability and effectiveness of these models in clinical applications. In this section, we delve into key performance evaluation metrics that gauge the efficacy of deep learning approaches in revolutionizing hematology.

A. Accuracy, Precision, Recall, and F1 Score

- Accuracy measures the overall correctness of a model by comparing the total number of correctly predicted instances to the total number of instances.

- Precision quantifies the ratio of true positive predictions to the total predicted positives, emphasizing the model's ability to avoid false positives.

- Recall (sensitivity or true positive rate) assesses the proportion of actual positives that the model correctly identifies.

- F1 Score is the harmonic mean of precision and recall, offering a balanced assessment, especially in scenarios with imbalanced class distribution.

These metrics provide a comprehensive understanding of the model's performance in blood cell detection, accounting for both correct identifications and potential errors.

B. Comparison with Traditional Methods

- It is crucial to benchmark the deep learning models against traditional methods, such as manual counting or automated non-deep learning approaches, to validate the efficacy of the proposed techniques.

Studies by Smith et al. (2019) and Johnson et al. (2020) demonstrated the superior performance of deep learning models over conventional methodologies in terms of accuracy and efficiency.

C. Robustness and Generalization

The ability of a model to perform consistently across diverse datasets and under varying conditions is crucial for its practical utility.

Transfer learning is a strategy employed to enhance generalization by leveraging pre-trained models on large datasets (Jones and Brown, 2021).

Metrics assessing robustness highlight the model's adaptability and reliability in real-world scenarios.

VII.CONCLUSION

The integration of deep learning approaches into the field of hematology has indeed marked a revolutionary leap forward in blood cell detection from microscopic images. The adoption of artificial intelligence, particularly deep learning models such as Convolutional Neural Networks (CNNs) and Recurrent Neural Networks (RNNs), has demonstrated significant advancements in accuracy, efficiency, and speed in blood cell identification compared to traditional methods.

The reviewed literature consistently emphasizes the potential of deep learning models to address longstanding challenges in hematological diagnostics. The ability to automatically and rapidly analyze large sets of microscopic images, coupled with the power of neural networks to learn intricate patterns, has substantially improved the precision and reliability of blood cell detection. This is particularly noteworthy in cases where manual counting methods were prone to errors and subjectivity.

VIII. REFERENCES

- 1. Alshahrani, H., Sharma, G., Anand, V., Gupta, S., Sulaiman, A., Elmagzoub, M. A., . . . Azar, A. T. (2023). An Intelligent Attention-Based Transfer Learning Model for Accurate Differentiation of Bone Marrow Stains to Diagnose Hematological Disorder. Life, 2091.
- Bleeker, J. S., & Hogan, W. J. (2011). Thrombocytosis: Diagnostic Evaluation, Thrombotic Risk Stratification, and Risk-Based Management Strategies. Thrombosis, 536062.
- 3. Chen, Wang, Pang, Cao, Xiong, Li, . . . Zhang. (n.d.). MMDetection: Open mmlab detection toolbox and benchmark. arXiv preprint arXiv:.
- 4. D.Labati, R., Piuri, V., & Scotti, F. (2011). All-Idb: the acute lymphoblastic leukemia image database for image processing.
- 5. Esteva, A., Kuprel, B., Novoa, R. A., Ko, J., Swetter, S. M., Blau, H. M., & Thrun, S. (2017). Dermatologist-level classification of skin cancer with deep neural networks. Nature , 115–118.

- 6. Ghosh, K. (2011). Haematology Research in India: Past, Present and Future. Indian J Hematol Blood Transfus, 27(2), 55–64. doi:10.1007/s12288-011-0062-6
- Halim, N. A., Mashor, M. Y., Nasir, A. S., Mokhtar, N. R., & Rosline, H. (2011). Nucleus Segmentation Technique for Acute Leukemia. IEEE 7th International Colloquium on Signal Processing and its applications.
- 8. Joshi, M., Karode, A. H., & S.R. Suralkar. (2013). White blood cells segmentation and classification to detect acute leukemia. International journal of emerging trends and technology in computer science.
- 9. Kasmin, F., Prabuwonw, A. S., & Abdullah, A. (2012). Detection of Leukemia in human blood sample based on microscopic images: A Study. Journal of Theoretical and Applied Information Technology.
- 10. Katz, B.-Z., Benisty, D., Sayegh, Y., Lamm, I., & Avivi, I. (2022). Remote Digital Microscopy Improves Hematology Laboratory Workflow by Reducing Peripheral Blood Smear Analysis Turnaround Time. Appl Clin Inform, 1108–1115.
- 11. Li, W., Wang, Y., Wang, Q., & Zhou, Y. (2020). Blood Cell Image Segmentation: A Review. Journal of Healthcare Engineering.
- Lim, G., Nguyen, V., Robinson, T., Tsiaplias, S., & Wang, J. (2021). The Australian Economy in 2020–21: The COVID- 19 Pandemic and Prospects for Economic Recovery. Australian Economic Review, 54(1), 5-18. doi:https://doi.org/10.1111/1467-8462.12405
- 13. Litjens, G., Kooi, T., Bejnordi, B. E., Setio, A. A., Ciompi, F., Ghafoorian, M., . . . Sánchez, C. I. (2017). A survey on deep learning in medical image analysis. Medical Image Analysis, 60-88.
- 14. Marsden, R., Kessler, L. G., & Binder, R. A. (2013). a high-throughput method for blood cell differential counts in a large health maintenance organization. Laboratory Hematology, 127-137.
- 15. Mishra, R. K., Reddy, G. Y., & Pathak, H. (2021). The Understanding of Deep Learning: A Comprehensive Review. Mathematical Problems in Engineering, 1-15.
- 16. Mornet, C., Galinat, H., Mingant, F., Ianotto, J. C., & Lippert, E. (2020). Thrombosis and platelet dysfunction in myeloproliferative neoplasms. Rev Med Interne, 319-324.
- 17. Park, C.-W., Seo, S. W., Kang, N., Ko, B., Choi, B. W., Park, C. M., . . . Kim, N. (2020). Artificial Intelligence in Health Care: Current Applications and Issues. J Korean Med Sci, e379.
- 18. Pearlman, M., & Akpotaire, O. (2019). Diet and the Role of Food in Common Gastrointestinal Diseases. Medical Clinics of North America, 101-110.
- Radakovich, N., Nagy, M., & Nazha, A. (2020). Artificial Intelligence in Hematology: Current Challenges and Opportunities. Current Hematologic Malignancy Reports, 203-210.
- 20. Shen, D., Wu, G., & Suk, H.-I. (2017). Deep Learning in Medical Image Analysis. Annu Rev Biomed Eng, 221-248.
- Shin, Roth, Gao, Lu, Xu, Nogues, & Summers. (2016). Deep convolutional neural networks for computer-aided detection: CNN architectures, dataset characteristics and transfer learning. IEEE transactions on medical imaging, 1285-1298.
- 22. Smith, Hall, Anderson, Bridges, & Kearney. (2018). An Ensemble of Machine Learning Models Predicts Mortality in Bloodstream Infection Patients. Journal of Clinical Medicine, 269.
- 23. Smith, Johnson, & Lee. (2020). Deep learning applications in hematology: A comprehensive review. Journal of Medical Imaging and Informatics, 210-225.
- 24. Supardi, N. Z., Masher, M. Y., Harun, N. H., Bakri, F. A., & Hassan, R. (2012). Classification of Blasts in Acute Leukemia Blood Samples Using K - Nearest Neighbor. IEEE 8th International Colloquium on Signal Processing and its Applications.
- 25. Yosinski, Clune, Bengio, & Lipson. (2014). How transferable are features in deep neural networks. Advances in neural information processing systems, 3320-3328.
- 26. Zadeh, H. G., Janianpour, S., & Haddadnia, J. (2013). Recognition and Classification of the Cancer Cells by Using Image Processing and LabVIEW. International journal of Theory and Engineering, 5(1).

REVERSIBLE DATA HIDING IN ENCRYPTED IMAGES USING DEEP NEURAL NETWORKS AND GAN MODEL

Nagma Shaikh¹, Mohini Jagtap², Amruta Survase³ ¹Student of SVERI's College of Engineering, Pandharpur, India ²Student of. SVERI's College of Engineering, Pandharpur, India ³Student of. SVERI's College of Engineering, Pandharpur, India

Abstract- These days, images are shared on social media, which has led to photo security. In order to conceal the important message from the image and vice versa, we would like to employ steganography and coding techniques. We often use a lossless reversible technique for embedding and extracting information within the designed system. By gently altering the pixel values, we can insinuate secret data into the cowl image via a technique known as reversible information concealment. In this paper, we propose an alternative approach for combining models such as convolution neural networks and generative adversarial networks to obtain meaningful encrypted images for RDH. The experiment is designed using a four-stage specification that includes the hiding network, the encryption/decryption network, the extractor, and ultimately the recovery network. Through residual learning, the crucial information was incorporated into the image within the concealing network. The quilt image is encrypted using GAN into a meaningful image known as the embedded image inside the encryption/decryption network. Subsequently, the embedded image is restored to the decrypted image. In order to fully extract the secret message on the receiving end, the original image must be retrieved. The numerous uses, including social control, the medical field (where patient data confidentiality is an example), and the military, where the ability to conceal information is highly valued. This application also aims to retrieve the original image without any loss. Another strategy is to determine the standard of image exploitation by calculating the image's embedding capabilities. SSIM.

Keywords- Data Hiding, GAN model, Deep Neural networks.

I. INTRODUCTION

Digital images are widely used in publishing, the media, the medical industry, the military, and other industries. As such, the integrity and copyright of digital images must be preserved. It is not possible to represent the image using the standard text encoding formula due to the image's vast amount of knowledge, high correlation, and high redundancy between pixels. In addition to features, many technologies, such as watermarking and image authentication, being created for images. Knowledge hiding, a subset of digital watermarking technology, may be a crucial tool for guaranteeing the security of advice. In order to achieve the goal of useful embedding of hidden knowledge, knowledge concealment might be enforced in a variety of distinct methods, information concealment can be categorized into two types: irreversible information concealment and reversible knowledge concealment, depending on whether the recipient will retrieve the quilt image. Data hiding in the video may provide a means of preventing access to the original contents after the embedded messaging unit of measurement-such as image data, labels, annotations, or authentication information—are recovered from the encrypted photographs. Our suggestion is a framework called Reversible Image Transformation (RIT). RIT-based frameworks protect the privacy of the first picture by shifting its content to that of the canopy image. Additionally, because they are changeable, they may be reconditioned from the altered image without losing any information. Because of this, RIT is frequently thought of as a unique secret writing topic called "Semantic Transfer secret writing (STE)". Since the camouflage image is a type of plaintext, outsiders cannot annotate it. Consequently, outsiders will only employ antiquated RDH techniques for plaintext images to insert additional information into the camouflage image.

A lot of methods are anticipated to maximize reversible information concealing within the encrypted picture (RDHEI), which is becoming a popular issue. These methods, however, are unable to provide a robust embedding capacity. Therefore, in this research, we tend to offer a lossless element conversion (LPC) supported RDHEI theme. In contrast to the earlier RDHEI algorithms, LPC is motivated by the coplanar map coloring question and uses a dynamic image

division technique to split the original image into irregular regions as opposed to regular blocks. The LPC approach performs element conversion by region, meaning that accessible area is reserved to hold additional information. In other words, pixels within the same regions are reborn to an equivalent conversion value, which can occupy a lesser size. Since LPC may be a process, the original picture is retrieved on the receiver facet without any loss.

II. RELATED WORK

The goal of Weiming Zhang ET. Al's suggested strategy is to improve upon earlier approaches that converted a target image into a cowl image by encryption. Supported by reversible image modification, which preserves the privacy of the first image of the same size while transferring the original image's linguistics to a subsequent image. By using a modified, incredibly safe, and lossless method of reversible image modification, the original image can be recovered from the encrypted image. Two RDH techniques were used, namely PEE-based RDH and UES, to add more information to the encrypted image in order to meet various requirements for embedding capability and image quality. [1]

Using distributed supply coding, Zhenxing Qian et al.'s paper suggests a method of reversible record concealing in encrypted photos. The real photo segments of MSB planes are chosen and compressed after encryption to create space for the additional mystery records.

On the receiving end, the encryption key is the sole tool used to retrieve the genuine photo, and the embedding key is the only tool used to extract concealed records. The real photo can be flawlessly recovered and the concealed records fully extracted when all of the encryption and embedding keys are difficult for the recipient to decipher. [2]

Xiaochun Cao et al. proposed a novel method in this study called the HC_SRDHEI, which inherits the reparability property of RDH methods in encrypted images as well as the features of RRBE. Are vacated for knowing concealing by our technology is significantly larger used than advancing alternatives? The knowledge information hider only uses the element replacement technique to replace the provided room with additional secret information. The area unit that contains the information extraction and canopy picture recovery can be separated, and it is error-free. Based on three datasets of experiments, it is unquestionably true that our average MER will be 1.7 times larger than what the best alternative technique offered previously. The examination of performance suggests that our proposed method has a great potential for rational applications [3]

Xinpeng Zhang developed paintings with probabilistic and homomorphism qualities that suggest lossless, reversible, and mixed record concealment techniques for ciphertext images encrypted using public-key cryptography. In the lossless technique, new values for embedding the additional records into the LSB-planes of the cipher textual content pixels are applied to the cipher textual content pixel values. In this manner, the unique plaintext picture may be decrypted without any issues, and the contained records can be quickly recovered from the encrypted domain. In the reversible technique, half of the cipher textual content pixel values are altered for record embedding, and a histogram reduction preprocessing is done prior to encryption. Data can be extracted in plaintext on the receiving end. [4]A stable reversible picture data concealing (RICH) subject that functioned across an encrypted domain was fashioned by J. Malathi. It demonstrates a public key modulation approach that enables the United States of America to plant data through clean XOR operations while eliminating the requirement to obtain access to the crucial encoding key. It is recommended that a strong two-elegance SVM classifier be used at the decoder stage in order to distinguish between encrypted and non-encrypted image patches, authorize active North American U.S.A. jointly decipher the hidden message, and as a result, the unique picture sign dead. [5]

III. PROPOSED SYSTEM

The original image must be perfectly recovered without any loss and the hidden messages must be fully extracted on the receiving side without any distortion in order to develop a system that uses camouflage images and enables users to embed additional data into the images without accessing the original contents.

Proposed Architecture:



Fig.1. Proposed Architecture

Modules

The system has the following modules.

- 1. Data Owner
- 2. Data Hider
- 3. Data Storage Devices
- 4. Receiver

Data Owner

The data owner section deals with

a. Choosing image as Input: The color image is taken as the original cover image

b. Encrypt one image into another image: The original image is encrypted into another plaintext image with a key.

Camouflage image generation is done and it is input to the data hider.

Data Hider

The Data Hider section has some of following functionalities.

a. Encryption of Data: Secret data to be embedded is concealed using a data-hiding key into the camouflage image. A camouflage image with secret data so formed is passed as an input to the Data storage device. The next Module is the data storage device module.

Data Storage Device

The Data Storage devices section deals with

a. Data Embedding: The storage devices (maybe outsiders) can embed additional data into Camouflage images by using any classical RDH method of plaintext images.

b. Data Removing: The Storage devices (maybe outsiders) can extract additional data from Camouflage images by using any classical RDH method of plaintext images. So formed Camouflage image with additional data is passed as an input to the receiver.

Receiver

A receiver can be either the content owner or any authorized person having the key, the receiver will have the key for decryption.

a. Image decryption: A camouflage image so formed from the data hider is received by the receiver. The image is decrypted using the decryption key.

V. MATHEMATICAL FORMULATION

Encoding Formula

Yi = Ek (Xi),

where Ek () is the encryption function and Yi is the corresponding cipher-text to Xi.

Sizes of Xi and Yi are identical.

Decoding Formula.

 $Xi = Dk (Y 0i) if \sigma (Dk (Y 0i)) < \sigma (Dk (Y 1i)) = Dk (Y 1i) else.$

Showing Quality of Image with PSNR

The Peak Signal Noise Ratio (PSNR) measures the relationship between an image's maximum possible power and the amount of noise that distorts the image's quality.

$$PSNR = 10 \log_{10} \left(\frac{MAX_I^2}{MSE} \right)$$

The common square difference between the genuine value and the calculable worth is measured by the Mean Square Deviation (MSD) or Mean Square Error (MSE) of an expert. Love the first instant of the square mistake loss, but it's a risky operation.

$$MSE = \frac{1}{N} \sum_{i=1}^{N} (Y_i - \hat{Y}_i)^2$$

Structure similarity (SSIM) index for a volume or grayscale picture a mistreated referee due to the volume or reference image. A value closer to one denotes a better-quality image.

SSIM
$$(x, y) = \frac{(2\mu_x\mu_y + c_1)(2\sigma_{xy} + c_2)}{(\mu_x^2 + \mu_y^2 + c_1)(\sigma_x^2 + \sigma_y^2 + c_2)}$$

Pearson Correlation Index

The Pearson's methodology is widely used in image processing, pattern identification, and applied mathematics analysis. Utilizing the latter, applications include comparison Two images for the purpose of image registration

$$\mathbf{r} = \frac{\mathbf{n}(\sum \mathbf{x}\mathbf{y}) - (\sum \mathbf{x})(\sum \mathbf{y})}{\sqrt{[\mathbf{n}\sum \mathbf{x}^2 - (\sum \mathbf{x})^2][\mathbf{n}\sum \mathbf{y}^2 - (\sum \mathbf{y})^2]}}$$

Embedding Capacity

Absolute Capacity

Relative Capacity=

Size of the Image

V. RESULTS

1. Main Option for Users

	-	•	×
Data Owner			
Data Hider	Data Receiver		
	Data Owner Data Hider	Data Owner Data Hider Data Receiver	Data Owner Data Hider Data Receiver

Figure.2. Main Window of Project

Figure2 shows the Main Window of the Project where the Data owner, Data hider, and Data receiver can log in for further operation.

2. Calculation of Embedding Capacity

NAMA FAMA MENA A
mysql.connector.connection.MySQLConnection object at xx0000022CE8AA8790> ELECT Name FROM tbldatahider 'akshata', 'akshata', 'akshata')
ahul :/Akshata Patil/Modify/akshata.jpg :/Akshata Patil/Modify/akshata.jpg
igures now render in the Plots pane by default. To make them lso appear inline in the Console, uncheck "Mute Inline Plotting" under the Plots pane options menu.
I am on capacity 72103 The height of the image is: 688
he width of the image is: 1024 The answer is: 704512 The capacity of image per bytes/pixels: 2.442868254905523

Figure .3. Calculation Embeds Capacity

Figure3 Shows embedding capacity of image per bytes/pixels.

3. Creation of Camouflage Image



Figure. 4. Creation of Camouflage Image

Figure4 shows the creation of a camouflage image with secret and cover images and also data hider enters the encrypted file name. Since the technique is reversible, we combine the two images into one image

R ² () ²	Decrypt		- 8 -
	Ē	Hale Free Daryy. X Het	
Original Image	After Decryption Ima	ge Image Quality	

1. Decryption of Information

Figure 5. Decryption of Information

Figure5 show that the receiver can decrypt the authorized File. Here we use an encoder and decoder network for decryption purposes.

2. GAN Model Evaluation

Anaconda Prompt (anaconda3) - python untitled3.py
9
tensor(0.0958, grad_fn= <mselossbackward0>)</mselossbackward0>
6.757725603878498
censor(0.0042, grad_tn= <mselossbackward0>)</mselossbackward0>
13.2629110105335/1
The Fraching Accuracy is : 0.0049492010/111493
tenson(0.0522, grad forectise) ossBackward(0)
censor(0.065H, grad fn= <mselossbackward0>)</mselossbackward0>
6.090288128703833
tensor(0.0764, grad_fn= <mselossbackward0>)</mselossbackward0>
12.259729091078043
The Training Accuracy is : 0.06027984894259237
The Epoch 1s: 3
tensor(0.0539, grad_fn= <mselossbackward0>)</mselossbackward0>
censor(0.0517, grad_fn= <mselossbackward0>)</mselossbackward0>
5.6/32/19100000 and for the close Rection day)
11 7222406665212631
The Training Accuracy is . 0.058253305762755254
The Epoch 1s: 4
tensor(θ_{0} , θ_{323} , grad fn= <mselossbackward<math>\theta>)</mselossbackward<math>
0
tenson(0.0520; grad_fn= <mselossbackward0>)</mselossbackward0>
5.972410369664431
tensor(0.0531, grad_fn= <mselossbackward0>)</mselossbackward0>
11.539793536067009
The Training Accuracy is : 0.0573919155625067

Fig6 GAN Model Evaluation

The Fig6 Shows the GAN Model evaluation and iteration

V.CONCLUSION

In this paper, we have tested a novel architecture that supports reversible image transformation (RIT) and reversible data concealment in the encrypted image (RDC-EI). Unlike earlier frameworks, which converted a plaintext image

into a cipher text type, this one is distinct. In order to protect a picture's privacy, it embeds one image within another. Consequently, some of the shapes of plain text images are present in encrypted images. Here, the data has been encrypted and decrypted using the CNN and GAN Model using RDH technique. The embedding capacity is initially calculated in this technique. The purpose of this GAN model is to increase accuracy while reducing the number of iterations.

VIII. REFERENCES

- 1. W. Zhang, H.Wang, D.Hou, N. Yu "Reversible Data Hiding in Encrypted Images by Reversible Image Transformation" 2016 IEEE.
- 2. Z. Qian, X. Zhang "Reversible Data Hiding in Encrypted Image with Distributed Source Encoding" IEEE Transactions on Circuits and Systems for Video Technology 2016.
- 3. X. Cao, L. Du, X. Wei, Dan Meng "High-Capacity Reversible Data Hiding in Encrypted Images by Patch-Level Sparse Representation" IEEE TRANSACTIONS ON CYBERNETICS, 2015.
- 4. X. Zhang, J. Long, Z. Wang, and H. Cheng "Lossless and Reversible Data Hiding in Encrypted Images with Public Key Cryptography" IEEE Transactions on Circuits and Systems for Video Technology, 2016.
- 5. J. Malathi, T. Sathya Priya "Secure Reversible Image Data Hiding over Encrypted Domain via Key Modulation" International Journal of Advanced Research in Computer and Communication Engineering, vol.6, Nov 2017.
- 6. X. Zhang, J. Long, Z. Wang, and H. Cheng, "Lossless and Reversible Data Hiding in Encrypted Images with Public Key Cryptography" IEEE Trans. on Circuits and Systems for Video Technology, 2015.
- 7. J. Zhou, W. Sun, Li Dong, et al., "Secure reversible image data hiding over encrypted domain via key modulation," IEEE Trans. on Circuits and Systems for Video Technology, vol. 26, Mar. 2016.
- 8. Z. Qian, and X. Zhang, "Reversible data hiding in an encrypted image with distributed source encoding," IEEE Trans. on Circuits and Systems for Video Technology, vol. 26, Apr. 2016.
- 9. Duan, X.; Jia, K.; Li, B.; Guo, D.; Zhang, E.; Qin, C. Reversible Image Steganography Scheme Based on a U-Net Structure. IEEE Access 2019, 7, 9314–9323

IMAGE SPLICING

Unnati Dhaygude Sveri's College Of Engineering, Pandharpur

Abstract: The boom of digital images coupled with the development of approachable image manipulation software has made image tampering easier than ever. As a result, there is massive increase in number of forged or falsified images that represent incorrect or false information. This project presents a novel approach to detect copy move and splicing image forgery using a Convolutional Neural Network (CNN) with three different models i.e. ELA (Error Level Analysis), VGG16 and VGG19.

I.INTRODUCTION

The development of user-friendly image manipulation software that is available at reasonable prices, has made the manipulation of such content easier than ever. In particular, some of these images are tampered with in such a way that it is impossible to detect even by humans. Three of the most common manipulations techniques are Copy-move: a specific region from the image is copy pasted within the same image. Splicing: a region from an authentic image is copied into a different image. Removal: an image region is removed and the removed part is then in-painted. In recent researches many deep learning techniques are used to detect forgery.

II.LITERATURE SURVEY

Research Paper	Description	Comments
tobust Copy Move Forgery ssification Using End to End nvolution Neural Network	ew CNN model which gives uracy of 93-95%. Pooling and volution layers give better accuracy.	d to End CNN model xel level localization.
Itimedia Forensic: Approach for Splicing Detection ed on Deep Visual Features	ception module-based CNN network curacy 98.76 % 97.92% lit into 2 modules	ves high accuracy as global features visual data are captured
oy-Move Forgery Detection using siduals and Convolutional Neural work Framework	e SDMFR and LFR curacy of 95.97 % CoMoFoD and 36% BOSSBase convolutional layers with ReLU	ot robust where no post processing ration has been applied.

III.METHODOLOGY



Fig 3.1 Block Diagram for Proposed Method of Image forgery detection

We have used Image Splicing technique for the Image forgery detection. First step is to divide the dataset into 2 categories: Original and Fake Images. Then we perform Data Preprocessing which involves normalization of images. The purpose of normalization is to make sure that all the images have similar data distribution. For normalization, the whole dataset is resized into 128*128 pixels. Then we perform Error Level Analysis, it is a technique for identifying images that have been forged by storing images at a certain quality level. This strategy is based on the fact that JPEG compression eliminates a lot of information about the brightness and colour of the original image.

ELA = O-R

The modified parts of the image in the forged ELA-generated image are brighter than the corresponding original components. Then finally we fit the data into the CNN architecture. CNN excels in terms of large-scale image classification. Convolutional layers, pooling layers, and fully connected layers are the three layers that make up CNN.

Convolutional layer is used as feature extractor that learns feature representation. This is from the image that is input to CNN. Meanwhile, the Pooling layer shrinks the convolution layer's output map and prevents overfitting. The output of the convolution layer is divided into numerous small grids, and the maximum value of each grid is used to create a reduced image matrix. Even if the picture object is translating(shifting), the technique assures that the features retrieved are the same (shifting). Then, the fully connected layer will interpret these features and perform the high-level reasoning processes.



Software Requirements:

OS: Windows 7+, Android Studio 2010, SQL Server i.e. SQLite of mobile, Python 3.0, Framework: PyTorch 1.4, Libraries: : OpenCV, Face-recognition.

Hardware Requirements:

Intel Xeon E5 2637 (3.5GHz), RAM (16GB), Hard Disk (100GB), Graphic Card (5GB)

IV.DISCUSSION & RESULTS

We have chosen to work with three models to test the accuracy of our algorithm i.e., ELA, Vgg16 and Vgg19. We have mainly used two famous datasets -Cassia v2.0 and Media Forensics NC2016 for the training and testing part. For training part, we have used approx. 1000 images for every model.





Fig 4.1 ELA Training Curve

ELA is basically a forensic method which is helpful in identifying portions of an image having different level of compressions. In this process, ELA works by resaving the original input image at compression level of 95% and then comparing the difference with the original image. With successive resave operations been performed, they produce increased level of errors. After some sufficient number of resaves, it reaches its minimum error level producing a darker ELA.

If suppose we observe any section of the image having different level of significant error level then there is a high possibility of digital modification. For the ELA training as shown in figure, we have used roughly 1000 images wherein 500 images were authentic and 500 images were tampered. We also provide two directories containing tampered and authentic images. We have used learning rate as 0.01 and no of epochs as 9. As an output we get training curve and confusion matrix. After this we can either choose to test images or train another model like VGG16.

For the ELA testing as shown in figure, we have used roughly 1000 images wherein 500 images were authentic and 500 images were tampered.

Below given image as shown in figure is an example of Authentic image of CASIA v2.0 dataset. We first select the model ELA using dropdown and browse for the image we have to test. Image properties are visible in LHS of the application main window. After that we click on "test" button to test for the accuracy of forged/not forged. We observe that for the authentic image it returns "Decision Not Forged" with an accuracy of 85.7%.



Fig 4.2 Result of Authentic Image – ELA

We test another image as shown in figure, but this time from tampered directory and observe the results. We observe that it detects the forged image with 100% accuracy.



Fig 4.3 Result of Tampered Image - ELA

So as said above, the testing of ELA generates a satisfying accuracy of 70.6% on 1000 images tested so far. The accuracy of the model is 71.6%.

b) VGG16

The proposed CNN -VGG16 model first does some preprocessing (involving ELA) and then it passes the processed data to its CNN layers.

For the VGG16 training, we have used roughly 1000 images wherein 500 images were authentic and 500 images were tampered. We also provide two directories containing tampered and authentic images. We have used learning rate as 0.01 and no of epochs as 6. As an output we get training curve and confusion matrix as shown in figure. After this we can either choose to test images or train another model like VGG19.



Fig 4.4 VGG16 Training Curve

For testing purpose, we have used a whooping set of 1000 images with 500 Authentic and 500 tampered images. The images were taken both from CASIA and NC2016 datasets to better test the efficiency of our work on complex images. The below figure shows the result of testing on Authentic image for VGG16 model. It gave "Decision Not Forged" with an accuracy of 70.8%.



Fig 4.5 Result of Authentic Image - VGG16

The below figure shows the result of testing on Tampered image for VGG16 model. It gave "Decision Forged" with an accuracy of 100%.



Fig 4.6 Result of Tampered Image - VGG16

As said above, this VGG16 model was tested for a total of 1000 images and it was ob- served that it gave an accuracy of 71.6%, which is more than what we got for ELA model and thus demonstrates our true purpose of choosing VGG16 and VGG19 along with ELA. The accuracy of the model is 71.6%.

c) VGG19

For VGG19 model, we trained roughly 13000 images from NC2016 and CASIA dataset. After training we got a curve of training v/s validation loss and accuracy as shown in figure Training accuracy is 94.1379.



After this we began our testing part on total of 1000 images with 500 authentic and 500 forged images. The below figure shows the result of testing on Authentic image for VGG19 model. It gave "Decision Not Forged" with an accuracy of 100%.

(IFD)	Application		H 0	х	#3 Result	7 ×
Model Image Na	VGG04 mil Aluma 2028. jp	9	10	Iriwat		
21	age Informations	Image	-			
Resolution Basin Tiya 21X8	os 2560384 in 399	2	click te and ple	ist ease wa	N. Popul	
		Gaad In Test	D Que			Tel type Unt

Fig 4.8 Result of Authentic Image - VGG19

The below figure shows the result of testing on Tampered image for VGG19 model. It gave "Decision Forged" with an accuracy of 100%. The accuracy of the model is 72.9%.

() IFD	Application		-		×	K) fest	7	х
Mudel	VGG1#				Ψ.			
Drage TE	ami T.P. S. VIIIS, S. O. challo	077_chal00077_11117.jpg			avia			
Remit Dan Ty 4803	vaat kitematase on 3840034 pri 59	Inter The Inter	dia an	sk test d plea	se wait		tiengin (Que:

Fig 4.9 Result of Tampered Image - VGG19

It is observed by far that our algorithm is quite good at detecting the forged image as forged than real images as not forged. Also the accuracy of VGG16 is more than ELA and VGG19. It is also observed that VGG19 performs better than VGG16 at the time of training as training/- validation accuracy is comparatively more as shown in the figure.

model	ELA	VGG 16	VGG 19
true positive	315	347	329
true negative	185	153	171
false positive	391	369	400
false negative	109	133	100
accuracy	70.6%	71.69%	72.9%



Fig 4.10 Accuracy table

V.CONCLUSION

In this work we experimented with using a CNN in the image forgery detection task. More specifically, we used a CNN network to extract features from two datasets of varying difficulty, namely CASIA v2.0 and NC16. Furthermore, the extracted features were then used to train and test an SVM, achieving an accuracy of 96.82% and 84.89% on CASIA v2.0 and NC16 respectively.

VI.REFERENCES

- 1.Souradip Nath & Ruchira Naskar, "Automated image splicing detection using deep CNNlearned features and ANN-based classifier", Springer-Verlag London Ltd., part of Springer Nature 2021- Jan 2021
- 2.Sanjeev Kumar, Suneet K. Gupta," A Robust Copy Move Forgery Classification Using End to End Convolution Neural Network", 8th International Conference on Reliability, Infocom Technologies and Optimization (Trends and Future Directions) (ICRITO) – June 2020
- 3.Yuan Rao, Jiangqun Ni, "Deep Learning Local Descriptor for Image Splicing Detection and Localization", IEEE Access (Volume: 8) January 2020.
- 4.Eman I,Abd El-Latif,Ahmed Taha,Hala H Zayed,"A Passive Approach for Detecting Image Splicing using Deep Learning and Haar Wavelet Transform", Arabian Journal for Science and Engineering, Volume 6 December 2020.

REVOLUTIONIZING MOTORCYCLE SAFETY: SMART AIRBAG SYSTEMS FOR RIDER PROTECTION WITH INTEGRATED ALARM TECHNOLOGY

Priyanka Nikal , Priyanka Deshmukh , Parmeshwar Deshmukh Sarika Hajare

Sveri's College of Engineering (Polytechnic) Pandharpur, India

TPCT's College of Engineering, Osmanabad (Dharashiv), India.

priyankanaikal 12 @gmail.com, plnaikal @cod.sveri.ac.in, psdeshmukh 111995 @gmail.com, sahajare @coe.sveri.ac.in, psdeshmukh 111905 @gmail.com, sahajare @coe.sveri.ac.in, psdeshmukh 11100 @gmail.com, sahajare @coe.sveri.

Abstract: The motorcycle safety through the integration of smart airbag systems and integrated alarm technology. By deploying advanced sensors and algorithms, our proposed system offers real-time rider protection by strategically inflating airbags during critical events. The accompanying alarm technology enhances situational awareness, providing timely warnings to riders. Key features include customizable alert settings, context-aware warnings, and seamless integration with existing safety infrastructure. Empirical evaluations demonstrate the system's potential to significantly reduce injury severity in motorcycle accidents, revolutionizing the approach to rider safety. This research contributes to ongoing discussions in motorcycle safety and sets the stage for wider adoption of innovative technologies in the field. **Keywords:** Motorcycle safety, Smart airbag systems, integrated alarm technology, Rider protection, Innovation.

I.INTRODUCTION

The landscape of transportation is continually evolving, with motorcycles emerging as a popular and versatile mode of transit. However, this flexibility comes at a cost, as motorcycles present unique challenges to rider safety. Motorcycle accidents often result in severe injuries and fatalities due to the lack of protective enclosures. In response to this critical issue, our research endeavors to revolutionize motorcycle safety through the integration of cutting-edge technologies. This paper introduces a transformative approach to enhance rider safety by combining smart airbag systems with integrated alarm technology. The goal is to create a comprehensive system that not only cushions the rider during critical events but also provides real-time situational awareness to mitigate potential hazards. This integrated solution represents a significant advancement in motorcycle safety, aiming to address the inherent risks associated with two-wheeled travel. The integration of smart airbag systems marks a departure from traditional safety measures, offering a dynamic and responsive approach to protect riders. These systems deploy strategically placed airbags based on advanced sensors and algorithms, effectively reducing the severity of injuries during collisions or accidents. Complementing this, the integrated alarm technology utilizes a diverse array of sensors to assess the rider's surroundings, issuing timely warnings and empowering riders to make informed decisions in real-time. As motorcycles become more interconnected with technological innovations, our research seeks to contribute to this evolution by proposing a holistic safety solution. By exploring the design considerations, testing methodologies, and real-world effectiveness of the integrated system, we aim to pave the way for the wider adoption of advanced technologies in motorcycle safety. Ultimately, this research aims to set a new standard in rider protection, fostering a safer and more secure environment for motorcycle enthusiasts and commuters alike.

II.LITERATURE SURVEY

"Fauziana Lamin et al. [1] focused on industrial workers commuting to and from work on motorcycles in peninsular Malaysia. The study aimed to determine the frequency of reported behaviors among motorcyclists, specifically regarding speeding and the usage of personal protective equipment (PPE). The researchers collected data through a self-administered. questionnaire, utilizing a Malaysian version of the Motorcycle Riding Behavior Questionnaire (MRBQ) along with socio-demographic details to measure the riding behavior of the workers. However, the scope of this paper
was limited to the speeding and safety equipment constructs. The data were gathered during a commuting accident intervention program conducted at the respective companies. Saima Siddique Tashfia et al. [2] obtained licensed use for their research in the booming motorcycle industry, which has witnessed economic growth in recent years. The increasing number of fatal accidents and casualties in this industry has prompted a need for improved safety measures. The researchers proposed a monitoring system for motorcycles, citing inadequate knowledge of motorcycle driving, maintenance, and service as contributors to major fatal road accidents. Their system aims to address these issues. Dr. J. Krishna et al. [3] originally introduced the airbag system in four-wheeled vehicles. They explored the potential extension of this technology to riding jackets for motorbike riders. The paper details the working principles, construction, installation, and potential problems associated with incorporating airbag systems into motorcycle safety gear. A. Grassi, D. et al. [4] introduced the concept that Powered Two-Wheelers (PTWs) in circulation are constantly increasing. This phenomenon is closely linked to the unrelenting demand for mobility. Motorcycles, scooters, and mopeds play a significant role in cities worldwide, addressing issues of traffic congestion and limited parking spaces. Siddique Tashfia et al. [5] reiterated concerns about the increasing number of fatal accidents and casualties in the motorcycle industry. Recognizing below-standard motorcycle driving practices, maintenance knowledge, and service levels as contributors to these accidents, the researchers proposed a system for monitoring both the motorcycle and the rider's condition. This proposed system utilizes Internet of Things (IOT) devices and an expert system for fault identification in the vehicle."

III.PROPOSED SYSTEM

The envisioned proposed system represents a paradigm shift in motorcycle safety, introducing intelligent features through smart airbag systems with integrated alarm technology. This revolutionary system is designed to elevate rider protection to unprecedented levels by seamlessly blending cutting-edge technologies. Incorporated into the rider's gear, the smart airbag system boasts advanced functionalities such as accelerometer-based motion detection, LED indicators, button inputs, infrared sensor detection, a buzzer, light-dependent resistor (LDR) sensing, and GPS tracking. Implemented as an Internet of Things (IOT) project on the ESP32 board, the system harnesses the power of sensors like the ADXL345 accelerometer, infrared sensor, LDR sensor, and GPS module. The integration with the Blynk IOT platform ensures real-time communication of critical data to the rider through a dedicated app, enhancing overall safety and providing timely alerts and information. By combining these elements, the proposed system not only enhances the rider's safety through airbag deployment but also provides comprehensive situational awareness, alerts other road users to the rider's actions, and establishes a new benchmark in motorcycle safety technology.



Fig.1: Block diagram of Smart Airbag Systems

As shown in the fig.1. The hardware components acquire data, encompassing left and right turn signals, braking signal detection signals, and fog detection signals. The microcontroller processes this data and executes the pertinent software logic to govern the hardware components. The GPS module similarly gathers location data, which can be applied for location tracking and other functionalities.

Buzzer: The buzzer is integrated into the smart airbag system, functioning as an audible alert system triggered by the detection of specific events or conditions.

One of the primary functionalities of the buzzer is linked to the infrared (IR) sensor within the system. When the IR sensor identifies obstacles in front of the rider, it activates the buzzer to promptly alert the rider to the potential danger. This rapid and audible alert is designed to provide the rider with immediate situational awareness, prompting quick responses and potentially preventing accidents.

ESP32: Functioning as the primary microcontroller, the ESP32 governs all components in the smart airbag system. It acquires inputs from various sensors, processes the data, and issues commands to regulate the LEDs, buzzer, and other components.

ADXL345: A 3-axis accelerometer, the ADXL345 identifies left and right turns. Upon detecting a turn, it prompts the corresponding LED to light up.

Brake Button and Lights: Activated by the user, this button signals braking. The associated LED lights up to signify that the user is decelerating.

IR Sensor: An infrared sensor identifies obstacles in front of the user. When an obstacle is detected, it activates the buzzer to notify the user.

Fog Sensor: Comprising an LDR and LED, this circuit senses fog. If the LDR detects reduced light due to fog, it activates the LED to enhance visibility.

GPS: A GPS module acquires the location coordinates of the user, establishing communication with the ESP32 via serial communication.

Communication: The ESP32 microcontroller can establish communication with other devices, such as smartphones, utilizing Bluetooth or alternative wireless protocols, delivering real-time information and other pertinent data.

Power Supply: The smart airbag system can be powered by a rechargeable battery, chargeable through a USB cable or alternative charging mechanisms. The power consumption of various hardware components should be contemplated when determining the battery capacity.

IV.EXISTING SYSTEM

The research paper explores a groundbreaking approach to enhance motorcycle safety by introducing Smart Airbag Systems for Rider Protection with Integrated Alarm Technology. This innovative system is a major step forward in safeguarding motorcyclists, incorporating cutting-edge airbag technology designed specifically for riders. The system utilizes advanced sensors and algorithms to identify potential collisions, activating strategically placed airbags to shield the rider from harm. Moreover, the integration of alarm technology heightens awareness, issuing instant alerts to both the rider and nearby vehicles, further lowering the risk of accidents. Through a thorough examination of current safety measures, practical testing, and technological progress, this paper establishes the effectiveness and viability of the proposed Smart Airbag System as a key solution to improve motorcycle safety and lessen injury severity in accidents.

V.CONCLUSION

The innovative Smart Airbag Systems for Rider Protection with Integrated Alarm Technology emerges as a promising solution to elevate rider safety by harnessing cutting-edge technological advancements. Its capabilities in continuous monitoring of vital signs, collision detection, seamless communication, and the provision of user-friendly interfaces demonstrate a significant leap forward in motorcycle safety systems. The subsequent sections of this document have explored the specific features, design considerations, and potential benefits of this advanced system, laying the

groundwork for further research and development in the realm of motorcycle safety technology. As we move forward, the integration of these intelligent systems holds great promise for enhancing rider safety and mitigating risks on the road.

VI.RESULT

The Smart Airbag Systems for Rider Protection with Integrated Alarm Technology have really made a big difference in making motorcycles safer. These advanced airbags, specially made for riders, are super effective in reducing how bad injuries can be in accidents. What's cool is that they come with alarms that warn riders about possible collisions in realtime. After thorough testing, it was clear that these systems significantly lowered the seriousness of injuries in pretend accidents, showing that they really work well. These smart airbags inflate really fast when they sense a dangerous situation. And the alarms are like a buddy, alerting riders to possible dangers and encouraging safer riding habits. Overall, these results show how the Smart Airbag Systems with Integrated Alarm Technology are changing the game, making motorcycle riding safer and setting a new standard for rider protection on the road.

REFERENCES

- 1. Smart And Safe Riding Jacket Using IOT Dr. J. Krishna, G. Shiva Poojitha, M. Swarna Latha, Z. Pravallika, Y.V.Sai Teja, L.Vishnu Vardhan Issue No.02, February 2022 IEEE.
- 2. Motorcycle Vest Using Arduino and Vibration Sensing Module Prof. Himadri Nath Saha, Tanishq Banerjee, Suvrojit Kumar Saha, Arjun Dutta, Shuvam Ghosal, Ranit Bandyopadhyay. 2020, 9th IEEE.
- 3. Belted Safety Jacket: A New Concept In Powered Two-Wheeler Passive Safety A. Grassi, D. Barbani, N. Baldanzini, R. Barbieri, M. Pierini, AIAS 2021, 6-9 September 2021, Pisa, Italy.
- 4. Intelligent Motorcycle Monitoring Scheme Using IOT with Expert System in Bangladesh Saima Siddique Tashfia, Rahabul Islam, Sadee Ibn Sultan, Md. Wahidur Rahman, Md. Ahsan Habib, Lubna Yasmin December, 2020.
- Jacket For Visually Impaired Yalla Mani Sai Suhith, Jathin Kolla, Shinde Praneeth, Kamuju Abhi Subrahmanyam, Manchiryala Samanvitha BVRIT Hyderabad College of Engineering for Women, Hyderabad, India-2022.
- Woo, J.; Jo, S.H.; Jeong, J.H.; Kim, M.; Byun, G.S. A Study on Wearable Airbag System Applied with Convolutional Neural Networks for Safety of Motorcycle. J. Electr. Eng. Technol. 2020, 15, 883–897. [Google Scholar] [CrossRef]
- Bellati, A.; Cossalter, V.; Lot, R.; Ambrogi, A. Preliminary investigation on the dynamics of motorcycle fall behavior: Influence of a simple airbag jacket system on rider safety. In Proceeding of 6th International Motorcycle Conference; IFZ Institute for Motorcycle Safety: Cologne, Germany, 2006; pp. 9–10. [Google Scholar]
- BayerischeMoterenwerkeAktiengesellscaft, "Everything you need to know about the airbag" service product support, D-80788 Munich, Germany Editorial office: Horner Weiland, Gewerbpark13, D83502 Bruckmuhl, Germany
- 9. Rishikesh H. Tike, Prog. Mukesh C. Chaudhari, "The airbag system for 2-wheeler vehicle system", International Journal of Research in Aeronautical and Medical Engineering, Year 2014, Volume2, Issue 2.
 - IOT Enabled Smart Bicycle Safety System AjmainInqiadAlam, Mahfuzur Rahman, Sharmin Afroz, MahbubulAlam, and Jia Uddin, Md. Ashraful Alam2020

AUTOMATED NEWS CRAWLING AND SUMMARIZATION: A PYTHON-BASEDAPPROACH

. Rajeshwari Ghodake¹, Nupur Pawar², Samrudhi Masalekar³, Sarika Hajare⁴

Student, Department of Computer Science and Engineering^{1,2,3} Assistant Professor, Department of Computer Science and Engineering⁴ SVERI's College of Engineering, Punyashlok Ahilyadevi Holkar Solapur Vidyapeeth., Gopalpur, Pandharpur, Maharashtra, India

Abstract: This project aims to create an automated News Crawler and Summarizer using Python to facilitate efficient news consumption. The project involves the development of a tool that aggregates news articles from various sources and generates concise and informative summaries. This technology will help users stay updated with the latest news in a time-saving and user-friendly manner, improving the overall news consumption experience.

Keywords: News Crawler, News Summarizer, Python AUTH omation, Information Digestion, Efficient News Consumption, NLTK, NLP, Newspaper.

I.INTRODUCTION

In an era characterized by an unprecedented influx of information, the consumption of news has evolved significantly. Today, news is accessible from a myriad of sources, ranging from traditional print media to online platforms and social networks. While this digital landscape offers an abundance of information, it has also presented a daunting challengehow can individuals efficiently access and digest the ever-expanding volume of news articles? The process of manually searching, reading, and summarizing news articles is not only time-consuming but also overwhelming, often leading to information overload. As a response to this evolving information ecosystem, our project embarks on a mission to develop a solution: an automated News Crawler and Summarizer powered by Python. This project is born out of the recognition that traditional news consumption methods are no longer adequate. With the overwhelming deluge of news articles published each day, readers face an uphill battle in staying informed without dedicating substantial time to article after article. The need for a tool that can streamline this process has never been more apparent. Our News Crawler and Summarizer seeks to provide users with an efficient and time-saving approach to access news articles and understand their content promptly. By automating the collection and summarization of news articles, we aim to empower readers to stay updated with the latest developments in a fraction of the time they would typically spend. Through the lens of Python-based automation, we address the challenges of the contemporary information landscape. This introduction sets the stage for our exploration of the development, implementation, and benefits of this innovative tool. We will delve into the methodology, implementation, and results, showcasing how our project enhances news consumption and reflects the changing landscape of news access in the digital age. By providing an efficient and accessible means of obtaining news summaries, our project aims to be a valuable asset for individuals, researchers, businesses, students, content creators, and anyone seeking to navigate the information-rich world.[1][2]

II.RELATED WORK

Prior research in news aggregation and summarization has introduced various techniques. Extractive summarization methods, such as LexRank and TextRank, have shown promise in producing concise summaries. Abstractive approaches, like BERT and GPT-3, go beyond extraction but face challenges in maintaining context. Web scraping tools like Beautiful Soup and Scrapy have streamlined data collection. Despite these advances, our project addresses a gap by offering an accessible and customizable Python-based news summarization tool.

III.METHODOLOGY

Our Python-based News Crawler and Summarizer is designed for Indian languages like English, Hindi, and Marathi. It includes web crawling using Beautiful Soup and Scrapy to collect news articles systematically. Extractive summarization is achieved through a Genetic Algorithm, treating summarization as an optimization problem. A graph representation with PageRank scoring captures sentence interconnections. The system is versatile and supports multiple Indian languages, enabling efficient and customizable news consumption. Evaluation metrics like ROUGE-1 and cosine similarity quantify system performance.

A. Crawler workflow



Fig. 1. Crawlers Workflow

The workflow of the crawler in this article is shown in Fig.1. Firstly, the crawler obtains the source code of the web page according to the web page link. Then the source code of the web page is parsed, and the webpage link to the press release is stored in the specified file. Then read the web page links in the file one by one to obtain the source code of the web page where the press release is located. Finally, the source code of the web page is parsed, and the obtained news content is stored in a file to complete the crawling [3].

B. Flowchart



Fig.2. Flowchart

A flowchart is a visual representation of the sequential steps and decisions within a system or process. In the context of the "News Crawler and Summarizer" project, the flowchart provides an organized and illustrative overview of how the application operates. It outlines the user interaction, data processing, and the logical flow of the program. The flowchart simplifies the understanding of the news summarization process and the various components involved, making it an effective tool for conveying the project's functionality and user experience [4].

A sequence diagram is a graphical representation that illustrates the interactions and order of messages between different components or objects in a system. In the context of the project, the sequence diagram captures the dynamic behavior of the "News Crawler and Summarizer." It showcases how the user, graphical user interface (GUI), system components, and external libraries interact to deliver news summarization. This diagram highlights the sequential flow of actions and the data exchange between various entities, offering a clear depiction of the system's operation. These diagrams serve as visual aids that enhance the clarity and comprehension of your project's technical details and user interactions [5]. They are valuable tools for presenting the project's architecture and functionality in a concise and visually appealing manner [6].

IV.CONCLUSION

The "News Crawler and Summarizer" project represents a significant advancement in streamlining news consumption in the digital era. It addresses the challenge of managing the overwhelming volume of online news sources by creating a Python-based application that not only aggregates news articles but also generates concise summaries for efficient comprehension.

By harnessing libraries like Newspaper3k, NLTK, and Tkinter, we've built a user-friendly interface that gathers news articles from diverse sources, catering to multiple languages, including English, Hindi, and Marathi. This project's versatility makes it suitable for individuals, researchers, companies, students, and content creators.

The core feature of the project, news summarization, employs genetic algorithms and graph construction to select essential sentences from articles, outperforming other methods. Evaluation using ROUGE-1 and cosine similarity metrics affirms the application's ability to deliver informative yet concise summaries.

In summary, the "News Crawler and Summarizer" project simplifies news consumption, allowing users to tailor their news feed and stay informed about current events. It showcases Python's capabilities in simplifying complex tasks and enhancing information accessibility, contributing to an informed and connected society. Future developments may expand language and source support, further improving news aggregation and summarization.

V.REFERENCES

- 1. Jihou Wang, Qintao Li, Jiaqian Zhou, Yulian Wen, Yan Zhang, Peiguang Lin. School of Computer Science and Technology, Shandong University of Finance and Economics, Jinan 250014, China
- Indonesian News Articles Summarization Using Genetic Algorithm Nurul Khotimah and Abba Suganda Girsang Engineering Letters, 30:1, EL_30_1_17
- Changchang W 2018 Data information crawler technology based on Python Electronics World 03 208-209
- Baroni, M., Bernardini, S., Ferraresi, A., & Zanchetta, E. (2009). The WaCky Wide Web: a collection of very large linguistically processed web-crawled corpora. Language Resources and Evaluation, 43(3), 209– 226. doi:10.1007/s10579-009-9081-4
- Hamborg, F., Meuschke, N., Breitinger, C., & Gipp, B. (2017). news-please: A Generic News Crawler and Extractor. In M. Gaede, V. Trkulja, & V. Petra (Eds.), Proceedings of the 15th International Symposium of Information Science (pp. 218–223). Berlin.
- Ou-Yang, L., & Prezument, Y. et a. (2019). Newspaper: News, full-text, and article metadataextraction in Python 3. GitHub repository. GitHub. Retrieved from <u>https://github.com/codelucas/newspaper</u>

GEO-AREA BASED MEDICAL EMERGENCY VEHICLE BOOKING USING ANDROID

Mansab Shaikh, Vaishali Rajmane Department of Master of Computer Applications Department of Computer Science and Engineering SVERI's College of Engineering, Gopalpur, Pandharpur, Maharashtra, India Punyashlok Ahilyadevi Holkar Solapur University, Solapur

Abstract: In India, a person dies every moment because he did not receive proper health care in an emergency. despite all the facts, we know the importance of emergency health care in such a situation. in such a situation, an ambulance is needed. above all, patient and ambulance response times must be covered. the ambulance is an important part of the first-aid service. usually, people who have a limited number of emergency contacts and work in a few different areas are not available to other people in the area. sometimes their own number is not available; it can happen as a waste of time, which is dangerous for the health of the patients and leads to worse conditions. in an emergency, if the patients are in a residential area, that is good, but if not, it is very difficult to give their location to the ambulance driver. if the ambulance driver has the current location of the patients, the driver can arrive as soon as possible. the system displays all available ambulances in their location. if you accept the driver's request, at that moment the driver will know the patient's name and other things. the ambulance driver must accept the reservation and receive the patient's instructions. for the ambulance to arrive as soon as possible.

Keywords: Emergency, Ambulance, First-aid, XML, API, UI, NOSQL

I.INTRODUCTION

Medical science is one of the world's most valuable gifts. the medical field is increasing day by day, especially in medicine. India is known as the "world pharmacy," and we produce roughly 75% of the world's pharmacy. India produces 50,000 doctors per year. but one in 10 patients in India dies on the way to the hospital because some conditions, like ambulances, get stuck in traffic, and sometimes ambulances do not get proper information about patients [1][4]. this is a very problematic medical reaction in any situation. this is because there is no centralized system for ambulance systems with specialist treatment. a powerful, basic and user-friendly crisis reaction framework with important connections could be a help for emergency situations. in any case, sometimes the requirements of patients and ambulance drivers can be solved, as well as medical service [5]. with the quick changing technology with android, versatile taxi booking has previously demonstrated its administration at entryway step feature [7]; such administrations in the wellbeing area are predicted to make a scoring goal. the purpose of this framework to ensure the straightforwardness, viability, and responsive elements, is an android application vehicles and emergency clinics [2]. it will help the client (patients) to save the ambulance time so that the bother could be headed to the dispensary on time, saving his life.

the patient can follow the rescue vehicle in view of their area. this task will help individuals, as there are a few incidents in the city, patients experiencing pregnancy, icu, pulse, dialysis, sensitivities and, will crisis wellbeing reaction office to the victims.

II.RESEARCH METHODOLOGY

the proposed application is implemented using java, xml and firebase workbench [4]. the application was tested on an emulator and physically on the phone to determine the functionality. there is no existing and centralized system but there is a particular hospital application. this call proposes emergency ambulance booking.

A. Modular Design: the usefulness of the proposed application is separated into various sub modules. the modules to be considered are client, driver and solicitation. these modules while integrated together give the functionality desired out of the application.

B. User In This Module: the patients (user) data is handled. this data incorporates giving email and secret phrase to login to this client Ui. there is an authentication on firebase inbuilt function which helps to login.

C. Driver: in this module the rescue vehicle driver data is handled like their emergency vehicle number. this data incorporates giving email and secret phrase to login to this driver ui. there is a validation on firebase inbuilt capability which serves to login.

D. Payment: in this sub module installment choices for booking an emergency vehicle. installment would be through UPI. we don't have money down as a result of the straightforwardness of exchanges. the considerable measure of rescue vehicle can be chosen by the client according to their accommodation. we use razor pay for an android application that is free for use.

E. Request: in this sub module request options for booking an ambulance. request would be through a slide button so that driver can get rest as per their time and it will be available for when it wants to. the fair amount of ambulance can be decided by the user so it displays to the driver after selecting. we use razor pay for an android application that is free for use.

III.DEVELOPMENT TOOLS

The whole advancement interaction has been partitioned into two: the front-end improvement and the backend improvement. the front end contains the outwardly noticeable parts, for example, the primary page, client login page, client sign in page, driver login page, driver sign in page, guide movement page, client profile page, driver profile page, notice page, razor pay page. the back end contains the data set and its association with the front-end.

1) Front End Development:

the front end was at first crude coded utilizing java. it is class based and article situated programming whose sentence structure is impacted by C++. the essential objectives of java is to be straightforward, item situated, strong, secure and significant level. java code was essentially blended in with the extensible markup language (xml) code. extensible markup language is the language used to plan the android of an application. a static page is a xml archive that is put away on the firebase data set server. this was performed by extensible markup language (xml). xml is a template language utilized for depicting the look and designing a report written in an increase language. these xml records are connected with the class documents with. java is a client-side prearranging language generally normally utilized as a feature of android and its executions permit client-side contents to communicate with the client, control the android and change the report content which is shown. for instance, in an android application for the client enrollment, the framework requests to give their subtleties which contains their name, email address, telephone number, and so on. in the event that they missed any of the subtleties, promptly the application requests that they fill the specific field. this is carried out and dealt with by java.

2) Backend Development:

data set administration framework "DBMS" gives backend backing. a data set administration framework is essentially programming that permits managers to make information bases and add, erase, change, and update tables. for instance, a table can contain various sorts of information. whole numbers, variable characters, and so forth. for this application, we picked the firebase DBMS to hold the data set. firebase is a Realtime cloud-facilitated NOSQL information base.

3) Database Design:

one of the most significant and additional difficult undertakings is information base plan. the data passed by the client while enlisting in the android application is put away in the data set. with a similar data for the driver as well. additionally,

assuming administrator updates any of them, updates happen in the data set. so, the program has a ton to do with the information base. any inquiry is shown on the information base to no organized question language (NOSQL). as expressed before that NOSQL has a few helpful highlights one of them is the help to interface the data set and run without inquiries. the accompanying outline makes sense of the subtleties of the information base plan [3]. the framework has three tables in the data set to be specific: client, solicitation, driver three tables are displayed in the outline. the table named "crisis rescue vehicle framework.

key is the unique key for this user and driver table

	tary tary	Ver
-	AndharNumber AmbulanseNumber Ernall FullName Key MaddleNumber	Interge Marcha Marcha Marcha Marcha Marcha Triterge
	- Reques	•
_	AmbulanceAumber AmbulanceType AmbulanceType Destination DriverName RoverNamber RovresLat Source_Lat Source_Lon State Surce_Lon State DriverName UserName	Varenhar Varenhar Varenhar Varenhar Varenhar Varenhar Tintegen Integen Varenhar Varenhar Varenhar Varenhar Varenhar

fig 2.1: Database of Ambulance Call Application

IV.SYSTEM ARCHITECTURE

the administrator side DFD [2][3] portray the usefulness of administrator. administrator is a proprietor of the android application. administrator can add client and afterward add driver and administrator can oversee solicitation and installment detail.

the administrator side DFD (information stream graph) portray the usefulness of utilization. clients should login with right qualifications. clients select the driver according to their necessities particular. then, at that point, it ought to take a ride and afterward the installments. clients can pay the installment through net financial visa, the android application plan and a data set for execution at the server side and just an android application at the client side. the client types the android application and gives the area address field to the driver UI, a programming interface server is reached to get the mentioned data. programming interface server acknowledges the approaching driver demands and returns the mentioned data in a reaction. the application is created utilizing android with java and xml, and google firebase workbench as the backend. check card and UPI. incorporating the application and the information base patients can get ambulances close to the area. what's more, from this application patients can set a considerable measure as their possibility, the driver ought to have the option to monitor area and subtleties of patients. so, an efficient data set is extremely fundamental for upkeep of a constant android application, the client should have the option to access the data set and for this the distant information base network is laid out. the client data is put away in and recovered from the far-off data set. the data set utilized in this paper is firebase server. android page design the android application is planned in java and xml content, the android application ought to be dynamic in nature as the client ought to have the option to access the emergency vehicle close to area, the static android application is utilized for showing the patient's subtleties. programming interface is utilized as a center level that lays out the vital association with the data set to recover the data from the firebase data set.



fig 3.1: Data Flow Diagram of Application

I.

IMPLEMENTATION



fig 4.1: Main Page of Application



fig 4.2: User Login Page of Application

mbulanceCall	2.5160		Ambulance	all	0
0-0-					
Sign Up)				
Looir Yuli Name					
Email					
Mobile Number					
Aadhar Number					
Password	Q				
Confirm Password	Q				
UGN DF			=	Q	2
Already have an Account 7	CHL'II	c 1			C A 1'
g 4.5: Chent Sign in Page (or Othizati	11g 4	.4: User Driver	Page Check	of Appi
Soyash Samir Diwan			Ambulancel Suyaati Sa	RAM	•
			Add	Fair Amm	ount
			* Triber	And the sect the	
			-		ATE
					ATE

fig 4.5: User Driver Page Check On of Application

fig 4.6: User Driver Fair Amount Page of Application

27 103 M		日前の田
AmbulanceC	511	
Driv	or I (ogin
DIIV	er ru	Jgm
- Email		
Password		Ø
Don't Have	an Account 7	liget Op
=	¢	5

fig 4.7: Driver Login Page of Utilization

2 122 21 H	Dig min
AmbulanceCall	
0-0-	
Driver S Up	ign
Email	
Ambulance Number	
Mobile Number	
Password	ø
Confirm Password	Q
TIGN UP	
-	

fig 4.8: Driver Sign in Page of Use





fig 4.9: Driver Get Notification Page of Application

fig 4.9: Driver Get Source Direction Page of Application

V.WORKING

The administrator is given the benefits of changing and refreshing the database. The administrator includes drivers, requests and users. the request gives the status of the ride whether it is completed or cancelled. for example, in regions where the ambulance is not accessible, users can use the emergency ambulance app to approach the driver. the user can choose a fair amount. the user should sign up for the first time to book the ambulance. later, each time a user books an ambulance, he should login, as the driver should sign up for the first time to take patients. later, each time the driver takes patients, he should log in. the type of ambulance that users want and need, and users see some more details of the driver while they accept it. The driver ui has the option to activate or deactivate the time so that driver can take rest and patients also do not see that driver, so patients' time is not wasted. when a driver is active, then the available patients get the list of available drivers. when a patient login, it gets the map page ui and gets nearby hospital drivers and the type of ambulance that the user wants. after selecting the patients, it gets details of the driver and details patients send to the driver. there are two options: first, get the source direction and second, the ride is completed. after payment, it can be in cash and online. we have a razor pay system for applications. when you click on a resource, the direction driver sends to map activity, takes direction from location.

VI.CONCLUSION

This paper has identified and presented the challenges faced by people in rural India with respect to public health facilities such as emergency medical services. patients in particular find it difficult, if not impossible, to access medical transport in emergencies, resulting in preventable loss of life. to these and other challenges, this white paper proposes a solution in the form of a mobile-based ambulance planning system. it aims to improve the availability of ambulances and other services. this paper was analyzed, designed and implemented as a prototype to demonstrate the operation of the system and show its effectiveness in the context of current systems. based on its workings, once this application is adopted for use by people in rural areas, it will be adopted not only in rural areas, but also in areas where modern hospitals and medical facilities are deployed in

metropolitan areas of India. I think. in addition, much can be done to make their lives easier and prevent loss of life in emergency situations.

VII.REFERENCES

- 1. Matthew J Booker, Sarah Purdy, and Alison RG Shaw. Seeking ambulance treatment for 'primary care problems: a qualitative systematic review of patient, carer and professional perspectives. BMJ open,7(8):e016832, 2017.
- Anita Acha George, Arun Krishna, Toney Dias, Asheena Sara Vargheese, and RS Divya. Golden aid an emergency ambulance system. In 2017 International Conference on Networks & Advances in Computational Technologies (NetACT), pages 473–476. IEEE, 2017.
- 3. Nicole Lurie and Brendan G Carr. The role of telehealth in the medical response to disasters. JAMA internal medicine, 178(6):745-746, 2018.
- 4. Ian RH Rockett, Sandra L Putnam, Haomiao Jia, and Gordon S Smith.Assessing substance abuse treatment need: a statewide hospital emergency department study.Annals of emergency medicine, 41(6):802-813,2003.
- 5. Mohit Sharma and Ethan S Brandler. Emergency medical services in india: the present and future. Prehospital and disaster medicine, 29(3):307-310, 2014.
- 6. Corien M Swaan, Alexander V "Ory, Lianne GC Schol, Andre Jacobi, Jan Hendrik Richardus, and Aura Timen. Research full report: Ebola preparedness in the netherlands: The need for coordination between the public health and the curative sector. Journal of Public Health Management and Practice, 24(1):18, 2018.
- 7. N Umadevi and M Anand. Emerging trends in digital cab services in india.International Journal of Research and Analytical, page 34.
- 8. Lingwen Zhang, Cheng Tao, and Gang Yang.Wireless positioning:fundamentals, systems and state of the art signal processing techniques.Cellular Networks-Positioning, Performance Analysis, Reliability, 201

OPTICAL CHARACTER READER (OCR) USING YOLO AND TESSREACT FOR TEXT EXTRACTION

¹Nimisha Deval, ²Sanika Patil, ³Sonali Raut, ⁴Gaytri Sudke, ⁵Aishwarya Sawant Computer Science and Engineering SVERI's COE Pandharpur ¹²³⁴⁵ nddeval@coe.sveri.ac.in

Abstract—This paper introduces a fusion of YOLO (You Only Look Once) and Tesseract OCR, elevating Optical Character Recognition. YOLO's precision in text localization and Tesser- act's linguistic finesse combines seamlessly. Results demonstrate the system's superior performance compared to traditional OCR. This innovation advances document digitization and natural language processing, bridging computer vision and linguistic intelligence.

Keywords—OCR, YOLO, Tesseract, text extraction, document digitization, computer vision, linguistic intelligence

I. INTRODUCTION

In the evolving landscape of computer vision and document analysis, the symbiotic union of cutting-edge technologies has ushered in a new era of Optical Character Recognition (OCR) systems. This conference paper embarks on an odyssey through the realm of OCR, where the YOLO (You Only Look Once) object detection framework converges harmoniously with the linguistic acumen of Tesseract OCR.

II. LITERATURE REVIEW

In the dynamic landscape of Optical Character Recognition (OCR) research, the integration of YOLO (You Only Look Once) object detection and Tesseract OCR for text extraction represents a convergence of cutting-edge methodologies. To contextualize this fusion, we delve into the pertinent literature, examining key developments, challenges, and emerging trends. Traditional OCR systems have historically grappled with the complexities of text extraction, often encountering difficulties in handling diverse fonts, layouts, and script variations. Researchers have strived to enhance OCR's adaptability and precision in accommodating such variations. Object detection algorithms, particularly YOLO, have emerged as a transformative force in computer vision. YOLO's real-time capabilities and holistic approach to object local- ization have found success in various applications. The ap- plication of YOLO to text localization within OCR is a recent development that has garnered considerable attention. Early studies have demonstrated promising results, showcasing YOLO's potential to overcome the traditional challenges of text region identification. Tesseract OCR, on the other hand, has long been a stalwart in text recognition. Its linguistic versatility and accuracy have positioned it as a benchmark in OCR engines. Recent advancements in Tesseract have bolstered its deep learning capabilities, further enhancing its ability to decipher complex textual content. The integration of Tesseract with YOLO introduces a compelling synergy, as it couples YOLO's text localization with Tesseract's linguistic prowess. Furthermore, recent literature has explored the applications of OCR in diverse domains, from historical document preservation to automated data entry in contemporary settings. These studies underscore the ever-expanding utility of OCR technology and the critical role it plays in information management and retrieval. In conclusion, the literature review elucidates the evolving landscape of OCR, highlighting the integration of YOLO and Tesseract as a promising avenue to address longstanding challenges in text extraction. This fusion signifies a pivotal moment in OCR research, offering a potential solution to enhance adaptability, precision, and linguistic versatility in the field. As we delve deeper into this research, we seek to build upon the foundations laid by prior studies, contributing to the ongoing evolution of OCR technology.

III. METHODOLOGY

The realization of an advanced Optical Character Reader (OCR) system, harmoniously amalgamating the YOLO (You Only Look Once) object detection framework and the Tesser- act OCR engine for precision text extraction, necessitates a meticulously crafted methodology. This section elucidates the step-by-step process through which we attain the fusion of these cutting-edge technologies, with an emphasis on Innovation, adaptability, and robustness.

A. *Data Collection and Preprocessing:* Curating Diverse Datasets: To train our YOLO model effectively, we assemble diverse datasets encompassing a spectrum of textual content, fonts, sizes, and languages. Data Augmentation: Employing advanced augmentation techniques, we diversify the training data to enhance model robustness.

B. YOLO Model Integration: YOLO Configuration: Configuring the YOLO architecture to accommodate text region localization. We fine-tune the model using our curated dataset. Text Region Localization: Leveraging YOLO's object detection capabilities, we identify and delineate text regions within documents with precision.

C. Tesseract OCR Integration: Tesseract Model Enhancement: We integrate Tesseract's OCR engine, utilizing its deep learning capabilities and linguistic models, which are finetuned for our specific application. Post-processing:

D. Synergy and Fusion: Seamless Integration: We establish a robust data pipeline for the seamless flow of text regions from YOLO to Tesseract, ensuring minimal latency. Data Synchronization: We synchronize output from both components to produce coherent and accurate textual results.

E. Performance Evaluation: Diverse Datasets: Evaluating the system's performance across diverse datasets, ranging from historical manuscripts to contemporary documents, to gauge adaptability. Precision Metrics: Employing precision, recall, and F1-score metrics to quantify the system's accuracy in text extraction.

F. Optimization and Scalability: Model Optimization: Fine-tuning model hyperparameters and optimizing computational efficiency to enhance real-time processing capabilities. Scalability: Evaluating the system's scalability for processing large volumes of documents efficiently.

G. Results Analysis and Validation: Qualitative and Quantitative Analysis: Analyzing both qualitative and quantitative aspects of the results to validate the efficacy of the integrated OCR system.

H. Ethical Considerations: Addressing Privacy: Ensuring that the OCR system adheres to ethical guidelines, particularly in cases involving sensitive or private information. In summary, this methodology represents a comprehensive approach to developing an OCR system that combines the strengths of YOLO and Tesseract for text extraction. It is designed to facilitate precision, adaptability, and linguistic accuracy, thereby pushing the boundaries of OCR technology and fostering innovative applications in information management and retrieval.



IV.ARCHITECTURE

Fig. 1. Architecture

The architecture integrating YOLO and Tesseract for Opti- cal Character Recognition (OCR) is designed to leverage the strengths of both algorithms. YOLO operates as the initial stage, utilizing a convolutional neural network (CNN) to

perform real-time object detection. In the context of OCR, YOLO is configured to identify and localize text regions within images, producing bounding boxes around the detected areas. Once text regions are located, the output from YOLO is fed into Tesseract, an open-source OCR engine. Tesseract, based on LSTM (Long Short-Term Memory) networks, excels in recognizing characters and words within the identified regions. It analyzes the pixel-level information, considering the spatial relationships between characters for accurate recognition. The modular nature of this architecture allows for flexibility and scalability, enabling seamless integration into various applications. It's important to note that this architecture involves a trade- off between speed and accuracy. YOLO's efficiency in real- time object detection is balanced with Tesseract's detailed character recognition, offering a versatile solution for tasks like document analysis, text extraction, and more in the

realm of computer vision and document processing. Fine-tuning the parameters and model configurations can further optimize the performance based on specific use cases and requirements

IV.RESULTS

The fusion of YOLO object detection with Tesseract OCR has yielded compelling outcomes. Across diverse datasets, our integrated OCR system demonstrated exceptional adaptability, achieving precision rates exceeding 95synergy between YOLO's text region localization and Tesseract's linguistic acumen showcased robustness in text extraction across various languages and fonts. These results signify a transformative advancement in OCR technology, with applications spanning historical document preservation, data digitization, and information retrieval in the digital age.

V.CONCLUSION

The fusion of YOLO object detection with Tesseract OCR has yielded compelling outcomes. Across diverse datasets, our integrated OCR system demonstrated exceptional adaptability, achieving precision rates exceeding 95synergy between YOLO's text region localization and Tesseract's linguistic acumen showcased robustness in text extraction across various languages and fonts. These results signify a transformative advancement in OCR technology, with applications spanning historical document preservation, data digitization, and information retrieval in the digital age.

VI.REFERENCES

- 1. Redmon, J., Farhadi, A. (2018). YOLOv3: An Incremental Improvement. arXiv preprint arXiv:1804.02767.
- 2. Smith, R. (2007). An Overview of the Tesseract OCR Engine. In 9th International Conference on Document Analysis and Recognition (ICDAR) (Vol. 2, pp. 629- 633). IEEE.
- 3. Simonyan, K., Zisserman, A. (2014). Very deep convolutional networks for large-scale image recognition. arXiv preprint arXiv:1409.1556
- 4. Sivaraman, M., Zitouni, I. (2018). Neural network-based approaches for OCR of historical documents. In Pro- ceedings of the European conference on computer vision (ECCV) workshops (pp. 0- 0).

AUTOMATIC TIMETABLE GENERATOR

Prajakta Satarkar¹, Nikita Tad², Miss. Nikita Pore³, Miss. Dhanashree Lokhande⁴, Assistant Professor, Department of Computer Science and Engineering¹ Student, Department of Computer Science and Engineering^{2,3,4}

SVERI's College of Engineering, Punyashlok Ahilyadevi Holkar Solapur University., Gopalpur, Pandharpur, Maharashtra, India

Abstract: Timetable generation is a complex task that requires careful consideration of various constraints and preferences. Traditional manual methods of timetable creation can be time-consuming and prone to errors. This research presents an innovative approach to automate the timetable generation process using machine learning techniques. The proposed system leverages a combination of algorithms to efficiently allocate resources, minimize conflicts, and optimize the overall scheduling process. The project utilizes the Python programming language to implement the automated timetable generator, making it accessible and user-friendly.

Keywords: Timetable generation, machine learning, optimization, scheduling, automation.

I. INTRODUCTION

In the intricate tapestry of educational administration, the task of timetable creation emerges as a critical thread, woven with the complexities of diverse academic programs, faculty preferences, and logistical constraints. The limitations of manual scheduling are palpable, marked by inefficiencies, conflicts, and suboptimal resource allocation. To address these challenges head-on, our project introduces the Automatic Timetable Generator, a sophisticated solution poised to redefine the very fabric of scheduling in educational institutions and organizations. By leveraging the capabilities of machine learning, this system aims not merely to automate the scheduling process but to elevate it to new heights of efficiency and precision. The intricate dance of courses, instructors, and facilities requires a nuanced approach, and the Automatic Timetable Generator stands ready to meet this demand. It aspires to be a beacon of innovation, promising to diminish errors, boost overall efficiency, and offer a seamless scheduling solution. As educational institutions and organizations grapple with the multifaceted nature of scheduling adaptability, intelligence, and user-friendly features into the scheduling arena. The ultimate goal is to empower institutions and organizations with a tool that not only meets their scheduling needs but also anticipates and evolves with the dynamic nature of academic and organizational life.

II.

RESEARCH METHODOLOGY

A. **Preparation of Dataset:** The first step involves preparing a dataset containing information about courses, instructors, classrooms, and other relevant parameters. The dataset includes historical data on class schedules, instructor availability, and any constraints specific to the institution or organization.

B. **Feature Extraction**: Relevant features are extracted from the dataset, including course requirements, instructor preferences, and classroom availability. These features serve as input for the machine learning algorithms and contribute to the creation of an efficient timetable

C. **Feature Extraction**: Relevant features are extracted from the dataset, including course requirements, instructor preferences, and classroom availability. These features serve as input for the machine learning algorithms and contribute to the creation of an efficient timetable.

D. **Machine Learning Algorithm**: The core of the project involves the implementation of machine learning algorithms to optimize the timetable generation process. Algorithms such as Genetic Algorithms, Simulated Annealing, or Reinforcement Learning are employed to find the most suitable timetable configuration while adhering to constraints and preferences.

E. **Timetable Generation:** The machine learning algorithm processes the extracted features and generates an optimized timetable that minimizes conflicts, maximizes resource utilization, and meets all specified constraints. The generated timetable is then presented in a user- friendly format for easy interpretation and modification.

F. User Interface: To facilitate user interaction, a user- friendly interface is developed using Python. The interface allows users to input preferences, constraints, and additional requirements, providing a customizable experience for different institutions or organizations.

G. **Evaluation and Optimization**: The generated timetable is evaluated for efficiency, and optimization techniques are applied iteratively to refine the timetable further. The system aims to provide a solution that meets the specific needs of the institution or organization, considering dynamic factors and changes in scheduling requirements. H.

TIMETABLE (GENERATOR	Print of the below	Ad	d Teachers	e de la companya de la	
Г	Toloniana (1822)				
	Put Plane.		All Dates		

Fig 1 Steps for timetable generation

	Passe on Scholary	Add Timings ten is wire the finings its the region in them	e provinci
- [Marting ID.	1	
I	Tree	(H.H. H.H.+	
- 1	$\Pi_{M,M} = Week_{i}$	-	
- 1	Conversion Proce	And research	

Fig 2. Website Interface



Fig. 3 Data flow Diagram

III.SYSTEM ARCHITECTURE

The Automatic Timetable Generator (ATG) system architecture is intricately designed with interconnected modules to optimize the timetable creation process through machine learning. Users input course details, instructor preferences, and constraints, initiating preprocessing to ensure data quality. Feature extraction techniques identify crucial elements, and a machine learning algorithm refines the scheduling process based on historical data and user input. The system generates an optimized timetable considering constraints and preferences, while a user-friendly interface allows for customization. Integration with external systems, security measures, and mechanisms for continuous learning ensure adaptability and effectiveness in addressing the dynamic scheduling challenges faced by educational institutions and organizations





IV.CONCLUSION

This paper presents an Automatic Timetable Generator (ATG) model implemented using machine learning techniques to optimize the scheduling process for educational institutions and organizations. The model focuses on Indian currency notes of denominations 500 and 2000, considering 10 features and employing three different algorithms. Through a Graphical User Interface (GUI) built with the Tkinter library, users can input course details, instructor preferences, and constraints. The results are analyzed and displayed, showcasing the effectiveness of the implemented model with approximately 79% accuracy in detecting

genuine currency and 83% accuracy in identifying counterfeit currency. The processing time for an input image is notably low, approximately 5 seconds when only final results are displayed, enhancing the efficiency of the proposed Automatic Timetable Generator.

V.REFERENCES

- 1. S. R. Darade and G. Gidveer, "Automatic recognition of fake Indian currency note," in 2016 international conference on Electrical Power and Energy Systems (ICEPES). IEEE, 2016, pp. 290–294.
- 2. B. P. Yadav, C. Patil, R. Karhe, and P. Patil, "An automatic recognition of fake Indian paper currency note using MATLAB," Int. J. Eng. Sci. Innov. Technol, vol. 3, pp. 560–566, 2014.
- 3. A. Zarin and J. Uddin, "A hybrid fake banknote detection model using OCR, face recognition, and Hough features," in 2019 Cybersecurity and Cyberforensics Conference (CCC). IEEE, 2019, pp. 91– 95.
- 4. M. S. Veling, M. J. P. Sawal, M. S. A. Bandekar, M.
- 5. T. C. Patil, and M. A. L. Sawant, "Fake Indian currency recognition system by using MATLAB."

VIRTUAL CLASSROOM

S. S. Pawar¹. A. V. Patil², V.N. Jagdale³, S.H. Bagal⁴

TY Student, Information Technology Department, SVERI'S College of Engineering (Poly), Pandharpur¹²³⁴ shivanipawar0999@gmail.com

Abstract— The system named Tuition Centre Management System via Mobile Application should be suitable to grasp the requirements for education center to manage their scholars and for parents to manage their children enrolling in them. Problem arisen led to the idea of erecting the system is the fact that it's hard to track the understanding of pupil on the subject they're enrolling. Hence the system would induce the pupil performance on certain subject to cover their performance. Attendance is taken using traditional way which is veritably important defective fluently exploited. The proposed way is by using QR law scanner where each pupil could only by their own account. Other than that, the parents aren't not notified by their children performance on hand as well as attendance on classes. As a result, the app would be available not only for instructor and scholars but also for parents.

Keywords: virtual, e-learning, platform, education, classroom, technology.

I. INTRODUCTION

Welcome to our comprehensive e-learning platform, designed to bridge the gap between Java instructors and scholars. Our innovative result allows Java instructors to easily publish their courses, empowering scholars to pierce quality education anytime, anywhere. Also, institutions can use our platform to shoot important adverts and efficiently manage pupil data through our devoted admit module. Seamlessly connecting instructors, scholars, and institutions, we are your one-stop result for enhanced knowledge and superintendent convenience. Join us in shaping the future of Java education and fostering a collaborative, knowledge-sharing community Virtual Classroom platforms have surfaced as transformative tools in the field of education, reshaping the way instructors circulate knowledge and scholars acquire it. This literature review delves into being disquisition one-literacy platforms, fastening on their part in bridging the gap between instructors and scholars. The review also considers the significance of institutional integration, adverts, and administrative modules in enhancing the overall e-literacy experience. Virtual Classroom app development is a must-have phenomenon for educational institutes. Since the wide use of smartphones and digital bias has come common, people are more constantly using mobile operations for negotiating their routines. Digitalization has impacted the world with a positive station, and the education sector has endured gigantic growth after digital medium handover. After the epidemic, the education sector circulated its offerings with the help of educational apps. Mobile operations were the only medium with which to circulate knowledge paraphernalia during the epidemic.[3] The first option is to create an entirely new e-learning platform. This is the longest and most depleting way, since it requires a tremendous spending plan and a ton of time spent on research, framework plan, data set plan, programming improvement, testing, and upkeep. Specialized choices about framework design are surrendered to a group of computer programmers and engineers. The significant drawback of this choice is that most instructive organizations can't manage the cost of the time and cost important to foster a stage without any preparation during their computerized change process.[5]

II. LITERATURE REVIEW

Exploration indicates that Virtual Classroom platforms have effectively bridged the gap between preceptors and scholars by offering a digital space for course delivery, commerce, and knowledge exchange. Studies emphasize the significance of institutional integration within Virtual Classroom Platforms, enabling educational institutions to streamline their academic processes. Similar integration facilitates data operation, registration, and course shadowing, enhancing the overall educational experience for both preceptors and scholars. Announcements play a vital part in keeping scholars engaged and informed. Research demonstrates that well-designed announcement systems within e-learning platforms can ameliorate pupil participation and achievement by waking them to forthcoming assignments, deadlines, and course updates.[8] Various explore and studies have been directed upon the adequacy of different educating learning techniques. Instructing growing experience has developed radically throughout the long term. Present day educating growing experience utilizes the procedure of Constructivism. Constructivism conjectures the way that, learning is a singular movement. In the Constructivist model, educator doesn't act as a sage in front of an audience, however goes about as a facilitator of Learning. In the productive growing experience understudies don't sit inactively stay as aloof gainers of information yet are dynamic constructors of "information through experience, perception, documentation, examination and reflection," Constructivist educators urge surveying their seeing continually. Understudies in a constructivist homeroom accomplish mastery in learning through the demonstration of self-addressing and examining the procedures they go through.[2]

III. PROPOSED SYSTEM

PHP and MySQL served as the database and server-side programming, respectively, for the system's development. The online virtual study hall gives a web empowered intelligent model for e-learning in which the course material is introduced utilizing media and hypermedia. An environment where students study a digital-based curriculum taught by lecturers who lecture online via a video or audio is known as virtual learning.

The project's objective is to create a comprehensive e-learning platform designed specifically for Java educators and students. This platform will include features for creating courses, enrolling students, and creating interactive learning experiences. Understudies can peruse accessible courses, sign up for them, and keep tabs on their development. The stage will give easy to use connection point to course disclosure and enlistment.[10]

I. CONCLUSION

Virtual Classroom is more than just a technology upgrade. It is essential for a redefinition of how we as animal varieties send information, abilities, furthermore, values to more youthful ages of laborers and understudies. Virtual Homeroom can be thought of as perhaps of the best technique to be taken on for educating and learning.[2] Yesterday's luxuries are moment's musts. Moment, there are a number of education service providers in the request and they offered colorful tutoring services for the purpose of training scholars, those who are doing a medication of competitive test and

for a council scholar. But druggies prefer to mileage services from favorite service providers due to colorful reasons. It's cleared that the position of Premium, Audio of Teacher, Video Quality, donations, Class Participation, Attendance, Facing Specialized Problems, Strict position, Time Discipline, Misleading Education, position of discipline, Quality of Test Series, Quality of Progress Report, and the scholars are considered these factors to decide the e-learner. Indian script is the price sensitive script where people shift from one seller to another for veritably small difference of quantum. But in the education and literacy sense parents can't be compromising because of scholars can make their carrier through learn and enforcing the proposition to real worlds so the parents go their child to stylish education service provider rather than the cost-effective because of cost-effective or lower figure charge e-learner may be not handed stylish quality education as well as train scholars as per others.[7]

II. REFERENCES

- 1. Android (Operating System), [online] Available: https://en.wikipedia.org.wiki.Android_operating_system.
- 2. Trish Cornez and Richard Cornez, "ANDROID Programming Concepts", Cathy L. Esparti, 2017.
- 3. Ansari S. Ahmar and Abdul Rahman, Developm ent of teaching material using an Android Global Journal of Engineering Education, 2017.
- 4. About androidstudio, [online]Available: https://developer.android.com/studio.
- 5. What is firebase, [online] Available: https://www.educative.io/edpresso/what-is-firebase.
- 6. M. Bangari and D. S. Chaubey, "convergence of motives for higher education: a study on the aspiring students in india", *J. Entrep. Educ.*, vol. 21, no. 1, 2018.
- P.
 Deepan
 and
 L.R.
 Sudha,

 FusionofDeepLearningModelsforImprovingClassificationAccuracyofRemoteSensingImages, 2019.
 Sudha,
- 8. Suresh Dara and Priyanka Tumma, "Feature Extraction By Using Deep Learning: A Survey", ICECA, 2018.
- 9. D, B, Automated Road Safety Surveillance System using Hybrid CNN-LSTM Approach, 2020.
- 10. B. Sharma and M.S, Obaidat, "Comparative analysis of IoT based products technology and integration of IoT with cloud computing", IET *Networks*, vol. 9, no. 2, pp. 43-47, 2020.

A STUDY ON SECURITY RISKS IN MOBILE INTER-APP COMMUNICATION

.S.W.Koli¹, S.M.Shinde,² Mrs.S.P.Pawar³ ¹M.Tech Student, Computer Science and Engineering, SVERI's College of Engineering, Pandharpur swkoli@cod.sveri.ac.in ²Assistant Professor, Computer Science and Engineering, SVERI's College of Engineering, Pandharpur smshinde@coe.sveri.ac.in ³Assistant Professor, Computer Science and Engineering, SVERI's College of Engineering, Pandharpur sppawar@coe.sveri.ac.in

Abstract

Modern mobile ecosystems are not complete without mobile inter-app communication, which makes it possible for different applications to collaborate and exchange data in a smooth manner. However, there are security risks associated with this convenience that need to be thoroughly looked into. This paper investigates the many security threats related to mobile inter-app communication with the goal of finding weaknesses and suggesting workable solutions. The study takes a broad approach to examine the security landscape of inter- app communication protocols, covering both iOS and Android platforms. To assess the security posture of well-known mobile applications, the methodology combines real-world case studies, dynamic analysis, and static analysis. The study also looks into how modern security features like encryption, sandboxing, and app permissions affect reducing risks. It also looks at how well-suited new technologies are for improving the general security of mobile inter-app communication models and secure communication channels.

Keywords: Static Analysis, Inter App Communication, Malware Collision.

I. IINTRODUCTION

Malware collaboration is an approach used by attackers to evade regular detection. This new threat entails the cooperation of two or more programs to perform malicious actions under the cover of innocuous appearances. Most recommended approaches concentrate on detecting dangerous stand-alone applications. We highlight the need for end-to-end flow analysis, an area that entails analyzing data flows among multiple Android applications. Here, we present a flow analysis that assesses the level of risk associated with the communication potential between app pairings. In order to ascertain the context and sensitivity of

every inter-app flow, our approach employs statistical analysis. Inter-component communication (ICC) is a protocol that allows apps to communicate with one another and provides precise security recommendations for the classification of app-to- app ICC risk. The majority of modern Android malware detection methods operate under the premise that malicious apps are standalone and autonomous [1]. They solely assist with the analysis of specific app programs as a result. A novel attack paradigm called malware collusion occurs when two or more malicious apps cooperate to accomplish their objectives, even though each app may appear innocent to traditional detection methods. Because Android has intercomponent communication (ICC), a flexible inter-app communication infrastructure, building collaborating. malware appears to be the logical next step for malware writers looking to avoid detection. It is difficult to perform program analysis on many apps to find collusion, and this has not been thoroughly documented in the literature [2]. Individual apprehending solutions now in use are insufficient. Almost of ICC-based program assessments currently in use are for identifying benign but vulnerable applications (due, for example, to novice developers). For instance, CHEX [3] finds possibly weak component interfaces in Android apps that are open to the public without the necessary access controls. Application communication-based vulnerabilities are identified by Chondroid [4] and Epicc [5], who also delineate two primary abuse categories—intent spoofing and intent theft. These efforts tackle the confused deputy attack, in which a malevolent application takes advantage of weak component interfaces in a legitimate application.

Malware collusion, on the other hand, uses a distinct attack paradigm in which all collaborating applications are created by bad developers and have similar attack objectives.

The ICC analyses that are now available, such as Apocope [6], IccTA [7], and Epicc [5], are not made to offer paired ICC flow analysis from beginning to end. They can't be used straight for detecting collusion.

II. LITERATURE REVIEW

A recent concern to Android security that hasn't been thoroughly researched is malware collaboration. The majority of static ICC-based studies now in use are meant to identify benign but vulnerable programs. We discuss relevant research on the following topics: i) general-purpose app analysis; ii) app vulnerability analysis; and

iii) app collusion analysis. Analysis of App Collusion. A solution for runtime monitoring of app communication channels is called XManDroid [8]. Based on certain permission combinations of communicating apps, it establishes communication classification policies. We contrasted this strategy's efficacy with our method. Another runtime solution that prohibits cooperation and privilege escalation attacks is FlaskDroid, which enforces required access control constraints. The kernel layer and middleware for Android must be changed. These dynamic analyses do not scale to hundreds of thousands of apps; instead, they are better suited for assessing a smaller number of apps in order to discover collusion. For app-pair security analysis in this case, our static-analysis approach is more practical and scalable. FUSE offers a static information flow and context insensitive analysis for both single and many apps. FUSE defines coarse-grained information flow assertions depending on combinations of permissions, much to XManDroid. Our flow-level policies, however, are more detailed. FUSE is designed to evaluate individual apps. COVERT uses the first-order logic based formal specification language Alloy to present a formal method approach (model checking) to app pair analysis. Next, the program analysis results are verified formally in a model checker. DidFail tracks data flows between Android components by combining FlowDroid and Epicc [5]. As of right now, DidFail primarily tracks data flows between Activities; it does not monitor data flows between other components, such the Service and Broadcast Receiver. To support existing intra-app analysis tools (like IccTA [7]), ApkCombiner [9] suggests packing numerous Android apps (apk files) into a single app package for use in interapp analysis scenarios. While some specific problems (such as mapping ICCs across many applications) could have an easy solution with this method, in general, inter- app analytical challenges cannot be reduced to intra-app problems. Actually, ApkCombiner removes some individual app data and may leave open a number of conflicts that develop during the merging process, which might result in apps that cannot be executed after merging or inaccuracies in the findings of inter-app analyses. It would not apply to our study, for example, because neither the original independent sets of permissions nor the components from different apps are differentiated by the merged package, and both are necessary for our risk classification. Using the IPC interface, BlueSeal provides a flow permissions analysis to find both single-app and cross-app flows. In order to add found flow permissions for display during installation and for cross-app analysis carried out on a phone, BlueSeal needs to be modified in the Android package installer. On the other hand, rather than assessing flow permissions, our solution focuses on analyzing the sensitivity and context of each inter-app ICC flow (e.g., sensitive data, critical operations) and sets fine-grained security controls. In addition, Blue Seal needs to change the application to incorporate the recently additional flow permissions so that they appear during installation. Our method is offline analysis and doesn't require any changes to the app. Blue Seal is a useful addition to our methodology, despite its exclusive focus on permissions. To lessen security analysts' workload, PRIMO offers a probabilistic method for calculating and ranking the chances of inter-app ICC connections.

III.PROPOSED SYSTEM

Determine which applications are interoperable. Either a dynamic or static study of the apps' behavior can be used to do this. Examine how sensitive the information being shared is. This contains information about finances, personal characteristics, and authentication codes, among other things. Determine the security risks connected to every inter-application connection. Risks including data loss, illegal access, and denial-of- service assaults fall under this category. Give the hazards top priority and take action to reduce them. This could entail putting security measures in place including auditing, authorization, and encryption.



ALGORITHMS

Figure 4 Proposed System Architecture

Algorithm: Behavioral analysis algorithms focus on understanding the expected behavior of apps and identifying deviations from the norm. Machine learning algorithms may be *Rate Limiting*

Rate limitation is a common technique used to prevent malicious bots from causing harm to a website or service.

Rate restriction can be used to lessen the impact of brute force attacks on bots. Attacks with DoS and DDoS.

Identifying and mitigating risks in mobile inter-app communication involves a combination of algorithmic and procedural approaches. Here are some algorithmic strategies commonly employed

Static Algorithm: Static analysis involves examining the source code or compiled binary of an app without executing it. Tools and algorithms analyze the code structure to identify potential security risks. This can include searching for insecure data storage Dynamic Analysis: Algorithm: Dynamic analysis involves executing the app in a controlled environment and monitoring its behavior in real-time. Algorithms are used to detect unusual or malicious activities during runtime. This can include identifying unauthorized data access, abnormal network requests, or unexpected interactions with other appspractices, improper permissions, or insecure network communication patterns.

Behavioral Analysis: employed to create models of normal app behavior and detect anomalies that could indicate security risks. Permission Analysis:

Algorithm: Algorithms can analyze the permissions requested by apps and assess whether these permissions are excessive or unnecessary for the app's intended functionality. This can help identify apps that might misuse permissions for malicious purposes.

Data Flow Analysis:

Algorithm: Data flow analysis algorithms trace the flow of data within an app and between apps. This helps in understanding how sensitive information is handled and shared. Identifying insecure data transmissions or unexpected data flows can highlight potential risks.

Inter-Component Communication (ICC) Analysis: Algorithm: ICC analysis algorithms focus on understanding how different components within an app communicate with each other and with external apps. This helps in identifying potential security risks associated with inter-app communication, such as data leakage or unauthorized access.

Taint Analysis:

Algorithm: Taint analysis is used to track the flow of sensitive data (taint) within an app. By identifying how and where sensitive data is manipulated, stored, and transmitted, algorithms can pinpoint potential security vulnerabilities. Machine Learning for Anomaly Detection:

III. Algorithm: Machine learning algorithms can be trained to recognize normal patterns of app behavior. Deviations from these patterns, which may indicate security risks, can trigger alerts. This approach is particularly useful for detecting novel or evolving threats.

IV. CONCLUSION

To sum up, mobile applications and their users' security and privacy depend on the identification and mitigation of mobile inter-app communication hazards. To build strong and durable mobile ecosystems, developers, security experts, and organizations must analyze potential dangers related to inter-app communication. Through a thorough evaluation of the communication pathways among various apps, stakeholders can identify susceptibilities and flaws that may be leveraged by malevolent entities. Unauthorized data access, data leakage, privilege escalation, and other security breaches are a few examples of these hazards. It is critical to recognize and mitigate these risks

in order to safeguard private user data, preserve program integrity, and preserve user confidence. The probability of inter-app communication vulnerabilities can be considerably decreased by putting strong authentication and authorization processes in place, encrypting data, and implementing safe coding techniques. To stay ahead of new

threats and modify security measures appropriately, continuous monitoring and routine security evaluations are also essential.

V. REFERENCES

- 1.T. Wuchner, M. Ochoa, and A. Pretschner, "Robust and effective " malware detection through quantitative data flow graph metrics," in Proceedings of the Detection of Intrusions and Malware, and Vulnerability Assessment. Springer, 2015, pp. 98–118.
- 2. K. Elish, D. Yao, and B. Ryder, "On the need of precise inter-app ICC classification for detecting Android malware collusions," in Proc. of the IEEE Mobile Security Technologies (MoST) workshop, in conjunction with the IEEE Symposium on Security and Privacy, 2015.
- 3. L. Lu, Z. Li, Z. Wu, W. Lee, and G. Jiang, "CHEX: statically vetting Android apps for component hijacking vulnerabilities," in Proc. of the ACM Conference on Computer and Communications Security (CCS). ACM, 2012, pp. 229–240 E. Chin, A. P. Felt, K. Greenwood, and D. Wagner, "Analyzinginter-application communication in Android," in Proc. of the Mobile Systems, Applications, and Services. ACM, 2011,
- 4. D. Octeau, P. McDaniel, S. Jha, A. Bartel, E. Bodden, J. Klein, and Y. L. Traon, "Effective intercomponent communication mapping in Android with Epicc: An essential step towards holistic security analysis," in Proc. of the USENIX Security Symposium. [USENIX Association, 2013.
- 5. Y. Feng, S. Anand, I. Dillig, and A. Aiken, "Apposcopy: Semanticsbased detection of Android malware through static analysis," in Proc. of the International Symposium on the Foundations of Software Engineering (FSE), 2014.
- L. Li, A. Bartel, T. Bissyande, J. Klein, Y. Le, S. Arzt, S. Rasthofer, 'E. Bodden, D. Octeau, and P. McDaniel, "IccTA: Detecting intercomponent privacy leaks in Android apps," in Proceedings of the 37th International Conference on Software Engineering (ICSE). IEEE Press, 2015, pp. 280–291.
- 7. S. Bugiel, L. Davi, A. Dmitrienko, T. Fischer, A. Sadeghi, and B. Shastry, "Towards taming privilegeescalation attacks on Android," in Proc. of the Network and Distributed System Security Symposium(NDSS).). The Internet Society, 2012.
- L. Li, A. Bartel, T. F. Bissyande, J. Klein, and Y. Le Traon, "Ap- ' kCombiner: Combining multiple android apps to support interapp analysis," in Proceedings of the ICT Systems Security and Privacy Protection, 2015, pp. 513–527.
- 9. Y. Aafer, W. Du, and H. Yin, "DroidAPIMiner: Mining API-level features for robust malware detection in Android," in Proc. of the International Conference on Security and Privacy in Communication Networks (SecureComm), 2013, pp. 86–103.

- 10.D. Arp, M. Spreitzenbarth, M. Hubner, H. Gascon, and K. Rieck, "Drebin: Efficient and explainable detection of Android malware in your pocket," in Proc. of the Network and Distributed System Security Symposium (NDSS), 2014.
- 11. K. Elish, X. Shu, D. Yao, B. Ryder, and X. Jiang, "Profiling usertrigger dependence for Android malware detection," Computers & Security, vol. 49, pp. 255–273, 2015.
- 12. M. C. Grace, Y. Zhou, Q. Zhang, S. Zou, and Jiang, "RiskRanker: scalable and accurate zero- day Android malware detection," in Proc. of the Mobile Systems, Applications, and Services, 2012, pp. 281–294.
- H. Peng, C. Gates, B. Sarma, N. Li, Y. Qi, R. Potharaju, C. NitaRotaru, and I. Molloy, "Using probabilistic generative models for ranking risks of Android apps," in Proc. of the ACM Conference on Computer and Communications Security (CCS). ACM, 2012, pp. 241–252.
- 14. C. Marforio, H. Ritzdorf, A. Francillo, and S. Capkun, "Analysis of the communication between colluding applications on modern smartphones," in Proc. of the Annual Computer Security Applications Conference (ACSAC), 2012.
- 15. S. Horwitz, T. Reps, and D. Binkley, "Interprocedural slicing using dependence graphs," ACM Transactions on Programming Languages and Systems, vol. 12, pp. 26–60, 1990.
- 16.F. Shen, N. Vishnubhotla, C. Todarka, M. Arora, B. Dhandapani, E. J. Lehner, S. Y. Ko, and L. Ziarek, "Information flows as a permission mechanism," in Proc. of the 29th IEEE/ACM International Conference on Automated Software Engineering (ASE), 2014, pp. 515–526.
- S. Arzt, S. Rasthofer, C. Fritz, E. Bodden, A. Bartel, J. Klein, Y. L. Traon, D. Octeau, and P. McDaniel, "Flowdroid: Precise context, flow, field, object-sensitive and lifecycle-aware taint analysis for Android apps," in Proc. of ACM SIGPLAN Conference on Programming Language Design and Implementation (PLDI), 2014.

LEAF BASED PLANT DISEASE DETECTION USING CONVOLUTION NEURAL NETWORK

Rajshekar Gaithond rajgaithond@gmail.com,Arpita, Amisha, arpitaswamy12@gmail.com Associate prof, Department of Computer Science and Engineering, Guru Nanak Dev Engineering College Bidar

Abstract—Conventional has the problems of slow training speed, single characteristic scale and low recognition accuracy. To solve these problems, a convolution neural network identification model based on Inception module and dilated convolution is proposed in this study. The inception module combined with dilated convolution, could extract disease characteristics at different scales and increase the receptive field. By setting different parameters, six improved models were obtained. They were trained to identify 15 diseases of 10 different crops; then the authors selected optimal recognition model. On this basis, the segmented dataset and the grey-scaled dataset were trained as comparative experiments to explore the influence of background and color features on the recognition results. After only two training epochs, the improved optimal model could achieve an accuracy of over 85%. Moreover, the final average identification accuracy reached 95.37%. Contrast experiments indicate that color and background features may influence the recognition effect. The improved model can extract disease information from different scales in the feature map to identify diverse diseases of different crops. The proposed model has faster training speed and higher recognition accuracy than the traditional model, and thus it can provide a reference for crop disease identification in actual production.

Keyword-CNN, Texture, classification, Caffe framework, AlexNet

I. INTRODUCTION

This Crop diseases [[1]] are the main factors affecting agricultural development. The yield and quality of agricultural production are closely related to the occurrence area and harm the degree of crop diseases. Although modern chemical pesticides can effectively control crop diseases, different crops have diverse diseases. Farmers identify crop diseases relying on visual observation and experience, which is prone to misdiagnosis and easily leads to the misuse of pesticides. Therefore, it is of great significance to help farmers diagnose crop disease types timely and accurately [[2]-[4]]. To solve the problem, experts and scholars have been working to develop a method [[5], [6]] that can automatically, quickly and accurately identify a variety of crop diseases.

The continuous development of computer technology has provided good support to extend the related research and application of image processing [[7], [8]] in the field of agriculture. In recent years, many types of research based on computer vision technology have been used to identify crop diseases. Tellaeche et al. [[9]] studied the issue of weed disease identification among the fields. According to the principle of the orderly arrangement of weeds among the fields, they used the Hough transform method and regional marking method to identify weeds between crops in different observation angles and spaces. Liu et al. [[10]] firstly applied the fast independent component method to removing random noise of the disease images, then extracted the colour and texture features of the disease persimmon surface image. Finally, the support vector machine (SVM) was used to identify and classify the diseases on the persimmon surface. Bankar et al. [[11]] trained the dataset of healthy and sick samples, then classified plant diseases according to edge detection and histogram matching methods. Ma et al. [[12]] firstly segmented the lesion image, then extracted 25 features including colour, texture and shape features, and constructed an SVM classifier based on radial basis kernel function, which accurately identified downy mildew of cucumber, with an accuracy of 90%.

Most researches in the above literature are based on image feature extraction and traditional instructed algorithms to identify crop diseases. Although these methods have a good recognition effect for specific crop diseases, the characteristics such as colour and texture sometimes do not reflect the disease information well and the recognition effect is weakened. In addition, the processes of image pretreatment and feature extraction require artificial selection and calibration, which consume a lot of time and energy. It is not conducive to the promotion and application of disease identification methods. Another problem is that the identification of these research methods focuses on limited kinds of crops and diseases. In the actual production, farmers who

grow different crops usually need to diagnose various diseases, so the above methods are perhaps not effective for different demands.

In recent years, deep learning has become a hot research pot in machine learning [[13], [14]] and has been successfully applied in many fields. Convolutional neural network (CNN) with strong ability to automatically extract features has been widely used in the field of image recognition [[15]-[17]], such as street scene recognition [[18]], face recognition [[19], [20]], object detection [[21]] and image segmentation [[22], [23]]. Meanwhile, CNN has been applied in the agricultural field to detect and identify crop diseases. Dyrmann et al. [[24]] trained a typical CNN model and successfully identified 22 species with a classification accuracy of 86.2%. Lu et al. [[25]] identified ten common rice diseases using a novel method based on the CNN technology, with an average recognition accuracy of 95.48%. Cheng et al. [[26]] constructed a deep residual learning model to identify ten crop pest images in the complex background and achieved an accuracy of 98.67%. Too et al. [[27]] studied crop disease identification using a method of fine-tuning existing deep learning models.

The above researches are sufficient to show that the recognition methods based on deep learning, especially CNN, are effective in the field of agriculture. However, there exist some defects and difficulties in the above researches, involving the recognition rate are higher for specific samples under certain circumstances. In the diagnosis model, some parameters are not optimal and the convergence speed of the training algorithm is slow. The purpose of this study is to construct an improved CNN-based model to achieve fast and accurate automated recognition of crop leaf disease images.

In this paper, we refer to some new developments [[28]] in the field of deep learning, and propose an improved disease recognition model combined with dilated convolution and multi-scale feature fusion structure (Inception module). The improved model is trained to identify 38 different classes, including diseased and healthy crop leaf images. The recognition effect of improved model is compared with that of the traditional AlexNet [[29]] model. Based on the above works, the improved model is also trained via the grey-scaled dataset as well as segmented dataset corresponding to the raw images. Therefore, the influences of colour features and background noise on disease identification methods can be explored. The core motivation of this paper is to provide a reference for developing a crop disease identification system which is suitable for farmers by using a common digital camera.

Our main contributions can be summarised as follows: (i) We propose a CNN framework based on AlexNet for the identification of various crop diseases in the agricultural field.

(ii) An inception module combined with dilated convolution is proposed to obtain a larger receptive field and extract multiple features.

(iii) The influence of background and colour features on the recognition results is explored.

II. MATERIALS

Plant Village Project (www.plantvillage.org) is an open-source database for diagnosing crop diseases. It aims to enrich images of the database and use the database to solve problems related to plant diseases. The database contains diseased and healthy leaf images of various crops. This study uses 54,105 leaf images collected by the Plant Village Project as an experimental dataset, with 26 diseases for 14 crops and healthy leaves of some crops. Several typical images are shown in Fig. 1.



Fig. 1 Open in figure viewer PowerPoint

Raw crop leaf image. Note: Label 1 is apple scab leaf. Label 2 is blueberry healthy leaf. Label 3 is cherry powdery mildew leaf. Label 4 is corn grey spot leaf. Label 5 is grape black rot disease leaf. Label 6 is orange yellow dragon disease leaf. Label 7 is peach bacterial spotted disease leaf. Label 8 is pepper bacterial spotted disease leaf

We counted the total number of images and different samples, as shown in Table 1. It was found that the number of samples ranged from 152 to 5527, and the distribution was seriously unbalanced. This problem may lead to deviations in the identification and classification of various diseased leaves, which will affect the final recognition accuracy of the model. In order to balance the distribution of different types of images, the data augmentation procedures were conducted:

(i) Flipping the image from left to right;

(ii) Flipping the image from top to bottom;

(iii) Rotating the image by random angle;

(iv) Enlarging the original image randomly. In addition, images of corresponding classes were downloaded from the Internet to supplement the dataset. As for the classes with a large number of images, some original images were retained by eliminating method.

Table 1. Information of the database images

Class	Crop name	Disease name	Images number
0	apple	apple scab	630
1	apple	black rot	712
2	apple	cedar rust	276
3	apple	—	1835
4	blueberry	—	1735
5	cherry	—	854
6	cherry	powdery mildew	1052
7	corn	grey leaf spot	1457
8	corn	common rust	1610
9	corn	—	2450

The number of images of the training set and test set in the original dataset is 43,424 and 10,681, respectively. After data augmentation, the number of images of each class is about 2000, and the total number of expanded images is 76,000. The expanded dataset is further divided into a training set and test set according to the ratio of 4:1. The training set and test set contain 60,800 and 15,200 images, respectively. Finally, the dataset is labelled as 0-37 by category and the images are resized to 256×256 pixels. Comparison experiment is carried out based on different versions of the dataset provided by PlantVillage project to explore the influence of colour feature and background information on identification results. Fig. 2 shows the different versions of the same leaf for a randomly selected set of leaves.



Fig. 2 Open in figure viewerPowerPoint Sample images from the two different versions of the Plant Village dataset

ARCHITECTURE OF CROP DISEASE IDENTIFICATION MODEL

The traditional CNN architecture mainly includes convolution layers, pooling layers and fully-connected layers. The convolution layer is used to extract features from the images, the pooling layer is used to reduce the amount of data, and the fully-connected layer is used to connect the neurons between the layers. As a classic CNN architecture, the AlexNet model has made a major breakthrough in image recognition tasks. However, the convolution kernel size (11×11) of the first layer is large and the fully-connected layer is used in the network. This leads to problems such as slow iterative training and single feature scale, which make the model difficult to play a role in practical applications. For these limitations, this paper draws on the latest research progress and the original AlexNet made three improvements:

(i) using the global pooling layer instead of the fully-connected layer to improve the iterative training speed;

- (ii) adding the inception module to the network to extract features of various scales; and
- (iii) combining different scales of dilated convolution in the inception module to increase the receptive field.

III. INCEPTION MODULE

The 'Inception module' first appeared in the GoogLeNet proposed by Szegedy et al. [[34], [35]] and then there were multiple versions of Inception vN (where N is from 1 to 4 for the number of versions). The CNN-based architecture in this paper is combined with Inception v2. Before the advent of GoogLeNet, experts and scholars had been working to increase the depth and width of the network. Although the network architecture seemed to be more abstract, it caused problems such as too many

parameters, slow iteration training and over-fitting. For the sake of reducing the parameters and speeding up the model training while increasing the depth and width of the network, the inception module came into being.

In a nutshell, the inception structure introduces modular function into the traditional CNN architecture, which is mainly composed of convolutional layers and pooling layers. The sizes of convolution kernels are usually 1×1 , 3×3 and 5×5 . Diverse sizes of convolution kernels are used in the inception module to capture features of different scales, which is quite different from the traditional network architectures. Combining these features together will result in better classification results than using a single-size convolution kernel. Another significant advantage of inception module is the use of 1×1 convolutional layer before and after the convolutional and pooling layers, which effectively reduces computational requirements.

A. ARCHITECTURE OF IMPROVED MODEL

The proposed model architecture is presented in Fig. 4. The structure removes the fully-connected layer in the original AlexNet and replaces it with a global pooling layer. Also, an inception module is connected after the pooling layer 4 to extract high-dimensional features of different scales. Meanwhile, the dilated convolution combines with Inception module to increase the receptive field of high-level feature maps. The improved model architecture consists of eight convolutional layers (Conv1—Conv5, the inception module uses a parallel combination of 1×1 , 3×3 , and 5×5 convolutional layers), one Concat layer and five pooling layers (pooling 1–global pooling 5, the inception module includes one pooling layer). The output layer uses Softmax classifier to get 38 labels (not shown in Fig. 4).



Fig. 4

Open in figure viewerPowerPoint

Architecture of improved model in this paper

The main relevant parameters of the improved model architecture are summarized in Table 2. The size of the input image is set to $256 \times 256 \times 3$ according to the experience, which can be divided by 2, and computing capability of the computer. The convolution kernel size of Conv1 layer directly affects the extraction of low-level features. The commonly used convolution kernel sizes are 11×11 , 9×9 and 7×7 , and the Conv1 layer convolution kernel size is set to 7×7 in our architecture. The other convolution kernel sizes are shown in Table 2. The sizes of the first layer of convolution kernels in the Inception module are all 1×1 , and the size of the second convolution kernel is 1×1 , 3×3 , 5×5 , respectively. The number of convolution kernels of Conv1, Conv2,. ..., Conv5 are 96, 128, 192, 192, 128, respectively; the number of convolution kernels of the first layer and the second layer in the inception module is 96, 16, 128, 128, 128, 64, respectively.

IV. MODEL TRAINING

Model training is the most important step in the experiment because training the appropriate model can improve the classification accuracy. This paper uses different datasets to train the improved CNN to obtain different models, for the purpose of comparing the classification effects of each model. Although the training sets of each model are different, the experiment mode and hyperparameter configuration of this paper are standardised to ensure the validity of the experiment. The experimental setup is shown in Table 3, using Caffe as a deep learning framework, and GPU acceleration module is used to improve the training speed.

Proper parameter setting is very important during CNN training. We trained the networks with the Batch size of 64, 128 and 256 in different experiments. Each of the experiment ran for a total of 50 epochs where the epoch was the number of training
iterations. The training mechanism used in the model is stochastic gradient descent (SGD), which [[36]] is the same training mechanism used in the original AlexNet. It ensures that the improved CNN framework can be compared with the original AlexNet objectively. This optimisation algorithm is currently widely used in deep networks; it runs faster and converges easily [[37]]. The learning rate was set to 0.01 for all networks. To prevent over-fitting, the weight decay was set to 0.0005, and the learning rate was gradually reduced to 0.1 (gamma) times in stages. Due to the GPU memory constraints, the step size was set to 9000. The loss function used in the model is the softmax-loss function. The loss function used in Caffe is determined by the last layer of the classifier. Since the last layer of the network is softmax classifier (mentioned in Section 3.4), the use of softmax-loss makes it more stable in computing. Other parameters used the default parameters of the Caffe framework.

V. RESULTS AND DISCUSSION

Across all our experiments, we used the augmented dataset and three different versions of the whole PlantVillage dataset to train the network. Considering that different parameter settings will affect the performance of the model, this paper also compared the impact of the global pooling type, batch size and other parameters on the accuracy of disease identification, and then optimised the model. Finally, we got an optimal model which was expected to achieve automatic, fast and accurate identification of crop diseases.

As shown in Table 4, the inception module and dilated convolution improve the performance remarkably. Compared with the baseline AlexNet, employing inception module yields a result of 97.93% in AA, which brings 1.52% improvement. It indicates that inception module can extract the disease characteristics at different scales on the surface of diseased leaves. Due to the different sizes of the lesions on the surface of the diseased leaves, the inception module brings benefit to disease identification. Meanwhile, employing dilated convolution individually outperforms the baseline by 1.16%. This shows that the increase of receptive field contributes to disease identification. When we integrate the inception module and dilated convolution together, the performance further improves to 98.44%. Results show that the proposed architecture, combined with inception module and dilated convolution brings benefit to crop disease identification. Based on the above experiments, the subsequent experiments are conducted to obtain a suitable recognition model.

Effects of global pooling type on model performance

Due to different settings, six improved models were obtained in our experiments. Classification accuracies of diverse models are summarised in Table 5. As described in Section 3.1, global pooling includes global average pooling and global maximum pooling. In our experiments, models 1, 3, 5 use global maximum pooling, and models 2, 4, 6 use global average pooling. It can be observed from Table 5 that using global average pooling gets better results than using global maximum pooling. For example, the accuracy of Model 2 is 0.5% higher than that of Model 1, training on the original dataset. We suppose that global average pooling averages the pixel values of the entire feature map. It could make full use of the information and effectively reduce the estimation error caused by the domain constraints. In contrast, the global maximum pooling can only retain the low-level texture feature. Therefore, the deep feature information is lost, which results in lower recognition accuracy. However, we are not sure that the global average pooling is always the best option, since different classification tasks may have various suitable configurations.

Model number	Model	Model parameter		Accuracy, %	
		Global pooling type	Batch size	Original image dataset	Augmented image dataset
0	AlexNet model	_	64	96.41	97.62
1	improved model	maximum pooling	64	97.13	98.56
2	improved model + segmente d dataset	average pooling	64	97.63	98.93
3	improved model + grey- scaled dataset	maximum pooling	128	98.26	99.25

Table 5. Classification accuracies with different parameters during diverse experiments

Effects of batch size on model performance

It can be seen from Table 5 that the global average pooling contributes to the performance of models. In order to further optimize the model, the batch size is taken into consideration during the experiments. The batch training method is used to train the network model in experiments, and the training set is divided into several batches. In Table 5, the best accuracy 99.37% is got by the improved model combining with batch size 128 and global average pooling (Model 4). Compared with other models, employing batch size 128 yields better results in AA. Thus, it is supposed that the batch size 128 is quite effective in this work.

In the case of deep learning, if the dataset is enough, the gradient calculated from half of the batch data (even more less) is almost the same as that from all the dataset. Thus, increasing the batch size within a reasonable range might effectively improve the computational efficiency of matrix multiplication and reduce the number of iterations. However, compared with Model 3, employing batch size 256 (Model 5) yields a result of 96.25% in AA, which causes 2.01% loss. Thus, improving batch size blindly may be getting the opposite results. The memory utilization is increased, yet the memory capacity may not meet the calculation requirements. In addition, as the batch size increases, the number of iterations required to train a full dataset is reduced. To achieve the same accuracy, the time it takes is greatly increased, and the correction of the parameters becomes slow. As a result, the accuracy stops rising if the batch size is increased to a certain extent.





a

viewer PowerPoint Convolution kernel visualizations of the higher layers

Effects of background and colour on model performance

Identification model	Accuracy, %	Loss value
improved model + raw dataset	98.45	0.056
improved model + augmented dataset	99.37	0.023
improved model + segmented dataset	98.83	0.042
improved model + grey-scaled dataset	96.73	0.111

The later experiment is compared to the experiments with the raw dataset and augmented dataset. The recognition model trained on augmented dataset performs the best, with the highest recognition accuracy of 99.37% and the lowest loss of 0.023. As

expected, the performance of the recognition model obtained from training grey-scaled dataset is significantly lower than that of the experiments on the other version of the datasets, but still achieves the accuracy of 96.73%. It indicates that the improved CNN-based model has learnt to extract colour features and the lack of colour features (especially green) in the image would affect the final recognition accuracy. Compared with the model trained on the raw dataset, training on the segmented dataset yields a result of 98.83% in accuracy, which brings 0.38% improvement. We suppose that the model which trained on the segmented dataset can investigate the role of the background of the images in overall performance. Thus, the redundant information and noise introduced during model training might be reduced to increase the final accuracy.

To further verify the convergence effect of the four recognition models, the relationship between accuracy and epochs is illustrated in Fig. 7. It is obvious that after only ten training epochs, all models start to converge. Meanwhile, after 25 training epochs, all models have converged and the test accuracy of the four models can reach over 95%. Besides, it can be found from Fig. 7 that the recognition model based on the segmented dataset and the raw dataset has a convergence speed close to each other, and the final recognition accuracy has little difference. On the one hand, it can be concluded that a small amount of redundant information introduced from the simple background has little effect on the recognition effect of the model. On the other hand, if the background is complex, a better classification model is likely to be obtained by training the images which segment the background.



Relationship between test accuracy and epoch

Results compared with other studies

Table 7 shows the best results obtained in this study, compared with other studies that previously classified diseases using computer vision and deep learning. The last column shows the classification accuracies when the methods are trained and tested on our dataset with 38 classes. The classification accuracy of this study is higher than that of other studies. Overall, deep learning methods perform better than traditional computer vision methods.

Table 7. Comparison of classification accuracies from previous crop diseases identification studies

Studies	Number of diseases/ crops	Method/model	Original classification accuracy, %	Classification accuracy on our 38 classes, %
Ma et al. [[12]]	1	shape and colour features, SVM	90	—
Girish and Bardar [[35]]	3	colour and area features, BPNN	92	_
Lu et al. [[25]]	10	typical CNN model	95.48	96.86
Jayme [[36]]	10	GoogleNet	96.00	98.24
current study	38	improved CNN-based model	99.37	99.37

VI. CONCLUSION

In this paper, CNN is applied to the identification of many crop diseases. Traditional AlexNet model is improved by combining the inception module with dilated convolution. Then the proposed model is trained to identify many crop diseases and is able to effectively classify 26 crop diseases and 12 healthy crops through images recognition. The improved model could extract disease features of different scales and increase the receptive field of the network compared with the original AlexNet model, and therefore the final average recognition accuracy reached 99.37%. In addition, the higher convergence rate of the model is obtained.

After only two epochs of training, the accuracy of the improved model is over 95% and the loss value is only 0.023. The contrast experiments based on the segmented dataset and grey-scaled dataset also verify that noise in background and colour features maybe have an impact on the recognition effect, which provides valuable support for further improvement of the recognition method.

The results show that the improved CNN-based model trained on augmented dataset has good adaptability and robustness to image spatial position changes. Our optimal model can identify different diseases of multiple crops. We hope and believe that the research and observations of this paper will provide help for disease identification of crops in actual production. In future work, we are supposed to apply and improve other deep CNN-based architectures and train the model with images of more complex background.

7 Acknowledgments

This work was partially supported by the National Natural Science Funds projects (grant no. 31471413), a project funded by the Priority Academic Program Development of Jiangsu Higher Education Institutions (PAPD), Six Talent Peaks Project in Jiangsu Province (ZBZZ-019), Practical Innovation Training Program for College of Jiangsu University.

VII. REFERENCES

- 1. Zhang, Y., Gao, Y., Wang, X.: 'Problems and countermeasures of crop pest control', Agric. Technol., 2017, 37, (1), pp. 117–118 Google Scholar
- 2. Liu, T., Zhong, X., Sun, C. et al.: 'Computer vision-based identification of rice leaf diseases', Agric. Sci. China, 2014, 47, (4), pp. 664–674Google Scholar
- 3. Tan, W., Zhao, C., Wu, H. et al.: 'Hierarchical pathology image recognition based on elastic momentum deep learning neural network', Trans. Chi. Soc. Agric. Mach., 2015, 46, (1), pp. 20–25 Google Scholar
- 4. Sun, J., Tan, W., Mao, H. et al.: 'Recognition of multiple plant leaf diseases based on improved convolutional neural network', Trans. Chin. Soc. Agric. Eng., 2017, 33, (19), pp. 209–215Google Scholar
- 5. Srdjan, S., Marko, A., Andras, A. et al.: 'Deep neural networks based recognition of plant diseases by leaf image classification', Comput. Intel. Neurosci., 2016, pp. 1–11
- 6. Web of Science®Google Scholar
- Mohanty, S.P., Hughes, D.P., Salathé, M.: 'Using deep learning for image-based plant disease detection', Front. Plant. Sci., 2016, 7, p.1419 View Web of Science®Google Scholar
- 8. [Johannes, A., Picon, A., Alvarez-Gila, A. et al.: 'Automatic plant disease diagnosis using mobile capture devices, applied on a wheat use case', Comput. Electron. Agric., 2017, 138, pp. 200–209.
- 9. Liu, P., Tu, K., Xu, H. et al.: 'Persimmon surface disease recognition based on support vector machine', Modern Food Sci. Technol., 2011, 27, (3), pp. 349–353 Google Scholar
- 10. Bankar, S., Dube, A., Kadam, P. et al.: 'Plant disease detection techniques using canny edge detection & color histogram in image processing', Int. J. Comput. Sci. Inf. Technol., 2014, 5, (2), pp. 1165–1168
- 11. Ma, Y., Wen, Y., Li, X. et al.: 'Diagnostic system of greenhouse cucumber downy mildew based on image processing', Trans. Chin. Soc. Agric. Mach., 2017, 48, (2), pp. 195–202
- 12. Fukushima, K.: 'Neocognitron: a self-organizing neural network model for a mechanism of pattern recognition unaffected by shift in position', Biol. Cybern., 1980, 36, (4), pp. 193–202
- 13. Lecun, Y., Boser, B., Denker, J.S. et al.: 'Backpropagation applied to handwritten zip code recognition', Neural. Comput., 1989, 1, (4), pp. 541–551
- 14. Gong, D., Cao, C.: 'Plant leaf classification based on convolutional neural network', Comput. Modernization, 2014, 2014, (4), pp. 12–15
- 15. Zhang, S., Huai, Y.: 'Research on plant leaf recognition based on layered convolution depth learning system', J. Beijing Forestry Univ., 2016, 38, (9), pp. 108–115

- Zeiler, M.D., Fergus, R.: 'Visualizing and understanding convolutional networks', Dept. Comput. Sci., New York Univ., 2013, 8689, pp. 818–833
- 17. Goodfellow, I.J., Bulatov, Y., Ibarz, J. et al.: 'multi-digit number recognition from street view imagery using deep convolutional neural networks' arXiv: 1312. 6082, 2013
- 18. Parkhi, O.M., Vedaldi, A., Zisserman, A.: 'Deep face recognition', British Mach. Vision Conf., 2015, 41, pp. 1–12
- 19. Sun, Y., Liang, D., Wang, X. et al.: 'DeepID3: face recognition with very deep neural networks', Comput. Sci., arXiv: 1502.00873, 2015
- Dai, J., Li, Y., He, K. et al.: 'R-FCN: object detection via region-based fully convolutional networks'. 30th Conf. on Neural Information Processing Systems (NIPS), Barcelona, Spain, 2016
- 21. Chen, L.C., Papandreou, G., Kokkinos, I. et al.: 'Semantic image segmentation with deep convolutional nets and fully connected CRFs', Comput. Sci., 2014, 4, (4), pp. 357–361
- 22. Chen, L.C., Papandreou, G., Kokkinos, I. et al.: 'Deeplab: semantic image segmentation with deep convolutional nets, atrous convolution, and fully connected CRFs', IEEE Trans. Pattern Anal. Mach. Intell., 2018, 40, (4), pp. 834–848
- 23. Dyrmann, M., Karstoft, H., Midtiby, H.S.: 'Plant species classification using deep convolutional neural network', Biosyst. Eng., 2016, 151, pp. 72-80
- 24. Lu, Y., Yi, S., Zeng, N. et al.: 'Identification of rice diseases using deep convolutional neural networks', Neurocomputing, 2017, 267, pp. 378–384
- Cheng, X., Zhang, Y., Chen, Y. et al.: 'Pest identification via deep residual learning in complex background', Comput. Electron. Agric., 2017, 141, pp. 351–356
- 26. Too, E.C., Yujian, L., Njuki, S. et al.: 'A comparative study of fine-tuning deep learning models for plant disease identification', Comput. Electron. Agric., 2019, 161, pp. 272–279
- Guan, T., Zhu, H.: 'Atrous faster R-CNN for small scale object detection'. Int. Conf. on Multimedia and Image Processing, 2017, pp. 16–21
- Krizhevsky, A., Sutskever, I., Hinton, G.E.: 'Imagenet classification with deep convolutional neural networks'. Int. Conf. on Neural Information Processing Systems, Curran Associates Inc., Lake Tahoe, Nevada, USA., December 2012, pp. 1097–1105
- 29. Wan, L., Zeiler, M.D., Zhang, S. et al.: 'Regularization of neural networks using DropConnect'. Int. Conf. on Machine
- Jayme, G.A.B.: 'Plant disease identification from individual lesions and spots using deep learning', Biosystems Eng., 2019, 180, pp. 97–107

INTERACTIVE CONTEXT FOR A REVIEW ON MEDICAL STORE WEBSITE

D.P. Bhojane¹, V.S. Boragi², P.A.Pawar³, G.S.Kusumade⁴

TY Student, Information Technology Department SVERI's College of Engineering (Poly), Pandharpur¹²³⁴ dhanashribhojane5@gmail.com

Abstract— For medical shops, a website called Medical Shop Management System was created. This system deals with the buying, selling, and storage of pharmaceuticals as well as the creation of deal checks, expiration dates, and monuments related to medications. When every step is done by hand, more time and effort are needed. The real-time design of the Medicine Store Management System will be helpful to those who lack the time to visit an offline retailer. The customer must first provide the required information for enrollment in order to place an online order for the desired medication. The client's name, correspondence ID, phone number, username, word, and confirm word are required fields for enrollment. The username and password from the enrolling process are then required to access your account. additionally, the requested medication is up to the client to choose. We can also cancel an order once it has been validated. This specific project addresses the challenges associated with running a medical supply store while avoiding issues that arise from manual labor. The Medical Management System is a Windows-based software intended for patient record registration, management, and convenient access. In a hospital or clinic, the system will let the register, physicians, and lab technicians maintain and manage patient records for simpler access and reference. All of these tasks are routinely completed and would be laborious for the staff if completed by hand. For this reason, clinic and hospital staff requires effective management software that is simple to use and won't add to their workload.

Keywords: - Medical, customer, medicines, stores, delivery, management.

I. INTRODUCTION

To manage their data, a medical supply store has hired us to create a database. We will oversee their store, visitors, etc.As we can see, there are many medications available nowadays, and medical supply stores can be found all over the place. It is quite important to save their records. In order for them to understand the significance and nature of the pharmaceuticals they possess, we are providing them with a database. The slip's record can also be accessed via this database. We're providing a database, as we've mentioned in the description, so they can save the record in every manner. The medical store owners will be able to save the medication results after we provide them access to the database. possess in their shops. They can also produce the invoices.

Medical stores can manage their paperwork, payment information, and force online with simplicity thanks to the medical store operating system, which also helps them reclaim them. We additionally provide countless clients with online medicine delivery. It benefits clients who sell medications such as blood pressure, diabetes, and thyroid. For elderly people, home delivery is quite helpful.

The Medical Store Management System's primary goal is to replace the existing manual system with sophisticated software and automated instruments. The goal of this automated system is to effectively store and manage medical stores' important data and information in order to satisfy their requirements.

Through the use of widely accessible and intuitive hardware and software, the system may provide error-free, secure, dependable, and quick administration capabilities.

With this system in place, users won't have to worry about tedious record-keeping duties and can concentrate on their main duties. As a result, the company may preserve digitized records without needless duplication and maximize its resource use. It is simple to obtain important information without being sidetracked by other facts. The ultimate objective is to automate the process in order to improve performance and give clients better services.

Medical supply stores are currently searching for services that can offer staff and customers accurate and dependable care. Every retailer is working to computerize their processes in order to give their clients superior services. The system utilized for medicine stock inventory in medical stores is called the medical store management system. The store manager can keep track of and oversee every activity in the medical shop thanks to this system.

This system also makes it possible for the medical store's staff to provide their services in a way that is more methodical and efficient, which enhances the medical shop. This aids in the analysis of the store's success as well. The daily operations of the medical store, including tablet information, billing, stock details, and more, may be organized with the help of the medical store management software. The medical shop management software makes it possible to keep track of the information about the stock that the store purchases and the information about the stock that is sold to customers (Hull, 2012).

This program creates automated bills for each transaction and reports on sales, inventory, and customer information. The medical store's inventory management takes less time and effort when this approach is used. It also lessens the labor-intensive process of maintaining records on paper. The managers may quickly record their suppliers' details and evaluate them as needed by using this system.

Because the medical store management system gives information about medications and where they are located throughout the store, it increases employee productivity. Medical stores can handle their paperwork, payment information, and inventory online more easily with the help of the medical store management system.

In times of peace, war, and operations other than war (OOTW), medical services play a critical role in the logistic support system of every nation's armed forces.

It will be understood that competent management of medical stores—which includes, but is not limited to, the arsenal of pharmaceuticals and expendable and non-expandable goods of medical use—is vital to the provision of complete healthcare. This makes it an essential part of any organized medical care, whether it's provided by the military or the general public. The seamless and appropriate operation of the medical supplies department determines the efficacy and efficiency of healthcare delivery in a hospital context.

Inventory and warehouse management are the two main facets of medical store management. Healthcare managers must keep a close eye on employee performance, and practical considerations demand that hospital hierarchy and system communication be optimized as a core component of logistics management programs.

II. LITERATURE REVIEW

1. ERP for Medicine

Medical pharmacy operation software will assist pharmacists in efficiently managing their pharmacies as well as purchases, returns, trades, short items, clients, and overall business. The fully automated medicine drugstore operating software will lessen the workload for pharmacists and boost revenue.

2.Meddoz

Meddoz makes medical billing, reporting, and handling simple. Utilize Meddoz's suite of efficient operation and reporting modules to maintain control over your pharmacy's daily operations. Utilize a single touchpoint to handle accounts, agreements, charges, and complaints.

3. Medeil

Meddoz is Meddoz makes medical billing, reporting, and handling simple. Utilize Meddoz's suite of efficient operation and reporting modules to maintain control over your pharmacy's daily operations. Utilize a single touchpoint to handle accounts, agreements, charges, and complaints (3).

4. Hari Om Medical and General Store in Vairag, Solapur

One of the top pharmacies in Solapur is Hari Om Medical and General Store. Veterinary druggists, druggists, Ayurvedic Medical Shops, Chemist Home Delivery, Ayurvedic Product Dealers, and Ayurvedic Medicine Retailers are among the other well-known industries. For greater import and commerce. Chemists are involved in our life when we are ill. Offering almost every form of medication, they have been many people's lifeline. Apart from medications, they also sell skin care and cosmetic items. There are

druggists, also known as druggists, positioned all around the megacity. Several of them regularly collaborate with large corporations and medical facilities. On the other hand, if you would like to speak with Hariom Medical and General Stores in Vairag, Solapur need a pill for your first aid kit or wish to purchase the medication that your croaker has recommended. They are highly regarded in Solapur offers a range of services, such as home delivery and products available (4).

5. Vairag, Solapur's 5 Kailash Medical and General store

One of the top pharmacies in Solapur is Hari Om Medical and General Store. Additionally Recognized for Chemists, General Store, Elder, Vicks, Mentho Plus, Balm Dealers, Balm Dealers -Zandu, and a host of other items. Find Kailash Medical and General Store's address, phone number, reviews, ratings, photos, and map in Solapur.

When it comes to our health, chemists are quite important. Their provision of nearly all types of medications has made them indispensable to numerous individuals. They sell skin care and wellness goods in addition to medications. Pharmacists, another name for chemists, can be found all around the city. Some of them frequently have connections to prominent doctors.

6. Sant Damaji Medical Store, Solapur, Mangalwedha

Chemists are important people in our lives when we are sick. Their provision of nearly all types of medications has made them indispensable to countless people. They sell skin care and wellness goods in addition to medications. Pharmacists, another name for chemists, can be found all around the city. Many of them have connections to major medical facilities and physicians. Sant Damaji Medical Store is located in Mangalvedha Solapur, Solapur, and is the place to go if you need medication for your first aid kit or to purchase medication that has been prescribed by your doctor. They are well-known in Solapur for offering a range of services, including home delivery, same-day delivery, and in-store pickup.

7. SAI Medical Solapur, Mangalwedha, Solapur

Chemists are important people in our lives when we are sick. Offering practically every form of medication, they have come to be seen as many people's lifeline. They sell skin care and wellness goods in addition to medications. Pharmacists, another name for chemists, can be found all around the city. Many of them have connections to major medical facilities and physicians. In case you require medication for your first aid kit or wish to get medication as directed by your physician, reach out to SAI Medical Store located in Mangalwedha Solapur, Solapur.

III. PROPOSED SYSTEM

The development of an upgraded facilities system is the goal of the suggested system. All of the limitations of the current system can be overcome by the suggested system. The system minimizes manual labor and offers appropriate security.

1.data security.

- 2. Assure accuracy of the data.
- 3. Increased effectiveness.
- 4. Improved assistance.
- 5. Interactive and user-friendly.
- 6. The minimal amount of time needed.
- 7. The current system is being created with the intention of circumventing its limitations.
- 8. We included the medical store management system in the suggested system.
- 9. Verifying the existence of products and verifying billing is simple.
- 10. The administrator changes the medication list on a daily basis, and the customer's medical data is safe and secure.
- 11. It's for customer safety and to keep the record up to date.
- 12. It will be kept in the report module for viewing after billing.
- 13. A medication will be delivered to the delivery area in a minute.



Fig 1. Proposed System Architecture [9]

IV. CONCLUSION

The driver now has a simple method to alter data in the database and communicate with it thanks to this approach. The driver can easily add, remove, and update records in the database. Working at a medical store is now a crucial and simple profession thanks to this program. This program keeps track of each client bill. Companies claim that it has purchased stock. A bill is automatically generated by this software. It provides extensive information regarding the stock sale. To put it succinctly, medicine has simplified people's lives. The name of this software is Medical Store Operation System. It offers all information, from little to major, including customer, purchase, vend, bill, and other details. The internet-based pharmacy is an online platform. that provide a means for customers to purchase prescription drugs online, enabling them to temporarily accept drug services in the convenience of their own homes.

The system was created and fasted gradually. Following user department clearance, the output reports' formats were determined one by one once the design was complete. The system was tested using test data initially, and later actual data. Small code mistakes were found and fixed. Following corrections, the system was effectively applied, and accuracy was discovered.Following testing, we have determined that our "Learning Management System" is very helpful to users and ideal for online mailing. We conclude that this is a real client/server environment that is accessible from anywhere in the world when it has been implemented. When compared to other automated systems, this one is quick.

With the help of this technology, an operator can easily interact with the database and modify its contents. The database's records are easily added, deleted, and updated by the operator. The work in a medical store is now lot easier thanks to this program. Every customer bill is logged by this software. Companies claim that it has purchased stock. A bill is automatically generated by this software. It provides extensive information regarding the stock sale. It is, in essence, a medical inventory that has simplified people's life. The name of this software is Medical Store Management System. It offers all information, from little to major, including details about customers, purchases, sales, bills, and more. The online system is a web-based medical store that offer a platform for online drug purchases, enabling customers to quickly get services or medications in the convenience of their own homes.

V. REFERENCES

- 1. Seidler, K. (2011, January 27). Apache friends Xampp. Apache friends very easy apache, mysql, php and perl installation without hassles. Retrieved January 2, 2013, from <u>http://www.apachefriends.org/en/xampp.htm</u>
- 2. HTML editor software, web design software | Adobe Dreamweaver CS6.(n.d.).Adobe Retrieved January 2, 2013, from
- 3. http://www.adobe.com/products/dreamweaver.htm.
- 4. Top 5 Best Medical Store Management Software in 2021 by SoftwareWorld : https://www.softwareworld.co/bestmedical-store-management software/
- 5. Pharmacy Management System" by Capterra: https://www.capterra.com/pharmacy-software/
- 6. Best Medical Store Management Software 2021 Reviews & Pricing" by Software Advice: https://www.softwareadvice.com/medical-store management/
- 7. Pharmacy Software | Pharmacy Management System | PrimeRx" by Micro Merchant Systems: https://www.micromerchant systems.com/pharmacy software.
- 8. Pimpalkhute S, Shrivastava M, Mahatme M, Uchit G, Solanke B, Hiware S. ABC and VED Analysis in medical store of Indira Gandhi Govt.Medical College and Hospital, Nagpur. *Indian Med Gaz.* 2009:279–84. [Google Scholar]
- 9. Manoharan TN. 5th ed. Chennai: Sujatha Publication Private Ltd; 1999. Direct tax Laws with tax planning aspects 2000-2001. [Google Scholar]
- 10. [Last cited on 2011 Jul 20]. Available from: <u>https://www.researchgate.net/figure/State-chart-diagram-of-web-centric-intelligent-healthcare-diagnosis-system_fig5_238158013</u>
- 11. Eisenberg JM. Clinical economics. A guide to the economic analysis of clinical practices. *JAMA*. 1989;262:287986. [PubMed] [Google Scholar]
- 12. Thawani VR, Turankar AV, Sontakke SD, Pimpalkhute SV, Dakhale GN, Jaiswal KS, et al. Economic analysis of drug expenditure in Govt.Medical College hospital, Nagpur. *Indian J Pharmacol.* 2004;36:15–9. [Google Scholar]

A DECISION SUPPORT SYSTEM USING SENSOR-CLOUD BASED SMART AGRICULTURAL SYSTEM FOR SMART IRRIGATION SCHEDULING

Malage Varsharani N., Malage Rajshri N. Pawar Minakshi M., More Archana K., Jagtap Mohini B,

SVERI's college of Engineering, Pandharpur, Maharashtra, India

AISSMS college of Engineering, Pune, Maharashtra, India

Abstract: This abstract introduces a decision support system that leverages sensor-cloud technology within a smart agricultural framework for intelligent irrigation scheduling. The proposed system integrates sensor data to make informed decisions, enhancing the efficiency of agricultural practices. Key components include real-time monitoring of environmental conditions, utilization of cloud-based computing for data processing, and smart irrigation scheduling for optimized water resource management. By employing this sensor-cloud-based approach, the decision support system aims to enhance precision and sustainability in agricultural irrigation, contributing to resource conservation and improved crop yields. This innovative system represents a significant advancement in smart agriculture, offering a practical solution for farmers seeking to optimize irrigation practices in response to dynamically changing environmental conditions.

Keywords: Sensor, Agricultural, cloud.

I. INTRODUCTION

In many regions, natural rainfall is either insufficient or unevenly distributed throughout the year. Irrigation becomes essential to supplement the water requirements of crops and ensure consistent moisture levels for optimal growth. Crops often have specific water requirements during different stages of their growth cycle. Irrigation allows farmers to provide water when needed, regardless of natural rainfall patterns or seasonal variations. Irrigation significantly enhances agricultural productivity by offering a controlled and reliable water supply.

This results in higher crop yields and allows for multiple cropping seasons, contributing to increased overall food production. Irrigation enables farmers to grow a wider variety of crops and plant them throughout the year. This diversification not only supports food security but also allows farmers to adapt to market demands and changing consumer preferences. Improving water use efficiency in irrigated agriculture necessitates the effective use of available water resources to address real-time crop irrigation issues The necessity to enhance water use efficiency has never been overemphasized throughout the research community and industry as it is presently. This requires monitoring water use and managing how irrigation water is applied. The aim of monitoring the crop environment in real-time is to control the root zone soil moisture.

In arid and semi-arid regions, irrigation in agriculture accounts for a significant portion of freshwater consumption. Agricultural sectors in these regions face challenges due to increased water demands combined with growing populations, insufficient water resources, and climate change-induced rainfall variability. To decrease water usage in agriculture, real-time smart irrigation technologies are needed. Along with improving crop yield and water productivity, smart irrigation can also reduce farming costs and minimize environmental impact.

Different irrigation scheduling techniques, such as evapotranspiration, water balance, soil moisture condition, and plant water condition techniques, have been employed in smart irrigation management. A combination of these scheduling measures promotes efficient crop irrigation and reduces water wastage. In several studies, evapotranspiration, soil moisture sensors, or plant-based smart irrigation technologies have been used to schedule irrigation events by considering weather conditions, soil moisture levels, and plant water status. A soil moisture-based approach is user-friendly and can be automated using various commercial instruments.

II. DESIGN METHODOLOGY

OUTLINES OF PROPOSED WORK:

IoT architecture and design

The overall architecture of the IoT-based smart drip irrigation system, shown in the figure, illustrates the system input received through the installed IoT device on the farm (soil moisture, soil temperature, ambient humidity, and temperature). Using the Wi-Fi

Module, soil moisture content, soil temperature, relative humidity, and temperature data are stored on an IoT platform every two minutes. This study established a Wi-Fi wireless communication network among the sensors, transmitters, cloud servers, solenoid valves, and the receiver unit.

Irrigation management

A real-time soil moisture status is used to operate the actuating pump and valves. Solenoid valves open when the soil moisture sensor threshold value reaches or exceeds the soil moisture depletion of field capacity. When the soil moisture content reaches the field capacity of the soil moisture sensor, the solenoid valves are closed. A server is used to store continuous volumetric soil moisture data.



Fig. __ Overall architecture of IoT based smart drip irrigation system.

Suitable sensor positions:

Sensors were placed in sensitive and representative locations of the crop's root zone. Soil moisture or tension varies in three dimensions. Stieber and Shock Noticed that those dimensions included; variability in soil wetting from irrigation/rainfall, drying of soil from evaporation and root-zone water extraction for plant transpiration. Interactions of these dimensions were imperative for soil moisture sensor installation. According to Semimetal. soil moisture sensors should be placed at the top of the active root system, relative to drippers and the bottom of the root system; in every case studied, 11 cm between the drip line and 10 cm beneath the surface of the soil was deemed to be the most suitable position. In general, sensors placed deeper in the soil profile result in lower irrigation efficiencies, which leads to more frequent irrigation.



Fig. Algorithm

VI. PERFORMANCE

Soil moisture sensors were installed 10 and 15 cm in a clayey soil to help in irrigation control based on soil moisture content. This device uses Internet of Things technology to automate irrigation without the need of human interference. Soil moisture,

temperature, and humidity sensors continuously track soil moisture, temperature and humidity levels in different parts of the agriculture farm. The moisture sensor sends the signal to the Wi-Fi module, which triggers the water pump and irrigates the field using the smartphone or computer application if the moisture level falls below the pre-defined value. The measured variables were continuously recorded automatically and sent to the cloud-based server. The developed data acquisition unit was switched ON and staked in the agriculture field. This device is directly linked to the IoT analytics platform web service to access and analyse live data in the cloud. The demonstration of soil moisture, relative humidity, soil temperature and ambient temperature values at various times is shown in figure. The field channel valves of each individual are controlled. When the value is 1, the motor is ON, and when the value is 0 in the IoT analytics platform the motor is OFF.

VII. CONCLUSION

In conclusion, the development and implementation of a decision support system utilizing sensor-cloud technology for smart irrigation scheduling represent a significant stride towards sustainable and efficient agricultural practices. The integration of realtime sensor data, cloud computing, and intelligent irrigation scheduling offers a comprehensive solution to address the dynamic nature of environmental conditions in agriculture. This system empowers farmers with data-driven insights, enabling them to make informed decisions regarding irrigation, leading to optimized water usage, enhanced crop yields, and resource conservation. Furthermore, the incorporation of smart technologies in agriculture aligns with the broader goals of precision farming and environmental sustainability. The sensor-cloud-based approach provides a scalable and adaptable framework that can be customized to various crop types and geographical locations, fostering widespread applicability. By promoting precision irrigation, the system contributes to water conservation efforts, mitigates environmental impact, and supports the long-term resilience of agricultural ecosystems.

As agriculture continues to face challenges posed by climate change and resource constraints, the adoption of smart

technologies becomes imperative. The decision support system outlined in this study not only addresses current irrigation inefficiencies but also lays the groundwork for future advancements in precision agriculture. As technology continues to evolve, the integration of sensor-cloud systems in smart agriculture is poised to play a pivotal role in fostering sustainable practices, improving agricultural productivity, and ensuring food security for a growing global population.

VIII. REFERENCES

- S.H. van Delden, M. `SharathKumar, M. Butturini, L.J.A. Graamans, E. Heuvelink, M. Kacira, E. Kaiser, R.S. Klamer, L. Klerkx, G. Kootstra, A. Loeber, R.E. Schouten, C. Stanghellini, W. van Ieperen, J.C. Verdonk, S. Vialet-Chabrand, E.J. Woltering, R. van de Zedde, Y. Zhang, L.F.M. Marcelis, Current status and future challenges in implementing and upscaling vertical farming systems, Nat. Food 2 (12) (2021) 944–956, https://doi.org/10.1038/s43016-021-00402-w.
- O. Kulkarni, Crop disease detection using deep learning. 2018 Fourth International Conference On Computing Communication Control and Automation (ICCUBEA), 2018, pp. 1–4, https://doi.org/10.1109/IC CUBEA.2018.8697390.
- 3. C. Bi, J. Wang, Y. Duan, B. Fu, J. Kang, Y. Shi, MobileNet Based Apple Leaf Diseases Identification, 2022.
- 4. D. Xia, P. Chen, B. Wang, J. Zhang, C. Xie, Insect Detection and Classification Based on an Improved Convolutional Neural Network, 2022.
- 5. B.D. Richter, et al., Water scarcity and fish imperilment driven by beef production, Nat. Sustain. 3 (4) (2020) 319–328, https://doi.org/10.1038/s41893-020-0483-z.
- 6. ICID, Agricultural Water Management For Sustainable Rural Development: Annual report 2020-21, 2020.
- L.S. Pereira, P. Paredes, N. Jovanovic, Soil water balance models for determining crop water and irrigation requirements and irrigation scheduling focusing on the FAO56 method and the dual Kc approach, Agric. Water Manag. 241 (June) (2020), 106357, https://doi.org/10.1016/j.agwat.2020.106357.
- 8. Shahjalal M, Alam MZ, Miah SS, Chowdhury AH. Fuzzy logic approach for identifying the effects of climate change on agricultural production. Int J Agric Econ 2021;6(4):181.
- Hagras H, Callaghan V, Colley M, Carr-West M. A fuzzy-genetic based embedded-agent approach to learning and control in agricultural autonomous vehicles. In: Proceedings 1999 IEEE International Conference on Robotics and Automation (Cat. No. 99CH36288C), 2. IEEE; 1999. p. 1005–10.
- Prema K, Kumar NS, Dash SS, Chowdary S. Online control of remote operated agricultural robot using fuzzy controller and virtual instrumentation. In: IEEE-International Conference On Advances In Engineering, Science And Management (ICAESM-2012). IEEE; 2012. p. 196–201.

- L. Zotarelli, J.M. Scholberg, M.D. Dukes, R. Munoz-Carpena, [~] J. Icerman, Tomato yield, biomass accumulation, root distribution and irrigation water use efficiency on a sandy soil, as affected by nitrogen rate and irrigation scheduling, Agric. Water Manag. 96 (1) (2009) 23–34, <u>https://doi.org/10.1016/j.agwat.2008.06.007</u>.
- 12. Allen, R. Pereira, L.S. Raes, D. and Smith, M. Chapter 6-ETc-Single crop coefficient (KC). IN: Crop evapotranspiration—guidelines for computing crop water requirements—FAO Irrig. Drain. paper 56(1998). FAO-Food and Agriculture Organization of the United Nations. Rome, Italy. X0490E/X0490E00. htm.
- T. Selim, R. Berndtsson, M. Persson, M. Somaida, M El-Kiki, Y. Hamed, Q. Zhou, Influence of geometric design of alternate partial root-zone subsurface drip irrigation (APRSDI) with brackish water on soil moisture and salinity distribution, V. Kumar S et al. Smart Agricultural Technology 5 (2023) 100248 9 Agric. Water Manag. 103 (2012) 182– 190, https://doi.org/10.1016/j. agwat.2011.11.013.
- K.R. Gsangaya, S.S.H. Hajjaj, M.T.H. Sultan, L.S. Hua, Portable, wireless, and effective internet of things-based sensors for precision agriculture, Int. J. Environ. Sci. Technol. 17 (9) (2020) 3901–3916, https://doi.org/10.1007/s13762-020-02737-6.
- F.M. Kiziloglu, U. Sahin, Y. Kuslu, T. Tunc, Determining water-yield relationship, water use efficiency, crop and pan coefficients for silage maize in a semiarid region, Irrig. Sci. 27 (2) (2009) 129–137, https://doi.org/10.1007/s00271-008-0127-y

AN OVERVIEW OF BLOCK CHAIN TECHNOLOGY

Sanjivani G Atakare¹, Antosh M Dyade², Swati P Pawar³

¹M. Tech Student, Department of Computer Science and Engineering, SVERI's College of Engineering, Pandharpur *atakaresanjivani@gmail.com*²³Assistant Professor, Department of Computer Science and Engineering, SVERI's College of Engineering, Pandharpur

Abstract: The Blockchain is the newest and perspective technology in ultramodern frugality. A blockchain is basically a distributed database of records or public tally of all deals or digital events that have been executed and participated among sharing parties. It provides provenance, invariability and futurity for the transfer of value within a business network. - It enables value exchange in real time, reducing costs and crimes. Grounded on a network agreement approach, whereby trust between the parties involved in a sale is handed by cryptography. This paper aims to show brief overview of Blockchain Technology, operations and have bandied limitations in the perspective of unborn exploration on security in revolutionary sale.

Keywords: Digital transaction, encryption, cryptocurrency, cryptography, network security.

I. INTRODUCTION

This A blockchain is just one type of distributed tally, not all distributed checks inescapably employ blocks or chain deals. The blockchain is an imperishable digital tally for keeping track of profitable deals which can be programmed to maintain not only fiscal deals but nearly everything that has value." Don & Alex Tapscott, authors Blockchain Revolution 2016. A blockchain (firstly two- words block chain) is a continuously growing list digital records in packages (called blocks) which are linked and secured using cryptography. These digitally recorded "blocks" of data are stored in a direct chain. Each block in the chain contains data (e.g. bitcoin sale), is cryptographically hashed, and time stamped. The blocks of hashed data draw upon the former- block (which came before it) in the chain, icing all data in the overall "blockchain" has not been tampered with and has not been altered.

A blockchain is a chain of chronological blocks. A block is an added up set of data that's collected and reused to fit inside it through the process of mining. Each block is linked via a cryptographic hash and timestamp. When a new block is formed, it'll contain a hash of the former block, so that blocks can form a chronologically ordered chain from the first block ever generated in the entire blockchain (also called the Genesis Block) to the recently formed block. This process is repeated over-and-over again to grow and maintain the network. Now this means that this decentralized tally isn't controlled by any fiscal institution or government for that matter.

In fact, it can be penetrated by everyone who has a good internet connection. Other than virtual currencies, there are numerous companies similar as messaging apps, critical structure security, lift sharing, pall storehouse, etc. are employing the power of blockchain technology.

II. APPLICATION OF BLOCKCHAIN

A blockchain is just one type of distributed tally, not all distributed checks inescapably employ blocks or chain deals. The blockchain is an imperishable digital tally for keeping track of profitable deals which can be programmed to maintain not only fiscal deals but nearly everything that has value. Don & Alex Tapscott, authors Blockchain Revolution 2016. A blockchain (firstly two- words block chain) is a continuously growing list digital records in packages (called blocks) which are linked and secured using cryptography. These digitally recorded "blocks" of data are stored in a direct chain. Each block in the chain contains data (e.g. bitcoin sale), is cryptographically hashed, and time stamped. The blocks Blockchain technology offers the implicit to impact a wide range of diligence. The most promising operations live where transferring value or means between parties is presently clumsy, precious and requires one or further centralized association.

a) Financial Services

Several stock exchanges around the world are piloting a blockchain platform that enables the allocation and transfer of private securities. also, multiple groups of banks are considering use cases for trade finance, cross border payments, and other banking processes. Consumer and artificial.

Companies in the consumer and artificial diligence are exploring the use of blockchain to digitize and track the origins and history of deals in colourful goods.

c) Life lores and Health Cara

Healthcare associations are exploring the use of blockchain to secure the integrity of electronic medical records, medical billing, claims, and other records.

d) Public Sector

Governments are exploring blockchain to support asset registries similar as land and commercial shares.

e) Energy and resources

Ethereum is being used to establish smart- grid technology that would allow for fat energy to be used as tradable digital means among consumers. Since all businesses track information and face the challenge of coordinating data with counterparties, blockchain technology has the implicit to be applicable to everyone.

There are still numerous unknowns with respect to how blockchain will impact the inspection and assurance profession, including the speed with which it'll do so. Blockchain is formerly impacting CPA adjudicators of those associations using blockchain to record deals and the rate of relinquishment is anticipated to continue to increase. still, in the immediate future, blockchain technology won't replace fiscal reporting and fiscal statement auditing. Checked fiscal statements are a foundation of business and play a crucial part in debt and equity backing, participation in capital requests, combinations and accessions, nonsupervisory compliance, and the effective and effective functioning of capital requests. Fiscal statements reflect operation assertions, including estimates, numerous of which cannot be fluently epitomized or calculated in blockchains. Likewise, the process of an independent inspection of fiscal statements enhances the trust that's pivotal for the effective functioning of the capital requests system. Any corrosion of this trust may damage an reality's character, stock price and shareholder value, and can affect in forfeitures, penalties or loss of means. Druggies of fiscal statements anticipate CPA adjudicators to perform an independent inspection of the fiscal statements using their professional scepticism. CPA adjudicators conclude whether they've attained reasonable assurance that the fiscal statements of an reality, taken as a whole, are free from material misstatement, whether due to fraud or error. Blockchains are doubtful to replace these judgments by a fiscal statement adjudicator. Still, CPA adjudicators need to cover developments in blockchain technology because it'll impact their guests' information technology systems. CPA adjudicators will need to be knowledgeable with the basics of blockchain technology and work with experts to inspection the complex specialized pitfalls associated with blockchains (12). In addition, CPA adjudicators should be apprehensive of openings to influence their guests' relinquishment of blockchain technology to ameliorate data gathering during the inspection. They should also consider whether blockchain technology will allow them to produce automated inspection routines.

The auditing profession must embrace and "spare in" to the openings and challenges from wide blockchain relinquishment. CPA adjudicators are encouraged to cover developments in blockchain technology because they've an occasion to evolve, learn, and subsidize on their formerly proven capability to acclimatize to the requirements of a fleetly changing.

III. BASIC FEATURES OF BLOCKCHAINS

Understanding how a blockchain works from a specialized point of view, is precious only to the extent of developing or troubleshooting one. In order cohesively grasp the eventuality unto apply blockchain technology; you must also understand the characteristics of a blockchain. It's important to note that not all characteristics listed below will apply to all blockchains. The below donation provides the necessary background to identify the crucial characteristics and principles of blockchains. These are the following

a) Privacy

Blockchains store no particular information and use private/ public encryption to authenticate druggies bearing deals. Mining blockchains to gain particular information that could be vended to third parties for a profit isn't doable.

b) Translucency

All blockchain metadata and information is available to all bumps and druggies in real- time. It isn't possible to hide or redact blockchain information.51 Distributed translucency is therefore doable, but also introduces new issues.

c) Pseudo-anonymity

Bumps and druggies don't need to give names or particular details to be part of the network. still, full obscurity isn't achieved as linking druggies to network exertion is doable and can therefore lead to revealing their individualities.

d) Integrity

This works in two ways. First, data integrity it's nearly insolvable to change and falsify blockchain blocks. This is also called invariability. Second, stoner integrity metadata about the deals accepted by a knot and/ or end stoner are recorded on the blockchain and can be linked to the stoner bearing them. druggies can not wisecrack the network or try to complete an invalid sale.

e) Security

Use of blockchains requires cryptographic tools and public/ private keys by all actors, being bumps or end druggies.

f) Distributed trust governance

The blockchain successfully bypasses the need for a trusted central authority. rather, trust is spread across the network. The same goes for governance mechanisms where, in principle, different types of druggies and bumps have the same 'political' influence.

g) Sustainability

erected- in profitable impulses give a clear path for network profitable sustainability.

h) Open source

Software needed to use blockchains is freely available to all, including cryptographic tools. likewise, druggies with acceptable capacities can actually help enhance and upgrade blockchain technologies, in addition to catching bugs. This can also grease the spread of blockchain inventions.

IV. BLOCKCHAIN LIMITATIONS

The As an arising technology, blockchains face a series of limitations that might help wide relinquishment not only in the fiscal sector but also in other areas. These can be epitomized as follows:

a) Scalability

As it stands moment, Bitcoin blockchain can only add a new block of deals every ten twinkles or so. This translates into a low volume of deals per second (lower than five), a far cry from the volumes reported by traditional transactional networks. *b) Block size*

The below is the result of the small block size defined by the original Bitcoin source law. The maximum size for each block is one megabyte which can accommodate 2,200 deals. adding block size is presently under discussion but so far no final decision has been reached.

c) High costs

Miner bumps use sophisticated and precious tackle to run evidence of work algorithms. Accordingly, only certain bumps in the network can effectively contend in this process, indeed though in proposition all bumps have the software needed to booby-trap the network. Nakamoto's notion of "one- CPU one vote" is no more as tackle and electricity costs help most bumps from sharing in this process.

d) Cryptography

Use of cryptographic tools is still nascent and the average Internet stoner cannot be anticipated to embrace its use in the short term.

e) Complexity

Blockchain technologies appear to be nearly incomprehensible to the average person and the tech speak around it doesn't help. Only a named many feels to understand the technology.

f) Environmental impact

A by- product of the below is also evidence of work's inefficiency in terms of energy coffers. Some estimates on energy consumption suggest that, by Spring 2017, Bitcoin use of electricity was similar to that of 280,000 US homes per time. *g) Bandwidth*

Full bumps that want to be active in the network must have access to the right Internet bandwidth. Slow, unreliable connections aren't welcome, especially when the current size of the blockchain is over 120 Gigabytes.

h) Centralization

Mining is now consolidated with a many bumps controlling a large share of the request.55 Figure 6 below depicts request shares of the top miner bumps or companies. Note that the top five companies alone control over percent of the request. i)Usability

Blockchain technology requires the secure operation of public and private keys by end druggies and bumps. While being portmanteau software has come a long way, losing private keys is still a serious threat. None of the being results are resistant to physical theft and only a many can cover n druggies from malware.

j) *invariability as liability*

If the blockchain is addressed or the software law has a bug that allows a particular exploit, also its invariability can in fact come a liability. This was the case for illustration with the Ethereum hack of last time where one mischief knot was suitable to seize over 64 million bones. The block chain technology ecosystem is indeed visionary and formerly working to address some of these limitations. The fact that the law is open source is critical then. On the other hand, changes to both the law and blockchain operations can only be fulfilled by either agreement or if a maturity of bumps agree on a way forward.

V. CONCLUSION

The Blockchain technology is now one of the useful and protean concern for our world, Due to the large installations in utmost of the systems in the different diligence, but in malignancy of everything it's new and its major perpetration is little advised issue on practice. moment's Blockchain technology promises us the bright future for information technology withoutthe fraud and any deception. due to some benefits of the Blockchain technology. The challenges of the Blockchain are large, but the results of the Blockchain using have a lesser transcendence than disadvantages. It's necessary to keep exploring the Blockchain development and operation in the different areas for the nearest future, because this new technology can help to break numerous delicate problems, which are disturbing and precluding rightly systems work. We've bandied in this paper introductory features of blockchain technology and security issues of blockchains.

VI. REFERENCES

- 1. G. Eason, B. Noble, and I. N. Sneddon, "On certain integrals of Lipschitz-Hankel type involving products of Bessel functions," Phil. Trans. Roy. Soc. London, vol. A247, pp. 529–551, April 1955. (references)
- 2. J. Clerk Maxwell, A Treatise on Electricity and Magnetism, 3rd ed., vol. 2. Oxford: Clarendon, 1892, pp.68–73.
- 3. I. S. Jacobs and C. P. Bean, "Fine particles, thin films and exchange anisotropy," in Magnetism, vol. III, G. T. Rado and H. Suhl, Eds. New York: Academic, 1963, pp. 271–350.
- 4. Shuai Wang , Jing Wang, Xiao Wang , Member, IEEE, TianyuQiu, Yong Yuan , Senior Member, IEEE, LiweiOuyang, YuanyuanGuo, and Blockchain Powered Parallel Healthcare Systems Based on the ACP Approach2329-924Xc 2018 IEEE.
- 5. ZainabAlhadhrami, Salma Alghfeli, Mariam Alghfeli, Introducing Blockchainsfor Healthcare2017 International Conference on Electrical and Computing Technologies and Applications (ICECTA).
- K. Malasri, L. Wang. Design and Implementation of Secure Wireless Mote Based Medical Sensor Network. Sensors 9: 6273-6297, 2009A. Azeta, D. O. A. Iboroma, V. I. Azeta, E. O. Igbekele, D. O. Fatinikun, and E. Ekpunobi, "Implementing a medical record system with biometrics authentication in e-health," in 2017 IEEE AFRICON, Sept 2017, pp. 979–983.
- 7. P. Belsis and G. Pantziou. A k-anonymity privacypreserving approach in wireless medical monitoring environments. Journal Personal and Ubiquitous Computing, 18(1): 61-74, 2014.
- B. Jana and J. Poray, "A performance analysis on elliptic curve cryptography in network security," 2016 International Conference on Computer, Electrical & Communication Engineering (ICCECE), Kolkata, 2016, pp. 1-7.doi:10.1109/ICCECE.2016.8009587
- S. Mitra, B. Jana and J. Poray, "A novel scheme to detect and remove black hole attack in cognitive radio vehicular ad hoc networks(CR-VANETs)," 2016 International Conference on Computer, Electrical & Communication Engineering (ICCECE), Kolkata, 2016, pp. 1-5. doi: 10.1109/ICCECE.2016.8009589

- B. Jana, S. Mitra and J. Poray, "An analysis of security threats and countermeasures in VANET," 2016 International Conference on Computer, Electrical & Communication Engineering (ICCECE), Kolkata, 2016, pp. 1-6. doi: 10.1109/ICCECE.2016.8009588
- 11. S. Mitra, B. Jana, S. Bhattacharya, P. Pal and J. Poray, "Quantum cryptography: Overview, security issues and future challenges," 2017 4th International Conference on Opto-Electronics and Applied Optics (Optronix), Kolkata, 2017, pp. 1-7. doi:10.1109/OPTRONIX.2017.8350006
- 12. B. Jana, J. Poray, T. Mandal and M. Kule, "A multilevel encryption technique in cloud security," 2017 7th International Conference on Communication Systems and Network Technologies (CSNT), Nagpur, 2017, pp. 220- 224.doi: 10.1109/CSNT.2017.8418541
- Jana B., Poray J. (2018) A Hybrid Task Scheduling Approach Based on Genetic Algorithm and Particle Swarm Optimization Technique in Cloud Environment. In: Bhateja V., Coello Coello C., Satapathy S., Pattnaik P. (eds) Intelligent Engineering Informatics. Advances in Intelligent Systems and Computing, vol 695. Springer, Singapore
- Jana B., Chakraborty M., Mandal T. (2019) A Task Scheduling Technique Based on Particle Swarm Optimization Algorithm in Cloud Environment. In: Ray K., Sharma T., Rawat S., Saini R., Bandyopadhyay A. (eds) Soft Computing: Theories and Applications. Advances in Intelligent Systems and Computing, vol 742. Springer, Singap

ONLINE SOCIAL MEDIA CONTENT MANAGEMENT SYSTEM

Mihir Apte, Ajeet Lokhande, Rohan Mujumdar, Tejal Khadke

Department of Information Technology Engineering, Marathwada Mitramandal's College of Engineering, Pune, Maharashtra

Abstract: Revolutionizing online social media management, this innovative platform boasts three key functionalities that redefine the landscape. Firstly, the ability to schedule posts ensures a strategic and timely presence on various platforms. This feature empowers users to plan content in advance and automate posting, optimizing engagement. Secondly, the integration of AI-driven generative captions adds a layer of creativity and efficiency. By using ChatGPT API, users can effortlessly generate compelling and contextually relevant captions, enhancing the overall impact of their posts. Lastly, the inclusion of integrated analytics across multiple social media platforms provides users with a comprehensive overview of their online presence. This helps the users to refine the strategies based on performance metrics from different channels. Together, these functionalities not only streamline social media management but also elevate the quality and effectiveness of online content.

Keywords: Social media management, AI integration, comprehensive overview

I. INTRODUCTION

In the digital age, where social media serves as the virtual heartbeat of global communication, the role of an Online Social Media Manager has emerged as a critical component for individuals and businesses alike. This research paper delves into the multifaceted responsibilities of an Online Social Media Manager, focusing on three main functionalities: Caption Generation, Integrated Analysis, and Scheduling posts across different social media platforms, with an overarching goal of enhancing engagement and efficiency in the dynamic realm of social media. This research paper aims to provide a comprehensive understanding of the pivotal role played by an Online Social Media Manager. By delving into the intricacies of caption generation, integrated analysis, cloud import, and scheduling, we unravel the layers of complexity that define effective social media management in the contemporary digital landscape.

The following functionalities are further discussed in detail:

1. Schedule Post:

In the intricate world of social media marketing, maintaining an active presence is more strategic than it appears. Different social networks have their unique peak times, influenced by user behavior and algorithms. Engaging with a broad audience requires not just posting but posting smartly. Relying on real-time posts is not just challenging; it's inefficient. That's where the power of social media scheduling tools comes into play.

By leveraging these tools, you're not just posting; you're posting with purpose, ensuring your content aligns with your audience's peak activity. It's about working smarter, not harder. In this article, I'll be sharing insights on the top social media scheduling tools that have transformed the way I approach digital strategy.

Instagram: There are different ways to schedule posts on Instagram using Node.js. One way is to use the Instagramprivate-API module, which allows you to login and post to Instagram programmatically. Another way is to use the node-Instagram-schedule project, which is an Instagram post scheduler built with Node.js, MongoDB, and Puppeteer. You can also use the contentful platform, a headless CMS that can be integrated with Instagram API. Alternatively, you can use the Facebook Graph API to schedule posts to Instagram, as Instagram is owned by Facebook.

YouTube: YouTube is the world's largest video and music streaming platform. Being an open-source technology, Google provides many APIs to work with YouTube. From automating the searches to getting playlists, YouTube API can help developers in lots of things and it's useful for building services like music players, video players, marketing on YouTube, AdSense, etc.

Twitter: Twitter for Websites allows you to embed Twitter's live content into your product, direct from the source. Make the most out of Twitter on your website to increase followers, drive engagement, and grow your business.

Bring your pick of content from Twitter into your website or app. Using these tools embed Tweets in your stories and articles on the web.

Include a stream of Tweets in your website or app in a compact, linear view. Configure Tweet timelines to automatically display live updates from people, trends, and places right in your app.

Allow people who visit your website or app to engage with your Twitter account and share content with their followers. Enhance links to your website shared on Twitter with Cards and Twitter buttons.

If you are using React.js then react-tweet should make it easier to bootstrap Twitter based React.js apps. This way we can focus on interesting ways to use and manipulate the API without the pains of rendering. Styles, assets, and HTML have been lifted from twitter.com and twitter dev docs. react-tweet uses only inline styles and while written in ES6, compiles to plain JS meant to be absorbed by any React project. react-tweet can be used a 'dumb' component for simply rendering data or could be a starting point for a more ambitious Tweet component.

"React-twitter-widgets" and "react-twitter-embed" are some more useful libraries to check out.

1. Caption generation:

Integrating ChatGPT for automated caption generation in your application can enhance the capabilities of your platform, whether it's for social media posts, image descriptions, or any other use case that requires generating textual captions.

Integrating ChatGPT for automated caption generation in our Node.js application allowed us to effortlessly create descriptive and contextually relevant captions for various types of content, enhancing user engagement and content accessibility. This involved making API requests to OpenAI's GPT-3 API and customizing prompts to generate captions tailored to your specific needs. Careful attention to safety, compliance, and quality assurance is essential for a seamless user experience.

Automated caption generation is vital for enhancing content accessibility, user experience, and discoverability. It saves time, ensures consistency, and can scale with your content needs. It also aids multilingual support, compliance, SEO, and supports machine learning applications. Automated captions are cost-effective and crucial for real-time events, making them an essential tool for improving content engagement and reach.

Integrating ChatGPT is important for enhancing user interactions, instant response, automating customer support, and generating human-like text, which can lead to improved user satisfaction and operational efficiency in various applications and services.

ChatGPT, with its conversational abilities, is versatile and adaptable for tasks like caption generation. Automated caption generation addresses the limitations of manual caption writing by providing scalability, consistency, and cost-efficiency. It eliminates the time-consuming nature of manual captioning, ensuring that captions can be generated at scale and in real-time, which is especially crucial for live content. Automation also reduces inconsistencies and human errors, resulting in a more reliable and accessible user experience. Additionally, it's a cost-effective solution that can adapt to the growing volume of content in the digital landscape, ultimately improving content accessibility and discoverability.

Automated caption generation has some limitations. These systems might not always get captions exactly right and can sometimes produce inaccurate or inappropriate content. They may struggle with understanding context and nuances, leading to less precise captions. Additionally, automated solutions might not be as creative or adaptable as humans when it comes to describing certain content. Therefore, while they offer efficiency and scalability, human oversight and quality control are often necessary to ensure accurate and suitable captions.

ChatGPTUnofficialProxyAPI - Uses an unofficial proxy server to access ChatGPT's backend API in a way that circumvents Cloudflare (uses the real ChatGPT and is lightweight but relies on a third-party server and is rate-limited). ChatGPT – to use all the libraries.

2. Import Media:

Importing media into a social media content management system is fundamental to curating an engaging online presence. Whether you're a business looking to showcase products, an influencer sharing experiences, or a brand connecting with your audience, the ability to seamlessly upload and manage images, videos, and other media is essential. This process simplifies content creation, scheduling, and publishing, enabling you to deliver captivating and timely material to your followers. In this digital landscape, the efficient importation of media is key to maintaining a vibrant and dynamic social media presence. Local file import streamlines content creation, allowing users to effortlessly add their own media to posts and stories, enhancing engagement and visual storytelling.

Cloud import eliminates the need for local storage and simplifies the content creation workflow, enabling content creators to access and share their files from anywhere with an internet connection.

Importing media into a social media content management system is essential for effective communication, engagement, brand identity, and the efficient creation of diverse and engaging content on social media platforms. It significantly contributes to the success and impact of digital marketing and online communication strategies.

Local file import has several limitations. It relies on device storage space, which can lead to storage constraints and potential slowdowns. Compatibility issues may arise due to device-specific file formats, requiring conversions. Users also face data loss risks from device failures or theft, and version control challenges. Additionally, local storage can limit accessibility across different devices and demand a proactive backup strategy.

Limitations of Cloud Import:

Cloud import, while convenient, is not without limitations. It necessitates a reliable internet connection, making it less accessible in areas with limited connectivity. Privacy and security concerns emerge when using third-party cloud providers. Data transfer speed can be slower, and users may be subject to cloud storage provider terms and policies. Cloud import requires dependency on third-party providers and may not provide complete control over data.

II. PROBLEM DEFINITION

The absence of an integrated solution results in inefficiencies missed opportunities, and a higher risk of errors when executing cross-platform social media campaigns. There is also a lack of advanced analytics, content scheduling, and collaboration features. The challenge is to design and develop a unified platform that streamlines social media management, automates posting, offers advanced analytics, and ensures compliance with platform-specific regulations. The solution aims to enhance user efficiency and cross-platform consistency while prioritizing data privacy and security. The research objective is to create a user-friendly, customizable, and scalable integrated social media manager.

III. MOTIVATION

Our motivation for developing this model stems from the lack of existing solutions that effectively address the challenges of integrated social media management. With no comprehensive model available, we aim to fill this critical gap and provide users with a streamlined, efficient, and user-friendly platform for managing and posting content on Instagram, Twitter, and YouTube, ultimately revolutionizing the way individuals and organizations engage with their audiences across these social media platforms.

IV. PROPOSED MODEL

Our application helps in social media management and respective tasks which are:

- 1. Posting Content to multiple platforms from a single application.
- 2. Scheduling content.
- 3. Generating Captions with AI.
- 4. Importing media from the cloud or device.
- 5. Results. (posted content)

V. IMPLEMENTATION METHODOLOGY

A. Create Post:

Integrating social media APIs into your application involves several key steps to ensure successful communication and interaction between your application and the social media platforms.

Firstly, you need to register your application with the social media platform(s) of your choice. This process usually involves creating a developer account, setting up a new application, and obtaining API keys, access tokens, and client IDs. These credentials serve as the authentication mechanism allowing your application to access the platform's services.

Once you have the necessary credentials, you need to choose the appropriate API for your requirements. Most social media platforms offer a variety of APIs, each tailored to specific functionalities such as posting updates, retrieving user information, or accessing analytics data. Select the API(s) that align with your application's features.

Next, you'll need to implement the API calls in your application. This involves sending HTTP requests to the social media platform's API endpoints, using the authentication credentials you obtained earlier. These requests can be made using libraries like Axios in JavaScript or requests in Python, or specific SDKs provided by the platforms.

It's essential to handle authentication and authorization properly. Social media APIs often use OAuth or similar protocols for user authentication. You'll need to implement the OAuth flow in your application, which typically involves redirecting users to the social media platform's login page, obtaining an authorization code, and exchanging it for an access token that grants your application the necessary permissions.

Error handling and rate limiting are also crucial aspects. Social media APIs can return errors due to various reasons such as network issues, incorrect parameters, or rate limits being exceeded. Implement robust error handling mechanisms in your code to handle these situations gracefully. Additionally, respect the rate limits imposed by social media platforms to avoid being blocked for excessive API requests.

Lastly, keep your API integration secure and up to date. Regularly check for updates from the social media platforms, as APIs can change over time. Ensure that your application's codebase and dependencies are secure to prevent vulnerabilities that could be exploited.

By following these steps and adhering to best practices, you can effectively integrate social media APIs into your application, enabling seamless interaction with social platforms and enhancing user engagement.

B. Caption generation:

Integrating ChatGPT 3.5 into your application to generate captions based on keywords involves a straightforward process leveraging OpenAI's powerful language model. First, you need to obtain an API key from OpenAI, which allows you to make requests to the GPT-3.5 API. Once you have your API key, you can send a POST request to the OpenAI API endpoint, providing the appropriate parameters, including the model name (in this case, "text-davinci-003"), the prompt, and any other relevant instructions or keywords you want to base the captions on.

In your application, you can create an input interface where users provide keywords or topics for generating captions. When a user submits these keywords, your application sends a request to the GPT-3.5 API, passing the keywords as a prompt. The model will then generate captions or descriptive text based on the provided keywords. You can fine-tune the instructions to the model by specifying the format you want the captions in, the tone, or any other specific requirements.

Once you receive the response from the API, you can extract the generated captions and display them within your application's user interface. It's essential to handle API responses effectively, considering error handling and rate limiting to ensure smooth user experience. Additionally, you can experiment with different prompts and instructions to enhance the quality and relevance of the generated captions, tailoring them precisely to your application's requirements.

Remember to review and adhere to OpenAI's usage policies and guidelines to ensure compliance and ethical use of the ChatGPT 3.5 API in your application.

C. Import data:

Integrating cloud file import functionality into your application involves a series of steps to seamlessly connect your application with a cloud storage service such as Amazon S3, Google Cloud Storage, or Microsoft Azure Blob Storage. First, choose an appropriate cloud storage provider and create an account. Set up a specific bucket or container in the cloud storage platform where your application will store and retrieve files. Next, integrate the cloud storage provider's official SDK or API into your application, allowing you to perform operations like uploading and downloading files. Authenticate your application securely using API keys or access tokens provided by the cloud storage provider to ensure authorized access. Implement file upload functionality, enabling users to upload files from their devices to the designated cloud storage location. Additionally, create features that allow users to download files from the cloud when needed. Implement robust error handling mechanisms to deal with various scenarios, including network issues and file upload failures. Ensure security by encrypting sensitive data, validating user input, and implementing access controls to prevent unauthorized access. Thoroughly test the implemented functionality, including edge cases and concurrent user interactions, to guarantee a seamless user experience. Finally, document the integration process and provide clear instructions for users on how to import and export files using the cloud storage service within your application. By following these steps and adhering to best practices, you can successfully implement cloud file import functionality, enhancing the capabilities of your application.

VI. RESULTS

A. Sign-up authentication

8	NAME
۵	MAILID
	PHONE NUMBER
a	PASSWORD
۵	CONFIRM PASSWORD
	SIGN UP
0	SIGN UP WITH FACEBOOK
G	SIGN UP WITH GOOGLE

C. Caption Generation

Fig 1: Sign-up page

Using Google O-Auth for sig-in and authentication or our dedicated authentication system user can safely and securely authenticate and login.

B. Import media

	Fig 2:
A Inan Ra	Import
Uptional My Drive Denset	media
	option
	We ca
	import
Concession in the local division of the loca	media
ar thag a file here	from t
	cloud ;
	well as

Fig 3. Autogenerate captions

With the help of OpenAI integration, we can generate automated captions for the media.

VII. CONCLUSION

This project helped us to learn about the complicated system of publicly available APIs to the fullest. It works correctly and fulfills all the requirements of users or content creators also we got some insights on how to extend a social media platform. This system properly helps the users to fulfill their tasks related to social media posting.

VIII. REFERENCE

- V. Volobuev and P. Y. Afonichkina, "Generating Photo Captions for Instagram," 2021 IEEE Conference of Russian Young Researchers in Electrical and Electronic Engineering (ElConRus), St. Petersburg, Moscow, Russia, 2021, pp. 735-738, doi: 10.1109/ElConRus51938.2021.9396441.
- Z. Yang, "Probabilistic model-driven video scheduling for the social media application in the Federated cloud," 2014 IEEE International Conference on Multimedia and Expo Workshops (ICMEW), Chengdu, China, 2014, pp. 1-6, doi: 10.1109/ICMEW.2014.6890563.
- 3. Kanuri, Vamsi & Chen, Yixing & Sridhar, Shrihari. (2018). Scheduling Content on Social Media: Theory, Evidence, and Application. Journal of Marketing. 86. 89-108. 10.1177/0022242918805411.

FIRE EYE: A REAL-TIME FOREST FIRE AND SMOKE DETECTION BY USING MACHINE LEARNING

Sonali Wadekar¹, V. D. Jadhav², Swati Pawar³

¹M-Tech Student, Computer Science and Engineering, SVERI's College of Engineering, Pandharpur, ²Assistant Professor, Computer Science and Engineering, SVERI's College of Engineering, Pandharpur ³Associate Professor, Computer Science and Engineering, SVERI's College of Engineering Pandharpur

Abstract— This paper provides a comprehensive review of the current landscape of forest fire detection and prediction, emphasizing the integration of computer vision, machine learning, and deep learning algorithms. Traditional methods for forest fire management are initially discussed, highlighting their limitations and motivating the exploration of more advanced technological solutions. The review explores the role of computer vision techniques in image and video-based forest fire detection, emphasizing the utilization of various features and image processing methods. Machine learning algorithms are then examined for their application in predicting forest fire occurrences, with detailed discussions on model architectures, input features, and evaluation metrics. The study also explores the field of deep learning, demonstrating how various deep learning designs, such as recurrent neural networks (RNNs) and convolutional neural networks (CNNs), can improve the detection and prediction of forest fires. Case studies are presented to illustrate the practical effectiveness of these models. Additionally, the paper explores the integration of multiple modalities, where data from diverse sources, including satellite imagery, weather conditions, and ground sensors, are amalgamated for more accurate forest fire management. Challenges associated with current approaches are addressed, and potential future directions are outlined. In conclusion, the paper emphasizes the critical role of advanced technologies in forest fire management and encourages ongoing research efforts to develop robust systems to mitigate the devastating impact of wildfires on the environment and human safety.

Keywords: Monitoring System, deep learning algorithms, YOLO-v5, early warning system, response time reduction, life and property protection, uncontrollable spread prevention, accuracy.

I. .INTRODUCTION

Wildfires, driven by a complex interplay of environmental factors and human activities, constitute a global challenge with severe repercussions for ecosystems, biodiversity, and human settlements. As climate change exacerbates the frequency and intensity of these events, there is an urgent need for advanced methodologies in forest fire detection and prediction to mitigate their devastating impact. This paper thoroughlyexplores the current research in this field, specifically focusing on integrating computer vision, machine learning, and deep learning algorithms. Colloquially known as wildfires, forest fires have emerged as a pressing environmental concern, transcending geographical boundaries and impacting diverse ecosystems. Their occurrence results from a dynamic interplay between climatic conditions, vegetation patterns, and human activities. Climate change-induced shifts in weather patterns, prolonged droughts, and rising temperatures contribute to an environment conducive to the rapid spread of wildfires. Moreover, anthropogenic factors such as land-use changes, inadequate firemanagement practices, and human negligence amplify the wildfire risk.

The consequences of uncontrolled wildfires are profound. Ecosystems, characterized by delicate balances among flora and fauna, face severe disruption. Biodiversity loss, soil degradation, and altered landscape structures become enduring legacies of these destructive events. Beyond ecological ramifications, wildfires significantly threaten human life, property, and infrastructure. The intensity and speed at which wildfires can spread make early detection and prediction paramount in formulating effective response strategies. Traditional forest fire detection and prediction methods have proven insufficient in addressing the evolving challenges posed by modern wildfire dynamics. While historically integral, ground-based observation systems and manual monitoring lack the scalability and real-time capabilities required to keep pace with wildfires' rapid onset and progression. Consequently, the scientific community has turned to advanced technologies, leveraging computer vision, machine learning, and deep learning algorithms to enhance our ability to detect and predict forest fires.

Computer vision, as a subset of artificial intelligence (AI), plays a pivotal role in processing and interpreting visual data for forest fire detection. Image and video- based analysis, combined with sophisticated algorithms, enable the identification of smoke plumes, flamepatterns, and other visual cues indicative of a wildfire. Integrating machine learning algorithms further refines the detection process by allowing systems to learn from historical data, adapt to changing conditions, and improve accuracy over time. Machine learning, a broader field encompassing algorithms that can learn and make predictions from data, provides a fertile ground for developing predictive models in the context of forest fires. These models leverage diverse data sources, including meteorological data, topographical information, and historical fire records. By discerning patterns

and relationships within this data, machine learning algorithms contribute to creating predictive models capable of forecasting potential wildfire events.

Deep learning is a subset of machine learning that uses deep neural networks, which are multi-layered neural networks, to identify intricate patterns. It draws inspiration from the composition and functions of the human brain. Deep learning models like convolutional neural networks (CNNs) and recurrent neural networks (RNNs) have unparalleled capacity to handle complex spatial and temporal data in managing forest fires. Because of their capacity to automatically extract features from unprocessed data, they are especially good at identifying minor signs that point to potential wildfire hazards. This work aims to compile and critically assess research on deep learning, machine learning, and computer vision to detect and predict forest fires. We hope to further the current conversation about cutting- edge technology development for efficient forest fire control by analyzing the advantages and disadvantages of these strategies. The subsequent sections will delve into the specifics of each technology, presenting case studies, discussing challenges, and proposing potential avenues for future research.

As the global community grapples with the escalating challenges posed by wildfires, the intersection of technology and environmental science emerges as a beacon of hope. By harnessing the capabilities of computer vision, machine learning, and deep learning, we stand at the precipice of a new era in forest fire management—one where predictive accuracy and early detection become not just aspirations but tangible tools for safeguarding our natural heritage and human habitation. Fire Eye is a system developed for detecting and monitoring forest fires using two cutting-edge deep learning algorithms (YOLO-v5 and Inception-v3), designed to detect fires at the fieriest stages. This early warning system critically analyses the scope of artificial intelligence, reduces the response time of fire authorities, and helps them mitigate the loss of life and property caused by forest fires before they spread uncontrollably. This system increases the reliability and accuracy of fire detection and monitoring operations and contributes to forest ecosystems' overall sustainability and resilience. The landscape of forest fire detection has witnessed a significant evolution in response to the escalatin

Challenges posed by climate change and human activities. Early detection systems have become pivotal in mitigating.

Climate change has made forest fires more frequentglobally, causing significant financial losses and ecological devastation. Waste, other biomes and human carelessness bring on summer forest fires. Forest fires can be man-made or natural. Wildfires can benefit the area's flora, fauna, and ecosystems, but they can also seriously harm property and result in fatalities. The number of forest fire incidents has been steadily rising inrecent years. To prevent the devastation of forests, there has thus been an increase in interest in implementing systems for automated observation and detection of forest fires. Many conventional and innovative fire and smoke detection methods have been made available to reduce the damage caused by fire disasters. The research community has focused much on sensor- and vision- based smoke detection systems among these techniques. Depending on the kinds and uses of the sensors employed, the fire detection approach is separated into five primary groups: smoke-sensitive, light-sensitive, gas-sensitive, temperature-sensitive, and composite. For this, temperature and smoke sensors are commonly employed. The speed and range of detection are two significant limits of the sensor-based technique.

Color information has also been combined with motion data. These techniques have increased the dependency on fire detection technologies. However, using photos fromsecurity cameras has brought up a new problem with image processing. Continuous streams of images from video cameras need to be processed and stored, which would incur additional costs. Various fire detection systems and approaches have been offered to make the system as autonomous and accurate as feasible. Machine vision image processing and video surveillance technology have progressed in recent years, resulting in faster transmission and sensing. Thus, the development of computer vision-based fire and smoke detection systems has enabled a greater variety of fire detection methodologies. Using video surveillance to collect and extract data from fire and smoke images, computer vision-based fire and smoke detection systems can build a detection model based on these images. Conventional machine learning and deep learning-based computer vision algorithms have been promoted to assess whether fire and smoke are present in pictures.

II.METHODS FOR FOREST FIRE DETECTION

Different methods were implemented to detect forest fires, including machine learning and deep learning. This section describes the different approaches for forest fire detection in detail.

A. Machine Learning-Based Forest Fire Detection

B. Pradeep Kumar G et al. [1] detection of forest fires usingmachine learning methods has used random forest and kto detect the forest fire. Users discuss an innovative fire detection policy that uses all technological tools. Moreover, proposes a platform for artificial intelligence–based. Computer vision for smoke and fire recognition and detection is based on still imagery or streaming information in the pictures taken. To determine the result, using machine learning. These algorithms have implemented data sets and split them into test and train sets, affecting dependability.

To detect forest fires, lasso, ridge, and linear regression are presented by Ahmed M. Elshewey et al. [2]. This study presents three machine learning approaches: ridge regression, lasso regression, and linear regression, using a data set of 517 entries with 13 attributes per row. This work has two versions; the first has all the features, while the second has only 70% of the functionality. The work also uses the test set, which makes up 30% of the data set, and the training set, which makes up 70%. The linear regression approach is more accurate than ridge regression and lasso regression algorithms.

David A. Wood [3] published the Forest Fire Prediction and Data Mining paper: A highly effective data matching and mining system offers insightful information. The multiple forecasting approach offers an instructive feature selection that determines the relative impacts of the crucial parameters on forecast burned-area numbers. Data analysis for each total fired incident requires complementary information, which would be supplied through maximizing with MAE and RMSE as different goal activities. By increasing comprehension of the critical factors that affect each burn occurrence, people will receive insight into farming, biodiversity, the environment, and forestry. Each maximized solution's suitability for correctly guessing burned-area events from a particular type may be evaluated using specific insight obtained from dataset mining the contrasting squared and cumulative total error trends. Such effectiveness and understanding help build strength in the methodology used to make each prediction. It gives the knowledge to behave quickly and minimize individual burn situations as they happen. By attempting to stop specific sorts of burn accidents from occurring once again or spreading, such educated interventions should both have short- and long-term, multifaceted benefits.

Pragati et al. [4] Wireless Sensor Networks (WSN) are becoming more important in current research fields as technology progresses due to their effectiveness in preventing accidents and helping people. An incident is recognized using the sensor devices positioned at various places once an uncommon occurrence in the networks is discovered. The base station receives this event detection information and makes a decision. Such sensor data in WSNs typically generate false alarms because of their static design. Due to their efficient dynamic configuration and autonomous operation, machine learning algorithms can be used to prevent false alarms.

Adithi M. Shrouthy et al. [5] The approach for predicting the danger of forest fires using SVM, Decision trees, Random Forest, KNN, and Logistic Regression are discussed. The results demonstrate that the threat of forest fires can be predicted with some degree of precision. So, based on various characteristics like data input on oxygen, temperature, and humidity by the un front end, designers suggested a system to forecast the percentage of fire occurrence using machine learning methods and algorithms such as Decision Tree, RandomForest, KNN, SVC, and Logistic Regression.

K.G. Shangavi et al. [6] researchers used sensors to examine forest fires at various locations and times in this study. By examining the time it takes to produce the output, the error rate, and a variety of other factors, we can determine whether rattling is an effective instrument. According to the findings, the decision tree is more effective than random forest regarding time and mistakerate.

Mohana Kumar S. et al. [7] Due to several factors, including the rapid advancement of digital camera technology, image processing has been used in this study. When covering large regions, the camera performs exceptionally well, and its response time for handling photographs is faster than that of existing sensor architectures. Moreover, picture preparation frameworks are generally less expensive than sensor frameworks. Final findings reveal that the suggested strategy for detecting forest fires has a high detection rate (92%) and a low percentage of false alarms (8%).

Gomathy, C. K., et al. [8] This article compares machine learning methods like neural networks, SVM, regression, and decision trees. Compared to other machine learning techniques, this paper requires less time to detect forest fires immediately. This study proposes a fire detection method that divides datasets into months and uses regression. The algorithm can achieve high R-squared and low root mean square error.

Ramasubramani Srinivas [9] This study proposes several machine learning algorithms, such as bagging, gradient boosting, SVR, random forest, linear regression, logistic regression, and random forest, for predicting the amount of forest land burned. Consequently, the fire breakouts in Portugal's northeast are used to develop the predictive model. In this research, researchers create a model for predicting the area burned during forest fires using Big Data and machine learning approaches.

Suhas G et al. [10] With this work, researchers want to advance deep learning's ability to detect fire in videos. Artificial neural networks-based deep learning is a new idea that has produced outstanding results in several domains, including computer vision.

A. Deep Learning-Based Forest Fire Detection

Priya and Vani's work [11] explored using InceptionV3 to enhance the satellite image classification of forest fires. 239 fire and 295 non-fire satellite photos, totaling 534, were used to validate their work. The performance of the DL model for each manuscript is detailed, along with augmentation.

The classification job in intelligent forest monitoring systems is one of the oldest and most studied [12]. Imageclassification is the process of labeling and classifying groups of pixels or vectors inside an image based on predefined criteria, as defined by Shinozuka and Mansouri [95]. They contended that one or more of an image's spectral or textural characteristics could be used to

create the categorization rule [13]. The main objective of picture categorization is to guarantee that every photois categorized by its relevant sectors or categories [14].

Using a portable Raspberry Pi platform, Arteaga et al.

[15] evaluate many pre-trained CNN models for forest fire classification rather than focusing on a single ResNetmodel. In this inquiry, 1800 images from a medium-sized database gathered from internet secondary sources were employed. The authors applied and added a set of augmented data to the training dataset. After applying a cropping approach to increase the data's width to 224 pixels, it was rotated horizontally with a 50% chance, and then its standard deviation and mean values from the ImageNet database were used to normalize it. The authors investigated several pre-trained VGG and ResNet model variations. The outcomes demonstrated that ResNet-18 processed data in less than 2.12 seconds and produced an excellent accuracy performance of 0.9950. Furthermore, their research demonstrated that the ResNet-34, ResNet-101, ResNet-50, and ResNet-18 models are better suited for implementation on mobile platforms than the VGG variants for detecting forest fires. The authors must conduct experiments with larger datasets to determine if the algorithms can handle big datasets in actual forest fire scenarios.

Ban et al. [16] employed their CNN model to extract the temporal backscatter variations information from Synthetic Aperture Radar (SAR) imagery obtained during wildfire episodes and SAR imaging time-series data gathered before the occurrences to automatically detect burnt zones. Additionally, an inventory map consisting of 10,000 points representing burned and unburned areas has been created using Sentinel-2 photographs to verify and confirm their findings. Additionally, they further employed visual comparisons to enhance the datasets obtained from Sentinel-2's burned area maps and SAR-based progress maps. Burnt confidence maps are created by fitting and training the CNN model, which will aid in identifying burned areas. The coarse binary transition map was used to create the training images automatically. The Otsu thresholding technique will then be used to binarize these confidence maps in order to increase output dependability and certainty. Subsequently, the maps will be gradually amalgamated. Sentinel-1 SAR data has limitations, including signal deterioration in certain environmental conditions and spatial resolution, although these are not discussed.

Benzekri et al. [17] presented a novel DL model in this work that predicts the final label by employing an RNN,LSTM, or GRU in the output layer and two hidden layers with 50 neurons each. The network backpropagated the loss function using the Adam optimizer. Four wrong predictions were made by the LSTM model, two by the simple RNN model, and one by the GRU model. The authors used about 2000 sample data points to analyze the three models. The LSTM model yielded 0.0298 loss and 99.82 percent accuracy on the test data. The accuracy of the baseline RNN model was 99.77 percent, with a loss of 0.0062. For early forest fire detection, the most accurate and appropriate model is the GRU model. According to the authors, the approach is more accurate than traditional monitoring techniques. In contrast to thereal world, a short dataset was used to test the high- accuracy results; a larger sample is required.

A comparison of three DL designs was done by Rahul etal. [18]. The authors utilized the ResNet-50, VGG-16, and DenseNet-121 models in their study to investigate forest fire detection. Flipping, shearing, and other methods are applied to the input pictures once they have been scaled to 224 pixels in width. A softmax layer, a pooling layer, a batch normalization layer, a convolutional layer, and a ReLu activation layer with dropout make up the traditional CNN layer architecture for image classification. The findings determined the best update backpropagation method with the best globalextremum as the stochastic gradient descent (SGD) optimizer. Overall, ResNet-50 performed better than DenseNet-121 and VGG-16. According to the results, theSGD optimizer is more effective at identifying forest fires than the Adam optimizer. However, assessing how broadly the findings may be challenging because the study does not specify the dataset used to train and test the model.

Jiang et al. [19] adjusted the CNN model hyperparameters using a genetic algorithm (GA) to predict fire occurrences accurately. The authors used the backpropagation (BP), support vector machine (SVM), GA-CNN, and CNN techniques as benchmarks to compare their methodology. The development dataset comprises 1900 images from the testing and training datasets. Positive and negative imagery of smoke and fire situations are included in the bulk of the photographs. It functions well under various assessment scenarios concerning accuracy, false alarm, and true-positive levels. The accuracy values of the unoptimized CNN, BPneural network, and SVM algorithms are 73 percent, 75 percent, and 85 percent, respectively.

In comparison, the optimized GA-CNN method's accuracy value is 95 percent (90 percent). Additionally, the study found that the GA-CNN approach might potentially identify forest fires. Given that an overfitting issue might arise from the imbalanced data, the authors ought to take the accuracy and recall measures intoaccount for more accurate result interpretation.

Gayathri et al. [20] used LSTM and CNN in a hybrid configuration of a bidirectional algorithm instead of integrating CNN with regular LSTM. The method took advantage of Google's Firebase, which can be connected to mobile or Internet of Things devices through alerts to convey warnings. The recommended model's accuracy rate for the training dataset is 96%, while for the test dataset, it is 92%. The findings imply that superior classification results for forest fires can be obtained by combining two DL models. The study's overfitting problem is apparent from the findings, which show that the precision and recall were poor even with a high accuracy value.

Instead of utilizing a single model, Ghosh and Kumar

[21] integrated CNN and RNN networks to extract the features, which are then sent to two fully linked layers for classification. The Mivia collection comprises 22,500 images, 12,000 of which feature scenes involving smokeor fire, and the remaining 10,000 do not. The Kaggle dataset comprises 1000 photographs, of which 755 are classified as fire photos and the other 245 as regular photos.

The study examines many approaches to forest fire detection in this thorough analysis, focusing on combining deep learning (DL) and machine learning(ML) techniques. The study encompasses both ML and DL-based approaches, providing an in-depth analysis of various techniques employed in the literature.

A variety of algorithms, including random forest, k- nearest neighbors, linear regression, ridge regression, lasso regression, and ensemble techniques, are included in the ML-based approaches that have been examined. Researchers propose innovative fire detection policies, artificial intelligence-based platforms, and dynamic wireless sensor networks to prevent false alarms. Using regression models with feature selection and data mining techniques reveals insightful information for predicting burned areas and understanding the impact of crucial parameters.

The DL-based methods investigate the use of hybrid models, recurrent neural networks (RNNs), and convolutional neural networks (CNNs) in detecting forest fires. Researchers aim to improve classification performance and prediction accuracy by utilizing advanced DL architectures like InceptionV3, ResNet-18, and genetic algorithm-optimized CNNs. They delve into integrating synthetic aperture radar (SAR) imagery, bidirectional algorithms, and combinations of RNN and

III. A FIGURES AND TABLES

CNN networks to enhance feature extraction and model performance. The discussion emphasizes the strengths of these approaches, including increased accuracy, efficiency, and adaptability to dynamic scenarios. ML models leverage historical data and real-time sensor information effectively, while DL models excel in image classification, feature extraction, and predictive accuracy.

Challenges and considerations for future research are highlighted, encompassing issues such as model interpretability, scalability, biases in datasets, and generalizability across diverse ecological contexts. The discussion underscores the importance of continued collaboration between researchers, technologists, and environmental experts to address these challenges and fully realize the potential of ML and DL technologies inforest fire detection.

Reference	Approach	Algorithms Used	Dataset Used	Key Findings
[1]Pradeep Kumar G et al.	ML	Random Forest, KNN	Not specified	Innovative fir detection technological tools. Platform policy using for AI-based computer vision.
[2] Ahmed M. Elsheweyet al.	ML	Lasso, Ridge, Linear Regression	517 entries, 13 attributes	Linear regression more accurate than ridge and lasso regression.
[3] David A.	ML	Not specified	Not specified	Effective data matching and mining system. Insightful information for forecasting burned- area numbers.
[4] Pragati et al.	ML	Machine Learning Algorithms	Wireless Sensor Networks	ML algorithms prevent false alarms in WSNs due to dynamic configuration.
[5]AdithiM.Shrou thy et al.	ML	SVM, Decision Trees, Random Forest, KNN, Logistic Regression	Not specified	Predicting forest fire danger using various ML algorithms with precision.
[6] K.G. Shangavi	ML	Decision Tree	Sensor data	Decision tree more effective than random forest regarding time and mistake rate.
[7] Mohana Kumar S. et al.	ML	Image Processing	Digital camera technology	High detection rate (92%) and low false alarms (8%) using suggested forest fire detection strategy.

Table I: Summary of the recent study for forest fire detection using ML and DL algorithms

[8] Gomathy, C.K., et al.	ML	Neural Networks, SVM,Regression, Decision Trees	Not specified	Proposed fire detection method with regression achieves high R-squared and low RMSE.
[9] Ramasubramani Srinivas	ML	Bagging, Gradient Boosting, SVR, Random Forest, Linear Regression, Logistic Regression	Fire breakouts in Portugal's northeast	Model for predicting burned area during forest fires using Big Data and ML approaches.
[10] Suhas G et al.	ML	Deep Learning (ANNs)	Videos	Advancing deep learning for fire detection in videos using artificial neural networks.
[11] Praia and Vani	DL	InceptionV3	534 satellite photos	textural characteristics

[12] Shinozu kaand Mansouri	DL	Not specified	Not specified	ResNet-18 performs well for forest fire classification with high accuracy.
[15] Arteaga et al.	DL	Pre-trained CNN models (VGG,ResNet)	1800 images	CNN model extracts temporal backscatter variations from SAR imagery to detect burntzones.
[16] Ban et al.	DL	CNN	SAR imagery, Sentinel-2 photographs	GRU model proves most accurate for earlyforest fire detection.
[17] Benzekri et al.	DL	RNN, LSTM, GRU	2000 sample datapoints	ResNet-50 outperforms other models in forestfire detection. SGD optimizer more effective.
[18] Rahul et al.	DL	ResNet-50, VGG-16, DenseNet-121	Not specified	GA-CNN method achieves 95% accuracy forfire occurrence prediction.
[19] Jiang et al.	DL	CNN, GA-CNN	1900 images	Hybrid configuration of LSTM and CNN for forest fire classification. Superior results butpotential overfitting.
[20] Gayathri et al.	DL	LSTM, CNN (bidirectional)	Not specified	CNN and RNN networks combination achieves high accuracy in identifying forest fires.
[83] Ghosh and Kumar	DL	CNN, RNN	Mivia collection,Kaggle dataset	early forest fire detection.





in Fig.1 factors. Fires.as shown fig

Fig.2 and its intensity, fires and the degree of damage

Intense, large-scale forest fires are damaging and very challenging to control. Locations, where various types of fire behavior occur, vary depending on environmental factors fires.as shown

IV. CONCLUSION

This review underscores the significant strides in forest fire detection by integrating machine learning (ML) and deep learning (DL) algorithms. The amalgamation of innovative methodologies and advanced technologies has enhanced our ability to predict, detect, and manage forest fires, marking a paradigm shift in wildfire management strategies. The collective findings from the examined literature showcase the diverse applications and effectiveness of both ML and DL approaches, each contributing unique strengths to the overall landscape of forest fire detection. Machine learning-based approaches, ranging from ensemble methods like random forest to regression models and data mining techniques, have demonstrated promising results. These techniques leverage historical data, sensor networks, and dynamic wireless sensor configurations, providing insights into predictive modeling and feature selection. The exploration of ML algorithms enhances the accuracy of forest fire predictions and addresses the critical issue of false alarms, a common challenge in traditional detection systems.

With its robust convolutional neural networks (CNNs) and recurrent neural networks (RNNs), deep learning offers robust solutions for image classification, feature extraction, and complex pattern recognition. Using sophisticated DL architectures, such as InceptionV3 and ResNet-18, signifies a breakthrough in satellite image analysis and mobile platform implementations. The integration of SAR imagery, bidirectional algorithms, and combinations of RNN and CNN networks showcases the adaptability and effectiveness of DL models in diverse scenarios.

While the advancements are promising, challenges persist. Critical considerations include model interpretability, scalability to handle diverse datasets, and addressing biases in training data. Future research directions should focus on integrating multiple modalities, including satellite imagery, ground-based sensor data, and real-time weather information, to further enhance the accuracy and reliability of forest fire predictions

V. REFERENCES

- 1. Pradeep Kumar G, Rahul R, Ravindharan N "Early Forest Fire Detection Using Machine Learning Algorithms" International Journal of New Technology and Research (IJNTR) ISSN: 2454-4116, Volume-7, Issue-4, April 2021.
- 2. Ahmed M. Elshewey, Amira. A. Elsonbaty "Forest Fires Detection Using Machine Learning Techniques" Journal of Xi'an University of Architecture & Technology ISSN No: 1006-7930 Volume XII, Issue
- 3. David A. Wood "Prediction and data mining of burned areas of forest fires: Optimized data matching and mining algorithm provides valuable insight" Artificial Intelligence in Agriculture 5 (2021).
- 4. Pragati, Sejal Shambhuwani, Piyusha Umbrajkar "Forest fire detection using machine learning" International Journal Advances Scientific Research and Engineering Trends Volume 4, Special Issue 12, ICCEME 2019-2020.
- Adithi M. Shrouthy, Syed Matheen Pasha, Yamini S.R. E, Navya Shree S, Lisha U. "Forest fire predictionusing ML and AI" International Journal of Advance Research, Ideas and Innovations in Technology ISSN: 2454-132X Impact Factor: 6.078 Volume 7, Issue 3.
- 6. K.G. Shangavi, K.K. Sindu, S. Gomathi, "Forest fire prediction with machine learning" Vol. 6, Issue 6 www.jetir.org (ISSN-2349-5162), June 2019.
- Mohana Kumar S, Sowmya B J 2, Priyanka S3, Ruchita Sharma4, ShivankTej5 Spoorthi Ashok Karani6 "Forest Fire Prediction Using Image Processing and Machine Learning" Nat. Volatiles & Essent. Oils, 2021; 8(4): 13116-13134.
- Dr. C. K. Gomathy1, Mr. Majjari Pedda Chandrasekhar2, Mr. Kalavakuri Mallikarjun3 "Forest Fires Detection using Machine Learning" International Research Journal of Engineering and Technology (IRJET) e-ISSN: 2395-0056 Volume: 08Issue: 10 | Oct 2021.
- 9. Srinivas Ramasubramanian "Predicting the burnedarea in forest using Machine learning techniques" 2014 International Conference on, IEEE, pp. 312–316.

- 10. Suhas G, Chetan Kumar, Abhishek B S, Digvijay Gowda K A, Prajwal R., "Fire Detection Using Deep Learning" International Journal of Progressive Research in Science and Engineering Volume-1, Issue-5, August-2020.
- 11. R.S. Priya, K. Vani, Deep learning based forest fire classification and detection in satellite images, in Proceedings of the 11th International Conference on Advanced Computing, ICoAC 2019 2019, pp. 61–65.
- H. Liao, S.K. Zhou, J. Luo, Classification: lesion and disease recognition, in Deep Network Design for Medical Image Computing, 2023, pp. 27–58.
- M. Shinozuka, B. Mansouri, Synthetic Aperture Radar and Remote Sensing Technologies for Structural Health Monitoring of Civil Infrastructure Systems, Structural Health Monitoring of Civil Infrastructure Systems, 2009, pp. 113–151.
- 14. M.A. Abu, N.H. Indra, A.H.A. Rahman, N.A. Sapiee, Ahmad, A study on image classification based on deep learning and TensorFlow, Int. J. Eng. Res. Technol. 12 (4) (2019) 563–569.
- 15. Arteaga, M. Diaz, M. Jojoa, Deep Learning Applied to Forest Fire Detection. 2020 IEEE International Symposium on Signal Processing and Information Technology, 2020. ISSPIT 2020.
- 16. Y. Ban, P. Zhang, A. Nascetti, A.R. Bevington, M.A. Wulder, Near real-time wildfire progression monitoring with sentinel-1 SAR time series and deep learning, Sci. Rep. 10 (1) (2020) 1–16. A. Saleh et al.
- 17. W. Benzekri, A. El Moussati, O. Moussaoui, M. Berrajaa, Early Forest fire detection system using wireless sensor network and deep learning, Int. J. Adv. Comput. Sci. Appl. 11 (5) (2020) 496–503.
- M. Rahul, K. Shiva Saketh, A. Sanjeet, N. Srinivas Naik, Early detection of forest fire using deep learning, in IEEE Region 10 Annual International Conference, Proceedings/TENCON, 2020-Novem, 2020, pp. 1136–1140.
- 19. Y. Jiang, R. Wei, J. Chen, G. Wang, Deep learning of qinling forest fire anomaly detection based on genetic algorithm optimization, UPB Scientific. Bull., Ser. C: Electr. Eng. Comp. Sci. 83 (4) (2021) 75–84.
- S. Gayathri, P.V. Ajay Karthi, S. Sunil, Prediction and detection of forest fires based on deep learning approach, J. Pharm. Negat. Results 13 (3) (2022) 429–433.
- R. Ghosh, A. Kumar, A hybrid deep learning model by combining convolutional neural network and recurrent neural network to detect forest fire, Multimed. Tool. Appl. 81 (27) (2022) 386

DATA-POWERED VOICE MATIC CHATBOT CONVERSATION FOR ENHANCED LEARNING

Mohini Jagtap1,. Archana More2, Nagma Shaikh3 ¹²³ Student of SVERI's College of Engineering, Pandharpur, India Punyashlok Ahilyadevi Holkar, Solapur University Solapur

Abstract — In the contemporary landscape of education and technology, the demand for innovative learning solutions has paved the way for the development of cutting-edge projects. One such endeavor is the creation of a Data-Powered VoiceMatic Chatbot Conversation designed to enhance the learning experience. This project converges advancements in chatbot technology, machine learning, and Android application development to offer a dynamic and personalized learning platform.

The primary objective of this project is to revolutionize traditional learning paradigms by introducing an intelligent and conversational interface. Users can engage in interactive conversations with the Voice Matic Chatbot, receiving personalized guidance, educational content, and real-time feedback. Through theutilization of machine learning, the chatbot continuously refines its understanding user preferences, thereby optimizing the learning journey for each individual. **Keywords:** Voice Matic Chatbot, Machine Learning, Android Application, Real-time Feedback

I. INTRODUCTION

In essence, the project envisions the creation of a Voice Matic Chatbot—a sophisticated and intelligent conversational interface—that harnesses the power of machine learning to offer an enriched and personalized learning experience. By being data- powered, the chatbot leverages insights from user interactions to continuously adapt and refine its responses, tailoring the learning journey to individual preferences and needs. The emphasis on voice interaction adds a dynamic layer to the traditional learning paradigm, making the educational process more engaging and accessible. This project aligns with the contemporary shift towards user-centric learning, acknowledging the diverse ways in which individuals absorb information and providing a tailored, responsive, and interactive learning environment.

II. LITERATURE SURVEY

The literature survey for the project "Data-Powered voice Matic Chatbot Conversation for Enhanced Learning" from a machine learning perspective involves exploring key domains within the field. Initial research focuses on the integration of chatbots, emphasizing studies that employ machine learning techniques to enhance conversational capabilities in educational contexts. Investigations into natural language processing (NLP) contribute insights into how advanced language models can improve the chatbot's understanding and response generation. The survey extends to machine learning applications in adaptive learning systems, analyzing approaches that dynamically adjust content based on user interactions. Additionally, studies on voice interaction in machine learning models are explored, emphasizing how voice-enabled technologies leverage ML algorithms to simulate human-like communication patterns. The literature review also incorporates research on data-driven learning analytics, highlighting howmachine learning algorithms can extract valuable insights from user interactions to optimize the learning factors such as model interpretability and ethical considerations. The survey concludes by examining emerging trends in machine learning within educational technology, providing a foundation for the development of the proposed Data-Powered voice Matic Chatbot for Enhanced Learning.

III. PROBLEM STATEMENT

"Data-Powered voice Matic Chatbot Conversation for Enhanced Learning"

OBJECTIVE

Natural Voice Interaction: Develop a chatbot that enables users to engage in conversations using spoken language, creating a more intuitive and human-like interaction experience. Enhance Accessibility: Create a chatbot that caters to individuals who may face challenges with text-based interfaces, ensuring inclusivity and accessibility for adiverse user base. Seamless

Multitasking: Design thechatbot to accommodate users' multitasking needs by allowing them to interact and gather information while performing other tasks.

IV. .FUTURE SCOPE

This project uses artificial intelligence and machine learning techniques to create chatbots. Chapter Harassment Chapter chatbot learns as itprogresses because it is different from human- created relationships Chapter Rapid change Thelanguage used in this project is in Python. ArticleThe field of machine learning and artificial intelligence is one of the current trends in computer science and technology. In these sectors, 4,444 new technology announcements are made every year. Chapter Research in these areas is still ongoing. However, they are not fullydeveloped and are still "learning". Our chatbot can learn new things every day. So there are no fixed standards for chatbot.



Fig. chatbot

V. PROPOSED SYSTEM

24/7 Availability: Chatbots can provide instant responses and support round the clock, enhancing customer service and user engagement.

Efficiency: Chatbots can handle multiple conversations simultaneously, improving efficiency and reducing waiting times for users.

Cost Savings: Automated interactions with chatbots reduce the need for human customer support, resulting in cost savings for businesses.

Consistent Responses: Chatbots provide consistent and accurate information to users, minimizing the risk of human error.

Instant Information: Chatbots can quickly retrieve and present information, helping users access relevant data without delays.

Scalability: Chatbots can scale easily toaccommodate increased user demand without significant resource allocation.

VI. CONCLUSION

The integration of a voice chatbot represents a significant advancement in conversational AI technology, offering a seamless and natural interaction between users and digital systems. Through the utilization of voice-based interfaces, the chatbot facilitates a more intuitive and user- friendly experience, eliminating the need for traditional text-based input. The project's exploration of voice chatbot technology has shown promising results, opening avenues for enhanced accessibility and engagement. The adaptability of voice interactions in diverse applications, from customer service to interactive learning, highlights the versatility and potential impact of this technology. As the project recognizes theimportance of continuous improvement and user feedback, the voice chatbot emerges as a dynamic tool with the capacity to evolve and meet evolving user needs. Overall, the project underscores the transformative role of voice chatbot technology in shaping the future of human-computer interactions, promising a more natural, efficient, and user-centric digital experience.

VII. REFERENCES

 $\label{eq:linear} 1. https://www.google.com/search?q=International+Research+Journal+of+Engineering+and+Technology+(IRJET)&oq=International+Research+Journal+of+Engineering+and+Technology+(IRJET)&aqs=chrome..69i57j0i512.691j0j7&sourceid=chrome&ie=UTF-8\\ \end{tabular}$
INCLUSIVE PUBLIC TRANSPORTATION: A HOLISTIC APPROACH TO ACCESSIBILITY AND USEREXPERIENCE FOR VISUALLY IMPAIRED TRAVELERS

Atharva Hande ^{*1}, Shruti Deshpande^{*2}, Vaibhavi Gawade^{*3}, HarshGhodgaonkar^{*4}, Mr. Jitendra Chavan^{*5} Department Of Information Technology, Marathwada Mitra Mandal College OfEngineering, Pune, India.

Abstract— The integrated mobile solution for visually impaired individuals represents a cutting-edge technology designed to significantly enhance mobility and independence, particularly in the realm of public transportation. By leveraging artificial intelligence (AI) and location-based services, the system offers a comprehensive suite of features, including advanced bus number recognition, precise map navigation, and valuable information on key points of interest such as bus stops, restaurants, hospitals, banks/ATMs, educational institutions, and public buildings. In addition to its navigation capabilities, the system endeavors to assist visually impaired users in providing a holistic solution to improve overall accessibility and empower individuals in navigating both transportation systems and daily activities. This research paper explores the comprehensive scope, objectives, and technological components of this innovative solution, emphasizing its potential to positively impact thequality of life for visually impaired individuals. **Keywords:** OCR, Edge detection, GPS.

I.INTRODUCTION

The primary purpose is to provide visually impaired individuals with equal access to essential services and public transportation. By recognizing bus numbers, identifying nearby points of interest, and offering navigation guidance, the system ensures that visually impaired individuals can participate more fully in society. The system empowers visually impaired users to navigate public transportation systems confidently. It assists inlocating bus stops, reading bus numbers, and providing real-time information, reducing the risk of accidents or getting lost. The system's integration with various points of interest, such as restaurants, hospitals, banks/ATMs, educational institutions, and public buildings, enhances the quality of life by making daily tasks and activities more accessible and manageable.

The scope of the bus number recognition system for visually impaired individuals is comprehensive and focuses on enhancing accessibility, independence, and overall quality of life. The system's primary objectives include recognizing bus numbers through advanced AI and computer vision, assisting users in identifying the correct buses, and integrating seamlessly with mapping and navigation services to provide real-time directions and routes [17]. It will also identify nearby bus stops, provide essential bus stop information, and assist users in reaching their destinations efficiently. Furthermore, the system extends its capabilities to include locating nearby restaurants, hospitals, banks/ATMs, educational institutions, and public buildings, offering users valuable information, directions, and the ability to filter options based on their preferences. To ensure a user-friendly experience, the system supports various accessible means of interaction, such as voice commands, while prioritizing data security, privacy, scalability, and reliability. Extensive testing and quality assurance will be integral to the project to guarantee the accuracy and usability of all features, making the system an indispensable tool for visually impaired individuals seeking to navigate and access essential services seamlessly

II.METHODOLOGY

the essential details, making it easier for OCR to precisely identify and recognize bus numbers. This contributes to the overall accuracy of the system, ensuring visually

The Optical Character Recognition (OCR) component within the integrated mobile solution for visually impaired individuals plays a pivotal role in enhancing accessibility. The system's OCR functionality utilizes advanced artificial intelligence (AI) and computer vision techniques to recognize and announce bus numbers, facilitating independent navigation for visually impaired users

Here's an overview of the OCR information for the research paper:

OCR Component:

The OCR component, integral to the system's functionality, employs cutting-edge AI and computer vision techniques for recognizing text elements, specifically focusing on bus route numbers. This component, referred to as the 'Scanner,' captures and processes images, employing techniques like edge detection to enhance features and remove distortions. Following image processing, Optical Character Recognition (OCR) is applied to extract relevant text information, such as bus numbers, from the scanned images. This OCR process enables the system to effectively read and communicate bus route numbers to visually impaired users.

OCR Workflow:

- Image Capture: The 'Scanner' captures images of vehicles, specifically designed for scanning bus route numbers. 1.
- 2. Image Processing: Image processing techniques, including edge detection, enhance features and remove distortions to optimize text recognition.
- Optical Character Recognition (OCR): After image processing, OCR is applied to extract text, such as bus route 3. numbers, from the images [10].
- Announcement: The recognized text is then announced to the visually impaired user, providing real-time information 4. about bus numbers and ensuring accurate identification.
- Edge Detection 5.
- Edge detection plays a crucial role in our project for visually impaired individuals, specifically in the bus number 6 recognition system (Scanner). It serves to enhance image quality, reduce noise, and improve feature extraction for accurate Optical Character Recognition (OCR). By highlighting edges, the algorithm focuses on
- impaired users can confidently and independently locate the correct buses with minimal risk of errors or 7. misinterpretations.
- 8. Image Enhancement: Edge detection is crucial to improve image features and reduce distortions before recognizing bus numbers, enhancing clarity for accurate identification.
- 9. Noise Reduction: Edge detection filters out unnecessary details and noise in real-world images, ensuring accurate Optical Character Recognition (OCR) by focusing on essential bus number features.
- 10. Improved Feature Extraction: Edge detection isolates and emphasizes text or numbers on the bus, contributing to accurate feature extraction for precise OCR during recognition.
- Facilitating OCR Processing: Identified edges guide the OCR algorithm, allowing it to focus specifically on 11. highlighted edges, streamlining the identification and recognition of characters like bus numbers.
- 12. Enhancing Overall System Accuracy: Edge detection improves image quality and feature focus, crucial for the accuracy of the bus number recognition system. This enhances the confidence and independence of visually impaired users in identifying the correct bus.

III.MODELING AND ANALYSIS

The modeling and analysis for the bus number recognition system involve developing image processing models with edge detection algorithms to enhance clarity and reduce noise in real-world bus images. Integration of OCR with these processed images ensures accurate bus number identification. A feature extraction model emphasizes text edges on buses, contributing to precise OCR. The navigation and routing model integrates location-based services, while the user-friendly interaction model supports voice commands. Seamless integration with points of interest is achieved for enhanced accessibility. A robust testing and quality assurance model ensures accuracy and reliability, evaluating system performance in diverse scenarios, making the system an effective tool for

visually impaired individuals.





Figure 1: System Architecture

1. Scanner: The system uses artificial intelligence and computer vision techniques to recognize and announce the bus number to visually impaired users. This helps them locate and board the correct bus with ease, reducing the need for external assistance. The 'Scanner' will capture images, specifically designed to scan vehicles for their bus route number. First the image will be processed to enhance features or remove distortions using edge detection. Optical Character Recognition (OCR) is used after edge detection to read text from images, such as numbers on a vehicle.

2. **Navigator:** The 'Navigator' is part of the system that deals with providing directions and routing. The system integrates with location-based services, such as the Google Maps API, to provide navigation guidance. It helps visually impaired individuals identify their current location, find nearby bus stops, and navigate to their desired destinations. It likely includes GPS technology to assist with real-time location tracking and guidance. It takes input directly from the 'Map Data Provider', which suggests that it uses both real-world scanning information and digital map data to function. The navigator would process this information, using a 'Routing Engine' to determine the best course of travel, and then integrate this with 'GPS Integration' for real-time navigation support. The processed navigation data likely contributes to the final output of the system, which could be visual directions on a map or spoken navigation cues.

Information about Points of Interest: The system integrates with various points of interest, such as restaurants, hospitals, banks/ATMs, educational institutions, and public buildings. It provides visually impaired users with information about these locations, making daily tasks and activities more accessible and manageable.

3. **Word Speller:** The system assists visually impaired individuals in spelling words correctly. This feature can be particularly helpful when they need to input or verify text information. The Word Speller' refers to a component in a speech recognition system that takes spoken input and converts it into its corresponding spelled-out form. In the diagram, it receives speech input, which it sends to the 'Feature Extraction' process. This part of the speech recognition process where distinguishing features of the spoken words are extracted to be analyzed. Lexicon and Acoustic Model are databases which are used for speech recognition. The lexicon matches sounds to words, and the acoustic model interprets the audio signals. Decoder processes the extracted features using the lexicon and acoustic model to decode the spoken words into text. After speech is decoded, it goes through a pattern classification process, likely to understand the context or meaning [12].



Figure 2: DFD Level 0

IV.RESULTS AND DISCUSSION

We have used Tesseract OCR for model image detection that yielded promising results with an achieved accuracy of 87%. Tesseract, an open-source OCR engine developed by Google, has proven effective in extracting text from images. In this scenario, the model was tasked with recognizing and extracting textual information from images, and the achieved accuracy of 87% reflects the ability to accurately decipher and interpret the text present in the images. The success can be attributed to Tesseract's robust character recognition capabilities, which enable it to handle various fonts, sizes, and styles of text. The 87% accuracy indicates a strong performance in accurately transcribing the textual content from the model images, showcasing Tesseract's suitability for OCR tasks in image- based data processing. Fine-tuning or preprocessing techniques may further enhance accuracy for specific use cases or challenging image conditions.

V. CONCLUSION

Our proposed framework for visually impaired people marks a significant advancement in utilizing technology to cater to their specific needs within the realm of public transportation. This cutting-edge solution, driven by a strong sense of social responsibility and a commitment to technological innovation, combines artificial intelligence and location-based services. By seamlessly integrating bus number recognition, maps navigation, and information about key points of interest, the system empowers individuals with visual impairments, facilitating independent use of public transportation and enhancing overall mobility. The incorporation of Optical Character Recognition (OCR) technology further extends its utility, enabling users to interact with textual information independently. This comprehensive and inclusive approach aims to reduce reliance on external assistance, fostering greater autonomy for visually impaired individuals.

VI. REFERENCES

- Sarmad Majeed Malik and Rehan Hafiz, "Automatic Number Plate Recognition Based on Connected Components Analysis Technique", IEEE 2nd International Conference on Emerging Trends in Engineering and Technology (ICE TET'2014), pp. 30-31, 2014.
- 2. Automatic number-plate recognition", Wikipedia Wikimedia Foundation, February 2021.
- 3. Visual impairment and blindness", [online] Available: <u>http://www.who.int/mediacentre/factsheets/fs282/en/</u>
- 4. P. Viola and M. J. Jones, "Robust Real-Time Face Detection", Int. J. Comput. Vis., vol. 57, no. 2, pp. 137-154, May 2004.
- D. Lee, H. Yoon, C. Park, J. Kim and C. H. Park, "Automatic number recognition for bus route information aid for the visuallyimpaired", 2013 10th International Conference on Ubiquitous Robots and Ambient Intelligence (URAI), pp. 280-284, 2013.
- K.N. Natei, J. Viradiya and S. Sasikumar, "Extracting text from image document and displaying its related information", J. Eng. Res. Appl, vol. 8, no. 5, pp. 27-33, 2018.
- A. Kumar, "An efficient approach for text extraction in images and video frames using gabor filter", International Journal of Computer and Electrical Engineering, vol. 6, no. 4, pp. 316, 20
- M. Z. H. Noor, I. Ismail and M. F. Saaid, "Bus detection device for the blind using RFID application", 5th International Colloquium on Signal Processing Its Applications 2009. CSPA 2009, pp. 247-249, 2009.
- 9. S. Ren, K. He, R. Girshick and J. Sun, "Faster R-CNN: Towards Real-Time Object Detection with Region Proposal Networks" in Advances in Neural Information Processing Systems 28, Curran Associates, Inc, pp. 91-99, 2015.
- 10. Shivani Bansal, Meenu Gupta and Amit Kumar Tyagi, "A necessary review on optical character recognition (ocr) system

 for
 vehicular applications",
 2020
 Second International Conference on Inventive

 Research in Computing Applications (ICIRCA), pp. 918-922, 2020.
 Second International Conference on Inventive
- L. Neumann and J. Matas, "Real-time scene text localization and recognition," Int. Conf. on Computer Vision and Pattern Recognition, pp. 3538-3545, 2012. [12]Ayushi Trivedi, Navya Pant, Pinal Shah, Simran Sonik, Supriya Agrawal and Navya Pant, Speech to text and text to speech recognition systems-A review Artificialintelligence(AI) sometimes
- 12. Manoj Kumar, Sanjay Singh, Urmila Pilania, Ankur Aggarwal, "High-Security Registration Plate Detection using OpenCV and Python", 2022 7th International Conference on Communication and Electronics Systems (ICCES), pp.835-839, 2022.
- 13. J. W. Hsieh, S. H. Yu, and Y. S. Chen. Morphology
- 14. based license plate detection from complex scenes. 16th International Conference on Pattern Recognition (ICPR'02), pp. 79-179, 2002.
- C. Guida, D. Comanducci and C. Colombo, "Automatic Bus Line Number Localization and Recognition on Mobile Phones—A Computer Vision Aid for the Visually Impaired", Image Analysis and Processing

 ICIAP 2011, pp. 323-332, 2011.
- C. M. Tsai and Z. M. Yeh, "Text detection in bus panel for visually impaired people "seeing" bus route number", 2013 InternationalConference on Machine Learning and Cybernetics (ICMLC), vol. 03, pp. 1234-1239, 2013.
- 17. H. Pan, C. Yi and Y. Tian, "A primary travelling assistant system of bus detection and recognition for visually impaired people", 2013 IEEE International Conference on Multimedia and Expo Workshops (ICMEW), pp. 1-6, 2013.

DISCOVER YOUR STRENGTHS WITH ACHIEVEAI'S NLP-POWERED PLATFORM

Malhar Gadade¹, Vinayak Jadhav², Om Dixi¹³, Pramod Mane⁴

SVERI's College of Engineering, Punyashlok Ahilyadevi Holkar Solapur Vidyapeeth., Gopalpur, Pandharpur-413304,

Maharashtra, India

Abstract: "AchieveAI: Uncover Your Strengths with NLP" addresses the escalating need for a transformative selfassessment platform in the digital age. This innovative web application allows users to upload academic and personal documents, utilizing advanced Natural Language Processing (NLP) algorithms for analysis. The paper outlines methodologies, technologies, and NLP algorithms, emphasizing the platform's significance in education, career development, and personal growth. "AchieveAI" generates personalized reports, evaluating strengths in key areas like communication skills and project ideation. By integrating NLP and machine learning, the project provides actionable insights, fostering personal and professional development with implications in education, human resources, and personal growth.

Keywords: AchieveAI, NLP, self-assessment, personal development, machine learning, education, career development, communication skills, project ideation.

I. INTRODUCTION

In the contemporary digital landscape, the exponential accumulation of personal and academic documents has brought to The fore the pressing need for a transformative self-assessment platform. This conference paper introduces "AchieveAI: Uncover Your Strengths with NLP," an innovative web application designed to address this need comprehensively. In an era marked by data abundance, this project redefines the boundaries of self-assessment and personal development. The significance of "AchieveAI" lies in its ability to harness the power of Natural Language Processing (NLP) and artificial intelligence (AI) to provide registered users with a dynamic, introspective journey. Through the seamless uploading and analysis of an array of documents, including academic transcripts, certificates, and achievement records, this platform unlocks a deeper understanding of one's strengths and potential areas for growth [5]. By focusing on criteria such as communication skills, stage daring, and project ideation strengths, "AchieveAI" generates personalized reports that offer actionable insights, illuminating users' unique capabilities. This paper provides a comprehensive overview of the project's methodologies, technologies, and NLP algorithms, underlining its transformative potential in education, career development, and personal growth. As we navigate the complex terrain of personal and professional development, "AchieveAI" emerges as a beacon of empowerment, fostering self-awareness and offering a roadmap to unlock untapped potential. This conference paper explores the multifaceted contributions and prospects of "AchieveAI" in the context of self- assessment, NLP, and personal growth [4].

RELATED WORK

The development of "AchieveAI: Uncover Your Strengths with NLP" draws upon a substantial body of prior research in several related domains:

NLP in Document Analysis

A considerable body of research has delved into the application of Natural Language Processing (NLP) for text analysis. Research in sentiment analysis, keyword extraction, and entity recognition has been pivotal in shaping the NLP components of our project.

Personal Assessment Platforms

Previous projects and platforms have explored self- assessment and personal development. E-portfolios, career assessment tools, and digital resume builders have paved the way for our endeavor, indicating a growing need for comprehensive self-evaluation tools [2].

Educational Technology and E-Learning

The field of educational technology has seen an evolution in e-learning platforms and digital education tools. These have set a precedent for integrating technology into the educational and self-assessment processes, aligning with the educational applications of our project.

Web Development and User Interface Design

The realm of web development and user interface design has offered insights into building user-friendly interfaces. Lessons learned from previous projects in web design and user experience have informed the design of our application.

Machine Learning in Personalization

Projects focused on recommendation systems and personalization have demonstrated the power of machine learning in tailoring user experiences. Our project's recommendation of areas for improvement leverages the principles of machine learning in personalized guidance[3].

II. METHODOLOGY

The development of "AchieveAI: Uncover Your Strengths with NLP" involved a structured and systematic approach that integrated web development, NLP algorithms, and machine learning techniques. This section outlines the key steps and methodologies utilized in building the project:

Frontend Development

User Interface Design: The project began with the creation of a user-friendly web interface using HTML, CSS, and JavaScript. The design aimed to provide a seamless and intuitive experience for users.

User Registration: User registration and authentication mechanisms were implemented to ensure secure access to the platform.

Document Upload: The system was designed to allow users to upload a diverse range of documents, from academic transcripts to certificates and awards.

User Dashboard: A user dashboard was created to enable users to manage their uploaded documents and access the generated reports.

Report Visualization: JavaScript libraries were utilized to visualize and present the data in the generated reports, ensuring a user-friendly and interactive display.

Backend Development

Database Setup: A secure database system (e.g., MySQL) was established to store user data, uploaded documents, and generated reports.

API Development: Backend APIs were developed using a framework such as Node.js with Express.js.

Integration with NLP Algorithms: NLP algorithms and libraries such as spaCy and NLTK were integrated into the backend to analyze document content and extract valuable insights.

Report Generation: A module was developed to create personalized reports based on the results of NLP analysis. User Management: Functionalities for managing user accounts, including profile updates and password resets, were integrated into the backend.

I. NLP Algorithms

Text Preprocessing: Document content underwent preprocessing to clean and format text for analysis.

Sentiment Analysis: Sentiment analysis was applied to assess the emotional tone expressed within the documents.

Keyword Extraction: Keywords and key phrases were extracted to highlight significant achievements and skills.

Named Entity Recognition (NER): NER algorithms identified and classified important entities within the text, such as names of people, organizations, and places.

Classification Algorithms: Classification algorithms categorized strengths into specific domains, such as communication skills, project ideation, and other criteria.

This structured methodology ensured the development of "AchieveAI" as a comprehensive and user-centric platform that merges web development, NLP capabilities, and machine learning techniques to offer actionable insights into individual strengths and areas for improvement [1].

III.DISCUSSION

The development and implementation of "AchieveAI: Uncover Your Strengths with NLP" represent a significant step towards empowering individuals with a transformative self-assessment tool. In this discussion, we delve into the multifaceted implications and the potential contributions of this innovative platform.

Firstly, "AchieveAI" is poised to bridge the gap between the digital age and personal development. In an era marked

by data abundance, the ability to comprehensively analyze and extract meaningful insights from diverse documents, spanning academic transcripts to achievement certificates, is of paramount significance. The integration of Natural Language Processing (NLP) algorithms, coupled with machine learning techniques, offers users a unique opportunity to reflect on their strengths and potential areas for growth.

From an educational perspective, "AchieveAI" holds the potential to reshape the way students approach their learning journeys. By providing personalized reports that pinpoint their communication skills, stage daring, and project ideation strengths, students can make more informed decisions about their educational and career paths. Similarly, for professionals, the platform offers a critical self-assessment mechanism to enhance career development. The discussion extends to the implications for the broader context of NLP and AI. The successful integration of NLP algorithms into a user-friendly web application signifies the practical applications of NLP in personal development, education, and professional growth.

IV.CONCLUSION

In an era defined by the digital transformation of our daily lives, "AchieveAI: Uncover Your Strengths with NLP" stands as a pioneering platform with the potential to reshape how individuals perceive their personal and academic achievements. This conference paper has presented the development, methodologies, and implications of "AchieveAI," and it is evident that this innovative web application brings forth a multitude of possibilities. The significance of this project lies not only in its sophisticated integration of Natural Language Processing (NLP) algorithms and AI but also in its user- centric design. "AchieveAI" provides a simple yet powerful means for individuals to gain insights into their strengths and areas for improvement. By analyzing a wide spectrum of documents, from academic transcripts to certificates and beyond, users can uncover facets of their achievements they might have overlooked. For educational institutions, "AchieveAI" offers an invaluable tool for career counseling and student development. For professionals, it provides a path to self-improvement and career advancement. In the context of NLP and AI, this project demonstrates the real-world applications of these technologies in enhancing personal development. The journey of "AchieveAI" is far from over; it is a testament to the ongoing evolution of technology in serving individuals' needs. As we stand at the intersection of digital innovation and personal growth, "AchieveAI" emerges as a beacon, illuminating the way forward for individuals looking to understand and maximize their unique potential. In doing so, it not only addresses the immediate problem of self-assessment but also signifies the promise of a more informed, empowered, and confident generation.

V.REFERENCES

- 1. Darling-Hammond, L., & Snyder, J. (2000). Authentic assessment of teaching in context. Teaching and Teacher Education, 16(5-6), 523-545. https://doi.org/ 10.1016/S0742-051X(00)00015-9.
- Kalia, V., Patel, A. K., Moriarity, A. K., & Canon, C. L. (2017). Personal branding: As a primer for radiology trainees and radiologists. Journal of the American College of Radiology, 14(7), 971-975. https://doi.org/ 10.1016/j.jacr.2017.03.017.
- McCaig, C., & Adnett, N. (2009). English universities, additional fee income and access agreements: Their impact on widening participation and fair access. British Journal of Educational Studies, 57(1), 18-36. https://doi.org/10.1111/j.1467-8527.2009. 00428.x.

- 4. Trochim, W. M., Rubio, D. M., Thomas, V. G. (2013). Evaluation guidelines for the clinical and translational science awards (CTSAs). Clinical and Translational Science, 6(4), 303-309.https://doi.org/10.1111/cts.12036.
- 5. Ullah, A., Liu, Y., Wang, Y., Gao, H., Wang, H., Zhang, J., & Li, G. (2022). E-Taste: Taste sensations and flavors based on tongue's electrical and thermal stimulation

5G AND BEYOND COMMUNICATION

.UjwalaVijayKumar Bhagat¹, Vinayak M. Sale²

¹Assistant Professor AI&DS Department, FTC, COER Sangola, Maharashtra (India).

²Ph. D. Scholar (AI & DS)*, DMIHER Sawangi (Wardha), Maharashtra(India).

Abstract:

This article provides an overview of the 5th Generation of cellular communication (5G) and beyond. It present the transmission techniques of current 5G communication and those expected of future development .study of non-orthogonal multiple access(NOMA) using the single carrier with frequency domain equalization block transmission technique, evidencing its value in terms of spectral efficiency. An introduction to 6th generation of cellular communication (6G) is also provided. The 5G mobile networks have revolution of communication but are not yet realized globally. The 5G promises to provide higher data rates, higher capacity, low latency, massive devices connectivity consists QoE (quality of experience), reduced cost as compared with 4G and take mobile communication to a new standard. The communication industry is rapidly advancing towards 5G and beyond 5G (B5G) wireless technologies in order to fulfill the ever –growth needs for higher data rates and improved quality –of –service (QoS).

Keywords: 5G mobile communication, computer architecture, MIMO communication, antenna array, microprocessor, NOMA, IoT, millimeter wave.

I. INTRODUCTION

Artificial intelligence for 5G and beyond 5G implementations, algorithm and optimizations. Wireless communications and networks for 5G and beyond enables to access various multimedia services such as stored multimedia (e.g. Video on demand), live streaming (e.g. Internet live sport networks, internet radio station) and real-time interactive streaming (e.g. Online games, video conference ,e-education). Consequently wireless communications technology has rapidly gain a role become an important aspect of life. There is a strong, credible body of evidence, suggesting that mobile network operators are facing many formidable task.5G cellular network which achieve 1000 times data rates,50 billion devices connected to internet. Data rates are projected to increase by factor of ten every year. IoT connected trillion of devices across the globe. Wireless communication and network of 5G and beyond.

II. RELATED WORK

Today 5G support human to human (H2H) interaction data for multimedia content and signal exchange. Providing full function for IoT when combining device to device (D2D) communication, collecting of sensing data, data analyzing during an interaction.

QoS / QoE mechanism for wireless communication and networking 5G wireless heterogeneous network design and optimizing 5G cognitive network and IoT, experimental result prototype and test beds for 5G wireless communication provide security and privacy. NOMA Full duplex, massive MIMO, and green 5G multimedia upcoming 5G wireless network joint design shown in Fig no 1. The 5G is based on OFDM (Orthogonal frequency-division multiplexing), a method of modulating a digital signal across several different channels to reduce interference. 5G uses wider bandwidth technologies such as sub -6 GHz and5G NR air interface alongside OFDM principles. 5G also uses wider bandwidth technologies such as sub-6 GHz and mm Wave.



Fig no 1: Artificial intelligence enables wireless networking

The 5G protocol is designed more efficient, reliable and secure than previous generation of wireless technology such as 4G LTE .it is based on the latest advancement in technology including artificial intelligence, edge computing and internet of things (IoT), increase friendliness and higher data rates are achieved when 5g-tcp is employed in predefined simulation environment protocol defined to be used in future 5G data networks engaged MAC Layer. It defined several response functions for variety of speed up to 400 Gbps.

The 5G Network Architecture: the architecture of the 5G network is based on three main components: the User Equipment (UE), the radio access network

(RAN) and core network (CN). The UE is device that connects to 5G networks.

III. BEYOND 5G COMMUNICATIONS

As the commercial deployment of 5G of cellular networks (5G) is well underway in many countries the world. Academia and industrial research organization turn their attention to comes next develop a new cellular communication standard. Foundation for 6G and moving from 4G to 5G changes to physical layer support a set of diverse application with different throughputs and latency and reliability requirement thanks to flexible OFDM by supporting large bandwidth and antenna array at base station that is massive MIMO.

The 6G will be driven of mixed past trends. 5G services have been commercially deployed since 2019 but realize the benefits of ultra-high speed and ultra-lowlatency by technology in carrier aggregation which boosts communication throughput utilizing multiple frequency bands simultaneously and TDK's ultra-compact DC-DC converters are helping to bring it reality by powering the FPGA's and next generation chipsets in networks.



Fig 2: 5G/B5G Intelligent network.

Challenges facing carrier aggregation –



Fig 3: Schematic diagram of the intelligent throughput optimization approach.

IV. A key technology in beyond 5G

A 5th generation mobile communication services rolled out globally around 2020 people benefits of 5g to improve our lives and society more substantially numerous challenges remain addressed, including the build out of base system based on 5G and featuring even advanced capability referred to beyond 5G around 2023 .reliability against security threats and disaster autonomy that allows working together without human.

V. Overview of Beyond 5G

This refers to mobile communication system that follows 5G features capability that will drive new value creation such as ultra-low power consumption high speed. 6G smart network and services industry association -5G and beyond 5G ecosystem business model .application of block chain in 5G and beyond network ,taxonomy ,field –trials ,challenges and opportunities.

VI. 5G Specifications

- Up to 10 Gbps data rate -10 to 100x speed improvement over 4G and 4.5G network.
- 1-millisecond latency
- 1000x bandwidth per unit area
- Up to 100x number of connected devices per unit area (compact with 4G LTE)
- 99.999% availability
- 100% coverage
- Up to 10 year battery life for low-power IoT device.
- 5G IS 10 *100 FATER THAN 4G

VII. WHAT MAKES 5G faster

According to communication the shorter the frequency the larger the bandwidth. Using shorter frequency (millimeter wave between 30 GHz and 300GHz) for 5G network is 5G faster high speed 5G spectrum provides the expected boost in speed and capacity, low latency and quality.

VIII. HOW FASTER IS 5G

• Use cases associated with low latency are: V2X (vehicle-to-everything) communication: V2V (vehicle-to-vehicle) V2I (vehicle-to-infrastructure) autonomous, connected cars, immersive virtual reality gaming (5Gwill bring VR to masses), remote surgical operations, simultaneous translating.

- Fixed wireless access(from 2018-2019 onwards)
- Enhanced mobile broadband with 4G fallback (from 2019-2020-2021)
- Massive M2M /IoT (from 2021-2022)
- Ultra-low -latency IoT critical communication (FROM 2024-2025).

IX. 5G Rollouts

Over 4 years since its launch 5G technology has significantly matured, achieving broader and faster coverage.as of June 2023 5G has reached 1.1.billion subscriptions globally, with an addition of 125 million in the first quarter 240 services provides have

established 5G network and 35 have launched 5G standalone network.as of early 2023 users have over 700 5G smartphone models to choose from GSMA intelligence predict rapid 5G expansion with 2 billion by 2025 outpacing 3G and 4G rollouts. By 2028, 5G is expected to encompass 4.6 billion subscriptions, overtaking half of all mobile subscriptions

X. 5G in India

India Telco's started their 5G rollouts in October 2022 the number of users now exceeds 100 million according to the main operators Bharati Airtel and Reliance Jio driving the push its one –sixth of the 600 million phone users according to the GSMA median 5G download speed are 25 faster than 4G in the country.

XI. What's the relation between 5G and satellites?

A new generation of satellites will bring on board 5G capabilities to ensure full 5G coverage of the earth and terrestrial mobile 5G networks. With space –based system offer higher accessibility, reliability resilience and broadcasting and multibroadcasting capability.5G support a wide variety of application (e.g. Agriculture logistic public safety) so everybody in the world can be connected along with space the effort of 3Gpp to develop solution embracing the vision of a single global – based network fully integrated with mobile operators G network. With the increased impact of IoT services the need for 5g security privacy and trust will be as strong as for 4g if not stronger.

XII. How will 5G network use cases change the world: its set to change our world many ways?

Healthcare: imagine having a doctor appointment over a crystal –clear video call or surgeons operating remotely with robotics precision -5G makes it possible with its fast and stable connection.

Smart Cities: city life will become more efficient as 5G helps manage traffic hanks to the quick reliable data 5G can used.

Driverless Car: these vehicles will be able to communicate with each other to avoid accident and ease traffic thanks to quick and reliable data 5G can send.

Manufacturing: factories will become more automated and efficient.

Education: student should take virtual trips or get hands on with complex subject through arguments reality all facility by 5G speedy connections.

Energy: managing power grid more helping

Emergency services: faster communication

Conclusion: Although the network key performance indicators of the 5th G mobile network have been optimized to support drone use cases for both high data rates and low latency application ,future aerial transport system will require stricter network key performance indicators to support the expected massive.5G network have revolutionized communication has improve 5G mobile devices provide higher data rates, higher capacity, low latency, massive device connectivity consistent QoE (quality of experience) reduce cost compare with 4G.

IV.REFERENCES

1. Duong T.Q. S. Editorial: wireless communication

2. and network for 5G and beyond Mobile Networking 24,443-446.

3.5G and beyond published in IEEE Nov. 2018.

- 4. S. Shinde, A. Nikam and S. Joshi, "An Overview of 5G Technology" (IRJET) Vol. 3 No 4 April 2016
- 5. T. Dateki, Hiroyuki Seki and M. Minowa, "from LTE Advanced to 5G mobile access system in progress", FUJITISU Sci J. Vol, No 2, April 2016.
- 6.S. Suyama, T. Okuyama, Y .Inoue and Y. KISHIYAMA. "Technical Report Special Articles on 5G Technology towards 2020 deployment NTT DOCOMO technical

A DATA-DRIVEN APPROACH USING PLACEMENT ASSISTANCE SYSTEMS

Shraddha B. Gavhane¹, Avinash K. Parkhe², Dr. S. P. Pawar³ ¹shraddhagavhane98@gmail.com ²akparkhe@coe.sveri.ac.in ³sppawar@coe.sveri.ac.in

 ¹M.Tech. Students, Department of Computer Science & Engineering, SVERI's College of Engineering, Pandharpur, India
 ²Assistant Professor, Department of Mechanical Engineering, SVERI's College of Engineering, Pandharpur, India
 ³Associate Professor, Department of Computer Science & Engineering, SVERI's College of Engineering, Pandharpur, India

Abstract - Information systems based on computers are made to enhance current systems. Since automated systems are becoming more and more advantageous, many manual processes are now automated. Colleges and other educational facilities needed their manual systems to work on computers because automated systems are now required. Placement automation is one such system that is crucial for recruiting on campuses. The goal of this project is to create a web application for the college's placement department. The college's placement officer can use this system as an application to manage the student data related to placement. Students who log in can upload a resume and their personal data. As a result, all of the data will be stored about the students, including their educational background, personal information, marks from their universities, and resume uploads. Students can view the training schedule and upcoming campus events after logging in. The system makes it easier for the college's placement officer to access student data. The chosen student's information, including their offer letter and company, can then be listed in another table by him. Through email, the placement officer can inform the students about the training schedule and campus. Colleges can use this system as an application to manage student records based on placement.

Keywords - Placement, Assistance System, Colleges, Diagrams.

I. INTRODUCTION

The main issue with the current system is that data filtering is done manually, takes a lot of time, and is subject to human error. The fact that the entire system must be maintained by hand makes it difficult and time-consuming to store, maintain, and retrieve information, which is one of the many problems with the current setup. The records were never kept in a methodical sequence. Information systems based on computers are made to enhance current systems. Since automated systems are becoming more and more advantageous, many manual processes are now automated. Colleges and other educational facilities needed their manual systems to work on computers because automated systems are now required. Placement automation is one such system that is crucial for recruiting on campuses. The purpose of this paper is to develop a web application for the college's Placement Department. The college's placement officer can use this system as an application to manage the student data related to placement. [1,2,3]

Students who log in can upload a resume and their personal data. As a result, all of the data will be stored about the students, including their educational background, personal information, marks from their universities, and resume uploads. Students can view the training schedule and upcoming campus events after logging in.

In terms of placement, the system facilitates access to student data for the college Placement Officer. Then he can create a table with the offer letter and company details of the chosen students. By emailing the students, the placement officer can inform them about the training schedule and campus. An application for managing student records based on placement can be used with this system by colleges.

II. LITERATURE REVIEW

Many researches have done study on placement management system as below:

Nicholine et al. has presented system allows the students in the college to view a list of companies who have posted for vacancy. The admin has the complete control over the system and can modify/delete anything that doesn't meet the standards of the college management. The system maintains the database of the students as well as the company details efficiently and displays the data to the respective ends.

Banu et al. has studied system can be used by institutions to manage student information and placement details. Students should be able to access their accounts and upload their work. This project helps students to keep track of their personal and academic information. It saves time by decreasing manual labor and reducing the amount of paper used.

Dhopavkar et al. has presented the details about development of a "Training & Placement Portal" system. The objective to develop such system is to automate the Training and Placement management system. This developed system helps students and people working at the Training and Placement cell to easily carry out regular tasks with high precision and in less time. Thai-Nghehs et al. presented One of the educational data mining tasks is predicting student performance (PSP), where the goal is to determine the students' level of knowledge and their ability to complete the tasks (or exercises) correctly. Jose et al. has presented creating an application for the college's placement office. The system is an application that, with the right login enabled, can be accessed and used efficiently across the entire organization. It can also be used as an application by the college's placement officers to manage student placement data, which cuts down on manual labor and paper consumption. Additionally, the system allows users to view the student's academic and personal data

NEED OF PLACEMENT ASSISTANCE SYSTEMS

Since the entire system had to be maintained by hand, the process of storing, maintaining, and retrieving the information was very time-consuming and laborious. The old manual system had a number of disadvantages. The records were never kept in a methodical sequence. It used to be very difficult to connect any given transaction with a specific context. Report generation would never exist; one had to search through various registers and documents in order to find any information. When entering and retrieving records, time would always be spent needlessly. Another issue was how challenging it was to identify mistakes made when entering the records. It was very difficult to update the records once they were entered. [4,5,6]

III. PROPOSED SYSTEM

This paper is aimed at creating a web application for the college's Placement Division. All of the schedule and events are accessible to everyone as soon as you launch the front end of this web application. There are three primary categories of users: students, college placement officers (administrators), and department staff. Being the master user, the administrator has more priorities than any other user. The duties involved in the role of Placement Officer include viewing all student information, creating and sending notifications, adding department personnel, and viewing specific student information. Placement Officer after login will be able to see admin panel where he will have number of options to go with. He can view details of all registered students of respective departments. He will mark the students who are selected in recent campus drive. He will also be able to view the non-selected students list. Placement Officer will have a right to add one department staff called sub admin of each department with their Name, Email, contact number and department. This sub admin will have authentication to view all registered students of his department. He can also view the notification sent by Placement Officer. [7,8]

Student registration will be the very primary work to do by the students in this system. Every student will register independently, and each student will have a unique profile. There is need to register their all details like personal details, academic details, Email, contact number etc. Only the registered students can log in to student panel. Students when logged in will go to their student panel. Students will have the options of View new notification; view their own profile; to download their CV uploaded at the time of registration and editing previous details. Students can edit their Resume and update them constantly. Students can look up and view company and vacancy details with a great deal of flexibility. Students have access to online resources that are relevant to them. The current student will benefit from this since other students will be sharing their ideas. Additionally, students will use the system to read important announcements in the newly created notification field, get information on assessments, and view the system-recorded results of assessments. Every user submitted a unique authorization. Furthermore, the Placement Officer has the ability to search for eligible students based on company criteria and produce a list of them. The system has the ability to automatically email the qualified student. All students for whom the Placement Officer has the necessary authorizations can have their information gathered. In the cutthroat world of today, this aids the organization in winning the war.



Fig. 1 Block diagram for Online Placement System

The administration module, student module, and department staff module are the three modules that make up the system. Every login module has an identical login page with a user ID and password field; the user must log in to the system by entering a value in those fields shown by above figure 1. [2]

1) **Student:** The notification appears on the homepage of the student module after the student logs in. Adding all the information allows them to manage the profile. The resources uploaded by users are available for download.

2) **Placement Officer:** The Placement Officer has the ability to create company schedules and events and upload them to the front page and every student's login. They have the option to get in touch with the business and request detailed information.

3) Department Employees: A single authorized employee may sign up for an account on the system. Any notice that the placement officer sends out is visible to the staff. Workers' logins allow them to view the schedule or event. An enrolled student's status is visible to staff.

A. USECASE DIAGRAM:

Use case diagrams show the relationships between the actors and use cases. The system or subsystem of an application is modeled using the diagram. Specific system functionality is captured in a single use case diagram. As a result, several use case diagrams are used to model the complete system.



Fig. 2 Usecase diagram for Online Placement System

B. SEQUENCE DIAGRAM:

An interaction's participants and the messages that flow between them are depicted in a sequence diagram. A sequence diagram illustrates how an actor and system interact to carry out all or a portion of a use case. Figure. The Admin Login Flow Chart's Sequence Diagram 3 essentially depicts a collection of classes, interfaces, partnerships, and the connections that exist between these classes and interfaces. Their primary purpose is to model object-oriented systems. In class models, classes, associations, and generalizations are the most crucial concepts. Groups of related objects are described by classes. A set of related connections between objects is referred to as an association.

fernin :		System	Databa
	1 Enter Username	3.Gheck	
	Password 4. Invalid	admin	
4	A Login Successfuly 6 Update	5. Give access	_
¢	SubAdmin 8 Sub add	B. Create Id	
<	10 Piter Stolent 12 Send	+ 11.Update	
	Notification	+ 13.Update	
¢	12 Mana Details	13. View	
-	14 Exit		14. 444

Fig. 3 Sequence diagram for Admin Login



Fig. 4 Sequence diagram for Students Login

C. SOFTWARE DESIGN FLOW CHART:

A flow chart essentially displays a collection of classes, interfaces, teamwork, and the connections that exist between these classes and interfaces. They are primarily used in object-oriented system modeling. Classes, associations, and generalizations are the key concepts in class models. A class is a collection of related objects. Furthermore, association denotes a collection of related relationships between objects. [1]



Fig. 5 Flow Chart for Software Design

D. SYSTEM ARCHITECTURE/ALGORITHM



Fig. 6 System Architecture/Algorithm [1]

SOFTWARE IMPLEMENTATION

In 'Online Placement System' everything is made computerized where Placement Officer can send all campus and training schedules through the notification option provided in system to students via Emails. The Training and Placement officer can also add the departmental staff who will be the placement coordinator of the department. All the students need to register their all details like personal details and educational details at the time of registration. Only registered students are allowed to login. After login students will able to see new notification if available, download their CV, edit details and also can change login credentials. Hence the system is less labored and paperless.

II. RESULTS AND DISCUSSION

Student can create their profile by filling the form with valid information, once registered admin need to accept or reject the candidate profile by checking whether it is authorized user.

0.01.00		(100	 Real Projection	
	PERSONAL INFOR	0647328		
	-			
	-			

Fig. 7 Students profile dashboard

The role of the admin would be to oversee and manage the system's operations. The key responsibilities of an admin in a placement management system are data

management, system configuration, user management, analytics and reporting, security and compliance etc.



Fig. 8 Admin dashboard

After filling the data by the students the final results will be received in form of excel file shown below:



ig. 9 Students data output

IV. CONCLUSION

The following conclusions are drawn through this study:

- 1. Any modifications to the current Placement system take time because it is primarily done by hand, prone to error, and complicated.
- 2. One of the main issues is the lack of a notification system for students other than the notice board, which makes it difficult to find, sort, and update student data.
- 3. The online enrollment process for every student is automated by the proposed system. It is possible for the administrator to view student data, create a student list based on company standards, give students access to company information, and quickly generate results by sorting students. In
- 4. During the recruitment process, the system's easy accessibility and portal functionality will aid in managing the allocation process.
- 5. We can expect there to be a high demand for these portals in the near future due to the growing need for digitalization in all facets of daily life and the comfort it will bring to everyone.

V. REFERENCES

- 1. Punitha Nicholine J, Lakshmi Priya B, Ilakkiya S, Kartheeswari M, Kethrin Malar D., Placement Management System, International Journal of Progressive Research in Engineering Management and Science (JJPREMS), Vol. 03, Issue 04, April 2023, pp : 120-124.
- Alfiya Banu, Dr. Manju Bargavi S. K., A Research on Placement Management System, Journal For Research in Applied Science and Engineering Technology, <u>https://doi.org/10.22214/ijraset.2022.41657</u>.
- 3. Dr. Gauri Dhopavkar, Toshit Kale, Rajat Gaikwad, Saurabh Chavan, Gaurang Kumar, 6Himanshu Shendare, Harsh Akre, An Integrated Web Application for Training and Placement, Mathematical Statistician and Engineering Applications, Vol. 71 No. 4 (2022).
- 4. Ajin P Jose, Hanshida P, Shejina N.M, Placement Management System for Campus Recruitment, IJARIIE-ISSN (O)-2395-4396, Vol-8 Issue-6 2022.
- 5. Nguyen Thai-Nghe, Tomas Horvath, Lars Schmidt-Thieme 2011, "Factorization models for forecasting student performance", Proceedings of the 4th International Conference on Educational Data Mining, Eindhoven, The Netherlands, July 6-8, 2011.
- 6. Rajnish Tripathii, Raghvendra Singh, Ms. Jaweria Usmani," Campus Recruitment and Placement System", In International Conference on Recent Innovations in Science and Engineering (ICRISE-18), April, 2018.
- 7. Mythili M, Aishwarya R, Shenbagam P, Sandhiya C," E Placement Management", International Journal of Pure and Applied Mathematics (IJPAM), Volume 119 No. 10 2018,1823-1834.
- Molina, M. M., Luna, J. M., Romero, C., & Ventura, S., 2012, "Meta-learning approach for automatic parameter tuning: A case study with educational datasets" In Proceedings of the 5th International Conference on Educational Data Mining, of EDM 2012, pages 180-183, Chania, Greece, 2012.

ELECTRICAL ENGINEERING

Adaptive Load Frequency Controller using Fuzzy Logic Technique for Parallel Resonant SIC Inverter in Surface Hardening Application

¹Vijay Patil, ²Anwar Mulla

¹Department of Electrical Engineering, Shivaji University, Kolhapur, Maharashtra, India ²Department of Electrical Engineering, Daulatrao Aher College of Engineering, Karad, Maharashtra, India

> ¹profmpvijaypatil2013@gmail.com ²ammaitp@rediffmail.com

Abstract— For converting DC electrical power into AC inverters are preferred choice. They are constructed in various forms, size and types according to the requirements of the given application. The parallel resonant inverter with key element of resonant tank circuit which finds applications in various domains, including power supplies, renewable energy systems, and induction heating. The resonant tank circuit is used to achieve soft switching of switching devices which results in reduced switching losses and improved efficiency. However, parallel resonant inverters are susceptible to load transients which causes voltage fluctuations. Hence, this paper focuses on frequency control of parallel resonant inverter with fuzzy logic control. The principle of fuzzy logic control is used to frequencyregulation which improves the dynamic response of the system. The details of MATLAB simulation and results are presented which demonstrates that the fuzzy logic control is a viablesolution for enhancing frequency control in parallel resonant inverters, enabling them to meet the demanding requirements of modern power conversion systems.

Keywords: parallel resonant inverter, frequency control, fuzzy logic, induction heating

I.INTRODUCTION

The choice between a current source inverter (CSI) and a voltage source inverter (VSI) depends on the specific application and system requirements. Both types of inverters have their advantages and disadvantages. For induction heating applications, CSI is preferred over VSI on account of better current control compared to VSI, robustness to load variations, fault tolerance and low dv/dt stress on switching devices.

A parallel resonant inverter is a type of power electronic device that converts DC into AC at a certain frequency. It is commonly used in applications like power supplies, renewable energy systems, and induction heating. A parallel resonant inverter consists of a DC power supply, a switch or switching matrix, a parallel resonant circuit, and a load. The switch is controlled to turn on and off at the resonant frequency of the parallel resonant circuit. This causes the current in the circuit to flow in a sinusoidal manner, with zero voltage at the switch terminals. The parallel resonant circuit consists of an inductor and a capacitor. The inductor stores energy when the switch is turned on, and the capacitor releases this energy when the switch is turned off. The combination of the inductor and capacitor creates a resonant circuit, which has a single frequency at which it resonates. The load is connected to the parallel resonant circuit. The current from the parallel resonant circuit flows through the load, providing power to the load. The circuit diagram of a parallel resonant inverter is shown in figure 1.

In parallel resonant inverter, the switching frequency of the inverter is often determined by the LC tank circuit in the load side. The switching pulses are to be provided to the semiconductor switches i.e. MOSFETs such that the inverter can supply constant power to the load [1]. The parallel resonant circuit is used to improve the efficiency of the inverter by reducing the switching losses. The resonant circuit also helps to improve the electromagnetic compatibility of the system by reducing the emissions of the inverter [3]. The switching frequency of current fed parallel resonant inverter can be controlled with phase locked loop (PLL). The PLL compares the phase of the output voltage to the phase of the output current, and generates a control signal that is used to adjust the switching frequency. This allows the output power of the inverter to be controlled accurately, even in the presence of changes in the circuit parameters [2]. The laboratory setup [4] of parallel resonant inverter for induction heating consists of a full bridge series resonant inverter, a power supply, a load, and a controller. It is used for heating the metal work piece. If the frequency of the inverter is controlled, the temperature of the metal work piece can be controlled accordingly. A low-cost DSP based PI controller [5] can regulate the output voltage of the inverter accurately with high efficiency. The pulse density modulation [6] based controller always maintains soft switching condition irrespective of load variation and transfers maximum power to the induction heating load but It requires a high-speed DSP processor to generate and decode the pulse stream and can be susceptible to interference from other signals. The PLL based control [7] allows output power control by adjusting the on time i.e. duty cycle. For induction heating applications, to achieve precise temperature, frequency control is desired. The inverter efficiency is maximum at resonance and decreases significantly off resonance conditions [8]. Hence precise

control of frequency is demanded for parallel resonant inverters. This paper describes fuzzy logic controlled current source parallel resonant inverter for induction heating applications. MATLAB simulation results are presented to demonstrate the tracking of resonant frequency.



Fig. 1: circuit diagram of parallel resonant inverter



Fig. 2: MATLAB model of current source fed parallel resonant inverter.



Fig. 3: Fuzzy logic controller-based control scheme.



Fig. 4: Fuzzy logic controller.



Fig. 5: Fuzzy membership function for error and change in error.







Fig. 10: Frequency change regulated by fuzzy logic controller at t= 0.3122s

I. PARALLEL RESONANT INVERTER

Figure 2 shows the MATLAB model of current source parallel resonant inverter. The system consist of a diode bridge rectifier to convert AC from grid into DC followed by a DC to DC converter. The DC to DC converter adjusts the DC voltage to the required value. A large inductor the input side of the H-bridge is used to converter the voltage source into current source and hence making the current source fed parallel resonant inverter. The parallel resonance is obtained in the LC tank circuit. The fuzzy logic based controller is shown in figure 3. The X-OR gate acts asphase detector. It takes inputs from zero crossing detectors of voltage and current and produced a pulse whose pulse width is proportional to the phase difference between voltage and current. This phase difference is further give to a fuzzy logic controller as shown in figure 4. It processes this pulse width as error signal. Taking into account the fuzzy rule table of figure 5, the controller generates switching pulses for gate terminals of the MOSFET in the inverter bridge. The changes in frequency are tracked by the fuzzy logic controller based frequency controller and the system responds in single cycle. The change in frequency takes place whenever there is change in load. The efficiency of inverter with fuzzy logic based frequency control is observed to be 89.4%.

Source	Voltage = $26V$, 3 phase
	Frequency = 50Hz
Rectifier	Diode bridge rectifier with capacitor falter
	$C1 = 2200 \mu F$
DC – DC Converter	Buck Converter with capacitor filter
	$C2 = 1000 \ \mu F$
Inductor for current source	20mH

Tank circuit and load

Table 1: Specifications of the parallel resonant inver
--

III.CONCLUSION

 $R = 5\Omega, L = 63.33 \text{mH}, C = 1 \ \mu\text{F}$

Frequency control of a current source fed parallel resonant inverter with fuzzy logic control is discussed in this paper. The fuzzy logic controller for frequency control is effective in the regulation of switching frequency. The controller responds to change in load and regulates the frequency of parallel resonant inverter in single cycle. The protocol implementation of the proposed controller can be made with DSP controller.

IV.REFERENCES

- 1. J. Mućko, "Parallel resonant inverter with auxiliary AC/DC converter used for induction heating," 2011 7th International Conference- Workshop Compatibility and Power Electronics (CPE), Tallinn, Estonia, 2011, pp. 415-419, doi: 10.1109/CPE.2011.5942271.
- N. Ponwiangkum and S. Kittiratsatcha, "Switching Frequency Control based on Phase-locked Loop for a 2. Current-fed Parallel Resonant Inverter," 2007 Power Conversion Conference - Nagoya, Nagoya, Japan, 2007, pp. 157-161, doi: 10.1109/PCCON.2007.372962.
- H.javadi, A. Shoulaie, "Analysis of H-bridge Current Source Parallel Resonant Inverter for Induction 3. Heating", Iranian Conference on Electrical Engineering, Iran university of science and technology, 2009
- Oncu, S., Unal, K., & Tuncer, U., "Laboratory setup for teaching resonant converters and induction heating". 4. International Journal of Engineering Science and Technology. Elsevier. 2021. doi:10.1016/j.jestch.2021.05.016
- G. Martin-Segura, J. Lopez-Mestre, H. Rivas-Guerrero, M. Teixido- Casas and D. Montesinos-Miracle, "A 5. low cost DSP-based control method for high frequency parallel resonant current-fed inverters," 2009 13th European Conference on Power Electronics and Applications, Barcelona, Spain, 2009, pp. 1-7.
- 6. Kumar, P.K. Sadhu, R. Raman, J. Singh, Design analysis of full- bridge parallel resonant inverter for induction heating application using pulse density modulation technique, 2018 Int, Conf. Power Energy, Environ. Intell. Control. PEEIC 2018 (2019) 398-402, https://doi.org/10.1109/PEEIC.2018.8665571.
- 7. J.-W. Kim M. Lee J.-S. Lai A new control method for series resonant inverter with inherently phase-locked coil current with induction cookware applications, in, IEEE Appl. Power Electron. Conf. Expo IEEE 2018 2018 3517 3522 10.1109/APEC.2018.8341611
- M. Matsuo, T. Suetsugu, S. Mori and I. Sasase, "Class DE current- source parallel resonant inverter," in 8.

IEEE Transactions on Industrial Electronics, vol. 46, no. 2, pp. 242-248, April 1999, doi: 10.1109/41.753762.

A Fuzzy based Approach for Battery Controller for Microgrid

Mitkari Mohit M.¹, Bawage Ankita S.², Mantri Lata R.³ Pate Pawan V.⁴

¹Department of Electrical Engineering, M.S. Bidve Engineering College Latur, Maharashtra, India ²Department of Electrical Engineering, V.D.F. School of Engineering and Technology Latur, Maharashtra India. 3Department of Electrical Engineering, M.S. Bidve Engineering College Latur, Maharashtra, India ⁴Department of Electrical Engineering, M.S. Bidve Engineering College Latur, Maharashtra, India

> ¹mohitmitkari25109 4@gmail.com ²ankita.sb2225@gmail.com ³latatawani@gmail.com ⁴pawanpate1913@gmail.com

Abstract— This paper presents a fuzzy-based approach for designing a charging-discharging controller for lithium-ion batteries in microgrid applications. The goal is to enhance the efficiency and performance of battery systems within microgrids. The proposed controller utilizes fuzzy logic techniques to handle uncertainties and imprecise information, providing robust and adaptive control in real-time scenarios. The controller's fuzzy rules consider factors such as battery state of charge, load demand, and renewable energy availability to determine optimal charging and discharging strategies. Simulation results demonstrate the effectiveness of the fuzzy-based controller in improving battery utilization, ensuring stable microgrid operation, and enhancing overall system performance. This research contributes to the advancement of battery control strategies in microgrids, promoting more efficient and sustainable energy management systems.

Keywords— fuzzy logic, charging-discharging controller, lithium-ion battery, microgrid applications, efficiency, performance, robust control, adaptive control, battery state of charge, load demand, renewable energy, optimal strategies, simulation results, system performance, energy management.

I. INTRODUCTION

Lithium-ion batteries have become increasingly popular forenergy storage in microgrid applications due to their high energy density, long cycle life, and fast charging capabilities. Effective control of these batteries is essential to ensure efficient utilization and reliable operation within microgrid systems. However, traditional control methods like proportional-integral-derivative (PID) control often struggle tohandle the uncertainties and dynamic nature of microgrid environments [1-3]. To address these challenges, fuzzy logic control has emerged as a promising alternative due to its abilityto handle imprecise and uncertain information.

The main objective of this research is to develop a charging-discharging controller for lithium-ion batteries in microgrid applications using a fuzzy-based approach. Fuzzy logic control offers advantages such as adaptability, robustness, and flexibility, making it well-suited for dynamic and complex microgrid scenarios. By incorporating fuzzy logic techniques, the controller can effectively handle uncertainties arising from variations in renewable energy availability, fluctuating load demand, and battery state of charge (SOC) fluctuations. [3-6]

The proposed fuzzy-based controller aims to optimize the charging and discharging strategies of the lithium-ion battery, considering multiple factors that influence battery performance and microgrid operation. These factors include the current SOC of the battery, the demand for electrical energy from the microgrid, and the availability of renewable energy sources. The fuzzy rules embedded in the controller are carefully designed to adaptively adjust the battery charging and discharging rates based on real-time system conditions [7-8].

Simulation studies are conducted to evaluate theperformance of the proposed fuzzy-based controller compared to traditional control methods. The simulation results demonstrate the superiority of the fuzzy-based approach in terms of battery utilization, microgrid stability, and overall system performance. The findings of this research contribute to the advancement of battery control strategies in microgrid applications, aiming to enhance the efficiency and sustainability of energy management systems [9].

In the subsequent sections, the methodology used for developing the fuzzy-based charging-discharging controller will be discussed in detail. The design considerations, formulation of fuzzy rules, and control algorithm will be

elaborated upon. Following that, the simulation setup and experimental results will be presented and analysed. Finally, the paper will conclude with a summary of the achieved outcomes, limitations, and potential avenues for future researchin this field [10-12].

II. LITERATURE REVIEW

- The development of efficient charging-discharging controllers for lithium-ion batteries in microgrid applications has been the subject of extensive research in recent years. Numerous studies have investigated various control strategies and optimization techniques to improve battery performance, enhance microgrid stability, and achieve optimal energy management [13].
- One common approach in battery control is the use of PID control, which relies on proportional, integral, and derivative terms to regulate the charging and discharging processes. PID control has been widely applied in microgrid systems due to its simplicity and ease of implementation. However, it may struggle to adapt to changing operating conditions and handle uncertainties effectively [14].
- To overcome the limitations of PID control, researchers have turned to fuzzy logic control as an alternative. Fuzzy logic enables the representation of imprecise and uncertain information using linguistic variables and fuzzy rules. This allows for more robust and adaptive control in dynamic microgrid environments [15].
- Several studies have employed fuzzy logic techniques to develop charging-discharging controllers for lithiumion batteries. For instance, Zhang et al. (2018) proposed a fuzzy-based controller that considered battery SOC, load demand, and renewable energy availability to optimize the battery operation in a microgrid. The results showed improved battery utilization and enhanced microgrid stability compared to traditional control methods.
- In another study by Li et al. (2019), a fuzzy logic-based energy management system was developed for a hybrid microgrid with renewable energy sources and energy storage. The fuzzy controller effectively regulated the battery charging and discharging rates based on real-time system conditions, leading to optimal battery utilization and improved energy management efficiency [16].
- Additionally, machine learning techniques, such as fuzzy neural networks and adaptive fuzzy systems, have been explored for battery control in microgrids. These approaches combine the advantages of fuzzy logic and neural networks to enhance control accuracy and adaptability. For example, Guo et al. (2020) proposed a fuzzy neural network-based controller for lithium-ion batteries in microgrids, achieving accurate and adaptive control under various operating conditions [17].
- Furthermore, optimization algorithms, including genetic algorithms and particle swarm optimization, have been integrated with fuzzy control to optimize battery charging and discharging strategies. These algorithms enable the search for optimal control parameters, considering multiple objectives such as battery lifespan, energy efficiency, and microgrid stability.
- Overall, the literature highlights the effectiveness of fuzzy- based charging-discharging controllers for lithiumion batteries in microgrid applications. Fuzzy logic techniques offer robustness, adaptability, and the ability to handle uncertainties, leading to improved battery performance and microgrid operation. However, further research is still required to investigate advanced control strategies, optimization techniques, and integration with emerging technologies such as artificial intelligence and machine learning, to further enhance the capabilities of battery control in microgrids.

III. METHODOLOGY AND ALGORITHM

Figure 2 depicts a typical configuration of a PV panel, consisting of PV cell strings connected in series and divided into three sections by bypass diodes. These bypass diodes serve the purpose of preventing the occurrence of hot spots and protecting the PV module from potential damage. To ensure optimal performance, the PV module is combined with a maximum power point tracking (MPPT) converter.



Fig 1 Fuzzy based MPPT of Photovoltaic power system.[6]



Fig 2 V-I Characteristics (Source: CPES) [7]

This converter continuously adjusts the operating point of the PV module to maintain it at the maximum power point. Consequently, the combination of the MPPT converter and the PV module acts as a constant power source, where the power output is determined by the peak power of the PV module. The output side of the converter accommodates a relatively wide range of voltage and current, allowing for series or parallel connection with other converters. Essentially, the distributed MPPT converter expands the range of voltage and current values, transforming the single voltage/current point of the PV panel into a broader range, as shown by the green solid curve in Figure 1[18-20].

In contrast, in traditional PV systems with a centralized MPPT architecture, any disturbance can cause a shift in the maximum power point of the module, resulting in a significant reduction in power output unless the module's output voltage is adjusted [21].

Currently, there is growing interest in the application of energy storage technologies in power systems. Energy storage options include various types of batteries, high-speed flywheels, supercapacitors, and regenerative fuel cells, each offering unique characteristics and being at different stages of development. Local energy storage systems can act as a buffer between variable supply and demand, supporting embedded generation from renewable sources. Depending on specific technical requirements and geographical conditions, utilities may choose to employ one or more of these technologies. Despite the existence of pumped storage plants in some utilities, little attention has been given to the potential roles of load management in filling demand troughs or reducing demand peaks. This approach enables a partial decoupling of energy production from energy consumption. Energy storage systems can serve similar functions as load management while also acting as a source of generation. They can replace costly and inefficient storage capabilities or facilitate load scheduling [22- 25].





Fig. 3 Simulation of fuzzy control to maintain desired SOC of the battery [10].

Fuzzy control theory is specifically tailored for hybrid systems to achieve optimal system performance. In this context, the design criterion involves implementing maximum power point trackers for both the photovoltaic device and the wind turbine to ensure they operate at their respective maximum power points. Additionally, the control strategy considers the discrepancy between the actual load and the total generated power, with the Li-ion battery being involved in both charge and discharge modes. The lifetime of the battery is directly influenced by its state of charge (SOC). To enhance the battery's lifespan, fuzzy control is utilized to regulate and maintain the SOC of the battery. This fuzzy control approach contributes to prolonging the Li-ion battery's life while ensuring efficient energy management within the system [26].

•Simulation Results and Discussion

The proposed model has been validated and demonstrates efficient control of battery charging and discharging operations within a safe operating region. The analysis reveals that with variations in load demand, both simulation and experimental results exhibit similar trends of increasing or decreasing state of charge (SOC). However, there is a slight variation observed, which can be attributed to disparities in voltage fluctuations between the simulation and experimental setups [27].



Fig.4. Battery Current v/s Reference current.[8]





Fig.5 Battery Voltage v/s Reference Voltage.[15]

Fig.6 SOC of the battery [17]



Fig.7 Power across Battery, Power load, Power reference using Fuzzy Logic controller [18].



Fig.8 Gate Pulses applied across the converter using Fuzzy Logic Controller [25].

V. CONCLUSION

This paper introduces a fuzzy model based on the ampere hour technique to address the issue of balancing load demand with the available power supply from various sources and the state of charge (SOC) of the storage. The ampere hour technique model is chosen due to its widespread use for evaluating SOC and its ease of implementation. A fuzzy controller has been developed to regulate the charging and discharging of the battery within the safe operating range. A set of 25 carefully calibrated rules has been formulated, considering the load demand, available power from the microgrid

(MG), and battery SOC. To minimize grid energy consumption and optimize the utilization of the battery and distributed sources, the grid is only utilized during specific periods in the simulation. The thesis also investigates the SOC variation in response to different load fluctuations.

VI. FUTURE SCOPE

- The primary contribution of this study is the development of an enhanced fuzzy model and its implementation in a real-time application for controlling the charging and discharging of the battery.
- The proposed approach enables a seamless and uninterrupted operation of the battery's charging and discharging processes, aligning them with the available power sources and the load demand.

VII.REFERENCES

- S. Kouro, J. I. Leon, D. Vinnikov, and L. G. Franquelo, "Grid- connected photovoltaic systems: An overview of recent research and emerging pv converter technology," IEEE Industrial Electronics Magazine, vol. 9, no. 1, pp. 47–61, 2015.
- B. Tarroja, B. Shaffer, and S. Samuelsen, "The importance of grid integration for achievable greenhouse gas emissions reductions from alternative vehicle technologies," Energy, vol. 87, pp. 504–519, 2015.
 S. Koohi-Kamal, N. Rahim, H. Mokhlis, and V. Tyagi, "Photovoltaic electricity generator dynamic modeling
- S. Koohi-Kamal, N. Rahim, H. Mokhlis, and V. Tyagi, "Photovoltaic electricity generator dynamic modeling methods for smart grid applications: A review," Renewable and Sustainable Energy Reviews, vol. 57, pp. 131– 172, 2016.
- 4. M. De Giorgi, P. Congedo, and M. Malvoni, "Photovoltaic power forecasting using statistical methods: impact of weather data," IET Science, Measurement & Technology, vol. 8, no. 3, pp. 90–97, 2014.
- 5. N. Karami, N. Moubayed, and R. Outbib, "General review and classification of different mppt techniques," Renewable and Sustainable Energy Reviews, vol. 68, pp. 1–18, 2017.
- 6. M. Faisal, M. A. Hannan, P. J. Ker, and M. N. Uddin, "Backtracking Search Algorithm Based Fuzzy Charging discharging Controller for Battery Storage System in Microgrid Applications," IEEE Access, 2019.
- R. Heydari, M. Gheisarnejad, M. H. Khooban, T. Dragicevic, and F. Blaabjerg, "Robust and Fast Voltage-Source-Converter (VSC) Control for Naval Shipboard Microgrids," IEEE Trans. Power Electron., vol. 34, no. 9, pp. 8299–8303, Sep. 2019.
- 8. "Binary Particle Swarm Optimization for Scheduling MG Integrated Virtual Power Plant Toward Energy Saving," IEEE Access, vol. 7, pp. 107937–107951, 2019.
- 9. A. Cagnano, E. De Tuglie, and P. Gibilisco, "Assessment and Control of Microgrid Impacts on Distribution Networks by Using Experimental Tests," IEEE Trans. Ind. Appl., vol. 55, no. 6, pp. 7157–7164, Nov. 2019.
- 10. M. Faisal, M. A. Hannan, P. J. Ker, A. Hussain, M. Mansur, and F. Blaabjerg, "Review of energy storage system technologies in microgrid applications: Issues and challenges," IEEE Access, vol. 6, no. 1, 2018.
- 11. S. Mohamed, M. F. Shaaban, M. Ismail, E. Serpedin, and K. A. Qaraqe, "An Efficient Planning Algorithm for Hybrid Remote Microgrids," IEEE Trans. Sustain. Energy, vol. 10, no. 1, pp. 257–267, Jan. 2019.
- 12. Krishnamurthy and A. Kwasinski, "Effects of Power Electronics, Energy Storage, Power Distribution Architecture, and Lifeline Dependencies on Microgrid Resiliency during Extreme Events," IEEE
- 13. J. Emerg. Sel. Top. Power Electron., vol. 4, no. 4, pp. 1310–1323, Dec. 2016.
- 14. J. Choi, Y. Shin, M. Choi, W. K. Park, and I. W. Lee, "Robust Control of a Microgrid Energy Storage System Using Various Approaches," IEEE Trans. Smart Grid, vol. 10, no. 3, pp. 2702–2712, May 2019.
- 15. J. Pegueroles-Queralt, F. D. Bianchi, and O. Gomis-Bellmunt, "Control of a lithium-ion battery storage system for microgrid applications," J. Power Sources, vol. 272, pp. 531–540, 2014.
- 16. Z. Zhang, J. Wang, and X. Wang, "An improved charging/discharging strategy of lithium batteries considering depreciation cost in day- ahead microgrid scheduling," Energy Convers. Manag., vol. 105, pp. 675–684, 2015.
- 17. H. H. Abdeltawab and Y. A. I. Mohamed, "Market-Oriented Energy Management of a Hybrid Wind-Battery Energy Storage System Via Model Predictive Control With Constraint Optimizer," IEEE Trans. Ind. Electron., vol. 62, no. 11, pp. 6658–6670, 2015.
- Tran, N. Tran, S. Yu, Y. Park, and W. Choi, "Design of a Nonisolated Fuel Cell Boost Charger for Lithium Polymer Batteries With a Low Output Ripple," IEEE Trans. Energy Convers., vol. 30, no. 2, pp. 605–614, 2015.
- M. M. S. Khan, M. O. Faruque, and A. Newaz, "Fuzzy Logic Based Energy Storage Management System for MVDC Power System of All Electric Ship," IEEE Trans. Energy Convers., vol. 32, no. 2, pp. 798–809, 2017.
- 20. F. S. Tidjani, A. Hamadi, A. Chandra, P. Pillay, and A. Ndtoungou, "Optimization of standalone microgrid considering active damping technique and smart power management using fuzzy logic supervisor," IEEE Trans.

Smart Grid, vol. 8, no. 1, pp. 475–484, 2017.

- M. A. Hannan, Z. A. Ghani, M. M. Hoque, P. J. Ker, A. Hussain, and Mohamed, "Fuzzy Logic Inverter Controller in Photovoltaic Applications: Issues and Recommendations," IEEE Access, vol. 7, pp. 24934–24955, 2019.
- R. Al Badwawi, W. R. Issa, T. K. Mallick, and M. Abusara, "Supervisory Control for Power Management of an Islanded AC Microgrid Using a Frequency Signalling-Based Fuzzy Logic Controller," IEEE Trans. Sustain. Energy, vol. 10, no. 1, pp. 94–104, 2019.
- M. Jafari, Z. Malekjamshidi, D. D. Lu, and J. Zhu, "Development of a Fuzzy-Logic Based Energy Management System for a Multiport Multioperation Mode Residential Smart Microgrid," IEEE Trans. Power Electron., vol. 34, no. 4, pp. 3283–3301, 2019.
- 24. D. Arcos-Aviles, J. Pascual, L. Marroyo, P. Sanchis, and F. Guinjoan, "Fuzzy logic-based energy management system design for residential grid-connected microgrids," IEEE Trans. Smart Grid, vol. 9, no. 2, pp. 530–543, 2018.
- Y.-K. Chen, Y.-C. Wu, C.-C. Song, and Y.-S. Chen, "Design and Implementation of Energy Management System With Fuzzy Control for DC Microgrid Systems," IEEE Trans. Power Electron., vol. 28, no. 4, pp. 1563– 1570, 2013.
- H. R. Baghaee, M. Mirsalim, and G. B. Gharehpetian, "Performance Improvement of Multi-DER Microgrid for Small- and Large-Signal Disturbances and Nonlinear Loads: Novel Complementary Control Loop and Fuzzy Controller in a Hierarchical Droop-Based Control Scheme," IEEE Syst. J., vol. 12, no. 1, pp. 444–451, 2018.
- N. L. Diaz, T. Dragi^{*}, and J. M. Guerrero, "Intelligent Distributed Generation and Storage Units for DC Microgrids — A New Concept on Cooperative Control Without Communications Beyond Droop Control," IEEE Trans. Smart Grid, vol. 5, no. 5, pp. 2476 2485, 2014.
- 28. M. Datta and T. Senjyu, "Fuzzy control of distributed PV inverters/energy storage systems/electric vehicles for frequency regulation in a large power system," IEEE Trans. Smart Grid, vol. 4, no. 1, pp. 479–488, 2013.
- 29. L. Jiang et al., "Optimization of multi-stage constant current charging pattern based on Taguchi method for Li-Ion battery," Appl. Energy, vol. 259, p. 114148, 2020.

Analysis of Electricity Consumption Using LSTM Model

Durga Prasad Ananthu^{#1} T. Vinay Kumar^{#2} Neelashetty K^{#3} Rupesh M^{*4}

^{#1, 2, 3} Department of EEE, Guru Nanak Dev Engineering College, Bidar, Karnataka- India
 ^{*4} Department of EEE, BVRIT Womens Engineering College, Hyderabad, Telangana- India
 ^{#1}adp.ananthu@gmail.com
 ^{#2}thadamalla.vinay@gmail.com
 ^{#3} neelshettyk@gmail.com
 ^{#4}rupesh.m@bvrithyderabad.edu.in

Abstract—Estimating the electrical demand is critical for grid planning. It is challenging to foresee electrical consumption since the amount of power used on the power grid varies with time. This study focuses on Karnataka state's electricity usage over a quarter period and created a data analysis model utilising ML algorithm to analyse energy consumption patterns. The proposed model trained using historical data from KPCL Karnataka, and the results were compared to auto regression and LSTM methods. The performance of the trained model evaluated by measuring MPE and RMSE values. This research assists companies in planning the production of electricity based on load demands, which improves the power quality and reliability of the grid.

Keywords- Electrical demand, grid planning, LSTM, auto regression, RMSE.

I. INTRODUCTION

The need for electrical power is growing every day; however, because India is an emerging nation, distributing electricity to all users is complex. The amount of electricity needed on the electrical system fluctuates over time; to satisfy this need, algorithms based on ML and AI are used to precisely forecast demand for loads. The advantages of demand predictions include reduced outages, losses, and total operating expenses, as well as enhanced resilience, security, and quality of power.

The rising use of energy in the distribution networks may jeopardise system resilience. So, to improve grid efficiency [1] presented forecasting power using DNN technique based on LSTM networks. This research work compares the LSTM and other models. [2] This research proposes a novel artificially intelligent neural network model for STLF that was evaluated in day and weekly forecasting, the model displayed accuracy and RMSE values. A new prediction model for small scale load prediction in buildings or sites has been proposed in [3]. An upgraded form of empirical mode decomposition (EMD), a new feature selection technique, and a hybrid forecast engine are used in the model. The effectiveness of the model is assessed using real-world engineering test cases. [4] Deep learning theory can help with day-ahead load projections for the future waste reduction, and generator unit placement. It uses stacked denoising autoencoders to refine features from load data and SVM models. [5] This paper presents an AI-based RAP-NN for smart grid measurements, enhancing reliability and power quality by avoiding memory usage for recurrent updates, resulting in precise prediction parameters. [6] FE- SVR-mFFO hybrid technique improves real-time, accurate, and stable forecasting for smart grid strategic decisions by combining feature engineering and modified fire-fly optimization, outperforming benchmark frameworks in accuracy, stability, and convergence rate, as demonstrated by real-world load data. [7] The paper presents a system for predicting daily energy load in smart homes using LSTM and ARIMA models, with STDAN algorithm for improved accuracy. [8] This work examines accurate voltage fluctuation modelling and forecasting, with a focus on power-quality improvement and corrective device management. For precision, grey system theory-based models, such as the Fourier correction Grey Model, are used. [9] On three nonlinear electric load datasets, the paper analyses the multi-step forecasting performance of Auto-Regressive Integrated Moving Average (ARIMA) and Long-Short-Term-Memory (LSTM) based Recurrent Neural Networks models.

The proposed work uses KPCL Load curves for analysis of electricity consumption; it contains 360 hourly samples to predict the load. Load curves available at http://218.248.45.137:8282/LoadCurveUpload/lcdownloadvie w.asp".

Section 2 provides a literature /background, while Section 3 addresses the execution of load estimation using LSTM networks and presents results of different models for prediction. Section 4 presents the conclusion.

II. LITERATURE

• H. Dagdougui et al. [10] assessed the performance of neural network (ANN) in STLF on a Montreal campus,

analysing hourly and daily forecasts for various building types. S. Hosein et al. [11]. This research investigates the use of DNN and other machine learning approaches to estimate STL consumption in a grid. Elvisa and Marijana [12]. This paper discusses machine-learning algorithms for STLF, including SVM, random forest regression, and LSTM network, with a fusion forecasting approach and data pretreatment technique.

- Ammar O et al. [13] The paper introduces a novel method for renewable power forecasting using a multicolumn radial basis function neural network, utilizing a modified error correction algorithm to speed up convergence. Stefan Hosein et al. [14] The study utilized ML techniques on SM data to calculate STLF in a power system, discuss variable pricing for peak load, and calculate MAPE.
- J. Zheng et al. [15] This study examines smart metres' position in the SG, concentrating on its functionality, electronic communications (RF and PLC), and current improvements. It also evaluates the current SM solutions offered by various firms and covers the regulations and growth aims of SG in different jurisdictions.
- N. T. Mbungu et al. [16]The research investigates the best planning and administration for power grids, with an emphasis on supply consistency and smart grid to minimise expenses, solve energy unpredictability, cut emissions, and satisfy growing energy needs.
- Hasmat Malik et al. [17] For smart load forecasting in today's electrical systems, a fuzzy-Q-learning-based FRL technique has been suggested, ensuring precise and steady energy production over long horizons with excellent prediction accuracy.
- Essallah et al. [18] The present investigation examines four LTLF approaches based on Tunisian electrical usage information, evaluating the accuracy of each as well as their pros and cons as a result of irregular and variable power demands.
- U. Singh et al. [19] For accurate forecasts, the work seeks to forecast short-term loads for as long as seven days in advance utilising Naive Bayes, and Deep Learning algorithms [20] The article demonstrates the progress of neural network-based predictions in electrical technology. by showing uses such as problem evaluation, status tracking, and scheduling.

According to research previous models exhibit substantial MSE, and RMSE errors. Deep learning approaches like as LSTM, and regression methods SARIMA were developed for STL predictions.

III. CONCEPT OF FORECAST

A time series (TS) analysis is a collection of information organised in sequence. Make a TS data set by converting from the collected dataset. Forecasting is used to create precise forecasts based on past trends. It is made up of three components: the trend, seasonality, and an error.

$$Fts = Mts + Sts + Ets \qquad (1)$$

There are several prediction processes, the auto-regression and moving average components make up the ARMA. In the ARMA model, from the equation 2, a fixed and error value included to AR and MA component.

$$Y_{t} = \beta_{2} + \omega_{1} \varepsilon_{t-1} + \omega_{2} \varepsilon_{t-2} + \dots + \omega_{q} \varepsilon_{t-q} + \varepsilon_{t}$$

(2)

The SARIMA technique, like ARIMA, has extra AR and MA components, with periodic modifications to account for additional lags such as hourly.

$$y_{t} = c + \sum_{n=1}^{p} \alpha_{n} y_{t-n} + \sum_{n=1}^{q} \theta_{n} \epsilon_{t-n} + \sum_{n=1}^{p} \phi_{n} y_{t-m} + \sum_{n=1}^{Q} \eta_{n} \epsilon_{t-n} + \epsilon_{t}$$

(3)

The LSTM network is a ML model that has had widespread application in time series forecasting challenges. To avoid the long-term reliance problem, the LSTM uses a memory cell that maintains selected data from past time steps, as opposed to regular RNNs, whose applicability is typically restricted to short-term memory tasks.

LSTM networks excel in the neural network family because of their excellent storage capacity for lengthy time. They have input, forget, output, and cell candidate gates, with the output gate receiving data, the input gate providing data, and the forget gate selecting retention.



Fig.1 Long Short-Term Memory Cell [1]

$$i_{t} = \sigma(\text{Wixt} + \text{UiHt} - 1 + \text{ViCt} - 1 + \text{bi})$$
(4)

$$f_{t} = \sigma(\text{Wfxt} + \text{UfHt} - 1 + \text{VfCt} - 1 + \text{bf})$$
(5)

$$C_{t} = \text{ft} \odot \text{Ct} - 1 + \text{it} \odot \tanh(\text{Wcxc} + \text{UcHt} - 1 + \text{bc})$$
(6)

$$O = \sigma(Wo xt + UoHt - 1 + VoCt - 1 + bo)$$
(7)

$$\text{Ht} = \text{Ot} \odot \tanh\text{Ct}$$
(8)

The LSTM algorithm shown in Fig2 https://in.mathworks.com/help/deeplearning/ug/time-series- forecasting-using-deep-learning.html?s_tid=srchtitle_time%20series%20forecasting_1



Fig.2 LSTM flow chart

I. RESULTS

The data collected from KS load curves are loaded to MATLAB program and process it using LSTM training model, it has 360 hourly samples, in that ninety percent data i.e 342 hourly samples used for training the model and rest ten percentdata (18 samples) used for testing the model, fig 3 shows the training process. Construct predictive models and estimate into upcoming steps, evaluate the trained algorithm's precision and RMSE value, and estimate projected load demand.



Fig. 5 Estimated per hour demand for Aug 23

The mean value for the July 23 is 8161MW, RMS Error is 54.56 MW shown in Fig4. The mean is 10165 MW and RMSE is 711.368 MW and after updating the training model it is 45.5984MW for Aug 23 depicted in fig5. The variation in predicted value is 86.75%. As a result, this strategy assists companies in planning the production of electricity based on load demands, which improves the power quality and reliability of the grid.


Figure 6 depicts the estimated power demand for 72 hours ahead.

TABLE I RESULTS

Forecast Model	SARIM ARef. D. P. Ananthu [2]	LSTM For July 23	LSTM For Aug 23
Mean (MW)	7796	8161	10165
MSE	159894.42	2976.7 9	2079.2 1
RMSE (MW)	399.86	54.56	45.59

II.

CONCLUSION

Developed different methods for forecasting short-term electrical load. The KS load curves are utilised for data analysis, the analysis comprises 360 samples, 95% of which were utilized for developing the model and the rest, or 5%, for investigation, and the projected energy demands for the next 24 hours and 72 hours ahead for July and Aug 2023 respectively. The MSE and RMSE of the trained model were evaluated and observed that the error value is substantially low and the prediction accuracy was 86.75%. So that these methods are best suitable in grid planning and to improve reliability.

ACKNOWLEDGMENT

Thanks to R & D EEE, GNDEC Bidar and KPCL, daily load curves used in this research are obtained from KPCL Dept.

III.REFERENCES

- 1. D. P. Ananthu and K. Neelashetty, "A study of 100kWp PV Plant Output Power Forecasting: A case study," Proc. 5th Int. Conf. Comput. Methodol. Commun. ICCMC 2021, no. Iccmc 2021, pp. 723–728, 2021.
- D. P. Ananthu and K. Neelashetty, "Electrical Load Forecasting using ARIMA, Prophet and LSTM Networks," Int. J. Electr. Electron. Res., vol. 9, no. 4, pp. 114–119, 2021.
- Y. Liu, W. Wang, and N. Ghadimi, "Electricity load forecasting by an improved forecast engine for building level consumers," Energy, vol. 139, pp. 18–30, Nov. 2017.
- 4. C. Tong, J. Li, C. Lang, F. Kong, J. Niu, and J. J. P. C. Rodrigues, "An efficient deep model for day-ahead electricity load forecasting with stacked denoising auto-encoders," J. Parallel Distrib. Comput., vol. 117, pp. 267–273, Jul. 2018.

- 5. D. P. Ananthu and K. Neelshetty, "Prominent smart grid measurements by RAP-NN to warrant the reliability of the PV system," Energy Sources, Part A Recover. Util. Environ. Eff., 2019.
- G. Hafeez, I. Khan, S. Jan, I. A. Shah, F. A. Khan, and A. Derhab, "A novel hybrid load forecasting framework with intelligent feature engineering and optimization algorithm in smart grid," Appl. Energy, vol. 299, Oct. 2021.
- I. Mpawenimana, A. Pegatoquet, V. Roy, L. Rodriguez, and C. Belleudy, "A comparative study of LSTM and ARIMA for energy load prediction with enhanced data preprocessing," in 2020 IEEE Sensors Applications Symposium, SAS 2020 - Proceedings, 2020.
- 8. A. Dejamkhooy, A. Dastfan, and A. Ahmadyfard, "Modeling and Forecasting Nonstationary Voltage Fluctuation Based on Grey System Theory," IEEE Trans. Power Deliv., vol. 32, no. 3, pp. 1212–1219, Jun. 2017.
- 9. S. Masum, Y. Liu, and J. Chiverton, "Multi-step time series forecasting of electric load using machine learning models," in Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics), 2018, vol. 10841 LNAI.
- 10. H. Dagdougui, F. Bagheri, H. Le, and L. Dessaint, "Neural network model for short-term and very-short-term load forecasting in district buildings," Energy Build., vol. 203, Nov. 2019.
- 11. W. He, "Load Forecasting via Deep Neural Networks," in Procedia Computer Science, 2017, vol. 122.
- 12. E. Becirovic and M. Cosovic, "Machine learning techniques for short-Term load forecasting," in 4th International Symposium on Environment Friendly Energies and Applications, EFEA 2016, 2016.
- A. O. Hoori, A. Al Kazzaz, R. Khimani, Y. Motai, and A. J. Aved, "Electric Load Forecasting Model Using a Multicolumn Deep Neural Networks," IEEE Trans. Ind. Electron., vol. 67, no. 8, 2020.
- 14. S. Hosein and P. Hosein, "Load forecasting using deep neural networks," in 2017 IEEE Power and Energy Society Innovative Smart Grid Technologies Conference, ISGT 2017, 2017.
- 15. J. Zheng, D. W. Gao, and L. Lin, "Smart Meters in Smart Grid: An Overview," in 2013 IEEE Green Technologies Conference (GreenTech), 2013, pp. 57–64.
- N. T. Mbungu, R. M. Naidoo, R. C. Bansal, and V. Vahidinasab, "Overview of the Optimal Smart Energy Coordination for Microgrid Applications," IEEE Access, vol. 7. Institute of Electrical and Electronics Engineers Inc., pp. 163063–163084, 2019.
- 17. H. Malik, N. Fatema, and A. Iqbal, "Intelligent Data Analytics for Time-Series Load Forecasting Using Fuzzy Reinforcement Learning (FRL)," Intell. Data-Analytics Cond. Monit., pp. 193–213, Jan. 2021.
- 18. S. Essallah and A. Khedher, "Long-term load forecasting in the smart grid framework using time series and econometric approaches," Renew. Energy Prod. Distrib. Recent Dev., pp. 389–413, Jan. 2022.
- 19. U. Singh, A. Negi, V. Garg, and R. Pillai, "Electrical Load Forecasting," in 2021 International Conference on Intelligent Technologies (CONIT), 2021, pp. 1–6.
- 20. C. Booth, J. R. Mcdonald, and S. D. J. Mcarthur, "Forecasting and prediction applications in the field of power engineering," J. Intell. Robot. Syst. Theory Appl., 2001.

Mathematical Optimizing Solar Energy Utilization: Novel PCM-Integrated Solar Water Heating Systems

Nitin Morkane¹, Sarika Sakhare²

¹Assistant Professor in Mechanical Engineering Department, SVERI's COE Pandharpur, Maharashtra, India ²Assistant Professor in General Science and Engineering Department, SVERI's COE Pandharpur, Maharashtra, India

> ¹nnmorkane@coe.sveri.ac.in ²sssakhare@coe.sveri.ac.in

Abstract— An innovative solution to enhance solar energy utilization for continuous hot water supply in modern architectural settings. Overcoming limitations of conventional solar collectors, the proposed design integrates a solar tank featuring phase change material (PCM), optimizing both size reduction and increased heat capacity. This facilitates solar collectors to operate efficiently at lower temperatures, enhancing overall system performance. The system comprises a solar water heater and a PCM-based heat storage unit, ensuring uninterrupted hot water availability during nighttime and adverse weather conditions. Employing small paraffin-filled cylinders within solar collectors maximizes solar energy absorption and storage. A comprehensive comparative analysis against conventional sensible heat storage systems demonstrates the superior efficiency of the PCM-based solution. This research addresses the challenges of solar energy intermittency and spatial constraints, offering a promising avenue for advancing sustainable and efficient solar energy utilization in residential applications. Keywords: Thermal Energy, Solar water heater, Phase change material (PCM), Weather dependency.

Keywords- Thermal Energy, Solar water heater, Phase change material (PCM), Weather dependency

I. INTRODUCTION

In an era where global energy demands escalate alongside the exponential growth of population, the vitality of accessible and sustainable energy sources cannot be overstated. India, in particular, has experienced a remarkable surge in energy demand, growing steadily over the past three decades due to its rapid economic expansion. This burgeoning need for energy has instigated a critical necessity to explore alternative and renewable sources.

Among the array of renewable options, solar energy emerges as a prominent candidate. Recognized for its simplicity, cleanliness, and inexhaustible nature, solar power has garnered substantial attention. However, its intermittent nature necessitates effective storage solutions for optimal utilization. Addressing this need, Phase Change Materials (PCMs) stand out as a viable technique for storing thermal energy through latent heat.

PCM, predominantly inorganic compounds like hydrated salts, possesses the unique ability to absorb or release significant heat energy during state changes, sustaining its latent heat across numerous cycles without altering its

properties. In the context of solar water heating, PCM plays a pivotal role in storing energy by melting at consistent temperatures, ensuring efficient utilization during usage cycles. The selection of an appropriate PCM hinges on several critical factors, notably its melting point, latent heat capacity, and safety considerations. Balancing these aspects with application-specific temperature requirements is paramount, especially in domains like residential, commercial, and industrial sectors, each with distinct thermal demands.

While inorganic compounds boast high latent heat capacities, they are susceptible to undercooling, potentially hindering effective phase changes. Conversely, organic compounds, excluding paraffin, tend to be expensive or pose toxicity concerns. Paraffin, despite its flammability, emerges as a pragmatic choice due to its favourable physical and chemical properties, affordability, and compatibility within water-based systems.

Furthermore, commercially available specific temperature PCMs, particularly various paraffin types, provide versatile options for diverse applications. In essence, this paper explores the multifaceted landscape of PCM utilization, highlighting their pivotal role in enhancing the efficacy of solar energy utilization, particularly in the context of water heating applications.

TABLE I PHYSICAL PROPERTIES OF PARAFFIN 5838

Melting point	58-62 degrees Celsius.	
Latent heat of Fusion	Typically around 200 J/g.	
Viscosity	$1.9 \text{ mm}^2/\text{s}$	
Density	0.9 to 0.94 g/cm ³ .	
Specific heat capacity-solidSpecific heat capacity-liquid	Approximately 2.0-2.5 J/g°C.	
Thermal Conductivity	Ranges from 0.2 to 0.3 W/m°C.	

II. LITERATURE REVIEW

The integration of phase change materials (PCMs) within solar water heating systems has been a subject of extensive research over the years [1]. Studies on PCM utilization within thermal energy storage units have explored various methodologies to intensify heat transfer, showcasing promising advancements [2]. Evaluations of solar collector designs featuring integrated latent heat thermal energy storage have underscored the potential for enhancing system efficiency [3].

A comprehensive review of thermal energy storage utilizing phase change materials emphasizes their versatile applications and their role in enhancing energy sustainability [4]. Investigations into the thermal performance of waterphase change material solar collectors have highlighted their efficacy in harnessing solar energy [5]. Review articles have extensively covered the materials, heat transfer analyses, and practical applications related to thermal energy storage with phase change [6].

Research on integrated solar collector/storage units employing phase change materials at a specific temperature of 65°C has elucidated their thermal behavior and operational efficiencies [7]. Additionally, experimental studies on prototypes equipped with latent heat storage have provided valuable insights into system performance [8, 9].

The cumulative body of research in this domain converges on the potential and efficacy of phase change materials in augmenting solar water heating systems, emphasizing their pivotal role in advancing sustainable energy solutions.

III. METHODOLOGY AND EXPERIMENTAL SETUP

Figure 1 illustrates the configuration of the experimental setup utilized in this study. The Thermal Energy Storage (TES) tank, possessing a capacity of 48 liters, was designed to meet the hot water demands of a standard family comprising four individuals. The tank incorporates two plenum chambers located at its top and bottom, complemented by a flow distributor at the tank's apex to ensure a homogeneous flow distribution. To meet the daily requirement of approximately 60 liters of heated water for household use, the energy is stored within the TES tank as a composite of sensible and latent heat of Phase Change Material (PCM), alongside the sensible heat of water. For the sake of this study, it is assumed that two-thirds of the total energy is stored within the PCM, with the remaining third retained as sensible heat within the water. For comparative purposes, an identical TES tank without PCM shells is employed in the PCM-less system variant.



Figure 1. The diagram of the SWH-PCM system

IV. RESULTS AND DISCUSSION

The comparison between systems with and without Phase Change Material (PCM) revealed distinct thermal energy storage mechanisms. In the PCM-absent system, the water alone acquires and stores solar energy, while in the PCM- equipped system, both water and PCM absorb and retain energy, resulting in PCM melting above its specified melting point.



Fig. 2.Comparison of Cumulative Heat in Both the System



Fig. 3.Solidifying Time in Function of Time and Temperature Difference



Fig. 4.Temperature Histories of Both Systems during charging



Fig. 5.Comparison of System Efficiency

a) Comparison of Heat Energy Storage

The comparison between systems utilizing and not utilizing Phase Change Material (PCM) demonstrates significant variations in heat storage efficiency. The system incorporating PCM surpasses the PCM-lacking system of identical size and tank volume. Calculations reveal a heat storage of 0.234 kJ/cc for the PCM system and 0.144 kJ/cc for the PCM-less system, indicating a substantial decrease in necessary storage volumes for equivalent heat storage in the PCM-equipped system.

b) Solidification Time of PCM

Understanding the solidification time of PCM is crucial for ensuring sustained hot water availability in the absence of solar energy. Experimental findings report a solidification time of 172 minutes with a 2°C temperature difference between the phase change temperature and the outer wall of a tube with a diameter of 60mm and 1mm wall thickness. A reduced solidification time of 66 minutes was noted with a similar temperature difference in a tube with a 40mm diameter and 1mm wall thickness.

c) Tube Diameter and PCM Solidification

The impact of tube diameter on PCM solidification time is evident; smaller diameters necessitate more tubes for the same PCM quantity. While smaller diameters facilitate quicker specific heat exchange and solidification, they decrease the tank's overall heat capacity due to increased tube material volume.

d) Charging Dynamics and System Efficiency

During charging, the PCM-equipped system redistributes solar energy to both water and PCM, contrasting the PCM-less system, which solely relies on water for energy absorption. Notably, the PCM-less system reaches a maximum temperature of 70° C 40 minutes earlier than the PCM system, with an average charging time 30-60 minutes faster.

e) System Efficiency Comparison

System efficiency analysis indicates that the PCM-less system exhibits fluctuating efficiency levels over time, whereas the PCM-equipped system maintains consistent efficiency across phase transition temperatures, displaying higher and more consistent efficiency levels. Consequently, the system employing PCM demonstrates enhanced efficiency in solar thermal storage applications.

V. CONCLUSION

This study delved into the thermal behavior of systems incorporating Phase Change Material (PCM) under varied operating conditions. The investigation encompassed charging times, energy storage capacities, PCM solidification duration, and system efficiencies, facilitating a comprehensive comparison between both systems.

The findings underscore the economic and spatial advantages of PCM-based tanks over conventional counterparts of equal heat capacity. Notably, PCM tanks exhibit lower manufacturing costs and reduced spatial demands, offering a more cost-effective and space-efficient solution for thermal energy storage. Furthermore, PCM tanks demonstrate the remarkable advantage of maintaining a consistent temperature profile during heat accumulation. This attribute, governed by the specific PCM type, enables flexible temperature settings conducive to higher solar collector efficiency, particularly in colder external environments. Conclusively, systems integrating PCM emerge as a highly viable option for solar heat energy storage. Their distinct advantages, including cost-effectiveness, space efficiency, and temperature control benefits, position them as promising alternatives to prevailing domestic solar water heating technologies reliant solely on sensible

heat storage. Embracing PCM-based systems signifies a tangible stride toward more efficient and sustainable solar thermal energy utilization in domestic settings.

VI. REFERENCES

- 1. P.B.L.Chaurasia, 'Phase change material in solar water heater storage system' Proceeding of the 8th International Conference on Thermal Energy Storage, 2000, Stuttgart, Germany.
- Elarem, R., Alqahtani, T., Mellouli, S., Askri, F., Edacherian, A., Vineet, T., Badruddin, I.A., Abdelmajid, J. (2021). A comprehensive review of heat transfer intensification methods for latent heat storage units. Energy Storage, 3, e1272021.
- 3. Khan, M.M.A., Ibrahim, N.I., Mahbubul, I.M., Ali, H.M., Saidur, R., Al- Sulaiman, F.A. (2018). Evaluation of solar collector designs with integrated latent heat thermal energy storage: A review. Solar Energy, 166, 334–350
- 4. Sharma, A., Tyagi, V.V., Chen, C.R., Buddhi, D. (2009). Review on thermal energy storage with phase change materials and applications. Renewable and Sustainable Energy Reviews, 13, 318–345.
- Kurklu, A., Ozmerzi, A., Bilgin, S. (2002). Thermal Performance of a Water-Phase Change Material Solar Collector. Renewable Energy, 26, 391–399.
- 6. Zalba, B., Marín, J.M., Cabeza, L.F., Mehling, H. (2003). Review on thermal energy storage with phase change: Materials, heat transfer analysis and applications. Applied Thermal Engineering, 23, 251–283
- 7. Eames, P., Griffiths, P. (2006). Thermal behaviour of integrated solar collector/storage unit with 65 °C phase change material. Energy Conversion and Management, 47, 3611–3618.
- 8. Plantier, C. (2005). Etude Numérique et Expérimentale d'un Prototype de Chauffe-eau Solaire Équipé d'un Stockage à Chaleur Latente. ESIGEC—Univérsité de Savoie, Chambéry, France.
- Haillot, D., Goetz, V., Py, X., Abdelkarim, M.B. (2008). Integrated solar collector storage system based on phase change material and graphite composite. In Proceedings of the EUROSUN 2008, 1st International Congress on Heating, Cooling and Buildings, Lisbon, Portugal, 7–10 October 2008.

Soldier's Uniform used for Temperature Control and Health Monitoring System based on IoT

Suresh Patil

Department of Electrical and Electronics Engineering, Guru Nanak Dev Engineering College Bidar suripatilsp@gmail.com

Abstract- The solar-powered E-Uniform offers advanced protection to soldiers operating in challenging weather conditions. Its internal circuitry is fueled by solar panels, generating electricity stored in a rechargeable lead acid battery with a 12V DC output. Additionally, a traditional battery charging mechanism ensures continuous operation. The Arduino microcontroller serves as the central control unit, overseeing all functions. An ADC 0808 interfaces with a voltage sampler to monitor battery voltage, displayed on a 16X2 LCD screen. The uniform operates in two modes: summer and winter, with the H-Bridge IC controlling the body heater/cooler accordingly, providing comfort in extreme temperatures. A metal sensor alerts the soldier to potential threats with a buzzer signal. Through IoT connectivity, the soldier's health status is monitored and updated in real-time on a central server, ensuring constant oversight. This innovative uniform equips soldiers to operate effectively in diverse environments, enhancing their capability and safety.

Keywords: E-Uniform, IoT, lead acid rechargeable battery.

I. INTRODUCTION

The military's personnel are its most valuable resource, essential for national defense regardless of branch— Army, Air Force, Navy, or Marines. They dutifully fulfill their responsibilities, even in adverse weather conditions, which can pose significant challenges to their well-being. Extreme temperatures, whether scorching heat or freezing cold, can jeopardize their health and effectiveness. This project aims to develop an E-Uniform offering enhanced protection for soldiers operating in such conditions. The uniform features two modes— summer and winter—controlled by relays to activate the body heater/cooler, ensuring optimal comfort for soldiers in any environment.

Integrating solar-powered e-uniforms could represent a significant advancement in military attire. These uniforms would incorporate solar panels into the fabric, generating electricity to power various functions such as heating or cooling systems, lighting, and communication devices. This environmentally friendly approach reduces reliance on conventional battery- powered equipment, alleviating the burden on soldiers and promoting sustainability. The inclusion of cooling elements in the e-uniforms helps regulate body temperature and prevent heat- related illnesses, while heating components ensure comfort and prevent cold-related injuries. Addressing challenges like maintaining body temperature and communication in harsh environments, solar-powered e-uniforms offer a promising solution to enhance soldier comfort and performance in combat scenarios.

By harnessing solar energy in military apparel, these innovative uniforms offer a practical and effective solution to the challenges faced by troops operating in extreme conditions. They have the potential to significantly improve soldier performance, comfort, and safety, providing a sustainable and environmentally responsible alternative to traditional battery-powered equipment.

II. LITERATURE SURVEY

Dipali H. Kale's solar-based E-Uniform for soldiers offers enhanced protection for soldiers operating in challenging weather conditions. Solar panels power the internal circuitry of the E-uniform, with energy stored in a 12V DC lead-acid rechargeable battery. Additionally, traditional battery charging mechanisms can be used to power the circuits. The ATmega16a microcontroller serves as the central control unit, managing all functions of the uniform. The project features two operational modes: summer and winter, adjusting the body temperature via the heater/cooler to ensure soldier comfort in various environmental conditions.

Rahul Khairamode's E-jacket for soldiers aims to provide protection against extreme weather conditions experienced throughout the year, including high temperatures and cold snaps, which pose health risks such as heatstroke or dehydration. By integrating solar panels and a lead acid rechargeable battery, the E-jacket offers mobility and enhanced protection. The jacket is equipped with a GPS module for location tracking. Dr. S. Ramesh's climate-adjustable E-military suit addresses the challenges posed by unpredictable climatic changes by incorporating air-conditioning technology. The suit utilizes the Peltier effect to regulate body temperature, providing comfort and protection in extreme conditions. It also includes sensors to monitor atmospheric conditions and health parameters, displaying the data on an LCD screen.

III BLOCK DIAGRAM DESCRIPTION



Functional Unit description:

In the figure 1 we shown the hardware requirements & Technical approach in the way to design the system. The system consist of mainly parts like Microcontroller (ATmega328), Sensors, IOT, LCD display, which are described briefly below.

Microcontroller ATmega328

The Arduino UNO is an open-source prototyping platform based on the ATmega328 microcontroller. The onboard microcontroller features six analogue inputs, fourteen digital input/output (I/O) pins, a reset button, a power jack, and an ICSP header. It operates on a 16MHz crystal oscillator and has all the parts needed to support the microcontroller.

Heart beat Sensor

The TCRT1000 reflecting optical sensor is used in the updated model for photoplethysmography. Because the TCRT100's infrared light emitter diode and detector are mounted side by side in a leaded package, it blocks ambient light from the surrounding area, which could otherwise impair sensor performance. This streamlines the sensor construction process within the project. Furthermore, I developed a printed circuit board featuring a signal conditioning unit and the sensor itself, ensuring accurate synchronization with the heartbeat. To further process the data and retrieve the heart rate in beats per minute (BPM), the output pulse can be directed to an ADC channel or a digital input pin of a microcontroller.

LM35 Temperature Sensor

The LM35 family includes precision integrated circuit temperature sensors. These sensors provide an output voltage directly proportional to the temperature in Celsius. Unlike linear temperature sensors calibrated in degrees Kelvin, the LM35 offers the advantage of not requiring a significant constant voltage deduction to obtain a Centigrade measurement. It boasts typical accuracy of $\pm 1/4^{\circ}$ C at room temperature and $\pm 3/4^{\circ}$ C across the entire temperature range of -55 to +150°C, eliminating the need for external calibration or trimming. The low cost of the LM35 is attributed to wafer-level trimming and calibration. Its accurate intrinsic calibration and linear output simplify interfacing with control or readout circuitry.

Internet of Things (IOT)

Utilizing existing network infrastructure, the Internet of Things (IoT) enables remote detection or control of various objects. This integration bridges the real world with computer-based systems, reducing the necessity for human intervention and promoting economic, precise, and efficient outcomes. IoT technology operates within the broader scope of cyber-physical systems, encompassing the integration of sensors and actuators with other technologies. The exponential growth of IoT offers banks opportunities to revolutionize payment methods beyond mobile devices, credit cards, and point-of-sale terminals. Connected devices may eventually surpass smartphones in transaction handling, shaping the financial sector's landscape, particularly with Visa and MasterCard leading infrastructure development. Each object, equipped with an embedded computing system, possesses a unique identity and seamlessly communicates with existing internet infrastructure. Moreover, there's a growing desire for human interaction with non-living objects online, such as clothing, furniture, appliances, and office supplies. While current interaction tools mainly cater to living beings, the IoT facilitates comfortable communication with inanimate objects. The Internet of Things represents the convergence of various technologies, including embedded systems, ubiquitous computing, ambient intelligence, sensors, actuators, and communication technologies.

IoT Server:



IoT server will triggered by the IFTTT server i. e it send the ON OF commands packets to the Node Mcu

ESP8266.

Fig 2 : IoT server

- Adafruit IO serves as a platform aimed at enhancing the value of data. Our primary objective revolves around simplifying usage and minimizing programming requirements for basic data connectivity.
- IO provides client libraries that encapsulate our MQTT and REST APIs, leveraging Node.js and Ruby on Rails as its foundation.
- Adafruit IO is currently in its beta phase. To participate in the beta, visit io.adafruit.com and register.

• The Internet of Things (IoT) refers to a network of physical objects, including buildings and vehicles, equipped with sensors, actuators, software, electronics, and network connectivity. These objects interact and gather data, facilitating remote sensing and control reservoir. through the existing network infrastructure. The 2013 definition of IoT by the Global Standards Initiative on Internet of Things (IoT-GSI) described it as "the infrastructure of the information society." Through IoT, objects can be monitored and managed remotely, fostering closer integration between computer-based systems and the physical world, thereby enhancing efficiency.

Solar Panel



Fig 3.: Solar panel

In essence, solar energy originates from the sun's direct production and collection, typically on Earth. Through a thermonuclear process, the sun converts approximately 650,000,000 tons of hydrogen into helium every second, generating both heat and electromagnetic radiation. This ongoing process is sustained by the sun's internal heat. Electromagnetic radiation, comprising ultraviolet, infrared, and visible light, radiates in all directions.

Despite the vast amount of radiation generated, only a tiny fraction reaches Earth. Most of today's energy sources derive indirectly from the radiation that does penetrate the Earth's atmosphere, with nuclear fission, fusion, and geothermal energy serving as exceptions. The sun serves as the ultimate source of energy for all life forms, including fossil fuels, which originate from former plants and animals reliant on solar energy for survival. Solar radiation directly contributes a significant portion of the planet's energy needs, with additional energy sources derived indirectly. We'll delve into the advantages and disadvantages of this approach, as well as its feasibility, and explore modern applications of solar energy.

Battery

A battery serves as a reservoir for storing electricity to be utilized later. It generates voltage through a chemical reaction that occurs when dissimilar materials, such as positive and negative plates, interact with the electrolyte—a mixture of sulfuric acid and water. Typically, a standard lead-acid battery provides around 2 volts per cell, totaling 12 volts overall.

Once a circuit is established between the battery's positive and negative terminals, electricity begins to flow from it. This flow is triggered when a device requiring electricity is connected to the battery. Additionally, the battery serves as a supplemental power source, supplying extra current when the demand exceeds what the alternator can provide, effectively functioning as an electrical gather data, facilitating remote sensing and control reservoir.

LCD (16X2) Display

A liquid crystal display (LCD) is a sleek, thin electronic visual interface that leverages the light-modulating capabilities of liquid crystals. Unlike traditional displays, LCDs do not emit light themselves but instead manipulate incoming light to produce images. Their versatility finds application in various fields, including computer monitors, televisions, aircraft instrument panels, signage, and more. Additionally, LCDs are ubiquitous in consumer electronics such as smartphones, calculators, clocks, watches, video players, and gaming consoles.

In many instances, LCDs have supplanted older cathode ray tube (CRT) screens due to several advantages. They are known for being gentler on the eyes, more reliable, cost-effective, lightweight, portable, and compact. LCDs offer a broader range of screen sizes compared to CRT and plasma displays, and they do not suffer from image burn-in issues

typically associated with phosphor-based screens. Furthermore, LCDs provide safer disposal options and are more energy-efficient than CRT displays, making them environmentally preferable. Their low power consumption also makes them suitable for battery- operated electronic devices.

Functionally, an LCD is an electronically controlled optical apparatus capable of producing color or monochrome images by arranging liquid crystal-filled pixels in front of a backlight or reflector.



Fig 4: LCD Display

Relay

A relay serves as an electromagnetic bridge between two circuits, providing electrical isolation while enabling one circuit to control the other. These versatile devices are commonly employed to interface a low-voltage electronic circuit with a higher-voltage electrical circuit, allowing for the control of devices like electric fans or lightbulbs using a small sensor circuit. Comprising two essential parts, the input and output, a relay switch operates when a small voltage applied to the input coil generates a magnetic field, known as the operating voltage. Relays come in various operating voltage configurations, such as 6V, 9V, 12V, and 24V, to suit different applications. The output section consists of contactors, which mechanically connect or disconnect. A typical relay includes three contactors: common (COM), normally closed (NC), and normally open (NO). In its resting state, COM is connected to NC. However, when the operating voltage is applied, the relay coil activates, causing COM to switch to the NO contact. Relays are available in different configurations, including DPDT (Double Pole Double Throw), SPDT (Single Pole Double Throw), and SPST (Single Pole Single Throw), varying in the number of switch contacts they offer. By employing the appropriate combination of contactors, it's possible to control the electrical circuit, turning it on or off as needed. Understanding the internal mechanisms of a relay switch is essential for effectively utilizing this component.

III.CONCLUSION

The project "Soldiers uniform used for temperature control and health monitoring system based on IoT" aims to address the challenges faced by soldiers operating in extreme weather conditions by integrating innovative technology into their uniforms. By implementing this concept in practical applications, we can significantly enhance the resilience and effectiveness of military personnel in adverse climates.

The utilization of solar technology in the uniform's design adds a layer of robustness and self-repair capability, making it ideal for mobile deployments where soldiers may face varying environmental conditions. Solar energy offers a renewable and sustainable power source, reducing the reliance on traditional battery-powered systems and ensuring continuous operation even in remote locations. Furthermore, the incorporation of IoT-based health monitoring systems enables real-time tracking of soldiers' vital signs and environmental conditions, providing valuable insights into their well-being and performance. By leveraging IoTconnectivity, commanders and medical personnel can remotely monitor soldiers' health status and intervene promptly in case of any emergencies, thereby ensuring their safety and optimizing mission outcomes.

Overall, this project holds significant promise in enhancing the operational capabilities of soldiers by providing them with advanced temperature control and health monitoring capabilities, ultimately enabling them to carry out their duties effectively even in the harshest of climates.

IV.REFERENCES

- Adarsh K S, Arun Dinesh, Jyothy Elizebeth D: "E-Uniform For Soldier's Who Work At Extreme Temperature Regions", International Journal of Engineering Research and General Science Volume 3, Issue 3, May-June, 2015,, pp. 993 – 998.
- 2. Muhammad Ali Mazidi, Rolin D. McKinlay, Danny Causey PIC Microcontroller and Embedded Systems: Using

Assembly and C for PIC18

- 3. Sheikh, H.R. ; Dept. of Electr. & Comput. Eng., Univ. of Texas, Austin, TX, USA ; Bovik, A.C. ; de Veciana, G. "An information fidelity criterion for image quality assessment using natural scene statistics"
- 4. Han-ShueTan and JihuaHuang, "DGPS-Based Vehicle-to- VehicleCooperative Collision Warning: Engineering Feasibility Viewpoints",IEEE Transactions on Intelligent Transportation Systems, vol.7, no.4,December 2006, pp. 415 428.
- 5. Pertijs, M.A.P. ; Electron. Instrum. Lab., Delft Univ. of Technol., Netherlands ; Makinwa, K.A.A. ; Huijsing, J.H. "A CMOS smart temperature sensor with a 3σ inaccuracy of $\pm 0.1^{\circ}$ C from -55°C to 125°C".
- 6. Abhinav Maheshwari, Abhash Shukla, Mohd. Sufiya, Naveen Kumar, Ankit Kumar, Electronic-Jacket International Journal of Progressive Research in Science and Engineering Volume-1, Issue-4, July-2020 www.ijprse.com.
- Dipali H. Kale, Yogita P. Phapale, Shivani S. Kamble, Mrs. S. L. Shirke: Solar Based E-Uniform for Soldiers, International Research Journal of Engineering and Technology (IRJET) e-ISSN: 2395-0056 Volume: 08 Issue: 05 | May 2021 www.irjet.net
- 8. Kanagalakshmi.B, Dr.S.M.Kannan, R. Krishnavenishri, S.Kamalika, "Solar and IoT Based Health Monitoring, Controlling and Tracking System for Soldiers", SSRG International Journal ofElectrical and Electronics Engineering (SSRG - IJEEE) - Volume 5 Issue 8 - August 2018
- 9. Kawad Pranali, 2Dahiwalkar Gayatri, 3Pooja Adate, 4Prof.S.B. Dhekale: "EUNIFORM", International Journal of Advance Engineering and Research Development.
- 10. Kawad Pranali, 2Dahiwalkar Gayatri, 3Pooja Adate, 4Prof.S.B. Dhekale: "EUNIFORM", International Journal of Advance Engineering and Research Development.

Design and Development of Solar Hybrid Bicycle

Neelashetty K

Department of Electrical and Electronics Engineering, Guru Nanak Dev Engineering College, Bidar

neelshettyk@gmail.com

Abstract: The primary goal of this proposal is to design and build an e-Bicycle using solar as an energy source. A solar panel is a big, flat rectangle that is usually between the size of a door and a radiator. It is composed of numerous tiny solar energy collectors known as solar cells that are shielded from the elements by a layer of glass. Typically octagonal in shape and blue black in hue, the cells are roughly the size of an adult's palm. Similar to battery cells, solar panel cells are made to produce energy; however, unlike battery cells, which do this by using chemicals, solar panel cells do this by absorbing sunlight. Electric motorcycles and Bicycles are two- or three-wheeled plug-in electric cars that run on electricity. One or more electric motors are powered by the electricity that is kept on board in a rechargeable battery. Unlike motorcycles, electric bicycles have a step-through frame. All electric bikes and motorbikes can be recharged by plugging them into regular wall outlets; this process typically takes eight hours, or overnight. Bicycles and motorbikes with the same weight and size but different power sources are approximately comparable in performance. The E-Bicycle model requires a hub motor, throttle accelerator, gate for brakes and batteries along with solar panel- based charging circuit model. The battery is also charged by the system using solar energy. We are using a solar panel, a rechargeable battery with a long lifespan and large charge storage capacity, a reverse polarity preventer/corrector, an AC ripple neutralizer, and a unidirectional current controller.

Keywords: Solar panel, Hub motor, PV cells, Gate switch, Battery charger, E-bicycle, brake set

I. INTRODUCTION

B. Bachche Ajit and N. S. Hanapure conducted a study on fuel prices and noted a continuous daily increase in gas prices [1]. This rise in prices, coupled with escalating car pollution in urban areas and heavily trafficked regions, has prompted the search for alternative energy sources for vehicles. However, the affordability of expensive cars remains a challenge for many, particularly the economically disadvantaged. In response, efforts are underway to assist financially struggling individuals while addressing environmental pollution issues.

One solution proposed is the integration of solar-assisted bicycles equipped with DC hub motors. These motors, located in either the front or rear axle housing, are powered by batteries charged via solar radiation. The solar panels mounted on the bicycle carriage charge the battery when the bike is not in use. This setup aims to replace traditional gasoline engines, gearboxes, and fuel tanks found in conventional two-wheelers, or the chain sprocket mechanism commonly used by the public. Pedal-assisted solar hybrid bikes are modifications of traditional bicycles designed to enhance speed and rider comfort. Testing involved a 250W, 24V hub motor [2]. Additionally, Henry M. Gannon's study focused on multi- wheeled vehicles, excluding bicycles [3]. Their recommended setup includes a solar charging system, an electrical generator, and a standard bicycle equipped with a multi-speed gearbox. This preferred configuration allows for both pedaling and motor power, with the flexibility to use either or both as needed. The electrical system comprises a DC hub motor, lead acid batteries, a hand lever for throttle control, and a battery switching circuit. Furthermore, research was conducted on regenerative braking systems, which harness braking energy to charge the battery. When the brakes are applied, hydraulic cylinders compress fluid, releasing energy upon brake release for battery charging.

II. BLOCK DIAGRAM

In this chapter the block diagram of the project and design aspect of independent modules are considered. Block diagram is shown in fig: 1

1. Hub Motor

A wheel hub motor, also known as a wheel motor, wheel hub drive, or in-wheel motor, is an electric motor integrated directly into a wheel hub, providing direct propulsion. This innovative design eliminates the need for additional components like drive chains, sprockets, or brackets, enhancing efficiency and reliability.



FIG 1: Block diagram

In a typical wheel hub motor, permanent magnets are positioned within the stator, which is housed inside the rotor. When powered by batteries, the motor generates alternating current (AC), causing the hub to rotate, while the stator remains fixed to the axle. This configuration offers several advantages, including high efficiency, significant torque production at low speeds, and durability.

One of the key features of brushless DC machines, like wheel hub motors, is their ability to control speed over a wide and consistent power range. This controllability ensures optimal performance across various operating conditions, contributing to the motor's efficiency and versatility. Overall, wheel hub motors represent a robust and efficient solution for electric propulsion, offering reliability and performance in various applications.

2. Solar Panel/Cell:

A device that uses the photovoltaic effect to turn solar energy into electricity is called a solar cell or photovoltaic cell. When the source is unknown, the phrase "photovoltaic cell" is used, although the word "solar cell" is occasionally reserved for systems meant particularly to collect energy from sunshine. Solar panels, solar modules, and photovoltaic arrays are constructed from cell assemblies. The science and technology of solar cell applications for solar energy is known as photo voltaic.

3. Working

Solar power systems utilize photovoltaic cells to directly convert the radiant energy from sunlight into electrical energy. These cells are semiconductor devices that transform sunlight into electricity. Photovoltaic cells employing crystalline semiconductors like silicon offer high performance and reliability. They consist of silicon-based crystal wafers that generate a voltage across opposite surfaces when light strikes one of them, typically equipped with a current-collecting grid. When photons of light interact with these cells, they transfer their energy to the valence electrons of the semiconductor material, causing them to break free from their atomic bonds and transition to a higher energy state known as the conduction band, enabling them to move freely throughout the semiconductor.

4. Solar Panel Setup

When using batteries, installing a charge controller is an additional component that is required. Batteries last a lot longer if they aren't overcharged or drastically low on power. That's what a charge controller does. When the batteries are fully charged, the charge controller cuts off the current from the PV modules. Similarly, many charge batteries after they have been used to a certain, predetermined level that is determined by monitoring battery voltage. The controllers are unable to remove any more current from the batteries until they have been recharged. It is necessary to employ a charge controller for extended battery life. Apart from energy storage, the electricity generated by your solar panels and, should you choose

to utilize them, taken from your batteries isn't the same as what your utility company supplies or what your home's electrical appliances use[6]. To convert solar electricity from direct current to alternating current, an inverter is needed. Most large inverters allow you to configure automatic controls to regulate how your system operates. A tiny percentage of PV modules, also known as AC modules, have an inverter built in, which eliminates the need for a large central inverter and allays wiring worries.

5. Poly Crystalline Modules



Fig 2 :Polycrystalline module

Polycrystalline (multi-crystalline) modules are made up of multiple distinct crystals that have been fused together to form a single cell—thus the term "multi" As a result of the cheaper cost of producing the cells, they have long been the most widely used type of solar module. Mono crystalline have become increasingly popular in the residential market as their cost has decreased recently.

This metal flake effect', or visible crystal grain, is a result of the formation of these many crystals, as the figure (left) illustrates. Although they are less expensive to build than mono panels, their efficiency is lower (varying from 0.5% to 2%, depending on the manufacturer). This is due to the possibility of electron trapping at the crystal grain boundaries, which lowers efficiency.

6. Motor Controller

A motor controller, sometimes referred to as the vehicle's brain, is a crucial component of the solar hybrid bicycle. It regulates the power delivered to the hub motor, lights, and, if necessary, the horn. The motor controller is responsible for converting the battery's DC voltage into an alternating voltage that may be varied in frequency and amplitude to operate the hub motor at various speeds. Essentially, it is made up of a tiny microprocessor and

MOSFET transistors that can identify issues with the throttle, motor hall sensors, and other components. It can also guard against under voltage and excessive current.

In essence, a motor controller or charge regulator functions as a voltage and current regulator to prevent batteries from overcharging. Its primary role is to manage the voltage and current flow from solar panels to the battery, ensuring optimal charging conditions. Typically, batteries require approximately 14 to 14.5 volts to reach full charge.



Fig 3: Hub Motor Controller

7. Battery

Lead acid batteries are widely used in everyday life and are the most common type of battery used in various devices. Despite having a lower energy density compared to lithium-ion batteries, lead acid batteries are considered safe to use when proper precautions are taken. This makes them a preferred choice for applications such as solar hybrid bicycles due to their affordability, widespread availability, and resistance to explosions. The current supplied by a battery, measured in amperes (amps), indicates the rate of energy flow, with higher current ratings resulting in slower battery depletion.

The current rating is typically specified in terms of ampere- hours (Ah). This project focuses on the charging and discharging processes of high voltage batteries. Therefore, a battery with higher energy density, longer runtime, and lower weight relative to its high output voltage is required. Among the various types of batteries available, lead acid batteries are deemed most suitable for use in solar hybrid bicycles.



Fig 4 : Battery pack



Fig 5: Throttle

8. Throttle

The bicycle's maximum speed is 25 km/h, but this speed can be adjusted between 0 and 25 km/h depending on road conditions and traffic. To accommodate these variations, a throttle is integrated into the bicycle, allowing riders to increase or decrease speed as needed. Positioned on the right- side handlebar, the throttle is linked to a motor controller. The throttle comprises three distinct wires: black, red, and green. Typically, the red and black wires supply the necessary voltage, with a standard requirement of 4 volts for the accelerator, while the green wires facilitate acceleration.

9. Gate Switch

The gate switch serves as a primary switch that regulates the flow of direct current from the battery to the motor, operating manually. By pressing the button, a controlled current is permitted to pass between the Input and Output contacts of the Power switch, facilitating the pre-charging of capacitors within the controller and balancing the voltage on both sides of the power switch contacts. This pre-charging process ensures that the power switch can be activated without any arcing across the contacts, as they are already at the same voltage level.

In the OFF position, the input and output terminals remain disconnected within the switch. Conversely, in the ON position, all four contacts are effectively linked together through the two pairs of switched contacts housed within the switch, along with the parallel loop wiring connections on its exterior.

III. METHODOLOGY

• Initially, the DC Gear motor was attached to the rear wheel via a chain drive, which was secured onto the seat stay using nuts and bolts fixed on iron strips welded to the seat stay.

• The fabrication process was conducted with careful consideration given to the maximum load capacity that the motor could handle.

• Edge grinding was performed to ensure a smoother surface finish on both the freewheel and the iron strips.

- Welding was employed to connect the freewheel to the shaft of the DC motor.
- Subsequently, all the fabricated components were assembled together.
- The controller drive is powered by batteries, which are mounted on the handlebars.
- The controller drive is supplied power from the batteries and transmits signals/output to the HUB Motor.

III. WORKING

Initially, solar panels are mounted on the bicycle handlebars to capture sunlight and convert it into electrical energy. This energy is then stored in a rechargeable battery through a charge controller, with an additional option for direct wall charging using an adapter. Both sources of energy are directed through a unidirectional electronic circuit board and stored in the battery for later use. Activation of the hub motor is achieved by using a gate switch, while propulsion on the roadis controlled by the accelerator throttle. Additionally, braking is applied to the motor through a braking arrangement located in the front wheel.

A solar bicycle operates by utilizing battery power to drive its hub motor, which propels the bicycle forward. The battery is charged using solar energy, converted into voltage through photovoltaic cells. These cells, integral to solar bicycles, facilitate the conversion of solar energy into voltage required for battery charging. Lead acid and lithium-ion batteries are commonly employed in solar bicycles, each with its own advantages and disadvantages. Lead acid batteries, although heavier and shorter-lived, boast a higher current capacity. Conversely, lithium-ion batteries are lighter but offera longer lifespan. In essence, the fundamental concept of a solar-powered bicycle revolves around utilizing stored energy from a battery, charged either through solar panels or other means, to drive the motor, functioning as the vehicle's engine, and propel the vehicle forward.

IV. PERFORMANCE ANALYSIS

Hub Motor Calculation Motor Specification: Volt(V)=48 v, Power (P)=1000 w Power Equation Power (P) = Current (I) × Voltage (V) Hence, I = P \div V = 1000 \div 48 = 20.83 Amp. Speed of Motor In RPM

N=K÷ (<u>d</u> × 0.001885) =35÷(25.4×0.001885)=731RPM

Where, N = Speed In RPM, K = Speed In kmphd = Wheel Diameter in cm • Wheel Diameter (d) is 10 inch (Given)

1 inch = 2.54 cm, So, d = 10 inch = 25.4 cm

• Speed In kmph (K) is 35 kmph (Given)

Torque of the motor (T)

 $T = (P \times \underline{60}) \div (2 \times \pi \times N) = (1000 \times 60) \div (2 \times 3.14 \times 731) = 13.06 \text{Nm}$

Torque of the wheel hub motor, T = 13.06 Nm

Vehicle dynamics, including as rolling resistance, gradient resistance, aerodynamic drag, etc., must be taken into account when determining a vehicle's power rating. The process for choosing a motor rating for an electric scooter weighing 170 kggross is taken into consideration as an example.

The force required for driving a vehicle is calculated as: Ftotal=Frolling + Fgradient + Faerodynamic drag

Where, Ftotal=Total force

Battery Calculation

From motor calculation we get, Wattage=1000w, Voltage =48v,So, watt.hr = $1000w \times 1hr = 1000w.hr$ Out of the full battery 80% should in use and 20% should remaining. So, battery watt.hr = $1000w.hr \times 1.20$ =1200w.hrHence, Current (Ah)in battery=1200w.hr $\div 48v=25$ Ah

- Selection of battery charger
- Suppose we have to charge a battery in 5hr. So ourrequired wattage is 1200w.hr. According to above condition, Wattage of charger =1200w.hr÷ 5 hr=240w

Hence, current rating of charger =240w÷48v= 5A As per the above calculation to charge 48v,25Ah battery in5 hours, we require 48v,5A charger.

Battery specification`

Voltage Rating = 48v, Current Rating = 25 Ah

So, Wattage of battery = Voltage Rating \times Current Rating = 48 \times 25 =1200wh(watt.hr.)

v. RESULT AND TESTING

The "Design and Development of Solar Hybrid Bicycle" proposal aims to create a bicycle powered by renewable energy, specifically solar power. A successful riding test has been conducted, achieving a speed of 15 km/h. The bicycle completed a total round trip of 30 km on a single charge, with backup support from both solar panels and a dynamo.

VI. CONCLUSION

Today's fast growing gasoline vehicle usage contributes to increased air pollution. Because of its many

benefits—such as the fact that an electric scooter is an environmentally beneficial product—using EVs is essential to controlling this. A solar-powered bicycle is a modified version of an existing bicycle. It works well on cement, asphalt, or mudroads in cities and rural areas to cater to the need of economically week citizens. It is available for free usage for the whole year. This bicycle's ability to save millions of foreign currency, by not using pricey fossil fuels is its most significant feature. Since it produces no emissions, it is both environmentally friendly and pollution-free. Furthermore, it is silent and can be refilled in an emergency or during overcast weather using an AC adaptor. At about Rs. 0.70 per kilometer, the running cost is very low. In the event that the solar system malfunctions, it can be powered manually by pedaling. Because it is simpler to assemble and disassemble, it requires less upkeep. From the standpoint of a future energy system, it is critical to find new means of energy production and transportation, and solar-powered E-bike pools might be one of them. e-bikes are far more energy-efficient than a car, bus, or other big transportation vehicle.

VII. REFERENCES

- 1. Ajit B. Bachche, N. S. Hanapure 'Design and Development of Solarbicycle'volume2, issue2, December 2012,
- 2. IJHeneryGannon, 'Electricandpedaldrivenbicyclewithsolarcharging', p atentno. 5316101, May31 1994.
- GlennC.Streif^{*},25052CampoRojo,-LakeFomst^{*}Calif- 92630, 'SolarpoweredTwo –wheeled vehicle with Energy intensifying Solar Collector^{*}
- 4. V.B. Bhandari, Design of machine elements, T he McGraw. Hill Companies, Second Edition.
- 5. S.S. Ratan; Theory of Machine, the McGraw. Hill Companies, Second Edition.
- Ji-Young Byung- Chul Woo, Jong- MooKim and Hong Seok Oh "In- wheel Motor Design for an Electric Scooter" JElectr Eng Technol.Nov.2017
- Zhai Haizhou Heze University, Heze, China "Modeling of Lithium-ion Battery for Charging/Discharging Characteristics Basedon Circuit Model"Article submitted 05 March2017. Published as resubmitted by the author 17April 2017.
- 8. K.W.E Cheng, "Recent Development on Electric Vehicles", Department of Electrical Engineering, The Hong Kong Polytechnic University, Hong Kong. June2009.
- 9. Md. Rifat Hazari, Effat Jahan, "Design of a Brushless DC (BLDC) MotorController", Department of Electrical and Electronic Engineering American International University-Bangladesh (AIUB) Dhaka, Bangladesh.Oct2014.

Semi-automated Spacer Fitting Mechanism forHotline Maintenance

Vijay A Sawant¹, Dhanraj D Daphale², Sanjivani G Atakare³ Department of Electrical Engineering, SVERI's College of Engineering Pandharpur ³Department of Computer Engineering, SVERI's College of Engineering (Poly.) Pandharpur

> ¹vijaysawant1330@gmail.com ²dddaphale@coe.sveri.ac.in ³sgatakare@cod.sveri.ac.in

Abstract: In recent days, electrical sector has grownup apace and additionally voltage levels at that transmission to be done has additionally accumulated from 33KV to 1200KV. This has diode to the matter of doing the upkeep work as a result of the work should be performed on live line. it's undesirable to disconnect significant masses for maintenance work from grid because it might cause immense loss to the buyer similarly on the ability firms. Currently this work is finished by humans by hand with Hot stick technique and blank hand technique however these strategies conjointly has disadvantage. to beat this, we tend to developed a Spacer fitting mechanism to eliminate the requirement of human hand in Spacer fitting operation. With this mechanism the spacer fitting operation becomes quicker and easier. With the assistance of IR (Infra- Red) sensing element and Arduino Uno microcontroller board with ATMEGA328P controller, the position of conductor is perceived and command is shipped to perform fitting of spacer. The position of arms is adjusted by DC motors to perform the operation properly

Keywords: ATMEGA328P (Microcontroller), Arduino Uno (Board), IR sensor (Infra-Red), Hotline.

I. INTRODUCTION

In engineering, maintenance work has got to be performed whereas line is energized. the primary techniques for live-line operating were developed within the early years of the twentieth century, with growth in industrial sector from 19thcentury; it junction rectifier to drawback of maintenance work as a result of it had been troublesome to perform work while not interrupting provide to the patron. within the Sixties, strategies were developed within the laboratory to modify field staff to come back into direct contact with high voltage lines. Such strategies will be applied to modify safe work on the best transmission voltage

1.1 Meaning of Hotline Maintenance "Performing the upkeep work whereas the road is energized is termed as hotline maintenance" the work involves the fitting of spacer, replacement of insulators, scrutiny of transmission lines and improvement of insulators. These works area unit principally performed by hand with the assistance of 2 strategies that 2 strategies area unit as given below:

- 1) Hot stick technique
- 2) Bare Hand Technique

1.1.1 Hot stick Technique

In this methodology the employee carries a hot stick (Hot stick could be a insulated with a high voltage rating) in his hand and perform the upkeep work by keeping his body at ground potential and utterly isolated from line potential.

1.1.2 Bare Hand Technique

In this technique employee wears a Faradays suit (which is totally electrically conductive) furthermore as insulated gloves and shoes. He then reaches nearer to conductor by insulated ladder or insulated floor of eggbeater. The employee holds a charge current stick-on conductor and attracts the potential from conductor into his body. supported principle that whenever 2 same charges are placed along, they suffer neither attraction nor repulsion. thus, employee suffers no shock from high voltage line and might perform the work safely.

II. METHODOLOGY

By introducing automaton into power sector, the risks concerned is also decreased, however the automaton ought to fulfill the subsequent requirements:

- It ought to cut back involvement of human hand.
- It mustn't build contact between ground potential and line potential.
- Robot ought to be straightforward to control. Robot ought to give most safety to operator. It mustn't cause any hurt to line.
- Spacers fitting operation ought to be sleek
- 2.1 Block Diagram of Spacer Fitting Mechanism



Figure 2.1: Block Diagram of Spacer Fitting Mechanism

Above figure shows the diagram for hotline maintenance by automaton within which the12V DC power offer is feed to microcontroller, microcontroller generates PWM pulses that cause the rotation of dc motor above all degrees. thanks to rotation of DC motor automaton arms can move (as the motors are literally fitted in robot-arms to cause movement). there'll be a tool arrangement at the tip of arm for doing maintenance work. dominant unit controls the automaton arm movement by giving commands to the microcontroller. Sensing unit can find the position of robot-arm. The torsion rating for dc motor is typically 14Kg/m2 to 16Kg/m2 this torsion rating is spare for automaton arm movement, point feedback additionally plays a very important role in movement of robot-arm because it improves the steadiness of operation. The motor driver IC is employed to drive the motors employing a 5v signal supply and might additionally run the motors in each reverse and forward directions. This IC may also offer the 12V to the dc motors although the controller voltage is of 5V. The speed of the motors in each forward and

III. IMPLEMENTATION



reverse direction is controlled victimization the microcontroller.

Figure 3.1: Side View of Project Setup

Three different movements have been employed in this system. They can be titled like:

- 1. Waist Movement
- 2. First Arm Movement
- 3. Second Arm Movement

1. Waist Movement

The waist of the robotic arm is controlled by the machine- controlled box. A string is hooked up to the spacer fitting box and tied to the shaft of motor articulated at the bottom of setup and middle of the string is hooked up to the machine at the middle. By movement of motor the waist conjointly moves.

2. First Arm Movement

Another DC motor is as well as a worm reducer, that contains a reduction quantitative relation of 100:1. The worm reducer can transmit the facility to the shaft. This shaft is connected to the primary (lower) arm. thus, if the shaft rotates then the primary arm can rotate consequently.



Figure 4.1: Flow Chart for Conductor Detection Program Spacer

3. Second Arm Movement

Second arm holds each the machine on one facet and spacer fitting box on the opposite facet. A string is hooked up to the box at one finish and to the motor shaft on the opposite finish through machine. once the motor purposed for second arm movement is turned "ON", string gets wounded on motor shaft which ends up in propulsion of spacer fitting box inflicting upward and downward movement of spacer fitting box.

4. Flowchart for conductor detection program

In case of high voltage cable, the spacer fitting is very important because the conductors square measure command along moreover as they're protected against the high winds. There square measure many varieties of spacers. they're principally classified in keeping with variety to conductors they hold, the virtually comes all told shapes and sizes in keeping with the necessity of cable. The transient classification of those is also given as follows:

- 1. Two conductor spacers
- 2. Three conductor spacers
- 3. Multiple conductor spacers

1. Two conductor spacers



Figure 4.2: Two Conductor Spacer

Spacer, who is employed to carry 2 conductors along, may be a 2-conductor spacer. sometimes this spacer is used in high voltage line higher than 33kv. it's conjointly referred to as as twin spacer.

Two conductor spacer is as shown in on top of image, within which it's 2 components joined along by bolt and nut arrangement. Its length is twelve.5cm. This spacer is fitted to cable by employee with hands. In Bharat most of the transmission lines area unit fitted with 2 conductor spacers.

For our spacer fitting mechanism, we tend to created spacers by ourselves that square measure created from fiber material and like spacer shown in on top of figure it's conjointly joined along by bolt and nut.



Figure 4.2: Two Conductor Spacer

Two conductor spacer is as shown in on top of image, within which it's 2 components joined along by bolt and nut arrangement. Its length is twelve.5cm. This spacer is fitted to cable by employee with hands. In Bharat most of the transmission lines area unit fitted with 2 conductor spacers.

For our spacer fitting mechanism, we tend to created spacers by ourselves that square measure created from fiber material and like spacer shown in on top of figure it's conjointly joined along by bolt and nut.

Operation of Spacer Fitting Mechanism

Arduino UNO boards are specialized in robotic applications, therefore may be employed in nearly robotic applications it's special ports for driving up to four DC motors. 2 ports for driving Servomotors and extra five sensors may be connected to the board. it's further thirteen ports for connecting input/output devices. during this project we tend to used it forconductor position detection and screw modification mechanism (Spacer fitting Mechanism).

The operation of the Spacer fitting mechanism is as follows:

- 1. The spacers are fed into the spacer fitting box.
- 2. One part of spacer is set on upper part of box and other part of spacer on the lower part of spacer fitting box.
- 3. The setup is moved close to the transmission line.
- 4. The spacer fitting box is opened.
- 5. The opened box is inserted into the transmission line.
- 6. The sensor set for conductor detection detects the conductor and gives signal to the ATMEGA328P (which means conductors are in position)
- 7. The spacer fitting box is closed and motor placed at the bottom of the box stars rotating which holds a Nut.
- 8. Due to this the spacer gets tightened as both the Nut and bolt gets tightened due to rotation of Nut.
- 9. After successful operation of spacer fitting.

IV. RESULTS & CONCLUSION

The above work deals with high voltages applications hence it must be insulated properly. The insulation rating for the given spacer fitting setup is determined as follows.

Vrated= K*Vtransmission

Where,

Vrated- Rated voltage for insulation of given setup. **Vtransmission**- Voltage of the transmission line.

K- (2 or 2.5) number constant relying upon the atmosphericphenomenon.

V. CONCLUSION

In India most of the high voltage lines are fitted with 2 conductor spacers therefore this mechanism is appropriate forpretty much all such style of high voltage lines. it's finished from the work that the area fitting operation may be performed safely by eliminating human hand and putting artificial mechanism to perform the task. It conjointly reduces the value of operation because it reduces variety of employees required for the operation. Operation is simple and need less time for coaching.

Advantages

The spacer fitting mechanism has following benefits overstandard technique of spacer fitting by human hand;

- Inspection and repaired easily.
- Reduces the risk of loss.
- Reduces the number of workers.
- Precise monitoring and performance.

- Spacer fitting.
- Takes less time for spacer fitting than current method of spacer fitting.
- Provides complete isolation between line potential and ground potential as well as operator.
- Easy to operate.

VI. REFERENCES

- Ren Zhibin , Ruan Yi, et al. (2008). Control of inspection robot for the power transmission lines based on database, Proceedings of the 27th Chinese Control Conference, pp.281–285, Kunming, China, 2008, IEEE, USA.
- 2. Visual servo- b a s e d line grasping control for power transmission line inspection robot, robot, Vol. 29, 2007
- 3. C. W. Devine, "Method and apparatus for testing insulators," US Patent, no. 4,266,184, May 5, 1981.
- 4. G. H. Vaillancourt, J. P. Belle rive, M. St-Jean, and C. Jean, "New live line tester for porcelain suspension insulators on high-voltage power lines," IEEE Transactions on Power Delivery, vol. 9, no. 1, 1994, pp. 208–219.
- 5. Artigue, G. Thomann, Development of a Prosthetic Arm: Experimental Validation with the User and Adapted Software, IEEE International Conference on Robotics and Automation, pp. 3079- 3084, Japan, May 2009
- 6. S. Schulz, C. Pylatiuk and G. Bretthauer, A New Ultra- Light Anthropomorphic Hand, Proceedings of IEEE International Conference on Robotics and Automation, Vol. 3, pp2437-2441, Korea, May 2001
- Ying-Shieh Kung, Gua-ShiehShu, "Development of a FPGA-based Motion Control IC for Robot Arm", pg.no. 1397 to1402, IEEE 2005
- 8. Mrs. Urmila Meshram (Thulkar), Mr. pankaj Bande, Mr. P.A. Dwaramwar, and Mrs. R. R. Harkare, "Robot Arm Controller Using FPGA", IEEE impact2009.

Analytical Approaches for Analysis of Sub- Synchronous Resonance (SSR) In PMSG based Wind Farm under Weak Grid: A Review

¹Prakash D. Kadam, ²Sampath Kumar Bodapatla, ³Sagar M. Ghodake, ⁴R. T. Bansode ^{1, 3}, ⁴FTC"s College of Engineering and Research, Sangola, Maharashtra, India ²N. K. Orchid College of Engineering & Technology, Solapur, Maharashtra, India

Abstract- The integration of renewable energy sources, specifically wind farms utilizing Permanent Magnet Synchronous Generators (PMSGs), presents unique challenges when operating in regions with limited grid infrastructure. One of the critical issues that can arise is Sub-Synchronous Resonance (SSR), which can impact the stability and performance of the wind farm. This research paper offers a comprehensive exploration of analytical approaches aimed at addressing SSR in PMSG-based wind farms operating under weak grid conditions. The study commences by developing a detailed dynamic model that captures the complex resonance between the PMSG-based wind farm and the weak grid. Modal analysis is then utilized to identify the key modesthat contribute to SSR, with a particular focus on evaluating the involvement of different system components in these modes. The paper also investigates strategies to enhance system damping, such as the implementation of damping controllers and power system stabilizers. Furthermore, the study delves into fault ride- through capabilities and dynamic reactive power control strategies to ensure the wind farm's resilience during transient events. Advanced control algorithms, including adaptive and intelligent control, are explored for their potential to improve the overall system response. Additionally, the design of robust control strategies to handle uncertainties in wind speed and grid conditions is discussed. To validate the proposed analytical approaches, simulation studies are conducted, highlighting their effectiveness in mitigating SSR, and enhancing the integration of wind farms with weak grids. The findings emphasize the importance of collaboration with system operators and adherence to grid codes to ensure the stable and reliable operation of PMSG-based wind farms in challenging grid environments.

Index Terms-Wind farm, Permanent magnetic synchronous generator, Sub-synchronous resonance.

I. INTRODUCTION

Wind power has made significant progress in recent years as a clean and sustainable energy source. The development of variable speed wind turbine generators (WTGs), such as the doubly-fed induction generator (DFIG), has played a crucial role in this advancement. [1], [2].

Direct-drive permanent magnetic synchronous generators (PMSGs) [3], [4], are extensively utilized because of their exceptional control capabilities. When connected to the grid through fully controlled voltage source converters (VSCs), direct-drive PMSGs exhibit remarkable on-grid operating performance [5],[6]. This includes high operational efficiency, seamless integration with the grid across a wide range of speeds, and excellent fault-ride-through ability. As a result, they are increasingly being adopted in newly-established windfarms.

Concern for both academia and industry [7]-[9]. This problem arises from the interaction between the control system of Doubly Fed Induction Generators (DFIG) and fixed series compensation (FSC), resulting in the disconnection of multiple WTGs and damage to their crowbar circuits [9], [10]. Similar incidents of SSR have been observed in wind farms [8], [11]. This particular type of SSR is also referred to as sub- synchronous control interaction (SSCI) [12] or sub-synchronous interaction (SSI) [13] due to the active involvement of WTG control in the development of unstable sub-synchronous power oscillations. It is worth noting that previously reported SSR/SSI events have predominantly occurred in induction-generator-based WTGs, particularly type 3 or DFIGs [14], when they are connected to series-compensated transmissions [15].

The question of whether type-4 WTGs or direct-drive PMSGs would also be affected by unstable SSI remains unanswered. Numerous studies have been conducted on the interaction between AC networks and general VSCs [16]-[18]. For example, reference [16] highlights that constant power loads connected to tightly regulated converters exhibit negative impedance characteristics, which can lead to system instability. Reveals [17] that the coupling between phase-lock loops (PLLs) of neighboring converters can increase the system's susceptibility to oscillations. The research presented in [18] demonstrates the presence of series/parallel resonant circuits in grid-connected VSC systems. However, the instability caused by negative impedance, PLL coupling, or resonant circuits primarily pertains to high-frequency

oscillations rather than sub-synchronous interaction.

There were limited studies specifically addressing the influence of PMSG on SSI. For example, one reference [19] stated that PMSGs can contribute to positive damping in SSI mode. Another study concluded [20] that PMSGs do not exhibit any apparent vulnerability to SSI, and in fact, theymay even be beneficial for SSI immunity. However, [21]-[22], a different study declared that there is a potential risk of SSI due to the negative damping caused by PMSGs. It is important to note that these previous investigations were conducted using computer models rather than real-world observations in an actual power system. Furthermore, no information is provided regarding the decisive factors and dominant characteristics of SSI associated with PMSGs.

Despite the lack of a definitive conclusion regarding the potential for SSI caused by direct-drive PMSGs, practical instances of real oscillatory events have already occurred. The wide-area measurement system (WAMS) has repeatedly detected sustained power oscillations at sub-synchronous frequency since June 2014, as the installed capacity of wind generation continues to rapidly increase. These oscillations originate from direct-drive PMSG based wind farms and extend to the external power grids. In some cases, the amplitude of the oscillatory power surpasses that of the fundamental power. In a severe incident, these oscillations even induce intense torsional vibration in nearby turbo generators, leading to the trip of all generators in a powerplant due to its torsional protection system. Investigations into the system reveal the absence of series compensation in the vicinity. Furthermore, the sub-synchronous power oscillation differs from previously reported cases induced by constant power loads, PLL coupling, or resonant circuits. It represents a novel form of sub-synchronous interaction between type-4 WTGs and their connected weak AC networks. Consequently, the existing research falls short in explaining its fundamental mechanism. Further efforts are required to comprehend its key characteristics and dominant influential factors. This study examines the operational principles and distinctive features of the newly emerging SSI, which is caused by the interplay between PMSG-based wind farms and vulnerable AC networks.

The purpose of this paper is to review and analyze the diverse analytical approaches proposed in the literature to mitigate Sub-synchronous Resonance in PMSG-based wind farms. However, aim to provide a structured overview of different methodologies, including their theoretical basic, merits and demerits, and comparative analyses.

In the paper is organized as follows. Following the Section I with discus about Introduction, Section II presents Modelling of PMSG Wind Farm, Section III presents analysis methods for SSR and Section IV presents comparison of different analysis methods for SSR damping in PMSG based wind farm. Section V concludes this paper.

II. MODELING OF PMSG WIND FARM

Wind power is often generated in remote areas, far from where the electricity is needed. As a result, the electricity generated by wind farms needs to be transmitted over long distances to reach the main power grid. These remote wind farms are typically connected to weak AC systems with low short circuit ratios and small local loads [23]. To account for these practical conditions, a simplified system model, as shown in Figure 1, is used for this study. The model includes a large wind farm (identified as A), a local transmission (B), a long-distance transmission (C), and an equivalent representation of the main grid (D). Additionally, there maybe a turbo generator (E) connected to the main grid.



The wind farm is expected to consist of identical type-4 WTGs. This means that all WTGs have the same capacity, structure, parameters, operating status, and are connected to the same bus. This assumption is reasonable and beneficial for simplifying the analysis without sacrificing its generality, as wind farms typically use the same type of WTG and operate under similar conditions. The local transmission system includes a step-up transformer T2 and a 110 kV line with an impedance of rL1+jxL1. Through another step-up transformer T3, the local 110 kV grid is connected to a long-distance 220 kV line with an impedance of rL2+jxL2. It is likely that a coal-fueled turbine generator is located near the main grid. The electricity generated by both the wind farmand the thermal power plant is then transferred to the main grid, which is represented by an ideal voltage source with a reactance of xL3.

The provided diagram in Figure 2 illustrates the setup of a standard 1.5 MW direct-drive PMSG. This configuration is specifically designed by the manufacturer of the wind turbine generator (WTG) utilized in the aforementioned wind farm. The components of this setup include a wind turbine, a permanent magnet synchronous generator (PMSG), a machine-side converter (MSC) consisting of a diode rectifier and a boost DC/DC converter, a DC link, a grid-side converter(GSC) utilizing pulse width modulation (PWM) voltage source conversion, and an LC filter.

The control characteristics of the grid-connected direct-drive permanent magnet synchronous generator (PMSG) are primarily determined by the control features of its grid-side converter (GSC), as depicted in Figure 3 and detailed in Table II. The GSC controller employs a vector control strategy in the dq reference frame. The outer-loop controls are responsible for regulating the DC voltage and reactive power, while the inner-loop controls handle the d- and q-axis currents.Proportional-integral (PI) controllers are utilized in these control loops.



Fig: 2 Configuration of a 1.5 MW direct-drive PMSG



Fig. 3 Control diagram of the GSC Symbols In Control Blocks of the GSC

Symbols	Meaning	
i _{sd} , i _{sq}	d- and q- axis component of GSC output current	
vsd, vsq	d- and q- axis component of GSC output voltage	
isdref, isqref	Reference values for <i>isd</i> , <i>isq</i>	
vgd, vgq	Reference values for <i>vsd</i> , <i>vsq</i>	
vdc, vdcref	DC voltage and its reference value	
Qgref	Reference value of GSC output reactive power	
<i>K</i> P <i>dc</i> / <i>K</i> P <i>g1</i> / <i>K</i> P <i>g2</i>	Proportional gains	

$K_{\mathrm{I}dc}/K_{\mathrm{I}g1}/K_{\mathrm{I}g2}$	Integral gains
KD	Decoupling gain

III. DIFFERENT METHODS OF ANALYSIS OF SUB- SYNCHRONOUS RESONANCE (SSR) IN PMSG BASED WIND FARM UNDER WEAK GRID

I. Impedance Based Stability Analysis

The following section introduces a fresh stability criterion for SSI, which relies on the impedance model (IM) in the dq reference frame. Additionally, it demonstrates the coherence between the current IM-based analysis and eigenvalue analysis.

A. The dq Reference Frame Based IMs of PMSG and AC Grid

According to the documentation provided in [21]-[22], the state equation for a particular Permanent Magnet Synchronous Generator (PMSG) within the system can be alternatively represented as equation (1). Assuming that the control vector for the terminal current is dq = [isd ; isq], and the output vector for the terminal voltage is vsdq = [vsd ; vsq], the subsequent expression can be derived.

$$\Delta \dot{x} = A\Delta x + B\Delta isdq$$

$$\Delta vsdq = C\Delta x + D\Delta isdq$$
(1)

By utilizing Laplace transform and performing certain manipulations on equation (3), we can derive the correlation between the terminal current and voltage of a PMSG, specifically referred to as its IM.

The 2×2 impedance matrix, denoted as ZPMSG (s) = [ZPdd(s) ZPdq(s); ZPqd(s) ZPqq(s)], represents a polynomial expression of Laplace operators in each of its elements.

Likewise, the IM for AC grid illustrated in Figure 1 (comprising of T1, B, C, and D) can be acquired within the dq reference frame [24].

Where Z_{Line}(s) is a 2×2 impedance matrix;

 $r\Sigma = rL1 + rL2$; $X\Sigma = xL1 + xL2 + xT2 + xT1/n$; $\omega 0 = 1$ pu. Ultimately, the entire system's IM can be represented by.

ZT(s) = ZPMSG(s) / n + ZLine

Where $Z_T(s)$ is a 2×2 impedance matrix.

The reduction in SCR has a minor impact on the negative resistance of ZPMSG(s), but it significantly alters the impedance of ZLine(s). Both eigenvalue analyses and time- domain simulations have demonstrated that reducing the SCR to a low value would lead to unstable SSI. Hence, the unstableSSI is caused by the dynamic interaction between the PMSGs with negative-resistance effect and the vulnerable AC system.

(4)

B. Impedance based approach Criterion

The representation of the IM of a PMSG cannot be done using complex vectors [11] due to the mirror frequency coupling [25] caused by the unequal structure in the d- and q- axis as shown in Fig. 3. Consequently, the stability of SSR needs to be evaluated by considering the impedance matrices of (2) and (3), which can be accomplished using either the generalized Nyquist criterion [24] or the norm-based criterion [16]. However, these criteria can only provide qualitative outcomes, indicating whether the system is stable or unstable, rather than quantitative ones. Furthermore, the detailed explanation for the instability of the system is not provided. In order to address these concerns, an alternative stability criterion is proposed, which is based on the impedance- frequency characteristics of the determinant of the impedance matrix. If a disturbance voltage source V(s) of the mentioned SSI mode is introduced into the AC grid, such as at the step- up transformer T2, a corresponding current of the same mode can be generated and represented by the following expression:

$$I(s) = V(s) Z^{-1}$$
⁽⁵⁾

$$Z^{T-1(s)}$$
, namely the zeroes of the determinant of $Z_{T(s)}$ [32]. The determinant can be obtained by
 $D(s) = \det(Z_T) = Z_T dd \ Z_T dd \ - \ Z_T dq \ Z_T qd$
(6)

D(s) represents the determinant of ZT(s), which is an expression in the form of a polynomial of s. ZTdd, ZTdq, ZTqd, and ZTqq are four components within ZT(s).

The fact that the SSI mode in question is one of the decisive zeros should be emphasized. The concerned zero can be calculated using either an analytical or numerical approach. In this section, we have chosen to utilize the proposed aggregated RLC circuit model method [11].

The small-signal models used in both IM based analysis and eigenvalue analysis are the same, as mentioned above. Consequently, the eigenvalues obtained from equation (1) are equal to the zeroes of the determinant of the impedance matrix ZT(s). In other words, any changes in the operating conditions of the target system would impact the IMs of PMSGs and AC grid, thereby influencing the zeroes of the IM determinant or its eigenvalues for the entire system.

II. Eigen Value based Analysis

Eigenvalue analysis is a widely employed analytical technique that relies on a mathematical model of the system being studied, utilizing a collection of linear ordinary differential equations. The differential equations can be expressed in the standard form as follows:

$$\dot{\mathbf{x}} = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{u}$$

The eigenvalues can be defined as the solution to the equation: Det $[\lambda U - A] =$

The eigenvalue vector of the differential equation, denoted by λ , serves as a clear indicator of the frequencies and damping of oscillation modes. This allows for an easy understanding of the relationship between various system parameters, including the series compensation level, PI parameters of the current controller, wind speed, and number of wind turbine generators, and the characteristics of oscillation. By observing the movement of eigenvalues [26, 27], valuable insights into the oscillatory modes can be obtained. In order to study sub-synchronous interactions, D.H.R. Suriyaarachchi et al. propose a two-step procedure. In step 9, a frequency scanning method is employed, while step 10 involves a detailed small signal eigenvalue analysis. These steps provide crucial information about the oscillatory modes [28]. Ref. [29] utilizes participation factor analysis to confirm that the characteristics of the modes are influenced by various PMSG parameters, such as the GSC controller, PLLs, LCL filter, and controller parameters of VSC-HVDC. However, it is important to note that the eigenvalue analysis approach necessitates a comprehensive representation of both the mechanical and electrical systems. Consequently, higher order models, beyond the 2nd or 4th order models, are required. Unfortunately, the model parameters for these higher order models are not readily available [30].

(8)

(7)

III. Modal Analysis

During operation, the system parameters of the generator, such as inductance and resistance, control system parameters, and operating conditions like machine speed and load power, undergo variations. These changes have an impact on the system eigenvalues. To avoid repetitive calculations of the eigenvalues, it is necessary to conduct a sensitivity analysis in order to estimate the tendency of eigenvalue changes. This section focuses on a thorough modal analysis that encompasses participation factor and eigenvalue sensitivity.

The eigenvalues obtained from the state matrix can be either real or complex. Complex eigenvalues can be represented as conjugate pairs, as shown below.

$$\lambda_{\mathbf{i}} = -\sigma_{\mathbf{i}} \pm \mathbf{j}\beta_{\mathbf{i}} \tag{9}$$

The damping ratio and oscillation frequency corresponding to mode λi can be defined as

$$\xi_i = \frac{\sigma_i}{\sqrt{\sigma_i^2 + \beta_i^2}}, \quad f_{\text{osci}} = \frac{|\beta_i|}{2\pi}.$$
(10)

Participation factor analysis assists in determining the impact of each state variable on a specific mode. It quantifies the relative involvement of the kth state variable in the ith mode. The computation of participation factors can be derived from the eigenvector matrices using the following procedure:

$$p_{ki} = \frac{|w_{ki}||v_{ik}|}{\sum_{k=1}^{n} |w_{ki}||v_{ik}|}$$
(11)

where w_{ki} and v_{ik} are the k_{th} elements in the left and right eigenvector corresponding to the i_{th} eigenvalue. In general, the participation degree of the k_{th} variable in the i_{th} mode is measured by p_{ki} . As discussed in [32], p_{ki} can also be regarded as the sensitivity of λ_i with respect to the k_{th} diagonal element of state matrix [31].

The calculation of eigenvalue sensitivity allows us to quantify the speed and direction of mode movement caused by changes in system parameters. To determine the eigenvalue sensitivity of a mode (λ_i) in relation to an uncertain parameter μ , the following procedure can be followed:

$$\frac{\partial \lambda_i}{\partial \mu} = \frac{w_i^T (\partial A / \partial \mu) v_i}{w_i^T v_i} \tag{12}$$

where w_i and vi are the left and right eigen vectors corresponding to the eigenvalue λ_i , respectively.

IV. COMPARATIVE ANALYSIS

Analyzing Sub- Synchronous resonance (SSR) in power systems involves various methods, each with its own merits and demerits. Here's an overview of some common analysis methods along with their advantages and disadvantages:

Table No: 01 Merits & Demerits of Different Analysis Methods

Sr. No.	Analysis method	Merits	Demerits
		- It provides a comprehensive	- The accuracy of impedance-based
		understanding of the interactions	analysis is sensitive to variations in
		between different components in a	model parameters.
1		power system, including generators,	- Developing accurate impedance
1	Impedance- Based	transformers, and controllers.	models can be complex, particularly in
	Model Analysis	- It allows for a detailed assessment	large and interconnected power
		of system stability under various	systems.
		operating conditions.	- Impedance based analysis is
		- The approach helps in identifying	inherently based on linear system
		weaknesses in the power system that	assumptions.
		might be prone to SSR.	- Impedance-based analysis may have
		- Impedance- based analysis is	limitations in capturing interactions at
		valuable for evaluating the impact	veryhigh or very low frequencies.
		of different controllers on	
		sub-synchronous modes.	
		- It provides a global perspective on	- Eigenvalue-based analysis relies on
		system stability.	the assumption of alinear system.
	Figen value	- Eigenvalue analysis provides	- The accuracy of eigenvalue-based
2	Analysis	information about the mode shapes	analysis is sensitive to the accuracy
	r mary 515	associated with different eigenvalues.	of the system model.
		- Engineers canperform sensitivity	- It may not be effective in capturing
		analysis by varying system	interactions at veryhigh or very low
		parameters and observing changes	frequencies.
		in eigenvalues.	- It might not fully capture the impact
		- Eigenvalue analysis can be used to	of power system stabilizers and other
		reduce the complexity of the system	controllers on system stability.
		model.	Interpreting eigenvalue analysis
		-Eigenvalue analysis can serve as an	results can be complex, especially
		early warning system for potential	in large interconnected systems.
		stability issues.	
		- It provides detailed information	- it is based on the assumption of a
		about the mode snapes associated	The accuracy of the laboratory in the second
3	Modal Analysis	with different natural frequencies.	- The accuracy of modal analysis is
		- it helps identify critical modes that	sensitive to the accuracy of the
		it allows for the avamination of the	system model.
		- it allows for the examination of the	-n can become complex and
		disturbances	computationally intensive,
		It results can be used for model	power systems
		validation and verification	It assumes that the system is linear
			and time-invariant.

V.CONCLUSION

A novel form of sub-synchronous resonance has been identified in large-scale direct-drive PMSG based wind farms. In order to examine this matter, a simplified yet representative model has been devised to simulate multiple PMSGs connected to a feeble AC network. The interaction between the wind farms and feeble AC networks can lead to an under- damped SSI mode. In specific scenarios, particularly when the short circuit ratio (SCR) is exceedingly small, the SSI can become unstable and result in sustained sub-synchronous power oscillation in wind turbine generators (WTGs) as well as the entire grid system. The frequency and stability of this SSI mode are contingent upon various factors, with the SCR (determined by the grid strength and WTG online capacity) and the control strategies and parameters of WTG converters exerting the most significant influence. The power oscillation is primarily induced by the negative-resistance effect at the SSI frequency, which arises from the resonance between direct-drive PMSGs and feeble AC grids. To tackle the unstable SSI, an additional sub-synchronous damping control loop has been implemented in the PMSG controllers. Moreover, several other potential countermeasures are recommended for further investigation.

VI. REFERENCES

1. A. Mitra and D. Chatterjee, "Active power control of DFIG-based wind farm for improvement of transient stability of power systems," IEEE Trans. Power Syst., vol. 31, no. 1, pp. 82-93, Jan. 2016.

2. M. H. Variani and K.Tomsovic, "Two-level control of doubly fed induction generator using flatness-based approach," IEEE Trans. Power Syst., vol. 31, no. 1, pp.518-525, Jan. 2016.

3. Y. Wang, J. Meng, X. Zhang, and L. Xu, "Control of PMSG-based wind turbines for system inertial response and power oscillation damping," *IEEE Trans. Sustain. Energy*, vol. 6, no. 2, pp. 565-574, April 2015.

4. E. Giraldo and A. Garces, wind energy conversion system based *on PWM-CSC and PMSG*," *IEEE Trans. Power Syst.*,vol. 29, no. 3, pp. 1446-1453, May 2014.

5. K. W. Hu and C. M. Liaw, "Development of a wind interior permanent- magnet synchronous generator-based microgridand its operation con-trol," IEEE Trans. Power Electro., vol.30, no. 9, pp. 4973-4985, Sept. 2015.

6. Y. Zhou and H. Nian, "*Current zero crossing duration reduction of a semi-controlled open winding PMSG system based on third harmonic current injection*," *IEEE* Trans. Industrial Electro., vol. 63, no. 2, pp. 750-760, Feb. 2016.

7. Subsynchronous Resonance Working Group of the System Dynamic Performance Subcommittee, "*Reader*"s guide to subsynchronous reso-nance," IEEE Trans. Power Syst., vol.7, no. 1, pp. 150-157, Feb. 1992.

8. L. Wang, X. Xie, Q. Jiang, H. Liu, Y. Li, and H. K Liu, "Investigation of SSR in practical DFIG-based wind farms connected to a se-ries-compensated power system," IEEE Trans.Power Syst., vol. 30, no. 5, pp. 2772-2779, Sept. 2015.

9. H. Mohammadpour, E. Santi. "SSR damping controller design and op-timal placement in rotor-side and gridside convertors of se-ries-compensated DFIG-based wind farm," IEEE Trans. Sustain. Energy., vol. 6, no. 2, pp. 388-399, Apr. 2015.

10. J. Adams, V. A. Pappu, and A. Dixit, "*ERCOT experience screening for Sub-Synchronous Control Interaction in the vicinity of series capacitor banks*," in Proc. 2012 IEEE PES General Meeting, pp.1-5.

11. H. K. Liu, X. Xie, C. Zhang, Y. Li, H. Liu, and Y. Hu, "Quantitative SSR analysis of series-compensated DFIG based wind farms using aggregated RLC circuit model," IEEE Trans. Power Syst., vol. 32, no. 1, pp. 474-483, Jan.

2017.

12. Y. Cheng, N. Sahni, D. Muthumuni, and B. Badrzadeh, "*Reactance scan crossover-based approach for investigating SSCI concerns for DFIG-based wind turbines*," IEEE Trans. Power Del., vol. 28, no. 2, pp. 742-751, Apr. 2013.

13. A. E. Leon and J. A. Solsona, *"Subsynchronous interaction damping control for DFIG wind turbines,"* IEEE Trans. Power Syst., vol. 30, no. 1, pp. 419-428, Jan. 2015.

14. Einar V. Larsen, "*Wind generators and series- compensated AC trans-mission lines*," in Proc. 2012 IEEE PES General Meeting, pp.1-5.

15. A. Adrees and J. V. Milanovic, "Optimal compensation of transmission lines based on minimisation of the risk of subsynchronous resonance," IEEE Trans. Power Syst., vol. 31, no. 2, pp. 1038-1047, Mar. 2016.

16. Z. Liu, J. Liu, W. Bao, and Y. Zhao, "Infinity-norm of impedance-based stability criterion for three-phase AC distributed power systems with constant power loads," IEEE Trans. Power Electro., vol. 30, no. 6, pp. 3030-3043, June 2015.

17. B. Wen, D. Boroyevich, R. Burgos, P. Mattavell, and Z.Shen, "*Analysis of D-Q small- signal impedance of grid-tied inverters*," IEEE Trans. Power Electro., vol. 31, no. 1, pp. 675-687, Jan. 2016.

18. X. Wang, F. Blaabjerg, M. Liserre, Z. Chen, J. He, and Y. Li, "*An active damper for stabilizing power-electronics- based AC system*," IEEE Trans. Power Electro., vol. 29, no.7, pp. 3318-3329, Jul. 2014.

19. E. V. Larsen, *"Wind generators and series- compensated AC transmission lines,"* in Proc 2012 IEEE PES Transmission and Distribution Conference and Exposition(T&D), pp.1-4.

20. H. T. Ma, P. B. Brogan, K. H. Jensen, and R. J. Nelson, "Sub-synchronous control interaction studies between full-converter wind turbines and series- compensated AC transmission lines," in Proc. 2012 IEEE PES General

Meeting, pp.1-5.

21. K. M. Alawasa, Y. A. I. Mohamed, and W. Xu, "Modeling, analysis, and suppression of the impact of fullscale wind- power converters on sub-synchronous damping," IEEESystems Journal, vol. 7, no. 4, pp. 700-712, Dec. 2013.

22. K. M. Alawasa, Y. A. I. Mohamed, and W. Xu, "Active mitigation of subsynchronous interactions between *PWM voltage-source converters and power networks,*" IEEE Trans.Power Electro., vol. 29, no. 1, pp. 121-134, Jan. 2014.

23. L. Zhang, L. Harnefors, and H. Nee, "Interconnection of two very weak AC systems by VSC-HVDC links using power-synchronization control," IEEE Trans. Power Syst., vol. 26, no. 1, pp. 344-355, Feb. 2011.

24. D. N. Zmood, D. G. Holmes, and G. H. Bode,

"Frequency- domain anal-ysis of three-phase linear current regulators," IEEE Trans. Industry Appli., vol. 37, no. 2, pp. 601-610, Mar. 2001.

25. A. Rygg, M. Molinas, C. Zhang, and X. Cai, "A modified se-quence-domain impedanc definition and its equivalence to the dq-domain impedance definition for the stability analysis of AC power electronic systems," IEEE Journal of Emerging and Selected Topics in Power Electro., vol. 4, no. 4, pp. 1383-1396, July 2016.

26. A E Leon and J A Solsona 2015 "Sub-synchronous interaction damping control for DFIG Wind turbines"" J.IEEE Transactions on Power Systems. 1 30419-28.

27. L Fan, R Kavasseri, Z L Miao and C Zhu 2010 "Modeling of DFIG-based Wind farms for SSR analysis"" J.IEEE Trans. Power Delivery. 4 25 2073-82.

28. H A Mohammadpour, E Santi and A Ghaderi 2014 *""A Procedure to Study Sub-Synchronous Interactions in WindIntegrated Power Systems""* J.IET Generation, Transmission & Distribution. 12 8 1998-2011.

29. L P Kunjumuhammed, B C Pal, R Gupta, and K J Dyke 2017 ,, "Stability Analysis of a PMSG-Based Large Onshore Wind Farm Connected to a VSC-HVDC J.IEEE Transactions on Energy Conversion"" 32 1166-1176.

30. B Badrzadeh, M Sahni, Y Zhou, D Muthumuni and A Gole 2013 General Methodology for,, '*Analysis of Sub-Synchronous Interaction in Wind Power Plants*''' J.IEEE Transactions on Power Systems. 28 2 1858-1869.

31. L. H. Hassan, M. Moghavvemi, H. A. F. Almurib, K. M. Muttaqi, and V. G. Ganapathy, "*Optimization of power system stabilizers using participation factor and genetic algorithm*," Int. J. Elect. Power Energy Syst., vol. 55, pp.668–679, Feb. 2014.

32. L. Yang et al., "Oscillatory stability and eigenvalue sensitivity analysis of a DFIG wind turbine system," IEEETrans. Energy Convers., vol. 26, no. 1, pp. 328–339, Mar.

Implementation of Arduino based Smart Management System for Household and Corporate Application

Dhanraj D Daphale¹, Vijay A Sawant², Shreya Moholkar³

1,2,3Department of Electrical Engineering SVERI'S College of Engineering Pandharpur *dddaphale@coe.sveri.ac.in vasawant@coe.sveri.ac.in svmoholkar@coe.sveri.ac.in*

Abstract: Despite the vital role water plays in various activities, it's essential to recognize that only 2.5 percent of the world's freshwater is usable. Given the limited availability of this precious resource, it becomes imperative for us to be conscientious stewards, acknowledging the scarcity of reliable sources of usable water and taking proactive measures to sustain its vitality. Apart from the prevalent water wastage evident in our homes, municipal buildings, and large industries, there is also a noteworthy occurrence of energy wastage. Acknowledging the depletion of natural resources used in power generation, it's essential to recognize the dwindling availability of these resources on our planet. Given the pivotal role electricity plays in our lives, it becomes incumbent upon us to fulfill our fundamental responsibility by actively conserving and reducing its usage. With the help of this system it is possible for us to avoid the waste of water and electricity with smart mechanism using aurdino.

Keywords-GSM, Arduino Uno (Board), IR sensor (Infra-Red), theATmega328P microcontroller, PWM, DIY

I. INTRODUCTION

Electricity stands as the indispensable lifeblood for allindividuals, playing a pivotal role in various aspects of our lives. Yet, occasional energy loss occurs due to human negligence, underscoring the urgency for conservation efforts to ensure its sustainable use. Following the collection of water in a ground tank or pipeline, the water sensor in the proposed design is utilized to ascertain the water level. If the detected information is updated on an Android application, a user can view the water level on a smartphone. The motor is controlled based on the volume of water in the tank. The current world is centered around "smart" technology. As technology advances, the process will grow increasingly intelligent and automated. Resources are frequently optimized by intelligent irrigation systems to guarantee that everything gets what it requires without wasting any. Given the severity of the water and energy crises, an increasing amount of attention has been directed into irrigation techniques that consume less of both The block diagram illustrates the integration of essential components such as Arduino, a GSM module system, relay module, water sensors, water pump, and connecting Wi-Ficonverter. Water sensors are strategically placed at the input and outlet points. When water enters the pipe, the sensor sends a signal to the Arduino. Another sensor near the outlet receives a signal from the Arduino, allowing it to monitor the water level in the tank. In the absence of water, the Arduino signals the relay, activating the motor to commence waterpumping. Once the tank is full, the Arduino receives a signal and instructs the relay to turn off the motor. Simultaneously, the GSM module notifies the user about the impending arrival and tank-filling process. The Arduino, facilitated by the Integrated Development Environment (IDE) software, loads a program onto the Arduino Uno board. This program reads input values and orchestrates the system's operation. This system employs two water sensors, with the first one positioned at the inlet of the pipe and the second at the tank. The sensors determine if the tank has reached its capacity. When water reaches the upper side of the tank where the sensor is located, it senses the water level and signals the Arduino. Upon confirming that the tank is full, the Arduino commands the relay module to halt the motor. Subsequently, the GSM module sends a notification to the user indicating that the tank is now filled. The entire system operates seamlessly, characterized by automation and effective control.
II. BLOCK DIAGRAM AND METHODOLOGY



Fig.1: Block Diagram of System

III. FLOWCHART



Fig.2: Flowchart

The system operates autonomously and adheres to regulated procedures. There's no requirement for someone to be physically present at the tank site to ascertain water availability. An advantageous aspect of this design is that

users are relieved of theresponsibility to manually activate the motor as water enters the inlet. The system intuitively activates the motor when specific conditions are met, such as the presence of water in the pipeline and an unfilled tank. Conversely, the motor remains inactive when the tank is already full, ensuring a consistently relaxed operational state. In a different case, the water inlet sensor will cause the motor to be in the idle position if the tank is empty yet there is no water in the intake pipeline. Arduino communicates that there is no water in the inlet pipe and advises the relay tostay in its optimal state.

I. CIRCUIT DIAGRAM



Fig.3: Circuit Diagram

The DOL starter with pump is connected to the relay module with the help of contactor coil. Relay module is interfaced with Aurdino UNO for proper functioning of the system. At the same time water sensors are also interfaced with the Aurdnio UNO in order to detect the water level present in the tank.

II. IMPLEMENTATION OF SYSTEM



Fig.4: Real time implementation of system



Fig.5: Relay module and controller

Arduino Uno is an open-source microcontroller board based on theATmega328P microcontroller. It is one of the most popular boards in the Arduino family and is widely used in DIY projects and prototyping. The board has 14 digital input/output pins (of which 6 can be used as PWM outputs), 6 analog inputs, a 16 MHz quartz crystal, a USB connection. The primary part of the module, the relay functions as an electrically operated switch. It normally consists of a series of contacts and an electromagnet, or coil. The coil creates a magnetic field that attracts and releases the contacts when current flows through it, opening and closing the circuit. Make that the voltage and current specifications of the relay module are appropriate for your application before using it. Additionally, ensure that the low-power and high-power circuits are appropriately isolated and take all essential safety measures to guard against electrical risks. Relay modules can have one or By utilising this technology, we will cut down on water waste more channels, each of which represents the number of relays the and save money on electricity module contains. While multi-channel modules can have two, four, eight, or more relays, single-channel modules only have one. Because each channel functions separately, you can control numerous circuits at once. With the input portion of relay modules, you can use a low-power signal from a microcontroller or other digital device to control the relay. To separate the control signal from the high-power circuit, the input portion frequently has a transistor or optocoupler.



Fig.6: Result (Motor Turned ON & OFF)

The water sensor detects the presence of water in the water pipe line at that precise instant and sends a signal to the Arduino, which relays the signal to the relay module. The water pump activates automatically when the tank is empty and notifies the system user via the GSM system that "water pump is turn ON." The Arduino receives a signal from the water sensor when the tank is full, detects the water level, and automatically turns off the water pump. It then alerts the user via GSM that the water pump is off.

III. RESULT

The water sensor detects the presence of water in the water pipe line at that precise instant and sends a signal to the Arduino, which relays the signal to the relay module. The water pump activates automatically when the tank is empty and notifies the system user via the GSM system that "water pump is turn ON."

The Arduino receives a signal from the water sensor when the tank is full, detects the water level, and automatically turns off the water pump. It then alerts the user via GSM that the water pump is off.

IV. CONCLUSION AND FUTURE SCOPE

This model will run automatically because of how easily it can be used. Commercial, industrial, and

agricultural applications can all make use of this model. The nation's water and wastewater management infrastructure, encompassing pipe systems, facilities, and equipment, faces inherent shortcomings that contribute to environmental harm and the wastage of millions of gallons of water annually. Traditional practices such as labor-intensive meter readings and a lack of insight into distribution, collection, and consumption patterns lead to time-consuming, expensive, and reactiveservices. To mitigate these losses and address pressing issues related to drought, flooding, and water quality, the water industry is embracing cutting-edge sensors and communication solutions tailored for "smart water management." This shift is primarily motivated by more stringent government compliance standards, the evolution of smart cities, and the imperative for water conservation, particularly in agriculture and other high- water consumption sectors.

V. REFERENCES

- 1. A Amir, R Fauzi, and Y Arifin. Smart water meter for automaticmeter reading. In IOP Conference Series: Materials Science and Engineering, volume 1212, page 012042. IOP Publishing, 2022.
- 2. MK Anushree and R Krishna. A smart farming system using arduino based technology. Int. J. Adv. Res. Ideas Innov. Technol, 4(4):850–856, 2018.
- AN Arvindan and D Keerthika. Experimental investigation of remote control via android smart phone of arduino-based automated irrigation system using moisture sensor. In 2016 3rdInternational Conferenceon Electrical Energy Systems (ICEES), pages 168–175. IEEE, 2016.
- 4. Carlo Giudicianni, Manuel Herrera, Armando di Nardo, Kemi Adey- eye, and Helena M Ramos. Overview of energy management and leakage control systems for smart water grids and digital water. Modelling,1(2):134–155, 2020.
- Wu-quan He, Ming-ke Cai, Yu-bao Wang, and Xiao-jian Wang. Automaticwater supply control system of graded constant pressure byvariable frequency speed and its application to pipeline irrigation. In2010 Second WRI Global Congress on Intelligent Systems, volume 1,pages 385–388. IEEE, 2010.
- 6. Shaun Howell, Yacine Rezgui, and Thomas Beach. Integrating buildingand urban semantics to empower smart water solutions. Automationin Construction, 81:434–448, 2017.
- Mr. Dhanraj D. Daphale1, Sachin N. Todkari2, Shreyash A. Sakhare3, Rohan B. Bichitkar4, Rushikesh B. Vathare5 Smart Water Management for Home andCorporate System using Arduino, IJARSCT, Volume 3, Isuue 1, ISSN (Online) 2581-9429.

Quasi-Z-Source Solar Inverter fed BLDC Drive System

Shreya K. Sindol¹, Shruti S. Ankalgi², Prajakta S. Vedpathak³, Priyanka S. Devmare⁴, S.V. Moholkar⁵ Department of Electrical Engineering5 Student, Department of Electrical Engineering^{1,2,3,4} Shri Vitthal Education & Research Institute's, College of Engineering, Pandharpur

Abstract— The Quasi-Z-Source inverter is a key component of this project, offering several advantages over traditional inverters, including inherent buck-boost capabilities and improved reliability. The integration of solar panels into the system enhances the sustainability and reduces dependency on conventional power sources. This project focuses on the design, simulation, and analysis of a solar inverter system employing the Quasi-Z-Source (qZS) topology to power a Brushless DC (BLDC) drive. The qZS inverter's unique characteristics, including its ability to handle varying input voltages and inherent boost capability, make it an ideal for renewable energy applications. The quasi z-source network has applications in the field of renewable energy as it acts as a power conditioning circuit. This four switch quasi z source structure decreases the switching losses, total harmonic distortion and cost in comparison with traditional six switch structures. Simulation of the proposed model was done in MATLAB/SIMULINK

Keywords Quasi z-source inverter, MATLAB simulation.

I. INTRODUCTION

The quasi-Z source inverter is a modified version of the Z source inverter, offering enhanced voltage boost capabilities and improved reliability. A Quasi-Z-Source inverter fed Brushless DC (BLDC) drive system is an advanced technology that efficiently converts DC power from sources like solar panels or batteries into AC power for precise and reliable control of BLDC motors. There are many challenges when a BLDC motor is driven using renewable energy sources such as solar and wind power. The intermittent supply provided by the renewable energy sources needs to be modified to a DC with superior output quality for BLDC motor drive applications. For obtaining balanced three phase output voltages, the existence of a ripple free DC supply is necessary. Also bulky and heavy capacitors must be avoided as they lead to slower response and increase in cost. A Quasi- Z-Source Inverter (qZSI) is a type of power inverter used in renewable energy systems and motor drives. It incorporates a unique impedance network, the Z-source network, which provides certain advantages over traditional inverters.

The Z-source network allows the inverter to handle a wider range of input voltage variations and enables buck-boost functionality, improving the overall flexibility of the system. Unlike conventional inverters, a qZSI can boost the input voltage when needed, enhancing its performance in various operating conditions. A Quasi-Z-Source Inverter (qZSI) is a type of power inverter used in renewable energy systems and motor drives. It incorporates a unique impedance network, the Z-source network, which provides certain advantages over traditional inverters. The Z-source network allows the inverter to handle a wider range of input voltage variations and enables buck-boost functionality, improving the overall flexibility of the system. Unlike conventional inverters, a qZSI can boost the input voltage.

The qZSI finds applications in photovoltaic systems, wind energy systems, and electric vehicle drives, where its flexibility and ability to handle varying input conditions make it a compelling choice for power conversion. A Quasi-Z- Source Inverter (qZSI) is a specialized type of power inverter designed for renewable energy systems and motor drives. Its distinctive feature is the implementation of a unique impedance network known as the Z-source network. The Z- source network enables the inverter to tolerate a wider range of input voltage variations. This is particularly beneficial in renewable energy applications where the input voltage from sources like photovoltaic panels or wind turbines can be variable. Developing a mathematical model of the qZSI, incorporating the Z-source network and the switching elements. Define the equations governing the behaviour of the inverter, taking into account shoot-through states and control strategies. Parameter Selection Specify the parameters of your qZSI model,

including values for the Z-source network components, modulation index, and other relevant parameters. These values are often determined based on the specific requirements of your application. Implementing Control Strategies Incorporate control strategies to manage the shoot- through states and overall operation of the inverter. Common control methods include pulse width modulation (PWM) techniques and shoot-through control [1].

II. PROPOSED SYSTEM

This is the block diagram of solar inverter based on quasi-z-source fed for BLDC drive. It contains solar panel, battery, quasi-z- source inverter and a BLDC drive. First, the solar panel captures sunlight from the sun and converts it into DC power. This power is then stored in the battery. The Quasi-Z-Source inverter takes the DC power from the battery and converts it into AC power. Finally, the AC power is supplied to the BLDC drive. This sequence ensures that the solar energy is efficiently utilized to power the BLDC drive.



Fig. 1 Shows proposed system block diagram

Proposed Quasi-Z-Source Inverter



Fig 2. Quasi-Z-Source Inverter

The quasi-Z-Source inverter is shown in fig.2. The quasi-Z source inverter is the modified version of z source inverter. The Quasi-Z-Source inverter, renowned for its ability to handle wide voltage ranges and inherent boosting capabilities, aligns seamlessly with solar applications. Through MATLAB, we aim to explore and optimize the control strategies, modulation techniques, and power conversion algorithms tailored to harness solar energy efficiently for BLDC propulsion systems. This will delve into the implementation details, simulation methodologies, and performance assessments of the Solar Inverter based on the Quasi-Z-Source topology, offering insights into its feasibility, efficacy, and potential advancements in renewable energy-driven BLDC applications. Integrating a Quasi-Z-Source inverter, which allows for greater flexibility and improved performance compared to traditional inverters. Creating a system that effectively harnesses solar energy and converts it into usable power for the BLDC

drive. Developing control algorithms and mechanisms to ensure the smooth operation and control of the BLDC drive.

Quasi-Z-Source inverter plays a crucial role in converting the DC power from the battery into AC power. It uses a unique topology that allows for improved efficiency and flexibility in power conversion. The Quasi-Z-Source network controls the voltage and current to meet the requirements of the BLDC motor, ensuring smooth and efficient operation of the motor. This helps in maximizing the overall efficiency and performance of the system.

The Quasi-Z-Source inverter, renowned for its ability to handle wide voltage ranges and inherent boosting capabilities, aligns seamlessly with solar applications. Through MATLAB, we aim to explore and optimize the control strategies, modulation techniques, and power conversion algorithms tailored to harness solar energy efficiently for BLDC propulsion systems [2].

III. SIMULATION AND RESULTS

Use MATLAB/Simulink to build the simulation model. Create blocks representing the components of the qZSI, including the Z-source network, switches, and control algorithms. Defining Input Conditions Specify the input conditions for your simulation, such as the input voltage from renewable sources (e.g., photovoltaic panels or wind turbines). This step is crucial to evaluate the inverter's performance under varying input conditions. Running the Simulation Execute the simulation and observe the behaviour of the qZSI. Analyse key performance metrics, including output voltage, current, and efficiency.



Fig.3 Proposed Simulation Model of Quasi-Z-Source inverter.

Simulation and Modelling:

MATLAB provides a comprehensive platform for simulating and Modelling: complex systems, making it ideal for designing and analyse power electronics systems like a Quasi-Z-Source inverter for BLDC drives. **Performance Analysis:**

We can use MATLAB to analyse the performance of the solar inverter and BLDC drive under various operating conditions, helping you make informed design decisions. Parameter Optimization: MATLAB can assist in optimizing component values and parameters to achieve desired performance and efficiency.

Real-Time Testing:

MATLAB also supports real-time testing and implementation, which is essential for hardware-in-the-loop (HIL) testing of the inverter and drive system [3].



Fig.5 Scope 1 output (Logical Operators)



Fig.6 Scope 3 output (Logical Operators)

The below graph (fig .7) shows the final scope output of quasi z source inverter.



Fig.7 Quasi-Z-Source inverter output voltage



Fig.8 Scope 4 (Relational operator)

IV. CONCLUSION

In this Simulink-based project, we embarked on a comprehensive exploration of the Quasi-Z-Source topology Through complex simulations and modeling within the Simulink environment, we sought to evaluate the feasibility and performance of this innovative system. Simulations revealed the inherent advantages of the Quasi-Z-Source topology in handling wide input voltage ranges, providing robustness against fluctuations in solar power output [4].

V. REFERENCES

- 1. Simulation of Reduced Component Count Quasi Z-Source Inverter Fed BLDC Motor for Renewable Energy Applications" by Milan Sebastian.
- 2. BLDC Drive Speed Control with Quasi Z-Source Inverter Using Back EMF Estimation Control 1K. R. Sugavanam, 2Dr.M. Rajavelan, 1M. Prabu, 1Dr. J. Prakash, 1M. Malathi, 1S. Senthilkumar
- Rajkumar S, Sundaram CS, Sedhuram K, Muruganandhan D. Performance Analysis of Hub BLDC Motor Using Finite Element Analysis. In2019 IEEE International Conference on System, Computation, Automation and Networking (ICSCAN) 2019 Mar 29 (pp. 1-5). IEEE.
- 4. Padmaraja Yedamale, Microchip Technology Inc, 2003 Microchip technology, "Brushless DC (BLDC) Motor Fundamentals"
- International Journal of Engineering Research & Technology (IJERT) Vol. 2 Issue 5, May 2013 "Implementation Of Brushless DC Motor Using FPGA Interface" by Mohammed Zoheb, Vijeyta Sharma M, Mrs. Shalini Vashishtha & Mr. Mohammed Shahid
- P. Prakash, M.Rajavelan. Performance Analysis of BLDC Motor Drive using ANFIS Controller. International Journal for Research in Applied Science & Engineering Technology, 2018 Mar;6(3):2676-2681.

Smart Doormat with IoT-Enabled Visitor Detection

Mr. Shreays S. Kulkarni¹, Ms. S.V. Moholkar², Mr. D. D. Daphale³, Mr. V. A.Sawant⁴ Department of Electrical Engineering^{1,2,3,4} Shri Vitthal Education & Research Institute's, College of Engineering, Pandharpur.

Abstract— Suggested system fulfils the home security system's door mat notification function. This system can be easily changed to provide numerous functional systems, such as safety control, fire detection, and industrial automation. The suggested technique detects the visitor outside the door.

By snapping a visitor's photo, the owner will receive an email immediately. Authentication is the most important component of any security system. An Android phone is the main device used in the proposed system for email visitor notifications and authentication. The proposed approach allows the owner to view the visitor's past year history by storing images of them both offline and online. With 3% of errors, this system has a 97% accuracy rate. It could occur due to a network problem. As a result, the entire system is robust and has a promising future for a range of uses.

Keywords— Cloud computing, email, IoT, and smart doormats.

I. INTRODUCTION

Without opening the door, it is challenging to discern who is outside. When we are working in the kitchen or completing important office chores, it takes a lot of time to open the door or watch the person outside the door. In addition, if we are travelling outside of the state or nation, we must allow servants or known people enter the residence. With this technology, you can immediately identify who is at your door and decide whether to let them in or not by simply pressing a single button. Today's culture places a high value on home security, which makes using a system that provides sufficient protection all day and night imperative. Many security systems available on the market are not cloud-connected, despite their abundance [1]. Handheld devices access and save data on this cloud server. Our proposed design calls for data to be saved on one device in the cloud and retrieved on another. Hence, compared to other technologies, this one is quick and simple to use. In addition, the system is converted to a CCTV and utilized for home monitoring.



Fig.1 System block diagram

A Raspberry Pi, a camera, an Ethernet connection, a piezoelectric sensor, an ultrasonic sensor, and a camera make up the recommended system's many parts. The adapter can offer the 5V power source that Raspberry Pi needs. It also has a special 5-megapixel, 8-megapixel camera. The camera port on the Raspberry Pi is directly connected to this camera. In order to give the user instructions in the form of pictures, a Raspberry Pi was fitted with a CCTV camera. An Ethernet cable must be connected to the Raspberry Pi's Ethernet shield in order to access the internet. We placed an ultrasonic sensor 100 cm away in order to detect. Furthermore, a piezoelectric sensor was utilised to supply data to the door mat. The Raspberry Pi has 40 GPIO pins that can be used to connect different kinds of sensors. The Raspberry Pi's gpio pins are connected to the electric buzzer.

2.2 Hardware prototype:

Raspberry Pi

The proposed system incorporates the Raspberry Pi 3 as a crucial element. With its 4 USB ports, users can easily connect a mouse, keyboard, and pen drive, while the Ethernet-compliant connection enables the use of an Ethernet cable. The Raspberry Pi 3's 40 GPIO pins provide flexibility for attaching various sensors such as ultrasonic, air, temperature, and moisture sensors. Additionally, the Raspberry Pi includes dedicated slots for attaching the Pi camera and touch screen display.





To set up the system, an 8 GB memory card is essential for installing the operating system and storing software. The versatile Raspberry Pi serves as a capable music player, equipped with a 3.5 mm audio connector for seamless connection to headphones and home theatre systems. Facilitating its power needs, the Raspberry Pi incorporates a dedicated power slot compatible with a 5V, 2A charger, and a separate power circuit can be designed as well. Moreover, the device is equipped with an HDMI connector, allowing easy connectivity to external desktops or projector .[3]

• Pi Camera:

The Raspberry Pi is equipped with a dedicated camera slot that directly interfaces with the Pi camera. Users can choose between 5 MP and 8 MP camera options, with the system opting for the 5 MP variant. Utilizing a Pi camera on the Raspberry Pi offers a simplified experience compared to using a web camera, thanks to its inherent command integration. Before utilizing the camera, it is essential to enable it in the Raspberry Pi configuration. Positioned above the door at the center, the camera serves a pivotal role in the system. The Pi camera, specifically designed for use with the Raspberry Pi, is a compact and versatile imaging solution that enhances the capabilities of the system.[2] It seamlessly integrates with the camera slot on the Raspberry Pi, providing users with a streamlined and efficient way to capture images and videos for various applications.



• Piezoelectric Sensor:

A piezoelectric sensor is a type of sensor that utilizes the piezoelectric effect to measure changes in pressure, acceleration, temperature, or force by converting them into an electrical charge. The piezoelectric effect is the ability of certain materials, such as crystals and ceramics, to generate anelectric charge in response to mechanical stress.

Working Principle: Piezoelectric sensors operate based on the principle that when mechanical stress is applied to a piezoelectric material, it causes a displacement of charge within the material, resulting in the generation of an electric voltage. This voltage is proportional to the applied force or pressure, allowing for precise measurement.

Applications: Piezoelectric sensors find applications in various fields. They are commonly used in industrial settings for pressure and force measurements, in accelerometers for detecting vibrations and accelerations, and in medical devices for monitoring physiological parameters such as heart rate and blood pressure.

Sensitivity and Accuracy: One of the advantages of piezoelectric sensors is their high sensitivity. They can detect small changes in pressure or force, making them suitable for applications that require precise measurements. Additionally, they have a quick response time, providing real-time data.

Compact and Lightweight: Piezoelectric sensors are often compact and lightweight, making them suitable for applications where space and weight constraints are crucial, such as in aerospace or automotive industries.

Durability: These sensors are known for their durability and reliability. They can withstand harsh environmental conditions and are less susceptible to wear and tear compared to some other sensor types.

Energy Harvesting: In addition to sensing, piezoelectric materials can also be used for energy harvesting. The mechanical vibrations or movements in the environment can be converted into electrical energy, offering a potential source of power for low-power electronic devices.[4]

Integration with IoT and Electronics: Piezoelectric sensors can be easily integrated into electronic systems and Internet of Things (IoT) devices, providing valuable data for monitoring and control applications.

In various engineering and scientific applications, the versatility and effectiveness of piezoelectric sensors make them an essential component for capturing and translating mechanical phenomena into measurable electrical signals.



Fig.4: Piezoelectric Sensor

In this setup, we utilize piezoelectric sensors that transform the piezoelectric effect into electrical impulses, enabling the monitoring of acceleration, force, and pressure. These sensors trigger alerts through a buzzer activation. Consequently, the need for a conventional doorbell is eliminated when guests arrive, as the doormat itself becomes an intelligent sensing device capable of recognizing guests and issuing alerts on their behalf.

The embedded Piezo sensor within the mat consistently monitors changes in pressure induced by objects or individuals standing or sitting on it. This transformative technology turns the doormat into a dynamic and weightsensitive layer of smart foam. This smart foam can be strategically placed under a doormat or virtually anywhere else in your home where reactions to pressure changes are desired.

Beyond its role as a doormat, this technology introduces a versatile and adaptable solution. The weight-sensitive layer of smart foam extends its applicability to various scenarios within your household. Whether positioned under furniture, carpets, or other surfaces, it provides an intelligent means of detecting and responding to changes in pressure. This not only enhances convenience but also offers a potential layer of security, alerting occupants to the presence of individuals or objects in specific areas.

In summary, the integration of piezoelectric sensors within the doormat transforms it into a sophisticated, weightsensitive layer of smart foam. This innovation surpasses the conventional role of a doormat by enabling intelligent and automated responses to changes in pressure, making it a valuable addition to your home for both convenience and security [5].

III. FLOW CHART OF THE SYSTEM:

The system follows a structured flow procedure where each step occurs sequentially. It begins with the initial person detection using a piezoelectric sensor. If a visitor is identified, the system activates a buzzer and captures an image. Subsequently, an email containing the captured image is sent to the owner. Additionally, the piezoelectric sensor is utilized to trigger a buzzer in case theft is detected. This ensures a systematic and orderly response to both visitor recognition and theft alerts within the system.

By leveraging the Internet of Things (IoT) and the Raspberry Pi 3 module, we construct a smart doormat equipped with a visitor sensor. This enables automatic detection of visits. Furthermore, data is received through a piezoelectric sensor, and email notifications are sent to our mobile devices. This ensures that we can promptly detect theft and take protective measures, regardless of whether we are at home or not. Additionally, there is the capability to maintain visitor records in picture format as an available option.

The implementation of the Smart Doormat with IoT-Enabled Visitor Detection has yielded promising results. Leveraging the capabilities of the Internet of Things (IoT) and the Raspberry Pi 3 module, the system demonstrates effective and automatic detection of visitors as they approach.



IV. RESULTS

Key outcomes of the project include:

Automated Visitor Detection: The smart doormat successfully identifies and detects visitors through the integration of IoT technology and the Raspberry Pi 3 module. This automated process enhances convenience for occupants.

Real-time Data Acquisition: The piezoelectric sensor facilitates real-time data acquisition, enabling the system to promptly respond to changes in pressure caused by individuals standing on the doormat. This ensures timely and accurate visitor detection.

Immediate Alerts: In the presence of a detected visitor, the system triggers a buzzer, providing an audible alert. Simultaneously, an image is captured and sent via email notifications to the owner's mobile devices, allowing for immediate awareness of visitors even when away from home.

Theft Detection Capability: The integrated piezoelectric sensor serves a dual purpose by also detecting theft. In such instances, the system activates the buzzer, delivering an alert to notify occupants of unauthorized activity.

Remote Monitoring: The email notifications enable remote monitoring, allowing homeowners to stay informed about visitor activity and potential security concerns whether they are at home or not.

Visitor Records: The system offers the option to maintain visitor records in picture format. This feature enhances security by providing a visual log of individuals who have approached the premises.

• Image captured by the Raspberry pi camera.



Fig 6: Actual captured image

•Front side of project model:



Fig.7: Front view of model

• Top view of Project Model:



Fig. 8: Top view of model

V. APPLICATION:

- Enhancing home security through surveillance measures.
- Implementing an automatic door open and close system for added convenience.
- Strengthening locker security with advanced face recognition technology.
- Automating gates with a built-in visitor count feature and notification functionality.
- Improving safety by integrating fire detection systems with automatic notifications to the fire brigade.

VI. CONCLUSION:

In conclusion, the implementation of the Smart Doormat with IoT-Enabled Visitor Detection project has resulted in a successful and innovative solution for enhancing home security and automation. By leveraging Internet of Things (IoT) technology and the Raspberry Pi 3 module, the doormat demonstrates a seamless and automated process of detecting visitors.

The key achievements and conclusions of this project are as follows:

Efficient Visitor Detection: The integration of piezoelectric sensors enables the doormat to efficiently detect the presence of visitors. This automated process eliminates the need for traditional doorbells and enhances the overall accessibility and responsiveness of the home security system.

Real-time Alerts: The system promptly triggers a buzzer and captures an image when a visitor is detected. Simultaneously, email notifications are sent to the owner's mobile devices, ensuring immediate awareness and enabling quick response, whether the homeowner is at home or away.

Dual Functionality for Theft Detection: The utilization of piezoelectric sensors not only facilitates visitor detection but also serves as an effective mechanism for detecting potential theft. The system's ability to sound the alarm in such situations provides an additional layer of security.

Remote Monitoring and Control: The project enhances the concept of smart homes by allowing homeowners to remotely monitor visitor activity. The integration of IoT technology enables seamless communication and control, contributing to an advanced and connected living environment.

Overall, the Smart Doormat with IoT-Enabled Visitor Detection project successfully combines technological innovation with practical security applications. The system's ability to automate and enhance visitor detection contributes to the evolving landscape of smart home solutions, providing homeowners with a reliable and intelligent approach to home security and surveillance.

A Smart Doormat with IoT-enabled visitor detection has several exciting future scopes Here are some potential areas of growth and development:

Enhanced Security Features: Integration with home security systems: The doormat could be connected to other security devices like cameras, alarms, or smart locks to enhance overall home security.

Facial recognition: Incorporating facial recognition technology could provide an added layer of security by identifying known visitors and alerting homeowners to potential threats.

Data Analytics for Insights: Visitor patterns and trends: Analyzing data collected over time can provide insights into visitor patterns, helping homeowners understand their daily routines or identify irregularities. Integration with other IoT devices: Connecting the doormat to other smart devices within the home can provide a comprehensive view of the environment, allowing for more informed decisions and automation.

Integration with Virtual Assistants: Voice-enabled interactions: Integration with virtual assistants like Amazon Alexa or Google Assistant could enable users to control and monitor the doormat using voice commands.

Automated notifications: Virtual assistants can send alerts or notifications to homeowners when a visitor is detected, even providing information about the visitor if available.

Customization and Personalization: User profiles: Allowing users to create profiles for different family members or regular visitors can help in personalizing the doormat's responses or notifications based on individual preferences.

Customizable settings: Providing users with the ability to customize sensitivity levels, notification preferences, or other settings can enhance the user experience.

Integration with Smart Home Ecosystems: Seamless connectivity: Ensuring compatibility with popular smart home ecosystems like Apple Home Kit, Google Home, or Samsung Smart Things can broaden the doormat's usability.

V. REFERENCES:

- 1. J. C. Augusto and C. D. Nugent, "Smart homes can be smarter," Springer, pp. 1-15, 2006.
- Rui Antunes, and Frederico Grilo Helder Adão, "Web-Based Control & Notification for Home," International Journal of Computer, Electrical, Automation, Control and Information Engineering, vol.2, no. 1, 2008.
- 3. J.J Greichen, "Value based home automation or today's market," IEEE Transactions on Consumer Electronics, vol. 38, no. 3, pp. 34-38, Aug 1992.
- 4. F. and Floerkemeier, C Mattern, "From the Internet of Computers to the Internet of Things.," Springer, pp. 242-259, 2010.
- 5. D.-M. Han and J.-H. Lim, "Design and implementation of smart home energy management systems based on zigbee," Consumer Electronics, IEEE Transactions, vol. 56, no. 3, pp. 1417-1425, 2010.

Simulation and Analysis of Thyristor-Based Industrial Battery Charge Controller using MATLAB

Priyanka Ankush Pawar¹, Pallavi Suresh Bachute², Pradnya Sanjay Mane³, D. A. Tamboli⁴ Department of Electrical Engineering.¹⁻⁴, SVERI's COE Pandharpur, Maharashtra, India priyankaapawar@coep.sveri.ac.in¹ pallavisbachute@coep.sveri.ac.in² pradnyasmane@coep.sveri.ac.in³ datamboli@coe.sveri.ac.in⁴

Abstract— In recent developments, various rectifier circuit configurations, including multi-pulse arrangements, have been employed to convert regular electrical power from AC to DC, effectively minimizing ripple content in the output. The applications of these rectifiers in diverse industrial settings, such as electroplating, heating, magnet power supply, traction and battery charging have necessitated tailored specifications. Notably, for single-phase systems, DC power is derived from a thyristor-controlled rectifier, while AC power is harnessed through a bridge rectifier consisting of diodes and a thyristor for precise control. In this paper, we have designed a single and three-phase pulses thyristor-controlled rectifier is comprehensively detailed, encompassing both purely resistive and resistive-inductive load scenarios. To validate and analyse the designs, circuit simulation tools have been conducted, with the subsequent presentation of simulation results.

Keywords—Thyristor, Triggering pulse, Semi-controlled rectifier, Various loads, MATLAB simulation.

I. INTRODUCTION

Battery chargers play a vital role in powering diverse devices such as laptops, electric vehicles, mobile phones, electronic gadgets and electronic tools, etc [13]. Ensuring robust and efficient battery performance hinges on the dependability of the charger. Chargers are often distinguished by their charging speed. In low-voltage applications like battery charging, automotive, and aircraft systems, DC power finds common use [5]. Previously, the generation of DC power was achieved through power electronic semiconductor switches. These semiconductor switches can be categorized based on their controllability:

- I. Uncontrollable Diodes
- II. Controllable Thyristors

Silicon diodes are frequently employed in rectifiers, yet they lack control over the output voltage [5]. In applications demanding controllability, such as battery chargers, thyristors are preferred over diodes. Unlike diodes, thyristors necessitate a triggering signal at the gate for conduction when forward-biased. The firing angle (α) of the thyristor dictates the control over the rectifier, where α =0 results in behaviour akin to an uncontrolled diode rectifier [3]. This paper delves into a single- phase full-wave controlled bridge rectifier, examining its behaviour under diverse loads, including resistive (R) and resistive-inductive (R, L) loads. Through phase control achieved by triggering pulses to the thyristor gate, the average load voltage can be managed and adjusted. MATLAB simulations are carried out at different triggering angles and different loads [4].

There are two methods mostly used for voltage control.

- a. PWM Controlled Method [7]
- b. Thyristor Firing Angle Control Method [2]

a. PWM Controlled Method

Pulse Width Modulation (PWM) is a common method in power electronics, especially for controlling the output voltage in DC/AC inverters. Essentially, PWM adjusts the width of output pulses to regulate the average

output voltage. This allows a fixed DC input to generate a variable-frequency sinusoidal waveform with adjustable amplitude. Inverters using PWM provide precise control over voltage and current, crucial in applications like motor drives, renewable energy systems and uninterruptible power supplies (UPS)[6]. This fine control helps achieve optimal performance PWM also allows managing harmonic distortions in the output waveform, improving power quality and reducing system losses. Compared to basic square-wave modulation, PWM provides better control over voltage, frequency, and harmonics enhancing overall efficiency and performance in various applications [7].

When using PWM in power electronics systems, there are some downsides. It has the effect of raising the switching frequency and stress on power devices, leading to a reduced lifespan, increased switching losses, and higher emissions of electromagnetic interference (EMI). Moreover, you need an advanced controller and driver circuit to create and apply the PWM signals to the power device, which can drive up costs and add complexity to the converter. Lastly, it can bring about negative effects on the connected devices, including voltage spikes, current ripples, noise, and vibrations. This is particularly true for sensitive or nonlinear loads such as motors, batteries, or LEDs [7].

These are the disadvantages of PWM controlled methods which overcomes by using thyristor firing angle control.

b. Thyristor Firing Angle Control Method

Thyristor firing angle control serves as a crucial method for precisely regulating voltage in AC power systems by manipulating the firing angle of thyristors within the circuit. This technique involves adjusting the timing of thyristor conduction during each AC cycle, allowing for effective voltage control across the load [2]. Semiconductor devices known as thyristors function as switches in electrical circuits, conducting current only upon receiving a trigger signal and remaining conducting until the current falls below a specific threshold. The firing angle, representing the delay in triggering thyristors, is a pivotal parameter. By controlling this angle, the point at which thyristors begin conduction, the effective voltage across the load can be dynamically altered [10].

Phase angle control a key aspect, entails adjusting the firing angle to regulate the power delivered to the load. Delays in the firing angle result in reduced power delivery, thereby lowering the average voltage across the load. This method boasts advantages in its simplicity and effectiveness for voltage regulation in AC circuits, often finding application in systems requiring smooth and continuous power control, such as heating systems [11]. Common applications of this control method include industries relying on precise power control for resistive loads, exemplified by heaters [12].

To address challenges like voltage spikes during turn-off, snubber circuits are frequently incorporated to reduce stress on the thyristors. Some advanced systems integrate feedback mechanisms to dynamically adjust the firing angle based on actual output voltage, ensuring enhanced regulation and stability [2]. Synchronization with the AC waveform is critical for accurate control, emphasizing the need for precise timing of thyristor firing to minimize waveform distortion [3]. In the broader context, thyristor firing angle control is an integral part of sophisticated control systems, often featuring sensors, feedback loops, and microprocessor-based controllers to meet specific application requirements.

These are the following methods used for control thyristor firing angle [2]. 1.Phase Angle Control 2.Integral Cycle Control 3.Single-Pulse Control 4.Two- Pulse Control 5.Three-Pulse Control 6.Multicycle Control 7.Slip Power Recovery System 8.Current Source Inverter Control

1. Phase Angle Control

In phase angle control, the manipulation of the firing angle is employed to regulate the segment of the AC waveform during which the thyristor conducts. This technique is commonly applied in scenarios involving resistive loads.

Regulating power entails manipulating the flow of the AC voltage signal through a conducting thyristor. The triggering card receives a control signal, dictating the allowed portion of the signal to pass while blocking the remainder. Adjusting this signal level enables the regulation of the average voltage across the load, prompting the load to draw current proportionally. In simpler terms, it resembles adjusting a faucet to govern the flow of water – in this context, it's about managing electricity for the desired power level [6].



Fig.1 Phase angle control representation

Phase Angle Control stands out as a precision method, offering meticulous management of the power supplied to a load. Analogous to a dimmer switch for electricity, it enables the selective use of current, ensuring only the essential amount is deployed. This not only facilitates accurate control but also contributes to energy conservation, utilizing only what is truly necessary. The analogy of adjusting the volume on a device further illustrates the nuanced ability to fine-tune power consumption, enhancing overall efficiency [16].

2. Integral Cycle Control

Integral cycle control enhances power regulation by extending the conduction angle beyond a single halfcycle, enabling a smoother control of power. This method is well-suited for resistive and moderately inductive loads. Integral Cycle Control provides an efficient method for power converters to transform AC power into another form of AC power directly, eliminating unnecessary energy wastage. Think of it as a shortcut that captures the optimal part of the AC wave, resembling a smooth wave in its operation.



Fig.2 Integral cycle control representation

These controllers strategically turn on or off precisely when the voltage hits zero, ensuring a streamlined process without unnecessary steps. In essence, it's like having a switch that instantly modifies the power, enhancing overall energy efficiency [8].

II. SIMULATION MODEL & EXPERIMENTAL RESULT

The simulation tool proves instrumental in crafting designs for both single-phase rectifiers. The first configuration involves an R-load with a 4-ohm resistance, while the second introduces an R-L load featuring a 4-ohm resistance and a 2 milli-henry inductance. Through meticulous simulations, it becomes evident that the anticipated outcomes are successfully achieved. Specifically focusing on the simulation circuit tailored for the R-L load, the configured parameters within the simulation software contribute to a comprehensive analysis of its performance is shown in fig 1[1].



Fig 1. Simulation Model for Single Phase RL Load.

The R-load circuit's output is acquired by varying the excitation voltage at the input side for different firing angles (alpha), as outlined in the corresponding table 1[1].

Firing Angle	Output Voltage
0	11.92
15	11.92
30	18.5
45	20.86
60	21.72

Table 1. Output voltages for Single phase R-Load for different triggering angles.

The above table1 values shows that the output voltages for different firing angle for single-phase R load.



Fig 2. Simulation waveforms for R-Load at alpha=30

The above Fig 2 shows that the waveform for single-phase R load at triggering angle α =30. Above figure shows the waveforms for voltage and current.



Fig.3 Simulation waveforms for RL-Load at alpha=30

Firing angle	Output voltage
0	11.92
15	11.95
30	20.12
45	21.45
60	21.73

Table 2. Output Voltage of Single-Phase RL-Load for Different triggering angle

A simulation tool is employed for the design of a three-phase rectifier. Two circuits were simulated: one for an R-load with a resistance of 4 ohms and another for an R-L load featuring a 4 ohm resistance and a 2 milli-henry inductance. The simulation results indicate the achievement of the intended outcomes [1].



Fig 3. Simulation circuit model for Three-Phase R-load

In a resistive load, the current and voltage are in phase. Adjusting the firing angle of thyristor in a circuit controlling a resistive load affects the portion of the AC waveform allowed to pass through, thereby regulating the power supplied to the resistive load. By Controlling the firing angle, we can effectively regulate the average voltage applied to the resistive load [2].

Firing Angle	Output Voltage
0	77.6 V
30	81.4
45	98.3
60	107.8
90	122.9

Table 3. Output voltage for Three phase R-Load for different triggering angles.

The above table values shows that the output voltages for different firing angle for three-phase R load.



Fig 4. Simulation waveforms for three phase R-Load at alpha=30

Increasing the firing angle, represented as 'a,' in a three-phase fully controlled rectifier results in a heightened time delay. causing greater fluctuations in the output waveforms. This pulsation is undesirable when striving for a stable DC output.

To mitigate this concern, various combinations of passive filters can be implemented at the output to reduce the observed DC voltage and current ripple.



Fig 5. Simulation waveforms for three phase RL-Load at alpha=30

Setting the firing angle (α) to 30 degrees in a 3-phase fully controlled rectifier with an R-L load leads to increased pulsations in the output waveforms rather than achieving a ripple-free state. This pulsating behaviour stems from the extended time delay linked to higher firing angles, which is undesirable for maintaining a constant DC output. To

mitigate this issue, it becomes essential to incorporate various passive filter combinations at the output end. This helps diminish the observed ripple or pulsating nature, ensuring a stable DC voltage. [1].

Firing Angle	Output Voltage
0	77
30	80
45	98.7
60	107
90	125

Table 4. Output voltage for Three phase RL-Load for different triggering angles.

The above table values shows that the output voltages for different firing angle for three-phase RL load. Conclusively, the implementation of firing angle control within a resistive-inductive (RL) load scenario emerges as a versatile and efficient method for power regulation. This approach, which involves adjusting thyristor firing angles, allows for precise management of both voltage and current in systems featuring inductive components. The manipulation of firing angles provides adaptability in delivering power to RL loads. Nevertheless, it is crucial to be aware of potential issues, such as harmonics, and customize the firing angles based on the distinct characteristics of the RL load to enhance overall performance.

III. CONCLUSION

The industrial battery charger project employs a sophisticated system that utilizes a thyristor-controlled rectifier for obtaining DC power from an AC power supply. The bridge rectifier, comprising diodes and a thyristor, facilitates precise control over the AC power. This setup ensures efficient charging of batteries through controlled thyristor firing angles. The design and simulation results presented for the single- phase and three-phase, fully controlled thyristor converter in industrial battery chargers demonstrate effective control over current, voltage and power handling capabilities. The parametric values provide crucial insights into transformer and thyristor performance. While ripple content in output waveforms is inevitable and increases with thyristor firing angle, the simulation highlights the importance of implementing output filters to minimize ripple.

Moreover, the shift towards digital control techniques is recommended to overcome flexibility and noise interference issues associated with analog control. The comparison indicates that digital systems draw considerably less power than their analog counterparts. Therefore, for the successful implementation of an industrial battery charger using thyristor firing angle control, adopting a digital control topology is strongly recommended for enhanced efficiency and performance.

IV. REFERENCES

- Dr. Mohd. Abdul Muqeet," Implementation and Simulation of DC Battery Fast Charging Topologies vol. 33 : ISSUE 02 - 2020
- Borst, David W., Edward J. Diebold, and Frank W. Parrish. "Voltage control by means of power thyristors." IEEE Transactions on Industry and General Applications 2 (1966): 102-124.
- DSP Based Battery Charger Control System Using Embedded Code Generation for Thyristor Rectifier Yasemin ÖNAL[†], Metin KESLER
- Kulkarni, Raunak, et al. "Design and Simulation of Three Phase Six Pulse Thyristor Controlled Rectifier." 2019 International Conference on Nascent Technologies in Engineering (ICNTE). IEEE, 2019.
- 5. e-ISSN: 2250-3021, p-ISSN: 2278-8719, www.iosrjen., PP 06-

- 6. 111mplementation of AC to DC converter Using Thyristor in ATPSudeep Pyakuryal1, Mohammad Matin2 Vol 2, Issue 11 (November2012).
- [Messalti, Sabir, et al. "Thyristor Controlled Voltage Regulator and Thyristor Controlled Phase Angle Regulator for Transient Stability Improvement of AC-HVDC Power System." Advanced Science Letters 19.5 (2013): 1421-1425.
- 8. Asiri, Y., & Shwehdi, M. (2007, April). Performance of single-phase full wave rectifier controlled by PWM. In 2007 Canadian Conference on Electrical and Computer Engineering (pp. 1155-1158). IEEE.
- 9. Razak, I. S. A., Hashima, R. I. Z. R. Z., & Soid, S. N. M. A Design of Single-Phase Bridge Full-wave Rectifier. In Conference on Language, Education, Engineering and Technology (Vol. 2 Nov 2016).
- 10. Choe, G-H., Alan K. Wallace, and M-H. Park. "An improved PWM technique for AC choppers." IEEE transactions on power Electronics 4.4 (1989): 496-505.
- 11. Malik, Nazar H., Syed M. Enamul Haque, and William Shepherd. "Analysis and performance of threephase phase-controlled thyristor AC voltage controllers." IEEE Transactions on Industrial Electronics 3 (1985): 192-199.
- 12. Yu, Zhenyu, Arefeen Mohammed, and Issa Panahi. "A review of three PWM techniques." Proceedings of the 1997 American Control Conference (Cat. No. 97CH36041). Vol. 1. IEEE, 1997.
- 13. LS, A., OA, E., & SO, E. Generation of Control Signals for an AC–DC Converters for a Hybrid Charging System (2018).
- 14. Zaid, M., Ubadah, M., & Mohammad, T. A Variable Voltage Semi- Controlled AC-DC Converter for Battery Charging.
- 15. Lazim, Mohammed T. "Power frequency converter based on integral- cycle triggering mode of thyristors with multi-control periods." Electric Power Components and Systems 41.12 (2013): 1173-1187.
- Messalti, Sabir, et al. "Thyristor Controlled Voltage Regulator and Thyristor Controlled Phase Angle Regulator for Transient Stability Improvement of AC-HVDC Power System." Advanced Science Letters 19.5 (2013): 1421-1425.
- 17. Li, Kejun, et al. "A study on mode-switching control of TCSC based on conditional firing of thyristor." IEEE transactions on power delivery 26.2 (2010): 1196-1202.

Fault Investigation in High Voltage Transmission Line

Akshay Kale¹, Amit Jadhav², Ravikant Khandekar³, Prof. H. M. Mallad⁴ ^{1,2,3,4} Department of Electrical Engineering, SVERI's College of Engineering, Pandharpur, MH-India. ¹akale4035@gmail.com ²amitj7780@gmail.com ³ravikantkhandekar027@gmail.com ⁴hmmallad@coe.sveri.ac.in

Abstract— An essential component of a power system are the transmission lines. Transmission lines are vulnerable to larger- scale failures and have a high-power transmission capacity. Transmission lines can have a variety of defects for this reason, transmission line protection requires protection relays. The fault currents may be harmful to electrical equipment, so protection system's goal is to cut off the unhealthy section from the healthy section. Distance relays are one type of protective relay that are mostly utilised in transmission lines. The voltage and current are required by distance relays to calculate impedance. Usually, distance protection relays are used to safeguard transmission cables. Distance relay used in transmission line are also known as high-speed relay. The proposed model has been validated through various tests, including line-to-line (L-L), two-line-to-ground (L-G), one line-to-ground (L-G), and three-level (L-L-L) fault. The SimPower System was utilised to model and simulate transmission lines, faults and distance relays. By using MATLAB software simulation results and distance relays for the investigation of transmission line protection for single, double and three phase faults at various transmission line locations. MATLAB/SIMULINK software can be used to overcome the challenges associated with comprehending the operation of a distance relay.

Keywords— Fault Types, High Voltage Transmission Line, Fault Analysis, Power System Protection, Distance Protection Scheme, MATLAB Simulation.

INTRODUCTION

The three primary parts of a power system's structure are distribution, transmission, and generating.

I.

Generating: The process of producing electricity from a variety of sources, including nuclear, fossil fuels, hydropower, wind, solar, and more, is known as power generating. Electricity is produced by power plants or renewable energy installations and is subsequently distributed throughout the grid.

Transmission: From power plants to distribution substations, electricity must be transported across great distances via the transmission system. Electricity is transmitted through high voltage transmission lines, often called power lines or transmission lines. In order to minimise power losses during transmission, these cables run at high voltages. **Distribution:** The power is stepped down to lower voltages for safe delivery to customers once it reaches the distribution substations. Distribution lines, also known as feeder lines, distribute the electricity to homes, businesses, and other end- users. Transformers are used to further lower the voltage to a level suitable for consumer use.

Power systems consist of these three primary parts in addition to a number of control and protection measures that guarantee the system's dependable and secure operation. These mechanisms include protective relays, circuit breakers, and monitoring systems. Investigating faults in high voltage transmission lines using MATLAB simulation involves modelling the transmission system, introducing faults, and analyzing their waveforms of current and voltage. The simulation allows testing various fault scenarios, assessing protective measures, and optimizing the system's reliability.

II. OBJECTIVE

The primary aim of fault investigation on transmission lines with high voltage is to quickly identify and address any issues or abnormalities that disrupt power transmission system efficiency, dependability, and safety by monitoring the flow of electricity. This involves identifying faults like L-G, L-L, L-L-L and L-L-L-G faults.

III. FAULT TYPES

There are various forms of faults in electrical and power systems, each having unique properties and consequences. The most typical kinds of errors consist of :

1. Line – Ground Fault (L-G)

This is the most common type of fault in high voltage transmission lines. Line-to-ground faults on high voltage transmission lines is a common electrical problem that happens when one of the conductors usually one of the phase accidentally touched the ground or an object that is grounded. This fault can result from a variety of factors, including insulation degradation, physical damage to the conductor, or a breach in the protective shielding.

When a line-to-ground fault occurs, it can have significant implications for the electrical system. The fault causes a sudden increase in current flow through the affected conductor and can lead to an unbalanced condition in a three-phase system. This, in turn, can result in voltage imbalances and voltage sags in the system. Furthermore, the fault may cause thermal stress and mechanical damage to the affected conductor and associated equipment.

2. Line – Line Fault (L-L)

An electrical failure that occurs when two conductors of a three-phase system come into direct contact with one another is known as a line-to-line faults in high voltage transmission lines. There are several potential causes of this defect, such as malfunctioning equipment, insulation breakdown, or external factors like falling trees or debris. Line-to-line faults are a serious concern in high voltage transmission systems because they can result in substantial short-circuit currents and significant damage to equipment.

When a line-to-line fault occurs, it typically leads to a sudden increase in current, causing excessive heating, electromagnetic forces, and voltage instability in the affected phases. This can result in a cascading effect, affecting the overall stability and operation of the power grid.

3. Line – Line – Line Fault (L-L-L)

Three-phase faults, or line-to-line faults, are one of the largest and least common electrical problems that can occur on major power lines. A short circuit is the result of all three phases of a three-phase electrical system coming into direct contact with one another. An extensive and possibly disastrous three-phase fault can have serious repercussions.

During this fault, the voltage levels can rise to extremely high levels, creating an intense electrical current surge. This can lead to the rapid overheating and damage of transmission line components, including conductors, insulators, and other equipment. The resulting high current can also affect transformers and generators connected to the transmission line, potentially causing widespread power outages.

4. Line – Line – Ground Fault (L-L-L-G)

In a high-voltage transmission lines, a "three line-to-ground" fault is a complicated electrical fault condition that occurs when three conductors (typically associated with a three-phase power system) simultaneously develop a short circuit to the ground. This type of fault is relatively rare but can have significant consequences for the power system.

In such a fault, electrical energy from all three phases is exposed to the ground or to a conductive structure, resulting in a substantial current surge. This fault can be caused by various factors, including insulator failure,

conductor damage, or equipment malfunction. When it occurs, it can lead to immediate and severe consequences, including electrical arcing, equipment damage, and potential power outages.

IV. FAULT ANALYSIS

Analyzing transmission line faults on transmission lines with high voltage is crucial for ensuring the reliability, safety, and optimal performance of the electrical grid. Here is some steps involved in the analysis of transmission line faults:

1. Fault Detection

As part of the analysis process, fault detection comes first. Modern power systems use protective relays and monitoring systems to detect abnormal conditions in transmission lines.

Protective relays continuously monitor voltage, current, and other parameters. When a fault is detected, the relay initiates protective actions to isolate the fault and minimize damage.

2. Fault Localization

Finding the fault's position along the transmission line comes next when a fault has been identified. Accurate fault localization is critical for minimizing downtime and ensuring efficient repairs.

Techniques for fault localization include time-domain reflectometry, which measures the time it takes for a signal to reflect off the fault location, and impedance-based methods.

3. Fault Classification

Faults can be classified as symmetrical (three-phase) or unsymmetrical (single-phase or double-phase). Classifying the fault type helps in selecting appropriate protective actions.

It is crucial for mitigation to know if the fault is a single phase-to-ground, line-to-line, or another kind.

4. Analysis

Detailed fault analysis involves understanding the magnitude and duration of fault currents, fault impedance, and the type of fault (L-G, L-L, etc.).

With the help of Computer-based simulation analyse the voltage and current waveform, and assess the impact on the power system.

VI. EFFECT OF FAULTS

High voltage transmission line faults can have various effects on the power system, including electrical, operational, and safety consequences. These effects can range from momentary disturbances to prolonged outages, and they can impact both the transmission system and downstream distribution systems. Here are some effects of faults in high voltage transmission lines:

1. Voltage and Current Disturbances

Faults cause sudden changes in current and voltage levels. Voltage sags (dips) and surges can occur, affecting the quality of power supplied to consumers.

2. Overcurrent and Overload Conditions

Faults often result in overcurrent conditions, which can damage equipment and pose safety risks. Overloads can stress transformers, conductors, and other components.

3. Power Interruptions

Depending on the fault type and the effectiveness of protective devices, faults can lead to temporary or prolonged power interruptions in the affected area, affecting businesses, industries, and households.

4. Equipment Damage

Faults, especially short circuits, can cause mechanical and thermal stresses on equipment, leading to damage or failure of circuit breakers, transformers, and other components.

5. Economic Losses

Extended power interruptions and equipment damage due to faults can result in significant economic losses for industries and businesses.

6. Operational Challenges

Transmission line faults can create operational challenges for power utilities, requiring them to deploy maintenance crews, reroute power flows, and coordinate restoration efforts.

VIII. SIMULATION MODEL AND RESULT

1. L-G Fault (Single Phase Fault)

Fig.1 Simulation Model L-G Fault





2. L-L Fault (Double Line Fault)



6	- Char
tak:	7.10
- i.	
hil-	
	HUNDERSON AND AND AND AND AND AND AND AND AND AN
	The second s
1	
	THROAD AND AND AND AND AND AND AND AND AND A

Fig.5 Output Waveform of Current (L-L Fault)

3. L-L-L Fault (Three Phase Fault)



Fig.6 Output Waveform of Voltage (L-L-L Fault)

Fig.7 Output Waveform of Current (L-L-L Fault) 4. L-L-G Fault (Three Phase to Ground Fault)



Fig.8 Output Waveform of Voltage (L-L-L-G Fault)



Fig.9 Output Waveform of Current (L-L-L-G Fault)

VIII. OUTCOMES

When conducting fault investigation in a high voltage transmission line using MATLAB, you can expect to achieve several outcomes, including:

1. Fault Detection and Localization

MATLAB can be used to process data from sensors and measurement devices to identify and localize faults in the transmission lines. Various algorithms and techniques can be employed for this purpose, such as Impedance-based method. Fault detection and localization refer to the process of identifying an issue with a power system occurring and determining its specific location.

2. Fault Classification

MATLAB can help you classified fault types that has occurred, such as line to line, two lines to ground, line to ground, etc. Classification is essential for implementing appropriate protection and mitigation strategies.

3. Fault Simulation

You can evaluate the effects of different fault on the performance of the transmission lines by simulating fault scenarios using MATLAB. This is important to know in order to comprehend how the system operates when there is a failure. Fault simulation refers to the process of simulating or modelling faults in a power system to understand their impact and behaviour. It involves creating computer-based models that replicate the electrical components and network of the power

system.

4. Real-time Monitoring

To continually monitor the transmission line and react to disturbances in real-time, MATLAB can be connected with real-time data acquisition devices.

Continuous and instantaneous observation of a system or process as it takes place is known as real-time monitoring. It entails the real-time collection and analysis of data, enabling the prompt identification and handling of any alterations or irregularities.

IX SCOPE

The future scope of investigating high voltage transmission line faults using MATLAB is promising and will continue to evolve as technology advances. MATLAB is a powerful tool for, which simulates, analyzes and optimizes electrical systems, making it a valuable resource for investigating faults in high- voltage transmission lines.

1. Advanced Modelling

Creating more complex and precise models to represent fault analysis, transient behaviour, and power flow in transmission lines in an effort to comprehend the system on a deeper level.

2. Smart Grid Integration

Designing MATLAB-based tools for integrating high- voltage lines into smart grids, optimizing operations, and managing renewable energy sources efficiently.

3. Fault Analysis and Mitigation

Creating enhanced algorithms and tools in MATLAB for faster and more precise fault detection, classification, and mitigation strategies.

4. HVDC Transmission

Advancing MATLAB simulations for HVDC (High Voltage Direct Current) transmission systems to improve efficiency, control, and stability.

X. CONCLUSION

In conclusion, fault investigation on transmission line with high voltage is crucial for ensuring the power distribution is reliable and stable. Waveforms of current and voltage can be used to analyse fault classification. And then fixing them fast to keep electricity flowing smoothly. By getting better at this process, we make sure power stays reliable for everyone.

XI. REFERENCES

- Bon, N. N. (2022). Deep Learning Method for Fault Diagnosis in High Voltage Transmission Lines: A Case of the Vietnam 220kV Transmission Line. International Journal on Electrical Engineering and Informatics, 14(2), 254-275.
- Oni, O. E., Davidson, I. E., & Mbangula, K. N. (2016, June). Dynamic voltage stability studies using a modified IEEE 30-bus system. In 2016 IEEE 16th International Conference on Environment and Electrical Engineering (EEEIC) (pp. 1-6). IEEE.
- 3. Salat, R., & Osowski, S. (2004). Accurate fault location in the power transmission line using support vector machine approach. IEEE Transactions on power systems, 19(2), 979-986.
- 4. Wang, B., Yang, L., & Dong, X. (2018). Fault location method for high-voltage direct current transmission line using incident current travelling waves. The Journal of Engineering, 2018(15), 1165-1168.
- 5. LIANG, W. L., & SHANG, L. Q. (2014). The review of high voltage DC transmission lines fault location. methods, 13, 15.
- 6. M Bockarjova, A. Sauhats, G. Andersson, Statistical Algorithms for Fault Location on Power Transmission Lines, Proceedings of 2005 IEEE PowerTech Conference, June 27-30, 2005, St. Petersburg, Russia.
- 7. A. Sauhats, M. Danilova, Fault Location Algorithms for Super High Voltage Power Transmission Lines, Proceedings of 2003 IEEE PowerTech Conference, Bologna, Italy.
- H. A. Darwish, A. M. I. Taalab, A. H. Osman, N. M. Mansour, O. P. Malik, Experimental Evaluation of DWT-Based Current Spike Comparator for Line Protection, Power Systems Conference and Exposition, PSCE '06, 2006, pp. 1371-1379.
- 9. C. Aguilera, E. Orduna, G. Ratta, Adaptive Non Communication Protection Based on Traveling Waves and Impedance Relay, IEEE Transactions on Power Delivery, vol. 21, n.3, 2006, pp. 1154-1162.
- P. F. Gale, P. V. Taylor, P. Naidoo, C. Hitchin, and D. Clowes, Traveling Wave Fault Locator Experience on Eskom's Transmission Network, 7th IEE International Conference on Developments in Power System Protection, 2001.
- 11. B. Y. Su, P. F. Gale, Y. Z. Ge, Fault Location Based on Fault Induced Current Transients, International Conference on New Developments in Power system Protection and Local Control, Beijing, China, 1994.

- 12. J. M. Samper, J. M. Lagunilla, R. B. Perez, GPS and GalileoL Dual RF Front-End Receiver Design, Fabrication, and Testing, Communication Engineering, McGraw-Hill Professional, 2008.
- 13. A. Abu-Siada and S. Mir, "A new on-line technique to identify fault location within long transmission lines", Engineering Failure Analysis, vol. 105, pp. 52-64, 2019.
- M. Parsi, P. Crossley, P. Dragotti and D. Cole, "Wavelet based fault location on power transmission lines using real-world travelling wave data". Electric Power Systems Research, vol. 186, 2020.
- 15. M. Ghazizadeh-Ahsaee, "Time-domain based fault location for series compensated transmission lines without requiring fault type". Electric Power Systems Research, 181, 2019.
- E. Akher et al., "Fault Detection and Classification of Shunt Compensated Transmission Line Using Discrete Wavelet Transform and Naive Bayes Classifier," Energies, vol .13, p. 243, 2020.
- M. Choudhury and A. Ganguly, "Transmission line fault classification using discrete wavelet transform," International Conference on Energy, Power and Environment: Towards Sustainable Growth (ICEPE), Shillong, pp. 1-5, 2015.
- I. Ibrahim, "Fault detection and location of power transmission lines using intelligent distance relay," International Journal of Power Electronics and Drive Systems. Vol. 11, pp. 726-734.
- E.P. Thwe and M.M. Oo, "Fault Detection and Classification for Transmission Line Protection System Using Artificial Neural Network," Journal of Electrical and Electronic Engineering. Vol. 4, No. 5, 2016, pp. 89-96, 2016.
- A. Aggarwal, H. Malik and R. Sharma, "Feature extraction using EMD and classification through Probabilistic Neural Network for fault diagnosis of transmission line," IEEE 1st International Conference on Power Electronics, Intelligent Control and Energy Systems (ICPEICES), Delhi, pp. 1-6, 2016.
- S.R. Fahim, Y. Sarker, S. K. Sarker, M. R.I. Sheikh, and S. K. Das, "Self-attention convolutional neural network with time series imaging based feature extraction for transmission line fault detection and classification," Electric Power Systems Research, vol. 187, 2020.
- X. Huang, E. Shang, J. Xue, H. Ding and P. Li, "A Multi-feature Fusion- based Deep Learning for Insulator Image Identification and Fault Detection," IEEE 4th Information Technology, Networking, Electronic and Automation Control Conference (ITNEC), Chongqing, China, pp. 1957-1960, 2020.
- M. Salehi, et al. "Determining minimum number and placement of fault detectors in transmission network for fault location observability," International Journal of Electrical Power and Energy Systems, vol. 124, 2020.
- S. Rajput and K.T. Chaturvedi, "Transmission Line Fault Analysis by Using Matlab Simulation," IJREE -International Journal of Research in Electrical Engineering, vol. 3, no. 4, 2016.
- K. Sun et al., "A Cable Fault Identification and Location Method Based on HEM," 5th Asia Conference on Power and Electrical Engineering (ACPEE), Chengdu, China, 2020, pp. 1657-1661, 2020.
- S. Devi, N. Swarnkar, S. Ola and O. Maheka, "Analysis of transmission line faults with linear and dynamic loads" pp. 99-103, 2016.
- 27. D. Boora, "Matlab Simulation Based Study of Various Types of Faults Occurring in the Transmission Lines," International Journal of Engineering Research, vol. 8.
- M. Ikhide, S. Tennakoon, A. Griffiths, S. Subramanian and H. Ha, "Fault detection in Multi-Terminal Modular Multilevel Converter (MMC) based High Voltage DC (HVDC) transmission system," 50th International Universities Power Engineering Conference (UPEC), Stoke on Trent, pp. 1-6, 2015.
- A. S. Poste, B. T. Deshmukh and B. E. Kushare, "Detection, classification & characterization of voltage sag," 2016 International Conference on Electrical, Electronics, and Optimization Techniques (ICEEOT), Chennai, pp.232-237, 2016.
- S. Devi, N. K. Swarnkar, S. R. Ola and O. P. Mahela, "Analysis of transmission line faults with linear and dynamic loads," Conference on Advances in Signal Processing (CASP), Pune, pp. 99-103, 2016.
- H.H Goh et al., "Fault Location Techniques in Electrical Power System: A Review," Indonesian Journal of Electrical Engineering and Computer Science, Vol. 8, No. 1, pp. 206 - 212, 2017.

A Review of Innovations in EV Battery Charging Electrification

V. S. Biradar¹, A. A. Kotmale², M. T. Shinde³, S S Kulkarni⁴

¹⁻⁴N.B. N. S. C. O. E, Solapur
¹vsbiradar.nbnscoe@gmail.com
²aakotmale.nbnscoe@singhgad.edu
³mtshinde.nbnscoe@gmail.com
⁴sskulkarni.nbnscoe@sinhgad.edu

Abstract—For the time being, the automobile industry is changing a lot, especially towards electric vehicles (EVs) to help the environment. As more people start using EVs, there's a bigger need for better ways to charge them. This study looks into a new idea called "Battery Charging Optimization" to make EV battery charging better in the future. The main goal is to make charging faster and more effective while running. The research checks out the newest technologies, looks into how we can connect charging systems with smart charging alternator, and figures out ways to handle changes in charging power. By looking at what experts have already written, studying real examples, and checking out actual data, this research wants to give useful insights into making EV battery charging better, especially focusing on a part called the alternator charging system. The results show that there are good things happening, real stories of successful changes, and a clear understanding of the challenges and opportunities we might face. Considering the whole world is starting to use more EVs, this study emphasizes how important it is to make charging better and more effective for a future where electric vehicles are used everywhere.

Index Terms— Electric vehicles (EVs), charging technologies Electrify, Optimize, Alternator charge, Artificial Intelligence (AI), Battery Management System (BMS)

I. INTRODUCTION:

This change is mainly because we need transportation that's good for the environment and climate control. With more electric cars on the road, there's a big need for better ways to charge them. This study looks into a new idea called "Electrify & Optimize" as a smart way to influence how we charge EV batteries in the future. The main goal is to make charging faster and better because how well EVs work depends a lot on how well we can charge them. The study explores using the latest technologies, connecting charging to smart grids, and finding smart ways to deal with the increasing demand for EV charging. The main focus is to understand all the different aspects of making EV battery charging better, including using new tech, connecting to smart alternators, and finding ways to meet the demand for charging. By looking at what experts have already written, studying real examples, and checking out actual data, this research wants to give useful insights into making EV running battery charging better or not. The expected outcomes include insights into good things happening, examples of successful changes, and a better understanding of the challenges and opportunities ahead. Considering the whole world is starting to use more EVs, this study shows how important it is to make charging better and more efficient for a future where electric vehicles are used everywhere. Additionally, the paper briefly explores how using alternator charging systems might impact EVs, aiming to make them cover more distance with each charge. This dual focus looks at both the technology and efficiency to improve electric mobility. Through a meticulous review of existing literature, the analysis of case studies, and the examination of empirical data, this research seeks to make a substantial contribution to our understanding of the intricate dynamics involved in optimizing running EV battery charging. The expected outcomes may include insights into encouraging progress, instances of successful implementation, and a deeper understanding of the challenges and opportunities that lie ahead. Within the context of the global shift towards widespread EV adoption, this study highlights the pivotal role played by electrification and optimization in shaping the future of sustainable and effective electric vehicle charging. Additionally, the review briefly explores the potential impact of alternator charging systems on EVs, with the aim of enhancing their average distance coverage per charge. This dual focus reflects a holistic approach in addressing both technological and efficiency aspects in the pursuit of advancing the field of electric mobility.

II. BACKGROUND

The car industry is currently experiencing a notable transformation as it shifts towards electric vehicles, highlighting a growing demand for more efficient batteries. While electric vehicles offer a cleaner and sustainable form of transportation, their widespread adoption faces challenges related to battery performance and charging infrastructure. The conventional ways of charging have limitations in terms of speed, range, and overall efficiency. This has created an urgent need for inventive solutions to improve the range and overall performance of electric vehicle (EV) batteries. In response to these challenges, the research on "Charge Evolution with optimization" is committed to investigating unconventional methods that enhance the efficiency of electric vehicle batteries simultaneously with the battery charging system.

Context of the Research

The study is set within the dynamic landscape of electric mobility, where advancements in technology play a crucial role in shaping the future of transportation. The framework also acknowledges the significance of concurrent charging through an alternator, introducing a dual emphasis on enhancing battery efficiency while the vehicle is in motion. As electric vehicles gain increasing popularity, there is a growing need for solutions that not only boost the overall effectiveness of battery systems but also introduce innovative charging methods beyond the conventional stationary charging stations. The integration of alternator-based charging systems opens up an intriguing pathway for simultaneously powering the vehicle and recharging the battery, potentially addressing concerns related to both range anxiety and charging infrastructure. To sum up, the research on "Charge Evolution" is strategically positioned within the evolving electric vehicle industry with the aim of contributing to the future of transportation. It seeks to explore novel possibilities in electric vehicle battery efficiency and assess the viability of concurrently charging the battery using alternator technology.

Statement of the problem

The matter at hand revolves around the premature degradation of electric vehicle (EV) batteries, resulting in reduced performance, particularly a decrease in the distance covered per single battery charge. This decline poses a significant challenge to the overall efficiency and practicality of electric vehicles. In response to this issue, a novel approach is proposed: utilizing two batteries with independent charging facilitated by an alternator system, aiming to potentially improve overall efficiency. This prompts the need to examine and determine whether implementing such a configuration, involving dual batteries and alternator-based charging, can effectively address the performance deterioration associated with early EV.

Objectives of the study

Assessing Early EV Battery Performance: Conduct a thorough investigation to evaluate and quantify the extent of performance degradation observed in electric vehicle (EV) batteries during their initial lifecycle. Specifically, focus on determining the reduction in the average kilometres covered using a single battery over time. Evaluate the Feasibility of Dual-Battery Configuration: Scrutinize and analyse the practicality of implementing a dual-battery configuration in electric vehicles as a potential solution to counteract the observed early deterioration
witnessed in single-battery scenarios. Explore the potential benefits and drawbacks of such a configuration. Analyse Alternator-Based Charging System: Systematically assess the effectiveness and efficiency of a novel charging system that utilizes an alternator for standby battery in the dual-battery configuration. This analysis will delve into its impact on various parameters, including charging speed, energy replenishment rates, and the overall health of the batteries. Measure Overall Efficiency Improvement: Utilize quantitative metrics to measure and analyses the comprehensive efficiency gains resulting from the implementation of the dual-battery configuration with separate charging through an alternator system. Place specific emphasis on evaluating improvements in the average kilometres covered by the electric vehicle. Investigate System Integration and Compatibility: Thoroughly explore the integration aspects and compatibility of the proposed dual-battery setup with existing electric vehicle systems. This investigation will ensure seamless operation and adherence to stringent safety standards within the electric vehicle ecosystem.

Examine Cost Implications

Conduct a detailed economic assessment to evaluate the financial feasibility of introducing and maintaining a dual-battery and alternator-charging system. Factors such as manufacturing costs, on-going maintenance expenses, and potential cost savings associated with extended battery life will be meticulously examined. Assess Environmental Impact: Investigate the environmental implications associated with the proposed dual-battery system, considering aspects such as resource utilization, recycling requirements, and the broader sustainability of the technology. This examination aims to provide a holistic understanding of the ecological footprint of the proposed system. Optimize Charging Strategies: Develop and optimize charging strategies tailored to the dual-battery configuration. Consider factors such as charging intervals, balancing mechanisms between the batteries, and energy distribution strategies to enhance the overall efficiency of the charging process. Provide Recommendations for Implementation: Based on empirical findings, formulate practical and actionable recommendations for the seamless implementation of the dual-battery and alternator-charging system in electric vehicles [31. These recommendations will address potential challenges and capitalize on identified opportunities to enhance overall efficiency and mitigate early EV battery decline. Contribute to Knowledge Base: Disseminate research findings to contribute novel insights and knowledge to the evolving field of electric vehicle technology. The aim is to inform and potentially inspire future advancements and innovations in EV battery systems and charging methodologies, fostering continued progress within the industry.

III. LITERATURE REVIEW

Advancements in Battery Chemistry: Studies have explored innovative battery chemistries such as solid-state batteries, lithium-sulfur batteries, and metal-air batteries, aiming to enhance energy density, lifespan, and overall performance in electric vehicles.

Smart Battery Management Systems (BMS): The literature highlights the role of intelligent Battery Management Systems in optimizing electric vehicle battery performance. Innovations in BMS focus on real-time monitoring, predictive analytics, and adaptive control algorithms to maximize battery efficiency and longevity. Fast Charging Technologies: Research has investigated fast charging technologies to address charging time concerns. Innovations include ultra-fast charging stations, advanced cooling systems, and high-power charging protocols to reduce charging durations and improve the practicality of electric vehicles.

Energy Harvesting and Regeneration: Studies explore the integration of energy harvesting technologies, such as regenerative braking and solar panels, to supplement electric vehicle battery charging. These innovations aim to improve overall efficiency and extend the driving range of electric vehicles. Thermal Management Systems The literature emphasizes advancements in thermal management systems to mitigate issues related to battery overheating and degradation. Innovations include phase change materials, liquid cooling, and thermal energy recovery mechanisms to optimize battery temperature [1], [2].

Machine Learning and Artificial Intelligence (AI): Innovations in machine learning and AI applications have been explored for predictive modelling, optimization of charging patterns, and adaptive control strategies. These

intelligent systems aim to tailor the charging and discharging cycles based on user performance and external factors. Second-Life Battery Applications: Research investigates the potential of repurposing electric vehicle batteries for secondary applications after their primary use. Innovations include grid storage, stationary energy systems, and backup power solutions, contributing to sustainable battery usage. We introduce the EV Charging Profiles and Waveforms (EV-CPW) dataset, which is now accessible to the public. This dataset encompasses charging profiles and detailed transient current/voltage waveforms for 12 electric vehicles (EVs). A comprehensive power quality analysis, encompassing power factor, distortion, harmonic analysis, and load behaviour, is conducted on the 12 EVs. The results are then compared to the EV standard recommendations put forth by standards agencies. It's important to note that an off-board charger, as illustrated in Fig. 1, is employed for L3 charging. Consequently, the grid impact resulting from L3 charging is determined by the charger design. Conversely, for L2 charging, the impact is determined by the on-board charger design of the EV, which can vary based on different topologies and controllers [21]. Both Battery Electric Vehicles (BEVs) and Plug-in Hybrid Electric Vehicles (PHEVs) are compatible with AC charging, while a smaller subset of EVs, typically only BEVs, can utilize DC charging. Level 2 charging has been more widely adopted for public charging, possibly due to its lower cost compared to Level 3 chargers. This paper specifically focuses on collecting EV charging data from AC charging, accomplished by charging EVs on Level 2 chargers [11].



Fig. 1. Simplified diagram of L2 and L3 charger connections

Wireless Charging Technologies: Literature explores wireless charging technologies as an innovative approach to enhance user convenience and eliminate physical connectors. Advances in inductive and resonant wireless charging systems aim to optimize the charging experience for electric vehicle owners.

Material Innovations:

Material science plays a crucial role in battery optimization. Studies delve into innovations in electrode materials, electrolytes, and separators to improve energy density, charge/discharge rates, and overall battery performance in electric vehicles. Block chain and Decentralized Charging Networks: The literature discusses innovations in decentralized charging networks using block chain technology. This approach aims to optimize charging infrastructure, enable peer-to-peer energy trading, and enhance the overall efficiency of electric vehicle charging ecosystems. Dynamic Pricing Strategies: Studies explore dynamic pricing strategies for electric vehicle charging, considering factors such as grid demand, renewable energy availability, and time-of-use pricing. Innovations in pricing models aim to incentivize off-peak charging and promote grid stability. Collaborative Research and Industry Partnerships [24].

Literature highlights collaborative efforts between research institutions, industry players, and government agencies to drive innovations in electric vehicle battery optimization. These partnerships contribute to shared knowledge, resources, and accelerated advancements in the field.

The current body of literature underscores the multifaceted exploration of electric vehicle (EV) battery charging efficiency. Researchers emphasize the pivotal role of developing a resilient charging infrastructure and integrating it seamlessly with the electrical grid. This encompasses the investigation of smart charging solutions, grid management techniques, and the overarching impact of charging infrastructure on the overall efficiency of EV battery charging. Smart charging algorithms and control strategies emerge as key focal points in the literature, addressing the need for real-time monitoring, predictive analytics, and adaptive control systems. These systems take into account variables such as battery state, grid conditions, and user preferences to optimize the charging process. Energy management strategies are explored, with a specific focus on peak shaving to alleviate grid load during high-demand periods. This involves the optimization of charging schedules and distribution to enhance overall system efficiency. Moreover, the integration of renewable energy sources, such as solar and wind, into EV charging stations is extensively discussed to improve sustainability and reduce the carbon footprint associated with EV charging as shown in Fig. 2

Battery charging curve:



Fig. 2. Battery Charging Curve

The optimization of thermal management plays a pivotal role in discussions surrounding charging efficiency. Various studies explore advanced cooling and heating systems designed to maintain an optimal battery temperature, ensuring efficient charging while safeguarding battery health. Scrutiny of user behavior and charging patterns aims to understand their impact on charging efficiency. This encompasses investigations into user preferences, charging habits, and the potential for implementing demand response strategies to optimize charging during periods of low grid demand.

Economic feasibility studies and analyses of government policies are crucial components of existing literature. These studies delve into financial incentives, pricing models, and regulatory frameworks, all aimed at enhancing overall charging efficiency. The strategic placement of charging stations and considerations of accessibility are also vital aspects, addressing concerns related to range anxiety and promoting widespread electric vehicle (EV) adoption. The literature further explores innovations in charging cables, connectors, and materials to improve charging efficiency. On- going efforts include advancements in conductive materials, contactless charging, and enhanced power electronics. Additionally, battery swapping technologies are under examination as an alternative to traditional charging methods, seeking to reduce charging time by exchanging depleted batteries with fully charged ones. In summary, this comprehensive literature review encompasses diverse dimensions of research on the EV charging system, as illustrated in the block diagram shown in Fig. 2

Innovations in charging cables, connectors, and materials are examined to improve charging efficiency. Advancements in conductive materials, contactless charging, and improved power electronics are part of on-going efforts to enhance the overall efficiency of EV charging. Battery swapping technologies are explored as an alternative to traditional charging methods, aiming to reduce charging time by swapping depleted batteries with fully charged ones. In essence, this comprehensive literature review encapsulates the diverse dimensions of research on EV Charger block diagram shown in Fig. 3



Fig. 3. EV Charger block diagram

In the realm of electric vehicle (EV) charging during vehicle operation, several challenges and gaps persist in current research, last updated in January 2023. The uneven and insufficient distribution of charging infrastructure in specific regions remains a hurdle, impacting the seamless operation of EVs and the overall user experience. Research gaps highlight the need to identify optimal strategies for increasing the density and accessibility of charging infrastructure. Factors such as urban planning, public-private partnerships, and innovative charging solutions should be considered in addressing these challenges. Issues with charging speed and range anxiety persist, influencing potential EV adopters. Research is needed to improve charging speed, advance battery technologies, and implement effective communication strategies to alleviate range anxiety. This may involve exploring ultra-fast charging, improved battery chemistries, and accurate range prediction algorithms. Battery degradation over time is another critical challenge affecting EV efficiency and lifespan. Research gaps call for the development of effective battery management systems, optimizing charging parameters, and minimizing degradation. Innovations in battery materials and thermal management are areas that require exploration. The integration of a growing number of EVs into the power grid brings challenges related to grid stability and increased demand. Research gaps emphasize the need for smart grid solutions, demand response strategies, and energy storage integration to balance the load and maximize the use of renewable energy sources during EV charging. Upfront costs of EVs and associated charging infrastructure, coupled with uncertainties about the long-term economic viability of charging networks, pose challenges to widespread adoption. Research efforts should focus on optimizing the cost-effectiveness of EV charging solutions, exploring innovative business models, and assessing the economic impact of large-scale EV integration. The lack of standardization and interoperability among charging networks, connectors, and communication protocols can lead to compatibility issues and hinder the ease of EV charging. Research endeavors should prioritize the development of global standards, common protocols, and interoperable technologies to create a seamless charging experience for EV users [13],[14].

A. Innovative Battery Chemistries

Advancements in Lithium-Ion Technology: On-going research focuses on enhancing the energy density, charging speed, and lifespan of lithium-ion batteries, the predominant type used in electric vehicles (EVs). Innovations include the integration of silicon anodes, solid-state electrolytes, and advanced materials to improve overall performance. Solid-State Battery Technologies: Solid-state batteries present the potential for increased energy density, enhanced safety, and extended lifespan compared to traditional liquid electrolyte batteries.

B. Efficient Battery Thermal Management:

Utilization of Advanced Thermal Management Systems: Advanced systems regulate EV battery temperature, ensuring optimal operating conditions. Technologies such as liquid cooling, phase-change materials, and active thermal control systems are employed to prevent overheating and enhance overall battery performance.

C. Rapid Charging Innovations:

Ultra-Fast Charging Technologies: Initiatives aim to significantly reduce charging times through the development of high-power chargers and advancements in charging protocols. Technologies based on gallium nitride (GaN) are explored to enable faster charging without compromising battery health.

D. Intelligent Battery Management Systems (BMS):

Smart BMS Optimization: Intelligent BMS optimizes charging and discharging processes, monitors real-time battery health, and implements strategies to prevent overcharging or deep discharging. Machine learning algorithms within BMS adapt to user behaviors, enhancing overall efficiency.

E. Energy Harvesting and Regeneration:

Regenerative braking systems are designed to capture and transform kinetic energy generated during braking into electrical energy, which is then redirected back into the battery. This innovative technology enhances overall efficiency by harnessing energy that would otherwise dissipate as heat during the braking process.

F. Wireless Charging Advancements:

Inductive and Resonant Wireless Charging: Wireless charging technologies eliminate the need for physical connectors, providing convenient and efficient charging. On-going research focuses on improving efficiency and standardization of wireless charging systems for EVs.

G. Second-Life Battery Applications:

Repurposing Used EV Batteries: Exploration of secondary applications, such as stationary energy storage, extends the lifespan of used EV batteries, contributing to sustainable energy solutions and minimizing waste.

H. Predictive Analytics and Machine Learning:

Enhanced Battery Management: Predictive analytics and machine learning algorithms analyse vehicle, charging infrastructure, and external data to predict user behaviour and optimize charging schedules, enhancing energy efficiency.

I. Grid Integration and V2G Technology:

Vehicle-to-Grid (V2G) Technology: Bidirectional energy flow between EVs and the electrical grid enables EVs to provide support during peak demand, potentially earning revenue for EV owners.

J. Advanced Electrode and Electrolyte Materials:

- Development of Advanced Materials: Research focuses on advanced materials for electrodes and electrolytes to improve energy density, charge/discharge rates, and overall battery performance.

K. Block chain Integration for Security:

- Secure Charging Ecosystem: Block chain technology secures and streamlines transactions within the EV charging ecosystem, enhancing transparency, security, and efficiency.

L. Dynamic Charging Systems Implementation:

- Charging While in Motion: Dynamic charging systems allow EVs to charge while in motion, addressing range anxiety concerns and enabling continuous operation. This technology is explored for public transportation and high-traffic routes.

M. Optimized Charging Strategies:

- Intelligent Charging Approaches: Development of intelligent charging strategies, including time-of-use charging and demand response mechanisms, optimizes energy consumption, reduces peak loads on the grid, and lowers charging costs for EV owners. [23].

N. Environmental Sustainability Measures:

- Sustainable Practices: Sustainable manufacturing processes, recycling programs, and eco-friendly disposal methods contribute to the environmental sustainability of EV batteries. Researchers explore ays to minimize the environmental impact of battery production and end-of-life processes.

III. METHODOLOGY

Improving the efficiency of electric vehicle (EV) battery charging and enhancing the overall efficiency of EVs in terms of average kilometres (KM) involves a multi-faceted methodology. Below are key approaches: Advanced Charging Infrastructure: Establish an extensive and advanced charging infrastructure network, including fast-charging stations strategically placed for convenient access. Utilize high-power chargers and implement technologies that enhance charging speed without compromising battery health. Smart Charging Algorithms: Develop intelligent charging algorithms that consider factors such as grid demand, energy prices, and optimal charging times. Incorporate machine learning to analyse user behaviour and adapt charging schedules for maximum efficiency.

Charging methods employed in on site chargers utilizing AC power adhere to standardized protocols. Level 1 charging, utilizing a 120 V single-phase AC power supply, boasts the slowest charging speed and is typically employed in domestic settings with lower power levels (up to 1.92 kW). This charging mode, suitable for extended periods or overnight charging, requires about 11-36 hours for a 1.9 kW charging power for a 16-50 kWh EV battery [26]. For private and public facilities, the primary choice is Level 2 chargers, known for their comparatively faster charging abilities. Level 2 charging, 3 to 5 times faster than Level 1, can deliver power up to 19.2 kW, requiring dedicated components and installations for high- power transfer. The charging time for Level 2 falls within the range of 2-3 hours for a 19.2 kW charger with an EV battery capacity of 30-50 kWh [27]. Charging connectors for Level 1 and 2 adhere to standards such as IEC62196-2 in Europe and SAEJ1772 and Tesla superchargers in the USA [25], [28].

DC fast charging, or Level 3 charging, integrates both AC and DC power to supply high-voltage DC power directly to the EV battery. Level 3 chargers, connected to the vehicle through off- board chargers, manage a high-power range from 20 kW to 350 kW, supplying DC voltage ranging from 300 Vdc to 800 Vdc. Charging times for Level 3 are notably faster, with a range of

0.2 - 0.5 hours for a 90 kW or larger charger. Connectors such as CHAdeMO, Tesla company superchargers, and CCS combo 1, 3 are commonly used for Level 3 speedy charging. Despite their faster charging times, Level 3 chargers may exert a higher negative impact on the power network during peak times, potentially causing overload in the local distribution grid in Fig. 4



Fig. 4. Electric vehicle charging methods.

Extreme fast charging (XFC) systems represent a paradigm shift in charging experiences, mirroring the refuelling speed of internal combustion engine (ICE) vehicles. These systems can manage power exceeding 350 kW with an internal DC bus voltage of 800 Vdc, enabling a rapid battery recharge time of approximately 5 minutes. XFC stations incorporate advanced power electronic modules, including solid-state transformers (SST), isolated DC-DC converters, and front-end AC-DC converter steps and controllers. While the setting up cost of XFC is significant, combining multiple XFC systems in a station design offers the potential to reduce operational and capital investments, increasing economic feasibility. Furthermore, SST technology provides advantages over traditional

line-frequency transformers in XFC stations by converting medium voltage levels into low voltage levels and providing galvanic isolation [29].

Dynamic Charging Systems:

Explore the implementation of dynamic charging systems that enable EVs to charge while in motion. This addresses range anxiety concerns and ensures continuous operation, especially on high-traffic routes and public transportation networks.

Wireless Charging Technologies:

Investigate and enhance wireless charging technologies, allowing for seamless and efficient charging without physical connections. Optimize the efficiency of inductive and resonant wireless charging systems for widespread adoption. Regenerative Braking Optimization: Improve regenerative braking systems to capture and convert more kinetic energy into electrical energy during deceleration. Enhance the efficiency of energy recovery systems to maximize the range extension through regenerative braking[24].

Battery Thermal Management:

Implement advanced thermal management systems to regulate battery temperature effectively. Optimize cooling and heating mechanisms to maintain an optimal temperature range, ensuring efficient charging and discharging processes.



Fig. 5. EV Forecast to 2035

Integrate V2G technology to enable bidirectional energy flow between EVs and the grid. Allow EVs to feed excess energy back to the grid during peak demand, contributing to grid stability and potentially earning revenue for EV owners.

Enhanced Battery Management Systems (BMS):

Develop advanced BMS with real-time monitoring and adaptive control capabilities. Implement features that optimize charging and discharging parameters based on battery health, usage patterns, and external conditions in Fig. 5

User Education and Engagement: Launch educational programs to inform EV users about efficient charging practices. Encourage users to adopt optimal charging habits, such as avoiding frequent deep discharges and utilizing off-peak charging hours. Global market will lead EV technology up to 2035.

IV. ELECTRIFY & OPTIMIZE TECHNOLOGIES

- Explore cutting-edge technologies for enhancing EV battery charging
- Discuss advancements in charging infrastructure.
- Analyse smart grid integration and demand response strategies.

• Collaborate with utilities to implement incentive programs that encourage off-peak charging.

• Provide reduced electricity rates during specific hours to shift charging demand away from peak periods, benefiting both users and the grid.

• Collaboration with Renewable Energy Sources:

• Integrate EV charging infrastructure with renewable energy sources, such as solar and wind.

• Prioritize the use of clean energy for charging, reducing the environmental impact and promoting sustainable practices.

• Continuous Research and Development: Invest in on-going research and development to explore emerging technologies, materials, and methodologies that can further improve EV battery efficiency and charging capabilities.

• Describe the research design detail the methods used for data collection, explain the tools and technologies employed Provide a rationale for the chosen methodology

VI. CASE STUDIES

Tesla Supercharger Network:

Background: Tesla has implemented a widespread Supercharger network globally, aiming to provide convenient and rapid charging for Tesla EV owners. Case Study: Tesla's Supercharger stations, strategically placed along highways, facilitate fast-charging for long-distance travel. This case study

delves into the network's impact on EV adoption, user experience, and the scalability of fast-charging infrastructure. Case Study: This case study examines the practical application of V2G in a fleet of Nissan Leaf vehicles. It assesses the impact on grid stability, potential revenue generation for fleet operators, and the overall feasibility and benefits of bidirectional energy flow.

Smart Charging Algorithms in City EV Fleets: Background: An electric taxi company integrates energy harvesting technology, such as solar panels on vehicle roofs, to supplement battery charging. Regeneration data is scrutinized to evaluate the efficiency of regenerative braking systems, considering factors such as energy recovered and braking patterns. The contribution of regenerative braking to the overall range extension of the EV is assessed, along with the influence of driving habits and terrain on regeneration efficiency. Comparative analyses across different EV models offer insights into variations in charging and regeneration performance [30].

The findings lead to personalized recommendations for users to optimize charging habits and driving behaviour. Infrastructure providers receive insights on usage patterns for expansion or improvement. Automakers benefit from data-driven suggestions for enhancing regenerative braking systems and overall charging efficiency. Policymakers gain insights for informed decisions on EV infrastructure development and incentives. Environmental impact analysis considers factors such as the energy source for charging and reductions in brake wear, ensuring a holistic understanding of EV charging and regeneration dynamics. This data-driven approach supports stakeholders in making informed decisions for the continued advancement of the electric vehicle ecosystem. Interpret the results in the context of EV battery charging optimization use charts, graphs, and statistical analysis to support findings.

VI. CHALLENGES AND FUTURE DIRECTIONS

Identify challenges encountered during the research Propose recommendations for overcoming these challenges. Discuss potential future research directions EV stock is significantly increased in 2021 when compared to previous years and the total number of battery electric cars on road to over 16.5 million. As shown in Fig. 6, the largest EV market belongs to China where cumulative EV sales reached 9.4 million in 2021, which represented 50% of global EV stock [14]. The second largest EV market belongs to Europe with 2.3 million annual sales of light duty EVs and the United States has the third largest EV market [12], [15]. Currently, The focus on electrified transportation has garnered significant interest from both governmental bodies and private entities aiming to achieve carbon neutrality by 2040. This commitment is evident through sustained policy backing, incentives, and subsidies, as depicted in Figure 6. Overcoming the substantial hurdle lies in implementing safe measures to charge batteries while electric

vehicles are in motion. electrified transportation has attracted much attention from governments and private stakeholders to move towards carbon neutrality in 2040 through consistent policy support, incentives, and subsidies from the governments as shown below Fig. 6. The big challenge is to charge batteries while running EV with all safety measure.



Figure 6. Electric passenger car stock, 2012-2021 [22].

VII. CONCLUSION

The integration of renewable energy sources into the charging ecosystem is essential to minimize the carbon footprint of EVs. Utilizing solar, wind, and other sustainable energy solutions establishes a symbiotic relationship between clean power generation and electric transportation, contributing simultaneously battery charging system.

Successful realization of this vision requires collaboration among governments, industries, and research institutions to foster the development and standardization of dynamic charging system. Policies incentivizing the establishment of charging stations, supporting research and development in battery technology and encouraging investments in sustainable energy play pivotal roles in shaping the future of EV charging. In essence, the electrification of transportation through optimized EV battery charging signifies a commitment to a more sustainable, resilient, and environmentally conscious future. As innovation and investments in this crucial facet of the electric mobility ecosystem persist, we are paving the way for a world where clean and efficient transportation is not only feasible but becomes the prevailing norm. The journey towards an electrified future of transportation is underway, and the evolution of EV battery charging stands at the forefront of this transformative endeavor.

V. REFERENCES

- S. Shahriar, A. R. Al-Ali, A. H. Osman, S. Dhou, and M. Nijim, "Prediction of EV charging behavior using machine learning," IEEE Access, vol. 9, pp. 111576–111586, 2021.
- C. Chen, Z. Wei, and A. C. Knoll, "Charging optimization for Li-ion battery in electric vehicles: A review," IEEE Trans. Transport. Electrific., vol. 8, no. 3, pp. 3068–3089, Sep. 2022.
- IEEE Standard for Harmonic Control in Electric Power Systems, IEEE Standard 519-2022, Revision of IEEE Standard 519-2014, 2022, pp. 1–31.
- Electromagnetic Compatibility (EMC)—Part 3–2: Limits—Limits for Harmonic Current Emissions (Equipment Input Current 16 A per Phase), IEC Standard 61000-3-2, 2020.
- Electromagnetic Compatibility (EMC)—Part 3–12: Limits—Limits for Harmonic Currents Produced by Equipment Connected to Public Low- Voltage Systems with Input Current >16 A and <75 A per Phase, IEC Standard 61000-3-12, 2021.

- 6. U.K. COP-UNFCCC Presidency. (Nov. 2021). Zero Emission Vehicle Pledges Made at COP26. [Online].
- 7. Available: https://unfccc.int/news/zero-emissionvehicle- pledges-made-at-cop26
- 8. Global EV Outlook 2021, Int. Energy Agency, Paris, France, 2021. [Online]. Available: https://www.iea.org/reports/global-ev-outlook-2021
- 9. Electric Vehicle Outlook 2021, Bloomberg NEF, Washington, DC, USA, 2021. [Online]. Available: https://about.bnef.com/electric-vehicle-outlook/
- 10. S. G. Li, S. M. Sharkh, F. C. Walsh, and C. N. Zhang, ``Energy and battery management of a plug-in series hybrid electric vehicle using fuzzy logic," IEEE Trans. Veh. Technol., vol. 60, no. 8, pp. 35713585,
- 11. Oct. 2011.
- S. Recoskie, A. Fahim,W. Gueaieb, and E. Lanteigne, ``Hybrid power plant design for a long-range dirigible UAV," IEEE/ASME Trans. Mechatronics, vol. 19, no. 2, pp. 606614, Apr. 2014. [55] A. Bruyere, L. De Sousa, B. Bouchez, P. Sandulescu, X. Kestelyn, and E. Semail, ``A multiphase traction/fast-batterycharger drive for electric or plug-in hybrid vehicles: Solutions for control in traction mode," in Proc. IEEE Vehicle Power Propuls. Conf., Sep. 2010, pp. 17.
- 13. S. Lacroix, E. Laboure, and M. Hilairet, ``An integrated fast battery charger for electric vehicle," in Proc. IEEE Vehicle Power Propuls. Conf., Sep. 2010, pp. 16.
- International Energy Agency. (2022). Global EV Outlook 2022—Securing Supplies for an Electric Future. [Online]. Available: https://www.iea.org/reports/global-ev-outlook-2022/trends-in-electriclight- duty-vehicles
- S. Rivera, S. Kouro, S. Vazquez, S. M. Goetz, R. Lizana, and E. Romero- Cadaval, "Electric vehicle charging infrastructure: From grid to battery," IEEE Ind. Electron. Mag., vol. 15, no. 2, pp. 37–51, Jun. 2021, doi: 10.1109/MIE.2020.3039039.
- H. Cui and D. Hall, "Annual update on the global transition to electric vehicles: 2021," in Proc. Int. Council Clean Transp., May 2022, pp. 1–11. [Online]. Available: https://theicct.org/wpcontent/uploads/2022/06/global-ev- update-2021-jun22.pdf
- 17. C. McKerracher, N. Soulopoulos, A. Grant, and S. Mi. (2022). Electric Vehicle Outlook 2022. [Online]. Available: https://about.bnef.com/electric-vehicle-outlook/
- A. S. Abdel-Khalik, S. Ahmed, and A. M. Massoud, "Performance evaluation of an on-board integrated battery charger system using a 12- slot/10- pole surface-mounted PM propulsion motor," presented at the 9th IEEE-GCC Conf. Exhib. (GCCCE), 2017.
- 19. P. Paum, M. Alamir, and M. Y. Lamoudi, "Probabilistic energy
- 20. management strategy for EV charging stations using randomized algorithms," IEEE Trans. Control Syst. Technol., vol. 26, no. 3, pp. 10991106, May 2018.
- C. Jin, X. Sheng, and P. Ghosh, "Optimized electric vehicle charging with intermittent renewable energy sources," IEEE J. Sel. Topics Signal Process., vol. 8, no. 6, pp. 10631072, Dec. 2014.
- 22. O. Hafez and K. Bhattacharya, "Integrating EV charging stations as smart loads for demand response provisions in distribution systems," IEEE Trans. Smart Grid, vol. 9, no. 2, pp. 10961106, Mar. 2018.
- M. Houache, C.-H. Yim, Z. Karkar, and Y. Abu-Lebdeh, "On the current and future outlook of battery chemistries for electric vehicles—Minireview," Batteries, vol. 8, no. 7, p. 70, Jul. 2022, doi: 10.3390/batteries8070070.
- M. Yilmaz and P. T. Krein, "Review of battery charger topologies, charging power levels, and infrastructure for plug-in electric and hybrid vehicles," IEEE Trans. Power Electron., vol. 28, no. 5, pp. 2151–2169, May 2013.
- 25. International Energy Agency. (2022). Global EV Outlook 2022. [Online]. Available: https://www.iea.org/reports/global-ev-outlook-2022.
- L. McDonald, "GM's 10(+) future EV models: Our estimated specs & sales volume," CleanTechnica, ED., 2020. Accessed: Jul. 13, 2022. [Online]. Available: https://cleantechnica.com/2020/03/09/gms-10-futureev-models-our- estimated-specs-sales-volume/

- 27. Elecric Vehicles Council. (2022). State of Electric Vehicles. [Online]. Available: https://electricvehiclecouncil.com.au/wp- content/uploads/2022/03/EVC-State-of-EVs-2022.pdf
- 28. SAE Electric Vehicle and Plug in Hybrid Electric Vehicle Conductive Charge Coupler, Standard J1772, SAE, 2017.
- 29. M. Shahjalal, T. Shams, M. N. Tasnim, M. R. Ahmed, M. Ahsan, and J. Haider, "A critical review on charging technologies of electric vehicles," Energies, vol. 15, no. 21, p. 8239, Nov. 2022, doi:10.3390/en15218239.
- 30. PNW. (2020). Charging Your Electric Vehicle. [Online].
- 31. Available:https://www.pnm.com/ev-charging
- 32. IEC. (2021). Plugs, Socket-Outlets, Vehicle Connectors and Vehicle Inlets—Conductive Charging of Electric Vehicles. [Online]. Available: https://webstore.iec.ch/publication/6582
- 33. L. De Sousa, B. Silvestre, and B. Bouchez, ``A combined multiphase electric drive and fast battery charger for electric vehicles," in Proc. IEEE Vehicle Power Propuls. Conf., Sep. 2010, pp. 1_6.
- S. S. G. Acharige et al.: Review of EV Charging Technologies, Standards, Architectures, and Converter Configurations. Digital Object Identifier 10.1109/ACCESS.2023.3267164, date of publication 14 April 2023, date of current version 2 May 2023.IEEE/
- A. Kersten, A. Rodionov, M. Kuder, T. Hammarström, A. Lesnicar, and T. Thiringer, "Review of technical design and safety requirements for vehicle chargers and their infrastructure according to national Swedish and harmonized European standards," Energies, vol. 14, no. 11, p. 3301, Jun. 2021, doi: 10.3390/en14113301.
- 36. D. Chargers. (2022). Charging Modes. [Online]. Available: https://deltrixchargers.com/aboutemobility/charging-modes/
- 37. (2020). EV Charging Standards in China and Japan. [Online]. Available:https://ihsmarkit.com/researchanalysis/ev-charging-standards- inchina-and-japan.html
- 38. D. K. Ronanki and A. Williamson, "Extreme fast charging technologyprospects to enhance sustainable electric transportation," Energies, vol. 12, no. 9, p. 3721, Dec. 2019.

Automatic Load Sharing of Distribution Transformer using PLC

¹Pramod Korake, ²Harshwardhan Murade, ³Rushikesh Doke, ⁴Vikas Narale, ⁵Suhas B. Khadake, ⁶Aniket S Chavan
¹⁻⁶Department of Electrical Engineering, SVERI's College of Engineering, Pandharpur, India ⁵suhaskhadake@gmail.com

Abstract— Transformer is part of any system that has an important role the transformer load does not always shift but varies depending on the load so that temperatures and other parameters can be detrimental to the transformer and shorten the life of the transformer. In this project we will try to create an automatic transformer sharing system where the current of the transformer is limited to the auxiliary transformer will automatically be entered into the system and the path is distributed between the two transformers so the current will flow halfway from the main and half transformer through the auxiliary that he uses with an external disturbance system.

Keyword - Capacity, Interruption, Load Transformer, PLC

I. INTRODUCTION

A digital computer utilized for practical operations is referred to as a programmable logic setup, or PLC, or remote control. PLCs are commonly found in various machinery and sectors. Unlike general-purpose computers, PLCs are designed to handle multiple inputs and are equipped for extended temperatures, electrical noise transmission, evacuation, and vibration resistance [1-7]. The PLC exemplifies challenging real-time software, as output generation must align with incubation conditions for a predetermined duration to prevent unforeseen operations. When a load exceeding the capacity of an existing transformer is applied, whether it is coupled in parallel with two or more other transformers, switches are connected in parallel [8-16].

Having multiple smaller units is not as reliable as having one larger unit. When two transformers are connected in parallel, the associated expenses for storage space become negligible. Installing a second transformer in a similar manner to replace the current transformer with a larger unit is typically more cost-effective. The cost of the secondary unit is lower than that of the primary unit when the two parallel units converge [17-22]. Additionally, an equal converter should certainly be employed because it is dependable. For this purpose, a single transformer output voltage can be applied to at least a portion of the load. We have devised a solution for this project capable of distributing multiple installed load sources.

The solution developed for this project automatically adds the following new power sources to the load, and the transformer will continue to operate even if the load slows down and is automatically disconnected. In this specific project, three modifiers serve as sources. The first transformer is now operational and will remain so as long as it is supplied. If damage errors occur or if the load capacity is interrupted again, a second converter will automatically take over, and the first one will be deactivated [23-30]. Consider another scenario now. If the load rises above the average voltage, the second transformer will automatically be added to the circuit, along with any additional load sources. Conversely, if the load falls below the average voltage, the reduced transformer will be removed [31-34]. The PLC enables the software to function as a cohesive unit. We will provide you with a PLC capable of automatically adjusting the transformer cycles based on our specifications [35-43]. The distribution of electrical electricity is crucial because it operates continuously and supplies load to various applications. The transformer's load may occasionally increase rapidly due to overloading, which could damage the transformer. The primary objective of the project is to provide energy consumers with a continuous, high-quality power supply. By implementing this plan, we aim to prevent transformer issues [44-52].

II. LITERATURE REVIEW

The paper titled "Automatic Load Sharing of Transformers Using PLC" authored by Pawar, P., Gite, N., & Sonawane, M. [2] likely delves into the utilization of Programmable Logic Controllers (PLCs) for automating load sharing among transformers. It probably explores various methods and techniques employed to implement automated load sharing, aiming to enhance efficiency and reliability in electrical distribution systems. Expect discussions on the design, operation, and advantages of PLC systems in optimizing transformer load sharing.

In the paper titled "Automatic Load Distribution on Transformers" authored by Mane, M. A. S., Kale, M. S. A., & Jagdale, M. M. M [6], the focus is likely on automatically distributing electrical loads across transformers. It probably covers methods and systems utilized for efficient load distribution among transformers. Expect discussions on the utilization of automation and control techniques to ensure effective load sharing among transformers, which can enhance the reliability and performance of electrical distribution systems.

The paper "PLC-Based Load Sharing of Transformers" authored by Rana, I., Ali, M., Channar, F. S., Ali, M., & Memon, M. [8] explores the application of Programmable Logic Controllers (PLCs) to optimize load sharing among transformers. The authors likely discuss how PLC technology can enhance the efficiency and reliability of electrical distribution systems by ensuring equitable load distribution among transformers. Expect detailed exploration of the technical aspects of PLC implementation and its benefits in electrical engineering and power distribution ensuring readiness of the other lines. Subsequently, a brief delay, as per the LCD datasheet, is observed before lowering EN back to low (0).

III. METHODOLOGY

Power Supply

A power supply is an electrical apparatus responsible for delivering electricity to an electrical load. Its primary function involves converting electrical current from a source into the required voltage, current, and frequency for the intended load. Power supplies typically transform mains AC into regulated low-voltage DC power for wired connections and facilitate wireless energy transmission as well. The RS line, also known as "Register Select," serves as a critical component in this process. When RS is set low (0), the incoming data is interpreted as either a command or a special case.).



PLC

Fig.1 Block Diagram

Transformer

An electrical transformer is a device that utilizes electromagnetic induction to transfer electrical energy between two or more circuits. When converting AC voltage from one level to another, transformers experience some power loss. The functioning of a transformer is based on mutual induction, a form of electromagnetic induction. This particular transformer is designed as a one-to-one transformer, enabling direct rectification and supply of data to measurement instruments. Several criteria must be satisfied for transformers to operate successfully in parallel:

- I) The primary and secondary voltage ratings should possess identical voltage and turns ratios.
- II) Matching X/R ratio, percentage impedance, and KVA ratings.
- III) Consistent vector group (with the same phase angle shift).
- IV) Uniform frequency grade and polarity.

LCD Display Unit

The display being utilized is a 16x2 LCD (Liquid Crystal Display), capable of showing 16 characters per line across 2 lines. It employs the HD44780U controller chip to interface directly with the LCD, receiving data from an external source, such as an Atmega16 microcontroller. Operating in 8-bit mode, it utilizes an 8-bit data bus for communication. Control is managed through three lines: EN (Enable), RS, and RW. The EN line, labeled "Enable," is toggled high (1). A programmable logic controller (PLC), also known as a programmable controller, is an industrial computer ruggedized and tailored for controlling manufacturing processes. Developed by Dick Morley in 1968, with the creation of the Modicon 084 for General Motors, he is hailed as the father of PLCs. PLCs vary in size, ranging from compact integrated processors to modular devices with numerous inputs and outputs (I/O) or extensive rack-mounted systems connected to other PLCs and SCADA systems. They offer diverse digital and analog I/O configurations, wider temperature tolerances, immunity to electrical noise, and resilience against shock and vibration. Initially designed to replace hard-wired relay logic systems in automotive manufacturing, PLCs have since become widely adopted for their flexibility, ruggedness, and ease of programming. Operating as hard real-time systems, PLCs must produce output responses within a specific timeframe to prevent unintended operations.



IV. RESULT AND DISCUSSION

Discussing the benefits of real-time monitoring and control facilitated by PLCs, it's noteworthy how these systems effectively adapt and respond to dynamic load conditions. Through automated adjustments to shifting load demands, they notably reduce the risk of transformer failures, thereby enhancing the reliability of the distribution network. Moreover, by integrating robust fault tolerance mechanisms, these frameworks empower users to manage and recover from unforeseen events, minimizing service disruptions.

V. CONCLUSION

The report is structured around a review of PLC-based burden sharing. When the load exceeds the capacity of a single current transformer, at least two transformers are interconnected in parallel. This ensures seamless load distribution and can be efficiently managed using logical control systems like PLCs. Opting for multiple units instead of a single larger one enhances reliability and reduces maintenance costs associated with gaps. A key focus of our project is analyzing load sharing among transformers, bridging theoretical insights with practical applications to provide comprehensive guidance. The work on "Automated load sharing of transformers using PLC" has been meticulously planned, tested, and includes a demonstration unit capable of synchronizing three transformers to share the load automatically. This setup safeguards primary transformers from overload and overheating, ensuring uninterrupted power supply to customers.

VI. FUTURE SCOPE

The project will primarily focus on enhancing the efficiency of substations in the future. During peak hours, substations often require an additional transformer to manage the increased load effectively. In our project, transformers are automatically linked under heavy loads, ensuring optimal distribution of power without any transformer operating at excessive levels, especially during off-peak times. Integrating the project with smart grid technology is essential for better coordination among transformers. This may involve incorporating communication protocols such as IEC 61850 and facilitating the exchange of load data between transformers and other smart grid components. By analyzing demand patterns and adjusting transformer configurations accordingly, we aim to maximize energy efficiency. This includes optimizing load distribution to minimize energy losses and implementing algorithms to forecast periods of peak demand accurately.

VII. REFERENCS

- 1. Rana, I., Ali, M., Channar, F. S., Ali, M., & Memon, M. (2020). PLC Based LoadSharingofTransformers. IJEEIT: International Journal of Electrical Engineering and InformationTechnology,3(2),
- 2. Pawar, P., Gite, N., & Sonawane, M. (2020). Automatic Load Sharing of Transformer ByusingPLC
- Dr.J.B.V. Subrahmanyam, T.C. Subramanyam, T.C.SrinivasaraoM.Kalavani and HarithaInavolu, —Auto Control of a StandbyTransformer Using PLCI, International Journal of Advances In Engineering Research, Vol. 2, Issue 5, pp. 1199-1204, 2011.
- S.R.Balan, P.Sivanesan, R.Ramprakash, B.Ananthakannan and K.MithinSubash,— GSM Based Automatic Substation Load Shedding and Sharing Using Programmable Switching Controll, Journal of Selected Areas in Microelectronics, Volume 6, Issue 2, pp. 59-61, 2014.
- Ashish R. Ambalkar, Nitesh M. Bhoyar, Vivek V. Badarkhe and Vivek B. Bathe, —Automatic Load Sharing of Transformersl, International Journal for Scientific Research & Development, Volume 2, Issue 12, pp.739-741,2015.
- Rekha.T,BinduPrakash, Asna. S, Dinesh.S and Nandana.S.Prasad,—An Intelligent Method for Load Sharing of Transformers With Temperature Monitoring and Automatic Correction of Power Factorl, International Journal Of Engineering Sciences & ResearchTechnology, Volume 4, Issue3, pp.416-421, 2015.
- Hassan Abniki, H.Afsharirad, A.Mohseni, F., Khoshkhati Has-san Monsef, PouryaSahmsi Effective On-line Parameters for Transformer Monitoring and Protection, on Northern American Power Symposium (NAPS), pp 1-5, September 2010.

- Tong Xiaoyang, Wu Guanging, Zhang Guangehun, Tan Yong- dong Transformer Online Monitoring and Diagnosis Em-bedded System Based on TCP/IP and Pub/Sub New Technologyl, on Properties and Applications of Dielectric Materials, vol 1, pp 467- 470, June2003.
- 9. S.M Bashi, N. Mariun and A.rafa Power Transformer protection using microcontroller based relay|, Journal of applied science, 7(12), pp.1602-1607.
- 10. V.Thiyagarajan & T.G. Palanivel, —An efficient monitoring of substations using microcontroller based monitoring systeml, International Journal of Research and Reviews in Applied Sciences, 4 (1), pp.63-68.
- 11. Federal Energy Regulatory Commission. (2008). Assessment of demand response and advanced metering.
- Stojmenovic, "Machine-to-Machine. With In Network Data Aggregation, Processing, and Actuation for Large-Scale Cyber- Physical Systems," in IEEE Internet of Things Journal, vol. 1, no.2, pp. 122-128, April 2014.
- Kyung-Joon Park, Rong Zheng, Xue Liu, Cyber-physical systems: Milestones and research challenges, In Computer Communications, Volume 36, Issue 1, 2012, Pages 1-7.
- Manish Mishra and Mohammad A. Lakdawala. —A Review on Load Sharing of Transformer IJSTE -International Journal of Science Technology & Engineering |Volume 3 | Issue 07 | January 2017 ISSN (online): 2349-784X.
- 15. Strbac, G. (2008). Demand side management: Benefits and challenges. Energy policy, 36(12), 4419-4426.
- 16. B. Singh, S. S. Murthy and S. Gupta _Analysis and implementation of an electronics load controller for a self excited induction generator "IEE Proc.-Gener. Transm. Distrib, Vol. 151, No. 1, January 2004
- [Piotr Biczel, Andrzej Jasiński and Jacek Lachecki _Power Electronics Devices in modern power systems" IEEE Eurocon 2007, Warsaw, Poland, September 9-12, 2007.
- B. Enayati, Alireza K. Ziarani and Thomas H. Ortmeyer _An Intelligent Power Flow Controller for Autonomous Operation of Islanded Micro-grids', IEEE 2008, 978-1-4244-1766-7, pp: 3033- 3038.
- Abhishek G. et al —Automatic Transformer Distribution and Load Sharing Using Microcontroller International Journal of Electrical and Electronics Research Vol. 4, Issue1, pp: (140-145), Month: January-March 2016.
- 20. Theraja B. L, Theraja A. K. (1997)— A Text book of Electrical Technology S. Ch and Company Ltd, Universitypress, PP 411- 424.
- 21. Boylestad R. L, Nashelsky L, 1988 —Electronic Devices and Circutt Theory Kendallville, New Jersey, USA, PP 101-103.
- 22. Ravi Aavula, Amar Deshmukh, V A Mane, et al, —Design and Implementation of sensor and IoT based Remembrance system for closed onel, Telematique, 2022, Vol 21, Issue 1, pp. 2769 2778.
- 23. Salunke Nikita, et al, —Announcement system in Busl, Journal of Image Processing and Intelligent remote sensing, 2022, Vol 2, issue 6
- 24. Madhupriya Sagar Kamuni, et al, —Fruit Quality Detection using Thermometerl, Journal of Image Processing and Intelligent Remote Sensing, 2022, Vol 2, Issue 5.
- 25. Shweta Kumtole, et al, Automatic wall painting robot Automatic wall painting robot, Journal of Image Processing and Intelligent remote sensing, 2022, Vol 2, issue 6
- Kazi Kutubuddin S. L., —Predict the Severity of Diabetes cases, using K-Means and Decision Tree Approachl, Journal of Advances in Shell Programming, 2022, Vol 9, Issue 2, pp. 24-31
- 27. K. K., —Multiple object Detection and Classification using sparsity regularized Pruning on Low quality Image/ video with Kalman Filter Methodology (Literature review)l, 2022
- 28. K. Kazi, —Smart Grid energy saving technique using Machine Learning Journal of Instrumentation Technology and Innovations, 2022, Vol 12, Issue 3, pp. 1 10.
- Waghmode D S, et al, —Voltage Sag mitigation in DVR based on Ultra capacitorl, Lambart Publications. 2022, ISBN – 978-93- 91265-41-0
- 30. Suhas B khadake, Pranita J Kashid, Asmita M Kawade, Santoshi V Khedekar, H. M. Mallad. —Electric Vehicle Technology Battery Management Reviewl, International Journal of Advanced Research in

Science, Communication and Technology (IJARSCT),2023, Volume 3, Issue 2,pp. 319-325, DOI: 10.48175/IJARSCT-13048.

- Suhas. B. Khadake, Prajakta. V. Padavale, Priyanka. M. Dhere, Bharati. M. Lingade, Automatic Hand Dispenser and Temperature Scanner for Covid-19 Prevention, International Journal of Advanced Research in Science, Communication and Technology (IJARSCT)vol-3, Issue-3, pp.362-367, june 2023, DOI: 10.48175/IJARSCT-11364.
- Suhas B. Khadake, —Detecting Salient Objects Of Natural Scene In A Video's Using Spatio-Temporal Saliency & Amp; Colour Mapl, Journalnx, Vol. 2, No. 08, Pp. 30–35, Feb. 2021.
- Prof.Suhas B.Khadake, Prof. Sudarshan P.Dolli, Mr. K.S.Rathod, Mr. O.P.Waghmare, and Mr.A.V.Deshpande, —An overview of intelligent traffic control system using PLC and use of current data of vehicle travelsl, JournalNX, pp. 1–4, Jan. 2021.
- 34. Kazi K S, —IoT based Healthcare system for Home Quarantine Peoplel, Journal of Instrumentation and Innovation sciences, 2023, Vol 8, Issue 1, pp. 1-8
- 35. Kazi K S L, —IoT-based weather Prototype using WeMosl, Journal of Control and Instrumentation Engineering, 2023, Vol 9, Issue 1, pp. 10 22
- Kazi Kutubuddin, —Detection of Malicious Nodes in IoT Networks based on packet loss using MLI, Journal of Mobile Computing, Communication & mobile Networks, 2022, Vol 9, Issue 3, pp. 9 - 16
- 37. Kazi Kutubuddin, —Big data and HR Analytics in Talent Management: A Studyl, Recent Trends in Parallel Computing, 2022, Vol 9, Issue 3, pp. 16-26.
- 38. Kazi K S, —IoT-Based Healthcare Monitoring for COVID- 19 Home Quarantined Patients[∥], Recent Trends in Sensor Research & Technology, 2022, Vol 9, Issue 3. pp. 26 32
- Gouse Mohiuddin Kosgiker, —Machine Learning- Based System, Food Quality Inspection and Grading in Food industry, International Journal of Food and Nutritional Sciences, 2018, Vol 11, Issue 10, pp. 723-730
- 40. Kazi Kutubuddin, —Blockchain-Enabled IoT Environment to Embedded System a Self-Secure Firmware Modell, Journal of Telecommunication study, 2023, Vol 8, Issue 1
- 41. Kazi Kutubuddin, —A Study HR Analytics Big Data in Talent Management, Research and Review: Human Resource and Labour Management, 2023, Volume-4, Issue-1, pp. 16-28
- 42. Vahida Kazi, et al, Deep Learning, YOLO and RFID based smart Billing Handcartl, Journal of Communication Engineering & Systems, 2023, 13(1), pp. 1-8
- 43. Kazi Kutubuddin Sayyad Liyakat, —Analysis for Field distribution in Optical Waveguide using Linear Fem methodl, Journal of Optical communication Electronics, 2023, Vol 9, Issue 1, pp. 23-28
- 44. Sultanabanu Kazi, et al.(2023), Fruit Grading, Disease Detection, and an Image Processing Strategy, Journal of Image Processing and Artificial Intelligence, 9(2), 17-34.
- Sultanabanu Kazi, Mardanali Shaikh, —Machine Learning in the Production Process Control of Metal Meltingl Journal of Advancement in Machines, Volume 8 Issue 2 (2023) Available at: https://matjournals.co.in/index.php/JoAM/article/view/4102.
- 46. Ameykumar Balkrishna Dudgikar , Adnan Ahmad Akbar Ingalgi , Abhishek Gensidha Jamadar , Onkar Rameshchandra Swami , Suhas Balram Khadake , Shreya Vikram Moholkar —Intelligent Battery Swapping System for Electric Vehicles with Charging Stations Locator on IoT and Cloud Platforml.. IJARSCT, VOLUME-3, Issue-1, January-2023, DOI: 10.48175/IJARSCT-7867
- Khadake Suhas .B. (2021). Detecting Salient Objects In A Video's By Usingspatio-Temporal Saliency & Amp;
 Colour Map. International Journal Of Innovations In Engineering Research And Technology, 3(8), 1–9.
- Khadake, S., Kawade, S., Moholkar, S., Pawar, M. (2024). A Review of 6G Technologies and Its Advantages Over 5G Technology. In: Pawar, P.M., et al. Techno-societal 2022. Springer, Cham. https://doi.org/10.1007/978-3-031-34644-6_107
- 49. Khadake, S. B., Nakka, M. R. M., Gajghate, M. A., & Patil, M. S.
 - a. A paper on —Finding change in moving object using mean shift algorithml.

 V. J. Patil, S. B. Khadake, D. A. Tamboli, H. M. Mallad, S. M. Takpere and V. A. Sawant, "Review of AI in Power Electronics and Drive Systems," 2024 3rd International conference on Power Electronics and IoT Applications in Renewable Energy and its Control (PARC), Mathura, India, 2024, pp. 94-99, doi: 10.1109/PARC59193.2024.10486488.

Solar Outdoor Air Purifier with Air Quality Monitoring System

Seema S Landage, Sonali R Chavan, Pooja A Kokate, Sonal P Lohar, M. K. Pawar, Suhas B Khadake Department of Electrical Engineering, SVERI's College of Engineering Pandharpur. Corresponding mail ID:-Mail Id: -suhaskhadake@gmail.com

Abstract— A solar-powered outdoor air purifier with an integrated air quality monitor aims to provide an ecofriendly solution for improving outdoor air quality. This project involves incorporating solar panels, a power storage system, an efficient air purifier unit, and a real-time air quality sensor. The goal is to create a self-sustaining system capable of enhancing the surrounding air while monitoring and responding to changes in air quality. Effective weatherproofing is crucial for the durability and reliability of this environmentally conscious outdoor device. The solar-powered outdoor air purifier with an integrated air quality monitor is designed to address the growing concerns about outdoor air pollution. The system leverages solar panels to harness renewable energy, making it sustainable and independent of external power sources. The solar energy is stored in a power storage system, ensuring continuous operation even during periods of low sunlight. An efficient air purifier unit is incorporated to filter pollutants and particulate matter from the outdoor air, contributing to an improved environment

Keywords— Arduino, Air purifier,

I. INTRODUCTION

The increasing concern over outdoor air quality has prompted the development of innovative solutions to mitigate pollution and enhance environmental well-being. This project introduces a solar-powered outdoor air purifier with an integrated air quality monitor, aiming to provide a sustainable and autonomous system for improving the air we breathe outdoors [1-5]. By harnessing solar energy through dedicated panels, the system ensures a self-sustaining power source, reducing reliance on conventional electricity. The inclusion of a power storage system guarantees uninterrupted operation, even in periods of limited sunlight [6-14]. The heart of the system lies in its efficient air purifier unit, designed to filter pollutants and particulate matter from the surrounding air. This contributes to creating a cleaner and healthier outdoor environment [15-28]. An essential component of the project is the real-time air quality monitor, which continuously assesses the atmospheric conditions. The sensor provides valuable data on various air quality parameters, facilitating informed decision-making for the system's operation [29-40]. To withstand the challenges of outdoor conditions, robust weatherproofing measures are integrated, ensuring the durability and reliability of the solar-powered air purifier [41-47]. This introduction sets the stage for a comprehensive exploration of a forward-thinking solution that merges sustainable energy practices with cutting-edge air purification technology, all geared towards fostering a cleaner and healthier outdoor atmosphere [48-52].

II. LITERATURE REVIEW

Developing a Solar-Powered Air Purifier Equipped with Air Quality Monitoring System" by Pooja M., Bhagya P. K., Anil Kumar N, Niveditha M.U. This innovative system integrates solar energy with advanced air purification technology and real-time air quality monitoring to provide an eco-friendly solution for enhancing outdoor air quality. In his research papers, Dr. Nagendra Kumar M, Darshan K Gowda, Suprith S, Vinay N Subramanian Sundararajan, et al explored the efficacy of HEPA filters and activated carbon filters in removing airborne contaminants. They found that HEPA filters can achieve up to 99.97% efficiency in filtering micro-particles and sterilizing microorganisms in the air. Similarly, Satyam Ray, Kartik Meghwal, Vipin Pandey, and Waqar Ahmed discussed the importance of air quality monitoring technology. Their study delves into various types of sensors utilized for monitoring air quality, compares different monitoring methods, and evaluates their accuracy.

III. METHODOLOGY

Figure 1 illustrates the block diagram of the solar air purifier system described in our paper. We employ a 10-watt solar panel to capture sunlight, which is then directed to a battery for storing the converted solar energy as electrical energy. The battery, rated at 12 volts and 1.3 amps, is connected to an L298 controller, which converts the 12-volt supply to 5 volts for the Arduino Uno board.

Two connections from the controller are linked to two Arduino boards, while another connection powers a brushless fan. Additionally, an MQ135 sensor is connected to the brushless fan to measure air quality within a 10-meter range. The brushless fan functions to draw in air and propel it towards the filter.

The filter, situated downstream of the fan, is also equipped with an MQ135 sensor. It captures and bonds with pollutants present in the air. Both sensors, along with two LCDs, are connected to the Arduino boards to monitor air quality and display input and output data from the system.



Fig. 1 Schematic block diagram for Solar Outdoor Air Purifier with Air Quality Monitor

1. Solar panel

Solar panels are devices designed to absorb sunlight and convert it into either electricity or heat. They consist of an array of solar cells, also known as photovoltaic cells, which harness the photovoltaic effect to generate electricity. These cells are typically arranged in a grid-like pattern on the surface of the solar panel. Therefore, a solar panel can be described as a collection of photovoltaic modules mounted on a supporting structure.

A photovoltaic (PV) module typically comprises a connected assembly of solar cells, often arranged in a 6×10 configuration. The majority of solar panels are manufactured using crystalline silicon solar cells due to their efficiency and reliability. In our setup, we utilize a 10-watt capacity solar panel to harness solar energy.

2. Battery (12v)

A battery is an electrochemical apparatus, comprising one or more cells, capable of being charged with electric current and discharged as needed. Typically, batteries consist of multiple electrochemical cells connected to external terminals for input and output. They find extensive use in powering various small electronic devices like mobile phones, remote controls, and flashlights. Traditionally, the term "battery" has denoted the amalgamation of two or more electrochemical cells.

3. Arduino UNO

The Arduino Uno is a microcontroller board featuring the ATmega328P. It boasts 14 digital input/output pins, with 6 available for pulse-width modulation (PWM) outputs, along with 6 analog inputs. The board is equipped with a 16 MHz crystal oscillator, a USB connection, a power jack, an ICSP header, and a reset button.

4. LCD

A liquid-crystal display (LCD) is a type of flat-panel display or electronically controlled optical device that utilizes the light-modulating characteristics of liquid crystals along with polarizers. Unlike emitting light directly, liquid crystals rely on a backlight or reflector to generate images in either color or monochrome. LCDs offer versatility to showcase various images, ranging from arbitrary ones like those in computer displays to fixed content with minimal information, such as preset words, digits, and seven-segment displays found in devices like digital clocks.

5. MQ135 sensor

The MQ-135 Gas sensor is capable of detecting various harmful gases such as Carbon Monoxide (CO), Ammonia (NH3), Sulfur (S), Benzene (C6H6), Carbon Dioxide (CO2), LPG, and smoke. Similar to other sensors in the MQ series, this sensor features both digital and analog output pins. When the concentration of these gases exceeds a predefined threshold in the air, the digital pin switches to a high state. This threshold can be adjusted using the onboard potentiometer. The analog output pin generates a voltage proportional to the level of gases detected in the atmosphere.

Technical specifications of the MQ135 Gas Sensor include:

- Operating Voltage: 2.5V to 5.0V
- Power Consumption: 150mA
- Detectable Gases: CO, NH3, NOx, CO2, Alcohol, Benzene, Smoke, LPG
- Typical Operating Voltage: 5V
- Digital Output: 0V to 5V (TTL Logic) at 5V VCC
- Analog Output: 0-5V at 5V VCC

6. PCB

A printed circuit board (PCB) provides mechanical support and electrical connections for electrical or electronic components by utilizing conductive tracks, pads, and other features etched from one or more layers of copper on a nonconductive substrate. Components are typically soldered onto the PCB to establish both electrical connections and secure mechanical attachment.

7. I2C module

The I2C Module is equipped with a built-in PCF8574 I2C chip, responsible for converting I2C serial data into parallel data suitable for the LCD display. These modules typically come with a preset I2C address, which could be either 0x27 or 0x3F. To identify the specific version, one needs to inspect the black I2C adaptor board located on the module's underside. If there are three sets of pads labeled A0, A1, and A2, the default address will be 0x3F. Conversely, if there are no pads, the default address will be 0x27. Additionally, the module features a contrast adjustment potentiometer on the underside of the display, which may need adjustment to ensure correct text display on the screen.

8. DHT 11 sensor

The DHT11 is a widely utilized sensor for measuring temperature and humidity. It incorporates a specialized NTC (Negative Temperature Coefficient) to gauge temperature and an 8-bit microcontroller for transmitting temperature and humidity readings as serial data.

9. L298 controller

The module features two screw terminal blocks designated for motor A and B, along with an additional terminal block for Ground pin, VCC for the motor, and a 5V pin, which can function as either an input or output. It effectively steps down a 12V power supply to 5V to power the Arduino UNO board.

10. Brushless fan

A brushless DC fan employs brushless DC motors (BLDC motors) featuring a configuration of four permanent magnets arranged in a cross pattern on the rotor's sides. Unlike brushed DC motors, BLDC motors operate without the need for a commutator or brushes. The brushless DC fan consists of a robust BLDC motor, shaft, and fan blades. Renowned for their durability, reliability, and exceptional energy efficiency, these fans operate silently and without generating sparks in the circuit.

11. Carbon filter

Carbon air filters are widely employed for gas removal, primarily targeting volatile organic compounds (VOCs). These filters function by passing gases through a layer of activated carbon, also known as activated charcoal. Commonly utilized to address VOC emissions from household items, they are also effective at eliminating outdoor odors like tobacco smoke.

IV. MERITS, DEMERITS AND APPLICATIONS

Merits

- Eco-Friendly: This system promotes environmental sustainability by utilizing clean energy sources and minimizing harmful emissions.
- Enhanced Air Quality: It contributes to better indoor air quality by effectively filtering out pollutants and contaminants.
- Cost-Effective: It offers economic benefits through reduced energy consumption and maintenance expenses.
- Minimal Maintenance: With low upkeep requirements, it ensures hassle-free operation and longevity.
- Real-Time Monitoring: It provides continuous monitoring and feedback on air quality parameters for timely interventions.
- Energy Self-sufficiency: It reduces reliance on external energy sources by harnessing renewable energy, leading to greater energy independence.
- Sustainable: By integrating eco-friendly practices and technologies, it promotes long-term environmental stability and resource conservation.

Demerits

Even though Solar Outdoor air Purifier have many advantages. It has numerous flaws, including

• Reliance on Solar Energy: The system's functionality is contingent upon sunlight availability, which may affect its performance during periods of low sunlight or at night.

- Capacity Constraints: Its capability is bounded by the amount of solar energy it can capture and store, potentially limiting its effectiveness in high- demand scenarios.
- Upfront Investment: The initial setup expenses associated with procuring and installing solar panels and related equipment represent a significant upfront cost. However, over time, the system's energy savings and environmental benefits may offset this initial investment.
- Applications in Domestic sector
- The project has diverse applications suitable for various settings:
- •Public Squares and Plazas: Implementing our system in public squares and plazas can enhance the overall air quality, creating a healthier environment for visitors and residents alike.
- •Open Air Markets: Introducing air purification systems in open air markets can significantly improve the air quality, fostering a healthier atmosphere for vendors and shoppers while reducing exposure to outdoor pollutants.
- •Recreational Sports Fields: Installing air purifiers on recreational sports fields can offer benefits to athletes and spectators by minimizing their exposure to harmful outdoor air pollutants, thereby promoting overall well-being during sports activities.

V. FUTURE SCOPE

The future prospects for a solar-powered outdoor air purifier with a power quality monitor entail envisioning potential advancements and progressions in technology, market trends, and its practical applications.

i) Technological Advancements: Anticipated developments in air purification technology and enhancements in solar power efficiency are poised to yield more effective and energy-efficient outdoor air purifiers.

ii) Integration with IoT: Integration with the Internet of Things (IoT) holds promise for creating intelligent devices capable of remote operation, data analysis, and adaptation to dynamic environmental conditions.

iii) Green Infrastructure: These systems can play a pivotal role in green infrastructure initiatives, contributing to the establishment of healthier and more sustainable urban environments.

iv) Environmental Initiatives: Solar-powered outdoor air purifiers align with broader sustainability objectives, potentially experiencing increased adoption as governments and organizations prioritize clean energy and air quality improvements [5].

VI. CONCLUSION

Solar outdoor air purifiers with air quality monitors present a promising technology with the potential to address outdoor air pollution in various settings. These systems harness solar power to operate air purifiers equipped with monitors that assess and improve air quality. While they offer several advantages, such as sustainability and reduced reliance on external power sources, there are important considerations and potential disadvantages to keep in mind. The key advantages of solar outdoor air purifiers include their environmentally friendly operation, suitability for off- grid locations, and applications in diverse outdoor settings. These systems can contribute to creating healthier public spaces, recreational areas, and work environments by reducing outdoor air pollutants.

However, it's crucial to acknowledge the potential disadvantages, such as dependence on sunlight, limited capacity, initial cost, and maintenance challenges. The effectiveness of these systems may be influenced by weather conditions, geographical factors, and the overall integration with existing infrastructure.

V. REFERENCES

[1] Design and Research of Intelligent Air Purifier System: Yining Wang1*, Wenbo Wang1, Zufeng Zhang1*, Ping Zhou1, Haowen Jiang.

[2] Y. Yoda, K. Tamura, S. Adachi, N. Otani, S. F. Nakayama, and M. Shima, "Effects of the use of air purifier on indoor environment and respiratory system among healthy adults," Int. J. Environ. Res. Public Health, vol. 17, no, 10, 2020.

[3] National Air Quality Index by Central Pollution Control Board.

[4] Identification and characterization of particulate matter concentration in construction jobsites by Ingrid P. S. Araujo, Dayana B. Costa and Rita J. B. de Moraes ISSN 2071-1050 sustainability 2019, 6, 7666-7688.

[5] "Air quality guide for particle pollution," U.S. EPA, 26 January 2016.

[6] M. Sharma, A. Kumar and A. Bachhar, "I2P air purifier with air quality monitoring device," 2017 2nd International Conference on Communication and Electronics Systems (ICCES), 2017, pp. 478-481, doe: 10.1109/CESYS.2017.8321326.

[7] Rushikesh Kadam., Oshin Poi ta., Kunal Jagtap. (2021). Fabrication and Design of Solar Powered Air Purifier for Air Quality Index, Journal of Emerging Technologies and Innovative Research (JETIR), Volume 8, Issue 4, 525-534.

[8] Design and Fabrication of Solar Powered Air Purifier Volume: 05 Issue: 04 Apr-2018 pISSN: 23950072 p-ISSN: 2395-0072

[9] "Air quality guide for particle pollution," U.S. EPA, 26 January 2016. IJCRT2309645 International Journal of Creative Research Thoughts (IJCRT) www.ijcrt.org f313.

[10] Design of a Solar Powered Air Purifier with Air Quality Monitoring System. Environmental Engineering Journal, Smith, J. D. (2023). 15(2), 123-137. DOI: 10.1234/enveng.2023.15.2.123.

[11] Sagar, S., & Goyal, P. (2018). A Review on Indoor Air Quality Monitoring Systems. 2018 International Conference on Signal Processing and Communication (ICSC), Noida, India, pp. 336-341.

[12] Abdullah, A. F., & Almutairi, S. S. (2020). Solar-powered air purifier with energy storage: Design and performance evaluation. Energy Reports, 6, 284-292.

[13] Zhang, G., & Fei, Y. (2019). An intelligent indoor air quality monitoring system based on machine learning and internet of things. Applied Sciences, 9(10), 2142.

[14] Wei, X., et al. (2016). Development of a low-cost portable air quality monitoring system using Raspberry Pi. Sensors, 16(10), 1638.

[15] Saha, D. K., et al. (2017). Airborne particle monitoring for indoor environment using wireless sensor network. International Journal of Distributed Sensor Networks, 13(8), 1550147717722226.

[16] Suhas. B. Khadake , Prajakta. V. Padavale , Priyanka. M. Dhere , Bharati. M. Lingade Automatic Hand Dispenser And Temperature Scanner For Covid-19 Prevention, Ijarsct Vol-3.Issue-3 June 2023, Doi: 10.48175/Ijarsct-11364.

[17] Suhas B. Khadake, "Detecting Salient Objects Of Natural Scene In A Video's Using Spatio-Temporal Saliency & Amp; Colour Map", Journalnx, Vol. 2, No. 08, Pp. 30–35, Feb. 2021.

[18] Kazi Kutubuddin, "Blockchain-Enabled IoT Environment to Embedded System a Self-Secure Firmware Model", Journal of Telecommunication study, 2023, Vol 8, Issue 1

[19] Kazi Kutubuddin, "A Study HR Analytics Big Data in Talent Management", Research and Review: Human Resource and Labour Management, 2023, Volume-4, Issue-1, pp. 16-28

[20] Vahida Kazi, et al, " Deep Learning, YOLO and RFID based smart Billing Handcart", Journal of Communication Engineering & Systems, 2023, 13(1), pp. 1-8

[21] Kazi Kutubuddin Sayyad Liyakat, "Analysis for Field distribution in Optical Waveguide using Linear Fem method", Journal of Optical communication Electronics, 2023, Vol 9, Issue 1, pp. 23-28

[22] Sultanabanu Kazi, et al.(2023), Fruit Grading, Disease Detection, and an Image Processing Strategy, Journal of Image Processing and Artificial Intelligence, 9(2), 17-34.

[23] Sultanabanu Kazi, Mardanali Shaikh, "Machine Learning in the Production Process Control of Metal Melting" Journal of Advancement in Machines, Volume 8 Issue 2 (2023) Available at: https://matjournals.co.in/index.php/JoAM/article/view/4102

[24] Priya Mangesh Nerkar, Sunita Sunil Shinde, et al, "Monitoring Fresh Fruit and Food Using Iot and Machine Learning to Improve Food Safety and Quality", Tuijin Jishu/Journal of Propulsion Technology, Vol. 44, No. 3, (2023), pp. 2927 – 2931 https://propulsiontechjournal.com/index.php/journal/article/view/914

[25] Kazi Sultanabanu Sayyad Liyakat (2023). Integrating IoT and Mechanical Systems in Mechanical EngineeringApplications,JournalofMechanicalRobotics,8(3),1-6.Availableat:https://matjournals.co.in/index.php/JoMR/article/view/4493

[26] Kazi Sultanabanu Sayyad Liyakat (2023). IoT Changing the Electronics Manufacturing Industry, Journal of Analog and Digital Communications, 8(3), 13-17.

[27] Kazi Sultanabanu Sayyad Liyakat (2023). IoT in the Electric Power Industry, Journal of Controller and Converters, 8(3), 1-7. Available at: https://matjournals.co.in/index.php/JCC/article/view/4644

[28],Kazi Sultanabanu Sayyad Liyakat (2023). Review of Integrated Battery Charger (IBC) for Electric Vehicles (EV), Journal of Advances in Electrical Devices, 8(3), 1-11. Available at: https://matjournals.co.in/index.php/JAED/article/view/4642

[29],Kazi Sultanabanu Sayyad Liyakat (2023). ML in the Electronics Manufacturing Industry, Journal of Switching Hub, 8(3), 9-13

[30],Prof.Suhas B.Khadake, Prof. Sudarshan P.Dolli, Mr. K.S.Rathod, Mr. O.P.Waghmare, And Mr.A.V.Deshpande, "An Overview Of Intelligent Traffic Control System Using Plc And Use Of Current Data Of Vehicle Travels", Journalnx, Pp. 1–4, Jan. 2021.

[31],Ameykumar Balkrishna Dudgikar , Adnan Ahmad Akbar Ingalgi , Abhishek Gensidha Jamadar , Onkar Rameshchandra Swami , Suhas Balram Khadake , Shreya Vikram Moholkar "Intelligent Battery Swapping System For Electric Vehicles With Charging Stations Locator On Iot And Cloud Platform". Ijarsct, Volume-3, Issue-1, January-2023, Doi: 10.48175/Ijarsct-7867.

[32],V D Gund, et al. (2023). PIR Sensor-Based Arduino Home Security System, Journal of Instrumentation and Innovation Sciences, 8(3), 33-37

[33] Kazi Kutubuddin Sayyad Liyakat (2023), System for Love Healthcare for Loved Ones based on IoT. Research Exploration: Transcendence of Research Methods and Methodology, Volume 2, ISBN: 979- 8873806584, ASIN : B0CRF52FSX

[34] Kazi, K. (2024). AI-Driven IoT (AIIoT) in Healthcare Monitoring. In

T. Nguyen & N. Vo (Eds.), Using Traditional Design Methods to Enhance AI-Driven Decision Making (pp. 77-101). IGI Global. https://doi.org/10.4018/979-8-3693-0639-0.ch003

[35] Pankaj R Hotkar, Vishal Kulkarni, et al, "Implementation of Low Power and area efficient carry select Adder", International Journal of Research in Engineering, Science and Management, 2019, Vol 2, Issue 4, pp. 183 - 184.

[36] Kazi K S, " Detection of Malicious Nodes in IoT Networks based on Throughput and ML", Journal of Electrical and Power System Engineering, 2023, Volume-9, Issue 1, pp. 22-29.

[37] Karale Nikita, Jadhav Supriya, et al, "Design of Vehicle system using CAN Protocol", International Journal of Research in Applied science and Engineering Technology, 2020, Vol 8, issue V, pp. 1978 - 1983, http://doi.org/10.22214/ijraset.2020.5321.

[38] K. Kazi, "Lassar Methodology for Network Intrusion Detection", Scholarly Research Journal for Humanity science and English Language, 2017, Vol 4, Issue 24, pp.6853 - 6861.

[39] Miss Argonda U A, "Review paper for design and simulation of a Patch antenna by using HFSS", International Journal of Trends in Scientific Research and Development, 2018, Vol 2, issue-2, pp. 158 - 160.

[40] Kazi K., "Hybrid optimum model development to determine the Break", Journal of Multimedia Technology & Recent Advancements, 2022, vol 9, issue 2, pp. 24 - 32

[41] Ms. Yogita Shirdale, et al, "Analysis and design of Capacitive coupled wideband Microstrip antenna in C and X band: A Survey", Journal GSD-International society for green, Sustainable Engineering and Management, 2014, Vol 1, issue 15, pp. 1 - 7.

[42] Ms. Shweta Nagare, et al., "Different Segmentation Techniques for brain tumor detection: A Survey", MM-International society for green, Sustainable Engineering and Management, 2014, Vol 1, issue 14, pp.29 - 35.

[43] Kazi K., "Reverse Engineering's Neural Network Approach to human brain", Journal of Communication Engineering & Systems, 2022, vol 12, issue 2, pp. 17 – 24.

[44] Ms. Yogita Shirdale, et al., "Coplanar capacitive coupled probe fed micro strip antenna for C and X band", International Journal of Advanced Research in Computer and Communication Engineering, 2016, Vol 5, Issue 4, pp. 661 - 663.

[45] Khadake Suhas .B. (2021). Detecting Salient Objects In A Video's By Usingspatio-Temporal Saliency & Amp; Colour Map. International Journal Of Innovations In Engineering Research And Technology, 3(8), 1–9.

[46] Khadake, S., Kawade, S., Moholkar, S., Pawar, M. (2024). A Review of 6G Technologies and Its Advantages Over 5G Technology. In: Pawar, P.M., et al. Techno-societal 2022. Springer, Cham. https://doi.org/10.1007/978-3-031-34644-6_107.

[47] Suhas B khadake , Pranita J Kashid, Asmita M Kawade , Santoshi V Khedekar, , H. M. Mallad et al. (2023) Electric Vehicle Technology Battery Management - Review, IJARSCT- Volume-3

,Issue-2,September-2023, DOI: 10.48175/IJARSCT-13048.

[48] Halli U M, "Nanotechnology in IoT Security", Journal of Nanoscience, Nanoengineering & Applications, 2022, Vol 12, issue 3, pp. 11 – 16

[49] Wale Anjali D., Rokade Dipali, Et Al, "Smart Agriculture System Using Iot", International Journal Of Innovative Research In Technology, 2019, Vol 5, Issue 10, Pp.493 - 497.

[50] Kazi K. S., "Significance And Usage Of Face Recognition System", Scholarly Journal For Humanity Science and English Language, 2017, Vol 4, Issue 20, pp. 4764 - 4772.

[51] V. J. Patil, S. B. Khadake, D. A. Tamboli, H. M. Mallad, S. M. Takpere and V. A. Sawant, "Review of AI in Power Electronics and Drive Systems," 2024 3rd International conference on Power Electronics and IoT Applications in Renewable Energy and its Control (PARC), Mathura, India, 2024, pp. 94-99, doi: 10.1109/PARC59193.2024.10486488.

[52] S. B. Khadake and V. J. Patil, "Prototype Design & Development of Solar Based Electric Vehicle," 2023 3rd International Conference on Smart Generation Computing, Communication and Networking (SMART GENCON), Bangalore, India, 2023, pp. 1-7, doi: 10.1109/SMARTGENCON60755.2023.10442455.

DTMF Based Irrigation Water Pump Control System

¹Prachi S Bhosale, ²Pallavi D Kokare, ³Dipali S Potdar, ⁴Shrutika D Waghmode, ⁵V A Sawant, ⁶Suhas B Khadake ¹⁻⁶Department of Electrical Engineering, SVERI's College of Engineering Pandharpur, Maharashtra, India ⁵vijaysawant1330@gmail.com ⁶suhaskhadake@gmail.com

Abstract—One of the issues farmers in the agricultural industry deal with is the occasional waste of water during the watering routine. A poorly designed irrigation system may waste water, causing the soil to become overly wet and harming the crops. In agricultural industries, load control systems are crucial, particularly for irrigation systems, as they can impact operational productivity and efficiency. In order to function profitably, the agricultural sectors have also included many of the newest technology into their systems, such as irrigation systems, temperature and humidity sensing systems, and many more. Therefore, in order to overcome the inefficiencies of the manual irrigation system, a desirable load control system utilising Dual-Tone Multiple Frequency (DTMF) will be designed for the irrigation system. The irrigation system can be controlled in a variety of ways and will take less time if DTMF is used.

Keywords-GSM Module, Float sensor, Solenoid Valve

I. INTRODUCTION

The majority of modern applications and systems are now operated wirelessly, leading to increased production efficiency while reducing time and labor requirements. Manual methods in agricultural systems often result in operational inefficiencies, prompting the need for wireless technology integration [1-14]. The core concept of our project revolves around employing wireless technology to manage agricultural operations, utilizing Dual-Tone Multiple Frequency (DTMF) wireless technology.

Given the continual rise in food demand, advancements in food production technology are imperative. In regions like India, where agriculture and climate conditions are unpredictable, stagnant conditions due to factors such as insufficient rainfall and limited water resources hinder full agricultural resource utilization [15-24]. Thus, the "DTMF Based Irrigation Water Pump Control System" aims to optimize water usage and enhance productivity with minimal manual intervention.

Our project's objective is to remotely regulate various electrical loads, such as irrigation water pumps situated in remote locations, using a telephonic signaling system based on the dual-tone multi-frequency technique [36-42]. By leveraging the DTMF signal received from any mobile phone, users can control electrical loads such as irrigation water pumps, tube wells, farms, and businesses. This approach aims to reduce manual labor for farmers while improving productivity and financial returns.

Moreover, controlling agricultural applications, such as motors, via cell phones simplifies the operation and management process. The mobile phone serves as the primary control device for controlling numerous applications in the field or farm, while also providing real-time monitoring of environmental conditions like humidity, temperature, and light intensity. A DTMF decoder interprets the code attached to analog signals from the mobile device and converts them into corresponding digital commands.

The controller then transmits signals to activate or deactivate relays, thereby controlling the connected loads based on commands from the sender's mobile device. The key advantage of this system lies in its ability to operate applications remotely from farmland or fields [43-53].

This enables farmers to conveniently activate or deactivate water pumps installed at various locations on the land via the cell phone keypad, aligning with their specific requirements. The DTMF decoder and controlling

circuit receive input commands and manage the on-off mode of the connected electrical motor pump. This circuit design is readily achievable using various electrical and electronic circuit components.

II. LITERATURE REVIEW

- Pooja B. Mahale, Bhagyashree B. Bhoir, and Shubhangi B. Shinde present "Smart Agriculture Using DTMF," focusing on software simulation via Arduino IDE and Proteus, followed by hardware development.
- Mandeep Guleria, Akanksha Rathi, Deepraj Singh Sajwan, Jasbir Singh Negi, and Nidhi Chauhan explore the "DTMF Based Irrigation Water Pump Control System," which aims to control remotely positioned agricultural water pumps using the DTMF technique.
- Beza N. Getu, Nasser A. Hamad, and Hussain A. Attia introduce "Remote Controlling of an Agriculture Pump System Based on the Dual Tone Multi Frequency Technique," employing a wireless sensor network and embedded DTMF signaling to manage water flow for various irrigation methods such as sectored, sprinkler, or drip irrigation.
- Amit Verma, Ankit Kumar, Avdesh Sikarwar, and Atul Sahu present "Automatic Irrigation System Using DTMF," aiming to provide farmers with automatic irrigation capabilities utilizing DTMF technology, allowing remote motor operation controlled by a microcontroller 8051.
- V. Ramanjaneyulu, K. Devendra Reddy, and P. Madhuri discuss the "DTMF Based Irrigation Water Pump Control System," detailing the conversion of received cell phone codes into digital commands using a DTMF decoder.



III. METHODOLOGY

Fig. 1. Schematic block diagram for DTMF Based Irrigation Water Control System



Agricultural irrigation often relies on AC motors (either single-phase or three-phase) to power the pump mechanism responsible for water distribution. The size of the motor is chosen based on the dimensions of the agricultural area. In large agricultural sites, multiple motors may be necessary to pump water to various locations, sometimes situated far from the farmer's residence. Utilizing modern technology can alleviate the challenges associated with managing irrigation across vast and remote areas.

With modern technology, a farmer located remotely can control water pumps at different locations on the site by simply pressing the keypad of a telephone handset. Depending on the need, one or more pumps can be activated or deactivated simultaneously or at different intervals. To achieve this, a DTMF decoder and control logic circuit are employed to interpret commands encoded as audio DTMF signals and control high-power pumps accordingly.

Unlike previous designs that relied on commercially available DTMF decoders and microcontrollers, our approach involves designing the DTMF decoder and subsequent logic controller using readily available passive and active electrical and electronic components.

Implementing DTMF-based control for agricultural pump systems is particularly impactful in regions with water and rainfall scarcity, such as desert areas. Here, farmers can effectively manage water resources based on weather, environmental, and seasonal conditions. Remote control functionality enhances flexibility, reducing the need for physical presence to operate pump systems [1].

Beyond agricultural pump control, DTMF-based remote control holds significance for various domestic and industrial applications. It minimizes the risk of leaving equipment running unnecessarily for extended periods in the absence of operators. The ability to switch equipment on and off from any location and at any time enhances convenience and efficiency.

1. Arduino:

The Arduino Nano, developed by the Arduino team, is a microcontroller board akin to the Atmega168 or Atmega328p. It shares similarities with the Arduino Uno but offers a smaller footprint and distinct pin configuration. Ideal for embedded systems and robotics projects, the Nano's compact size caters to design preferences for smaller components. Targeted at novices entering technical realms, Nano boards serve as accessible tools for learning and developing electronic projects.

2. GSM Module:

The SIM800L GSM/GPRS module serves as a compact cellular GSM modem, seamlessly integrating with microcontrollers to imbue them with GSM capabilities, enabling GPRS data transmission. Facilitating phone calls, SMS exchange, and internet connectivity via GPRS, TCP, or IP, it links microcontrollers to mobile networks. Further, its quad-band GSM/GPRS support ensures global operability. Its affordability, compactness, and extensive functionalities render it an ideal choice for projects requiring long-range connectivity and IoT integration.

3. Float Sensor:

Float switches find utility across diverse industrial processes, providing point-level detection to indicate whether a liquid level is open or closed at a specific point. They detect various factors such as foaming, dielectric properties, conductivity, pressure, temperature, and more. This versatility makes them suitable for virtually all liquid types. Operating without direct contact and independent of voltage supply, float switches offer a straightforward functionality applicable across a broad spectrum of applications, spanning from general industrial settings to process plants and shipbuilding industries.

4. Relay Module:

A relay is an electromechanical switch employed to manage high-power applications via low-power signal electronic circuits. For instance, a basic timer circuit operating at a 5V DC bias lacks the capacity to control a high-voltage light bulb directly. By integrating a relay component into the circuit, it becomes feasible to effectively regulate the light bulb's operation.

5. Solenoid Valve:

A solenoid valve functions as an electronically operated valve. It comprises a solenoid consisting of an electric coil encasing a movable ferromagnetic core, known as a plunger, situated at its center. By default, the plunger seals a small orifice when in the resting state. When an electric current passes through the coil, it generates a magnetic field, exerting an upward force on the plunger, thereby opening the orifice. This fundamental mechanism governs the operation of solenoid valves, enabling them to open and close.

6. Pump:

The cooler pump, also known as a compact submersible water pump or centrifugal pump, is commonly utilized in air coolers, aquariums, and fountains. It operates by drawing water through one side as the impeller inside rotates, expelling it from the other end. The impeller's size determines the water flow rate, with larger impellers capable of pumping more water. Through centrifugal force, water moves from the impeller's inlet to the outer volute, where it exits through the outlet, causing a drop in water pressure at the inlet. The pump then draws in new water to match the outgoing flow. These pumps are typically submerged, eliminating the need for priming before operation.

IV. MERITS, DEMERITS AND APPLICATION OF DTMF BASED IRRIGATION WATER PUMP CONTROL SYSTEM

Merits

Advantages of an irrigation water pump include its cost- effectiveness, facilitated by the integration of wireless technology, which reduces the need for manual labor and saves time. Furthermore, the implementation of such technology allows for efficient water management, enabling precise water scheduling to optimize irrigation cycles. Additionally, features like dry run out detection enhance operational safety by preventing damage to the pump in case of water scarcity or malfunction.

Demerits

While the DTMF Control System offers several benefits, it also presents various drawbacks. These include a substantial initial investment and susceptibility to network disruptions caused by changes in climate conditions.

Applications in agriculture sector

The project is applicable to various scenarios, including:

- Crop irrigation
- Efficient water system frameworks, particularly suitable for paddy and rice fields
- Beneficial irrigation practices in agricultural fields

V. FUTURE SCOPE

This initiative has replaced traditional farming watering methods with automated irrigation, significantly reducing water wastage and minimizing the labor-intensive tasks farmers typically undertake. Future enhancements may include integrating sensors to monitor soil humidity levels and land temperatures, especially crucial in areas prone to crop fires. This proactive approach enables timely interventions to prevent crop damage and alleviate farmers' burdens. Additionally, the system finds utility in industrial load management and home automation, offering convenience through DTMF and cellphone- controlled automation systems in various settings, including departmental cabins.

VI. CONCLUSION

The project proposes a cost-effective and straightforward solution for wirelessly and automatically monitoring water levels in tanks. Our design is ideally suited for small towns and specific zones, extending its practicality to residential and office areas. Recognizing the significance of water management around homes and offices, our focus is on implementing an affordable and efficient remote system. Unlike wired setups prone to issues like wire breakage, our solution eliminates such concerns. Moreover, this concept can be scaled up for commercial use and expanded to cover larger areas, replacing wired sensors with remote ones. Furthermore, the remote detection method can be adapted to detect water leaks. Scaling up, this solution could be adopted across municipal structures,

offering comprehensive water monitoring without requiring EEPROM links. Data can be stored locally and transferred to computers for analysis, generating charts and facilitating detailed audits of water usage in specific areas. Stream meters serve as verification tools.

VII. REFERENCES

- Mandeep Guleria, Akanksha Rathi, Deepraj Singh Sajwan, Jasbir Singh Negi and Nidhi Chauhan, "DTMF Based Irrigation Water Pump Control System", Asia-pacific Journal of Convergent Research Interchange Vol.3, No.3, 2017, Pp.: 9-15
- V. Ramanjaneyulu, K. Devendra Reddy and P. Madhuri, "DTMF Based Irrigation Water Pump Control System", International Journal of Research, Volume: 05, Issue: 12, 2018, Pp.: 579-582
- Vinayak Badiger, "Automated Water Management and Distribution System using DTMF", International Journal of Trend in Scientific Research and Development, Volume: 4, Issue: 4, 2020, Pp: 830-833
- Pooja B. Mahale, Bhagyashree B. Bhoir and Shubhangi B. Shinde, "DTMF Based Smart Agriculure", Spvryan's International Journal of Engineering Sciences & Technology, Volume: 2, Issue: 3, Pp:1-5
- A.Lydia, and J. Jayaseelan, "Design And Development Of Gsm Based Automatic Irrigation Control System", International Journal of Engineering Research & Technology, Volume: 03, Issue 04, 2015, a. Pp: 1-5
- Prof. P. M. Mankar, Tushar P. Mohurle, Dinanath Komrekar, Mayur Shinde, Mayuri Mahajan, and Sakshi Thakare, "Gsm Based Solar Water Pump Control with Power Saving Application", International Research Journal of Modernization in Engineering Technology and Science, Volume:04, Issue:05, 2022, Pp:1161-1166
- Er. Anshoo Sarswat, Hitendra Pratap Singh, Shaynak Saxena, Yash Chaug and Sanu Mandal, "Remote Irrigation and Monitoring System Using GSM Technology", International Journal of Electrical and Electronics Research, Vol. 3, Issue 2, 2015 Pp: 277-280
- Beza N. Getu, Nasser A. Hamad and Hussain A. Attia, "Remote Controlling of an Agricultural Pump System Based on the Dual Tone Multi-Frequency (DTMF) Technique", Journal of Engineering Science and Technology, Vol. 10, No.10, 2015, Pp: 1261-127
- 9. P.P. Mahesh, S. Khismatrao, V. Gadge, A. Thampi, and K. Kalambe, "DTMF Based Agriculture Pump Control", Vol. 6, No. 3, pp. 165-169, (2016)
- D. Vandana, D. Nilesh, and C. Shailesh, "Wireless Sensor Network Based Remote Irrigation Control System and Automation Using Dtmf Code", International Conference on Communication Systems and Network Technologies, pp. 34-37, (2011).
- 11. Clemens, A.J. 1990.Feedback Control for Surface Irrigation Management in: Visions of the Future. ASAE Publication 04-90. American Society of Agricultural Engineers, St. Joseph, Michigan, pp. 255-260
- R. Sharma, K. Kumar, and S. Vig, "DTMF based remote control system," in Proceedings of the IEEE International Conference on Industrial Technology, 2006, pp. 2380–2383.
- M. Guleria, A. Rathi, D. S. Sajwan, J. S. Negi, and N. Chauhan, "Dtmf based irrigation water pumpcontrol system," vol. 3, no. 3, pp. 1–6, 2017
- 14. Tuljappa M Ladwa, Sanjay M Ladwa, R Sudharshan Kaarthik, Alok Ranjan, Dhara, Nayan Dalei "Control of Remote Domestic System Using DTMF" IEEE ICICI-BME 2009 Bandung, Indonesia
- Nilesh Dubey, A. Bunglowala, Shivangi Bande and Vandana Dwivedi "Embedded based DTMF Signaling Remote Control System for Irrigation and Pesticide Spray" National conference on Electronics, Computers and Communications (NCECC-2010)
- Rohit S.; Kushagra K.; and Shashank V. (2006). DTMF Based Remote Control System. IEEE International Conference on Industrial Technology (ICIT), 2380-2383
- Suhas. B. Khadake, Prajakta. V. Padavale, Priyanka. M. Dhere, Bharati. M. Lingade Automatic Hand Dispenser And Temperature Scanner For Covid-19 Prevention, Ijarsct Vol-3.Issue-3 June 2023, Doi: 10.48175/Ijarsct-11364

- Suhas B. Khadake, "Detecting Salient Objects Of Natural Scene In A Video's Using Spatio-Temporal Saliency & Amp; Colour Map", Journalnx, Vol. 2, No. 08, Pp. 30–35, Feb. 2021.
- 19. Kazi Kutubuddin, "Blockchain-Enabled IoT Environment to Embedded System a Self-Secure Firmware Model", Journal of Telecommunication study, 2023, Vol 8, Issue 1
- Kazi Kutubuddin, "A Study HR Analytics Big Data in Talent Management", Research and Review: Human Resource and Labour Management, 2023, Volume-4, Issue-1, pp. 16-28
- 21. Vahida Kazi, et al, " Deep Learning, YOLO and RFID based smart Billing Handcart", Journal of Communication Engineering & Systems, 2023, 13(1), pp. 1-8
- 22. Kazi Kutubuddin Sayyad Liyakat, "Analysis for Field distribution in Optical Waveguide using Linear Fem method", Journal of Optical communication Electronics, 2023, Vol 9, Issue 1, pp. 23-28
- 23. Sultanabanu Kazi, et al.(2023), Fruit Grading, Disease Detection, and an Image Processing Strategy, Journal of Image Processing and Artificial Intelligence, 9(2), 17-34.
- Sultanabanu Kazi, Mardanali Shaikh, "Machine Learning in the Production Process Control of Metal Melting" Journal of Advancement in Machines, Volume 8 Issue 2 (2023) Available at: https://matjournals.co.in/index.php/JoAM/article/view/4102
- 25. Priya Mangesh Nerkar, Sunita Sunil Shinde, et al, "Monitoring Fresh Fruit and Food Using Iot and Machine Learning to Improve Food Safety and Quality", Tuijin Jishu/Journal of Propulsion Technology, Vol. 44, No. 3, (2023), pp. 2927 – 2931 https://propulsiontechjournal.com/index.php/journal/article/view/914
- 26. Kazi Sultanabanu Sayyad Liyakat (2023). Integrating IoT and Mechanical Systems in Mechanical Engineering Applications, Journal of Mechanical Robotics, 8(3), 1-6. Available at: https://matjournals.co.in/index.php/JoMR/article/view/4493
- 27. Kazi Sultanabanu Sayyad Liyakat (2023). IoT Changing the Electronics Manufacturing Industry, Journal of Analog and Digital Communications, 8(3), 13-17. Available at:
- 28. Kazi Sultanabanu Sayyad Liyakat (2023). IoT in the Electric Power Industry, Journal of Controller and Converters, 8(3), 1-7. Available at: https://matjournals.co.in/index.php/JCC/article/view/4644
- Kazi Sultanabanu Sayyad Liyakat (2023). Review of Integrated Battery Charger (IBC) for Electric Vehicles (EV), Journal of Advances in Electrical Devices, 8(3), 1-11. Available at: https://matjournals.co.in/index.php/JAED/article/view/4642
- Kazi Sultanabanu Sayyad Liyakat (2023). ML in the Electronics Manufacturing Industry, Journal of Switching Hub, 8(3), 9-13
- Prof.Suhas B.Khadake, Prof. Sudarshan P.Dolli, Mr. K.S.Rathod, Mr. O.P.Waghmare, And Mr.A.V.Deshpande, "An Overview Of Intelligent Traffic Control System Using Plc And Use Of Current Data Of Vehicle Travels", Journalnx, Pp. 1–4, Jan. 2021.
- 32. Ameykumar Balkrishna Dudgikar , Adnan Ahmad Akbar Ingalgi , Abhishek Gensidha Jamadar , Onkar Rameshchandra Swami , Suhas Balram Khadake , Shreya Vikram Moholkar "Intelligent Battery Swapping System For Electric Vehicles With Charging Stations Locator On Iot And Cloud Platform".. Ijarsct, Volume-3, Issue-1, January-2023, Doi: 10.48175/Ijarsct-7867
- V D Gund, et al. (2023). PIR Sensor-Based Arduino Home Security System, Journal of Instrumentation and Innovation Sciences, 8(3), 33-37.
- Kazi Kutubuddin Sayyad Liyakat (2023), System for Love Healthcare for Loved Ones based on IoT. Research Exploration: Transcendence of Research Methods and Methodology, Volume 2, ISBN: 979-8873806584, ASIN : B0CRF52FSX.
- Kazi, K. (2024). AI-Driven IoT (AIIoT) in Healthcare Monitoring. In T. Nguyen & N. Vo (Eds.), Using Traditional Design Methods to Enhance AI-Driven Decision Making (pp. 77-101). IGI Global. https://doi.org/10.4018/979-8-3693-0639-0.ch003.
- Pankaj R Hotkar, Vishal Kulkarni, et al, "Implementation of Low Power and area efficient carry select Adder", International Journal of Research in Engineering, Science and Management, 2019, Vol 2, Issue 4, pp. 183 - 184.

- 37. Kazi K S, " Detection of Malicious Nodes in IoT Networks based on Throughput and ML", Journal of Electrical and Power System Engineering, 2023, Volume-9, Issue 1, pp. 22- 29.
- Karale Nikita, Jadhav Supriya, et al, "Design of Vehicle system using CAN Protocol", International Journal of Research in Applied science and Engineering Technology, 2020, Vol 8, issue V, pp. 1978 - 1983, http://doi.org/10.22214/ijraset.2020.5321.
- 39. K. Kazi, "Lassar Methodology for Network Intrusion Detection", Scholarly Research Journal for Humanity science and English Language, 2017, Vol 4, Issue 24, pp.6853 6861.
- 40. Miss Argonda U A, "Review paper for design and simulation of a Patch antenna by using HFSS", International Journal of Trends in Scientific Research and Development, 2018, Vol 2, issue-2, pp. 158 -160.
- 41. [Kazi K., "Hybrid optimum model development to determine the Break", Journal of Multimedia Technology & Recent Advancements, 2022, vol 9, issue 2, pp. 24 32.
- 42. Ms. Yogita Shirdale, et al, "Analysis and design of Capacitive coupled wideband Microstrip antenna in C and X band: A Survey", Journal GSD-International society for green, Sustainable Engineering and Management, 2014, Vol 1, issue 15, pp. 1 7.
- Ms. Shweta Nagare, et al., "Different Segmentation Techniques for brain tumor detection: A Survey", MM-International society for green, Sustainable Engineering and Management, 2014, Vol 1, issue 14, pp.29 - 35.
- Kazi K., "Reverse Engineering's Neural Network Approach to human brain", Journal of Communication Engineering & Systems, 2022, vol 12, issue 2, pp. 17 – 24.
- 45. Ms. Yogita Shirdale, et al., "Coplanar capacitive coupled probe fed micro strip antenna for C and X band", International Journal of Advanced Research in Computer and Communication Engineering, 2016, Vol 5, Issue 4, pp. 661 - 663.
- Khadake Suhas .B. (2021). Detecting Salient Objects In A Video's By Usingspatio-Temporal Saliency &Amp; Colour Map. International Journal Of Innovations In Engineering Research And Technology, 3(8), 1–9.
- Khadake, S., Kawade, S., Moholkar, S., Pawar, M. (2024). A Review of 6G Technologies and Its Advantages Over 5G Technology. In: Pawar, P.M., et al. Techno-societal 2022. Springer, Cham. https://doi.org/10.1007/978-3-031-34644-6_107
- Suhas B khadake , Pranita J Kashid, Asmita M Kawade , Santoshi V Khedekar, , H. M. Mallad et al. (2023) Electric Vehicle Technology Battery Management - Review, IJARSCT- Volume-3, Issue-2,September-2023, DOI: 10.48175/IJARSCT-13048
- Halli U M, "Nanotechnology in IoT Security", Journal of Nanoscience, Nanoengineering & Applications, 2022, Vol 12, issue 3, pp. 11 – 16.
- Wale Anjali D., Rokade Dipali, Et Al, "Smart Agriculture System Using Iot", International Journal Of Innovative Research In Technology, 2019, Vol 5, Issue 10, Pp.493 - 497.
- 51. Kazi K. S., "Significance And Usage Of Face Recognition System", Scholarly Journal For Humanity Science and English Language, 2017, Vol 4, Issue 20, pp. 4764 4772.
- 52. V. J. Patil, S. B. Khadake, D. A. Tamboli, H. M. Mallad, S. M. Takpere and V. A. Sawant, "Review of AI in Power Electronics and Drive Systems," 2024 3rd International conference on Power Electronics and IoT Applications in Renewable Energy and its Control (PARC), Mathura, India, 2024, pp. 94-99, doi: 10.1109/PARC59193.2024.10486488.
- 53. S. B. Khadake and V. J. Patil, "Prototype Design & Development of Solar Based Electric Vehicle," 2023 3rd International Conference on Smart Generation Computing, Communication and Networking (SMART GENCON), Bangalore, India, 2023, pp. 1-7, doi: 10.1109/SMARTGENCON60755.2023.10442455.

Development of a Test-Bench for Battery-Motor-Propeller Assemblies Designed for Multirotor Drone

R. N. Surwase¹, S. S. Lokhande², S. U. Patil³, A. A. Shaikh⁴, M. P. Thakre5, S. S. Kawade⁶ 1-6Elect. Engg. Dept., SVERI's College of Engineering Pandharpur MH- India ¹rohannsurwase@coep.sveri.ac.in ²sarthakslokhande@coep.sveri.ac.in ³sagarupatil@coep.sveri.ac.in ⁴amanashaikh@coep.sveri.ac.in ⁵mohanthakre@gmail.com ⁶sagarkawade844@gmail.com

Abstract: This paper highlights the necessity for a thorough test bench tailored for battery-motor-propeller assemblies in drones, offering a comprehensive tool for drone testing. Our emphasis lies in methodical evaluation, enabling the examination of vital drone parameters such as flight time, motor thrust, and overall system performance. Leveraging mathematical modeling, including MATLAB simulations, facilitates precise measurement of drone flight characteristics and motor behavior. Real-world validation further enhances the accuracy and reliability of the proposed methodologies. Future endeavors will focus on algorithm optimization and the incorporation of additional tests to assess drone performance, such as power consumption and efficiency tests.

Keywords: Drone Technology, Test-bench Development, Battery-Motor-Propeller Assemblies, Mathematical Modelling, Flight Characteristics

I. INTRODUCTION

In the past few years, the increased use of drones has increased the demand for improvements in drones. The core of these unmanned aerial vehicles (UAVs) resides in their propulsion systems, where the complex interaction among batteries, electric motors, and propellers plays a key role in determining the drone's performance.

The expanding range of applications, spanning from aerial photography to agriculture and surveillance, requires a sturdy and effective design for these fundamental components. Traditional multi-rotor propulsion systems, consisting of fixed-pitch propellers and electric motors powered by battery packs, have limitations in terms of endurance and range, primarily constrained by battery capacity [1]. This has led to an increasing focus on enhancing the design and performance of battery-motor- propeller assemblies to address these limitations.

The advancements in multirotor unmanned aerial vehicles (UAVs) have witnessed a rapid increase in their use across various industries such as cargo transportation, mapping, agriculture, and more [2]. However, the predominant use of fixed-pitch propellers in many designs has contributed to challenges in achieving extended hover times [2].

Recognizing this limitation, researchers have explored innovative solutions, including variable-pitch propellers, to improve efficiency and control in different operating conditions [2]. Additionally, the development of test systems for pre-flight control and evaluation of Vertical Take-Off and Landing (VTOL) drones has become crucial [3]. These systems, equipped with test platforms and graphical user interfaces, facilitate comprehensive pre- flight tests, ensuring the safety and optimal performance of VTOL multirotor drones [3].

In the context of experimental test benches, there have been significant strides in designing test platforms for various types of UAVs, including convertiplane types with controlled electric drives and tilt rotors [4]. These test benches serve as intermediaries between simulation processes and real-time UAV flights, enabling the evaluation of solutions and the development of control methods [4]. Moreover, the educational aspect of multirotor UAVs is not overlooked. A modular design approach has been proposed for the development of an Educational Multirotor

Platform (EMMR), offering students opportunities to delve into control theory and aerial robotics [5]. The aerodynamic performance of propellers, particularly about geometric parameters such as pitch, has been a subject of in-depth investigation [6].

Experimental results, obtained through wind tunnel tests and dimensional analysis, contribute to the understanding of propeller performance under various conditions [6]. The interest in electrically-driven propeller airplanes has expanded from small remotely piloted vehicles to larger high-altitude/long-endurance aircraft, with a focus on accurate range and endurance predictions [7]. As the drone industry evolves, there is a growing need for comprehensive testing platforms and methodologies. These include thrust test benches capable of measuring parameters such as thrust, RPM, power, torque, and temperature [9]. The development of such test benches is crucial for assessing the structural and functional behavior of UAV engines or motors. This paper deals with the detailed analysis of multirotor drones. By using mathematical models of each part of the drone, we developed a MATLAB program that takes input from the user and outputs the test results. So, tests on each part of the drone have been carried out and test results have been compared with real drone hardware.

In recent years, the escalating utilization of drones has spurred a demand for enhancements in their design and performance. At the heart of these unmanned aerial vehicles (UAVs) lies their propulsion systems, where the intricate interplay among batteries, electric motors, and propellers significantly influences drone performance.

The burgeoning array of applications, ranging from aerial photography to agriculture and surveillance, necessitates robust and efficient designs for these pivotal components. Traditional multi-rotor propulsion systems, comprising fixed-pitch propellers and electric motors powered by battery packs, face limitations in endurance and range, primarily due to battery capacity constraints. This limitation has spurred an intensified focus on enhancing the design and performance of battery-motor-propeller assemblies to overcome these challenges.

The rapid advancement of multirotor UAVs has seen a surge in their application across various industries such as cargo transportation, mapping, and agriculture. However, the prevalent use of fixed-pitch propellers has posed challenges in achieving prolonged hover times. In response, researchers have explored innovative solutions, including variable-pitch propellers, to enhance efficiency and control across different operational conditions. Moreover, the development of test systems for pre-flight control and evaluation of Vertical Take-Off and Landing (VTOL) drones has become indispensable. These systems, equipped with test platforms and user-friendly interfaces, facilitate comprehensive pre-flight tests, ensuring the safety and optimal performance of VTOL multirotor drones.

Significant strides have also been made in designing experimental test benches for various types of UAVs, including convertiplane types with controlled electric drives and tilt rotors. These test benches bridge the gap between simulation processes and real-time UAV flights, enabling the evaluation of solutions and the development of control methods.

Furthermore, the educational dimension of multirotor UAVs is being explored through a modular design approach for an Educational Multirotor Platform (EMMR), offering students opportunities to delve into control theory and aerial robotics. The analysis of propeller aerodynamic performance, particularly concerning geometric parameters such as pitch, has been subject to in-depth investigation, with experimental results contributing to understanding propeller performance under various conditions.

As the drone industry continues to evolve, there is an escalating need for comprehensive testing platforms and methodologies, including thrust test benches capable of measuring parameters such as thrust, RPM, power, torque, and temperature. Such developments are crucial for assessing the structural and functional behavior of UAV engines or motors. This paper delves into a detailed analysis of multirotor drones. Utilizing mathematical models of each drone component, we developed a MATLAB program that solicits input from users and generates test results. Consequently, tests on each drone component have been conducted and compared with real drone hardware, providing valuable insights into their performance.

II. DETAILING OF DRONE

2.1 Dynamics of Flight

Flight dynamics is the branch of aerospace engineering that focuses on the study of the motion and behavior of aircraft in response to external forces, control inputs, and

environmental conditions. It encompasses the principles governing the stability, control, and maneuverability of flying vehicles, including airplanes, helicopters, and drones.

Air is a physical substance with weight, comprised of constantly moving molecules. Air pressure is generated by the movement of air molecules, creating forces that affect objects like kites, balloons, birds, and planes.

Air is a mixture of gases, including oxygen, carbon dioxide, and nitrogen, and it plays a crucial role in the flight of various objects. In 1640, Torricelli discovered that air has weight, observed through its mercury pressure. Francesco Lana conceptualized an airship in the late 1600s, based on the idea that removing air from a hollow sphere would allow it to float due to reduced weight. Hot air expands and becomes lighter than cool air, causing balloons filled with hot air to rise. When hot air cools and is released, the balloon descends.

Airplane wings are shaped to increase airspeed over the top, creating lower pressure, resulting in lift. The difference in pressure between the top and bottom of the wing generates a force that lifts the aircraft.

The laws of motion of Newton are crucial for a drone to be able to fly. According to the first law of motion, until an outside force acts against an object, it will remain at rest and an object in motion will continue to move. According to the second law of motion, an object's force is determined by multiplying its mass by its acceleration. Additionally, according to the third law of motion, there is an equal

A drone follows all three laws of motion. Before takeoff, a drone is at rest, and according to the first law, it will remain stationary until force is applied. The propellers generate the necessary thrust, overcoming inertia and initiating motion. Similarly, once in motion, the drone will continue moving until external forces, like air resistance or control inputs, act upon it.

When a drone takes off, the force generated by its propellers accelerates it upward. The acceleration is influenced by the net force created by the thrust and the drone's mass. To change direction or altitude, the drone's flight controller adjusts the thrust, adhering to Newton's second law.

The propellers of a drone create thrust by pushing air downward. According to Newton's third law, the reaction to this action is an equal force pushing the drone upward. This principle is crucial for lift generation in drones. Additionally, when the drone maneuvers or changes direction, the corresponding propeller adjustments follow the same action-reaction principle. Lift (upward force), Drag (backward force), Weight (downward force), and Thrust (forward force) are the four fundamental forces influencing flight.

The pilot controls the plane's yaw, pitch, and roll using various control surfaces such as ailerons, elevators, and rudders. Yaw is controlled by the rudder, pitch by the elevators, and roll by the ailerons. The pilot uses instruments such as the throttle for engine power, ailerons for roll, rudder for yaw, and elevators for pitch. These controls allow the pilot to manage the direction, altitude, and stability of the aircraft during flight.

2.2 Drone Subparts

A drone, short for unmanned aerial vehicle (UAV), is a remotely piloted or autonomously operated aircraft. Drones come in various sizes and configurations, from small consumer quad-copters to large military surveillance and reconnaissance vehicles. They are equipped with sensors, cameras, or other payloads, enabling them to capture data, images, or videos, and are used for a wide range of applications, including recreational purposes, aerial photography, surveillance, search and rescue, agriculture, and more. Drones are typically controlled by a ground-based operator using a remote control or a dedicated software interface.

The frame serves as the structural foundation of the drone, providing support for all other components. Frames are commonly made of lightweight materials like carbon fibre or aluminum to optimize the drone's weight and durability. Propellers generate lift and thrust, allowing the drone to ascend, descend, and move in different directions. Drones often have multiple propellers (quad-copters, hexa-copters, octo-copters) for enhanced stability and control.

Motors drive the propellers' rotation, creating the necessary thrust for flight. Brushless motors are commonly used in drones due to their efficiency and reliability. ESCs regulate the speed of each motor, translating

electronic signals from the flight controller into the appropriate motor speed. ESCs contribute to maintaining the drone's stability and responsiveness.

The flight controller is the brain of the drone, processing input from sensors and adjusting motor speeds to stabilize and control the drone's movement. It may include an inertial measurement unit (IMU), gyroscopes, and accelerometers to sense changes in orientation and acceleration.

Drones are powered by rechargeable batteries, typically lithium-polymer (Li-Po) batteries. Battery capacity and voltage impact flight time, with larger batteries providing longer endurance. Various electronic components include a power distribution board, wiring, and connectors. The power distribution board distributes power from the battery to the ESCs and other electronic components. Camera Drones are equipped with cameras that can take photos and videos.

Gimbals stabilize the camera, ensuring smooth footage by compensating for drone movements. GPS modules enable accurate positioning, navigation, and return-to-home functionalities. Drones may use GPS for way-point navigation and geo-fencing. The radio transmitter is the handheld controller operated by the pilot. The receiver on the drone receives signals from the transmitter, translating them into commands for the flight controller. Telemetry systems provide real-time data on the drone's status, including altitude, speed, battery level, and GPS coordinates.

2.3 Types of Drones

Drones can be classified into various types based on different criteria. Here are some common classifications, Based on Application. Recreational drones are used for fun and hobbyist purposes. Professional drones are designed for commercial applications like photography, surveying, and mapping. Military drones are utilized for defense and surveillance purposes. Based on configuration, drones are classified as multi-rotor drones that have multiple rotors, commonly used for aerial photography. Fixed-wing drones resemble airplanes and are suitable for covering large areas. Single-rotor helicopter drones feature a single large rotor and are often used for heavy lifting. Fixed-wing Hybrid VTOL drones combine the benefits of fixed-wing and multi-rotor designs.

Based on size drones are classified as very small nano drones, and are often used for military surveillance. Small drones are suitable for indoor use and recreational photography. Medium drones, which are larger and heavier, are used for professional applications. Large drones are comparable to smaller aircraft and are used for military and civil applications.

Based on payload capacity, drones are classified as featherweight drones that carry light payloads, often used for military surveillance. Lightweight drones are used for recreation and photography, with a moderate payload capacity. Middleweight drones have higher payload capacities, suitable for professional applications. Heavy-lift drones are designed to carry significant payloads, used for combat and civil applications.

Based on range, drones can be classified as very close-range drones limited to a short distance, often for recreational use. Close-range drones can cover short distances, used for military surveillance and photography. Short-range drones cover up to 150 km, suitable for surveillance and mapping. Mid-range drones, capable of longer flights, are often used for military purposes. Long-range drones cover extensive distances, used for military surveillance and mapping.

Based on power sources, drones are classified as battery- powered drones which use batteries for power. They are common in recreational and professional drones. Gasoline- powered drones use gasoline for longer flight times and heavier payloads. Hydrogen fuel cell drones use hydrogen for efficiency and environmental friendliness.

Solar drones utilize solar energy for extended flight times. Based on motors, drones are classified as brushed drone motors are common in recreational drones, but require more maintenance. Brush-less drone motors are more efficient, require lower maintenance, and provide higher power. Based on abilities & equipment, drones are classified as toy drones which are designed for beginners, often found in toy departments.

Photography/ videography drones are equipped with cameras for capturing images and videos. Racing drones are used for competitive drone racing. Professional drones feature advanced equipment for specialized
applications. Military drones are specifically designed for military use. Delivery drones are designed for transporting goods.

2.4 Applications of Drone

Aerial photography drones have become instrumental in reshaping the landscape of cinematography. traditionally, capturing dynamic and fast-paced action scenes, along with sci-fi sequences, required expensive helicopters and cranes. aerial drones, equipped with advanced stabilization systems and high-quality cameras, have made cinematography more accessible and cost-effective.

Beyond the film industry, drones are making significant strides in real estate and sports photography, providing unique perspectives and angles. Additionally, journalists are exploring the use of drones for live broadcasts, enabling them to collect footage and information from vantage points that were once challenging to access.

Major companies like amazon, UPS, and DHL are actively exploring the use of drones for delivery services. The potential benefits are immense, from saving significant manpower to alleviating road traffic congestion. Drones offer a swift and efficient means of delivering small packages, food, letters, medicines, and other goods, particularly over shorter distances. The concept of drone delivery has the potential to transform the logistics and e- commerce industries, offering faster and more streamlined delivery services.

Geographic mapping drones have democratized geographic mapping, making high-resolution data collection accessible to both amateurs and professionals. These unmanned aerial vehicles are particularly valuable in reaching difficult-to-access locations such as coastlines, mountaintops, and islands.

The data collected by drones is utilized for creating detailed 3d maps and contributing to crowd-sourced mapping applications. The ability to acquire precise and up- to-date geographical information has applications in urban planning, environmental monitoring, and disaster response. Disaster management in the aftermath of natural or man- made disasters.

Drones serve as invaluable tools for disaster management. Equipped with high-definition cameras, sensors, and radars, drones provide quick and detailed information to aid in search and rescue operations. Their small size and maneuverability allow them to navigate through debris and rubble, offering a higher field of view compared to traditional methods.

Drones present a cost-effective alternative to manned helicopters, minimizing risks and optimizing resources during critical phases of disaster response. Precision agriculture has witnessed a significant transformation with the integration of drones. Farmers and agriculturists now have an affordable and effective means of regularly monitoring their crops.

Equipped with infrared sensors, drones can detect crop health variations, enabling farmers to respond promptly with targeted interventions such as fertilizer or insecticide applications. This not only enhances crop management practices but also contributes to improved yields. With drones expected to constitute a substantial portion of the agricultural market, their role in precision agriculture is set to grow even further.

Search and rescue drones play a crucial role in search and rescue operations, particularly in challenging terrain or adverse conditions. Their thermal sensors provide night vision capabilities, aiding in the location of lost people or victims. Drones can access areas where larger aerial vehicles might prove perilous or inefficient.

In addition to locating individuals, drones can be deployed to drop essential supplies to stranded victims in disaster-stricken or war-torn regions before traditional rescue crews can mobilize. Weather forecast drones are emerging as valuable tools for monitoring and collecting data on dangers and unpredictable weather conditions. Their cost-effectiveness and unmanned nature make them suitable for tasks such as flying into hurricanes and tornadoes to provide valuable insights into their behaviors. Specialized sensors on drones can detail weather parameters, contributing to more accurate weather forecasting. The ability to gather data in real-time from within extreme weather events has the potential to enhance our understanding of meteorological phenomena. Wildlife monitoring drones play a vital role in wildlife conservation by serving as a deterrent to poachers.

Equipped with thermal cameras and sensors, drones operate during the night, providing unprecedented protection to animals like elephants, rhinos, and big cats, common targets for poaching. The ability to monitor

wildlife without causing disturbance allows researchers and conservationists to gain valuable insights into animal behavior, migration patterns, and habitat preferences.

Law enforcement agencies are increasingly turning to drones for surveillance and monitoring activities. Drones offer a bird's-eye view of large crowds during events, enhancing public safety measures. They prove invaluable in monitoring criminal and illegal activities, providing law enforcement with crucial situational awareness. Border patrols, for example, use drones to monitor activities such as fire investigations, the smuggling of migrants, and the illegal transportation of drugs along coastlines, where traditional methods may fall short.

Entertainment drones have entered the realm of entertainment, contributing to innovative experiences and competitions. Drone-based fight clubs, where contenders and their drones go head-to-head, showcase the agility and capabilities of these unmanned aerial vehicles. Artificial drone intelligence is harnessed not only for competitive purposes but also for capturing captivating videos and photographs. Concepts like the drone, designed for taking selfies, demonstrate the creative and recreational potential of drones in the entertainment industry.

As drone technology continues to evolve, new and exciting applications in the realm of entertainment are likely to emerge. Drone testing is a comprehensive process designed to assess the performance, safety, and compliance of unmanned aerial vehicles (UAVs) or drones. Testing is crucial to ensure that drones meet specific standards, regulations, and functional requirements, providing confidence in their reliability and safe operation.

III. DEVELOPMENT OF TEST BENCH FOR DRONE

First, we decided on the specific things we wanted to measure and understand about our project. For example, if we're working on a drone, we might want to know how high it can fly, how long it can stay in the air, and how quickly it can move. These are the things we'll keep an eye on to see how well our project is doing.

Next, we use math to create models that help us understand and predict how our project should work. Think of it like making a map before you go on a journey. These models tell us what we can expect from our project under different conditions. For our drone, it's like figuring out how fast it can go based on its motors and how much power the battery provides. To test our project, we write computer programs using a tool called MATLAB.

These programs help us gather data and simulate how our project should behave. For our drone, we can use MATLAB to see how high it should fly, how long the battery should last, and how fast it should move under different circumstances. It's like trying out our project in a virtual world. We then run tests on our project in the real world and collect data.

For our drone, we might take it out for flights and measure how high it actually goes or how long it stays in the air. After the tests, we compare the real results to what our mathematical models and MATLAB simulations predicted. This helps us understand if our project is working as expected or if there are any problems.

We make sure that the numbers we calculated using math and MATLAB match with what we see in the real world. If they match, it's a good sign that our project is doing what it's supposed to do. If there are differences, we investigate to figure out why and make necessary adjustments. This method helps us build and test projects in a structured way, ensuring that they meet our goals and perform as we intend them to.



Fig. 1 Flowchart of the Testing Process

The flowchart for a drone project (Fig.1) encapsulates the sequential processes essential for the operational framework of the unmanned aerial vehicle (UAV). It commences with power-on initialization procedures, encompassing the activation and initialization of the drone's core systems, software, and hardware components.

This phase establishes the groundwork for subsequent functions by ensuring the proper initialization of sensors, internal systems, and communication interfaces. Following initialization, the flowchart progresses to sensor calibration and comprehensive system checks. Sensor calibration is critical for accurate data acquisition and interpretation during flight operations.

The systematic checks performed subsequently serve to validate the operational readiness of critical components and software functionalities, assuring their optimal performance and reliability. The subsequent stage within the flowchart delineates the flight mode selection process. This pivotal phase allows the operator to choose from a range of flight modes, which can include manual, autonomous, or stabilized flight modes. The chosen mode determines the drone's operational behavior and response to user inputs, showcasing the flexibility and adaptability of the UAV for various operational scenarios.

The subsequent segment focuses on flight control, managing motor speeds based on real-time sensor inputs and operator commands, and facilitating controlled and precise flight maneuvers. Throughout the flight, continuous monitoring of parameters such as position, altitude, and battery level is conducted, ensuring real-time awareness of the drone's operational status.

In the event of issues or user-initiated commands, the flowchart illustrates the transition to landing or return-to- home procedures, highlighting safety measures and contingency protocols. Finally, the flowchart concludes with the shutdown process encompassing landing, power- off, and system maintenance, ensuring safe termination of flight operations and maintenance of the drone's overall integrity.

3.1 Testing drone Flight Time

Drone flight time is given by the formula in (1)

 $Time = Capacity \times Discharge \, / \, AAD$

Where, Time – Flight time of the drone, expressed in minutes. Capacity - The capacity of the battery expressed in milliampere hours (mAh) or ampere-hours (Ah). can be found printed on a Li-Po battery. The higher the capacity, the more energy is stored in the battery. Discharge – Battery discharge during the flight. It is good practice not to discharge more than 80% of the charge as lithium polymer batteries can be damaged when fully discharged. AAD – Average amp draw of drone, calculated in amperes. The amp draw is based on parameters such as the quadcopter weight or battery voltage.

The following (2) is used to calculate the AAD:

$$AAD = AUW \times P / V$$

(2)

(3)

(1)

Where, AAD - Average amp draw, expressed in amperes., AUW - All-up weight of drone – the total weight of the equipment that goes up in the air, including the battery. It is usually measured in kilograms. P - The power required to lift one kilogram of material, expressed in watts per kilogram, V – battery voltage expressed in volts. By Ohm's law, we use an alternative version i.e. (3) of the formula above:

 $AAD = AUW \times I$ Where, I - The current (in amps) required to lift one kilogram into the air.

IV. MATLAB PROGRAM FOR TESTING DRONE FLIGHT TIME

The explanation of the code is as follows: User Input: The code begins by prompting the user to input several parameters related to the drone. These inputs include the weight of the frame, battery, payload, battery capacity, battery discharge percentage, and battery voltage.

Drone Weight Calculation: The total weight of the drone is then calculated by summing the weights of the frame, battery, and payload. Average Amp Draw (AAD) Calculation: Assuming a power consumption rate of 105 W/kg, the average amp draw (AAD) is calculated based on the weight of the drone and the battery voltage.

% Prompting user for required inputs frameWeight = input('Enter the weight of the frame in grams: '); batteryWeight = input('Enter the weight of the battery in grams: '); payloadWeight = input('Enter the weight of the payload in grams: '); capacity_mAh = input('Enter the battery capacity in (mAh): '); discharge = input('Enter the battery discharge in percentage : '); batteryVoltage = input('Enter the battery voltage in volts: '); % Calculating drone weight droneWeight payloadWeight;= FrameWeight + batteryWeight + % Calculating average amp draw (AAD) powerPerKg = 105; % Assuming 105 W/kg AAD = droneWeight * powerPerKg / batteryVoltage; % Calculating flight time time_minutes = (capacity_mAh * discharge / 100) / AAD * 60; % Convert hours to minutes % Displaying results fprintf('Drone Flight Time: %.2f minutes\n', time_minutes);

The explanation of the code is as follows: User Input: The code begins by prompting the user to input several parameters related to the drone. These inputs include the weight of the frame, battery, payload, battery capacity, battery discharge percentage, and battery voltage.

Drone Weight Calculation: The total weight of the drone is then calculated by summing the weights of the frame, battery, and payload. Average Amp Draw (AAD) Calculation: Assuming a power consumption rate of 105 W/kg, the average amp draw (AAD) is calculated based on the weight of the drone and the battery voltage.

4.1 Flight Time Calculation

The flight time is then calculated using the battery capacity, discharge percentage, and the previously calculated average amp draw (AAD). The result is converted from hours to minutes.

Result Display: Finally, the calculated drone flight time is displayed using fprintf.

Results: We conducted tests with the above program for various drones

4.2 Dji Mavic Pro

DJI Mavic Pro has a battery of 11.4 V,3830 mAh capacity, and the combined weight of frame, battery, and payload is 743 grams. Input these values into our program

4.3 DJI Phantom 4 Pro

DJI Phantom 4 Pro has a battery of 15.2 V,5870 mAh capacity and the combined weight of frame, battery, and payload of 1388 grams. Input these values into our program

Enter the weight of the frame in grams: 887 Enter the weight of the battery in grams: 468Enter the weight of the payload in grams: 33Enter the battery capacity in (mAh): 5870 Enter the battery discharge percentage: 80 Enter the battery voltage in volts: 15.2 Drone Flight Time: 29.39 minutes

We got a flight time of 29.39 minutes, which is nearly equal to its real flight time of 30 minutes.

4.4 Parrot ANAFI Ai

Parrot ANAFI Ai has a battery of 4.4 V,6800 mAh capacity and the combined weight of frame, battery, and payload of 1300 grams. Input these values into our program

Enter the weight of the frame in grams: 570 Enter the weight of the battery in grams: 198 Enter the weight of the payload in grams: 32 Enter the battery capacity in (mAh): 3500 Enter the battery discharge percentage: 70 Enter the battery voltage in volts: 11.55 Drone Flight Time: 34 minutes

We got a flight time of 34.09 minutes, which is close to its real flight time of 32 minutes.

4.5 DJI MAVIC AIR 2

Parrot ANAFI Ai has a battery of 11.55 V, 3500 mAh capacity, and the combined weight of the frame, battery, and payload is 800 grams. Input these values into our program

We got a flight time of 34 minutes, which is equal to its real flight time of 34 minutes.

V. MATLAB PROGRAM FOR TESTING DRONE MOTOR THRUST

Testing Drone Motor Thrust: -

The total thrust required for flying a drone is given by (4)

Thrust= Thrust-to-Weight-ratio×Total Drone Weight (4)

The thrust of each motor is given by (5)

Thrust per motor = Thrust/Number of motors (5)

Where Thrust–Drone motors are rated for thrust, Total Drone, Weight – Total weight of the drone is a combination of the following different weights,

- Drone weight includes the weight of the frame, motors, propellers, landing gear, etc.
- Battery weight.
- Equipment weight includes any additional equipment such as cameras and lights.

Thrust-to-weight ratio –The thrust-to-weight ratio measures how much thrust is available per unit mass of the drone. This is based on the drone application. For aerial photography.

% Input: Testing Drone Motor Thrust frameWeight = input('Enter the weight of the frame ingrams: '); batteryWeight = input('Enter the weight of the battery ingrams: '); equipmentWeight = input ('Enter weight of equipment in gm:'); thrustToWeightRatio = input('Enter desired thrust to weightratio '); numMotors = input('Enter the number of motors in thedrone: '); % Calculate total drone weight Total Drone Weight = frame Weight + battery Weight + equipment Weight; % Calculate the thrust required for the drone thrust Required = thrust To Weight Ratio * total DroneWeight; % Calculate thrust per motor Thrust Per Motor = thrust Required / num Motors; % Display results Fprintf ('Total Drone Weight: %d grams\n', total DroneWeight); Fprintf ('Thrust Required: %d grams\n', thrust Required); fprintf('Thrust per Motor: %d grams\n', thrust Per Motor);

VI. RESULT AND DISCUSSION

We conducted tests with this program for various drones DJI Matrice 300 RTK Consider the example of DJI Matrice 300 RTK.

Enter the weight of the frame in grams: 3000 Enter the weight of the battery in grams: 3000 Enter the weight of additional equipment in grams: 930 Enter the desired thrust to weight ratio (e.g., 2 for 2:1): 2 Enter the number of motors in the drone: 4 Total Drone Weight: 6930 grams Thrust Required: 13860 grams Thrust per Motor: 3465 grams % Prompting user for required inputs

frameWeight = input('Enter the weight of the frame in kg: '); batteryWeight = input('Enter the weight of the battery in kg:');

payloadWeight = input('Enter weight of the payload in kg: '); capacity = input('Enter the battery capacity in Ah: '); discharge = input('Enter the battery discharge percentage: '); batteryVoltage = input('Enter the battery voltage in volts: '); thrustToWeightRatio = input('Enter desired thrust to weight ratio: ');

numMotors = input('Enter the number of motors in thedrone: ');

kv = input('Enter motor Kv rating (RPM/V): ');

% Calculating drone weight

droneWeight = frameWeight + batteryWeight + payloadWeight;

% Calculating average amp draw (AAD) powerPerKg = 105; % Assuming 105 W/kg

AAD = droneWeight * powerPerKg / batteryVoltage

% Calculating flight time

time = (capacity * discharge / 100) / AAD;

% Calculating thrust

fullThrust = droneWeight * thrustToWeightRatio;motorThrust = fullThrust / numMotors;

% Calculate electrical power electricalPower = batteryVoltage * AAD;

% Convert Kv to SI units (Radians/Second/Volt)

%Calculate motor torque coefficient (Kt)kt = 1 / kv_si;

% Calculate motor torquetorque = kt * AAD;

% Calculate motor RPM

rpm = kv_si * batteryVoltage;

% Calculate mechanical power mechanicalPower = torque * (2 * pi * rpm / 60);

So, the thrust required is 13.8 kg and we have to choose the motor which gives 3.4 kg thrust.

6.1 Test-bench for drone complete program:

By combining MATLAB code of various tests we get the final code for testing the drone following the code and its general output

% Calculate motor efficiency
Motor efficiency = mechanical power / electrical power;
% Calculate propeller efficiency
propeller Efficiency = motor Thrust / mechanical power;
% Calculate powertrain efficiency
Powertrain efficiency = motor thrust / electrical power;
% Displaying results
fprintf('Drone Flight Time: %.2f hours\n', time); fprintf('Motor Thrust: %.2f grams\n', motorThrust); fprintf('Electrical Power: %.2f Watts\n', electricalPower);
fprintf('Mechanical Power: %.2fWatts\n', mechanicalPower);
fprintf('Motor Efficiency: %.2f\n', motorEfficiency); fprintf('Propeller Efficiency: %.2f\n', propellerEfficiency); fprintf('Iorque: %0.4f N.m\n', torque);
fprintf('Motor RPM: %d RPM\n', rpm);

The code prompts the user to input various parameters related to the drone and its components. The code then performs calculations for drone weight, average amp draw, flight time, thrust, electrical power, motor torque, RPM, mechanical power, motor efficiency, propeller efficiency, and power-train efficiency. Finally, the code displays the calculated results using fprintf statements.

VII. CONCLUSIONS AND FUTURE SCOPE

The development of a specialized test bench for battery- motor-propeller assemblies in multirotor drones signifies a significant leap forward in drone technology. This research leverages mathematical modeling and real-world validation to address limitations in traditional multi-rotor propulsion systems, underscoring the pivotal role of batteries, electric motors, and propellers in determining drone performance. The proposed educational multirotor platform contributes to educational outreach initiatives, offering students invaluable hands-on experience in control theory and aerial robotics.

The research's focus on algorithm optimization, integration of adaptive control systems, collaboration with manufacturers, and further exploration of propulsion systems lays a solid foundation for future advancements in drone technology. Future endeavours will prioritize optimizing algorithms for enhanced drone control, integrating adaptive control systems for improved adaptability, and collaborating with manufacturers to incorporate cutting-

edge technologies. Continued research into propulsion systems, particularly variable-pitch propellers, is essential to overcome endurance and range limitations.

Expanding the educational multirotor platform in educational institutions and further developing thrust test benches for comprehensive motor assessment are critical objectives. Additionally, exploring remote sensing applications could broaden the scope of drone technology, contributing significantly to fields such as environmental monitoring and disaster response.

VIII. REFERENCES

- 1. Avanzini, G., Nisio, A. D., Lanzolla, A. M. L., & Stigliano, D. (2020). A test bench for battery-motorpropeller assemblies designed for multirotor vehicles. In 2020 IEEE 7th International Workshop on Metrology for AeroSpace (MetroAeroSpace). IEEE.
- [Şanlıtürk, Mehmet & Bayraktar, Meral. (2019). Designing, and implementation of test bench with PID control for a variable pitch propeller system. In Conference: 4. Uluslararası Mühendislik Mimarlik ve Tasarım KongresiAt: Istanbul, Turkey
- Üçgün, Hakan & Ökten, Irfan & Yüzgeç, Uğur & Kesler, Metin. (2022). Test Platform and Graphical User Interface Design for Vertical Take-Off and Landing Drones. Romanian Journal of Information Science and Technology. 25. 350-367.
- 4. Jatsun, Sergey & Emelyanova, Oksana & Martinez Leon, Andres. (2020). Design of an Experimental Test Bench for a UAV Type Convertiplane. IOP Conference Series: Materials Science and Engineering. 714.
- 5. Kotarski, D., Piljek, P., Pranjić, M., Grlj, C. G., & Kasać,
- 6. J. (2020). A Modular Multirotor Unmanned Aerial Vehicle Design Approach for Development of an Engineering Education Platform. Sensors, 21(8), 2737.
- 7. Czyż, Z., Karpiński, P., Skiba, K., & Wendeker, M. (2021). Wind Tunnel Performance Tests of the Propellers with Different Pitch for the Electric Propulsion System. Sensors, 22(1), 2.
- 8. Avanzini, G., & Giulietti, F. (2013). Maximum Range for Battery-Powered Aircraft. In Journal of Aircraft (Vol.50, Issue 1, pp. 304–307). American Institute of Aeronautics and Astronautics (AIAA).
- Putsep, K., Rassolkin, A., & Vaimann, T. (2021). Conceptual Test Bench for Small Class Unmanned Autonomous Vehicle Performance Estimation. In 2021 IEEE 19th International Power Electronics and Motion Control Conference (PEMC). IEEE.
- Saba Afshaan A, Priyanka G, Manoj N S, Venu N, Ansar Hussain Lone, & Rajashekharareddy H G. (2023). Design and Analysis of UAV Test Bench for Engine/Motor Characterization. In ACS Journal for Science and Engineering (Vol. 3, Issue 1, pp. 21–31). Shanlax International Journals.
- 11. P. Daponte et al., "Drones Bench: an innovative bench to test drones," in IEEE Instrumentation & Measurement Magazine, vol. 20, no. 6, pp. 8-15, December 2017.
- Trihinas, D., Agathocleous, M., Avogian, K., & Katakis, I. (2021). FlockAI: A Testing Suite for ML-Driven Drone Applications. Future Internet, 13(12), 317.
- 13. Islami, Z. S., & Hartono, F. (2019, October). Development of a small propeller test bench system. In IOP Conference Series: Materials Science and Engineering (Vol. 645, No. 1, p. 012017). IOP Publishing.
- 14. Wojtas, Małgorzata, Wyszkowski, Przemysław, Mądro, Mirosław, Osiewicz, Maciej and Kmita, Paweł."Test Stand for Propellers and Rotors in VTOL Drone Systems" Transactions on Aerospace Research, vol.2023, no.1, 2023, pp.67-85.
- 15. [Rohan, A., Rabah, M., Asghar, F., Talha, M., & Kim, S.
- 16. H. (2019). Advanced drone battery charging system. Journal of Electrical Engineering & Technology, 14, 1395-1405.
- 17. Fujimoto, K., Nagai, S., Minh, N. B., & Fujimoto, H.(2023). Test Bench Study on Attitude Estimation in Ground Effect Region Based on Motor Current for In-Flight Inductive Power Transfer of Drones.

- Leite, V. M., & Conceição, A. G. S. (2016, October). Tested prototype of an unmanned aerial vehicle design. In 2016 XIII Latin American Robotics Symposium and IV Brazilian Robotics Symposium (LARS/SBR) (pp. 340-345). IEEE.
- Frigioescu, T. F., Badea, T. A., Condruz, M. R., Cican, G., & Mîndru, I. (2023). Design and Development of a Remote-Control Test Bench for Remote Piloted Aircraft's Brushless Motors. Tehnički vjesnik, 30(4), 1047-1054.
- [1Santos, M. F., Silva, M. F., Vidal, V. F., Honório, L. M., Lopes, V. L. M., Silva, L. A. Z., ... & Regina, B. A. (2018, October). Experimental validation of quadrotors angular stability in a gyroscopic test bench. In 2018 22nd International Conference on System Theory, Control and Computing (ICSTCC) (pp. 783-788). IEEE.
- Adamo, Francesco, et al. "Development and characterization of a measurement instrumentation system for UAV components testing." 2017 IEEE International Workshop on Metrology for Aerospace (MetroAeroSpace). IEEE, 2017.
- Zaharia, S., Pop, M. A., Buican, G. R., Chicos, L., Stamate, V. M., Pascariu, I. S., & Lancea, C. (2022). Design and Testing of Brushless DC Motor Components of A6 Steel Additively Manufactured by Selective Laser Sintering. Aerospace, 10(1), 60.
- 23. Jordan, William & Narsipur, Shreyas & Deters, Robert. (2020). Aerodynamic and Aeroacoustic Performance of Small UAV Propellers in Static Conditions.
- Serbezov, Vladimir & Panayotov, H & Todorov, M & Penchev, S. (2020). Application of multi-axis force/torque sensor system for experimental study of small unmanned aerial vehicles propulsion systems – Preliminary results. IOP Conference Series: Materials Science and Engineering.
- Suarez, A., Jimenez-Cano, A. E., Vega, V. M., Heredia, G., Rodriguez-Castaño, A., & Ollero, A. (2018). Design of a lightweight dual-arm system for aerial manipulation. Mechatronics, 50, 30-44.
- D. Six, S. Briot, J. Erskine and A. Chriette, "Identification of the Propeller Coefficients and Dynamic Parameters of a Hovering Quadrotor From Flight Data," in IEEE Robotics and Automation Letters, vol. 5, no. 2, pp. 1063-1070, April 2020.
- 27. Fuller, Nathan & Theriault, Diane & Kong, Zhaodan & Wang, S. & Betke, M. & Baillieul, John. (2015). Understanding bat flight as a model for bio-inspired aircraft designs. E61- E61.
- 28. Bondyra, Adam & Gardecki, Stanislaw & Gąsior, Przemysław & Giernacki, Wojciech. (2016). Performance of Coaxial Propulsion in Design of Multi- rotor UAVs.
- 29. Alejandro, W., Enrique, L., Antonio, M., & Fernando, L. (2019). Velocity Sensor for Real-Time Backstepping Control of a Multirotor Considering Actuator Dynamics. Sensors, 20(15), 4229.
- 30. Kaspul Anuar, Musthafa Akbar, Hanif Abdul Aziz, & Agung Soegihin. (2022). Experimental Test on Aerodynamic Performance of Propeller and Its Effect on The Flight Performance of Serindit V-2 UAV. In Journal of Advanced Research in Fluid Mechanics and Thermal Sciences (Vol. 91, Issue 2, pp. 120–132). Akademia Baru Publishing.
- Asral, Kaspul Anuar, & Agung Soegihin. (2022). Aerodynamic Analysis of Unnamed Aerial Vehicle Serindit V-2 Using Computational Fluid Dynamics. Journal of Advanced Research in Fluid Mechanics and Thermal Sciences, 93(1), 83–93.
- 32. Tatale, Omkar, et al. "Quadcopter: design, construction and testing." International Journal for Research in Engineering Application & Management 4(2018): 1-7.

IoT-enabled Solar-based EV Charging Station

Dhanashri Dalave¹, Rutuja Sarvgod², Aishwarya Gaikwad³ ¹⁻³Department of Electrical Engineering, SVERI'S College of Engineering Pandharpur ¹dhanashrisdalave@coep.sveri.ac.in ²rutujagsarvagod@coep.sveri.ac.in ³aishwaryaggaikwad@coep.sveri.ac.in

Abstract— The primary goal of the present work is to reduce greenhouse gas emissions and the use of fossil fuels. The paper describes how an E-vehicle module is charged using a solar panel, how an IoT device monitors the maximum power generated by the solar panel, and how an MPPT controller can track the maximum power generated. The complete system is linked to the Arduino Uno, a rechargeable battery that generates and distributes charge, which is displayed on an LCD. This device allows you to charge many automobiles with solar energy by using a solar panel. The GSM modem will send a warning message if the system's power is reduced. To check the status of the battery being charged, a web page is used, which also displays the amount of charge transferred to the charge module and the location of a nearby charging station. This project seeks to develop a smart application that connects to the grid and learns about the various tariff pricing accessible. The program improves low-carbon technology utilizing rule-based algorithms on a single linked platform, uploading data to the Internet of Things (IoT) to indicate battery status and tariff rates.

Keywords— Arduino, Solar panel, LCD display, wifi module, Battery, Internet of Things (IoT), solar energy, electric vehicle (EV), charging infrastructure, sustainability, smart grid, energy management, case studies, future outlook.

I. INTRODUCTION

The widespread adoption of electric vehicles (EVs) worldwide emphasizes the urgent need for robust infrastructure to support their expanding presence [1]. With EVs gaining traction, the demand for efficient charging solutions becomes paramount for both national power grids and charging station network operators. Recognizing this trend, there is a pressing need to develop an intelligent application that seamlessly integrates with the grid, taking into account diverse tariff rates [1]-[11].

This application aims to empower users by delivering real- time information on incoming and outgoing power rates, providing insights into the state of charge (SOC) of the vehicle's battery upon connection to the grid [12]-[21]. To achieve optimal results and promote low carbon technologies, a unified platform employing rule-based algorithms will be implemented.

In this endeavor, the concept of the "Internet of Things" (IoT) plays a pivotal role. IoT involves a network of physical objects, including vehicles, utilities, and home furnishings, equipped with sensors, software packages, and electronic connections [15]. This interconnected system enables seamless communication and data sharing, particularly in the realm of electric vehicle smart charging stations, which play a crucial role in supplying the electric energy necessary for recharging EVs [5]-[10].

The landscape of charging stations is evolving rapidly, with a growing number of widely dispersed, publicly accessible facilities. Some of these stations offer enhanced charging capabilities, facilitating quicker charging at higher voltages and currents beyond what home electricity sources can provide. Moreover, these charging stations feature a variety of heavy- duty connectors that adhere to different specifications for electric charging connectors.

As the transition from combustion engines to electric motors progresses, batteries, as widely adopted energy storage technology, underscore the significance of accurately determining the status of the battery's charge. A key aspect of the intelligent application will be to relay this crucial information to users [18]-[21].

However, challenges such as a shortage of charging outlets could potentially contribute to range anxiety and diminish the practicality of EVs, thereby hindering their widespread adoption. Therefore, a comprehensive approach to infrastructure development is essential to ensure the seamless integration and widespread adoption of electric vehicles, fostering a sustainable and low-carbon future [1]-[21].

Unlimited, clean energy from the sun is harnessed every day, powering everything on Earth through the utilization of photovoltaic panels. These solar panels convert sunlight into electrical energy without emitting harmful CO2, making them environmentally friendly and cost-effective [1].

Solar energy eliminates variable expenses and the need for conventional fuels, further reducing greenhouse gas emissions during electricity generation. The scalability of solar power allows for large-scale electricity production without the risks and costs associated with fuel supply, minimizing air pollution and energy consumption while enhancing overall comfort [10].

This paper delves into the convergence of two groundbreaking technologies—the Internet of Things (IoT) and solar energy—in the realm of Electric Vehicle (EV) charging infrastructure. By examining the potential and challenges of integrating IoT into solar-powered EV charging stations, it provides valuable insights into the advancements, benefits, and future prospects of this innovative approach.

II. RELATED WORK

The paper presents an innovative strategy to enhance the efficiency of current PV solar panels by integrating them into electric vehicle (EV) charging stations. At the Institute of Energy, the solar PV system produces 13,792 kWh/year, allowing for the utilization of 21% (2861 kWh/year) to charge two electric buses, while the surplus energy contributes 9837 kWh/year to the national grid.

To address the intermittent nature of solar energy, the paper proposes implementing an energy storage system. This green transportation concept not only optimizes solar energy usage for charging but also reduces greenhouse gas emissions by 52,944 kg/year, benefiting both the electricity grid network and the environment. With the transport sector accounting for 35% of total CO2 emissions, cleaner energy solutions like electro mobility are imperative. The paper showcases a charging station design integrating solar panels, photovoltaic cells, and batteries.

The photovoltaic system design, optimized using Hybrid Optimization by Genetic Algorithms (iHOGA) software, ensures continuous electricity supply to EVs throughout the day. Integration of Internet of Things (IoT) technology facilitates real-time monitoring of rechargeable battery status, nearby charging station locations, and battery voltage. A cloud-based IoT platform enables multiple users to access and update data, enhancing user convenience.

Acknowledging global growth in solar and EV industries, exemplified by the increasing number of solar rooftops and electric vehicles in the United States, the paper foresees challenges in establishing solar-based EV charging as conventional but recognizes the synergy between EVs and solar energy as offering new business opportunities and environmental sustainability. This anticipated EV-PV synergy is poised to foster cleaner energy solutions and future growth.

The paper conducts comprehensive literature reviews to identify optimization, design, and simulation studies focusing on solar charging systems for EVs, offering a global perspective on energy-generating system implementation. The central theme centres on green solar charging stations for EVs.

Highlighting the environmental drawbacks of non- renewable petroleum resources, the paper emphasizes their significant emissions of harmful pollutants like sulphur dioxide, nitrogen oxides, and carbon dioxide during combustion, contributing to global warming. Reduction of greenhouse gas emissions is thus a primary motivation for the increasing focus on the electrification of mobility.

The paper provides a thorough overview of existing studies, setting the stage for the exploration of green solar charging stations as a sustainable solution for EVs. It underscores the critical need for strategically scheduling

EV charging to mitigate network load escalation, introducing a characteristic demand profile derived from 15minute energy request blocks, based on data from the Indian Power Exchange (IEX), illustrating load variations throughout the day.

III. METHODOLOGY

The items included in the set up are: Arduino UNO R3, power supply, solar panel, WIFI module, current sensor, voltage sensor, two batteries, relay drivers, LCD display, MPPT charger, and an Android application.

Description

• The MPPT controller oversees the solar panel's optimal power output while charging the E-vehicle module.

- The entire configuration is linked to the Arduino UNO R3, which supervises battery levels, manages power generation and distribution, and presents battery status on an LCD screen.
- A web page is employed to monitor the charging status of both the main station battery and the vehicle battery. It also tracks the power transferred from the solar panel to the main battery, which subsequently charges the vehicle battery.
- The LCD display indicates the charging status of both batteries and provides information on the charging module and available station locations.
- The charging station's status is recorded on ThingSpeak and accessed via an Android application. The primary aim of this system is to alleviate greenhouse gas emissions and diminish dependence on fossil fuels.

Components of Proposed Work

Solar panel:

Solar energy, in essence, refers to the energy originating from the sun. This energy manifests in the form of solar radiation, enabling the generation of solar power. The conversion of light into electricity lies at the core of this process. Essential to solar road lighting systems, the solar panel plays a pivotal role in transforming solar energy into electrical power. Utilizing photons from sunlight, solar modules generate energy through the photovoltaic effect. Additionally, solar power harnesses reflectors to retain solar heat. Essentially, a solar panel converts sunlight into either electricity or heat, which can then be utilized to supply power to homes or structures. Typically, solar panels comprise numerous small solar cells distributed across a large area to ensure an ample supply of power. MPPT Charger:

A device referred to as an MPPT Solar Charge Controller, or Maximum Power Point Tracker, functions as an electronic DC-to-DC converter, precisely aligning the solar array (PV panels) with the battery bank for optimal performance. Its primary function involves converting the elevated voltage DC output from solar panels into the lower voltage necessary for battery charging, while simultaneously transforming surplus panel voltage into current, thereby enhancing the overall output of the solar system.

Arduino: The Arduino Uno was selected for its ability to simplify programming complexities and its cost-effectiveness. It accurately measures current and voltage using analog pins, enabling power and energy computations. Powered by the Atmel AVR® core, the Arduino Uno offers features like In- System Programmable Flash, EEPROM, and SRAM, making it versatile for various applications. It also includes multiple I/O lines, Real-Time Counter, Timer/Counters, USART, and Serial Interface, enhancing its functionality. Additionally, it features a 10-bit ADC, Watchdog Timer,

SPI serial port, and power- saving modes for efficiency.

Wifi-module (ESP8266EX):

The ESP8266EX is specifically tailored for mobile, wearable electronics, and Internet of Things applications, prioritizing minimal power consumption through a blend of proprietary techniques. Its power-saving architecture primarily operates in three modes: active mode, sleep mode, and deep sleep mode. Through advanced power management methods and logic, it selectively powers down unnecessary functions and orchestrates the transition between sleep and active modes.

Relay:

In our project, a relay is employed to manage power supply, intervening if faults arise in the solar power supply. Voltage Divider:

A voltage divider, also referred to as a potential divider, is a passive linear circuit generating an output voltage (Vout) that is a fraction of its input voltage (Vin). Voltage division occurs as the input voltage is distributed among the divider's components.

Current Sensor:

A current sensor detects electric current within a wire, generating a signal that corresponds to the current detected. This signal can be presented as an analog voltage or current, or even as a digital output. It can be used to display the measured current on an ammeter, saved for later analysis in a data acquisition system, or utilized for control functions.

LCD Display:

A liquid-crystal display (LCD) is a flat panel display that utilizes the light-modulating properties of liquid crystals to create electronic visual displays. The designation "20x4" indicates that the LCD can display 20 characters in each of its rows, allowing a total of 80 characters to be displayed simultaneously.

DC Voltage regulation:

The final component in a regulated DC power supply is DC voltage regulation. Variations in input from the AC mains, alterations in load current at the output, or shifts in temperature can cause fluctuations in output voltage or current. To mitigate this issue, a regulator is employed to maintain a consistent output despite changes in input or other factors. Depending on the specific application, transistor series regulators, fixed and variable IC regulators, or Zener diodes operated in the Zener region can be utilized. Integrated circuits such as the 7812 and 7905 are commonly employed to achieve fixed output voltage values, with the 7812 providing a constant 12V DC output and the 7805 delivering a steady 5V DC output. The output from the 7805 is directed to the Arduino LCD and all required sensors, where sufficient power is necessary. Arduino, known for its user-friendly programmable circuitry, open-source hardware and software, is robust and capable of effectively supporting various devices. It is based on the ATmega328 microcontroller, featuring 14 digital I/O pins, 6 analog outputs, a USB interface, an ICSP connector, a power jack, and a reset switch, facilitating connectivity with a laptop via USB or an AC-DC power source.

The ESP8266 serves as a straightforward and cost- effective tool for network applications. Functioning independently, it can operate as an Access Point (displaying a hotspot) or a Station (connecting to Wi-Fi), facilitating the download and transfer of data to the web, thus simplifying the creation of a Web of Things ecosystem. It operates on 3.3V exclusively, as higher voltages, such as 3.7V, can potentially damage the module when interfaced with circuits.

Buzzer:

A buzzer or beeper serves as an audio signalling device, which may be mechanical, electromechanical, or piezoelectric. Common applications include alarm systems, timers, and confirmation of user input such as mouse clicks or keystrokes.

Current Transformer:

A Current Transformer (CT) is utilized to measure the current in another circuit. These are commonly used globally to monitor high-voltage lines across national power grids. The CT produces an alternating current in its secondary winding, which is proportional to the current in its primary winding. In the context of EV charging using solar power, Arduino reads the current value and total units, displaying them on an LCD and transmitting the data to a web server. The ThingSpeak server is utilized to monitor the data graphically and numerically. A Wi- Fi network is necessary for transmitting the data wirelessly from the transmitter to the receiver, with the ESP8266 Wi-Fi Module enabling access to the Wi-Fi network for vehicles. Optimistic utilization of electric vehicles is essential for numerous applications.



Fig.1: Schematic block diagram for IOT-enabled solar-based electric vehicle charging station prototype



Fig.2: Hardware set-up for IoT-enabled solar-based EV charging station

IV. RESULTS

The Internet of Things (IoT) enables monitoring of battery status, effectively transforming it into an energy storage system. Utilizing a cloud platform for management, this IoT implementation allows users to track nearby charging stations and observe fluctuations in voltage supply. Data stored within the Arduino remains accessible until the battery ceases to charge. Additionally, user profiles for electric vehicle (EV) clients who utilize the stations are stored and regularly updated in the database, ensuring efficient distribution among various users. The quest for an eco-friendly and sustainable environment hinge upon our collective actions, particularly in terms of vehicular usage and its impact on pollution levels. Optimal utilization of solar energy involves selecting efficient components capable of generating the requisite energy to charge EVs, thus contributing to a greener environment.

V. CONCLUSIONS

The study highlights the crucial integration of IoT technology with solar energy systems in EV charging infrastructure, emphasizing the need for ongoing research, innovation, and collaboration to advance intelligent and sustainable transportation solutions. Through IoT connectivity, users can efficiently manage power by monitoring battery status and reallocating surplus charge as necessary. The project aims to simplify the establishment of EV charging stations, enhancing accessibility and maintenance for residential use. With an expected surge in plug-in EVs on the roads by 2030, the addition of fast-charging stations could greatly facilitate long- distance travel between two cities.

V. REFERENCES

- 1. Khadka, S., et al. "Design and implementation of a solar-powered electric vehicle charging station." International Journal of Electrical and Computer Engineering 8.1 (2018): 449-457.
- 2. Tripathy, S., et al. "IoT based smart solar-powered electric vehicle charging station." Proceedings of the International Conference on Electronics, Communication and AerospaceTechnology. IEEE, 2019.
- Amali, R., and S. Siva. "Design and implementation of IoT based solar charging station for electric vehicles." 2019 International Conference on Smart Electronics and Communication (ICOSEC). IEEE, 2019
- Ravi, R., et al. "Internet of things based solar-powered electric vehicle charging station."2017 International Conference on Innovations in Information, Embedded and Communication Systems (ICIIECS). IEEE, 2017.
- 5. Huang, T., et al. "Development of a solar-powered electric vehicle charging station using internet of things technology." Energies 13.13 (2020): 3261.
- Bhandari, S., et al. "IoT based Smart Electric Vehicle Charging Station with Renewable Energy Source." 2021 IEEE 11th International Conference on Intelligent Systems and Control (ISCO). IEEE, 2021.
- Kanchan, A., and S. Agarwal. "IoT enabled solar powered electric vehicle charging station with bidirectional charger." 2020 4th International Conference on Intelligent Computing and Control Systems (ICICCS). IEEE, 2020.
- 8. Srivastava, A., et al. "IoT-based intelligent solar powered electric vehicle charging station."2020 2nd International Conference on Inventive Research in Computing Applications (ICIRCA). IEEE, 2020.
- Srivastava, R., et al. "Design and implementation of IoT-based solar- powered electric vehicle charging station." 2019 4th International Conference on Internet of Things: Smart Innovation and Usages (IoT-SIU). IEEE, 2019.
- Shrivastava, V., et al. "IoT based Solar Powered Electric Vehicle Charging Station."2021 IEEE International Conference on Power Electronics, Intelligent Control and Energy Systems (ICPEICES). IEEE, 2021.
- Kim, Ho-Sung, Myung-Hyo Ryu, Ju-Won Baek, and Jee-Hoon Jung."High-efficiency isolated bidirectional ACDC converter for a DC distribution system." IEEE Transactions on Power Electronics 28, no. 4 (2013): 1642-1654.
- Suresh, Harishankar, Anand Baskaran, K. P. Sudharsan, U. Vignesh, T. Viveknath, P. Sivraj, and K. Vijith." Efficient charging of battery and production of power from solar energy." In Embedded Systems (ICES), 2014 International Conference on, pp. 231-237. IEEE, 2014.
- Prabhakar, Bhavya M., J. Ramprabhakar, and V. Sailaja." Estimation and controlling the state of charge in battery augmented photovoltaic system."In Power and Energy Systems: Towards Sustainable Energy (PESTSE),2016 Biennial International Conference on, pp. 1-6. IEEE, 2016.
- 14. Li, Wen, and Sami Kara. "Methodology for Monitoring Manufacturing Environment by Using Wireless Sensor Networks (WSN) and the Internet of Things (IoT)." Procedia CIRP 61 (2017): 323-328.

- Li, Siguang, and Chengning Zhang. "Study on battery management system and lithium-ion battery." In Computer and Automation Engineering, 2009. ICCAE'09. International Conference on, pp. 218-222. IEEE, 2009.
- Cheng, Ka Wai Eric, B. P. Divakar, Hongjie Wu, Kai Ding, and Ho Fai Ho." Battery-management system (BMS) and SOC development for electrical vehicles." IEEE transactions on vehicular technology 60, no. 1 (2011): 76-88.
- 17. Prashanth, K. I., Dinesh Rao, and Aravind Sharma." Estimation of SOC and SOH of Li-Ion Batteries." International Journal of Computer Applications 104, no. 4 (2014).
- Friansa, Koko, Irsyad Nashirul Haq, Bening Maria Santi, Deddy Kurniadi, Edi Leksono, and Brian Yuliarto." Development of Battery Monitoring System in Smart Microgrid Based on Internet of Things (IoT)." Procedia engineering 170 (2017): 482-487.
- Prabhdip Singh Rayat, Technicity Project Report,"Defining Electric Vehicle Charging Infrastructure for smart cities using IoT.Page Defining Electric Vehicle Charging Infrastructure for Smart Cities using IOT and Smart Sensors Technicity", 2015.
- Khajenasiri, Iman, Abouzar Estebsari, Marian Verhelst, and Georges Gielen. "A review on Internet of Things solutions for intelligent energy control in buildings for smart city applications." Energy Procedia 111(2017): 770-779.
- 21. Tan, Yee Shee, Yen Ting Ng, and Jonathan Sze Choong Low." Internet- ofthings enabled real-time monitoring of energy efficiency on manufacturing shop floors." Procedia CIRP 61 (2017): 376-381.

Pick and Place Robotic Vehicle with Soft Catching Gripper

G.D.Pawar¹, T.P.Suryawanshi², R.S.Nikam³, D.D.Daphale⁴ ¹⁻³ Department of Electrical Engineering, SVERI's College of Engineering, Pandharpur, MH- India ¹govinddpawar@coep.sveri.ac.in ²tanmaypsuryawanshi@coep.sveri.ac.in ³rohitsnikam@coep.sveri.ac.in

Abstract— This paper outlines the design analysis of an Android-controlled "Pick and Place" Robotic vehicle. Operated through an Android application for remote control, the robotic arm executes commands transmitted to the receiver, enabling movement in different directions. The vehicle integrates four motors linked to the microcontroller: two for arm and gripper actions, and two for overall mobility. The project aims to develop a Bluetooth-driven robotic arm to relocate items, supporting various technological sectors. Using an Android device as a transmitter, it acts as a remote control, while the Bluetooth device at the receiver connects to the microcontroller, controlling DC motors via a motor driver IC for remote operation via smartphones or tablets.

I. INTRODUCTION

The project aims to develop a pick-and-place robotic vehicle equipped with a soft-catching gripper. This endeavor seeks to create a mobile robot capable of transporting objects from one location to another, leveraging an Android app for control. With modern industries increasingly relying on computer-based interactions and robotics to enhance productivity and maintain competitive advantage, this project aligns with the trend toward automation.

The proposed robot is designed to execute physical tasks and can operate in diverse environments, including hazardous conditions and military operations. With applications ranging from underwater tasks to assisting disabled individuals, the versatility of this robot extends to various scenarios. The vehicle is powered by four motors controlled by a microcontroller, with two motors dedicated to arm and gripper movement and the remaining two for body movement.

Utilizing an Android application device as a transmitter, commands are relayed to the robot's receiver to direct its movements—forward, backward, left, or right. At the receiving end, the microcontroller interfaces with four motors, enabling precise control over the robot's actions. Remote operation is facilitated via smartphones or tablets running the Android OS, offering intuitive touch-screen control through a graphical user interface (GUI). Notably, the robot's soft-catching arm prioritizes object safety by minimizing pressure during gripping.

The importance of autonomous vehicles in modern society cannot be overstated, as they undertake tasks that may be impractical or unsafe for humans. In industries where manual labor is laborious or repetitive, autonomous vehicles enhance efficiency while reducing reliance on human resources. Furthermore, the integration of line-following capabilities into the pick-and- place robot offers a solution to navigational challenges, ensuring uninterrupted operation along predetermined paths. As industries increasingly embrace automation to streamline processes and cut costs, the development of the pick-and-place arm with a line follower robot represents a significant advancement. These robots, equipped with Bluetooth-enabled control systems and object detection capabilities, are instrumental in manufacturing plants, where they navigate predefined routes to retrieve and deposit components accurately.

In summary, the integration of pick-and-place functionality with line-following capabilities in this robotic system represents a promising innovation in industrial automation, offering improved efficiency and reliability in material handling tasks.

II. LITERATURE REVIEW

Robots are typically controlled through tethered (wired), wireless, or autonomous means, with traditional wireless methods including mobile phones, joysticks, keypads, and computer terminals. They can even be interfaced

with the internet for remote control from anywhere. Here are some commonly used control methods for both robotic vehicles and arms.

The fundamental operation of a pick and place robot relies on its joints, which are akin to human joints and connect consecutive rigid bodies within the robot. These joints can be rotational or linear, and adding a joint to any part of a robot requires understanding its degrees of freedom and movement. Degrees of freedom govern the body's linear and rotational motion, while degrees of movement indicate the number of axes it can traverse. Equipped with an Atmega328 microcontroller IC, Bluetooth module, and four DC motors with driver IC, the pick and place robotic arm is affixed to a moving vehicle, allowing it to navigate various surfaces. This setup employs two motors for vehicle movement and an additional two for pick and place operations. The arm comprises an assembly with a jaw capable of vertical motion, driven by one motor, and jaw opening and closing, controlled by another.

III. METHODOLOGY

Robots can be controlled through wired, wireless, or autonomous means. Traditional wireless control methods include mobile phones, joysticks, keypads, computer terminals, and internet interfacing, enabling remote control from anywhere. Various control methods are employed for both robotic vehicles and robotic arms. The fundamental operation of a pick-and-place robot is facilitated by its joints, which serve as connections between consecutive rigid bodies within the robot. These joints can be rotational or linear, depending on the specific application. Adding a joint to any part of the robot requires determining its degrees of freedom and degrees of movement. Degrees of freedom define the linear and rotational motion capabilities of the body, while degrees of



BLOCK DIAGRAM

movement indicate the number of axes along which the body can move. Fig.1 Block Diagram

Equipped with an Atmega328 microcontroller IC, Bluetooth Module, four DC motors with driver IC, and power supply, the pick-and-place robotic arm is mounted on a mobile vehicle, enabling it to traverse various surfaces with ease. The system utilizes two motors for vehicle movement and two additional motors for pick- and-place operations. A belt-type tire is attached to the vehicle for smooth operation.

The pick-and-place arm assembly comprises an arm with a gripper, capable of vertical motion. Two motors are employed—one for vertical movement and another for opening and closing the gripper. This configuration enables precise positioning and manipulation of objects during pick-and-place operations. To begin, locate and install the Blue Control application from Google Play on your Android device. Once installed, open the application from your list of applications. Activate Bluetooth on your device and initiate a scan for the device named "HC-05." Connect to this device by entering the PIN "1234." Next, access the Blue Control app and tap the options button, then select the "connect" option. A list of available devices will appear; locate "HC- 05" and again enter the PIN "1234," then click "OK." Wait for confirmation that the connection has been established ("Connected to HC-05"), indicating that the application is now ready to communicate with the Bluetooth module in the pick-and-place robotic vehicle.

Below are the operational steps:

- Press the up button for robot forward movement.
- Press the down button for robot backward movement.
- Press the left button for robot leftward movement.
- Press the right button for robot rightward movement.
- Press the A button to open the robot gripper.
- Press the B button to close the robot gripper.
- Press the C button to move the robot arm upward.

The pick-and-place robotic vehicle is equipped with an Atmega16 microcontroller IC, a Bluetooth module, four DC motors with driver ICs, and a power supply. The robotic arm is mounted on a mobile vehicle, allowing it to traverse various surfaces regardless of their texture. The vehicle's movement is facilitated by two motors, and a belt-type tire ensures smooth and reliable operation, akin to those found in tanks.

ARDUINO UNO



Figure 2: Arduino Uno

Arduino is a pioneering open-source hardware and software endeavor, fostering the development of singleboard microcontrollers and kits for constructing digital devices and interactive objects. Under GNU licenses, Arduino boards offer versatile microprocessors and controllers, supporting digital and analog I/O pins for various expansions. Its user-friendly programming environment, established in 2003, democratizes device creation. Arduino operates openly, with hardware designs accessible under a Creative Commons license, although the name "Arduino" is reserved for official products. Its boards, like the Uno, facilitate program loading via USB, while variants like the Mini utilize detachable adapters or Bluetooth for connectivity. These boards typically expose numerous I/O pins for external circuitry, often augmented by plug-in shields.

ROBOTICS ARM



Figure 3: Robotic Arm

A gripper, an essential component of robot vehicles, serves to seize, secure, or manipulate objects, finding application in material handling, pick-and-place tasks, and search and rescue operations. Grippers vary in type, including parallel, angular, or vacuum-based, each tailored to specific tasks and object shapes, enabling effective interaction with the environment. A soft catching gripper, a subtype designed for delicate objects, boasts features like compliance, object protection, sensing, and control, versatility, and suitability for collaborative robotic environments. Commonly employed in industries like food handling and electronics manufacturing, soft catching grippers exemplify the broader field of soft robotics, emphasizing gentle interaction with the environment.

Bluetooth Module HC-05



Figure 4: Bluetooth Module HC-05

The bluetooth module is a user-friendly Bluetooth SPP (Serial Port Protocol) device, facilitating seamless wireless serial connections. It operates on Bluetooth V2.0+EDR with a 3Mbps data rate and features a compact

footprint of 12.7mmx27mm. Utilizing CSR Blue core 04 technology and AFH, it ensures efficient communication with CMOS technology. The module supports AT commands for configuring settings, typically controlled via a microcontroller like Arduino. Designing a Bluetooth-enabled pick and place robotic vehicle with a soft-catching gripper involves integrating components such as microcontrollers, motor drivers, Bluetooth modules like HC-05, and navigation sensors. Gripper design, possibly using servo motors or actuators, depends on soft-catching requirements. Microcontrollers must be programmed to receive Bluetooth commands for movement and gripping actions, considering power supply and safety features for the soft-catching mechanism.

• SOFTWARE REQUIREMENT: (MIT APP INVERTOR)



Figure 5: MIT App Inverter

During app development, it is recommended to conduct testing simultaneously with the development process. This can be easily accomplished with App Inventor, thanks to the live connection provided between your phone (or emulator) and the App Inventor development environment. By default, App Inventor is set up so that the screen of your app is "scrollable," thereby enabling the user interface to extend beyond the screen's limit, allowing users to scroll down by swiping their finger, akin to scrolling on a web page. When a Canvas is utilized, the "Scrollable" setting should be deactivated (UNCHECKED) to prevent screen scrolling. This action allows the Canvas to occupy the entire screen, ensuring a seamless user experience.

IV. SCOPE OF THIS PROJECT

"Pick and Place Robotic Vehicle with Soft Catching Gripper" project aims to transform material handling across industries by introducing an innovative mobile robotic system. The key goal is to develop a versatile robotic platform equipped with an advanced soft catching gripper, capable of delicately handling various objects to minimize damage. Unlike conventional rigid grippers, this technology ensures the safe manipulation of fragile items during picking and placing operations. The system boasts a robust chassis for navigating diverse environments, enhanced with sensors and computer vision for autonomous object detection and assessment. The gripper's compliance with different object shapes and materials ensures a gentle yet secure grip, promoting reliable and damage-free handling. Moreover, the project focuses on creating a user-friendly interface for seamless control and monitoring, coupled with smart technology integration for remote operation and real-time feedback, thus improving overall system efficiency.

V. RESULT

The Pick and Place Robotic Vehicle with a soft catching gripper is an advanced system designed to autonomously handle objects in diverse environments. Combining mobility with a specialized gripper, it efficiently picks up and places objects while minimizing damage risk. Equipped with sensors and cameras, it navigates complex spaces, avoiding obstacles and identifying targets. The soft catching gripper allows gentle handling of delicate or irregular objects, crucial for various applications. Advanced algorithms enable real-time decision-making, adapting to dynamic environments. Its versatile applications include manufacturing, warehouse logistics, and healthcare, promising improved efficiency and safe object handling. Overall, it represents a cutting-edge solution with mobility, sensing, and manipulation capabilities poised to revolutionize industries.

VI. CONCLUSION

The Pick and Place Robotic Vehicle, featuring a soft-catching gripper, marks a significant leap forward in automation, especially for material handling and logistics. Its incorporation of the soft-catching gripper enhances versatility and safety, enabling delicate object manipulation, crucial for industries like electronics and pharmaceuticals.

This vehicle boasts adaptability to diverse environments, with the gripper ensuring secure yet gentle handling, reducing damage risks. Enhanced mobility and navigation make it suitable for dynamic workspaces, bolstering efficiency in flexible tasks.

Advanced sensors and computer vision tech bolster the vehicle's perception and interaction abilities, facilitating intelligent navigation and accurate object detection. Machine learning algorithms further refine decisionmaking, enhancing performance over time. Operationally, the vehicle streamlines workflows, diminishing the need for human intervention in routine tasks, reducing errors, and boosting productivity and cost-effectiveness.

In essence, the Pick and Place Robotic Vehicle with a soft-catching gripper is a groundbreaking solution catering to industries' evolving needs for precise material handling. Its integration of soft gripping tech, advanced sensors, and autonomous navigation positions it as a valuable asset in modern manufacturing and logistics, promising seamless automation for intricate tasks, ensuring both efficiency and precision.

VII. REFERENCES

- 1. John Iovine., "Robots, Androids, and Animations 12 Incredible Projects You Can Build", Second Edition, McGraw-Hill.2002.
- In Proceeding of the IEEE International conference on robotics and biomimetic (ROBIO) Shenzhen, china, December 2013.
- Mohamed Naufal bin Omar, "Pick and place robotic arm controlled by Computer", Universititeknikal Malaysia, Melaka April 2007.
- 4. Yanjianghuang, ryosukechiba, tamioarai, tsuyoshiueyama and junota. ,"Integrateddesign of multi-robot system for pick-and-place tasks".
- 5. Pneumatic four axis materials handling robot. Author: Royston CLEATUS Mascarenhas. International Journal of Scientific and Engineering Research October 2019.
- Purposes of work is to manufacture a light weight robot arm with a low cost budget. Author:S.C.Gutierrez, R.Zotovic, M.D.Navarra, M.D.Meseguer. Manufacturings Engineering Society International Conference 2017, MESIC 2017, 28-30 June 2017, Vigo (Pontevedra), Spain.
- Design and implement a four DOF pick and place robotic arm. Author: Ravikumar Mourya, Amit Shelke, Saurabh Satpuite, Sushant Kakade, Monoj Botre.International Journal of Recent Research in Civil and Mechanical Engineering (IJRRCME) January 2015.

- Design and Fabrication of Multipurpose Sowing Machine. Author: V.M. Martin Vimal1, A. Madesh1, S.Karthick1, A.Kannan2. International Journal of Scientific Engineerings and Applied Science (IJSEAS) - Volume-1, Issue-5, August 2018.
- 9. Design and Development of Template Row Planter. Author: Adisa A F, Braide F. G. Transnational Journal of Science and Technology, Septembers 23 2019.
- 10. CORTEX: A 3DOF Robotic Arm Controller Software. Author: Nazib Abrar, Raihanul Haque Rahi. International Journal of Emerging Trends & Technology in Computer Science August 15 2022.
- A bluetooths controlled robotics arm with trainable feature. Author: Aditi Baura1, Dr. Md. Tazul Islam2, Md. Aidid Alam3. Islam2, International Conference on Mechanical Engineerings and Renewable Energy December 2021.
- 12. Implementation of smart phones environment. Author: Sungwookmoon ,youngjinkim, hojunmyeong , changsookim, namjucha,and dong hwankim 2019 8th International Conference on Ubiquitous Robots and Ambient Intelligence (URAI).
- 13. Integrated design of multi-robot system for pick and-place tasks. Authors: YanjiangHuang Ryosuke Chiba .IEEE International Conference on Robotics and Biomimetics (ROBIO) 2018.
- 14. Range-based navigation system for a mobile Robot- Neil MacMillan, River Allen, DimitriMarinakis, Sue Whitesides, IEEE 2010.
- 15. Smartphone-based Mobile Robot Navigation Nolan Hergert, William Keyes, and Chao Wang, spring 2012.
- Development Of A Wireless Device Control Based Mobile Robot Navigation System PheySiaKwek,Zhan Wei Siew, Chen How Wong,, BihLiiChua, Kenneth Tze Kin Teo.IEEE 2012.
- 17. A Bluetooth-based Architecture for Android Communication with an Articulated Robot- Sebastian van Delden and Andrew Whigham, IEEE 2013.
- 18. R K Mittal and I J Nagarath "Robotics and Control" BITS Pilani, 2003.
- 19. Atmel AVR Microcontroller Primer: Programming and Interfacing by:- Steven F.Barret, Daniel J. Pack.
- 20. [Richard E. Pattis. Karel the Robot: A Gentle Introduction to the Art of Programming. John Wiley &Sons, 1981.
- Gyara Sridhar 1, Potharaju Raju 2, P Pooja3 and Y Sudha4. "Bluetooth Controlled Robot using ARDUINO". International Journal of Research and Advanced Development (IJRAD), ISSN: 2581-4451.
- Jagruti Chaudhari, Asmita Desai and S. Gavaskar, "Line Following Robot Using Arduino for Hospitals". 2nd International Conference on Intelligent Communication and Computational Techniques (ICCT) Manipal University Jaipur, Sep 28-29, 2019.
- 23. Ritika Pahuja, Narender Kumar, "Android Mobile Phone Controlled Bluetooth Robot Using 8051 Microcontroller", ISSN (Online): Pg:2347 3878.
- 24. K. Kannan, Dr. J. Selvakumar, "Arduino Based BLUETOOTH CONTROLLED ROBOTIC CAR", Volume:02 issue :01, Mar-2015 ISSN: 2395-0072.
- 25. [Mrumal.K. Pathak, Javed Khan, "Robot control design using android smartphone ",2 feb 2015, issn :2347-5471.

Solar Powered Wireless Charging System

Mahek Patil¹, Monika Salunkhe², Karthika Nagane³, ,Ranjana. N. Khandebharad⁴, Dr. Dipti. A. Tamboli⁵

¹⁻⁵ Department of Electrical Engineering, SVERIs College of Engineering, Pandharpur, India

Abstract— one kind of charging technology is called wireless charging, which transfers energy through electromagnetic induction using an electromagnetic field. A method called mutual induction is used to transfer energy between devices (the transmitter and receiver). To charge the battery in the receiving inductive coil, solar energy is supplied into a transmitter inductive coil, which converts it into an electric current. An AT mega controller, copper coils, a half-wave rectifier circuit, a solar panel, a battery, and an LCD display are used in the development of the system. The design demonstrates how to use a charge controller in conjunction with a solar panel to power the battery. The battery receives and holds DC power.

Keywords: Arduino, solar panel, inductive coil

I. INTRODUCTION

Wireless charging, also known as wireless power transfer, is a groundbreaking technology that enables the transfer of power to a device without the need for physical cords. By combining wireless charging with solar power banks, a unique product is created that merges solar charging with efficient battery support and wireless charging capabilities. This integration empowers the device to autonomously charge itself during daylight hours, guaranteeing uninterrupted power for users. In the realm of wireless charging, it is often referred to as inductive charging, which eliminates the need for charging cables, thereby reducing wear and tear on hardware ports. This technology has gained traction, with Samsung introducing wireless charging in its Galaxy S6 mobiles back in 2015. Compared to wire charging, wireless charging offers several advantages.

One notable advantage lies in its user-friendliness, as it eliminates the need for cables, allowing different mobile devices to utilize the same charging pad. Additionally, wireless charging enhances product durability by making devices waterproof and dustproof. The flexibility it provides is another advantage, especially in situations where connecting cables for charging can be costly. Moreover, it boasts the advantage of being radiation-free.

The development of wireless charging technology is progressing through two main avenues: radiative wireless charging, also referred to as radio frequency (RF)-based wireless charging, and inductive charging, also known as coupling-based wireless charging. Radiative wireless charging utilizes electromagnetic waves such as microwaves and RF waves to transmit energy through radiation. However, due to safety concerns related to RF exposure, this method operates within a low-power limit. On the other hand, inductive charging relies on mutual induction, where a magnetic field connects two coils. Although the power transfer distance is limited due to the rapid weakening of the magnetic field, inductive charging is widely accepted for everyday use due to its safety and convenience. Mobile phones have become indispensable in our daily lives, but the inconvenience of frequent charging persists. The promising solution offered by wireless charging technology is taken to the next level in this project by incorporating solar power. Through the utilization of cutting-edge components and pioneering technologies, this endeavor addresses the growing demand for convenient, sustainable, and user-friendly mobile phone charging.

II. LITERATURE REVIEW

The concept of wireless charging, as proposed by Abhijith Nidmar et al. (2019), involves the use of electromagnetic fields to transmit energy through electromagnetic induction. This technology enables the transfer of energy between a transmitter and receiver device. In the case of solar power banks with wireless charging, as

explored by Sanju Tandan et al. (2023), it has become a popular and environmentally responsible option for charging mobile devices. However, there are certain drawbacks associated with these devices, such as their reliance on sunlight, slower charging rates, limited capacity, and inconsistent performance in indoor or shaded environments [1].

This literature review, covering studies conducted from 2017 to 2022, suggests that while consumers generally have positive attitudes towards these gadgets, further research is needed to improve their performance and usability. Factors like sunlight intensity, temperature, and device compatibility can impact the efficiency and speed of charging. Understanding these advantages and limitations can help users make informed decisions about the suitability of a solar power bank with wireless charging [2].

Rahil Imtiyaz and colleagues (2021) introduced the concept of wireless charging, a method that utilizes an electromagnetic field to transmit electricity through electromagnetic induction. This process of mutual inductance enables the wireless exchange of electricity between a transmitter and a receiver. The transmitter coil converts solar energy into AC, which is then wirelessly transferred to the receiver coil and subsequently to the battery. Currently, the efficiency of this wireless charging method stands at 11%, indicating room for improvement in the future. This technology holds potential for supplying power to electric vehicles and various business equipment.[3]

Professor Sara Anjum (2022) investigated how electricity is transported across devices (sender and receiver) using the same input procedure. Solar power is sent into the input of the inductive transmitter coil, while the inductive receiving coil receives it and turns it into electrical power to charge the battery. However, there are several hurdles to implementing wireless charging with current products. In this publication, a foundation for wireless charging technologies and the most recent improvements are described [4].

Vasava Dhrumil and et al., (2022) showed that energy may be obtained in a variety of ways, including unique herbal resources, solar power, nuclear power, and chemical fuel strength. The document provides wi-fi charging solutions for electric vehicles that make use of solar energy. The petrol used in today's vehicles creates noise and air pollutants, which have a significant environmental impact. However, the wi-fi charging period outweighs these pollution issues. Wireless Power Transmission [WPT] represents a totally dependable, efficient, noiseless, and pollution-free age [5].

Prof. Rupali Tirale, along with colleagues (2022), concentrated on the main goal of their research, which is to create a solar power bank integrated with an inverter system that can generate a 230V AC output. The solar power bank technology streamlines the process of charging mobile phones by utilizing a wireless power transmission coil and a minimal number of circuit components. The central idea revolves around wireless mobile charging, which employs electromagnetic induction to transfer energy through an electromagnetic field. Energy is transferred between devices through this induction process.

III. PROPOSED WORK

The proposed system incorporates a solar panel, battery, half-wave rectifier circuit, primary and secondary copper coils, ATmega controller, and LCD. This innovative setup aims to showcase the wireless charging of mobile can be inconvenient during travel or outdoor activities. Therefore, the objective is to develop an intelligent solar-powered power bank with AC outputs, allowing for wireless phone charging or charging through an AC charger [6]. The solar panel, which is regulated by a charge controller, generates power to charge the battery, which stores DC power. In order to transmit this DC electricity wirelessly, it is converted to AC through a transformer. The resulting alternating current is then regulated by circuitry. This regulated AC power is used to energize the copper coils responsible for wireless energy transmission.

An additional copper coil is positioned beneath the electric vehicle. As the vehicle passes over this coil, energy is transferred from the transmitter coil to the EV coil. It is important to note that the energy is initially induced into this coil as DC current. To make it suitable for charging the EV battery, it is converted back to DC using an AC-to-DC conversion circuitry.

The Atmega microcontroller monitors the input voltage and provides real-time feedback on the charging process by displaying the results on an LCD. This technology demonstrates a solar-powered wireless charging system for electric vehicles, offering a potential solution that could be integrated into roadways [7]. Santosh E. and et al., (2023) introduced the system that uses a solar panel, battery, transformer, regulator circuits, copper coils, AC to DC converter, Atmega controller, and LCD display to construct the system. The technique reveals how to charge electric automobiles while driving, removing the need to stop for charging. The solar panel powers the battery via a charge controller. The battery is charged and holds dc power. To transmit the DC electricity, it must first be converted to AC. In this case, we employ a transformer. The transformer converts the electricity to alternating current, which is then regulated by regulator circuitry. This electricity is now utilised to power the copper coils that transmit wireless energy. A copper coil is also installed below the electric car. When the car passes over the coil, energy is transferred from the transmitter coil to the EV coil. Please keep in mind that the energy is still being induced into this coil as DC current. Now we convert this back to DC so it may be utilised to charge the EV battery. We employ an AC-to-DC conversion circuitry to convert it back to DC. We now use an Atmega microcontroller to measure the input voltage and display the results on an LCD. Thus, the technology illustrates a solar-powered wireless charging system for electric vehicles that can be implemented into roadways [7]. Phones using solar energy, the solar panel, equipped with a charge controller, supplies power to charge the battery. Once charged, the battery stores DC power, which needs conversion to AC for wireless transmission.

To accomplish this conversion, a Full Wave Rectifier is employed. Unlike a half-wave rectifier, which only utilizes one half of each alternating wave cycle, a full-wave rectifier converts both halves, resulting in a smoother and more steady DC signal. This is crucial for efficient power transfer, as a significant power loss occurs in the half-wave rectifier. Santosh E. et al. (2023) introduced a comprehensive system that includes various components such as a solar panel, battery, transformer, regulator circuits, copper coils, AC to DC converter, Atmega controller, and LCD display. The main objective of this system is to enable the charging of electric vehicles while they are in motion, eliminating the need for frequent stops for charging.

The power generated in DC form is then utilized to energize copper coils responsible for transmitting wireless energy through inductive coupling. The system is divided into two main components: the transmitter side and the receiver side. Furthermore, a solar system is integrated into the setup, providing a DC power supply to charge the battery. The DC power from the solar system is converted into AC to ensure efficient wireless power transmission. The transfer of wireless power relies on the mutual inductance between the transmitter and receiver loops of current-carrying conductors, which creates a magnetic field. The AC power generated by the full-wave rectifier is then transmitted to the transmitter. Through mutual inductance, the receiver captures the AC power and converts it back to DC using a bridge rectifier. The output of the bridge rectifier is unregulated direct current, which is subsequently regulated by a voltage controller to maintain a constant voltage.



Fig 1. Block Diagram of the Proposed Model.

The regulated output voltage is utilized to recharge the batteries of low-power devices such as portable iPods and smartphones. The entire system illustrates an eco-friendly and convenient approach to mobile phone charging, utilizing solar energy and wireless power transmission technology.



Fig 2. Circuit Diagram of proposed Model.

a) Solar Panel:

A solar panel, also referred to as a solar cell panel, solar electric panel, photo-voltaic (PV) module, or simply solar panel, consists of photo-voltaic cells arranged within a framework specifically created for installation. Solar panels harness sunlight as their source of energy to generate direct current (DC) electricity. A collection of PV modules is called a PV panel, and a system of PV panels is called an array—arrays of a <u>photovoltaic system</u> supplying <u>solar</u> <u>electricity</u> to electrical equipment.



Fig. 3. Solar Panel

Photovoltaic modules harness sunlight to produce electricity using the photovoltaic effect. These modules are commonly made with wafer-based crystalline silicon cells or thin-film cells. The protective layer of a module can be located on either the front or back, safeguarding cells from damage and moisture. While most modules are rigid, there are also flexible options available, often utilizing thin-film cells. Electrically, cells are typically connected in series and then in parallel to achieve the desired voltage and increase current output. The power output of a module, measured in watts, is determined by multiplying its voltage (in volts) by current (in amperes). Manufacturing standards for solar panels are defined under standard conditions, which may differ from the actual conditions at the installation site.

Photovoltaic modules utilize sunlight to generate electricity through the photovoltaic effect. The majority of modules use either wafer-based crystalline silicon cells or thin-film cells. The load-carrying component of a module can be either the top or back layer. It is essential to protect the cells from mechanical damage and moisture. While most modules are rigid, there are also semi-flexible ones that are based on thin-film cells. The cells are typically connected in series and then in parallel to achieve the desired voltage and increase current. The module's power is determined by multiplying its voltage and current. Manufacturing specifications for solar panels are established under standard conditions, which may differ from the actual operating conditions at the installation site.

b) LCD

LCD (Liquid Crystal Display) technology is prevalent in modern devices such as CD players, DVD players, digital watches, and computers. LCDs have replaced Cathode Ray Tubes (CRTs) due to their lower power consumption and compact size. The LCD 16×2, commonly used in screens, operates by blocking light rather than dissipating it. This display module is widely used in various circuits and devices such as mobile phones, calculators, computers, and TV sets. LCDs are recognized for their affordability, ease of programming, and ability to showcase personalized characters, unique symbols, and animations.

c) Arduino Uno

Arduino UNO, a microcontroller board designed by the Arduino team, utilizes the Atmega168 or Atmega328p microcontroller. While similar to the Arduino Uno, the Nano board is smaller and preferred for its pin configuration and features. Arduino boards are widely used in electronic projects, embedded systems, and robotics. The Nano board is specifically introduced for beginners without a technical background who are involved in designing embedded systems with small-size components.

In the realm of power sources, an electric battery functions as the primary source of electric power. It is comprised of one or more electrochemical cells with external connections that are specifically designed to power electrical devices. When providing power, the battery's positive terminal acts as the cathode, while the negative terminal serves as the anode. Electrons move from the negative terminal to the positive terminal through an external electric circuit. When connected to an external electric load, a redox (reduction-oxidation) reaction takes place within the battery, transforming high-energy reactants into lower-energy products. The resulting difference in free energy from this reaction is then transferred to the external circuit as electrical energy. Over time, the term "battery" has evolved to include devices made up of a single cell, deviating from its original association with multiple cells.

Primary (single-use or "disposable") batteries are utilized once and then discarded due to the fact that the electrode materials are permanently altered upon discharge. A well-known example is the alkaline battery used in flashlights and various portable electronic devices. Secondary (rechargeable) batteries can be discharged and recharged multiple times with the application of an electric current, and reverse current can restore the original composition of the electrodes. Examples include lead-acid batteries used in vehicles and lithium-ion batteries used in portable devices like laptops and mobile phones. Batteries come in various shapes and sizes, ranging from small cells used in hearing aids and wristwatches to large battery banks the size of rooms that provide backup or emergency power for telephone exchanges and computer systems. Despite being heavier (in terms of mass) than conventional fuels like gasoline, the higher efficiency of electric motors in converting electrical energy to mechanical work, as opposed to combustion engines, somewhat balances this out in automobiles.

d) primary coil

Consists of two inductors: transmitting and receiving coils. The circuit operates primarily on the idea of mutual induction. The transmitting coil's dimensions are 22.1*13.1*3.2 mm, and its inductance is 3.7 uH.

e) Secondary Coil

The receiver module consists of several components, including an inductor coil, bridge rectifier, voltage regulator, and rechargeable battery. The bridge rectifier plays a crucial role in converting the AC signal received by the coil into a DC signal. However, the voltage produced by the bridge rectifier is not regulated and needs to be converted into a constant

voltage. This is where the voltage regulator IC 7805 comes into play. It ensures that the voltage remains stable and regulated. Additionally, the receiving inductor coil has specific dimensions of 24.2*9.38*5.36 mm and an inductance of 14uH.

f) Bridge Rectifier

The bridge rectifier converts alternating electricity to direct current. When compared to a center-tapped full wave rectifier, it has high rectification efficiency (82%), as well as a cheap implementation cost. The diode used is 1N4007.

g) LED

LED Indicator & Half Wave Rectifier: An LED, or light-emitting diode, is a semiconductor device that emits light when an electric current flows through it. When electrons and electron holes recombine within the semiconductor material, energy is released in the form of photons, resulting in light emission. The color of the emitted light is determined by the energy required for this recombination, which is influenced by the semiconductor's band gap. To achieve white light, different semiconductors can be combined or a layer of light-emitting phosphors can be applied to the LED. This approach enables the creation of a wide range of colors using LEDs.



h) Inductive coupling

Inductive coupling enables wireless power transmission across the system by utilizing mutual inductance. The magnetic field transfers the maximum energy possible, following Faraday's law of electromagnetic induction. This mechanism operates akin to resonant transformers, with alternating current voltage playing a crucial role. The primary coil is situated on the transmitter side, while the secondary coil is located on the receiver side. The inductance quantifies the connection between the two coils, allowing for an output of up to 5V.

i) Battery

This idea employs a battery to store electricity generated by solar power. The project utilizes a 12v/1.2Ah lead acid battery for this purpose. It effectively stores and transfers energy from the solar panels to the remaining components of the system.

IV.



METHODOLOGY

Fig 5. Diagrammatic Representation of Model.

The battery functions as a storage unit for solar energy conversion. The solar panel transmits electricity to the transmitter circuit, which consists of components like a BD139 transistor, capacitor, and resistor. The BD139 transistor monitors the charge voltage of the battery to ensure effective charging. Additionally, the transmitter circuit contains an inductive coil at the output, enabling wireless power transfer to the receiver section through inductive coupling. Inductive coupling operates based on mutual inductance, where current in one conductor induces a voltage in a nearby conductor, allowing for wireless electricity transfer. On the receiver end, a voltage regulator maintains a steady 5V output, ideal for charging mobile devices. This regulated voltage guarantees a reliable and secure charging process for the device being charged.



Fig 6. Final Output (wireless solar mobile charger).

VII. RESULT

The results of our efforts are outlined below. By increasing the number of coil turns, we observed a rise in the output voltage. For wireless electricity transfer through inductive coupling, we utilized 20-turn coils on both the transmitter and receiver ends. The 20-turn coil has a diameter of 4.5cm and can manage a maximum of 1.3 amps of electricity. The inductive coupling yielded a voltage of approximately 5v. Please refer to the table for detailed output information.

Sr .No	Value of Ranges	DC Output Voltage
1.	00	00
2.	0.5	0.10
3.	1.0	0.50
4.	1.5	0.89
5.	2.0	1.0
6.	2.5	1.10
7.	3.0	1.21
8.	3.5	2.20
9.	4.0	3.15
10.	4.5	3.50
11.	5.0	3.90

Table 1. Result

V. CONCLUSION

Solar power is a sustainable energy source that can be harnessed to generate electricity, like a portable wireless solar mobile charger. This innovative charger is ideal for off-grid locations and offers a safe and eco-friendly alternative to traditional chargers that often consume excessive power.

VI. REFERENCE

- 1. Abhijith Nidmar, Chetan Chougala, Kumaraswamy G M, Harshithgowda K, and Sharmila B S, "Solar Based Wireless Charging", International Journal of Engineering Research & Technology, NCRACES 2019 Conference Proceedings, Volume 7, Issue 10, 2019 pp:1-3
- 2. Sanju Tandan, Akash Chakraborty, and Khusboo Singh, Solar Power Bank With Wireless Charging, Eur. Chem. Bull. 2023, 12, (Special Issue 5), pp: 926-930
- 3. Rahil Imtiyaz, Aman Kumar, and Gitanjali Mehta, "Design and Implementation of Solar Powered Wireless Mobile Phone Battery Charger Using Electromagnetic Induction", International Journal of Progressive Research In Science and Engineering, Vol.2, No.5, 2021, pp: 110-113
- 4. Prof. Sara Anjum, "Wireless Solar Mobile Charger", International Journal of Research in Engineering and Science, Volume 10, Issue 6, 2022, pp. 1221-1224
- 5. Vasava Dhrumil, Vaghela Parth, Vishalkumar, Tandel Ayush and Rohit Rutvik, Wireless Charging E Car Using Solar Panel, International Research Journal of Modernization in Engineering Technology and Science Volume: 04, Issue: 05, 2022,
- 6. Prof. Rupali Tirale, Miss Aakansha Mahirale, Miss Rohini Ingale, Miss Dipali Ingale, Miss Snehal Ingale, Miss Gokula Punde, Miss Priya Suryawanshi, Miss Anjali Suralkar and Mr. Rohit Tekade, "Solar Power Bank with Wireless" Charging International Journal of Trend in Scientific Research and Development, Volume 6, Issue 4, 2022, pp:1478-1481
- Santosh E, Raj Kumar K, Vinod Raj B, Yogesh Gowda M., and Vasanth Kumar G, "Solar Wireless Electric Vehicle Non-Stoppable Charging System", International Journal of Research Publication and Reviews, Vol 4, no 6, 2023, pp: 2420-2425
- 8. Kang Miao, Bidirectional battery charger for electric vehicles, Asia (ISGT Asia) 2018. Pinto, J. G. Bidirectional battery charger with Grid-to-vehicle, Vehicle -to-Grid and Vehicle-to-Home technologies, IEEE 2020.
- 9. Bugatti Ram Vara Prasad, "Solar Powered BLDC Motor with HCC Fed Water Pumping System for Irrigation," Int. J. Res. Appl. Sci. Eng. Technol., vol. 7, no. 3, pp. 788–796, 2019
- 10. Gallardo-Lozano, Milanes-Monster, Guerrero Martinez, Three-phase bidirectional battery charger for smart electric vehicles, International Conference-Workshop 2021.
- 11. M. C. Kisacikoglu, "Vehicle-to-grid (V2G) reactive power operation analysis of the EV/PHEV bidirectional battery charger," Ph. D. dissertation, University of Tennessee, Knoxville, 2019.
- 12. Vehicle," i-manager's J. Digit. Signal Process., vol. 8, no. 1, p. 7, 2020
- 13. X. Zhou, S. Lukic, S. Bhattacharya and A. Huang, "Design and control of grid-connected converter in bidirectional battery charger for plug-in hybrid electric vehicle application," in Proc. IEEE Vehicle Power and Propulsion Conference (VPPC), 2019, pp. 1716–1721.
- 14. Bugatha Ram Vara prasad, D. V. S. J. Poojitha, and K. Suneetha, "Closed-Loop Control of BLDC Motor Driven Solar PV Array Using Zeta Converter Fed Water Pumping System," vol. 04, no. 17, pp. 2795–2803, 2017.
- 15. Sagolsem Kripachariya Singh, T. S. Hasarmani and R. M. Holmukhe Wireless transmission of electrical power overview of recent research and development, International Journal of Computer and Electrical Engineering, Vol.4, No.2, April 2019.
- 16. Bugatti Ram Vara Prasad, K. M. Babu, K. Sreekanth, K. Naveen, and C. V. Kumar, "Minimization of Torque Ripple of Brushless DC Motor Using HCC with DC-DC Converter," vol. 05, no. 12, pp. 110–117, 2018.
- 17. W. Green and J. T. Boys, "10KHz inductively coupled power transfer-concept and control," in roc. 5th Int. Conf. Power Electron. Variable-Speed Drives, Oct. 2019, pp. 694-699.
- 18. Bugatti Ram Vara Prasad T. Deepthi N. Satyavathi V. Satish Varma R. Hema Kumar, "Solar charging station for electric vehicles," Int. J. Adv. Res. Sci. Commun. Technol., vol. 7, no. 2, pp. 316–325, 2021
- 19. T. D. Nguyen, S. Li, W. Li, and C. Mi, "Feasibility study on bipolar pads for efficient wireless power chargers," in Proc. APEC Expo., Fort Worth, TX, USA 2020.
- 20. Bugatha Ram Vara prasad and K. Aswini, "Design of Bidirectional Battery Charger for Electric Vehicle," Int. J. Eng. Res. Technol., vol. 10, no. 7, pp. 410–415, 2021.
- 21. M. Singh, K. Thirugnanam, P. Kumar, I. Kar Real-time coordination of electric vehicles to support the grid at the distribution substation level IEEE Syst. J, 9 (2019), pp. 1000-1010.
- 22. International Research Journal of Engineering Technology (IRJET)Volume:03 Issue:05 May 2016

ELECTRONICS AND TELECOMMUNICATION ENGINEERING

Baby Care Monitoring System

Rahul Pawar¹, Priti Tawashe², Anjali Manakoji³ ^{1, 2, 3} SVERIs College Of Engineering, Pandharpur, India

Abstract— In this competitive world, both the parents are busy in their career. The main problem of today's parents is that they are unable to nurture their child because of their work. In India both the parents have to work and take care of their baby so there is more work pressure and stress on such parents especially on women. The number of mothers who are working has increased so baby care becomes a big challenge for many parents. So, mostly parents send their baby to their grandparent's house/relatives house or to the baby care centres. The parents are unable to continuously monitor their baby's condition in these situations. The solution to this problem is the design of a monitoring system which continuously monitors the baby and it should be of low cost. This project presents a Baby Care Monitoring system for busy and working parents so that they can take proper care and ensure safety of their baby. This baby monitoring system is able to detect crying of the baby automatically. For this, we use impact sound detection sensor. So that the system can monitor crying of the baby when the crying sound goes above the predetermined range then the GSM Module will send SMS to provided mobile number. The goal of our project is to save time, Provide maximum security and safety.

Keywords: Arduino, Cradle, Infant cry, Sound sensor, Buzzer, Camera, GSM Module

I.INTRODUCTION

Since last few years there is huge relocation of people in metropolitan cities in search of good job opportunities specially womens. Currently parents who are working face problems to give time & take care of their baby. Since COVID period people are working from home and it is difficult for them to maintain balance in both work and baby care. This problem not only hampers their career but also put a stress on them. We all know the problems faced by Parents to take care of their baby and especially in case where mother is working. In such case to give whole day for baby is not to possible. So, we need to create a system that can assist parents to continuously monitor the baby and notify about the same. Thus, we are presenting an idea to create a Baby care monitoring System which will help the parents to monitor their child even if they are not with baby & detect every movement of the baby and take proper care and nurturing of baby. Our project is a good idea to overcome this problem by creating "Baby Care Monitoring System" to keep watch & nurture baby in an proper way. We have to create a baby monitoring system to help parent in monitoring of their child irrespective of their presence whether they are at workplace or at home. It also has detection of baby crying sound and it will send notification to parents after certain time limit of baby cry. A camera module is attached to the cradle so that parent can have continuous watch on their baby while they away from baby.

II. Need of Project

Working Woman

Now a days every woman is working so it is impossible to then to balance workload and monitor the baby so to overcome this problem We are inventing "BABY CARE MONITORING SYSTEM USING IOT".

Ease For Parenting

Parents face many problems during parenting especially working parents. This system will make it easy to parents to nurture their baby.

Caring System

This system is not only for baby but we can also use it for patients who are at bed rest or for senior citizens. It will help to take care of them and monitor their activities.

Long Distance Monitoring

Sometimes parents go outside from their house for work keeping their child with their relatives. In this case parents require monitoring on their baby. This system helps parents to monitor baby from long distance.

III. PROBLEM DEFINITION

When both the parents are working and building their career for betterment of themselves and their family, it becomes next to impossible to take care of your baby and do work simultaneously because of these parents get workload and stress which is leading to various kinds of health issues to parents and lack of proper nurturing of baby. Sometimes they have to compromise with their career for taking care of the baby especially mother of that baby. Taking care and nurturing the baby should not be a problem in career of the parents. As a result we require a Baby care monitoring system which will help parents for proper care and nurturing of their baby. This system will help parents in various aspects for proper care of their baby.

IV. OBJECTIVES

- 1. To make ease for women to balance her work and nurturing the baby.
- 2. To make parenting easy.
- 3. To help parents to take care and nurture the baby.
- 4. Nurturing baby should not be obstacle for women to maintain their career.

V. SCOPE OF PROJECT

This system can be enhanced by adding Health monitoring system to it which will help to monitor baby's health & lead to parents give proper treatment to baby. To enhance our project we can use machine learning and artificial intelligence that will capture and record all the activities of baby for some days especially baby body cycle and then tells us when baby is hungry or it will urinate etc

VI. EXPECTED OUTCOME

Baby Care Monitoring System is a useful system for caring and nurturing of baby efficiently especially for working parents. It consists of cry detection module and music player also it has swinging of cradle on detecting baby cry. We are having an application for controlling of various modules. It can be an efficient way for working parents to take care and nurture their baby without disturbing their career.

VII.LITERATURE REVIEW

1] IOT Based Baby Monitoring System Smart Cradle by Senoj Joseph; Ajay Gautham.j

In this paper the author has proposed the method for creating baby monitoring system smart cradle by involving different modules like swinging, health monitoring, cry detection etc. This system is created with the help of components like sound sensor, servo motor, Arduino and main component is raspberry pi.

2] IRJET-IOT Based Baby Monitoring System Using Raspberry PI

In this paper a raspberry pi based baby monitoring system is created which results in efficient care and nurturing of baby with the help of raspberry pi. Different modules are involved in this system such as cry detection, Swinging of cradle etc. This system involves components like sound sensor, servo motor, Arduino.

3] IoT-BBMS: Internet of Things-Based Baby Monitoring System for Smart Cradle by Waheb A Jabbar

In this paper the methodology used for development of the IOT based Baby Monitoring system for smart cradle involves use of Node MCU, Arduino UNO, sound sensor, and tempreture and humidity sensor. This system helps in proper care and nurture of the baby in absence of the parents as it involves camera for monitoring the activities of the baby.

4] Smart Cradle by Dr.Isuru Nawinne; Dr.Mahanama

The given solution involves monitoring of the baby through a mobile app remotely. Sound Sensor for the detection of the baby's crying activity. When it detect crying, it send a message to parent, and plays a song. The Tempreture sensor notifies the parent about the environmental temperature near the baby and turn on the fan automatically with temperature

5] Baby Monitoring Smart cradle by Manas Dhruve In this paper the method given to create the baby monitoring smart cradle involve the components like Arduino mega, Raspberry pi, servo motor, sound sensor, etc. It also involves development of mobile app for controlling of the system.

VIII.METHODOLOGY

The given methodology, tries to solve the limitations of the currently available system. It mainly consists of sensors, hardware unit, Arduino, GSM module, Sound Sensor. The below block diagram shows how all element are connected with each other.



Fig.1 Methodology

The System takes input as baby's crying sound through sound sensor, by detecting this sound the system will run. This system consists of the servo motor which is used for swinging of cradle. It consists of SD card module which will play song on cry detection and it has 12V DC fan which is used to comfort the baby. Main component of the system is camera which is used for live monitoring of the System.

Step1:

Circuit Designing: Connection Of Arduino and NodeMCU with Different Modules Like sound sensors temperature and humidity sensor, servo motor, etc.

Step 2:

Arduino Program Coding: The different Modules interfaced with Arduino UNO by simple C programming for Arduino using Arduino IDE software.

Step 3:

Connection Of Components: The different components are connected to each other by connecting wires.

Step 4: Final Result Testing

IX. HARDWARE USED

Arduino

An Arduino is a microcontroller board which is an open source platform. It is based on easy to use hardware and software(Arduino IDE). It is used to build real world applications. It consists of circuit board which can be readily programmed using Arduino IDE software. It Uses microcontroller ATMega328p. We have used Arduino for interfacing of different devices to create this project.


Fig.2 Arduino

- Operating Voltage: 5V.
- Input Voltage (recommended): 7-12V
- Input Voltage (limit): 6-20V
- Digital I/O Pins: 14 (6 provide PWM output)
- PWM Digital I/O Pins: 6.

Sound Sensor

A sound sensor is an electronic sensor that detects sound through its intensity. Based on cry detection other modules will work. A crying child can expose caregivers and health care providers to sound pressures as high as 120 dB(A).



Fig.3 Sound Sensor

- Operating Voltage: 3.3V to 5V DC
- LM393 comparator with threshold preset
- PCB Size: 3.4cm * 1.6cm
- Induction distance: 0.5 Meter
- Operating current: 4~5 mA
- Microphone Sensitivity (1kHz): 52 to 48 dB
- Easy to use with Microcontrollers or even with normal Digital/Analog IC
- Small, cheap and easily available

GSM Module

GSM stands foe Global System for Mobile Communication. SIM900A GSM Module is the smaller and cheaper module for GSM communication. It is common with Arduino and microcontroller in most of the application. We have used GSM Module for sending SMS on parents mobile number. When sound sensor will detect the baby cry then GSM module will send message as "Baby is Crying" on given mobile number.



Fig.4 GSM Module

- Single supply voltage: 3.4V 4.5V
- Power saving mode: Typical power consumption in SLEEP mode is 1.5mA
- Frequency bands:SIM900A Dual- band: EGSM900, DCS1800. The SIM900A can search the two frequency bands automatically. The frequency bands also can be set by AT command.
- GSM class: Small MS
- GPRS connectivity: GPRS multi-slot class 10 (default), GPRS multi-slot class 8 (option)
- Transmitting power: Class 4 (2W) at EGSM 900, Class 1 (1W) at DCS 1800
- Operating Temperature: -30°C to
- +80°C
- Storage Temperature: -5°C to +90°C
- DATA GPRS: download transfer max is 85.6KBps, Upload transfer max 42.8KBps
- Supports CSD, USSD, SMS, FAX
- Supports MIC and Audio Input

Servo Motor

Servo Motor is closed loop servomechanism which is us for controlling motion and final position. In our project we have used servo motor for swinging of cradle. We ha SD Card Module programmed Arduino to interface the servo motor so that can be used for create the motion of cradle i.e swinging the cradle.



Fig.5 Servo Motor

- Operating Voltage is +5V typically
- Torque: 2.5kg/cm
- Operating speed is 0.1s/60°
- Gear Type: Plastic
- Rotation : 0° -180°
- Weight of motor : 9gm

Quantum USB Camera QHM495B

We have used this USB camera for continuous surveillance of baby. It will make live monitoring of baby activities. When parents are away from the baby it will help to monitor the baby.



Fig.6 Quantum USB Camera QHM495B

SD Card Module

SD Card Module is used for SD card processes the cradle, such as reading and writing with a microcontroller. The board can be used with microcontroller systems like Arduino. We have used sd card module to play a lullabuy when baby cry is detected. We have to save a song in wav format and sd card is directly inserted in to the sd card module.



Fig.7 SD Card Module

- Operating Voltage: 4.5V 5.5V DC
- Current Requirement: 0.2-200 mA
- 3.3 V on-board Voltage Regulator
- Supports FAT file system
- Supports micro SD up to 2GB
- Supports Micro SDHC up to 32GB

Speaker

Speakers are output devices that enable the people to hear to a sound as an output. We have used speakers to listen the lullabuy playing after baby cry detection.



Fig.8 Speaker X.RESULT

Thus we have developed Baby care monitoring system using Arduino and it assists parents in monitoring their baby. Because of using Node MCU overall system cost reduces. The experimental demonstration show the system works successfully and efficiently and results of the system is given below.



Fig.9 Hardware XII. CONCLUSION

The baby care monitoring system can be used in situations when we have to monitor the baby's activities. It can also be used when a person's health has to be checked. It will help working parents to proper care and nurture the baby. This system can be helpful in hospitals or in baby care centers which will reduce manpower.

XIII. REFERENCES

- 1. P. Hemalatha and S. Matilda, "Smart digital parenting using internet of things," in 2018 International Conference on Soft-Computing and Network Security (ICSNS), pp. 1–6, Coimbatore, India, 2018.
- 2. M. Burhan, R. A. Rehman, B. Khan, and B. S. Kim, "IoT elements, layered architectures and security issues: a comprehensive survey," Sensors, vol. 18, no. 9, p. 2796, 2018.
- 3. View at: Publisher Site | Google Scholar
- 4. L. Atzori, A. Iera, and G. Morabito, "The internet of things: a survey," Computer Networks, vol. 54, no. 15, pp. 2787–2805, 2010.
- 5. View at: Publisher Site | Google Scholar
- 6. M. T. Ubaid, M. Z. Khan, M. Rumaan,
- M. A. Arshed, M. U. G. Khan, and A. Darboe, "COVID-19 SOP's violations detection in terms of face mask using deep learning," in 2021 International Conference on Innovative Computing (ICIC), pp. 1–8, Lahore, Pakistan, 2021. View at: Publisher Site | Google Scholar
- 8. M. A. Arshed, W. Qureshi, M. Rumaan,
- M. T. Ubaid, A. Qudoos, and M. U. G. Khan, "Comparison of machine learning classifiers for breast cancer diagnosis," in 2021 International Conference on Innovative Computing (ICIC), pp. 1–6, Lahore, Pakistan, 2021. View at: Publisher Site | Google Scholar
- M. A. Arshed, W. Qureshi, M. U. Khan, and M. A. Jabbar, "Symptoms based Covid-19 disease diagnosis using machine learning approach," in 2021 International con-Ference on Innovative Computing (ICIC), Lahore, Pakistan, 2021.

Smart Farm Hub

¹Sohel Sayyad , Disha Bhattad , Ashitosh Mane, Kenny Santania, Nikhil Gaikwad , Kajal Ingale , ²Dr.Meenakshi M Pawar, ³Amol Chounde ¹UG Student, SVERIs College Of Engineering, Pandharpur, India ²Professor, SVERI's College of Engineering, Pandharpur, ³Amol Chounde

³Assistant Professor, SVERI's College of Engineering, Pandharpur

Abstract—The "SmartFarm Hub" project focuses on developing an IoT- enabled system for efficient farm management by enabling farmers to remotely oversee their farms. By implementing IoT technology, the system aims to streamline daily farming tasks and provide farmers with real-time data through a user- friendly mobile application. The raw data received from multiple sensors is processed and analyzed to offer actionable insights for optimized remote monitoring, controlling, and irrigation. The solution contributes to sustainable farming practices, improves productivity, and enhances profitability in the agricultural sector which summarizes the methods and techniques used in Wireless Database Management to create an IoT system tailored for farmers who can easily understand the farm condition in graphical forms.

Keywords:Arduino, node MCU, temperature sensor, humidity sensor, soil moisture sensor, electronic motor, relay, GSM module, solenoid valve, solar panel, electronic motor relay, battery, plugs, switches, pipes, wires, database, android application.

I. INTRODUCTION

The Internet of Things (IoT) will be the foundation of smart computing in the future. It plays a vital part in converting "Traditional Technology," which is used in offices and homes, into "Next Generation Everywhere Computing." The "Internet of Things" is becoming increasingly prominent in global research, particularly in the field of advanced wireless communications. IoT is currently affecting everyone and, from the perspective of the average user, is building the groundwork for the creation of several products, including automation, smart living, smart health services, and smart education in schools. Additionally, it is employed commercially in a variety of industries, including business management, transportation, manufacturing, and agriculture.

Agriculture is the IoT domain with the most research. Because the world's population is growing at an accelerating rate, it is vital to guarantee food security. When researchers initially began using ICT-based techniques in this field, they were helpful in certain ways but ultimately would not solve our problems. Thus, they are currently investigating IoT as a substitute for ICT in agriculture. Applications for agricultural products are required, such as supply chain management, infrastructure management, soil moisture monitoring, and environmental condition monitoring for temperature and moisture.

After the sensors' data is gathered, it is stored on a server, and mobile applications and a cloud-hosted database are created. A facility with instruments and equipment that provides information to make moisture detection in our surrounding environment is what is known as a moisture monitoring system. It will track temperature, humidity, precipitation total, and moisture content while providing real-time weather updates at a specific location, making the data accessible from anywhere in the globe. The system's goal is to deliver an effective environmental monitoring system by using Internet of Things (IoT) technology to measure and monitor weather data. It also displays weather data via a mobile application that users can easily and quickly access.

Raindrop module, moisture sensor, temperature and humidity sensor (DHT11), and other weather-related sensors are collected using Node MCU in this study. The real-time database is created using the sensor data, and an Android mobile application and cloud-hosted database are created to display the local weather in real-time for that area. The key benefit of this type of system is its lower cost advantage over other weather station projects of a similar nature due to its use of Node MCU, a module that provides low power consumption and cost-effective Wi-Fi connectivity for the Internet of Things applications. applications are developed to show the real-time weather conditions of a particular region on an Android mobile phone. The main preferences of such a system include lower cost advantage compared to other similar weather station projects because of using Node MCU, designed for Wi-Fi connectivity for IoT applications module with a cost-effective solution and low power consumption.

II. RELATED WORK

The project's goal is to use the Internet of Things (IoT) to build a smart farming system that will enable farmers to remotely manage their properties using a mobile app. We need to look at other projects and conduct research in several areas to do this effectively. We should first investigate the applications of IoT in agriculture, such as data collection via sensors and equipment control via remote access. Next, we need to comprehend the operation of wireless sensors on farms, particularly those that take measurements of soil moisture and temperature. We can also gain knowledge from current smart farming initiatives that employ technology to assist farmers in increasing crop yields and optimizing resource utilization. To make our app user-friendly, we must investigate mobile apps tailored for farming. We also need to figure out how databases store this data and how to interpret and process the data that is gathered from the sensors. We must learn how to make our devices energy-efficient since power conservation is essential to the Internet of Things. Different methods of communication between devices will be used, and we must be aware of the advantages and disadvantages of each. For accurate data, we should become familiar with the functioning of the sensors that we will be using to monitor the weather. We must compare the cost-effectiveness of our solution to others because cost is a significant consideration. Finally, we want to look at eco-friendly practices and make sure our app is user- friendly for farmers to make farming more sustainable. We can get ideas and improve our project by looking into these areas.

III. PROPOSED SYSTEM

3.1: Overview and Significance of SAMS:

The "SmartFarm Hub" project introduced a ground-breaking solution called the Smart Agriculture Management System (SAMS). It makes use of the Internet of Things (IoT) capabilities to revolutionize farm management. With IoT changing the face of technology, SAMS wants to empower farmers by giving them the ability to effectively manage their farms from a distance. The project acknowledges the increasing importance of the Internet of Things and concentrates on tackling particular issues in a range of industries, with a primary focus on agriculture.

As the world's population continues to rise quickly, it is critical to ensure food security. Although traditional ICT (information and communication technology) solutions have proved useful, they are unable to keep up with the changing demands of the agriculture industry. To monitor environmental conditions, manage infrastructure, improve agricultural supply chains, and monitor soil moisture, the project embraces IoT as a strong alternative.

3.2 : Essential Elements and Advantages of SAMS:

The main component of this project is SAMS, which is intended to make regular farming chores easier and give farmers access to real-time data via an intuitive mobile application. The principal functions of the system comprise the gathering, manipulation, and interpretation of data from multiple sensors, such as soil moisture, temperature, humidity, and specialty weather-related sensors like the Raindrop module and DHT11. To optimize remote monitoring, control, and irrigation, SAMS depends on this data to offer insightful information. IoT connectivity, wireless sensor networks, a user-friendly mobile application, sophisticated data processing, dependable database management, power-efficient design, support for several communication protocols, real-time weather monitoring, and an economical strategy are just a few of the crucial elements that the system integrates. This project, spearheaded by SAMS, represents a breakthrough in agricultural management. Providing farmers with up-to-date information, practical advice, and intuitive mobile applications, enhances agricultural output and sustainability. Accessible, economical, and in line with the evolving needs of the farming sector, SAMS encourages ecologically friendly farming methods and food security.

3.3 Smart Agriculture Management System

An essential part of the larger initiative to transform farm management is the Smart Agriculture Management App. This mobile application is essential to enable farmers to effectively manage their farms from a distance in a world where the Internet of Things (IoT) is having a profound impact. This application assumes a crucial role, acknowledging the growing significance of IoT and its potential to address the unique challenges within the agriculture sector. Food security has become increasingly important due to the world's population growth, and this app has the potential to support sustainable farming methods while also increasing agricultural productivity and profitability. Farmers can use the app to access real-time data that helps them make educated decisions. It provides insights into several aspects of their farms, including information on soil moisture, temperature, humidity, and specific weather conditions. This app allows farmers to remotely operate and control

agricultural machinery and irrigation systems, saving them a great deal of time and resources. It is more than just a passive source of data. It also distinguishes itself with its easy-to-use interface, which is meant to be understandable and accessible to farmers with different levels of technical expertise. In addition to being displayed, the data gathered from the farm's sensors is processed and analyzed in-depth by the app. With the help of this analysis, farmers will be able to make well-informed decisions and maximize the effectiveness of their farm management techniques.

Through the use of secure connections to databases hosted in the cloud, the application guarantees farmers access to historical and current farm data from almost anywhere, providing them with a level of convenience and flexibility never before possible. This app stands out in part because of its dedication to affordability. Its broad range of farmer customers is served by its affordable and easily accessible design, which helps to democratize sophisticated farm management tools. The Smart Agriculture Management App is a huge advancement in farm management. Real-time data, actionable insights, and smooth control options are provided to farmers through the application of IoT technology and advanced data processing. It is made to be as affordable as possible, serving a wide variety of farmers, and coordinating well with the changing needs of the agriculture sector. This app serves as a trailblazing illustration of smart agriculture, whereby the use of a mobile interface renders remote farm management not only practical but also effective and user-friendly.

IV. IOT AND INTELLIGENT SYSTEMS FOR WATER AND FERTILIZATION MANAGEMENT IN AGRICULTURE

4.1Cloud Technology

Because cloud technologies make data collection, storage, processing, and sharing more efficient, they have sparked a revolution in smart agriculture and irrigation systems. These technologies are the foundation for operations that are more accessible, efficient, and data- centric. and Google Cloud are examples of scalable and secure data storage platforms that enable the centralization, organization, and easy retrieval of agricultural data from a variety of sources. By providing farmers with insights, crop predictions, and optimal irrigation schedules, data analytics and machine learning raise crop yields and resource efficiency. Irrigation systems can be remotely monitored and controlled in real- time, from any location, allowing for quick schedule modifications to conserve water. Secure data sharing makes stakeholder collaboration easier, especially when tackling difficult crop-related issues.

Cloud solutions provide scalability and cost-efficiency because they scale with agricultural operations and because costs are directly correlated with resource consumption, reducing the need for upfront hardware purchases. Sensitive agricultural data is protected even in the event of a hardware failure thanks to strong data security and redundancy protocols. Furthermore, cloud-based applications, which are frequently accessed through web dashboards and mobile apps, simplify data interaction and intuitive control and monitoring through their user- friendly interfaces. Essentially, data-driven decision- making in smart irrigation and agriculture systems and increased agricultural efficiency have been made possible by cloud technologies.



4.2Architecture of SmartFarm Hub

i. Sensor of Soil Moisture:

An essential part of the project is the Soil Moisture Sensor, which uses a number of probes to gauge the soil's moisture content. Accurate readings are guaranteed by the strategic placement of this sensor within the target soil area. Its analog or digital output, which shows the moisture content of the soil, is closely linked to the ESP8266 microcontroller.

ii.Sensor of Humidity and Temperature:

By recording ambient temperature and humidity levels, the Temperature and Humidity Sensor improves the project's environmental monitoring capabilities. For applications like agriculture or indoor climate control, where weather conditions are important, this sensor is essential. Its output is seamlessly integrated into the ESP8266 as either analog or digital signals.

iii.Weather Station

Acting as a clever middleman, the Weather Station coordinates the data transfer between the sensors and the ESP8266 microcontroller. This part serves as a central hub for gathering, evaluating, and possibly displaying the environmental data and has data processing capabilities. An essential component of maintaining effective system communication is the Weather Station.



Fig.2 Block Diagram

iv.Microcontroller ESP8266:

The project's central nervous system is the ESP8266 microcontroller, a strong and adaptable part. It can send data to a specified platform or user interface after processing inputs from the temperature and humidity sensor and the soil moisture sensor. The microcontroller's capacity to manage Wi-Fi connectivity makes it possible to do real-time data analysis and remote monitoring.

v.Relay Module:

A relay module receives the output that the ESP8266 microcontroller produces. This part functions as a switch, allowing or denying connected devices power in response to commands from the ESP8266. It ensures precise control by acting as a vital connection between the operating components and the microcontroller.

vi.Power Supply:

The system's mainstay, the power supply unit gives all of the other parts a steady and dependable source of power. This guarantees uninterrupted and continuous operation. To ensure optimal functionality, the Soil Moisture Sensor, Temperature and Humidity Sensor, and ESP8266 are carefully connected to the power supply output.

vii.Motor Irrigation:

The motor irrigation system is connected to the output of the relay module. This part is in charge of managing the irrigation system, including turning on and off the water pump. With the Relay Module interface provided by the ESP8266, irrigation can be intelligently managed using environmental data.

viii.Fertigation Solenoid Valve:

The Fertigation Solenoid Valve is the recipient of another output that comes from the Relay Module. The irrigation system's flow of nutrients or fertilizers is managed by this valve. Based on real-time data from the sensors, the ESP8266's integration with the Relay Module enables automated and controlled fertigation processes.

ix.ESP8266 to Web Server:

The ESP8266 not only processes data but also acts as a communication link, connecting to a web server. This connection enables users to remotely monitor environmental conditions and control irrigation and fertigation processes through an online platform. In simpler terms, it allows users to keep track of and manage the system from anywhere using the internet.

x. Data Visualization:

The data visualization process is started by the ESP8266 when it receives data from the web server. Using this data, the server creates various graph types that dynamically depict temperature, humidity, soil moisture content, and irrigation cycles.

V. RESULT AND DISCUSSION

A strong and user-focused solution is produced by implementing this expanded Smart Agriculture System architecture. An easy-to-use web interface allows users to effortlessly monitor real-time data, such as temperature, humidity, soil moisture, irrigation activities, and fertigation processes. The system provides users with visually striking graphs that dynamically represent environmental conditions and system performance, going beyond simple data. Analyzing historical data is made possible by the web server's systematic spreadsheet storage of all transmitted data. Users can identify trends and patterns and make well- informed decisions about the best farming practices with the help of this historical archive. By enabling users to personalize data displays to suit their preferences, the user interface promotes interaction. Essentially, the outcome is a comprehensive and intelligent Smart Agriculture System that gathers, processes and converts data into useful insights for increased productivity and sustainable farming methods.

5.1 Performance Evaluation

The temperature sensor graph, spanning from 8 am to 2:15 pm, reveals distinct patterns in ambient temperature. Morning hours witness a gradual rise, followed by midday stability. Peak temperatures are observed around noon, providing valuable insights into environmental dynamics crucial for agricultural decision-making during this specific timeframe. Fig.5(a). The humidity sensor graph, covering the period from 8 am to 2:15 pm, illustrates the fluctuation in atmospheric moisture. Morning hours typically show a rise in humidity, while midday stability is observed. The graph's peak values around noon offer insights into crucial environmental conditions, providing valuable information for agricultural decision-making during this specific timeframe Fig.5(b). The soil moisture sensor graph, spanning from 8 am to 2:15 pm, delineates the dynamics of soil moisture levels. Morning hours may reveal changes influenced by overnight conditions. The graph showcases variations in soil moisture, with potential peaks or troughs around specific times. This data is instrumental for gauging optimal irrigation and agricultural practices during the specified timeframe Fig.5(c). The light intensity sensor graph, capturing data from 8 am to 2:15 pm, depicts the variations in natural light exposure. Morning hours mark a gradual increase, reaching a potential peak around midday. The graph offers insights into the duration and intensity of optimal light conditions crucial for plant growth, aiding in informed decisions for agricultural practices during this specific time frame Fig.5(d).



(a)Temperature





(c)Soil Moisture



(d)Light Intensity

Fig.3. weather station scheduled (a)Temperature (b)Humidity (c)Soil Moisture (d)Light Intensity







Fig.4. Insightful Dashboard

"We have a visually stunning interface with dynamic graphs on our mobile application's home screen, which gives users instant access to important data points. With its user- friendly comprehension system, the real-time information is readily available, giving users empowerment. Customizable options allow for easy customization of the experience to personal preferences. Users can explore the graphs interactively by zooming in, panning, and analyzing data with ease because they are touch-responsive. These graphical elements are carefully designed to provide an understandable and illuminating interpretation of the data, as demonstrated on the opening page. Our application's user-friendly features guarantee a seamless and enriching experience by giving users direct access to crucial data insights."

5.2 Protocol-Governed Service

"Thank you for visiting Smart Farma! You accept our Policy- Based Service by downloading and using our mobile application. It is expected of users to give true information, abide by all relevant laws, and handle the graphical exploration features with care. A seamless user experience is promoted by the interactive exploration of the touch- responsive customization options. Users who violate these policies may have their services terminated or suspended by Smart Farma. For information on how we use data, please read our Privacy Policy. Users will be informed of any modifications to these policies. Please get in touch with 12345 67890 with any queries or concerns. I appreciate your participation in Smart Farma.

VI. CONCLUSIONS

Using the Internet of Things, the "SmartFarm Hub" aims to revolutionize farming. Farmers are empowered with real-time data for remote management through the use of IoT by the Smart Agriculture Management System (SAMS). SAMS uses IoT for infrastructure management, environmental monitoring, and agricultural process optimization in order to address the issue of global food security.

SAMS combines wireless sensors, an intuitive mobile app, IoT connectivity, and effective data processing. In order to promote sustainable farming methods and higher productivity, it seeks to give farmers access to real-time data and insights. The Smart Agriculture Management App provides real-time environmental data for informed decision-making and enables remote farm control. Its easy- to-use interface makes sophisticated farm management tools accessible to all.

The system's architecture, using sensors, microcontrollers, relay modules, and a web server, forms a cohesive and intelligent framework. Performance evaluation demonstrates effective monitoring of temperature, humidity, soil moisture, and light intensity, guiding agricultural decisions.

Integration of cloud technology enhances data efficiency, scalability, and cost-effectiveness, facilitating remote monitoring and control for improved resource utilization. The system's protocol-governed service ensures responsible user engagement.

VII. REFERENCES

- 1. Doshi, P. J., Patel, T., & Bharti, S. K. (2022, August 12). Smart Farming using IoT: A Solution for Optimally Monitoring Farming Conditions.
- Nayyar, A., & Puri, V. (2016). Smart Farming: IoT-Based Smart Sensors Agriculture Stick for Live Temperature and Moisture Monitoring Using Arduino, Cloud Computing & Solar Technology. In The International Conference on Communication and Computing (ICCCS-2016).
- 3. Jegadeesan, S., & Prasanna Venkatesan, D. G. K. D. (2016, January-March). Smart Cow Health Monitoring: The Farm Environmental Monitoring and Control System Using Wireless Sensor Networks. International Journal of Advanced Engineering Technology.
- 4. Gorli, R., & Yamini, G. (2017). Future of Smart Farming with Internet of Things. Journal of Information Technology and Its Applications.
- 5. Tenzin, S., Siyang, S., Pobkrut, T., & Kerdcharoen, T. (2017, March 27). Low-cost Weather Station for Climate-Smart Agriculture. In 2017 9th International Conference on Knowledge and Smart Technology (KST).
- 6. Marwa, C., Othman, S. B., & Sakli, H. (2021, January 26). IoT-Based Low-cost Weather Station and Monitoring System for Smart Agriculture.
- 7. IoT and agriculture data analysis for smart farm Jirapond Muangprathub, Nathaphon Boonnam, Siriwan Kajornkasirat, Narongsak Lekbangpong, Apirat Wanichsombat, Pichetwut Nillaor
- 8. .Smart farming for improving agricultural management Elsayed Said Mohamed, AA Belal, Sameh Kotb Abd-Elmabod, Mohammed A El-Shirbeny, A Gad, Mohamed B Zahran
- 9. Design and implementation of a connected farm for smart farming systemMinwoo Ryu, Jaeseok Yun, Ting Miao, Il-Yeup Ahn, Sung- Chan Choi, Jaeho Kim2015 IEEE sensors, 1-4, 2015
- 10. Smart farm monitoring via the Blynk IoT platform: case study: humidity monitoring and data recordingPeerasak Serikul, Nuttapun Nakpong, Nitigan Nakjuatong 2018 16th International conference on ICT and knowledge engineering (ICT&KE), 1-6, 2018

Automatic seed Sowing Robot

¹Adsul Abhishek, ²Atar Sahil, ³Divane Saurabha, ⁴Ms. S S Pawar ^{1, 2, 3} UG Student, SVERIs College of Engineering, Pandharpur. ⁴Assistant Professor, SVERI's College of Engineering, Pandharpur.

Abstract: Farming is a vital aspect of human survival, and farmers dedicate a significant amount of their time to cultivating crops, irrigating them, and other tasks. Farmers will benefit greatly from the suggested system, which integrates robotics and agriculture. It can move across the field like a farmer, till the soil, plant seeds in the designated rows, and irrigate the field on its own. Conventional seeding, which is widely employed in farming, requires more work and time. Although the seed feed rate is higher, the overall operation takes longer.

I. INTRODUCTION

In India, agriculture is the primary industry for employment. A little over half of the population is employed in agriculture. It is our economy's skeleton. Despite the emphasis on industrialization, the Indian economy still heavily depends on agriculture, which generates millions of jobs nationwide and contributes significantly to GDP.

Crop productivity must rise in order to stimulate the economy [1]. The technology employed during the production and processing stages affects the volume of production in addition to capital investments and marketing plans. Modern technology are lacking for Indian farms.

Farmers engage in agricultural techniques, but with the right automation, agriculture becomes more efficient.[2] This device assists a farmer in automatically planting seeds, hence minimizing manual effort. An important factor in meeting the demands for agricultural productivity is automation. Technology adoption rates will rise and expenses will decrease as automation and agriculture gain traction. Farmers that practice autonomous agriculture can more efficiently increase precision while minimizing their negative effects on the environment.

II. PROBLEM STATEMENT

The inaccuracy and inefficiency of manually spreading seeds in agriculture is the issue. Due to uneven seed spacing and inadequate distribution, the current approach produces crop growth and yield that are below ideal levels. The project's goal is to create a method for automating the sowing of seeds and guaranteeing accurate and consistent seed placement.

III. LITERATURE SURVEY

The literature review highlights key research and developments in the field of automatic seed sowing agriculture robots. Various studies have explored the use of robotics to improve seed sowing efficiency. Researches have focused on navigation technique sensing technologies, and control algorithms to optimize seed placement and spacing. Notable advancements include computer vision for seed detection, GPS-based navigation and precision control systems. However there is still a need for comprehensive and cost-effective solutions that can be easily integrated into existing agricultural practices.

1] Novel Approach For Automatic Seed Sowing Machine - Aniruddha D Dharmadhikari-In this paper the author has proposed an Automatic Seed Sowing Machine to integrate the automation in agriculture sector. To reduce human efforts and to reduce the time taken by workers manually sowing seed. It uses an IR sensor which is capable of sensing any obstacle detected in path of seed sowing machine

2] Design and Development of Automatic Seed Sowing Machine - P. Premalatha-In this paper the author has proposed this project aims to increase the productivity and to reduce the time for seed sowing process and wastage of seeds. An ultrasonic sensor is also installed to detect the obstacle in the path and end of each row.

3] Design of automatic seed sowing Machine for agriculture sector- Ratnesh Kumar-In this paper the author has proposed. This machine will enable the farmers to engage in alternative activities that generate income. By this paper farmers will have an option that without their direct involvement, their land is being ploughed and seeds being sown.

4] Fabrication and Automatic Seed Sowing Machine- Senthilnathan N-In this paper the author has proposed this work a seed sowing machine has been developed that help the farmers in harvesting the best crop with least efforts. Agro technology is the process of implementing the recent technologies to develop the crops that are being produced.

5] Literature Review on Automatic Seed Feeder-A.O.Hannure In this paper the author has proposed, In plant nursery more time is required for plantation which is due to seed feeding process. For reducing these problems of plant nursery research of automatic seed feeder mechanism is used.

IV. SYSTEM DESIGN

The system design for an Automatic Seed Sowing Robot using Arduino involves several key components and subsystems working together to automate the seed sowing process in agriculture. Below is an overview of the system design:



Fig.1 block diagram

1) Hardware Component::-

Arduino UNO:- The arduino responsible for controlling the entire system.

Sensors:

Soil Moisture Sensors: Monitor soil conditions to determine the optimal planting time and depth. Motor Encoders: Track the movement of the robot and ensure accurate seed spacing.

Actuators:

Motors and Servos: Used for robot movement, including forward, backward, turning, and for seed placement. Seed Release Mechanism: Controlled by a servo to release seeds at specific intervals and depths.

Power Supply: Batteries or a rechargeable power source to provide energy for the robot's operation.

2) Software Component:-

Arduino IDE: The code for the robot is developed using the Arduino Integrated Development Environment.

Control Algorithms: Algorithms for navigation, seed placement, and data collection.

Bluetooth module: In mobile make one application and connect that app to the Bluetooth module to control the robot.

User Interface Software: Code for the user interface to input commands and monitor the robot's progress.

Data Analysis Software: If required, software for processing and analyzing the data collected during operation.

3)System Architecture:-

The Arduino uno serves as central hub connecting and controlling all the hardware components. Sensors continuously collect data and feed it to the Arduino, which processes the information and makes decision based on predefined algorithms. Actuators including motors and servo, execute commands from the Arduino to control the movement and seed placement.

The user interface allows the farmer to set parameters such as seed spacing, planting depth and field boundaries.

Data collected during operation is displayed on the user interface and if necessary transmitted for further analysis.

4)Operation sequence:-

The robot is placed in the field, and the farmer inputs the required parameters through the user interface. The robot uses GPS data for navigation and moves through the field in predefined patterns, ensuring complete coverage. Soil moisture sensors assess soil conditions, and based on this data, the robot determines the optimal time and depth for seed placement. The seed dispenser controlled by a servo, releases seeds at the predetermined intervals and depths. Motor encoders track the robot's movement and ensure uniform seed spacing. The robot provides real-time updates on its progress and any issues encountered via the user interface.

IV. **ADVANTAGES**

1. Efficiency: Automation of the seed sowing process significantly increases efficiency by reducing the time and labor required for this critical task. The robot can work tirelessly, day and night, without fatigue.

2. Precision: The robot ensures precise seed placement at optimal depths and spacing, resulting in improved crop yield and uniform growth. This precision minimizes resource wastage, such as seeds and water.

3. Labor Reduction: By automating seed sowing, the project reduces the dependency on manual labor, which can be costly and subject to human error. This can lead to substantial cost savings for farmers.

4. Versatility: The system is designed to adapt to various crop types and field conditions, making it versatile and applicable in diverse agricultural settings. This adaptability enhances its value to different farmers and regions.

VI.CONCLUSION

Our project, the Automatic Seed Sowing Robot Using Arduino, demonstrates how technology and agriculture can be combined to solve important problems that contemporary farmers face. We hope that automation, accuracy, and resource efficiency will help the world shift to more productive and sustainable farming methods.

VIII. REFERENCES

- 1. Aniruddha D Dharmadhikari., Surajmal Sharma., Makarand Baviskar., Shantanu Kakad.Manish Chandekar ., "Novel tomatic Seed Sowing Machine "Volume: 10,Issue: 05 May 2023.P. Premalatha ., Sharath Singh.," Design and Development of Automatic Seed Sowing Machine ".,Vol. 8, 2019.
- 3. Ratnesh Kumar, Aadhar Govil, Parth Daga, Shubh Goel, Saurabh Dewangan.,
- Senthilnathan N, Shivangi Gupta, Keshav Pureha and shreya Verma, "Fabrication & Automatic Seed Sowing 4. Machine."Volume 9, Issue 4, April 2018
- A.O.Hannure , S.P.Kshirsagar, V.S.Kodam , O.N.Patange , V.S.Nakod., "Literature Review on Automatic Seed 5. Feeder".,Volume 36 .,7- June 2016
- Anuja Mohalkar., Priti Mohite., Shubhangi Nagare., Prof.Sampada Tavase., " Automatic Seed Sowing Machine Using 6. Solar Panel.
- 7. Smith, A. B., et al. "Autonomous Seed Sowing Robot for Precision Agriculture." Journal of Agricultural Engineering Research.
- Patel, R., et al. "Design and Development of a GPS-Guided Agricultural Robot." International Conference on 8. Robotics and Automation.
- Kumar, S., et al. "Obstacle Avoidance Techniques for Agricultural Robots: A Review." Agricultural Engineering 9. Today. "Design of automatic seed sowing Machine for agriculture sector".,02 May 2022.

Crop Monitoring System Using Wireless Sensor

¹Shivani Atkale,²Aishwarya Bhosale, ³Dr.Nikhilesh k Mishra
^{1, 2,} UG Student, SVERIs College Of Engineering, Pandharpur.
³Assistant Professor, SVERI's College of Engineering, Pandharpur.

Abstract: Precision agriculture has drawn a lot of attention lately due to its potential to increase farming conversion rate. This abstract illustrates a methodical approach to "crop monitoring system using wireless sensor" technology. In modern sensing technologies and Wireless Sensor have made it possible to complete agricultural tasks accurately, promptly, and adequately. We call these methods "smart agriculture." We address several sensing technologies that support smart agriculture in this paper. This research project's main objective is to create a wireless sensor network-based agricultural monitoring system that will improve farming quality and productivity without requiring a lot of physical monitoring. Temperature, humidity, light intensity, and gas levels are the main factors in crop production that have an impact on plant growth, productivity, and quality. This goal is to use the Arduino tool to design and develop an agricultural system. Temperature, humidity, and soil moisture are examples of ecological parameters that are assessed using the Arduino chip in conjunction with sensors. In the majority of agricultural settings, we verify these variables by hand.

Keywords: Arduino UNO, Liquid Crystal Display(LCD), Temperature Sensor, Humidity Sensor, Soil Moiture,LDR Sensor,Gas Sensor.

I. INTRODUCTION

In contemporary agriculture, wireless sensor- based crop monitoring systems are becoming more and more common, completely changing how farmers tend to their crops. This novel strategy seeks to raise farming operations' general quality, maximize resource efficiency, and increase yield. A crop monitoring system's main objective is to provide a more automated and effective method of crop observation and management that eliminates the need for ongoing human intervention. Conventional monitoring techniques, which entailed sporadic manual inspections, are laborious and sometimes fail to deliver real-time data that is essential for prompt decision-making.

This problem is addressed by wireless sensor technology, which makes it possible to continuously monitor important environmental elements that have an impact on crop growth automatically.Plant growth, productivity, and quality are significantly influenced by temperature, humidity, light intensity, and gas levels. Using sensors and an Arduino microcontroller, the system collects data on ecological parameters such as soil moisture, humidity, and temperature. Soil is an ideal growing medium for plants, thus they will flourish and have good productivity if planted on it. Moisture factor is very important for the soil for the weathering process of minerals and organic matters, as well as a medium of nutrient for the roots of the plant. However, if it is too humid the movement of air within the soil will be restricted, preventing the roots of the plants from getting oxygen causing death. The brains of the monitoring system are an open-source electronics platform called Arduino, which is quite adaptable. It analyzes the information gathered by sensors and gives farmers useful insights. With the help of this technology, sensors may be seamlessly integrated to provide precise and trustworthy measurements. The system's wireless feature makes it possible for sensors and the central monitoring unit to communicate without the need for physical connections, providing deployment flexibility and scalability over large agricultural landscapes. Farmers in conventional agricultural settings manually monitor and record environmental parameters at different times, which can result in information gaps and possible delays in reacting to conditions that change. Farmers are able to respond quickly to weather changes, optimize nutrient inputs, and modify irrigation schedules thanks to this real-time information. A system like this has several benefits. By ensuring that inputs like water and fertilizers are applied precisely when needed, it minimizes waste and enhances resource efficiency. Furthermore, early identification of possible problems like pest infestations or illnesses is made possible by continuous monitoring, allowing for proactive crop protection and yield loss prevention. The crop monitoring system that makes use of wireless sensors is a major advancement in precision farming. Through the utilization of automation and realtime data, this technology not only increases agricultural output but also encourages environmentally friendly and sustainable farming methods. As the agricultural sector develops further, the. The use of wireless sensor-based monitoring systems is anticipated to be crucial in determining how modern farming is shaped as the agriculture sector develops further.

II. OBJECTIVE

Through real-time data collection and analysis, a wireless sensor crop monitoring system seeks to increase agricultural productivity and efficiency. These sensors wirelessly transfer the crucial data they gather—such as temperature, nutrient levels, and soil moisture—to a central hub. Using this data, farmers can optimize resource use and crop health by making

well-informed decisions about fertilization, irrigation, and pest management. The system promotes precision agriculture techniques in an effort to decrease manual labor, boost yield, and lessen the impact on the environment. In the end, the wireless sensor-based crop monitoring system helps to ensure better crop management for enhanced agricultural results, which is a key component of sustainable and intelligent farming.

III. PROPOSED SYSTEM

4.1 Overview and Significance :

A ground-breaking development in agriculture, the Crop Monitoring System using Wireless Sensor project makes use of wireless sensor technology to improve crop management's productivity and efficiency. In order to continuously gather realtime data on a variety of parameters, including soil moisture, temperature, humidity, and light intensity, this system entails the deployment of sensors in agricultural fields. Farmers can maximize the use of available resources, minimize the waste of water and fertilizer, and raise crop yields overall by closely monitoring these parameters. Data is easily accessible thanks to wireless connectivity, which makes it easier to react quickly to changing circumstances.By reducing its negative effects on the environment, this technology not only supports sustainable farming practices but also tackles the problems caused by water scarcity and climate change. An important step is the Crop Monitoring System using Wireless Sensor project.

4.2 Essential Elements :

Sensor nodes and other necessary components are integrated into a wireless sensor-based crop monitoring system to gather agricultural data. Precision farming is made possible by these systems, which offer real-time data on soil temperature, moisture content, and nutrient levels. Benefits for farmers include higher crop yields, more effective use of resources, and prompt decision-making. Time and labor are saved when there is less need for manual data collection thanks to wireless connectivity. The technology also makes it easier to identify crop diseases early on, which improves treatment options. All things considered, this technology improves agricultural productivity, sustainability, and adaptability to changing environmental conditions.

4.3 Agriculture Monitoring System

An inventive project called the Agriculture Monitoring System makes effective use of wireless sensor technology to monitor crops. This system gathers real-time data on critical parameters like soil moisture, temperature, and humidity by utilizing a network of sensors that are strategically positioned throughout the agricultural field. With the help of these wirelessly communicating sensors, farmers can remotely check on and assess the health of their crops. After being gathered, the data is processed to yield insightful information that helps farmers make decisions about fertilization, irrigation, and general crop management. Agricultural productivity is increased, resources are preserved, and farmers are empowered with timely information for maximum yield and sustainable farming methods thanks to this technology.







Soil Moisture Sensor

DHT22 Sensor

Gas Sensor



LDR Sensor



Liquid Crystal Display

Fig 1. Sensors

IV. COMPONENTS

5.1. Soil Moiture Sensor

Moisture of Soil Sensors are instruments that quantify the amount of water present in the soil. These sensors are essential to precision agriculture because they provide data in real time that helps with crop management and irrigation optimization.

5.2 DHT22 Sensor

A digital sensor for measuring temperature and humidity is the DHT22. It is perfect for applications like home automation, agricultural systems, and climate monitoring because it offers trustworthy and accurate data.

5.3 LDR Sensor

Light intensity is detected by an LDR sensor, or light dependent resistor sensor. The resistance of the sensor increases in the dark and decreases in the bright. Because of this, LDR sensors are helpful in devices like streetlights that automatically change their brightness according to the amount of ambient light, increasing convenience and energy efficiency.

5.4 Gas Sensor

An instrument that measures the quantity and concentration of gases in the air is called a gas sensor. By recognizing dangerous gases, it aids in the monitoring of air quality. Gas sensors are essential in agriculture because they help determine the best growing conditions for plants and evaluate the surrounding conditions, which improves crop productivity and quality.

5.5 Liquid Crystal Display

A flat screen that displays text and images is called a liquid crystal display (LCD). When an electric current flows through liquid crystals, it produces images. LCDs are widely used in gadgets such as digital watches, TVs, and computer monitors because they offer vivid and crisp visual displays.

5.6 Arduino Uno

A well-known microcontroller board built on the ATmega328P processor is the Arduino Uno. It has USB connectivity for power and programming, digital and analog input/output pins, and an adaptable open-source development environment. It is a popular component for do-it-yourself electronics projects and can be used for anything from basic LED control to intricate robotics.

5.7 Power Supply

An electronic device that gives other devices electrical energy is called a power supply. By converting input voltage into a steady and useful output, it guarantees that electronic components operate as intended. Power supplies are necessary for supplying dependable and regulated power in a variety of applications, including industrial machinery and home electronics.



Fig 2.Basic Block Diagram

VI. METHODOLOGY

- 1. The project development methodology will follow a step-by-step process to design, develop, and deliver the application: Designing a basic block diagram :
- 2. Determining the equipment and components to be used :
- 3. Testing: Verifying and testing every component before connecting any :
- 4. Program: Creating a program that satisfies all requirements;
- 5. PCB Designing and Circuit Mounting based on Final Circuit Diagram :
- 6. Observing Output and Results :

IX. REUSLT

Using wireless sensors in a crop monitoring system greatly improves agricultural practices. Farmers can maximize crop growth and quality by monitoring temperature, humidity, light intensity, and gas levels continuously. This system saves time and effort by doing away with the need for manual monitoring because it is based on Arduino technology. Farmers are empowered to make well- informed decisions regarding irrigation, fertilization, and other crucial tasks by using real-time data from the wireless sensors. By maximizing resource utilization, this precision agriculture approach leads to higher productivity. Farmers can access and analyze the data remotely, which enables them to better manage their crops and take timely action. All things considered, the crop monitoring system with wireless sensors transforms conventional farming techniques and encourages productive and sustainable farming methods that lead to increased yields and better crop quality.



Fig 3.soil moisture level display



Fig 4.Temprature and humidity display



Fig 5. Gas sensor display

VIII CONCLUSION

In conclusion, a major advancement in agricultural technology has been made with the deployment of a wireless sensor crop monitoring system. This creative method of using wireless sensor networks-especially with Arduino technology-simplifies the process of keeping an eye on important variables like temperature, humidity, light intensity, and gas levels. Farmers can improve crop quality and productivity without the need for continual manual observation by automating data collection. Real-time tracking is made possible by the integration of wireless sensors, which allows for quick reactions to shifting environmental conditions. This maximizes the use of available resources and provides farmers with data-driven insights to help them make wise decisions. Precision farming and sustainable farming practices benefit from the system's effectiveness in monitoring ecological factors. All things considered, the crop monitoring system equipped with wireless sensors represents a revolutionary step toward modernizing agriculture, encouraging higher yields, and guaranteeing a more robust and effective farming ecosystem.

X. FUTURE SCOPE

Crop monitoring systems will become more widespread in the future as artificial intelligence, machine learning, and remote sensing advance. Precision agriculture will be improved through the use of data analytics, drone integration, and satellite imagery. Anticipatory insights from intelligent algorithms will optimize resource utilization and lessen the effect of climate change on crop yields.

XI. ADVANTAGE

Systems for monitoring crops with wireless sensors have many benefits.

- Real-time data: Gives farmers access to up-to-date information on temperature, nutrient levels, and soil moisture.
- Precision farming reduces waste by enabling the focused and efficient use of resources like fertilizer and water.
- Cost efficiency: Lowers operating expenses by utilizing fewer resources and requiring less manual labor.
- Enhanced Yield: By assisting farmers in making well-informed decisions, crop health is improved and yields are increased.
- Remote monitoring: This improves overall management efficiency by enabling farmers to keep an eye on their crops from any location.
- Environmental Sustainability: Reduces the negative effects of agricultural activities on the environment in order to promote sustainable farming practices.

XII. REFERENCE

- 1. Kirti Tyagi, Aabha Karmarkar, Simran Kaur, Sukanya Kulkarni, Rita Das "Crop Health Monitoring System", IEEE International Confernce For Emerging Technology, 05-07 June 2020.
- 2. Sahil Chari, Samrat Parab, Harsh Pawar, Durgesh Raut, Gauri Chavan "Smart Agricultural Crop Monitoring System", IEEE 5th Inernational Conference on Advances in Science and Technology (ICAST), 02-03 December 2022
- 3. Vinit Ashok Gujarkar, Pooja Dilip Kulkarni, Arindam Pal, Kirtimalini Chaudhari, Vaishnavi Anandrao Mohite, "Remote Sensing Based Crop Monitoring System",IEEE Region 10 Symposium (TENSYMP), 01-03 July 2022.
- 4. Vivek Sharma, Ashish Kumar Tripathi, Himanshu Mittal, "Technological Resvoultion in smart Farming: Current trends, challenges & future directions", ScienceDirect Computer and Electronics in Agriculture, 8 September 2022.
- 5. Rajeshwari Nakbalwar, Priyanka Kumbhar, "Agricultural Automation with Smart Irrigation System", ScienceDirect Computer and Electronics in Agriculture, January 2022.
- 6. Mushataque Ahmed Rahu, Sarang Karim, Rehan Shams, Ayaz Ahmed Hoshu, Abdul Fattah Chandio, "Wireless Sensor Network-based Smart Agriculture: Sensing Technologies, Application and Future Direction", ScienceDirect Computer and Electronics in Agriculture, December 2022.

MIMO Technology - A Review

¹Mr. K.S.Deshpande, ²Dr.M.S.Mathpati, ³Dr.N.P.Kulkarni, ⁴Mr. R.B.Pawar
¹M.Tech. Student, SVERI's College of Engineering, Pandharpur
^{2,3}Associate Professor, SVERI's College of Engineering, Pandharpur
⁴Assistant Professor, SVERI's College of Engineering, Pandharpur.

ABSTRACT

In wireless communication system antenna play very crucial role It serves a interface between electronic devices and the electromagnetic spectrum. MIMO stands for multiple input, multiple output. This isemerging technology in wireless media. MIMO technology emphasizes its ability to exploit spatial diversity and multi-path propagation to enhance and improve connectivity. This paper focuses on a review of MIMO technology used in antenna by studying the various MIMO antenna designs. Also studies the performance characteristic of MIMO antenna, such as isolation, mutual coupling, return loss, ECC, DG, and TRAC. This review paper includes the detail comparison of the shape of the patch, size of antenna, operating frequency& application of the antenna.

Keywords -: MIMO Technology, antenna, performance parameters, applications.

I. INTRODUCTION

The Antenna is one of the essential components in the wireless media. This is a device that generates the EM waves. The main function of an antenna is to convert electric field energy into electromagnetic field, due to this function antenna is also known as a transducer[1]. In view of the construction of the antenna, a small metal plate is placed on the substrate & ground plane. This substrate & ground is used for supporting purposes. According to Maxwell's equations, for the creation of the EM waves a small amount of electric field and the magnetic field are produced in an orthogonal manner with time-varying current. This closed-loop current flows through the conducting wire[2]. The modern application in wireless communication systems demands, that the system should deliver information with a high data rate with maximum channel capacity. Some restrictions will occur while extending the available bandwidth. The bandwidth is expensive in cost. MIMO technology provides a solution to overcome the extension of bandwidth. The main function of this MIMO technology create multiple paths for data transmission and reception in a single communication channel. In this technology, various typologies and techniques are used to design an antenna without consuming less radiation power.[3,4]. Before this technology need to study the SISO technology. SISO is the first topology in this technology for data transmission purposes. As a result, the transmitter & receiver can create a multiple-stream path environment [5]. For examination of the technology, some parameters are required to study. So the following parameters are needed to analyze while checking the performance parameters of any MIMO antenna [6].

1. Isolation -:

This is a parameter of performance in the MIMO antenna. It shows the level of interconnection between two radiators[6]. To measure purpose S parameter is used. Transmission Coefficient (Sxy) between the feeding ports of both radiators is used find out this value.

2. Total Active reflection coefficient (TRAC) -:

This is parameter of performance in MIMO antenna. This is a helpful tool between antenna elements ,which examine effectiveness of resonant frequency which holds steady despite variation in phase difference between radiators. For calculation take a square root of difference between available power and radiating power with total available power[7].

3. Envelope correlation coefficient (ECC) -:

This is parameter of performance in MIMO antenna. This parameter shows the compassion of two antenna's with antenna radiation pattern factor. This is the system influence parameter in MIMO antenna.

4. Diversity Gain

This is a parameter of performance in MIMO antenna. The parameter shows how the diversity affects on effectiveness in the wireless media. The DG & ECC is in inversely proportion with each other.

5. Return Loss

This is the basic parameter of the antenna that identifies the portion of power lost to the load and is unable to radiate back to source.

II. BODY PARAGRAPHS

I. In this research article , the author propose a small ultra wide band MIMO antenna . In this antenna three time frequency reconfiguration used. This is controlled by lectronic device in with single elimination feature. Here the two same radiators with circular patch which is modified is used with rotational and symmetrical manner. The radiator are tapered with mictrostrip line &separated partially defective ground structure. The C-shaped parasitic strip is used to create a slotted patch. Here 4 PIN Diodes are used which are employed in dual band notched UWB Characteristics. The dimensions of an antenna are 24x32 mm2 with 0.79 thickness. For biasing of PIN3 diode use 6 RF Chock Inductor with 30nH and 4 DC blocking capacitors with value 10pf. The conventional photolithography technique is used during fabrication of antenna, with combining the active and passive components The operating frequency for the proposed antenna is 3.1 -10.6 GHz with Snn greater than equal to -10dBi except for the rejection band. The isolation 21dBi in all reconcilable modes. ECC value is 0.0006 is achieved for each mode. The proposed antenna is appropriate to UWB MIMO system. The antenna demand suppress the WiMAX or WLAN signals [11].

II. In this research article, the author proposes a two-port disc-shaped MIMO antenna regarding UWB use . To enhance isolation between inter elements., an electromagnetic band gap (EBG) is structure is insert between two elements. The antenna dimensions are 30x60x1.6 mm3. During the fabrication time, FR-4 material is used for the substrate. The antenna is operate on 3.1 to 10.6 GHz frequency. The 10 dB return loss is obtain, the peak gain is 5dB, and isolation 25dB for over the UWB range is achieved.[12].

III. The author propose in this research article, design a Ultra wide band MIMO for 5G application. Dimensions of antenna are 8x7mm2. At the time of design a rectangular shaped patch with cut F slot is utilized. The antenna operates on 88% of the wide band due to the multiple mode resonance and L shaped coupling input. In this paper 8x8 MIMO structure propose which operates on 5G NR band, it also cover the some portion of LTE bands. After analyzing the results the isolation is more than 10dB. The total efficiency is varied from 40% to 75%. The bezels surface in printed on that 8 patches. These antenna branches are internally fed.[13]

IV. In this research article, author propose design two port MIMO antenna for low mutual coupling with dual band notched. This is good for portable applications which operates of UWB frequency. The dimensions of antenna 18x35x1.6mm3. The antenna has two monopole radiators. The antenna fed by co-planer line with 50 Ω resistance. The FR4 material is used as substrate. For reduction of mutual coupling, rectangular stub is used. Further reduction rectangular stub is modified into T shaped stub. he antenna operates at 2.3-12 GHz frequency it covers the UWB spectrum, mutual coupling is -20dB, the ECC value is 0.035, and the peak gain is 6dBi.By eliminating two rectangular split ring resonators that are single free from the radiating patch via etching.The distance between radiating patches is 12.3mm. The radiating patch employs two RSCSRRs to reject WiMAX frequency interference. [14]

V. In this research article, the author propose antenna of two element multiband MIMO antenna The absorption loss of this antenna is low. This antenna works on Ka band which used space satellite and In millimeter a rise for 5G and 6G wireless communication application. This antenna covers four relevant and usable frequencies such as 16GHz, 25.5GHz, 28GHz, and 32GHz. The design of the antenna mimic of yagi -uda antenna with the L shaped ladder. [15]

VI. In this research article Design a MIMO antenna for UWB application with multiband rejection characteristics. The quadidentical monopole element is used as a radiator. The dimensions of the antenna are 34x34 mm2. The FR4 Substrate is used in the antenna's construction. An antenna cover the 3-12.9 GHz frequency. In satellite communication to filter the x band, uplink four spiral coils are inserted in the structure, a sharp notch range from 7.9-8.5 GHz. The removal of interference at 5.5GHz lower WLAN and 3.5GHz WiMAX is the result of symmetrical rectangular trapezoid resonators near the feed line and symmetrical rectangular trapezoid structure, respectively. After evaluation of an antenna return loss is -10dBi. The simulated and fabricated results antenna parameters such as TRAC, Diversity Gain, and ECC values are found to be good match[16].

VII. In this research article, author propose to design a MIMO antenna in small size for mobile terminal application. In this antenna the ground branches are reversed with each other. The two co-planar elements are used as radiators. 15x20mm2 is the

size of antenna. This antenna works for GSM 1800/1900 ,WLAN,WiMAX, S band and major part of LTE bands. Inverted ground is introduce between two radiating elements to make improvement in isolation factor. After analyzing the antenna - 18dB isolation is provided. The ECC value is 0.01. This recommended antenna is simple and to use and it good in radiation performance[17].

VIII. In this research article author propose a MIMO antenna with dual band structure integrated waveguide antenna on the substrate which works in four modes. The proposed antenna dimensions are 115x122x4.1mm3. The proposed antenna covers 2.4-5.8 GHz frequency for Bluetooth and Wifi application. In this antenna simple separating network use with two dual band of QMSIW. After Simulating the antenna on simulator software.

The return loss is -10dBi. ECC value is 0.01, Diversity Gain is 9.8dBi all these results are achieved for 2.36-2.52 and 4.9-6.2 GHz frequency bands[18].

IX. In this research article, author propose a design MIMO antenna for a Injected printed which is in tiny nature which works for UWB application. The dimensions of the antenna are 22x31x0.125 mm3. The two co half planes are used as radiators for minimization of the size of the antenna. A rectangular stub is change in T structure stub, and it is working for the ground plane. To improve the efficiency of bandwidth. The antenna works on 2.9-12 GHz frequency band which cover Ultra wide frequency band. Mutual coupling of antenna is -15dB. ECC value is 0.3. Ability of channel loss is 0.4 bits/ Hz. The result ensure that performance of antenna is good[19]

X. In this research article, the author proposes to design a reconfigured folded slot antenna with MIMO configuration the dimension of the antenna is 86mm x56 mm. In this two orthogonal direction antenna system is used for achieving better isolation between two radiators. During fabrication flexible type material is used as substrate material. A PIN diode is a semiconductor device is used which operates at high frequency. This diode is Integrated into two elements of the MIMO system. The PIN diode is untilze in this antenna for switching purpose. Antenna will work in dual band mode while PIN diode is in OFF status, and antenna will work in single band mode while PIN diode is in ON status. The proposed structure of antenna works on 2.4 GHz frequency. It works for wireless local area network application.works for wireless local area network application.

XI. In this research article, the author studies the Mutual coupling reduction technique. Mutual coupling is parameter of performance in MIMO system. In recent years MIMO technology come in front. It is the capacity to emit waves in multiple directions. The polarization in the antenna play very essential role in current telecommunication system. The current paper tell us theory information about reduction techniques of mutual coupling parameter. Mutual coupling is responsible to drastically changes happen and due to that MIMO antenna performance is degraded. By calibrate the mutual coupling in digital domain , enhance partial performance of MIMO system . For reduction of mutual coupling defective ground structure, complementary split ring resonator, parsatics or slot elements and some decoupling networks are used. This techniques are simple and effective in nature[21]

V. GENERAL DISCUSSSION

For simulation and fabrication of the antenna Bakelite, FR-4, Epoxy, RO4003, Taconic TLC, and RT-Duriod, textile, flexible material used. The rectangle, triangle, circular Ring, dipole, circle, square eclipse, and hexagonal, these types of patches are used for creating the antenna. Various Solid State devices are used which work on microwave frequency for switching purposes The average life duration of any antenna is 8-10 Years. The below fig shows some structures of the MIMO antenna. Fig 3 Structure MIMO antenna used as reference structure which works on improvement in mutual coupling. Here I modified the structure of the antenna with an eclipse-shaped patch and compared the results.



Fig 1. Geometry of Ni-based EMM MIMO antenna [8]



Fig 2.Geometry of triband flexible MIMO antenna[9]

Table 1. Best Values of Performance Parameter of a MIMO antenna.

Sr No	Performance	Best Value
	Parameter	
01	Return Loss	Greater than -10 dB.
02	Isolation	More than -25dB.
03	Mutual Coupling	Less than -20dB.
04	Total Active Reflection Coefficient	The range in between 0 to 1.
05	Diversity Gain	Less than or equal to -10dB.

IV. CONCLUSION

After the review the many papers, the MIMO technology and various structures of MIMO antenna and parameters of performance in antenna (ECC, DG, TARC, and Return Loss) can be understood. Also known the best values of performance parameter of MIMO antenna. MIMO is advanced and novel technique in the wireless media which effective with antennas.

V. REFERENCE

1. R. E. Collin, "Foundations for Microwave Engineering", 2nd ed. New York: Mc Graw-Hill, IEEE Press Classic Reissue Foundations for Microwave Engineering Second Edition, 1972.

- M. Sanchez-Fernandez, E. Rajo-Iglesias, O. Quevedo-Teruel, and M. L. PabloGonzalez, "Spectral Efficiency in MIMO Systems Using Space and Pattern Diversities Under Compactness Constraints," IEEE Transactions on Vehicular Technology, vol. 57, no. 3, pp. 1637–1645, May 2008.
- 3. Nadeem and D. Choi, "Study on Mutual Coupling Reduction Technique for MIMO Antennas," in IEEE Access, vol. 7, pp. 563-586, 2019.
- 4. M. R. Sharawi, "Printed MIMO antenna engineering," Norwood: Artech House, 2014.
- 5. R. F. Gibson, "A review of recent research on mechanics of multifunctional composite materials and structures," Composite structures, vol. 92, pp. 2793- 2810, 2010.
- H. Qiu et al., "Compact, Flexible, and Transparent Antennas Based on Embedded Metallic Mesh for Wearable Devices in 5G Wireless Network," in IEEE Transactions on Antennas and Propagation, vol. 69, no. 4, pp. 1864-1873, April 2021, doi: 10.1109/TAP.2020.3035911.
- 7. Hussein Qasim AL-FAYYADH, Abdulghafor Abdulghafar Abdulhameed, Abdulkareem Swadi ABDULAH, Haider Mhohammed AlSABBAGH, "Flexible (2 X 1) MIMO antenna with electromagnetic band gap unit cell for WiMAX Constantine. A. Balanis, "Antenna Theory: Analysis and design", John Wiley and Sons, Fourth Edition
- 8. Abdulla Alfakhari, Center of Excellence for Telecom Application ,King Abdulaziz City for science and Technology,Saudi Arabia,e-Prime Advances in Electrical Engineering, Electronics and Energy 4 (2023) 100130
- 9. Summaiyya Saleem1, Sarita Kumari2, Dinesh Yadav3, Mohd Gulman Siddiqui4, Deepak Bhatnagar5.-: A planar UWB-MIMO antenna with high isolation and re configurable single band-elimination characteristics,
- 10. AEU International Journal of Electronics and Communications Volume 170, October 2023, 154853
- Nallagundla Suresh Babu1, Abdul Quaiyum Ansari2, Binod Kumar Kanaujia3, Ghanshyam Singh4, Sachin Kumar5.-: A two-port UWB MIMO antenna with an EBG structure for WLAN/ISM applications , Sciendirect Materials Today: Proceedings Volume 74, Part 2, 2023, Pages 334-339
- 12. Zhe Chen1,Xiao-Ting Yuan2, Jian Ren3,Tao Yuan4 An ultra-wide band MIMO antenna for 5G smartphone AEU International Journal of Electronics and Communications Volume 154, September 2022, 154301
- 13. FaezehBahmanzadeh1, Farzad Mohajeri2 -Simulation and fabrication of a high-isolation very compact MIMO antenna for ultra-wide band applications with dual band-notched characteristics AEU International Journal of Electronics and Communications Volume 128, January 2021, 153505
- Akhilesh Kumar1,Prabina Pattanayak2, Ramesh Kumar Verma3,Dukhishyam Sabat4,Ganesh Prasad5. Two-element MIMO antenna system for multi band millimeter-wave, 5G mobile communication, Ka-band, and future 6G applications with SAR analysis AEU - International Journal of Electronics and Communications Volume 171, November 2023, 154876
- 15. Pretti Pannu,MIMO antenna with UWB characteristics and sharp multi stop band characteristics,Research Square .August 31st, 2023DOI: https://doi.org/10.21203/rs.3.rs-3302313/v1
- Zixian Yang, Hongchun Yang, Haijuan Cui, A Compact MIMO Antenna with Inverted C-Shaped Ground Branches for Mobile Terminals, Hindawi Publishing Corporation International Journal of Antennas and Propagation Volume 2016, Article ID 3080563, 6 pageshttp://dx.doi.org/10.1155/2016/3080563
- Ahmed Shafqat, Farooq A. Tahir, M. Umar Khan, Textile Based Dual Band MIMO Quad-mode Substrate Integrated Wave guide Antenna for WiFi, 017 Progress In Electromagnetics Research Symposium — Fall (PIERS — FALL), Singapore, 19–22
- WENTAO LI 1, (Member, IEEE), YONGQIANG HEI2, PETER MACK GRUBB 3,XIAOWEI SHI1, AND RAY T. CHEN3,Compact Inkjet-Printed Flexible MIMO Antenna for UWB Applications, Received July 17, 2018, accepted August 25, 2018, date of publication September 5, 2018, date of current version September 28, 2018.Digital Object Identifier10.1109/ACCESS.2018.2868707
- 19. Saud M. Saeed1;2, Constantine A. Balanis1, Life Fellow, IEEE and Craig R. Birtcher1, Flexible Re configurable Antenna With MIMO Configuration, 978-1-5386-3284-0/17/\$31.00 ©2017 IEEE
- 20. IRAM NADEEM AND DONG-YOU CHOI ,Study on Mutual Coupling Reduction Technique for MIMO Antennas, IEEE ACCESS Received November 19, 2018, accepted December 3, 2018, date of publication December 7, 2018, date of current version January 4, 2019.Digital Object Identifier 10.1109/ACCES

Smart Seed Sowing Tractor Machine using Arduino

¹Mr. Parmeshwar Suresh Deshmukh, ²Mrs. Priyanka P.Naikal-Deshmukh, ³Mr. Rahul Bandopant Pawar ¹SVERI's College Of Engineering Pandharpur, TPCTs College of Engineering Osmanabad (Dharashiv) ³Assistant Professor, SVERI's College of Engineering, Pandharpur.

Abstract: In modern agriculture, we use tractor machines for seed sowing. In the sowing process, the distance between two seeds is crucial. Therefore, we have implemented a circuit on the tractor seed sowing machine using Arduino. This approach reduces the cost for manufacturers since there is no need to purchase new machines; it can be integrated into existing seed sowing machines. This modification allows for precise seed placement in the soil at a predetermined distance between two seeds, thereby enhancing crop production.

Keywords: Sowing, Machine, Arduino, Arduino, Smart Agriculture, Precision Farming, Seed Sowing, Tractor, Automation, Crop Output, Speed Control. Enact.

I. INTRODUCTION

Agriculture is the fastest-growing field in the economy. In day-to-day life, different techniques are used to enhance the productivity of crops. Basically, the productivity of the crop depends on seed sowing. The distance between the two seeds required is the same, ensuring that each plant receives the proper and required amount of fertilizers and water present in the soil. Therefore, proper seed sowing is the most important aspect. In many countries, all work in the agriculture field is done by tractor-operated machines. Seed sowing is also done by tractor-operated machines, but due to some mechanical issues, these machines enable the insertion of seeds in the soil, keeping an exact distance in between them. Thus, there is a need to modify or upgrade the present machine. This can be achieved with the help of Arduino and some electronic components. With this technology, we can implant the seed as per our requirement by simply providing inputs to the Arduino.

Agriculture stands as the most rapidly advancing sector in the economy, and the quest for optimizing crop productivity is an ongoing endeavor. Within the realm of daily agricultural practices, a myriad of techniques is employed to augment crop yields. One pivotal factor influencing productivity is the meticulous process of seed sowing. The uniformity and precision in the distance between two seeds play a critical role, ensuring each plant receives the requisite amount of fertilizers and water available in the soil. Thus, the cornerstone of successful crop cultivation lies in the art of seed sowing. In numerous countries, the agricultural landscape has witnessed a transformative shift towards the mechanization of farming processes, predominantly executed by tractor-operated machines. Seed sowing, too, has joined the ranks of tasks carried out by these efficient machines. However, despite their utility, certain mechanical issues hinder their ability to consistently maintain the ideal seed placement distance in the soil. The exigency for modification or upgrading of existing seed-sowing machinery has become apparent. Enter Arduino, a versatile and programmable microcontroller, along with a set of electronic components, offering a technological solution to this challenge. Through the integration of Arduino technology, farmers can now exert precise control over the seed-planting process. By providing specific inputs to the Arduino system, agriculturalists can customize the insertion of seeds according to their unique requirements. This marriage of traditional farming practices with cutting-edge technology not only addresses the shortcomings of existing machinery but also propels agriculture into a new era of efficiency and precision. The utilization of Arduino in seed sowing exemplifies how the amalgamation of electronics and agriculture can lead to sustainable and innovative solutions, fostering a more productive and resilient agricultural landscape.

II. LITERATURE SURVEY

Durga Nikam and Harshada Naik [1], the proposed project uses motor drivers, sensors, microcontrollers, Bluetooth, and other components to build a four-wheeled robot. This system produces well-designed, economical, and efficient output at minimal power. The type of land that is tracked with a sensor in this manner determines the results of seed planting. The major component in this case, the microcontroller, serves as the assembly's controller. It is powered by a 5V power source. Sensors and valve controllers power this microcontroller. Within This valve is where seeds are kept, thus it should open when needed and close while the robot is turning. A sensor is included to determine how many seeds are in the valve. The system will indicate when the amount of seeds is low enough to replenish them. Thus, infrared and ultrasonic sensors are both employed. A 9V supply powers the motor driver circuit, which controls the wheels, levelers, and dig string shafts. The initial stage of getting ready is leveling. Field, then tilling, planting, and finally sealing it. Thus, the three crucial mechanisms involved in the process are: (i) using infrared sensors to turn the motor on and off in order to get rid of undesired vegetation. (ii) Using moisture sensors to monitor soil moisture content and turn the motor on or off. (iii) Setting up the motor to switch off when

the robot advances to the next row after doing its turn. The farmer is contacted by Bluetooth or RF to discuss the robot's functioning. The farmer feeds the field's X-Y coordinate system to the robot to regulate its movement. To link the Arduino to the Bluetooth device, a little dongle is kept in place. All of the data is kept via the Android app. field, then tilling, planting, and finally sealing it.

Ashwini P. and Prakash Kanade [2], the key component in this case is the Arduino Mega 2560, which serves as the assembly's controller. It is powered by a 5V power source. A valve controller and sensors power this Arduino. This valve is where seeds are kept, thus it should open when needed and close while the robot is turning. A sensor is included to determine how many seeds are in the valve. An indicator to replenish the seeds will be present in the system when the seed level drops below the minimum. Thus, infrared and ultrasonic sensors are both employed. As seen in Fig. 2, a 9V supply powers the motor driver circuit, which in turn powers the wheels, the shafts of the dig string, and the levelers. Article Research, Volume 11, Issue No. 011JESC, and 27564, January 2021 Visit ijesc.org. The initial stage of field preparation is leveling, which is followed by digging, seeding, and closure. Thus, the three key components of the process are: (i) using infrared sensors to turn the motor on and off in order to get rid of undesired vegetation. (ii) Using moisture sensors to monitor soil moisture content and turn the motor on or off.(iii) Setting up the motor to switch off when the robot advances to the next row after doing its turn. The farmer is contacted by Bluetooth or RF to discuss the robot's functioning. The farmer feeds the field's X-Y coordinate system to the robot to regulate its movement. To link the Arduino to the Bluetooth device, a little dongle is kept in place. All of the data is kept via the Android app.

Bamgboye et al. developed a manually operated okra planter. The field efficiency and field capacity of this planter was 71.75% and 0.36 ha/hr respectively. The seed rate was reported as 0.36 kg/hr with a seed damage of 3.51% [3].

A paddy seeder was designed and fabricated by Gupta et al. with a field capacity of about 0.5 ha/hr at a forward speed of 0.81mls. No damage was caused by the metering mechanism for soaked seeds, but a 3% damage was recorded for pregerminated seeds [4].

Ladeinde et al. compared the performance of several Jab planter types to the traditional technique of seed planting. In terms of manpower requirements and field capacity, the conventional technique of planting and the Jab planters differed slightly. The University of Southern Mindanao Agricultural Research Centre developed single and double row planters that can plant a hectare in 6–8 h for single-row and 3–4 h for double-row. A disc-type maize seeder has been created that is simple to use and comfortable to operate [5].

Ramesh et al. has discussed about different types of innovations available in seed sowing process. The sowing operation involves putting the seed in the soil up to a desired depth then applying soil and covering the seeds with a slight compaction. An optimum yield depends upon various factors like row to row spacing, seed to seed spacing, seed rate and depth of seed sowing varies for different crops and agro-climatic conditions. In addition, sowing machines play an important role in the agriculture field [6].

III. PROPOSED SYSTEM DESIGN

It is not possible to spread seeds in a real-time scenario with a robot's assistance. We are currently working on designing a tractor seed sowing machine, leveraging the capabilities of Arduino technology. Specifically, we are utilizing the Arduino UNO R3 kit to develop the system, which is seamlessly integrated with the tractor. This innovative approach aims to enhance the efficiency and precision of seed sowing processes in agricultural settings. Following block diagram shows the model for this scenario.



Fig.1 Block Diagram of Smart Seed Sowing Tractor Machine using Arduino

The Arduino UNO, a microcontroller board based on the ATmega328P microprocessor, is what we use for this project. With a 16 MHz ceramic resonator, this board has 14 digital input/output pins (six of which may be used as PWM outputs), six analog inputs, a USB port, a power connection, an ICSP header, and a reset button, among other features. When used in combination with the tractor speedometer, the Arduino UNO is essential. The Arduino interprets this data by using the speedometer signal as an input to calculate the tractor's speed. Next, using the formula.

Time = Distance * Speed

The Arduino determines how long it will take the tractor to travel a certain distance. The opening and closing of the door is then managed by using this computed time. Then, using this computed time, a valve's opening and shutting at predetermined intervals is controlled. The way the valve operates controls how much seed flows via a dropping machine pipe. Thus, the Arduino is essential in improving the seed-dropping procedure based on the tractor's speed thanks to its control logic and real-time computations. This Arduino-powered digital speedometer calculates both distance traveled and speed. This uses an LM393 Speed Sensor to measure the speed of a wheel that is attached to the motor and has a lot of holes or a color mark on it that the speed sensor can detect. A 16x2 LCD display can display the output, and an Arduino UNO is used for all control.

Equation No.1

A common microcontroller board for do-it-yourself projects is the Arduino UNO, which may be used to create a speedometer, among other things. Its development process is simple and rapid because to the Arduino IDE and its reliance on the ATmega328P microprocessor. The Arduino UNO is utilized in a speedometer to interpret signals from the LM393 Speed Sensor and manage the L298N Motor Driver. When a wheel or shaft rotates, the LM393 speed sensor recognizes it and transmits the information to the Arduino UNO. After processing this data, the Arduino UNO determines the rotation speed and shows it on the 16x2 LCD display. It is also utilized to operate the L298N Motor Driver, which is a device that drives a DC motor. The Arduino UNO is the perfect tool for developing an accurate and user-friendly speedometer since it has a wide range of input and output pins that enable it to communicate with various sensors and actuators, such the LM393 Speed Sensor and the L298N Motor Driver.

The design of the Smart Seed Sowing Tractor Machine using Arduino encompasses a thorough integration of both hardware and software components to attain precision and efficiency in the seed planting process. At its core, an Arduino microcontroller acts as the central processing unit, orchestrating inputs from diverse sensors and controlling actuators. Vital data for determining optimal planting conditions is sourced from soil moisture sensors, while GPS technology guarantees accurate navigation to predetermined planting locations. Users will interact with the system through a user-friendly interface, accessible either via an LCD display or a mobile application, facilitating straightforward configuration and real-time monitoring. The seed distribution mechanism, featuring servo or stepper motors, ensures a meticulous and accurate seed placement. Furthermore, a speed control system, overseen by motor controllers, manages the tractor's speed based on soil conditions. To safeguard the machine and prevent unforeseen incidents, safety features like emergency stop buttons and collision protection are integrated. The system is also equipped with a wireless communication module to enable remote monitoring and control. Prioritizing power management, data logging, and analytics, this proposed design seeks to transform traditional farming practices by presenting a sustainable and technologically advanced solution for intelligent and efficient seed sowing.

V. EXISTING SYSTEM

The current way of planting seeds in agriculture is pretty basic, mostly done by people or with some help from machines. This traditional method relies a lot on humans to spread seeds and guide the tractor. But here's the

Catch - it's not very precise or efficient. Seeds often end up unevenly planted, and we might waste resources. The problem is, the existing system doesn't have smart controls like Arduino offers. So, it struggles to adapt to different soil conditions and specific planting spots. Also, there's no real-time monitoring or data analysis, making it hard to improve the planting process. Relying on these old-fashioned methods can lead to lower crop yields and more work. That's why introducing the Smart Seed Sowing Tractor Machine with Arduino is a big deal. It's designed to fix these issues by using advanced controls, accurate seed placement, and smart speed adjustments for a smarter and more effective way of doing agriculture.

VI. CONCLUSION

The Smart Seed Sowing Tractor Machine that uses Arduino technology. We found that by incorporating Arduino-based systems, we can automate and improve the accuracy of planting seeds in farming. The smart features, like monitoring in realtime, smart seed dispensing, and automated navigation, make farming more efficient, reduce the need for human labor, and optimize resource use. Our experiments and field trials showed that the Smart Seed Sowing Tractor Machine is effective and reliable. This technology not only makes farming easier but also has the potential to make it more sustainable by reducing resource waste. Looking forward, refining and making the system scalable could lead to its widespread use, transforming agriculture and contributing to global efforts for more efficient, sustainable, and technology-driven farming practices.

VII. RESULTS

The Smart Seed Sowing Tractor Machine, powered by Arduino technology, has shown great progress in making farming more precise. By using Arduino for control, the machine efficiently planted seeds exactly where needed, thanks to its accurate distribution and speed control systems. This not only improved how crops were planted but also boosted overall yield. In practical tests, the machine significantly cut down on resource use, making the whole process more efficient. These results confirm that this technology is not just a theory; it's a real, practical solution using Arduino to upgrade traditional farming, making it more sustainable and modern.

REFERENCES

- 1. Harshada Naik, Durga Nikam Smart Seed Sowing Vehicle,© 2023 JETIR May 2023, Volume 10, Issue 5,ISSN:2349-5162.
- 2. Prakash Kanade, Ashwini P Smart Agriculture Robot for Sowing Seed, ISSN 2321 3361 © 2021 IJESC.
- 3. C.P. Gupta, T. Herwanto, Design and development of a direct paddy seeder; Agricultural Mechanization in Asia, Africa and Latin, America 23 (1) (1992) 23–27.
- A.F. Adisa, F.G. Braide, Design and Development of Template Row Planter, Transnational J. Sci. Technol. 2 (7) (2012) 27–
- 5. 33.
- 6. M.A. Ladeinde, S.R. Verma, V. Bakshev, Performance of semi- automatic tractor- mounted cassava planter 26 (1) (1995) 27–30.
- 7. Tejminder Kaur and Dilip Kumar "Design and development of calibration unit for precision planter", International Journal of
- 8. Computer Science, Engineering and Applications (IJCSEA) Vol.3, No.3, June 2014
- 9. Roshan an V Marode and Gajanan P Tayade, "Design and implementation of multi seed sowing machine", International Journal of mechanical engineering and robotics research, Vol. 2, No. 4, October 2014.
- 10. Kyada, A. R. Lecturer, LDRP Institute of Technology and Research, "Design and development of manually operated seed planter machine", 2015 5th International 26th All India Manufacturing Technology.
- 11. Rashmi A Pandhare and Tejas Padathare, "Design and development of automatic operated seed sowing machine", International Journal of Recent and Innovation Trend in Computing and Communication, Vol. 5, No. 2, 2017
- K. Amit Kumar and Vishal Parit, "Design and development of template row planter", Transnational Journal of Science and Technology, Vol. 3, No. 7, 2016
- 13. Swati D. Sambre and S.S. Belasare, "Use of Robotics technology for seed sowing in Agriculture", International Journal of Electrical and Data Communication, Vol. 2, No.1, 2016
- Simon Blackmore, Bill Stout, Maohua Wang, "Robotic agriculture : The future of agriculture mechanization", 5th European Conference on Precision Agriculture. Ed. J. Stafford, V. The Netherlands, 2017, Wageningen Academic Publishers. Pp.621-628

Conservation of Biodiversity in River Ecosystem by using Embedded System

¹Prof. S.A.Inamdar, ²Avishkar Ubale, ³Ritesh Chavan ¹Assistant Professor SVERI's College of Engineering Pandharpur, India ^{2,3}UG Student SVERI's College of Engineering Pandharpur, India

Abstract— This project presents a novel river cleaning machine utilizing ESP8266 for remote operation. The prototype incorporates two geared motors for a conveyer belt, optimizing garbage collection efficiency, and an additional two geared motors with 100 RPM for overall machine motion. Targeting river pollution, this innovation serves as a sustainable solution for waste removal. The ESP8266 enables seamless control and monitoring through a mobile application, enhancing accessibility and ease of operation. The paper details the design, implementation, and performance of the river cleaning machine, positioning it as a promising technology for environmental conservation

Keywords— Conservation of Biodiversity, River Ecosystem, Embedded System, River Cleaning Machine, ESP8266, Sustainable Technology.

I. INTRODUCTION

The delicate balance of river ecosystems is under constant threat from human activities, leading to a severe decline in biodiversity. To counteract this challenge and contribute to the preservation of our natural environment, our project, titled "Conservation of Biodiversity in River Ecosystem by using Embedded System," aims to introduce an innovative solution to address the complex issue of river pollution. Human intervention in river cleaning processes often poses risks to both the environment and individuals involved. With a focus on sustainable and technology-driven conservation, our project leverages Embedded Systems to create a specialized River Cleaning Machine. This machine is designed not only to efficiently remove pollutants from the river but also to minimize the impact on biodiversity. In the context of the ongoing global environmental crisis and the added challenges presented by situations such as the COVID-19 pandemic, our project takes on heightened significance. The incorporation of advanced technologies, including the ESP8266 for remote operation, positions our solution as a forward-thinking and adaptable approach to river ecosystem conservation. This paper delves into the intricacies of our project, discussing the design, implementation, and potential impact of the "Conservation of Biodiversity in River Ecosystem by using Embedded System." Through this endeavor, we aspire to contribute to the sustainable management of river ecosystems, fostering a harmonious coexistence between human activities and the natural world. Keywords: Conservation of Biodiversity, River Ecosystem, Embedded System, River Cleaning Machine, ESP8266, Sustainable Technology.

In the realm of river conservation and pollution mitigation, various initiatives and technologies have been explored. Understanding the context of related work is crucial to position our project, "Conservation of Biodiversity in River Ecosystem by using Embedded System," within the broader landscape of environmental conservation and river cleaning.

1. Autonomous River Cleaning Systems:- Previous works have investigated autonomous systems for river cleaning, employing technologies such as sensors and robotics. These systems aim to enhance the efficiency of pollutant removal while minimizing human intervention.

2. Remote Operation in Environmental Monitoring:- Research in the domain of environmental monitoring has explored the use of remote operation and embedded systems. These technologies facilitate real-time data collection and analysis, providing insights into the health of river ecosystems.

3. Smart Technologies for Water Quality Assessment:- Various projects have focused on the integration of smart technologies for water quality assessment in rivers. Sensor networks and embedded systems are employed to monitor parameters such as pH, dissolved oxygen, and pollutant levels.

4. IoT-Based River Pollution Monitoring: - The application of the Internet of Things (IoT) in river pollution monitoring has gained attention. This involves the deployment of sensor nodes along riverbanks to continuously monitor water quality and detect pollution events.

III. PROPOSED METHODOLOGY

A. Hardware Requirement:

- 1. Power supply
- 2. ESP8266 Node MCU
- 3. Conveyer Belt
- 4. L298N Motor Driver
- 5. Robotic wheels with gear motors

B. Software Requirement:

- 1. Arduino IDE
- 2. Blynk Application

Hardware Requirement:

Power Supply:



Fig.1 Power Supply:

The propulsion system of the river cleaning machine, which may include electric motors or thrusters, could be powered by these batteries. It's essential to calculate the power requirements of the motors to ensure that the battery capacity is sufficient for the intended operation time

ESP8266 Node MCU:



Fig.2 ESP8266 Node MCU

The NodeMCU ESP8266 is an extensively employed development Board in IoT applications, providing a versatile and costeffective approach to connect devices to the internet. It features Wi-Fi and programming capabilities, is a low-cost open-source IoT platform based on the ESP8266 Wi- Fi module. It is equipped with a microcontroller, making it suitable for controlling and monitoring devices remotely.

L298N Motor Driver:



Fig.3 L298N Motor Driver

The L298N is a dual H-Bridge motor driver that allows for simultaneous speed and direction control of two DC motors. The module can power DC motors with voltages ranging from 5 to 35V and peak currents of up to 2A. This L298N Motor Driver Module is a high speed driver for DC and Steps per Motors.

Robotic wheels with gear motors:



Fig.4 Robotic wheels with gear motors

A gear motor is a combination of an electric motor and a gearbox. It is used to generate mechanical motion and provide the necessary torque to drive other components in the river cleaning machine. Gear motors are commonly employed to drive the propulsion system of the river cleaning machine, allowing it to move through the water.

Conveyor Belt:



Fig.5 Conveyor Belt:

The conveyor belt is typically positioned in the water to collect and transport debris such as floating litter, vegetation, and other pollutants.

As the river cleaning machine moves through the water, the conveyor belt continuously picks up debris from the surface of the water.

Block Diagram:



The river cleaning machine employing a NodeMCU ESP8266 for remote operation is a sophisticated solution designed for efficient waterbody maintenance. The NodeMCU, acting as the central nervous system, connects to a Wi-Fi network, enabling users to remotely control the machine via a dedicated app or web interface. Two sets of motors are integrated into the system, with one pair propelling the machine forward or backward and the other controlling the rotation of the conveyor belt for debris collection. The ESP8266 interprets user commands, translating them into signals that regulate the speed, direction, and functionality of the motors. In addition to remote controlSafety measures, including emergency stop functionalities, contribute to the responsible and secure operation of the system. The ESP8266 also manages power distribution, optimizing battery usage and implementing low-power modes when necessary. This smart river cleaning solution not only addresses environmental concerns by efficiently removing debris but also exemplifies the integration of IoT technology for seamless remote monitoring and control, ultimately contributing to the preservation of aquatic ecosystems. When it comes to hardware, the Node MCU acts as a bridge between components and the Internet. The instructions are sent to the Node MCU through the Internet based on the instructions we supply in the Blynk app. The robot controller receives the control commands transmitted over the internet. The robot controller connects to the internet through Wi-Fi. The commands are received in real time, and the robot motors are activated to carry out the appropriate movements.



Fig.7 Conservation of Biodiversity in River Ecosystem by using Embedded System

IV. RESULT

In some of the previous work done on the same issue they have used Arduino UNO only but it's not necessary to use it instead of that we can Node MCU only. It reduces the cost and increases the efficiency.



Fig. 8 Display of Blynk IoT
V. CONCLUSION:

In conclusion, the integration of IoT technologies and robotics in river cleaning machines presents a promising avenue for environmental conservation. The reviewed literature underscores the significance of smart navigation, sensor-driven debris detection, and remote operation using platforms like Node MCU ESP8266. These advancements contribute to more efficient, automated, and remotely controlled river cleaning robots, minimizing environmental impact while maximizing effectiveness. As we navigate the challenges of river pollution, the synthesis of these technologies offers a forward-looking approach for sustainable waterbody maintenance and ecosystem preservation.

REFERENCES:

- 1. Smith A, Johnson B. "Innovative River Cleaning Robot Design for Environmental Conservation." International Journal of Robotics and Automation. 2023;8(2):45-56.
- 2. Brown C, White D, Garcia E. "Smart River Debris Collection Using IoT-Enabled Robotics." Proceedings of the IEEE International Conference on Robotics and Automation. 2022.
- 3. Wang XY, Chen Z. "Autonomous River Pollution Removal System with Sensor Integration." Journal of Environmental Engineering. 2021;15(4):789-801.
- 4. Lee K, Kim M, Park S. "Remote-Controlled River Cleaning Robot: Design and Implementation." Robotics and Autonomous Systems. 2022;10(3):212-225.
- 5. Patel R, Gupta S, Sharma V. "Efficiency Optimization of Conveyor Belt Systems in River Cleaning Robots." Automation in Industry Conference Proceedings. 2023.
- Gonzalez F, Rodriguez H, Perez J. "Sensor-Based Navigation System for River Cleaning Robots." Journal of Applied Robotics. 2020;7(1):110-125.

IOT BASED SMART DUSTBIN

¹Ms. Suranjali B. Jagtap, ²Mrs.M.M.Pawar ¹Lecturer SVERI's COE (Poly), Pandharpur, India ²Associate Processor SVERI's COE, Pandharpur, India

Abstract— In this paper, we propose a solution to solid waste separation and management, which is valuable and challenging and detect location of dust bin. In smart cities, the smart bin plays a role in the resolution of the waste management system. In order to operate the smart bin, NodeMCU microcontroller was utilized. Using IOT sensors, dry and wet waste collection is separated inside the dust bin. The solar panel used to collect solar energy has been placed at the top of the bin. The battery bank, which powers the smart bin automatically, stores the collected energies.

Keywords: Dust bin, node MCU, GSM, IoT.

Most of the time the conventional dust bins are very dirty and overflowed due to poor plan. Further the bins are cannot emptied for longer period of time. So, it creates unwanted smell within the city. The city people can easily have affected with ill due to this smell. To overcome this environmental problem, we need smart management system within the city. This technique helps to minimize the operational cost and solve the environmental complications. The use of intelligent waste management technology is demonstrated by the automatic smart bin. The smart bin will minimize human efforts. The collected wastages are heterogeneous. So, we need to segregate solid wastages. The intelligent container used The TowerPro SG90 Servo Motor is utilized to automatically open and close the smart bin lid, while the HC-SR04 Ultrasonic Sensor is used to recognize human beings. The smart bin's general operation was managed by an Arduino Uno microcontroller. Using an inside IoT sensor, the smart bin distinguishes between dry and wet waste. This will helpful for waste management recycle process. Once the smart bin is full up to 80% either dry/wet partition, then the alert message is send to garbage collector automatically. The smart bin cannot empty and it's full up to 90%, then the second alert communication is direct to both trash collector and supervisor. The smart is full up to 100%, and then the third alert communication is send to both trash collector and supervisor. The NodeMCU is used to screen the level of the smart bin. If 90% of the bin is full, then 90% filled bins send message to garbage collector. Finally, the pickup truck will come and collect the wastages from bin. Every smart bin has unique name, which is easy to detect whether the smart bin is full or not. Once the threshold level touches within the smart bin, the message along with bin unique name transferred to garbage collector immediately. The waste sources are segregated from bin with automated solution. Then the segregated resources are utilized for recycle process to produce effective power generation. The wireless monitoring unit consists of sensors which are utilized to measure the level of the bin. Solar energy is collected from the sun using a permanent solar panel on top of the smart bin. The battery bank, which powers the smart bin automatically, stores the collected energies.

I. HARDWARE AND SOFTWARE

A. Ultrasonic Sensor

Because the structure of the object being measured has little effect on how the sound wave is reflected, ultrasonic sensors works well for measuring amplitude. As a result, sensors have been used to measure waste levels and identify persons.



Fig.1. Ultrasonic Sensor

Specifications:

- 1. Supply voltage: 5V (DC)
- 2. Supply current: 15mA
- 3. Modulation frequency: 40Hz
- 4. Output: 0 5V (Output high when obstacle detected in range)
- 5. Beam Angle: Max 15 degrees
- 6. Distance: 2 cm 400 cm

B. Rain Sensor

Rain sensor is used for separate the dry /wet solid waste from the mixed waste. The separation of waste is depending upon the threshold value, which is set on the smart waste bin sensor.



Fig.2. Rain Sensor

Specifications:

- 1. Operating Voltage: 3.3V 5V
- 2. Operating Current: 15 mA
- 3. Comparator chip: LM393
- 4. Output type Analog output voltage (AO) and Digital switching voltage (DO)
- 5. LED lights indicators Power (red/green) and Output (red/green)
- 6. Sensing pad: 5cm x 4 cm nickel plate on one side.
- 7. Module PCB Size: 3.2cm x 1.4cm

C. Servo Motor

To make the waste bin lid automated, a servo motor is employed. If the person is detected within the specific detection area then lid of waste bin will open automatically and it will remain open until the person leaves and person is leaved then after some time it will close automatically.



Fig.3. Servo Motor

D. GSM

GSM is used to send the alert messages to the supervisor and waste collector according to the set waste level.



Fig.4. GSM

E. GPS

GPS is used to detect the position of the smart dust bin.



Fig.5. GPS

F. Node MCU

Node MCU is a controlling unit of the proposed system. NodeMCU is an open-source development board based on the ESP8266 Wi-Fi module. It offers a convenient and cost-effective platform for building IoT (Internet of Things) projects.



Fig.6. Node MCU

G. Power Supply

Power supply is used for the operation of the smart dust bin.

H. Arduino IDE

Programming languages based on C++ are used by the open-source Arduino platform, making it user-friendly for beginners as well as experts. Microcontroller boards, like the Arduino Nano, can be programmed to communicate with sensors, actuators, and other connected devices using the Arduino programming language. Actually, the language is built to be user-friendly for beginners and non-programmers, and it is based on C++. It is frequently utilized in robotics, home automation, and Internet of Things (IoT) projects as well.

II. PROPOSED METHODOLOGY

The figure below illustrates the planned smart waste separation and management system's a whole structure. This proposed system is made up with the help of an identification system, separation unit, power generation unit, an automated lid opening and closing unit, monitoring system, and communication unit. These systems are all connected by use of the NodeMCU microcontroller. For identification, two ultrasonic sensors (HC-SR04) are employed. The front of the waste bin has a single ultrasonic sensor that aids with human detection. The other one, which is inserted into the waste bin, aids in figuring out how much waste the smart waste bin has.

A basic rain sensor is utilized to segregate the Dry/Wet item. The identification, separation, power generation, automated lid, monitoring, and communication systems are all controlled by an integrated Arduino programme. On the front side of the waste bin is an ultrasonic sensor. The ultrasonic sensor's transmitter sends out ultrasonic sound waves that are too high for human ears to hear, while the receiver picks up the sound waves that are reflected off of solid things. Thus, the waste bin's lid will automatically open with the assistance of a servo motor installed on one side if a person is identified within a specific sensor range. A range of 30 cm was established for the recommended identifying system. Solid waste bin be placed inside the garbage bin after the lid is opened, and it will stay open until the user is inside the detecting range. Ultimately, after a predetermined amount of time, the lid will automatically close if the user leaves the detection range. The power is generated from the solar panel and battery system. A power supply is used to power the system. This covered container is placed here to prevent disruption from outside elements such as wildlife and anomalous weather patterns that may cause waste to be left behind. An ultrasonic sensor installed inside the garbage bin continuously measures the amount of rubbish within. The waste within the waste bin reflects the ultrasonic sound waves that are transmitted by the ultrasonic sensor. The transmission of ultrasonic sound and the reception of reflected sound waves are separated in time.

This time interval is used to calculate the proportion of the waste container that is filled. The calculated data is displayed with the help of the Blynk Mobile app. The waste bin level is 80% in the waste bin (dry or wet) then the alert message is send to the waste collector along with location and percentage filled status of the particular waste bin.

The waste bin level is 100% in the waste bin (dry or wet) then the alert message is send to the waste collector as well as supervisor or manager along with location and percentage filled status of the waste bin. After pickup truck will come and collect the separated waste. The automatic lid and separation mechanism will be switched back on, and the waste bin will be reused.



Fig.7 Block diagram

IV PROPOSED METHODOLOGY

This technique can help you in classifying the solid waste. This system gives the alert message to waste collector and supervisor according to the level of bin along with location of bin. This system shows real time status of bin. 1. When human is detected then lid of waste bin is open.



2. Separate dumped solid waste into dry and wet waste.



3. Monitor real time status of bin on Blynk IoT Mobile App.

8:50		18 8 5 80
\times	Smart Dustbin	000
	Drig studention	
	(18)) (
	(1)	<u>n</u>

4. Send alert message according to level of bin.



V ADVANTAGES

- Reduces human efforts.
- > To minimize the operational cost.
- ➢ Keeps the atmosphere fresh and tidy.

VI CONCLUSION

Most of the time the conventional dust bins are very dirty and overflowed due to poor plan. Further the bins are cannot emptied for longer period of time. A step toward automating the manual collection and identification of wastes is the installation of a garbage management system. The proposed system trace the location of the bin .The smart bin separates dry/wet garbage collection using IoT sensor within the smart bin. This will helpful for waste management recycle process. Once the smart bin is full up to 80% either dry/wet partition, then the alert message is send to garbage collector automatically.

REFERENCES

- 1. K. C. Saranya, Vijayaraj Sujan, Balasubramanian Abivishaq and K. Nithish Kann —Smart Bin with Automated Metal Segregation and Optimal Distribution of the Bins, 2020.
- Hong I, Park S, Lee B, Lee J, Jeong D, Park S, Lee Beomseok, Lee Jaekeun, Jeong Daebeom, Park Sehyun. IoT- based smart garbage system for efficient food waste management. Sci World J 2014;2014:1–13. https://doi.org/10.1155/2014/ 646953.
- 3. Zeb A, Ali Q, Saleem MQ, Awan KM, Alowayr AS, Uddin J, et al. A proposed IoT-enabled smart waste bin management system and efficient route selection. J Comput Netw Commun 2019;2019:1–9. https://doi.org/10.1155/2019/ 7043674.
- 4. Michael E, Otaru CO, Liman AD, Bomoi MI, Awotoye B. Design and development of a smart waste bin. Int J Sci Technol Res 2017;6(10):101–5.
- 5. Kalpana Devi P, Dharini D, Lakshmi Vandhana J. Design of IoT based garbage segregation for automatic smart trash bin using NI LabVIEW. Int J Innovat Technol EXplor Eng 2019;8(11):3897–901.
- 6. Abhay Shankar Bharadwaj, Rainer Rego, Anirban Chowdhury, —IoT based solid waste management system: a conceptual approach with an architectural solution as a smart city application.
- 7. Hannan MA, Arebey M, Begum RA, Basri Hassan, Abdulla Al Mcamun Md. Content-based image retrieval system for solid waste bin level detection and performance evaluation. Waste Manag 2016;50:10–9.
- Karadimas, D., Papalambrou, A., Gialelis, J., Koubias, S.: An integrated node for smart-city applications based on active RFID tags; Use case on waste-bins. In: 2016 IEEE 21st International Conference on Emerging Technologies and Factory Automation (ETFA), pp.1–7.IEEE (2016, September).
- Lella Jaydeep, Mandla Venkata Ravibabu, Zhu Xuan. Solid waste collection/ transport optimization and vegetation land cover estimation using Geographic Information System (GIS): a case study of a proposed smart-city. Sustain Cities Soc 2017;35:336–49.

Handwritten notes recognition using Machine learning

¹Sakshi Rananavare, ²Swarali Joshi, ³Aishwarya Shinde, ⁴Seema Atole ^{1,2,3}UG Student, SVERI's COE, Pandharpur ⁴Assistant Professor, SVERI's COE, Pandharpur

ABSTRACT

Handwritten Notes Recognition (HNR) represent pivotal convergence of traditional penmanship and advanced computational techniques. In an increasingly digitized world, the preservation and utilization of handwritten information are essential. HNR, leveraging artificial intelligence and machine learning algorithms, seeks to bridge the gap between the analog and digital realms by transforming handwritten contenting to machine-readable text. This process involves complex pattern recognition, character analysis, and context understanding, allowing handwritten notes, regardless of the handwriting style, to be accurately transcribed and stored electronically. The implications of HNR extend far beyond mere transcription; it plays a vital role in digitizing historical documents, enhancing accessibility for individuals with disabilities, and revolutionizing education by making handwritten materials easily shareable and editable .The abstract nature of handwriting, characterized by diverse styles, shapes, and sizes, presents a challenging yet intriguing problem for computer scientists, making HNR an area of continuous research and innovation. As technology advances, the potential applications of HNR in diverse fields ,such as archival preservation, education ,and administrative efficiency ,are limitless, promising a future where the richness of handwritten communication seamlessly integrates with the digital landscape.

Key-Words: machine learning, HTR

I. INTRODUCTION

In the strokes of a handwritten note that digital text can seldom replicate. Handwritten digital age, where technology is seamlessly integrated into our daily lives, the art of handwriting often takes a backseat. However, there is a profound beauty and intimacy in notes hold a sentimental value, carrying the personal touch and unique identity of the author. In recent years, advancements in artificial intelligence and machine learning have paved the way for innovative technologies, one of which is handwritten notes recognition. This fascinating field amalgamates the charm of traditional penmanship with the power of cutting-edge technology, transforming handwritten scribbles into digital data that can be analyzed, stored and utilized in various applications.

The recognition of handwritten notes involves the use of sophisticated algorithms and machine learning models to interpret and convert handwritten text or symbols into machine-readableformat. Thistechnologyhasfar-reachingimplications, not only in preserving historical manuscripts and documents but also in modernizing education, enhancing accessibility, and revolutionizing administrative tasks. Handwritten notes recognition serves as a bridge between the tangible, nostalgic world of pen and paper and the limitless possibilities offered by digital platforms. It enables the conversion of handwritten content into editable digital text, making it easier to search ,store, and share valuable information. Furthermore, it open avenues for individuals with disabilities, offering them equal access to handwritten materials, which was once challenging feature, Harmonizes seamlessly with the efficiency of digital processing, revolutionizing the way we perceive, utilize, and cherish heart of handwritten communication.

BALTHAZAR	SIMON	BENES	
NOM BALTHAJAR	51 NON	86465	
LA LOVE	DAPHNE	UCE	
NOM LALOUE	PRIME OFFICE	10 C I E	

Fig 1. Text Recognition

II. MEYHODOLOGY

Hand written notes recognition definition says that it is the ability of the model to recognize the notes written by the human. It is not easy for the machine to recognize the human written characters since each and every human being tends to have different handwriting along with it comes the different strokes used by each individual. Use of machine learning model is the easy way out of this problem. The recognition using the NumPy, Panda, Seaborn proves to be them efficient and useful solution for this problem. Use of datasets available in the Kaggle dataset proves to be one of the best useful dataset having variety of handwriting images.

Tensorflow being a open source platform for the machine learning operations. It is an open source platform for deep learning operations . It is a beautiful computational graph which helps build functions and variables by just giving Them Shape and size.



Fig. 2.Readable images



Fig.3.Unreadable images

Dataset consist of three different types of files particularly train, test, and validation. The first thing that we have done is to visualize it and see the description of the dataset. After that is to measure the dimension of the images presenting dataset along with its indices. Next is cleaning the data, getting rid of all the null values along with the images which are unreadable due to missing characters. Setting all the labels to uppercase for uniformity of the data and the cleaning is done. The images present in dataset are being loaded as grayscale and all are reshaped with the 265width and 64 height. If image is smaller than this dimension padding is added. The model is trained on 60000 images, validated on 3000 images and tested on 3000 images. The strings of name are converted into the numbers by using systematic sets of rule. Four variables are created for each sets i.e. train, test& validate (train_y, train name_len, train_input_len, train output.) Using CNN to process images and connecting it to RNN, we will employ deep CNN architecture (3 layers) and shallow RNN architecture (2 layers) to build our machine learning

model. Train the model for 10 epochs to get the higher accuracy and counting the c loss for each epoch. Them or the model run through epochs the more is the accuracy. After training our model we are validating it on our validation dataset to measure the accuracy. And now using it on the test dataset.



Fig.4.FlowChart

In a Handwritten Notes Recognition project using machine learning, Convolutional Neural Networks (CNNs) are commonly used for image processing tasks. CNNs excel at capturing hierarchical features in images, making them effective for recognizing patterns in handwritten notes. These networks are trained on labeled datasets to learn and identify specific patterns, allowing them to classify and recognize handwritten characters or words in notes. When training a model for handwritten notes recognition using

machine learning, determining the number of epochs involves finding the balance between underfitting and overfitting. Experimentation is often required to find the optimal number. Start with a reasonable number of epochs and monitor the training and validation performance. Underfitting: If the model hasn't learned enough, increasing the number of epochs may help it capture more patterns and improve performance. Overfitting: If the model starts performing well on the training data but poorly on new, unseen data, it might be overfitting. In such cases, reducing the number of epochs or incorporating regularization techniques may be necessary.Consider using techniques like early stopping to prevent overfitting and save computation time. Early stopping monitors the validation performance over epochs to understand the model's behavior and make informed decisions about the number of epochs.Recurrent Neural Networks (RNNs) can also be employed in handwritten notes recognition projects. RNNs are well-suited for sequence data, making them suitable for tasks where the order of information matters, such as recognizing handwriting strokes or the temporal aspect of handwritten notes. They can capture dependencies between different parts of the sequence, helping in understanding the context of characters or words within the handwritten text. However, for image-based tasks like handwritten notes recognition of CNNs and RNNs CRNN - might be more effective, as CNNs handle spatial features while RNNs capture sequential dependencies.

III.RESULT

In this study, we developed a robust Handwritten Notes Recognition system using Convolutional Neural Network (CNN) implemented with TensorFlow, a popular machine learning framework. The objective of the project was to accurately and efficiently transcribe handwritten notes into digital text, enabling seamless integration of handwritten content into digital platforms. Utilized a comprehensive dataset comprising diverse styles of handwritten notes collected from various sources. The dataset was preprocessed, including task such a normalization, and noise reduction, ensuring optimal input quality for the CNN model. The CNN architecture employed for this task consisted of multiple convolutional layers followed by maxpooling layers to extract essential features from handwritten images. Batch normalization and dropout techniques were applied to enhance model generalization and prevent overfitting. The model was trained using a carefully curated training set and validated on a separate validation set. During training, we employed stochastic gradient descent as the optimization algorithm, fine-tuning hyper parameters like learning rate and batch size through rigorous experimentation. The model's performance was continually monitored using metrics such as accuracy, loss, and validation scores. After extensive training,

the CNN-based Handwritten Notes Recognition system achieved remarkable results. The model demonstrated an accuracy of on the test dataset, indicating its proficiency in recognizing handwritten characters and symbols. The system also exhibited excellent generalization capabilities ,accurately transcribing notes with various writing styles and inconsistencies. Comparative analysis with existing handwritten recognition system showcase four CNN-based approach. The System out performed traditional methods and exhibited significant advancement in terms of accuracy, especially when dealing with complex and diverse handwritten styles. While the results were promising, challenges such as handling noisy or distorted inputs still exist. Future work will focus on integrating advanced preprocessing techniques and exploring state-of-the-art CNN architectures, such as attention mechanisms and transfer learning, to further enhance the system accuracy and robustness.

	and the second
NAID Ennergenergenergenergen	unnund - tess erinte - test kiers - arftest mans
ech 2/3#	
6/135 [*********************	
ech 3/18	
25/235 [**********************	neeneenee] - 1446s da/step - Joss: 2,4253 - val_loos; 2,9418
pich 4/18	
25/235 [
poch 5/38	
35/235 [] - 1447s de/step - less: 1.9776 - val_less: 2.4561
poch 6/18	
35/235 [mananaa) - 5461x Anister - 1045: 1.8275 - val 1008: 2.8578
auch 7/18	
35/235 (************************	
apph \$/28	
15/285 Tenenson and an and a second	unnersen - tails folates - last: 1.6100 - val lass: 2.1901
and \$15	
95/195 1	managed a 1446 database - Souri 1 Stati - and teach 1 800
with talts	
221533 Constantine services	unnunna) + 74412 Athlad - Talli Franz - AMF 1002: 11902

Fig. 5 Epochs



Fig.5. Output

IV.CONCLUSION

In conclusion, Handwritten Notes Recognition (HNR) stands as a testament to the symbiosis between human ingenuity and technological prowess. A smaller reflect on the transformative journey from ink on paper to pixels on screens, it becomes evident that HNR has ushered in an era where the intimacy of handwritten expression harmonizes flawlessly with the efficiency of digital processing. The ability to decode and digitize handwritten content not only preserves our cultural so democratize access to information. Through HNR, historical manuscripts once confined to the archives are now accessible to a global audience, fostering a deeper understanding four collective past. Furthermore, the impact of HNR on education cannot overstated, revolutionized classrooms, making educational resources more accessible and interactive. Students can now collaborate, edit, and share handwritten notes effortlessly, transcending the limitations. physical boundaries. The inclusivity it offers to differently-abled individuals, empowering them with equal access to knowledge, is a remarkable stride towards a more equitable society. As we peer into the future, the continuous evolution of HNR technology holds the promise of even greater achievements. The ongoing refinement of algorithms, coupled with the integration of emerging technologies like neural networks, ensures that HNR will become more accurate, adaptable, and ubiquitous. From aiding historical research to transforming how we learn, work, and communicate, Handwritten Notes Recognition stands as a testament to human innovation ,demonstrating that the essence of handwriting can endure and thrive in the digital age, enriching lives and expanding horizons for generations to come.

REFERENCES

- 1. U.Pal,B.B.Chaudhuri,"IndianScriptCharacterrecognition:Asurvey",PatternRecognition,vol.37,pp.1887-1899,2004.
- 2. Dholakia,K.,ASurvey on Handwritten Character Recognition Techniques for various Indian Languages, International Journal of Computer Applications,115(1),pp17–2015
- SuryaNath RS, and S. Afseena. "Handwritten Character Recognition-A Review." International Journal of Scientific and Research Publications (2015).
- 4. Salma Shofia Rosyda, and Tito Waluyo Purboyo."A Review of Various Handwriting Recognition Methods."International Journal of Applied Engineering Research13.2(2018):1155-1164.
- 5. Younus, S. B. S., S. Shajun Nisha, and M. Mohamed Sathik. "Comparative Analysis of Activation Functions in Neural Network for Handwritten Digits." Studies in Indian Place Names40.71(2020):793-799.
- 6. Jagan Mohan Reddy D, A Vishnuvardhan Reddy "Recognition of Handwritten Charactersusing Deep Convolutional Neural Network".
- 7. HrudayM."ImplementationofHandwrittenCharacterRecognitionusing Neural Network".
- 8. Meenu Mohan, and R. L. Jyothi. "Handwritten Character Recognition: Comprehensive Review on Geometrical Analysis".
- 9. YuvalNetzer,TaoWang Adam Coates Alessandro Bissacco Bo Wu Andrew Y.Ng."Reading digits in natural images with unsupervised feature learning."(2011).
- 10. Polaiah Bojja., Naga Sai Satya Teja Velpuri, Gautham Kumar Pandala, SD Lalitha Rao Sharma Polavarapu"HandwrittenText Recognition using Machine Learning Techniques in Application of NLP".
- 11. AhmedMahdiObaid, II Hazem M. ElBakry, III M.A.Eldosuky, IVA.I.Shehab"Handwritten text recognition system based on neural network." Int. J. Adv. Res. Comput. Sci.Technol.(IJARCST)4.1(2016):72-77.
- 12. S. M. Shamim, Mohammad Badrul Alam Miah, Angona Sarker, Masud Rana, Abdullah AlJobair "Handwritten digit recognition using machine learning algorithms." Global Journal Of Computer Science And Technology(2018).
- 13. P.ShankarRao, J.Aditya. "HandwritingRecognition-"Offline" Approach"

Adiabatic Technique Based Low Power 12t Static Ram For Data Communication Systems

¹Shilpa D B, ²Dr. Kishan Singh

Ph.D Scholar (Visvesvaraya Technological University) Professor(GNDEC,Bidar)

ABSTRACT-

The use of SRAM enormously varies based on the how repeatedly it is accessed. Like, when used at higher frequencies it's as control hungry as a DRAM, that results in several watts of power consumption at maximum bandwidth in certain IC's. In applications with microprocessors that are agreeably controlled, they are accustomed to operate at a lower speed, drawing nearly no power and it might aid pretty much unfavorable power consumption while in idle conditions that might range in microwatts. Incessant candidacy to compute constraints like functionality and capacity on compact hand-holder and systems that are battery-operated has kicked in to poor capacity memories. Hence, it becomes essential to introduce new methods to handle SRAM based memory architectures. In this paper an adiabatic SRAM is proposed and evaluated using EDA tool, that aims to produce a meaningful output for SRAM by using adiabatic power converters and flipping ideas to construct the design. The observations disclose that the proposed SRAM has higher output than the existing one.

Index Terms: Single-event upset (SEU), single-event multimode upsets (SEMNUs), critical charge, and radiation hardness, read stability, hold power, and write ability.

Deep space contains highly energetic particles, which impact the functionality of memory circuits [1,2]. On striking the substrate of an integrated circuit, such as semiconductor memory, an energetic particle generates electron-hole pairs. The electric field caused due to the reverse bias between the diffusion region and substrate/n-well appears to the strike-generated minority carriers as a forward field[3]. Hence, minority carriers drift towards the drain diffusion regions, and on accumulation, a positive or negative voltage spike is generated based on the type of minority carrier. If the level of the spike is beyond the threshold use for switching of the logic circuit and its duration is long enough, the stored content may flip, that results in a phenomenon called single-event upset (SEU) or soft-error [4], [5]. Furthermore, slightest spacing between devices of an integrated circuit decreasing drastically due to aggressive technology scaling, a strike by a single ion may affect multiple nodes, which may result in a single-event multi-node upset (SEMNU) [6]. To address the effects of SEUs on memory, triple modular redundancy (TMR) has been used. This method uses three copies of memory cells, with majority voting to select and output the correct value [7], [8]. If one copy is flipped, the other two will dominate the voting process, resulting in similar output. However, this technique incurs huge power & area penalties, making it unsuitable for most designs [8], [9]. Another way to mitigate the effects of SEUs

is to employ error correction codes (ECCs). However, ECCs incur huge power, area and delay overhead consequent to the requirement of redundancy and extra devices for encoding and decoding circuits [10], [11]. Therefore, soft-error-aware SRAMs are preferred over ECCs because they are a less power-area- and delay-consuming solution [12]. Furthermore, it is preferred that the Static RAM cell should have multi-node upset recoverability along with its SEU recovery ability [12]. With the positive feedback in cross-coupled inverters of the 6T SRAM cell, an SEU happening at one storage node automatically alters the content of the other storage node. Therefore, the 6 Transistor cell does not possess the characteristics that a soft-error-aware SRAM should.

I. EXISTING METHOD

Proposed in [11], is able to regain from a '1' \rightarrow '0' SEU. However, it is unable to accomplish from an SEU induced that is at the '0'-storing storage node. Moreover, it flashes an extreme write failure probability. The authors in [12] presented its modified version, QUATRO12T, to improve the write operation. However, QUATRO12T also shows only partial immunity to SEUs. Two further soft-error-aware SRAM cells, QUCCE10T and QUCCE12T, were proposed in [10]. However, QUCCE10T exhibits poor write performance. Moreover, it is unfit to regain from a '0' \rightarrow '1' SEU generated at its '0'-storing storage node, while QUCCE12T cannot recover from a '0' \rightarrow '1' SEU induced at both its '0'-storing storage and internal nodes. Furthermore, QUCCE12T, besides previously mentioned QUATRO12T, consumes high hold power. In short, all the above-mentioned cells are only halfway immune to SEUs and cannot regain at all from SEMNUs. To achieve recovery from SEMNUs, the authors in [13] proposed RHD12T, that is able to recover from SEMNUs occurring at its internal node-pair. Whereas, it's incapable to get back from SEUs induced at its '0'-storing storage node. The enhanced version of RHD12T, called RSP14T [5], can tolerate a higher charge at the '0'-storing storage node. It still cannot regain the data if a '0' \rightarrow '1' SEU of sufficient strength affects the node. Both types of SEUs, i.e., '1' \rightarrow '0' and '0' \rightarrow '1', induced at all the sensitive nodes and SEMNUs induced at one nodepair can be recovered by RHM12T [11] and RHPD12T [13]. However, in RHM12T, scaling down of the VDD, the supply voltage is limited in consequence to extreme stacking present in core inverters, while RHPD12T consumes a bigger area and higher power consequential to the use of large-size transistors. Also, it is to be point taken that in all the above-mentioned cells the '0'-storing storage and/or internal node(s) are directly accessed during read operation by the bitline and the node(s) are unable to recover from upset. Therefore, all the above-mentioned cells show deprived read stability. To treat the above-mentioned issues, we propose the Soft-Error-Aware Read-Stability-Enhanced Low-Power 12Transistor (SARP12T) Static RAM cell (Fig. 1) in this article. SARP12T has the below mentioned major features: 1) SARP12T is immune to SEUs of both polarities caused at any sensitive node. 2) The existing cell can recover from SEMNUs that occur at its storage node-pair. 3) SARP12T consumes the slightest hold power amid all the considered cells. 4) SARP12T shows enhanced read stability as the '0'-storing storage node, which is directly accessed by the bitline during read operation, can recover from any upset. 5) The existing cell presents greater write ability and shorter write delay in comparison to other cells.



Fig.1. Schematic of the proposed SARP12T SRAM cell

The schematic of SARP12T is exhibited in Fig. 1. SARP12T has two wordlines, WL and WWL, two storage nodes, Q and OB, and two internal nodes, S1 and S0. WL regulates the access transistors N7 and N8, that connect Q and QB the storage nodes with their corresponding bitlines BL and BLB. The internal nodes S1 and S0 are linked to their corresponding bitlines BL and BLB through their corresponding access transistors N9 and N10, that are controlled by WWL. Let us contemplate SARP12T with all the comparison cells storing '1', i.e., Q = '1' and QB = '0'. Thus, S0 and S1 are storing '0' and '1', respectively. All the primary operations of SARP12T are mentioned in this sub-section. 1) Hold Operation: Both WWL and WL are pulled down to GND by both pairs of access transistors by keeping them OFF during hold mode. To shorten the read delay, bitlines are kept precharged to VDD during hold mode. Therefore, when the cell is in hold state, transistors P1, N2, N3 & N6 are still ON, while the rest of the transistors keep OFF for the above mentioned case. Thus, SARP12T maintains its initial stored data. 2) Write Operation: During write operation, both the wordlines (WL and WWL) are activated. Therefore, both pairs of access transistors (N7/N8 and N9/N10) are turned ON. For altering the data to be stored (i.e., writing '0' at Q), BL is linked to GND, whereas BLB is clamped at VDD. Since BL is connected to GND, nodes S1 and Q nodes are pulled down by BL through N7 and N9, respectively. Subsequently, node Q turns ON P2 & turns OFF N6, whereas node S1 turns OFF N2 and N3. In the meantime, nodes QB and S0 are pulled up by BLB through N8 and N10, respectively. Consequently, node QB turns OFF P1 and turns ON N5. Similarly, node S0 turns ON N1 and N4. The cross-coupling between P1 and P2 amplifies the potential difference between Q and QB. Similarly, the cross-coupling between N3 and N4 enhances the potential difference between S1 and S0. Therefore, the write operation is executed successfully. 3). Read Operation: During read operation, WL is linked to VDD, whereas WWL is kept deactivated. Therefore, access transistors N7 and N8 are turned ON, while the other access transistors (N9 and N10) remain OFF. Bitlines, for read operations are precharged to VDD. Therefore, BLB discharges through N8, N2 and N3. As N1 and N4 are OFF, BL stays at VDD .A sense amplifier (not shown) can sense the stored data, once the voltage difference between BL and BLB reaches 50 mV that completes read operation.

II. PROPOSED METHOD

Dynamic Voltage Scaling (DVS) is used in adiabatic SRAM technology to improve capacity information while in neutral mode and reduce leakage current & quality of adiabatic Static RAM cells. When the working voltage is scaled to submicron levels, the resulting short channel reduces the spilling current. Pass transistors N4 & N5 provide standard and inactive ground flexibly pressures of the adiabatic SRAM cell.

Transistor N4 is given a positive voltage if the adiabatic Static RAM cell has moved out of gear, and when the cell is operating in dynamic mode, another transistor N5 is given a simulated premise. Memory cells use different working voltages to distinguish between active and passive modes, which allows for a significant increase in the spillage limit. The concept behind adiabatic Static RAM is to use 2 pass transistors to provide ground voltages explicitly to memory cells in both normal and rest states. When the cell is disabled, the positive ground flexibly voltage of these exchange transistors allows them to interface with the ground in the similar way as standard 6 T cells' cross-associated inverters do. When switching between dynamic & reserve modes, the operating voltages in a memory cell's configuration are discernible, and the spillage limit is drastically reduced. The bit line leaking is something that may be helped along by using high-Vt control transistors (M5, M6). High-Voltage, a control instrument that screens the flow from positive control voltage V to gnd via such 2 pass transistors, is employed at one of the passing entrances to guide the N-Metal Oxide Semiconductor transistor sources voltages on the coupling inverter. No hubs remain skimming when the cell is not in use, ensuring uniformity without adding complexity or expensive hardware to the capacity data. Since the efficiency of the field flexible lines is only slightly lower than that of the wells, there has been an increase in the correlation between the time and resources exchanged across the various lines. Indeed, the source voltage, as opposed to the sub-voltage, is used to guide the V, of N-Metal Oxide Semiconductor transistors during rest stage, preventing any inherent concerns with the body bending. There will be no door leakage in an adiabatic SRAM cell. The concept behind the adiabatic SRAM is to provide varying base rates in both dynamic and silent approaches to the memory cell. Out of gear mode, the positive strain at the adiabatic SRAM cell's entrance and under-edge flows is reduced to near-earth levels.



Fig: 2. Adiabatic based 12T SRAM

A low-power random-access memory (RAM) device that selects just the read bit lines to minimise preload and RAM power consumption. the preferred precharge bit in the pre-charge RAM circuit is the A precharge amplifier, linking the selected bit line and precharging it through a Mutiplexers board. In order to prevent RAM data corruption during loading, it is suggested

to use Preload Devices on each bit line in addition to the primary loaders. Because RAM leakage occurs over time, secondary precharge devices are quite tiny, often being just 1/20 the size of a conventional precharge unit.

The RAM system's pre-load trigger, column-select signals, word-line signals, randomly pre-loaded chosen bit line, and the omission of potentially wasteful DC current direction all contribute to the system's low power consumption.

Fig: 3. Existing method design.

Fig: 4. Simulation results for existing system.

- Fig: 5. Power results for existing system.
- Fig: 6. Proposed method design using adiabatic system.
- Fig: 7.Simulation results for proposed system.
- Fig: 8. Power result for proposed system.

VLSI programmers now have a major challenge in the form of rising energy costs. For use in aerospace, a low-power SRAM cell is presented with improved read stability that can account for soft errors. Even if a radiation hit flips the values of the sensitive nodes, SARP12T can restore the original data. Also, if several nodes are disrupted by an ion impact on the storage node-pair, SARP12T may recover. The suggested cell also has greater write performance than the majority of the comparator cells and displays the greatest RSNM while using the least amount of hold power. In addition, SARP12T outperforms all other modern cells in terms of EQM. As a result, the suggested SARP12T is preferable for use in aircraft. The suggested cell is robust against single-node upsets of any polarity and strength, and is also capable of recovering from multiple-node upsets caused by charge sharing on the fixed nodes. In addition, compared to earlier radiation hardened memory cells, the suggested cell offers similar or reduced expenditures in terms of stationary power, area, and access time.

REFERENCES

- N. Maroof and B. S. Kong, "10T SRAM Using Half-VDD Precharge and RowWise Dynamically Powered Read Port for Low Switching Power and Ultralow RBL Leakage," in IEEE Transactions on Very Large Scale Integration (VLSI) Systems, vol. 25, no. 4, pp. 1193-1203, April 2017, DOI: 10.1109/TVLSI.2016.2637918.
- C. C. Wang, D. S. Wang, C. H. Liao and S. Y. Chen, "A Leakage Compensation Design for Low Supply Voltage SRAM," in IEEE Transactions on Very Large Scale Integration (VLSI) Systems, vol. 24, no. 5, pp. 1761-1769, May 2016, DOI: 10.1109/TVLSI.2015.2484386.
- J. P. Kulkarni and K. Roy, —Ultralow-voltage process-variation-tolerant Schmitt-trigger-based SRAM design, IEEE Trans. Very Large Scale Integr. (VLSI) Syst., vol. 20, no. 2, pp. 319–332, Feb. 2012, DOI: 10.1109/TVLSI.2010.2100834.
- 4. Mr.NurdianiZamhari, Mr. Peter Voon, Miss/Mrs.KuryatiKipli, Mr.Kho Lee Chin, Mr.MaimunHujaHusin, "Comparison of Parallel Prefix Adder (PPA)",proceedings of the World Congress on Engineering 2012 Vol IIWCE 2012, July 4 - 6, 2012, London, U.K.
- 5. Anas Zainal Abidin, Syed Abdul Mutalid Al Junid, KhairulKhaiziMohdSharif, "A survey on 4 bit Brent kung PPA simulation study usingsilvaco EDA Tools".
- 6. Avinashshrivastava, ChandrahasSahu, "Performance Analysis of PPA based on FPGA, International Journal of engineering trends and technology (IJETT)-volume 21, number 6-march 2015.
- 7. B. H. Calhoun and A. P. Chandrakasan, "A 256-kb 65-nm subthreshold SRAM design for ultra-low-voltage operation", IEEE J. Solid-State Circuits, vol. 42, no. 3, pp. 680-688, Mar. 2007, DOI: 10.1109/JSSC.2006.891726.
- 8. S. Lin, Y.-B. Kim, F. Lombardi, "A low leakage 9t SRAM cell for ultra-low power operation", Proc. 18th ACM Great Lakes Symp. VLSI, pp. 123-126, 2008, DOI: 10.1145/1366110.1366141.
- 9. Z. Liu and V. Kursun, "High read stability and low leakage cache memory cell," IEEE Trans. Very Large Scale Integr. (VLSI) Syst., vol. 16, no. 4, pp. 488–492, 2008, DOI: 10.1109/ISCAS.2007.378628.
- I. J. Chang, J.-J. Kim, S. Park, and K. Roy, "A 32 kb 10T subthreshold SRAM array with bit-interleaving and differential read scheme in 90 nm CMOS," IEEE Journal of Solid State Circuits, Vol. 44, No. 2, pp. 650–658, 2009, DOI: 10.1109/JSSC.2008.2011972.
- 11. M.H. Tu, J.Y. Lin, M.C. Tsai, Shyh-JyeJou and Ching-Te Chuang "Single-Ended Subthreshold SRAM With Asymmetrical Write/Read-Assist" IEEE Trans. on circuits and systems-I: Regular Papers, vol. 57, no. 12, Dec. 2010, DOI: 10.1109/TCSI.2010.2071690.
- M.H. Chang, Y. T. Chiu and W. Hwang., "Design and Iso-Area VminAnalysis of 9T Subthreshold SRAM with Bit-Interleaving Scheme in 65-nm CMOS", IEEE Trans. on Circuits and Systems-II: Expr. Briefs, vol. 59, no. 7, pp.429-433, Jul. 2012, DOI: 10.1109/TCSII.2012.2198984.

Flexgrip: Enhancing Robotic Manipulation With Sensor-Enabled Gloves

¹Rahul Pawar1, ²Jeevan Musale, ³Aditya Ghavate, ⁴Sandip Khatake, ⁵Shubham More ^{1,2} Assistant Professor, SVERI's College of Engineering, Pandharpur ^{3,4,5}Student SVERI's College of Engineering, Pandharpur

Abstract— The study explores the emerging domain of hand gesture recognition within robotics and human-computer interaction, offering transformative applications in everyday tasks and intelligent workplaces. The research introduces an innovative system designed to facilitate hands-free operation, featuring a glove equipped with flexible force sensors and a 3D-printed robotic forearm. Users can seamlessly control the robotic arm's movements by wearing the glove, with the forearm mirroring the glove's motions in real-time. The 3D-printed forearm incorporates electronic components such as five flex sensors, & an Arduino UNO, Through rigorous testing, the system demonstrates precise responsiveness, showcasing potential benefits for individuals working in hazardous environments or requiring remote operation for safety.

Keywords: Robotic arm, Flex sensor, Arduino UNO

I.INTRODUCTION

Robotics is a multidisciplinary field that involves the design, construction, operation, and use of robots to perform various tasks autonomously or semi-autonomously. It merges aspects of mechanical engineering, electronics, computer science, and artificial intelligence to create intelligent machines capable of interacting with the physical world. Engaging in perilous activities like landmine demining, military clearance, operating hazardous vehicles, space exploration, or handling molten metals in industrial settings exposes individuals to significant risks, including life-threatening situations. The nature of these tasks entails potential dangers that could be fatal.[1] Developing systems or devices to replace human workers in hazardous tasks is crucial for ensuring safety. Over the past few decades, there has been a growing focus on and heightened concern for hand function due to its unique importance in daily activities. [[2], [3], [4], [5]]

Studies on hand functions can be categorized into two main areas: hand gesture recognition and hand motion controls. Several studies have explored hand gesture research, particularly in the realm of artificial muscle-based human-machine interfaces [[6], [7], [8]]. Diverse technologies, including electromyography (EMG) [9], computer vision [10], and accelerometers [9,11], have been employed to capture hand movements driven by forearm muscle contractions. A comprehensive review, encompassing 148 articles, aimed to delineate the characteristics of touchless or in-air hand gestures in interaction interfaces. Additionally, recent research has delved into advanced material technologies, focusing on tactile sensors to capture hand motion. One such study investigated wearable soft artificial skin, incorporating an array of embedded soft strain sensors to detect hand motions by measuring finger joint movements. The system presented in this paper introduces a 3D printed forearm designed to perform hazardous tasks remotely, effectively substituting for human hands. The robotic arm's fingers are electronically manipulated by five flex force sensors. These flexible sensors, known for their versatile capabilities in measuring angle displacement, pressure, and stress, find applications in wearable electronics and health monitoring. Comprising two primary hardware components—a sensor-equipped glove and the robotic forearm with motors—the flex sensors, strategically placed on the glove, serve to measure finger angles. The glove functions as a wireless controller, enabling seamless control over the actions of the 3D-printed robotic forearm.



Fig.1 Flex Sensor-Controlled Robotic arm

II. LITERATURE SURVEY

A scalable wearable robotic exo-glove designed to accommodate variations in hand kinematics. Unlike traditional methods, this glove deforms the robotic finger's skin, employing woven tendons for multi-DOF finger joint motions, mimicking both intrinsic and extrinsic muscles. By placing actuators on the surface, this biomimetic approach allows the exo-glove to transform any hand skeleton into a functional robotic hand, offering versatility for traditional robotic hands, prosthetic designs, and power-assisted hand gloves.[12] YoBu, a novel data glove designed for robotic hand-arm teleoperation. YoBu incorporates eighteen low-cost inertial and magnetic units to capture simultaneous human arm and hand motions. The system utilizes Bluetooth for seamless command transmission to the robot, ensuring stability, portability, and user-friendliness in various experiments validating its efficiency.[13] The application of robotic arms in diverse fields, including military, defense, medical surgeries, and industrial automation. It focuses on a flexible and gesture-controlled robotic arm system using accelerometers, enabling wireless control via Zigbee protocol. The technology allows the replication of human hand actions, making it suitable for hazardous environments like firework manufacturing and bomb diffusers.[14] This project addresses the growing need for safer and more affordable prosthetic arms in the face of increasing amputations due to industrialization and safety awareness issues. The prototype utilizes servo motors controlled by Arduino, integrating flex sensors and muscle signals for versatile hand functions. Designed for diverse applications, including bomb disposal, humanitarian aid, and surgery, the robotic hand offers a cost-effective solution in response to the rising demand for automated prosthetics and military technology.[15] 3D-printed components for the basic structure of an artificial hand, with finger movement controlled by a smart glove equipped with flex sensors. The entire system can be connected to a microcontroller and utilizes LabVIEW software for programming. The study focuses on deriving precise equations for each artificial finger, aiming to match the angles of human fingers. LabVIEW's digital gates are employed to implement these equations, and further modifications are made to enhance accuracy and eliminate errors in the movements.[16] A robotic arm is designed for challenging scenarios where human interaction is difficult or impossible, such as collecting data from a volcano or diffusing a bomb. The mechanical arm mimics natural human movements through accelerometers, and an Arduino-based control system ensures precise and smooth operation. The six-position arm, featuring bidirectional rotation, opening and closing, and various inclinations, is capable of handling tasks like picking up 150gms of weight, addressing challenges in handling both hazardous and non-hazardous objects from a distance.[17]

III. METHODOLOGY

The project initiation involves a comprehensive definition of the objectives, delineating specific tasks and movements for the Arduino UNO, servo motor, and flex sensor-controlled robotic arm. The system architecture is meticulously designed, specifying the strategic placement of the flex sensor, determining the necessary servo motors for the robotic arm's movements, and establishing communication protocols. For component selection, careful consideration is given to choosing appropriate hardware, including an Arduino UNO microcontroller, servo motors with suitable torque and speed characteristics, and a flex sensor with the capacity to capture a broad range of hand gestures. The flex sensor is then seamlessly integrated onto a wearable surface, typically a glove, ensuring optimal positioning for precise gesture recognition. Concurrently, the servo motors are configured according to the mechanical design of the robotic arm, aligning with the desired degrees of freedom for each joint. The crux of the project lies in the development of Arduino code, intricately tailored to interpret signals from the flex sensor and translate them into precise commands for the servo motors. This comprehensive methodology ensures a systematic and detailed approach to creating a functional and responsive robotic arm system.

1 Flex Sensor

A flex sensor, a versatile and adaptable component, operates on the principle of variable resistance, changing its electrical resistance in response to bending or flexing. Constructed from flexible materials like polyester, it typically features conductive elements on its surface. As the sensor bends, the distance between these conductive elements alters, thereby modifying the sensor's resistance. Widely employed in wearable technology, such as gloves or garments, flex sensors facilitate the measurement of joint movements and gestures. Their applications span across diverse fields, including robotics, medical devices, virtual reality systems, and human-machine interfaces. Integration with microcontrollers, such as Arduino, allows for easy interfacing with electronic circuits to capture and process changes in resistance. Renowned for their flexibility and durability, flex sensors conform to curved surfaces and endure repeated bending. While excelling at measuring bending or flexing, they may have limitations in capturing other forms of deformation, like torsion or compression. Overall, flex sensors offer a dynamic solution for applications requiring precise and responsive monitoring of physical movement.



Fig.2 Flex Sensor

2 Arduino UNO

Arduino UNO is a popular and widely used open-source microcontroller board that serves as a fundamental building block in electronics projects. Developed by Arduino, it is based on the ATmega328P microcontroller and features a user-friendly interface, making it accessible for both beginners and experienced developers. The Arduino UNO microcontroller plays a central role as the brain of the system. Arduino UNO provides a user-friendly platform for interfacing with electronic components, making it an ideal choice for integrating and processing signals from the flex sensor. With its ATmega328P microcontroller, the Arduino UNO facilitates precise control and real-time responsiveness, interpreting the flex sensor data and generating corresponding commands for the servo motors. Its open-source nature and compatibility with the Arduino IDE simplify the programming tasks, allowing developers to efficiently code the logic that translates hand gestures into robotic arm movements. The numerous digital and analog pins on the Arduino UNO enable seamless connectivity with sensors and motors, making it a versatile and powerful controller for this project. Overall, the Arduino UNO serves as a crucial component, providing the intelligence and control necessary for the successful implementation of the flex sensor-controlled robotic arm.



Fig.3 Arduino UNO

3 Servo Motor

The servo motor serves as the actuator responsible for translating the signals received from the Arduino UNO and flex sensor into physical movements. The servo motor's precise control and ability to rotate to specific angles make it well-suited for applications requiring accurate and controlled motion, such as mimicking human hand gestures. Integrated into the robotic arm's mechanical design, the servo motors execute the desired movements of the arm's joints and fingers based on the input from the flex sensor. The project benefits from the servo motor's compact size, high torque, and positional accuracy, allowing for a responsive and lifelike replication of hand gestures. The modular nature of servo motors also facilitates scalability, enabling the incorporation of multiple motors to achieve more complex and coordinated actions in the robotic arm. Overall,

the servo motor proves instrumental in translating the electronic signals into the physical actions required for the flex sensorcontrolled robotic arm.



Fig.4 Servo Motor

IV. RESULT

The implementation of the robotic arm controlled by Arduino UNO, servo motors, and a flex sensor yielded promising results. Through extensive testing, the system demonstrated accurate and responsive control over the robotic arm's movements in correspondence to the user's hand gestures. The flex sensor, integrated into a wearable glove, effectively captured a diverse range of hand motions, enabling seamless communication with the Arduino UNO microcontroller. The servo motors, configured based on the mechanical design, exhibited precise articulation, mimicking the intended degrees of freedom for each joint in the robotic arm. The system's real-time responsiveness was a key highlight, with minimal delays observed between the user's gestures and the corresponding robotic arm actions. Additionally, the ergonomic integration of the flex sensor on the glove ensured user comfort during prolonged operation. The project successfully achieved its objectives of creating a functional and controlled robotic arm system, showcasing the potential of combining Arduino UNO, servo motors, and flex sensors for intuitive and accurate human-machine interaction.

V. FUTURE SCOPE

The successful development of the robotic arm controlled by Arduino UNO, servo motors, and a flex sensor paves the way for exciting future advancements. Integration of advanced machine learning can enhance gesture recognition, expanding the range of hand movements. Further research may explore additional sensors like accelerometers or gyroscopes for nuanced gestures, and scalability for multiple robotic arms collaborating in industrial automation or complex tasks. Adding haptic feedback can improve user experience, and the open-source nature allows for community-driven enhancements, ensuring relevance amid evolving technology landscapes. The project's foundation can serve as a platform for emerging hardware and software solutions, fostering continuous improvement and innovation in human-machine interaction.

VI. CONCLUSION

The development of the Arduino UNO, servo motor, and flex sensor-controlled robotic arm represents a significant milestone in creating an interactive and responsive human-machine interface. The successful integration of these components demonstrates the feasibility of designing a robotic arm that accurately replicates human hand gestures. The system's potential applications in hazardous environments and remote operations highlight its practical value, while serving as a foundation for future innovations. The project emphasizes the critical role of effectively combining hardware components for precise control and real-time responsiveness. This success serves as a launching pad for ongoing research in gesture recognition, sensor integration, and collaborative robotics, contributing to the advancement of intuitive human-machine interaction through robotic systems.

REFERENCES

- 1. T. Vuletic, Systematic literature review of hand gestures used in human computer interaction interfaces, Int J Hum-Comput St.(2019)
- 2. Q.F. Shi et al.Minimalist and multifunctional human machine interface (HMI) using a flexible wearable triboelectric patch, Nano Energy.(2019)
- 3. Health and Safety Executive, Workplace fatal injuries in Great Britain (2019)
- 4. H. Kawasaki et al.Development of a hand motion assist robot for rehabilitation therapy by patient self-motion control
- 5. S. Marco et al. Patterns of hand motion during grasping and influence of sensory guidance, J. Neurosci.(2002)
- 6. W.L. Howard, P.A. Dean, M.G. Geoffrey, Game with hand motion control. US patent 9317110B2....
- 7. E.W. Daniel, Resolved motion rate control of manipulators and human prostheses, Trans. Man-Mach. Syst. (1969)
- 8. S.H. Carter et al. Artificial muscles from fishing line and sewing thread, Sci. (2014)
- 9. W.W. Lee et al. A neuro-inspired artificial peripheral nervous system for scalable electronic skins, Sci Robot. (2019)
- 10. M. Duduta et al. Realizing the potential of dielectric elastomer artificial muscles, Proc. Natl. Acad. Sci. U.S.A.(2018)
- 11. X. Zhang et al. A Framework for Hand Gesture Recognition Based on Accelerometer and EMG Sensors, Trans. Syst. Man Cybern. A.(2011)
- 12. Jaeyoung Park, Inchan Hwang, Woochan Lee, Wearable Robotic Glove Design Using Surface-Mounted Actuators, Front. Bioeng. Biotechnol., 25 September 2020, Sec. Bionics and Biomimetics, Volume 8 2020
- 13. Bin Fang, Di Guo, Fuchun Sun, Huaping Liu, Yupei Wu, A Robotic hand-arm teleoperation system using human arm/hand with a novel data glove.
- S. S. Dheeban, D. V. Harish, A. Harivignesh, M. Prasanna, N. Senthil Kumar, Gesture Controlled Robotic Arm, The International Journal Of Science & Technoledge (ISSN 2321 – 919X)
- 15. Anughna N, Ranjitha V, Tanuja G, Design and Implementation of Wireless Robotic Arm Model using Flex and Gyro Sensor, International Journal of Recent Technology and Engineering (IJRTE) ISSN: 2277-3878, Volume-8 Issue-5, January 2020 [12] U. D. Meshram and R. Harkare, "FPGA Based Five Axis Robot Arm Controller," International Journal of Electron-ics Engineering, Vol.2, No.1, 2010, pp. 209-211. [13] J. Bernstein, H. R. Everett, L. Feng, and D. Wehe, "Mobile Robot Positioning Sensors & Techniques," Journal of Robotic Systems, Special Issue on Mobile Robots, Vol. 14, No. 4, pp. 231-249.
- M. naveau, m.kudruss, o.stasse, c.kirches, k.mombaur, and p.soueres. A Reactive Walking Pattern Generator Based on Nonlinear Model Predictive Control. IEEE ROBOTICS AND AUTOMATION LETTERS, 2(2377-3766), 10 -17. 2016.

Anti-Theft Vehicle Tracking System

¹Shreyash Kamble, ²Amol Korea, ³Kunal Korade, ⁴Mr.A.A.Garad. ^{1,2,3}UG Student, SVERI's Collegeof Engineering, Pandharpur. ⁴Professor, SVERI's College of Engineering, Pandharpur,

Abstract : The Anti-Theft Vehicle Tracking System is designed to enhance vehicle security by integrating various components. The system utilizes an Arduino UNO as the central control unit, interfacing with a GSM Module (SIM800I) for communication and a GPS Module (NEO-6M) for accurate location tracking. An Ignition Switch and Ignition Sensor are incorporated to detect the vehicle's ignition status. A 12V Power Supply ensures continuous operation, while a DC to DC voltage converter maintains stable power for the components. The system features a 5-channel relay for controlling various vehicle functions and a BO Motor for implementing anti-theft mechanisms. Overall, this integrated system provides a comprehensive solution for tracking, securing, and managing vehicles through a combination of GPS technology and intelligent control components.

Keywords:- Anti-Theft, Vehicle Tracking, GPS Technology, Ignition Detection, Location Tracking, Security System, Realtime Monitoring, Intelligent Control, Vehicle Security

I. INTRODUCTION

The Anti-Theft Vehicle Tracking System is a sophisticated solution designed to address the rising concerns related to vehicle security. As instances of vehicle theft continue to escalate, there is an increasing need for advanced technologies that can provide robust protection and facilitate efficient tracking. This project introduces an integrated system centered around the Arduino UNO microcontroller, incorporating key components such as the GSM Module (SIM8001), GPS Module (NEO-6M), Ignition Switch, Ignition Sensor, 12V Power Supply, DC to DC Voltage Converter, BO Motor, and a 5-Channel Relay.

The primary objective of this project is to develop a comprehensive anti-theft system that not only tracks the location of the vehicle in real-time but also employs intelligent control mechanisms to enhance security. By leveraging GPS technology, the system ensures accurate and continuous tracking, enabling users to monitor their vehicles remotely. The integration of communication modules allows for seamless interaction with the system, facilitating commands and status updates through mobile networks.

Furthermore, the project incorporates features such as ignition detection, power regulation, and versatile control over various vehicle functions, adding layers of security to prevent unauthorized access. The introduction of an anti-theft mechanism, implemented through the BO Motor and the 5-Channel Relay, aims to deter potential thieves and enhance overall vehicle protection.

In summary, this project addresses the critical need for advanced anti-theft and tracking solutions in the realm of vehicle security. By combining cutting-edge technology and intelligent control, the Anti-Theft Vehicle Tracking System offers a comprehensive approach to safeguarding vehicles and providing peace of mind to owners.

II. LITERATURE REVIEW

The literature on vehicle tracking and anti-theft systems underscores the growing importance of integrating advanced technologies. Studies highlight the use of GPS for accurate real-time tracking, communication modules like GSM for remote interaction, and microcontrollers such as Arduino UNO for system management. Ignition detection, stable power supply, and intelligent control mechanisms are identified as key factors in developing effective anti-theft solutions. The existing literature emphasizes the need for integrated systems that go beyond tracking, incorporating features to enhance security and deter potential theft. The Anti-Theft Vehicle Tracking System presented in this project aligns with these findings, providing a comprehensive solution to address vehicle security challenges.

III. MATERIALS AND METHODS

3.1 Components:

The Anti-Theft Vehicle Tracking System comprises several integral components. At its core is the Arduino UNO, serving as the central microcontroller for processing and overall system control. Facilitating remote communication, the GSM Module (SIM800I) enables interaction with the system, while the GPS Module (NEO-6M) provides precise real-time location tracking. Detection of the vehicle's ignition status is achieved through the Ignition Switch and Ignition Sensor. Ensuring continuous operation, the 12V Power Supply is a vital element, regulated by the DC to DC Voltage Converter for stability. The BO Motor is implemented to enforce anti-theft measures within the vehicle, and the 5-Channel Relay allows versatile control over various functions. Together, these components form a comprehensive system aimed at enhancing vehicle security through advanced tracking, communication, and intelligent control mechanisms.

3.1.1 Arduino UNO:

The Arduino UNO is the central microcontroller at the heart of the Anti-Theft Vehicle Tracking System project. Its primary role is to process data and control various components within the system. Built around the ATmega328P microcontroller, the Arduino UNO boasts digital and analog pins for versatile input/output, programmable memory for code execution, and USB connectivity for both programming and power supply. In this project, the Arduino UNO serves as the central control unit, interfacing with the GPS module to track the vehicle's location, managing communication with the GSM module for remote interaction, and controlling output devices such as the BO Motor and 5-Channel Relay to implement anti-theft measures and manage different vehicle functions. Programmed using the Arduino IDE in C/C++, it offers flexibility, ease of integration, and compatibility with various sensors, making it a pivotal component in creating a comprehensive and intelligent vehicle tracking and security system.

3.1.2 GSM 800L Module:

The GSM Module (SIM8001) serves as a vital communication link in the Anti-Theft Vehicle Tracking System project. This compact module is designed to enable remote interaction with the system through GSM/GPRS communication. It requires a SIM card for mobile network connectivity and is equipped with an antenna connector for optimal signal reception. In the project, the GSM Module facilitates the exchange of commands between the system and users remotely, enhancing the overall functionality of the vehicle tracking system. Utilizing AT commands for communication, the module is integrated with the Arduino or microcontroller and operates within a specific voltage range. Its role extends to providing status updates and alerts through mobile networks, contributing to the system's effectiveness in real-time tracking and security management. Compact, flexible, and designed for easy integration, the GSM Module plays a pivotal role in establishing seamless communication for the project's success.

3.1.3 GPS NEO-6M Module:

The GPS Module (NEO-6M) is a critical component in the Anti-Theft Vehicle Tracking System project, providing precise real-time location tracking capabilities. This module is designed to interface with the Arduino UNO and offers the following key features:

The GPS Module (NEO-6M) is a crucial element in the Anti-Theft Vehicle Tracking System project, offering accurate realtime location tracking capabilities. Specifically designed to interface seamlessly with the Arduino UNO, this module provides essential features for the project's success. The NEO-6M relies on signals from GPS satellites to calculate and transmit the vehicle's coordinates, contributing to the system's ability to monitor the vehicle's whereabouts. Equipped with a UART interface, it allows straightforward communication with the Arduino, and its compact design facilitates easy integration into the project. With low power consumption and rapid time- to-first-fix characteristics, the NEO-6M ensures efficient and reliable tracking performance. Its role in the project is fundamental, forming the backbone of the location tracking functionality that enhances the overall security and management capabilities of the Anti-Theft Vehicle Tracking System.

3.1.4 Ignition Key/Sensor:

The Ignition Switch and Ignition Sensor are integral components in the Anti-Theft Vehicle Tracking System, contributing to the project's ability to detect and respond to the vehicle's ignition status. The Ignition Switch serves as a manual control, allowing users to interact with the system by manually activating or deactivating the vehicle's ignition. Complementing this, the Ignition Sensor provides automated feedback on the ignition status. This real-time information is crucial for anti-theft measures, enabling the system to respond promptly to changes in the vehicle's operational state. By integrating both the Ignition Switch and Sensor, the project gains a comprehensive understanding of the vehicle's ignition status, enhancing its capability to implement effective security measures and allowing for seamless interaction between users and the tracking system. This dynamic combination of manual and automated ignition detection significantly contributes to the overall security and functionality of the Anti-Theft Vehicle Tracking System.

3.2 Methodology:

The methodology for developing the Anti-Theft Vehicle Tracking System is characterized by a comprehensive and iterative approach, ensuring the integration of diverse components to create a sophisticated and reliable solution.

The project initiation involves an exhaustive analysis of requirements, encompassing the need for accurate real-time tracking, seamless remote communication, and robust anti-theft measures. This analysis serves as the foundation for subsequent decision-making and system design. Careful consideration is given to component selection, with the Arduino UNO chosen as the central control unit due to its versatility and widespread use in embedded systems. The GPS Module (NEO-6M) is integrated to provide precise real-time location data, while the GSM Module (SIM800I) facilitates bidirectional communication for remote interaction.

The architecture of the system is carefully conceptualized, and a block diagram is constructed to visually represent the interconnections between the various components. This diagram serves as a blueprint for the physical integration of modules and ensures a systematic approach to wiring. Wiring and connection are executed meticulously, with an emphasis on establishing reliable and efficient communication pathways between the Arduino UNO and each module. The physical integration of components is a critical step, ensuring the seamless interaction required for the system to function cohesively. Programming the Arduino involves the development of code that interprets GPS data, manages GSM communication, and controls anti-theft mechanisms. The code undergoes iterative refinement to optimize responsiveness and overall system efficiency.



Fig 1. Block Diagram



Fig 2. Hardware Structure

The integration of anti-theft measures involves the strategic incorporation of the BO Motor and 5- Channel Relay. These components are programmed to respond to specific commands, providing a means for remote control over vital vehicle functions to enhance security.

Rigorous testing is conducted at each stage of development, encompassing simulated and real- world scenarios. Calibration is performed to fine-tune components, particularly the GPS module, ensuring accuracy in location data and minimizing errors. Comprehensive documentation is maintained throughout the development process, encompassing detailed wiring diagrams, code explanations, and an operational guide. This documentation serves as a valuable resource for troubleshooting, maintenance, and future enhancements.

Iterative testing, including user feedback loops, refines the system's performance and usability. The project output includes real-time location tracking, remote communication capabilities through mobile networks, and effective implementation of anti-theft measures.

In essence, this methodology combines meticulous planning, precise integration, and continuous refinement to ensure the Anti-Theft Vehicle Tracking System is not only functional and reliable but also adaptable to the evolving needs of users and the challenges of vehicle security.

IV. RESULTS AND DISCUSSION

4.1 Result:

The implementation of the Anti-Theft Vehicle Tracking System has yielded notable results in enhancing vehicle security and tracking capabilities. The integration of key components, including the GPS Module (NEO-6M) and GSM Module (SIM8001), has proven successful in providing accurate and real-time information to users. The user-friendly interface, exemplified by the output of receiving updates through messages, significantly enhances the user experience. This direct communication method ensures seamless interaction with the system, empowering users with accessible and actionable information.





Fig 3 Output through Messages

4.2 Discussion:

Discussions surrounding the project's results emphasize the effective integration of components and the practical implications for vehicle security. The bidirectional communication not only provides real-time information but also grants users remote control capabilities, showcasing the project's practical utility. Looking forward, considerations for future enhancements and adaptations become relevant. Exploring additional features or incorporating emerging technologies could further elevate the system's capabilities, prompting ongoing discussions in the dynamic field of IoT-based security systems. Ethical considerations, particularly concerning user privacy and responsible technology use, emerge as critical discussion points. Ensuring transparency, obtaining user consent, and implementing responsible data handling practices are integral for the continued success and acceptance of such systems.

In conclusion, the Anti-Theft Vehicle Tracking System has delivered tangible results, sparking discussions about its practical implications, future developments, and ethical considerations within the context of IoT-based vehicle security systems.

V. CONCLUSION

The Anti-Theft Vehicle Tracking System project has successfully achieved its objectives by integrating components like the GPS Module and GSM Module. The system provides accurate real-time tracking, bidirectional communication through mobile messages, and features anti-theft measures. The user-friendly design, exemplified by updates received via messages, enhances usability. Future considerations include potential enhancements and ethical considerations regarding user privacy and responsible technology use. Overall, the project represents a practical and effective application of technology in addressing vehicle security challenges.

VI. REFERENCES

- 1. Ramadan M N, Al-Khedher M A, Al- Kheder S A. Intelligent Anti-Theft and Tracking System for Automobiles, International Journal of Machine Learning and Computing, 2012 Feb, 2(1), pp. 88-92.
- 2. Pooja S. Vehicle Tracking System Using GPS, International Journal of Science and Research, 2013 Sep, 2(9), pp. 128-30.
- 3. V.Ramya, B. Palaniappan, K. Karthick, "Embedded Controller for Vehicle In-Front Obstacle Detection and Cabin Safety Alert System", International Journal of Computer Science.
- 4. E. D. Kalpan, Understanding GPS: Principles and Applications, Artech house Publishers, ISBN 0890067937, February 1996.
- 5. Gupta P, Sutar S S. Study of Various Location Tracking Techniques for Centralized Location, Monitoring & Control System, IOSR Journal of Engineering, 2014 Mar, 4(3), pp. 27-30.
- 6. Akuwan Saleh, 2008, the Motor Vehicle Safety With Identification Method and SMS, a National Seminar on Information Technology Application (SNATI 2008) Yogyakarta, B7-B14.
- 7. N. Kiruthiga, L. latha, S. Thangasamy (2015) Real Time Biometrics based Vehicle Security System with GPS and GSM Technology. Procedia Computer Science 47, 471-479.
- 8. Montaser N. Ramadan, Mohammad A. Al-Khedher, Sharaf A. Al- Khedher (2012) Intelligent Anti-Theft and Tracking System for Au- tomobiles. International Journal of Machine learning and Computing 2 No.1.
- 9. J.R.Shaikh, S. M. Kate (2012) ARM7 based Smart Car Security System. International of Engineering and Trends and Technology 3 Issue2.
- 10. Gunaganti Ganesh (2014) Anti-Theft Contrl System of a car using ARM7. International Journal of Engineering and Science Research 4, 480-484 [5]

Indian Currency Detection And Recognization From Image By Using Matlab

¹Ms.Megha Arun Sontakke ¹Assistant Professor, SVERI's COE, Pandharpur, India

Abstract- There is Great technological advancement in printing and scanning industry made counterfeiting problem to grow more vigorously. Because of this, fake money has an impact on the economy and devalues real cash. Therefore, it is crucial to identify counterfeit money. The majority of the earlier approaches rely on image processing techniques and technology. Using these techniques to find fake money is less effective and takes longer. To address the aforementioned issue, we suggested a deep convolution neural network for detecting counterfeit cash. By looking at the photos of the currency, our study detects the counterfeit money. Two thousand, five hundred, two hundred and fifty Indian currency note data sets are used to train the transfer learnt convolution neural network in order to learn the feature map of the currencies. The network is prepared to detect fraudulent cash in real time after learning the feature map. With less time spent, the suggested method effectively detects fake currencies of 2000, 500, 200, and 50.

Keywords: Convolutional Neural Network, Deep Learning, Feature Extraction, Currency Detection

I. INTRODUCTION

The Reserve Bank of India deals with damaged or counterfeit currency notes every year. Managing a significant amount of fake currency presents more challenges. Therefore, the process of identifying notes is made simpler and more effective by utilizing machines with the help of human professionals. Every time they utilize the ultraviolet light-containing gadget, which is used to detect fake notes (using a bank as an example), they must be able to recognize the denomination. The bank employee places the paper money note on the device and attempts to determine the denomination and verify its authenticity by looking up the note's serial number, watermark, and other details. This increases the work of the employee Technology is growing very fast these daysAs a result, the banking industry is likewise becoming more and more modern. This creates a pressing demand for automatic teller machines and automatic goods seller machines to detect phony cash. Block 2 Therefore, it is crucial that we extract the features from the currency note image and use the right method to increase the note's recognition accuracy. Here, we employ a straightforward approach that functions well. Using a digital camera, the picture of the banknote was taken. The multiple feature extraction procedure is used to retrieve the note's hidden features. Processing of the obtained image now takes place using ideas such as image segmentation, edge information extraction, and feature extraction. The ideal tool for computational work is MATLAB., As a result, the banking industry is likewise becoming more and more modern. This creates a pressing demand for automatic teller machines and automatic goods seller machines to detect phony cash. Block 2 Therefore, it is crucial that we extract the features from the currency note image and use the right method to increase the note's recognition accuracy. Here, we employ a straightforward approach that functions well. Using a digital camera, the picture of the banknote was taken. The multiple feature extraction procedure is used to retrieve the note's hidden features. Processing of the obtained image now takes place using ideas such as image segmentation, edge information extraction, and feature extraction. The ideal tool for computational work is MATLAB.. Here, the image is captured by a simple digital camera such that all the features are highlighted .Manual testing of all notes in transactions is very time consuming, untidy process and also there is a chance of tearing while handing notes. Nobody's confidence in manual recognition can ever reach 100%. In monetary transactions, one of the main issues is counterfeit or fake currency. It is becoming a significant obstacle for a nation like India. With the use of the newest hardware, it is now simple for someone to produce phony notes thanks to advancements in printing and scanning technologies. The traditional procedure of identifying counterfeit notes is messy and time-consuming, hence automated methods are required for the efficient detection of cash.

III. REQUIREMENTS FOR SOFTWARE AND HARDWORK

Hardware and software specifications

- 1) Software
- 2) Tensor flow
- 3) Matlab IDE
- 4) Matlab Libraries
- 5) Sci kit Libraries
 - Windows 7 and Windows 10 are the operating systems for the hardware (PC with display).
 - Processors: Any Intel or AMD x86-64 CPU will do;

- Minimum disk space required for MATLAB is 2.6 GB;
- Minimum RAM: 4 GB Suggested 8 GB Graphics: You don't need a specialized graphics card. It is advised to use a hardware-accelerated graphics card with 1GB of GPU memory that supports Open GL 3.3..



III .PROPOSED METHODOLOGY

Fig. 1 Block diagram of proposed system

Image Acquisition: The image acquisition is to acquire a digital image. An image sensor and the ability to digitize the signal the sensor produces are needed. Since processing cannot occur without a picture, acquiring images is always the initial step in the work flow sequence for image processing. After the image has been obtained, various methods of processing can be applied to the image to perform the many different vision tasks. There are various ways to acquire images such as with the help of camera or scanner. The acquired image should retain all the features. Every feature should be preserved in the obtained image.

Preprocessing: Operations involving images at the lowest level of abstraction, when both the input and the output are intensity images, are commonly referred to as pre-processing. Pre-processing aims to improve the image data by reducing undesired distortions and enhancing certain key visual elements that are necessary for additional processing. The main goal of the pre-processing to enhance the visual appearance of images and improve the manipulation of data sets. The correction of distortion, deterioration, and noise introduced during the imaging process is known as image pre-processing, or picture restoration.Interpolation is the technique mostly used for tasks such as zooming, rotating, shrinking, and for geometric corrections. Removing the noise is an important step when processing is being performed. However, noise affects segmentation and pattern matching.

Gray scale conversion: conversion of colour image to a gray scale image requires more knowledge about the colour image. A pixel colour in an image is a combination of three colours red, green& blue (RGB) similarly a gray scale image can be viewed as a single layered image.

Segmentation: The technique of dividing a digital image into several parts is called segmentation. Usually, it's employed to set items apart from backdrops. Here, the image is segmented using an edge-based method. We must separate the small objects—the digits on the serial numbers—from the vast banknote images based on the binary images that were subjected to the morphological modifications. In order to accomplish this, the huge binary banknote images were first subjected to the Sobel operator to separate the regions of interest (ROIs) from the background. The ROIs were then extracted to yield the little digits, which are the objects utilized to build the templates of the digits and for recognition. The process of image segmentation subdivides an image into its individual objects or areas. The problem being solved determines the level of sub division that is applied. For monochrome images, segmentation algorithms are often based on one of two fundamental

characteristics of image intensity values: similarity or discontinuity. The first category uses a method of segmenting an image according to sharp shifts in intensity, such edges. The method used in the second category consists of dividing an image into similar parts based on a predetermined set of parameters.

Feature Extraction and Comparison: Comparing and Extracting Features: One particular type of dimensional reduction is feature extraction. An algorithm will convert input data into a reduced representation set of features when the amount of data is too big to handle and is thought to be highly redundant. Feature extraction is the process of converting the input data into a collection of features. It is anticipated that the features set will extract pertinent information from the input data in order to accomplish the intended task utilizing this smaller representation rather than the full-size input if the features extracted are correctly selected. Haar-like features are digital imaging features which are working only on image intensities (i.e) RGB values of each and every pixel in an image. A Haar-like feature adds up the pixel intensities of each surrounding rectangular region at a specific position inside a detection window and computes the difference between these sums. Subheadings within an image are then categorized using it. This work uses a currency-containing picture database.

Output : The currency recognition result will be provided in text format.

Methodology

Our objective in this proposed system is to concentrate on detecting counterfeit currencies that are prevalent in the Indian market. By removing the currency note's security thread feature, we are able to identify counterfeit money in our work. The most often used deep neural network technique for identifying counterfeit money is transfer learning with Alex net. Convolutions, max pooling, dropout, ReLU activations, and fully-connected layers make up the Alex net. The layers are shown in figure 1 below. The final three layers of transfer learning are adjusted to meet the needs of our suggested work. For the purpose of precisely extracting the latent feature from the image, the "fc8_ layer is adjusted using a weight learning factor and a bias learning factor. The weight gives the impact of the input on the network and bias is used to adjust the output with weighted sum of inputs to the neuron. The weight and bias learning factors increase the accuracy of the learning feature. Since the input image is typically two-dimensional, so is the object map. A predetermined number of filters are applied to the input image by the convolution layer. An array of numbers the same size as the input image makes up the filter. This filter multiplies each filter number by the matching pixel value in which it is placed, travels to each pixel value, and does multiplication by element calculations. As seen in Figure 2, each multiplication is added for each pixel position to create an element of the output matrix known as a feature map.



Fig. 2 Convolution operation

Convolutional process The stride is another factor to take into account. This is the number of pixels that are moved by the strainer matrix whenever it is shifted on the input image. Generally speaking, when the passage is 1, we can leave all of the spatial samples that are dropping for POOL levels. Following each convolution layer is a component of the smart activation function. In order to identify more intricate linear regression functions, network linearity is broken using the activation function. max (0, x) is the function that activates a rectified linear unit. The total values of the secondary matrix value and the spatial contact of the output data from the single feature map aid in describing the physical domain's value. Choose maximum grouping, which merely displays the region's maximum activation. the 2x2 matrix's maximum coupling procedure. max pooling method Fully connected layers are used to do high-level neural network reasoning after multiple layers of convolutional and maximum pooling. "Fully coupled" refers to a situation in which every neuron in the preceding level is connected to every neuron in the subsequent level. It is possible to extract high level features from the pooling and

convolutional layers. The determination of a fully connected layer is to use these characteristics to sort the original image into different types based on a training dataset.

Database:

The proposed method uses transfer learned Alex network with adam optimization. A database of 100 photographs per currency is used to test the constructed network. Fifty of the images are taken as captured images, and the remaining fifty are augmented images. The 50, 200, 500, and 2000 real notes of Indian currency are displayed in figures in the database.



Fig. 3 : Real currency rate detection

Training The Network

NETWORK TRAINING A database containing images of money notes is created in order to train the network. Using augmentation, 100 images are created for every note. The purpose of augmentation processes like resizing and rotating is to expand the number of data bases. Following the process of augmentation, all currency photos are annotated and labeled before being saved in a distinct folder. The images and network are now prepared for training. The network learns the characteristics of actual money notes of 2000, 500, 200, and 50 after the training procedure is complete. The Intel Core i3 64-bit processor running at 2GHz with 4GB of RAM, Windows 10 and Matlab 2018a is the system utilized for training. The goal of the suggested plan is to use the fully linked layer to learn more features. Hyperparameter is employed for this optimization in order to increase the learning feature. The technique is evaluated in real time with a camera; as soon as the image is taken, the

network begins to learn the attributes of the input dollar note, compares them with the features it has learnt, and outputs a result labeled as either "Fake Note" or "Real Note."



IV RESULT

Fig. 4 Training progress



Fig. 5 Output of 50rs



Fig. 6 Output of 100rs

V CONCLUSION

In this research, the technique of image processing is used to detect phony Indian rupee notes. This is the inexpensive setup. Additionally, the technology yields reliable and accurate findings. (Block 7) It takes little time or effort to identify a bogus note. This system uses a digital camera for input, and a PC is used to display the results. When compared to other instruments and machinery, image processing improves the accuracy of the analysis of Currency images while also being a more economical and time-efficient way. This investigation makes use of MATLAB software. Research in this area is conducted on a daily basis, and different image processing methods are created to produce results that are more accurate. The suggested method extracts features from photos of Indian currency with effectiveness. The image's extracted features will be utilized to both recognize and verify the worth of the cash. In future an android application can be developed where we can easily get the images by just accessing the camera which is very simple than storing in file and again doing all the tasks. There is a lot of scope in terms of future work in this area . Further more, the original idea of creating wearable device can be pursued.

REFERENCES

- 1. Laavanya, M., Vijayraghvan, V. (2019, December). Real Time Fake Currency Note Detection using Deep Learning. International Journal of Engineering and Advanced Technology, 9(1S5), ISSN: 2249-8958.
- 2. Vora, K., Shah, A., Mehta, J. (2015, April). A Review Paper on Currency Recognition System. International Journal of Computer Applications, 115(20), ISSN: 0975-8887.
- 3. Sahu, S., Verma, T. (2016, June). Identification of Paper Currency Techniques: A Survey. International Journal of Science Technology & Engineering, 2(12), ISSN (online): 2349-784X.
- 4. Pathrabe, G. T., Karmore, M. S. (2011, May to June). A Novel Approach of Embedded System for Indian Paper Currency Recognition. International Journal of Computer Trends and Technology, ISSN: 2231-2803.
- 5. Mirza, R., Nanda, V. (2012, April). Characteristic Extraction Parameters for Genuine Paper Currency Verification Based on Image Processing. IFRSA International Journal of Computing, 2(2).
- 6. Pathrabe, T., Bawane, N. G. (2011). Feature Extraction Parameters for Genuine Paper Currency Recognition & Verification. International Journal of Advanced Engineering Sciences and Technologies, 2, 85-89.
- 7. Prasanthi, B. S., Setty, D. R. (n.d.). Indian Paper Currency Authentication System using Image processing. International Journal of Scientific Research Engineering & Technology (IJSRET), ISSN 2278 – 0882.
- 8. Aruna, D. H., Bagga, M., Singh, B. (2015, March). A survey on Indian currency note denomination recognition system. International Journal of Advance Research In Science And Engineering, ISSN: 2319-8354(E).
- 9. Surya, S., Thailambal, G. (n.d.). Comparative Study on Currency Recognition System Using Image Processing. International Journal Of Engineering And Computer Science, ISSN: 2319-7242.
- 10. Suresh, I. A., & Narwade, P. P. (2016). Indian currency recognition and verification using imag e processing. International Research Journal of Engineering and Technology (IRJET), 3(6), 87-91.
- Agasti, T., Burand, G., Wade, P., & Chitra, P. (2017, November). Fake currency detection using image processing. In IOP Conference Series: Materials Science and Engineering (Vol. 263, No. 5, p. 052047). IOP Publishing
- 12. Babu, P. A., Sridhar, P., & Vallabhuni, R. R. (2022, February). Fake currency recognition system using edge detection. In 2022 Interdisciplinary Research in Technology and Management (IRTM) (pp. 1-5). IEEE.
- Tajane, A. U., Patil, J. M., Shahane, A. S., Dhulekar, P. A., Gandhe, S. T., & Phade, G. M. (2018, February). Deep learning based Indian currency coin recognition. In 2018 International Conference On Advances in Communication and Computing Technology (ICACCT) (pp. 130-134). IEEE
- Chowdhury, U. R., Jana, S., & Parekh, R. (2020, March). Automated system for Indian banknote recognition using image processing and deep learning. In 2020 International Conference on Computer Science, Engineering and Applications (ICCSEA) (pp. 1-5). IEEE.
- 15. Abburu, V., Gupta, S., Rimitha, S. R., Mulimani, M., & Koolagudi, S. G. (2017, August). Currency recognition system using image processing. In 2017 Tenth International Conference on Contemporary Computing(IC3)(pp.1-6)IEEE

Secure Attendance Tracking With Facial Recognition And Liveness Detection

¹Bhavana Bidarkar, ²Meer Hyder Ali, ³SK Tazim, ⁴MD Afnan Khan

^{1, 2,3,4}Dept of Computer Science and Engineering Guru Nanak Dev Engineering College Bidar, Karnataka

ABSTRACT— Biometric technology utilizing facial recognition has gained widespread acceptance. An effective face identification system needs to distinguish real faces from fraudulent attempts using printed photos or digital images. A reliable method to combat spoofing involves assessing face liveliness by monitoring eye blinking and lip movements. However, this approach proves inadequate against video-based replay attacks. To address this limitation, a novel approach is introduced in this study, combining face liveness detection with a CNN (Convolutional Neural Network) classifier. The anti-spoofing mechanism comprises two components: the first module analyzes eye blinking and lip movement, while the second module employs the CNN classifier. Training data for the CNN classifier can be sourced from various publicly accessible datasets. These two modules are integrated sequentially and implemented in a user-friendly facial recognition application on the Android platform. The experimental outcomes demonstrate the module's efficacy in detecting a wide array of facial spoofing tactics, including the use of posters, masks, and smartphones.

Keywords— face recognition, face spoofing, CNN classifier, face liveness detection, deep learning

I. INTRODUCTION

Biometrics stands as a widely embraced method for authentication in current applications, with a specific focus on the rise of facial recognition technology due to its convenience and precision. The incorporation of this technology into everyday devices like smartphones, tablets, and laptops reflects its broad acceptance. Operating by capturing facial images using a device's camera and analyzing them with specialized algorithms, facial recognition technology validates an individual's identity against an existing database.

Despite its effectiveness, the susceptibility of facial recognition systems to spoofing attacks presents a notable challenge. These attacks take advantage of the system's incapacity to distinguish between authentic faces and deceptive efforts, such as those involving masks, videos, or photographs. Adding to the problem is the easy access to facial data on social media, enabling individuals to manipulate the system.

The range of face spoofing attacks involves both static and dynamic methods, covering techniques like 2D and 3D demonstrations. While eye blink identification serves as a precise indicator of liveness, it proves insufficient against advanced video replays. Another approach requires challenges and responses during a video sequence, yet this method demands additional user input, potentially affecting the overall user experience.

Motion detection techniques aim to identify vital signs through the assessment of facial movements, but they may struggle against dynamic rendering attacks, particularly those involving videos. While 3D cameras and photo plethysmography offer dependable anti-spoofing measures, their limited availability on common devices like smartphones poses practical difficulties. Previous research has explored texture pattern analysis utilizing color variations, yet these methods are sensitive to room lighting conditions. Deep learning and convolutional neural networks (CNN) emerge as potential solutions, although concerns arise due to the inconsistency of features for CNN recognition.

In response to these challenges, this article suggests an advanced method for detecting face liveness. This method combines blink and lip movement detection with CNN analysis, presenting a multifaceted approach to distinguish between genuine and false faces. Importantly, this proposed method is known for its accuracy, independence from additional hardware, and strong performance in real-world situations. Its seamless implementation and real-time detection capabilities position it as a comprehensive solution to combat various spoofing attacks across different environments. The contributions of this study lie in the application of CNN and deep transfer learning for precise detection, providing an effective, adaptable, and user-friendly approach to address the evolving landscape of facial recognition security.

II. RELATED WORK

Face spoofing attacks often involve presenting a face using devices like smartphones or tablets, resulting in poor face textures that can be easily detected by assessing HSV feel and image quality. The limited color reproduction of screen media compared to real faces, coupled with color variations in represented faces, can be analyzed through examining chroma channel features. Extracting Local Binary Pattern (LBP) information from various color spaces, especially in poorly lit areas, is crucial in determining the most valuable micro-texture representation.

Software-based anti-spoofing detection methods have become popular due to their cost-effectiveness and high precision. Early systems required users to blink, move their lips, or follow specific instructions to counter print attacks but compromised user experience and did not effectively address video playback attacks. Researchers have explored hand-crafted feature analysis systems to address these challenges, although concerns about their applicability in real-world scenarios persist. A continuous study on liveness detection using various benchmark datasets aims to enhance detection accuracy.

Unlike software-based methods, hardware-based approaches employ distinctive sensors for image acquisition to enhance the differentiation between real faces and spoofing attacks, ensuring a more stable detection outcome. Techniques such as multispectral, infrared, and remote Photo Plethysmography leverage reflectivity differences to achieve accurate liveness detection. However, these methods have strict collection requirements and complex hardware installations, limiting their widespread use. Devices based on density, like time-of-flight cameras, effectively handle 2D attacks but struggle with 3D attacks. A liveness detection system utilizing a light field camera has been proposed, capable of detecting various spoofing attacks despite its high cost and sensitivity to lighting conditions. The primary drawback of hardware-based detection methods lies in the cost and complexity associated with acquiring equipment, installation processes, and broad adoption.

III. METHODOLOGY

Our anti-spoofing model consists of essential components: the liveness detection and CNN classifier. The operational sequence of this model is quite simple. The initial input is subjected to liveness verification, examining eye blinks and lip movements. If these cues are identified, the input proceeds to the CNN classifier unit to validate the face's authenticity. Verification of a genuine face occurs only when the input successfully passes through both sections. The methodology comprises several fundamental stages, with the CNN classifier unit's development including data gathering, preprocessing, model training, model assessment, and testing.

The module for detecting facial indications of liveness is further divided into two parts: blink detection and lip motion detection. For lip motion detection, the lip-movement-net module is employed with a basic Recurrent Neural Network (RNN) founded detection algorithm in Python. This algorithm evaluates lip movements in a 1-second video clip to determine if the subject is speaking in real-time, applicable to video content or webcam feeds. The process of lip movement detection involves establishing a filter to pinpoint the upper and lower lip positions and then computing the gap between them.

Regarding eye blink detection, a previously researched module is utilized. By using an eye area filter, this module detects blinking by observing the eye area within a person's facial image. Subsequently, an eye openness classification is implemented, estimating the likelihood of eye opening in the input image. The classification is evaluated based on the variation between the highest and lowest eye openness levels. A substantial difference signifies blinking, with a dataset of faces displaying both opened and closed eyes employed to create the eye classification module.

The dataset for training the CNN algorithm is compiled from numerous sources as referenced in Table 1. This dataset contains raw images depicting various spoofing techniques, such as posters, masks, videos, and pictures. The dataset displays a broad spectrum of room lighting, image quality, and backgrounds, as illustrated in Figure 1. The diversity within the dataset guarantees the model's resilience against various forms of spoof attempts.

Dataset	Subject	Item
NUAA Photo	16	5,000
Imposter		
CASIA-SURF	1,000	21,000
SiW	165	4,620
CSMAD	12	16,060
MSU-USSA	1,140	102
Total	2,333	46,782

TABLE I. DATASET ITEMS



Fig. 1 Sample Dataset

After the dataset is collected, we pre-process the data. Data processing is carried out in the following stages:

- 1. Extraction of mid-level features from images, we extract parts of the whole image such as eyes, nose, and mouth, separated in each image frame. One input image is extracted into nine different mid-level feature images.
- 2. Resize the image. All images are resized to 100x100 px size
- 3. Reducing image noise, we use a gaussian blur filter to reduce noise in the image.
- 4. Image segmentation, we are separating the background from the foreground on image objects.
- 5. Image augmentation, we do image augmentation by zooming +/-10% and rotation +/-10%
- 6. Split the dataset. The image dataset is divided by 80% for model training and 20% for model validation.

At the model training stage, everything is processed in thepre- processing stage. The data is used as input for training the model. We train the model by transferring learning from one dataset to another. All datasets are pre-processed and trained separately without being combined. Through this approach, the training process will be lighter and faster without affecting the overall results.

The main of our strategy is a modified dynamic CNN structure. In the following study, we will denote a multi- layer perceptron as MLP (m, n...), where m, n... are the number of parameters of each layer of the MLP. We define the shape parameter area and present parameter area as Sx R10 and P xR10, respectively Our model begins with just two EdgeConv [x] modules, and equally together with all k=10 nearest neighbors and maximum characteristic aggregation type. The primary module includes MLP (6,32,64,128) along with also the latter one MLP (64 + 128,256). Outputs of both EdgeConv modules are concatenated and passed forward. The version is subsequently broken up into two divisions, one regressing the present parameter SPxP and the form parameters, therefore xS of the input stage. The second division that arouses he form parameters xS divides the GP module using a Clustered Pooling module with an MLP (64 + 128,256) and can be accompanied closely by a different MLP sub-block MLP (1024,512,256,8). After passing through the neural network screen, the input will be processed at the pooling layer. The CNN infrastructure used has three neural network layers and three pooling layers. Since all the inputs have a similar file type, JPEG RBG, we use a matrix (4x4x3)for each pooling layer. Fig. 2 shows an overview of the CNN infrastructure in use.



Fig. 2 The CNN infrastructure model
Aside from Graph CNN, HSV Texture outcomes were examined from every frame recorded from the camera. This approach is required to expect video-based spoofing attacks in which the movie may escape the Graph CNN detector. Opportunely, there are essential differences in color texture, contrast, and light that differentiate video replays and actual faces. To distinguish this, the chromatic second feature concept is employed. Within this theory, every color channel HSV is figured. Standard deviation and skewness can be utilized to symbolize the chromatic attribute's value. This value reflects the amount of intensity or brightness of a color. Thus, this process is acceptable for spoof attacks from gadgets like tablets or smartphones. On the flip side, hue & saturation values are utilized to discover the subsequent shade patterns. Since printed photos commonly utilize CMYK (Cyan Magenta Yellow Key) color patterns, so the color patterns on the actual face will differ.

The face liveness detection module and CNN classification module are combined sequentially. First, facial input will be detected through the face liveness module. If there is a sign of facial activity, it will be forwarded to the CNN classification module. Face input is declared a real face if it passes the two modules. The merged module is converted to a TF-lite file for implementation on Android mobile devices. The module is implemented in an application based on a simple Android platform using a smartphone camera to test. The last process is testing the combined modules. We directly tested several face spoofing scenarios, such as using a poster or smartphone photo.

. During this research, the software tools used for data pre- processing, training, and model evaluation were OpenCV, Jupyter Notebook, TensorFlow, and Keras. Android Studio and TfLibrary are used for program deployment and testing. The development environment settings are on the Windows 10 platform, Intel Core I7 9750H, 32GB DDR-4 RAM, and NVidia Quadro T-1000 GPU 8GB GPU memory.

IV. RESULT AND DISCUSSION

In the data collection process, we found 46,782 images consisting of 2,333 subjects. This image consists of 26,546 spoof images and 20,236 real face images. These images have a variety of different resolutions, as well as backgrounds and lighting conditions. Before this image dataset is used to train the model, this image will be pre-processed in stages described in the methodology section. The results of this pre- processing stage can be seen in Fig. 3 & 4.



Fig. 4 Real face image dataset after pre-processing

From the accuracy table above, it is known that the algorithm has the highest average accuracy compared to the other three algorithms. Besides, the KNN algorithm also has an average accuracy level that is close to the average accuracy of the SVM algorithm.

Dataset	Training			Validation		
	Acc.	Pre.	Rec.	Acc	Pre	Rec
NUAA	78.67	77.89	79.56	76.61	75.90	77.56
CASIA	84.36	82.98	84.98	80.88	81.13	80.57
CSMAD	87.25	88.76	86.34	82.39	82.15	82.44
SiW	90,11	90,93	89,23	87.23	87.34	87,21
MSU	93,52	94,43	93,04	88.74	88.74	88.74

TABLE II. Training and validation result

After going through the pre-processing stage, the image dataset is used to train the model. The results of the model training are shown in table 3. As can be seen, the model was trained per the existing dataset by using transfer learning so that the knowledge gained from previous model training can be transferred. The results show that the better the dataset used to train the model results in the accuracy. After fitting the model with the existing dataset, the accuracy is 93.52%, 92.43% precision, and 94.04% recall on the training data.



Fig. 5 Blink detection frame by frame

While in validation data, get 88.74% accuracy, 88.74% precision, and 88.74% recall. Judging from the training results, we made a few changes to the designed model by adding a dropout layer of 0.1 in each after the pooling layer. We got slightly better validation results using the dropout layers at 90.39% accuracy, 90.84% precision, and 90.21% recall. We also test the face liveness detection module. This test is intended to ensure that the module works appropriately before integrating with the CNN analysis module and implemented. The results of this module are in the form of a wink or lip movement score. The score is a value from 0 to 1. This value is measured by the value of eye openness or the difference in the upper and lower lips' width. Based on the test results, we determined the value 0.5 as the threshold value. Where if the value =>0.5, then it is stated that there is a sign of activity on the face. Fig. 5 shows the sample test results of the face liveness detection module. The caption under each image shows the face liveness score. The first image (topmost) shows the image with the maximum eye openness value. Meanwhile, the bottom image shows minimal openness.

After testing the CNN analysis module and face liveness detection, we integrated the two modules. Integration is done sequentially. We created a simple simulation program using the Android platform's camera to test the final capabilities module integration. CNN module training results are exported in the TF-Lite form using the Tensor Flow Lite API then implemented into a native Android APK form. Table 4 shows the results of tests carried out with various real-world test scenarios. The test results show that the proposed model works well to block login attempts with face spoofing attacks such as using posters or replay attacks.

V. CONCLUSIONS

With so many devices using facial recognition biometric authentication, the need for face anti-spoof is an absolute must. This paper proposes using the CNN analysis model for image input combined with the face liveness detection module. Based on the results of module testing shows excellent results to prevent various types of face spoof attacks. We test various spoof face attacks tested included static attacks such as masks, photo posters or digital photos, and dynamic attacks such as video replays. Further research can explore parallel programming techniques that can speed up the time for facial recognition programs.

REFERENCES

- 1. P. Kavitha and K.Vijaya, "A Study on Spoofing Face Detection System," International Journal of Pure and Applied Mathematics, vol. 117, no. 22, pp. 205-208, 2017.
- 2. F. Sthevanie and R. kurniawan, "Spoofing detection on facial images recognition using LBP and GLCM combination," International Conference on Data and Information Science, Bandung, 2018.
- 3. D. Wen, "Face Spoof Detection with Image Distortion Analysis," IEEE Biometrics Compendium, vol. 10, no. 4, pp. 746-761, 2015.
- 4. Schiffman, H.R., "Sensation and Perception. An Integrated Approach", New York: John Wiley and Sons, Inc., 2001
- 5. B. Zinelabidine, K. Jukka and H. Abdenour, "Face Anti-spoofing based on Color Texture Analysis," 22nd IEEE International Conference on Image Processing (ICIP), Chicago, 2015.
- 6. Y. A. Rahman, M. Liu and L. M. Po, "Deep learning for face anti-spoofing: An end-to-end approach," Signal Processing: Algorithms, Architectures, Arrangements, and Applications (SPA), Hong Kong, 2017.
- 7. E. Alexey, "Algorithm for optimization of Viola–Jones object detection framework parameters," Journal of Physics Conference Series, no. 1, p. 945, 2018.
- 8. Z. Boulkenafet, J. Komulainen and A. Hadid, "Face Anti- Spoofing Based on Color Texture Analysis," Machine Vision Research, vol. 1, 2015.
- 9. R. Hasan, H. Mahmud and X. Y. Li, "Face Anti-Spoofing Using Texture-Based Techniques and Filtering Methods," Journal of Physics: Conference Series, 2019.
- I. B. Kusuma, A. Kartika, T. A. Budi, K. N. Ramadhani and F. Sthevanie, "Image Spoofing Detection Using Local Binary Pattern and Local Binary Pattern Variance," International Journal on Information and Communication Technology (IJoICT), vol. 4, no. 2, pp. 11-18, 2018.
- 11. K. Larbi, W. Ouarda, H. Drira, B. B. Amor and C. B. Amar, "DeepColorFASD: Face Anti Spoofing Solution Using a Multi Channeled Color Spaces CNN," International Conference on Systems, Man, and Cybernetics (SMC), Miyazaki, 2018.
- 12. A. Anjos, J. Komulainen, S. Marcel, A. Hadid and M. Pietik, "Face Anti-spoofing: Visual Approach," Handbook of Biometric Anti-Spoofing: Trusted Biometrics under Spoofing Attacks, London, Springer, 2014, pp. 65-82.

- 13. G. Pan, L. Sun and S. Lao, "Eyeblink-based Anti-Spoofing in Face Recognition from a Generic Webcamera," IEEE International Conference on Computer Vision, Rio de Janeiro, 2017.
- 14. S. Parveen, S. Mumtazah, M. Hanafi and W. Azizun, "Face anti- spoofing methods," Current Science, vol. 108, no. 8, pp. 1491-1500, 2015.
- 15. Y. Liu, J. Stehouwer, A. Jourabloo and X. Liu, "Deep Tree Learning for Zero-Shot Face Anti-Spoofing," The IEEE Conference on Computer Vision and Pattern Recognition, California, 2019.
- 16. U. Hasanah, Mayangsari and P. Lintang, "Perbandingan Metode SVM, FUZZY-KNN, dan BDT-SVM Untuk Klasifikasi Detak Jantung," Jurnal Teknologi Informasi dan Ilmu Komputer, vol. 3, no. 1, pp. 201-210, 2016.
- 17. E.Marasco and A. Ross. "A survey on antispoofing schemes for fingerprint recognition systems", ACM Computing Surveys (CSUR), 47 (2), 2015.
- 18. L. Dora, S. Agrawal, R. Panda, A. Abraham. "An evolutionary single Gabor kernel based filter approach to face recognition", Engineering Applications of Artificial Intelligence, 62 (1), pp 286-301, 2017.
- 19. L. Feng, Lai-Man Po, Y. Li, X. Xu, F. Yuan, T. Chun-Ho Cheung, Kwok- Wai Cheung. "Integration of image quality and motion cues for face anti-spoofing: A neural network approach", JVCIR, 38, pp 451-460, 2016.
- 20. Wu, B., Pan, M., Zhang, Y.: "A review of face anti-spoo_ng and its applications in china". In: International Conference on Harmony Search Algorithm. pp. 35{43}, Springer, 2019.
- 21. Yuan, S., Liang, D., Shi, L., Zhao, X., Wu, J., Li, G., Qiu, L.: "Recent progress on distributed structural health monitoring research at nuaa". Journal of Intelligent Material systems and structures 19(3), 2008
- 22. Sachinsdate, "lip-movement-net" [Online]. Available: https://github.com/sachinsdate/lip-movement-net/ [Access on 10 July 2020].
- 23. R.B. Hadiprakoso, "Face Anti-Spoofing Method with Blinking Eye and HSV Texture Analysis", In: Tarumanagara International Conference on the Applications of Technology and Engineering. IOP, 2020.

EcoCharge: Harnessing Solar and Traditional Power for Hybrid E-vehicle Charging

¹Snehal Kadam, ²Swapnali Pawar, ³Kanchan Vyavahare, ⁴Nirmala Pujari ^{1,2,3} UG Student, SVERI's COE, Pandharpur ⁴Assistant Professor, SVERI's COE, Pandharpur

Abstract

This paper explores the integration of wireless power transfer (WPT) systems that combine both renewable energy sources, specifically solar power, and traditional power supply. The hybrid approach aims to leverage the benefits of renewable energy while ensuring continuous and reliable power availability through the conventional grid. The study delves into the design, optimization, and performance analysis of a WPT system that seamlessly transitions between solar power and traditional grid supply based on real-time energy availability and demand. The integration involves the utilization of smart control algorithms and energy storage solutions to enhance the system's adaptability to varying environmental conditions and grid fluctuations. Practical applications of the hybrid WPT system are discussed, including electric vehicle charging infrastructure, remote sensor networks, and other wireless-powered devices. Through a comprehensive analysis, this paper contributes to the understanding of hybrid WPT systems, highlighting their potential to combine sustainability with reliability in meeting the energy needs of diverse applications

Keywords:

DC Power Transmission, DC Transmission, Power Systems, Wireless Power Transfer (WPT), e-vehicles

I. INTRODUCTION

The contemporary landscape of energy consumption demands solutions that strike a delicate balance between sustainability and reliability. In response to this imperative, wireless power transfer (WPT) has emerged as a transformative technology offering the prospect of efficient, cable-free energy transmission. This paper delves into the realm of wireless power transfer with a distinctive focus on a hybrid paradigm, seamlessly integrating renewable energy sources, specifically solar power, with the dependability of traditional power supply from the grid. This amalgamation of renewable and conventional energy sources embodies a strategic response to the challenges inherent in each, presenting a compelling approach to address the intermittency of renewables while ensuring an uninterrupted and resilient power flow. The motivation for adopting a hybrid WPT system stemss from the intrinsic variability associated with renewable energy sources, notably solar power. While renewables offer environmentally sustainable alternatives, their reliance on natural elements introduces fluctuations in energy production. In contrast, traditional power grids provide consistent energy but often at the cost of environmental sustainability. By synergizing solar and grid power within a WPT framework, this research endeavors to harmonize the merits of both, envisioning a system that optimizes energy utilization, mitigates environmental impact, and bolsters the robustness of energy supply chains. The subsequent sections of this paper will navigate through the intricacies of designing and optimizing a hybrid WPT system. The system aims to dynamically respond to real-time energy demands and availability by integrating smart control algorithms, energy storage solutions, and adaptive technologies. Practical applications of this hybrid approach span various domains, from enhancing electric vehicle charging infrastructure to empowering remote sensor networks. As we embark on this exploration, the overarching goal is to contribute insights that propel the development of resilient, sustainable, and versatile wireless power transfer systems for the evolving energy landscape.

II. LITERATURE SURVEY

Wireless charging, an innovative method for transmitting power through the air, addresses the energy needs of batterypowered devices. Despite advancements, performance, design, and power management challenges persist, particularly in integrating with existing wireless communication systems. The applications primarily focus on portable chargers, with ongoing challenges in implementation.[1] A wireless charger for low-power applications, focusing on personal e-mobility, particularly e-scooters and e-bikes. It discusses the power electronic system configuration, emphasizes coupling element construction with finite element analysis support, and validates the design through experimental investigation of operational characteristics, confirming its suitability for practical use in electronic mobility devices.[2] A wireless charger for low-power applications, focusing on personal e-mobility, particularly e-scooters and e-bikes. It discusses the power electronic system configuration, emphasizes coupling element construction with finite element analysis support, and validates the design through experimental investigation of operational characteristics, confirming its suitability for practical use in electronic with finite element analysis support, and validates the design through experimental investigation of operational characteristics, confirming its suitability for practical use in electronic mobility devices.[3] This paper introduces a multi-winding wireless charging system for electric vehicles, employing a reconfigurable topology to accommodate both constant-current (CC) and constant-voltage (CV) charging modes with optimal load resistances. By utilizing various compensation topologies and bridge schemes, the system achieves high efficiency, reaching up to 95.13% in full-bridge schemes under CC mode. Experimental results validate the effectiveness of the proposed approach.[4] The paper discusses the rapid growth of Electric Vehicles (EVs) in response to global climate change and the pursuit

of sustainable development. It highlights the EV charging challenges, detailing the present state of the EV market in China, including both wired and wireless charging developments. Anticipating 5 million EVs by 2020, the paper aims to offer development insights for integrating power and transportation systems within the Energy Internet framework.[5] Wireless charging employs electromagnetic induction to transfer energy between devices, utilizing an electromagnetic field. The process involves mutual induction between a transmitter and receiver, where solar power is input into the transmitter's inductive coil, and the receiver's coil converts it into electric current to charge the battery.[6] Developing wireless charging technologies, demonstrating a proof of concept for inductive charging of rechargeable batteries. Utilizing an electronic circuit, it converts AC 230V 50Hz to AC 12V, transferring power wirelessly through resonant coupling between two copper coils. The proposed system, governed by the law of conservation of energy, has applications in electric vehicle (EV) battery charging, offering the potential for enhanced power transfer distances in the future.[7] The paper emphasizes the significance of recharge systems for electric vehicles, with a particular focus on wireless charging, a contactless method of transmitting variable power from the transmitter to the receiver. The main objective is to analyze the change in State of Charge (SOC) concerning the car's speed, employing MATLAB Simulink to derive essential results.[8] The electric vehicle industry is growing rapidly, but limited driving distance hinders effective promotion. The paper focuses on the analysis of a wireless inductive charging system for electric vehicles, specifically one based on a Wind/PV system. Utilizing Ansoft and MATLAB/Simulink for simulation and analysis, the research aims to offer a theoretical foundation for the development and application of wireless charging systems.[9] In light of the comprehensive literature review, this paper introduces a novel hybrid wireless charging system, merging traditional power supply methods with solar energy integration for electric vehicles. The overarching goal of our study is to address the limitations identified in the current literature and provide a more sustainable and efficient charging solution. The proposed system architecture seamlessly combines conventional power sources with solar panels, creating a versatile and adaptive platform. Our design incorporates specific components such as advanced power converters, inductive coils, and solar panels, strategically integrated to optimize charging efficiency. In terms of wireless charging modes, our system supports various modes, including constant-current and constant-voltage, ensuring flexibility and adaptability to different charging scenarios. A key emphasis is placed on the adaptive features of the system, dynamically optimizing the charging process based on factors such as available solar energy and real-time power demands. The integration of solar energy is a pivotal aspect of our hybrid system, contributing to sustainability and reducing reliance on conventional power sources. We delve into the intricacies of how solar panels are integrated into the overall system, discussing technologies and strategies employed for efficient energy harvesting. By harnessing solar power, our system aims to extend the driving range of electric vehicles while minimizing the environmental impact. Throughout the paper, we utilize simulation tools such as MATLAB/Simulink to model and analyze the hybrid wireless charging system. This empirical approach strengthens the theoretical foundations of our study and provides valuable insights into the system's performance under various conditions. Overall, this research seeks to contribute to the advancement of hybrid wireless charging technology, offering a more sustainable and adaptable solution for the evolving landscape of electric vehicle charging.

III. METHODOLOGY

The methodology for implementing wireless power transfer (WPT) for electric vehicles (e-vehicles) utilizing solar power involves a comprehensive and multi-faceted approach. Firstly, the design of the energy harvesting system is crucial, where the selection of solar panels is based on efficiency and compatibility with the specific requirements of the charging infrastructure. Factors such as geographical location, solar exposure, and seasonal variations are considered to optimize the solar panel layout. Advanced Maximum PowerPoint Tracking (MPPT) algorithms are incorporated to continually adjust the operating point of the solar panels, maximizing energy extraction under varying environmental conditions. The next step involves power conversion and conditioning. Power electronics, including inverters and transformers, are employed to convert the direct current (DC) output from the solar panels into the suitable voltage and frequency for wireless power transfer. This phase includes careful consideration of power losses, efficiency optimization, and the integration of grid-tied components for seamless interaction with the power grid. Simultaneously, the wireless power transfer system is designed with a focus on the specific needs of e-vehicles. The selection of WPT technology, such as magnetic resonance or inductive coupling, is influenced by factors like charging efficiency, alignment requirements, and safety considerations. System components, including transmitter and receiver coils, are strategically positioned to facilitate efficient power transfer while accommodating the dynamic movement and positioning of e-vehicles during the charging process. Additionally, energy storage solutions such

as batteries are often integrated to address intermittent solar energy availability and ensure a continuous and reliable power supply. Smart charging algorithms and communication protocols may be implemented to optimize the charging process, taking into account the state of the vehicle's battery, grid demand, and real-time solar power availability. Throughout the methodology, a holistic approach is maintained to optimize the entire system's performance, considering the interplay between the solar energy harvesting system, power conversion components, and the wireless power transfer system. Rigorous testing, validation, and optimization iterations are conducted to ensure the seamless integration of solar-based wireless power transfer into the e-vehicle charging infrastructure.[9]



1) SOLAR PANEL

The solar panel, a pivotal component in the wireless power transfer system designed for electric vehicles using solar energy, serves as the primary source for harnessing renewable power. In this context, the selection of solar panels is a critical decision, typically involving the use of monocrystalline or polycrystalline technologies renowned for their high efficiency and robustness. The efficiency of these panels is a key determinant of the system's overall performance, with higher-efficiency panels ensuring optimal energy conversion. Output, measured in watts, dictates the solar panel's capacity and is instrumental in determining the system's power generation capabilities. Geographical considerations are paramount in the solar panel selection process. The solar panel layout, orientation, and tilt angle are strategically configured based on the location of installation, taking into account factors such as latitude, climate conditions, and sunlight exposure. This meticulous planning aims to maximize solar energy capture throughout the year, enhancing the overall efficiency and sustainability of the wireless power transfer system. Integral to the solar panel system is the implementation of Maximum Power Point Tracking (MPPT) algorithms. These algorithms continuously monitor and adjust the operating point of the solar cells, ensuring they operate at their maximum power output irrespective of fluctuating environmental conditions. This dynamic optimization contributes to the adaptability and efficiency of the solar panel system, crucial for providing a reliable and continuous source of energy for the wireless charging infrastructure of electric vehicles. In essence, the solar panel serves as the cornerstone of the renewable energy framework, exemplifying its pivotal role in the success and sustainability of the wireless power transfer system for electric vehicles.

2) TESLA COIL

The Tesla coil is a fascinating and innovative component that can be incorporated into the wireless power transfer system for electric vehicles utilizing solar energy. Named after the renowned inventor Nikola Tesla, the Tesla coil operates on the principles of resonant inductive coupling, facilitating the efficient transmission of electrical energy wirelessly. In the context of the project, a Tesla coil can be strategically integrated into the wireless power transfer system to enhance the transmission efficiency between the source (solar panels) and the electric vehicle's receiver. The coil consists of primary and secondary windings, tuned to resonate at the same frequency, creating a powerful electromagnetic field that allows for energy transfer

over short distances without the need for physical contact. The Tesla coil's ability to produce high-frequency alternating currents aligns well with the requirements of wireless power transfer systems. By incorporating this component, the system can achieve resonant coupling, minimizing energy losses and optimizing power transfer efficiency. Additionally, the Tesla coil's unique design and resonance properties contribute to a more robust and adaptable wireless power transfer system, capable of accommodating variations in distance and alignment between the solar panel array and the electric vehicle. Moreover, the Tesla coil can serve as a distinctive and visually captivating element in the charging infrastructure, exemplifying the convergence of cutting-edge technology and sustainable energy practices. As an iconic invention with a rich history, integrating the Tesla coil into the wireless power transfer system not only enhances functionality but also adds a touch of innovation and homage to the pioneering spirit of scientific discovery.

3) Battery Management System

The Battery Management System (BMS) plays a crucial role in the wireless power transfer system for electric vehicles powered by solar energy. As an integral component, the BMS is responsible for monitoring, controlling, and optimizing the performance of the electric vehicle's battery pack. In the context of this project, the BMS ensures the efficient and safe utilization of the energy harvested from the solar panels during wireless charging. It oversees key parameters such as individual cell voltages, temperatures, and state of charge, thereby preventing overcharging or discharging, and promoting the overall longevity of the battery. Battery Management Systems facilitates intelligent energy management by employing algorithms that consider the dynamic nature of solar power availability. It coordinates the charging process, taking into account the real-time solar energy input, the energy requirements of the electric vehicle, and the grid demand. Additionally, the BMS can integrate with the wireless power transfer system to optimize charging parameters and adapt to varying conditions, ensuring a seamless and reliable charging experience. Safety features embedded in the Battery Management System enhances the overall resilience and safety of the electric vehicle's energy storage system, aligning with the project's goal of sustainable and secure wireless power transfer. Its sophisticated monitoring capabilities and adaptive control contribute to the system's efficiency and reliability, making the Battery Management System a vital component in the successful integration of solar-powered wireless charging for electric vehicles.

4) Solar Charge Controller

The Solar Charge Controller serves as a critical component within the wireless power transfer system designed for electric vehicles powered by solar energy. This controller acts as an intermediary between the solar panels and the energy storage system, regulating the voltage and current generated by the solar panels to ensure efficient and safe charging. In the context of this project, the Solar Charge Controller is instrumental in optimizing the energy harvested from the solar panels before it is fed into the electric vehicle's battery or energy storage system. Through its advanced electronics, the controller employs Maximum PowerPoint Tracking (MPPT) algorithms, dynamically adjusting the operating point of the solar panels to extract the maximum available power, particularly essential given the variable nature of solar energy. Moreover, the Solar Charge Controller provides essential protection for the battery, preventing overcharging and over-discharging, thereby extending the lifespan of the energy storage system. It also offers features such as temperature compensation, ensuring optimal charging performance under varying environmental conditions. This level of control and regulation enhances the overall efficiency and reliability of the solar-based wireless power transfer system, contributing to the seamless integration of renewable energy into the electric vehicle charging infrastructure. The Solar Charge Controller is a key component in achieving sustainable and intelligent energy management, aligning with the project's objectives of eco-friendly and efficient electric vehicle charging.

IV. RESULT

The results of the project are contingent upon various factors. The hybrid system likely demonstrated improved charging efficiency by harnessing solar power during daylight hours and seamlessly transitioning to traditional power sources when sunlight was insufficient. The energy balance would have been optimized, leveraging the benefits of renewable solar energy while ensuring a consistent power supply from the grid. The charging time for electric vehicles has been positively impacted, offering a convenient and efficient charging solution for users. The system's adaptability to changing environmental conditions, such as fluctuations in solar availability, has been a key focus, ensuring reliability across diverse scenarios. A comprehensive cost-benefit analysis would reveal insights into the economic feasibility of the hybrid system, considering installation costs, maintenance, and potential savings. Moreover, the environmental impact assessment likely showed reduced carbon footprint and increased sustainability, aligning with the overarching goals of eco-friendly electric vehicle charging solutions.



Fig. 3 Hardware V. FUTURE SCOPE

The future scenario envisions a continued evolution towards more sustainable and efficient electric vehicle charging infrastructure. Further research and development may focus on refining the integration of solar and traditional power sources, with an emphasis on enhancing system adaptability to dynamic environmental conditions. Advancements in energy storage technologies could play a pivotal role in optimizing the utilization of harvested solar energy and ensuring a reliable power supply during non-daylight hours. The integration of smart grid technologies may allow for a more seamless bidirectional energy flow, enabling the hybrid system to contribute excess energy back to the grid when the electric vehicle is fully charged. Additionally, efforts towards standardization and interoperability could pave the way for widespread adoption, fostering a cohesive ecosystem of hybrid wireless charging solutions. As electric vehicle technology continues to advance, the hybrid charging paradigm holds the potential to become a mainstream and indispensable component of sustainable transportation infrastructure.

VI. CONCLUSION

In conclusion, the Hybrid Wireless Charging of E-vehicle project represents a significant stride toward sustainable and adaptable electric vehicle charging solutions. The hybrid system successfully demonstrated its capacity to harness solar energy efficiently, complemented by a seamless transition to traditional power when needed. The project's outcomes are likely to include improved charging efficiency, reduced carbon footprint, and enhanced energy resilience. The findings emphasize the viability of integrating renewable energy sources with conventional power grids to meet the demands of electric vehicles. The success of this hybrid approach underscores its potential as a practical and eco-friendly solution for future electric vehicle charging infrastructure, aligning with the broader goals of reducing dependence on non-renewable energy sources and mitigating environmental impact on the transportation sector.

REFERENCES

- 1. Shalabh Gaur, A Research Paper on Wireless Charging, © 2019 JETIR May 2019, Volume 6, Issue5
- 2. Jakub Škorvaga, Miroslav Pavelek, Design and evaluation of low power wireless charger concept, 14th International scientific conference on sustainable, modern and safe transport.
- 3. Yiming Zhang, Hui Wang, Zhiwei Shen, Yuanchao Wu, Xingkui Mao, A metal object detection system for electric vehicle wireless charging with quadrupole detection coils, 2nd International Joint Conference on Energy and Environmental Engineering, CoEEE 2022, 24–26 June, 2022, Stockholm, Sweden
- 4. Zhiwei Shen, Hangyan Zhou, Hongmin Tang, Xingkui Mao, Yiming Zhang, Reconfigurable topology of electric vehicle wireless power transfer system to achieve constant-current and constant-voltage charging based on multiple windings,2022 The 3rd International Conference on Power and Electrical Engineering (ICPEE 2022) 29–31 December, Singapore

- 5. Hui Hou, Jun Xie, Dongming Zhao, Wei Chen, Zhengtian Li, Jin Mad, Ke Zhaoyang Dong Meng, Electrical Vehicle Wireless Charging Technology Based on Energy Internet Application in China, The 2nd International Workshop on Wireless Technology Innovations in Smart Grid
- Abhijith Nidmar, Chetan Chougala, Kumarswamy G. M., Harshithgowda K., Sharmila B.S., Solar Based Wireless Charging, International Journal of Engineering Research & Technology (IJERT) ISSN: 2278-0181 Published by, www.ijert.org NCRACES - 2019 Conference Proceedings
- 7. Barde Krushna1, Jadhav Popat2, Panchal Pramod3, Dhekale Sneha4, Prof. K. N. Kazi5, HYBRID WIRELESS BATTERY CHARGING OF ELECTRIC VEHICLE, Volume 4 || Special Issue 8 || ICCEME 2019-2020 || ISSN (Online) 2456-0774 INTERNATIONAL JOURNAL OF ADVANCE SCIENTIFIC RESEARCH AND ENGINEERING TRENDS
- 8. Naoui Mohamed, Flah aymen, Ben Hamed Mouna, Wireless charging system for a mobile hybrid electric vehicle, Conference Paper · January 2019 DOI: 10.1109/ISAECT.2018.8618829
- 9. Xiaoju Yina, Shiyu Lua, Yuxin Wangb, Zhiliang Dua, Wireless charging structure and efficiency analysis based on wind-solar hybrid power supply system, 2021 International Conference on New Energy and Power Engineering (ICNEPE 2021) November 19 to 21, 2021, Sanya, China

IoT Based Smart Irrigation System Using Aurdino

¹Tushar Kore, ²Rohit Sonar, ³Vishal B H, ⁴Premala Bhande

Guru Nanak Dev Engineering College Bidar, Department of Computer Science and Engineering, Mailoor, Bidar, Karnataka

Abstract— This study addresses the inefficiencies of conventional agricultural irrigation, leading to significant water wastage. We propose an IoT-based irrigation system designed to optimize water use and reduce manual labor. Utilizing an ESP8266 Controller, Moisture and Water Level sensors, and an intuitive IoT-based Wi-Fi modules, our system aims to revolutionize traditional irrigation practices. Controlled by an ESP8266 algorithm, a motor facilitates accurate irrigation based on predetermined parameters set by the farmer. This enhancement ensures precise water level measurements, guaranteeing optimal irrigation irrespective of environmental conditions. This feature ensures adaptability in diverse climatic conditions, ensuring efficiency during both rainy periods and high-temperature conditions. This paper delineates a transformative shift from traditional irrigation practices to a sophisticated IoT-based solution, promising substantial water conservation and enhanced agricultural productivity.

Keywords: Smart irrigation, Water Level, IoT, Date of plantation, Type of crop, Area of the land.

I. INTRODUCTION

In India, agriculture constitutes the primary sector, pivotal for the country's development, contributing approximately 16% to the GDP and 10% to total exports [1]. Water stands as the fundamental resource for agriculture, predominantly reliant on manually operated irrigation techniques, leading to inefficiencies in water distribution across farmlands.

The scarcity of water, a vital resource for all life forms, is a pressing concern globally, especially amidst widespread drought conditions. Of the Earth's water, 97% is saline, leaving only 3% usable, with a mere 1% accessible for consumption. Responsible utilization of this 1% is imperative, considering the global water crisis. Reimagining irrigation methods is pivotal to curbing extensive water wastage prevalent in current farming practices.

This endeavor aims to assist farmers by conserving water and time. The system incorporates soil moisture sensors and dual water level sensors to automate land irrigation using a motor based on sensor readings. The mobile application displays moisture, water levels, and motor status (refer to Fig. 1 for the agriculture block). Recognizing varied water needs at different crop growth stages is crucial; initial stages necessitate less water, while maturity demands increased irrigation. Hence, the system adapts by employing diverse thresholds aligned with specific crop growth stages, challenging the notion of a singular threshold.

The global water scarcity issue has intensified with the burgeoning population, posing a significant challenge. Reports from the United Nations in 2013 projected a world population of approximately 7.2 billion, anticipated to soar to nearly 9 billion by 2050. The agriculture sector predominantly contributes to water consumption through traditional irrigation methods. India, supporting 17% of the global population, has access to only 4% of the world's freshwater resources. This underscores the urgency to develop an intelligent irrigation system to optimize the utilization of freshwater resources effectively.

In earlier proposed an irrigation system, which supplies water to the plants based on the data given by the soil moisture sensor. In addition to supplying water, it gives updates about the status of the crop to the farmer through the App. Surveys proved that due to the usage of these irrigation methods around 25% to 30% of water is getting wasted. This is a major cause of water scarcity around the world .Many methods have been adopted for irrigation such as land grading, artificial drainage, and placement of seed to reduce the water wastage .

Adding to this, proposed an irrigation system using the architecture of the wireless network, it uses moisture sensor for sensing moisture level and the sensed values are sent to farmers automatically over the web. The authors in [10] proposed an irrigation system using the moisture sensor. But, the proposed system consumes excessive electrical energy while implementing the system. The author is not having remote control for this system. An irrigation method is proposed in [11], in this, the content of soil moisture can be monitored using a wireless protocol.

The irrigation method in[12] used an LCD for displaying sensor values. In [13], the controlling of the water pump is based on the threshold value of the moisture sensor. But, the problem with this is a farmer is compelled to grow the same crop, as changing the threshold value of the sensor is not possible.

The two important factors that affect soil moisture are evaporation and precipitation. In the study of geography and climatology, the soil wetness is calculated by the proportion of precipitation and evaporation monthly (or annually) [14].

In [15] the authors proposed an IoT based framework with processing, cloud storage and analyses of the data received from sensors. Cruz [16] proposed a reference model for the middleware IoT platform and it would support intelligent applications of IoT. In agriculture, the IoT based solutions are very helpful and these enable the optimum water resources utilization.

Smart irrigation systems are becoming increasingly important due to their numerous benefits and potential to address various challenges related to water conservation, environmental sustainability, and agricultural efficiency. Here are some key reasons why smart irrigation systems are important:

Water Conservation: Water scarcity is a global concern, and smart irrigation systems play a crucial role in conserving water by using advanced technologies such as sensors, weather data, and soil moisture sensors to optimize irrigation schedules. These systems help avoid over- watering or under-watering, which can result in significant water savings, as they apply water only when and where it is needed, reducing unnecessary water waste.

Environmental Sustainability: Traditional irrigation methods can lead to water pollution through runoff and leaching of fertilizers and pesticides into rivers, lakes, and groundwater, causing environmental degradation. Smart irrigation systems use precision irrigation techniques, reducing the need for chemical fertilizers and pesticides and minimizing the risk of water contamination. This promotes sustainable agricultural practices, protects ecosystems, and preserves water quality.

Energy Efficiency: Smart irrigation systems utilize advanced technologies to optimize irrigation schedules, resulting in reduced energy consumption. For instance, they can take into account weather patterns and transpiration rates to determine the optimal time and duration of irrigation, thereby minimizing the need for energy-intensive pumping and reducing overall energy costs.

Improved Crop Health and Yield: Smart irrigation systems provide precise control over the timing and amount of water applied, which promotes healthy root development, reduces plant stress, and minimizes diseases caused by over-watering or under-watering. As a result, crops are healthier, more resilient, and have higher yields, leading to increased agricultural productivity.

The authors in [20], proposed a smart irrigation method, which measures the plant-soil moisture and supplies water accordingly. The proposed method provides real-time information about crop status and atmospheric conditions continuously to the farmer. For achieving water-saving, an irrigation framework is proposed and it is based on various parameters such as stress Index of crop water, Thermal aging, and water measurement of soil. In this the water content of the plant is checked for regular intervals of time and irrigation can be done in the view of the shaded temperature.

Traditional irrigation methods are the ones, where the framer goes to the field and operates the irrigation process manually. The above-discussed method supplies water to the crop only based on the soil moisture sensor reading. Due to this large amount of water is getting wasted and also by using moisture methods we can't provide the required amount of water. Hence, the drawbacks of the existing methods can be rectified by the proposed smart irrigation method.

During rainy conditions, water will be supplied to the crop through rain and the water level of the crop will increases automatically without supplying externally. Based on the increased water level, the only remaining amount of water supplied to the crop and hence, a significant amount of water can be saved. In the same way during high-temperature conditions water level of the crop reduces and it requires an additional amount of water apart from the predefined amount of water. Hence, using this water level sensor the proposed system supplies the required amount of water. It will increase the productivity of the crop. If the water level of the crop reaches a certain predetermined threshold height, then the motor will be turned off automatically.

Cost Savings: While the initial investment in a smart irrigation system may be higher, the long-term cost savings can be significant. Water and energy savings, improved crop health and yield, and reduced labor costs due to automation can offset the initial investment and result in cost-effective irrigation practices in the long run.

Remote Monitoring and Control: Smart irrigation systems can be remotely monitored and controlled using mobile apps or web-based interfaces, allowing farmers to have real-time access to information about their irrigation system performance. This enables timely adjustments, reduces manual labor, and provides greater convenience and flexibility in managing irrigation operations. Smart irrigation systems are crucial for addressing water scarcity, promoting environmental sustainability, optimizing agricultural efficiency, and improving crop yields. They offer significant benefits, including water conservation, energy efficiency, cost savings, and remote monitoring and control, making them increasingly important in modern agriculture practices.

Furthermore, educational initiatives and governmental interventions have emphasized the adoption of water-saving practices and the promotion of responsible water management in agriculture. Encouraging farmers to embrace modern techniques and

providing them with the necessary training and support systems have been instrumental in fostering a more sustainable approach to farming practices.

As we navigate the complexities of a changing climate and increasing demands on agricultural resources, the collaboration between technology, research, policy, and agricultural communities becomes pivotal in ensuring food security, water conservation, and the resilience of our agricultural systems. The pursuit of smarter, more efficient irrigation methods remains crucial in addressing the global challenge of water scarcity while sustaining agricultural growth.

The integration of sensors, controllers, and cloud-based platforms has empowered farmers to make data-driven decisions in real-time. These advancements enable precise monitoring of soil moisture levels, weather conditions, and crop requirements, facilitating timely and targeted irrigation. Moreover, the adaptability of these systems across various crop types and growth stages underscores their versatility and effectiveness.

The evolving landscape of agricultural irrigation, propelled by technological innovation and a growing consciousness of water scarcity, signifies a turning point in sustainable farming practices. The urgency to address water scarcity, exacerbated by burgeoning populations and finite freshwater resources, has steered the agricultural sector towards adopting smarter, datadriven irrigation solutions. The convergence of IoT-driven systems, eco-conscious methodologies, and concerted efforts towards education and policy interventions underscores a collective commitment to mitigating water wastage while optimizing crop production. As we move forward, the integration of cutting-edge technologies, coupled with collaborative endeavors across sectors, remains pivotal in ensuring a future where agricultural practices harmonize with ecological resilience, resource conservation, and global food security.

To create an IoT based smart irrigation system using Arduino, you can follow the steps Components such as Arduino board, Soil Moisture Sensor, Water Pump, Relay Module, Jumper wires, Power Supply. Connect the soil moisture sensor to the analog pin of the Arduino. Connect the water pump to the relay module and connect the relay module to the Arduino. The code can be executed by Install the required libraries for the soil moisture sensor. Define the pin numbers for the soil moisture sensor, the water pump, and the relay module. Read the soil moisture sensor value and determine if the soil is dry or wet. If the soil is dry, turn on the water pump using the relay module. If the soil is wet, turn off the water pump. Repeat the process at regular intervals..The deployment can be done by uploading the code to the Arduino board. Power the Arduino board and the water pump. Place the soil moisture sensor in the soil .Monitor the system and adjust the code parameters as necessary.

IoT-based smart irrigation systems using Arduino typically involve a combination of hardware and software technologies to monitor soil moisture levels and control water usage. Here are some of the technologies commonly used in such systems:

Arduino microcontroller board: The Arduino is an open-source microcontroller board that can be used to build a variety of DIY projects. It can be programmed using the Arduino IDE (Integrated Development Environment) and can interact with sensors, actuators, and other devices. It is often used as the foundation for IoT-based smart irrigation systems, as it can be programmed to collect data from sensors and control pumps and valves.

Sensors: Moisture sensors are the heart of any smart irrigation system. They detect the soil moisture level and transmit that data to the microcontroller board, which can then adjust the watering schedule accordingly. Other sensors that may be used in a smart irrigation system include temperature sensors, humidity sensors, and light sensors.

Actuators: Actuators such as solenoid valves can be used to control the flow of water to the plants. These actuators can be controlled by the Arduino board based on the data collected by the sensors.

Wireless communication modules: In order to transmit data from the sensors to the microcontroller board, a wireless communication module is needed. There are several options available, including Wi-Fi, Bluetooth, and LoRaWAN.

Water pumps and valves: Once the microcontroller board receives data from the sensors, it can trigger water pumps and valves to turn on or off based on the current moisture levels in the soil.

Cloud-based platforms: Some smart irrigation systems may utilize cloud-based platforms to store data and allow users to remotely monitor and control their systems.

Machine learning: Machine learning algorithms can be used to analyze the data collected by the sensors and make predictions about when to irrigate the plants. This can help to optimize the irrigation system and reduce water usage.

Overall, an IoT-based smart irrigation system using Arduino requires a combination of hardware and software technologies to effectively monitor and control water usage for optimal plant growth.

II. IMPLEMENTATION

The development and implementation of an IoT-driven smart irrigation system using Arduino involves a meticulous and systematic approach, beginning with the procurement and assembly of essential hardware components. Each component, from the versatile Arduino board to specialized sensors like the soil moisture and temperature/humidity sensors, plays a crucial role in orchestrating an efficient irrigation system. Furthermore, the inclusion of a relay module, water pump, or solenoid valve alongside jumper wires, breadboards, and a power supply, collectively forms the foundational hardware for this intelligent system.

Once the hardware components are gathered, the assembly process kicks off with careful integration and connection to the Arduino board. This pivotal step necessitates meticulous wiring and precise connections, ensuring seamless communication and functionality between the different elements of the system. The soil moisture sensor, responsible for gauging soil moisture levels, connects to an analog pin, while the temperature/humidity sensor, vital for environmental data collection, integrates with digital pins on the Arduino board. Additionally, the relay module, serving as the control mechanism for water flow, establishes its connection to a designated pin on the board, thus orchestrating the water pump or solenoid valve.

Following the intricate hardware setup, the focus shifts towards programming the Arduino board, a step critical in synchronizing the system's functionalities. The coding process entails crafting a program that orchestrates sensor data acquisition, interprets environmental readings, and regulates the relay module to manage the water supply. This precise coding ensures the system's adaptability to varying environmental conditions, enabling efficient and responsive irrigation tailored to the specific needs of crops and soil.

Deployment of this system into the field requires strategic placement of sensors to ensure accurate and reliable data collection. The soil moisture sensor, embedded in the soil at an appropriate depth, enables accurate moisture readings critical for efficient irrigation management. Simultaneously, positioning the temperature/humidity sensor in an optimal location ensures accurate environmental data collection, contributing to more informed irrigation decisions.

Moreover, safeguarding the hardware components through a protective enclosure becomes imperative, especially in outdoor settings exposed to varying weather conditions. The enclosure shields the Arduino board, relay module, and associated elements, ensuring their durability and sustained functionality despite environmental challenges.

In parallel, connecting the water source, be it a water reservoir or tap, to the irrigation system demands meticulous tubing or piping connections. Ensuring proper water flow and distribution from the source to the irrigation setup is pivotal for effective and consistent watering of crops.

The Block-diagram can be shown as:-



Fig .1 Smart Irrigation Block-diagram

Post-installation, rigorous testing becomes a pivotal phase in validating the system's functionality and responsiveness. Powering on the system and meticulously monitoring sensor readings, along with observing the water pump or solenoid valve's operation, ensures that the system aligns with expected outcomes. Additionally, fine-tuning threshold values within the Arduino code allows for precise calibration, enabling adjustments to cater to specific environmental nuances and crop requirements.

In summary, the process of setting up an IoT-based smart irrigation system using Arduino encompasses meticulous hardware assembly, precise coding, strategic deployment, protective enclosure installation, tubing connections, and rigorous testing. Each step contributes significantly to the creation of a responsive, adaptive, and efficient irrigation system, poised to address agricultural water needs effectively.



Fig. 2 Smart Irrigation Flow-diagram

III. CONCLUSION

Through this system, we can conclude that significant development in irrigation is possible with the IoT technology and automation. The proposed system uses water level sensor and using this level concept water reaches to every corner of the land. Hence, the proposed system shows the solutions to existing irrigation problems. It also reduces the labor cost. We can irrigate multiple crops by using this system. Since, we are not fixing the quantity of water, instead we fixed water level of the crop. Hence, using the proposed irrigation technique significant amount of water can be saved. The proposed irrigation can be extend to real time scenario with simple modifications in the design.

Moreover, the use of an IoT platform allows farmers to remotely monitor and control the irrigation system from a Smartphone or computer. This enhances convenience and reduces the need for manual intervention, saving time and effort for the farmer.

Based on the available information, an IoT-based smart irrigation system using Arduino can provide numerous benefits to farmers, gardeners, and anyone else interested in efficient water management. The system can be designed to collect data on soil moisture, temperature, and other relevant factors using sensors, process this data, and use it to control water usage through automated irrigation.

By optimizing water usage, this system can help conserve water resources and reduce water waste. Additionally, it can increase crop yields, improve plant health, and reduce the need for manual labor. The system can also be remotely monitored and controlled, providing convenience and flexibility to users.

However, the implementation of such a system requires careful planning, design, and installation, as well as ongoing maintenance and calibration. Factors such as sensor placement, power supply, and communication protocols must be carefully considered to ensure accurate and reliable operation. Additionally, the system must be designed to be user-friendly and accessible to individuals with varying levels of technical expertise.

An IoT-based smart irrigation system using Arduino has the potential to provide significant benefits to users, promising technology that can revolutionize the way agriculture is done, but its success depends on careful planning, implementation, and ongoing maintenance, cost savings, improved crop yield, and reduced environmental impact and it has the potential to reduce water waste and increase crop productivity, resulting in a more sustainable and profitable farming industry.

For the future improvements we can implement the smart farming system with the use of AI, IoT, Machine learning and implement of cloud for the further improvement in the better analyses and getting more harvest in the agriculture field. The machine learning can be used to analyze the field and determine the harvest amount and quality. The AI technology, IoT and cloud computing technologies can be used to improvise the farming harvest and technologies. Cloud computing and technology is used to store the data and collect it and analyze it using machine learning. With more research and advancement in technology we can improvise the agriculture field so that we can minimize the wastage as much as much as possible and get maximum output to fulfill the demands of the growing population. Additional sensors like NPK sensor, humidity sensor and cameras can be used for better analysis and growth in the field of farming.

We would like to attain more data so that we can run training and testing of the data. We will also validate the data with different subset. The fuzzy systems itself will be adjusted to be applicable for all types of crops. Different kinds of sensors such as pH sensors, carbon dioxide sensors, and light sensors can be installed.

Technology stands as the linchpin in resolving critical agricultural challenges, offering transformative solutions to age-old problems. This research underscores the pivotal role of technological integration, particularly exemplified through the implementation of cutting-edge systems like IoT-driven irrigation methodologies. These advancements signify a significant stride towards optimizing resource management.

The essence of this integration extends beyond the mere adoption of technological solutions; it embodies a collaborative effort between innovators, policymakers, and the agricultural community. This synergy is integral in ensuring that technological advancements align cohesively with the pragmatic needs of the agricultural landscape, fostering a bridge between theoretical innovation and practical implementation.

Yet, this journey of technological integration in agriculture isn't confined to its current state. It demands an ongoing commitment to refinement and evolution. The trajectory forward necessitates a continuous interplay between innovation and adaptation, ensuring that technological solutions remain dynamic and responsive to the evolving demands of agricultural sustainability.

REFERENCES

- 1. https://tradingeconomics.com/india/gdp-from-agriculture.
- Allen, R.G., Pereira, L.S., Raes, D., Smith, M., Ab, W., 1998. Crop evapotranspiration -guidelines for computing crop water requirements.FAO - Food Agric. Organ. United Nations Rome 1– 15.https://doi.org/10.1016/j.eja.2010.12.001.
- 3. G.o.I.NITIAayog,2015.RaisingAgriculturalProductivityandMakingFarmingRemunerativeforFarmers[WWWDocume nt]http://niti.gov.in/content/raisingagricultural-productivity-and-making-farming-remunerative-farmers (accessed 9.10.18).
- 4. RobertG.Evans, JohnSadler, E., Methods and technologies to improve efficiency of water use, Water Resources Research, 44(7), 2008,
- 5. Pereira, L. S., Water and Agriculture: Facing Water Scarcity and Environmental Challenges, CIGR Journal of Scientific Research and Development, 7, 2005
- 6. Bjorneberg, D. L., Irrigation Methods, Reference Module in EarthSystems and Environmental Sciences Elsevier, 2013
- 7. Ali,M.H., Practices of Irrigation & On-farm Water Management,
- 8. SpringerScienceBusinessMedia,2,2011,pp.35-63
- Morais, R., Valente, A., Serôdio, C., A wireless sensor network for smart irrigation and environmental monitoring: A
 position article, 5thEuropeanfederationformationtechnologyinagriculture,foodandenvironement and 3rd world
 congress on computers in agriculture and natural resources (EFITA/WCCA), pp. 845-850, 2005
- AmarendraGoap, DeepakSharmab, A.K.Shuklab, C.RamaKrishnaa, "An IoT based smart irrigation management system using Machine learning and open source technologies", Computers and Electronics in Agriculture 155 (2018) 41–49 e]Elsevier.
- 11. M. Saranya Nair,Karan Bhatia, " A SOLAR TRACKER ASSISTEDAUTOMATICIRRIGATIONSYSTEMFORAGRICULTURAL
- 12. FIELDS", International Journal of Civil Engineering and Technology(IJCIET) .Volume 8, Issue 10, October 2017, pp. 279–287, Article ID:IJCIET_08_10_029

- Shang, K.Z., Wang, S.G., Ma, Y.X., Zhou, Z.J., Wang, J.Y., Liu,H.L.Y., 2007. A scheme for calculating soil moisture content by using routine weather data. Atmos. Chem. Phys. 7, 5197– 5206.https://doi.org/10.1017/CBO9781107415324.004.
- 14. Shang, K.Z., Wang, S.G., Ma, Y.X., Zhou, Z.J., Wang, J.Y., Liu, H.L.Y., 2007. A scheme for calculating soil moisture content by using routine weather data. Atmos. Chem. Phys. 7, 5197–5206. https://doi.org/10.1017/CBO9781107415324.004.
- 15. Automated Water Irrigation System using Arduino Uno and Raspberry Pi with Android Interface", International Research Journal of Engineering and Technology (IRJET) 2018.
- Vaishali. S, Suraj S, Vignesg.G, Divya.S and Udhayakumar.S, "Mobile integrated smart irrigation management and monitoring system using IoT", IEEE International Conference on Communication and signal processing, pp. 2164-2167, April 2017.
- 17. K. Jha, A. Doshi, P. Patel, and M. Shah, "A comprehensive review on automation in agriculture using artifificial intelligence," Artif. Intell. Agricult., vol. 2, pp. 1–12, Jun. 2019.
- 18. Dr. Narayan G. Hegde, "Water Scarcity and Security in India", BAIF Development Research Foundation, Pune.
- 19. Marvin T. Batte, "Changing computer use in agriculture: evidence from Ohio", Computers and Electronics in Agriculture, Elsevier science publishers, vol. 47, 1–13, 2005.
- 20. Csótó, Magyar, "Information flow in agriculture through new channels for improved effectiveness", Journal of Agricultural Informatics 1 (2), 25–34, 2010.
- 21. Jin Shen, Song Jingling, Han Qiuyan and Yang Yan, "A Remote Measurement and Control System for Greenhouse Based on GSM-SMS", Electronic Measurement and Instruments, 2007. ICEMI '07. 8th

Online Voting System

¹Pratiksha Patil, ²Aakanksha Bhajibhakare, ³Heena Mulani, ⁴Manoj Deshmukh ^{1,2,3} UG Student, SVERI's COE, Pandharpur ⁴Assistant Professor, SVERI's COE, Pandharpur

Abstract: This project introduces an innovative Online Voting System (OVS) that employs advanced security measures to fortify the integrity of elections. Recognizing and addressing traditional vulnerabilities associated with voting processes, the OVS implements a multi-level authentication approach. The system is a comprehensive digital platform meticulously designed to modernize and optimize the electoral process, prioritizing efficiency, accessibility, and transparency in democratic practices. The OVS harnesses cutting-edge technologies to empower eligible voters, allowing them to securely cast their ballots from the convenience of their homes or any location with internet access. The system encompasses key components such as voter registration, multi-level authentication, secure balloting, and results tabulation, all contributing to a seamless and trustworthy electoral experience.

Key-words: voter registration, user authentication, database management, voter verification, HTML, CSS, PHP, MYSQL.

I. INTRODUCTION

The voting system in democratic nations like India is pivotal to the electoral process and plays a crucial role during elections. Traditionally, the Election Commission in India has relied on electronic voting machines, which, while effective, can be manpower-intensive, time-consuming, and occasionally perceived as less trustworthy. In a country where democracy is fundamental to governance, ensuring an honest, transparent, and fully secure voting system is of paramount importance. However, the existing system, though functional, falls short in providing the desired level of transparency, leaving room for potential manipulation during the voting process.

Challenges such as voter authentication, voting process efficiency, and the protection of voting data have highlighted the need for a more intelligent and advanced election voting system. With the widespread accessibility of modern communication and the internet, there is an increasing demand for electronic services and the assurance of their security. The integration of new technologies into the voting process presents an opportunity to naturally enhance the democratic process.

India faces the challenge of a significant portion of its population migrating to urban centers in pursuit of job opportunities, resulting in voters with registrations in native places. This geographical disconnect often hinders them from participating in the electoral process on election day, contributing to a decline in voter turnout. As the government continually seeks optimal solutions to address these complexities and uphold the democratic values and principles upon which the nation was founded, the need for a more sophisticated and accessible voting system becomes increasingly apparent. This project aims to address these challenges by proposing a modern, secure, and technologically advanced voting system that ensures the full and active participation of all eligible voters, regardless of their geographical location.

II. BACKGROUND STUDY

The proposed project delves into the current voting system in India, traditionally reliant on electronic voting machines (EVMs) during elections. While EVMs have played a crucial role, challenges such as significant manpower requirements, time-consuming processes, and concerns about transparency and trustworthiness persist. In democratic India, where the electoral system is fundamental to governance, ensuring an honest, transparent, and fully secure voting system is imperative. The voting process must instill confidence, uphold democratic principles, and safeguard the integrity of elections. Additionally, the surge in modern communications and internet usage has led to an increased demand for electronic services and robust security measures. The integration of new technologies into the voting process presents a natural opportunity to enhance democracy.

The digital age has spurred a reevaluation of traditional methods across various societal domains, with a focus on the democratic process. Traditional voting systems, marked by manual paper-based ballots and physical polling stations, face logistical challenges and limitations in accessibility and inclusivity. The emergence of online voting systems represents a transformative response to these challenges, utilizing the internet and advancements in cybersecurity. Driven by a pursuit of greater accessibility, efficiency, and cost-effectiveness, these systems aim to democratize the electoral process by enabling eligible voters to cast their ballots remotely. However, the adoption of online voting introduces complexities, including security concerns such as protection against cyber threats and the assurance of voter anonymity. Authentication technologies like biometrics and digital signatures, along with the integration of blockchain for transparency, have become integral components of online voting systems. Pilot programs and trials globally contribute to a growing body of knowledge,

emphasizing the need to balance technological innovation with public trust. The background study of online voting systems provides a nuanced exploration of the evolving intersection between technology, democracy, and the ongoing quest for secure and accessible electoral processes.

III. METHODOLOGY

The development and implementation of an online voting system necessitate a comprehensive methodology to ensure the security, reliability, and accessibility of the electoral process. The initial phase involves meticulous planning and analysis of the requirements, encompassing legal and regulatory considerations. To enhance security, a robust authentication system must be designed to verify voters' identities, incorporating multi-factor authentication. Additionally, encryption techniques should be employed to safeguard the confidentiality and integrity of the voting data. The development process should prioritize accessibility, leading to the creation of a user-friendly interface compatible with various devices.

To identify and address potential vulnerabilities, rigorous testing is imperative, including penetration testing and simulations of real-world scenarios. Continuous monitoring and auditing mechanisms should be implemented to promptly detect and respond to any suspicious activities. Collaboration with cybersecurity experts, election officials, and stakeholders is essential to gain insights and address concerns effectively. Public awareness campaigns should be conducted to educate voters about the online voting process, build trust, and encourage participation. Finally, a contingency plan should be in place to manage any unforeseen technical issues or cyber threats that may arise during the election period. Overall, a comprehensive and well-executed methodology is fundamental to the successful and secure implementation of an online voting system.

Flow chart:



Fig. 1 Flow chart of online Voting System

Algorithm/ Steps in developing the project:

1) Voter Registration:

Collect and verify voter information during the registration process. Assign a unique identifier (such as a cryptographic key) to each voter to ensure anonymity.

2) Authentication:

Implement a robust authentication system using multi-factor authentication. Use strong encryption protocols to protect voter credentials during the authentication process.

3) Ballot Generation:

Generate a unique, verifiable digital ballot for each voter. Ensure the anonymity of the voter while maintaining the integrity of the ballot.

4) Voting Process:

Present the digital ballot to the voter in a user-friendly interface. Allow voters to make selections and review their choices before submission

5) Vote Submission:

Encrypt and securely transmit the completed ballot from the voter to the server. Use secure protocols to protect the transmission of votes.

6) Vote Storage:

Store votes in a secure and tamper-evident manner, using cryptographic techniques to maintain integrity. Implement redundancy and backup mechanisms to prevent data loss.

7) Tallying and Results:

Use cryptographic methods to aggregate and tally votes while maintaining anonymity.

8)Security Measures:

Employ firewalls, intrusion detection systems, and other security measures to protect the online voting system from cyber threats.

8) Auditability:

Implement a comprehensive logging system to record all activities related to the voting process.

9) Contingency Planning:

Develop a contingency plan to address technical issues, cyber attacks, or other disruptions.

IV. RESULT



Fig. 2 Voting System output

The implementation of the online voting system represents a significant leap forward in improving voter accessibility, enabling individuals to participate conveniently from diverse locations. This inclusive approach holds the promise of fostering a more democratic process by eliminating geographical barriers, ensuring that citizens can engage in the electoral system with ease. The efficiency and speed of the electoral process have experienced substantial enhancements through the introduction of digital ballots and secure transmission mechanisms. This innovative approach has resulted in a quicker tabulation of results, expediting announcements and ultimately elevating the overall efficiency of the electoral system.

Moreover, the shift towards online voting has demonstrated potential cost savings by eliminating reliance on traditional methods such as printing and distributing physical ballots. This transition towards a paperless system not only aligns with contemporary environmental sustainability goals but also signifies a dedicated effort to diminish the ecological impact of electoral practices. In essence, the online voting system not only streamlines the democratic process but also contributes significantly to creating a more sustainable and environmentally conscious electoral landscape.

V. CONCLUSION

In the project presents a promising solution to the challenges inherent in India's current voting system by introducing a sophisticated online voting system. In conclusion, the advent of online voting systems presents both opportunities and challenges in reshaping the democratic electoral process. The positive outcomes, such as increased accessibility, convenience, faster results, and potential cost savings, hold the promise of a more inclusive and efficient democracy. However, these advantages must be carefully weighed against significant challenges, including cybersecurity risks, voter authentication concerns, and the digital divide. Building and maintaining public trust through transparency and robust security measures are paramount for the successful implementation of online voting. Addressing legal and regulatory challenges, ensuring privacy protection, and mitigating technical issues are crucial steps toward establishing a reliable and resilient online voting system. As technology evolves, ongoing vigilance and adaptability will be essential to stay ahead of emerging threats. While the journey towards online voting is fraught with complexities, a well-executed and secure system has the potential to enhance democratic participation, streamline processes, and contribute to the evolution of 21st-century electoral practices. As societies continue to explore these digital frontiers, a balanced and comprehensive approach is essential to realize the full benefits while safeguarding the integrity of democratic institutions.

REFERENCE

- 1. Malwade Nikita, Patil Chetan, Chava Suruchi, Prof. Raut S. Y, Secure Online Voting System Proposed By Biometrics And Steganography, Vol. 3, Issue 5, May 2017.
- 2. Ankit Anand, Pallavi Divya, An Efficient Online Voting System, Vol.2, Issue.4, July-Aug. 2019, pp- 2631-2634
- 3. Alaguvel.R, Gnanavel.G, Jagadhambal.K Biometrics Using Electronic Voting System with Embedded Security, Vol. 2, Issue. 3, March 2018.
- 4. Firas I. Hazzaa, Seifedine Kadry, Oussama Kassem Zein, Web-Based Voting System Using Fingerprint: Design and Implementation, Vol. 2, Issue.4, Dec 2019.
- 5. Alexander. Stakeholders: Who is your system for? IEEE: Computing and Control Engineering, 14(1):22{26,April 2003}.
- 6. K. P. Kaliyamurthie, R. Udayakumar, D. Parameswari and S. N. Mugunthan, "Highly Secured Online Voting System over Network," in Indian Journal of Scienceand Technology | Print ISSN: 0974-6846 |Online ISSN: 0974-5645.
- 7. Almyta Systems, Point of Sale Systems.http://systems.almyta.com/Point_of_Sale_, Software.a sp. Accessed on 20th October 2008.
- 8. Swaminathan B, and Dinesh J C D, "Highly secure online voting system with multi security using biometric andsteganography," in International Journal of Advanced Scientific Research and Technology, vol 2(2), 195–203.
- 9. Drew Springall, Travis Finkenauer, Zakir Durumeric, Jason Kitcat, Harri Hursti Margaret MacAlpine J. Alex Halderman, November 3–7, 2014, "Security Analysis of the Estonian Internet Voting System," in CCS"14, Scottsdale, Arizona, USA. ACM 978-1-4503-2957-6/14/11.
- 10. M A Imran, M S U Miah, H Rahman, May 2015, "Face Recognition using Eigenfaces," in International Journal of Computer Applications (0975 8887) Volume 118 No. 5.
- 11. Anand A, and Divya P, "An efficient online voting system," in International Journal of Modern Engineering Research, vol 2(4), 2631–2634.

Efficient Crowd Density Estimation Using Deep Learning

¹Ms. Shweta Jadhav, ²Prof.Ms.J.A. Kendule ¹PG Student, SVERI's College of Engineering, Pandharpur ²Assistant Professor, SVERI's College of Engineering, Pandharpur

Abstract:Efficient crowd density estimation (ECDE) is a very challenging task in various practical scenarios such as weather-degraded photos, dynamic backgrounds, etc. For security and protection by viral infection-based applications such as analysis and surveillance, ECDE is a highly challenging job. In this paper, propose an end-to-end recurrent network for ECDE. The proposed network extracts robust features through the trail of multi-column Convolution Neural network(MCNN) blocks. MCNN is deep learning model specially designed for crowd-counting tasks. Also, we have employed a feature refinement block to refine the features learned by the initial layers. crowd counting data saved do not adequately cover all the challenging events assume in our work, we have gathered and labelled a large new dataset that includes 1000 images with about 10,000 heads annotated. On this challenging all existing datasets, we conduct extensive performances to verify the proposed model. VGG-16 structure model is used for simplicity and performance various computer vision tasks, including image classification. A comparison of qualitative and quantitative experimental findings with state-of- the-art approaches shows the efficacy of the proposed ECDE process. The experimental analysis shows that the proposed network exceeds current state-of-the-art standards for ECDE.

Keywords: Efficient crowd density estimation, MCNN, Deep learning, VGG-16

I. INTRODUCTION

Crowd management has been a topic of concern for many years because accidents frequently occur in high- crowd areas. A finite amount of space is available during shows, protests, or other special occasions, and a high crowd density can present a clear danger for those in the area also our world witnessed major events that attracted a lot of attention toward the importance of automatic crowd scene analysis. An example is the COVID-19 breakout and public events require an automatic system to manage, count, secure, and track a crowd that shares the same area. Crowd analysis is done by the deep learning method. The methods are categorized as crowd counting and crowd action recognition. Moreover, crowd scene datasets are surveyed. The image dataset is then applied to a Gaussian kernel filter shown in Fig. 1 to create a ground truth density map where it will be fed along with its corresponding images to the neural network model for Google Colab training shown in Fig. 2.These challenges, employed and modified a three-tier multicolumn convolutional neural network (MCNN) system

architecture to precisely estimate crowd density. Based on the MCNN system architecture, detected changes in the size of a crowd according to a distance measure and examined additional features that can be incorporated to demonstrate their effects on crowd density maps.



Fig. 1 Gaussian Density Map Formations



Trained in Google colab

Fig. 2 Trained in Google colab

II. LITERATURE REVIEW

Khosro Rezaee [1] Quick and automated recognition of abnormal behaviors in crowded scenes is significantly worthwhile in increasing public security. The traditional procedure of identifying malformation in the Web of Thing (WoT) platform for monitoring the activities and describing the crowd properties such as density and motion pattern from the visual frames. Accordingly, in organization real-time security monitoring based on the WoT platform and machine learning algorithms would significantly increase the effective detection of abnormal behaviors in the crowds. This paper addresses different automatic and real-time surveillance methods for event detection to recognize the dynamic crowd behavior in security applications. The major aspect of security and protection of public places is that we cannot manually monitor un imaginable and complicated crowded environments.

EMAD A. FELEMBAN [2] Hajj, the annual pilgrimage to Makkah, is one of the most huge organizing events in the world. It is a compulsory religious activity, once in a lifetime for every sane well-off Muslim. Every year around 3 million Muslims across the bulky visit Makkah to perform religious activities, like circumambulation around the Kaaba and the tent at Mina, Arafat and Muzdalifah. The key ritual at Rami-al-Jamarat (the stoning-of- the-pillars) is a place seen as particularly crowded. The different nature of the crowd in terms of age, language and culture brings several managerial challenges for local organizers whose responsibility is to ensure the smooth gathering of the event. The entry of digital technologies has helped researches to propose modern methods for effective crowd administrative and control strategies to organize such a mass gathering. In this paper present a divided that summarizes our survey on the utilization of technology areas for providing services and developing crowd management during the Hajj season.

Wei-Teng Weng [3] Crowd administrative has been a topic of concern for many years because accidents frequently occur in events with a huge crowd density. With only a limited amount of space available during shows, protests, or other special events, a huge crowd density can present a clear hazard for those in the area. These challenges, we employed and divided a three-tier multicolumn convolutional neural network (MCNN) deep learning system architecture to precisely manage crowd density. We difference three regions from the near to far field to produce a crowd density map. Based on the MCNN deep learning system architecture, we find out changes in the size of a crowd according to a distance measure and examine more features that can be corporate to demonstrate their effects on crowd density maps.

Renhe Jiang [4] Assuming the density and flow of the crowd or traffic at a city level becomes possible by using the large data and cutting-edge AI technologies. It has been a very significant research point with more social impact, which can be frequently applied to emergency management, traffic management, and urban management. In particular, by meshing a large urban area to a number of fine- grained mesh grids, citywide crowd and traffic information in a continuous time period can be presented with a 4D tensor. Based on this idea, a series of methods have been proposed to address grid-based prediction for citywide huge crowds and traffic. In this research, we revisit the density and in-out flow assumption problem and publish a new human mobility dataset generated from a real-world smartphone feature.

III. CREATIVE MECHANISM DESIGN METHODOLOGY

ECDE is synthesized by the methodology Software Design Life Cycle (SDLC). The process is in a circular motion with 6 steps as shown in Fig. 3 below. The planning stage i s where the necessary requirements need to be found and the expected timeline of the whole process is charted. After that, the next stage is the Design stage where a rough outline of the design is made and is updated thoroughly. When that is finish, the development stage starts which means the programming and research begins. That process would be the one that takes the largest, so careful management is definitely needed beforehand. Next, when the project model is all done, it is then tested for any errors or unintended results. It's basically the debugging process. At the end of the first loop, the project model is working but will still need some more fine-tuning or results in the process. The consist 6 steps are following,]



Fig. 3 SDLC Process Flowchart

Requirement Analysis

To search for an existing design project model or study a present new design model with fulfills the required specifications and making the topological structure of these models is the first step of the methodology. The goal of this model for searching their equivalent mechanism skeleton and kinematic chain for developing new designs.

Planning

Planning for the quality assurance requirement and identification of the risk associated with the project is also done in the planning stage.

Architectural Design

The architectural design is important to stakeholders and is based on different parameters such as risk assessment and software. The best design approach is selected for the product for reliability, design modularity, budget, and time constraints.

Software development

Software development is the initial step of development start of the project. By requirements, the programming code is generated. If the design is performed in a detailed and organized manner, code generation can be completed without much hassle.

Testing

This stage is usually an important of all the stages as in the modern SDLC models, the testing process is mostly involved in all the stages of SDLC. However, this stage refers to the testing-only stage of the project where project defects are reported, tracked, fixed, update and retested until the product reaches the quality standards.

Development

Development is the last step of the project model. After the all steps project was developed with accuracy, and all requirements were completed result. This is the final stage

Requirement Analysis

The traditional approach of crowd density estimation is an extremely developed and popular field because of the powerful feature extraction and classification technique. However, crowd density estimation is a challenging problem with models for crowd management system by using image classification and a system of alarms to manage thousands of people during events.

During the development and testing of a neural network model, the developer has to create a separate file to test its accuracy which takes a lot of time during the development. There is rarely used a graphical user interface (GUI) for it too development by creating a ready-to-use system that could take in the parameter file, ground truth file, and image files to output it's estimated density map, estimated count from crowd, and a graph analysis of its accuracy result as well as the accuracy Mean Absolute Error (MAE) and Mean Squared Error (MSE). Furthermore, once the development is finished, the system can be used independent to determine the number of people present. The goal of this project is to improve my Python programming abilities while simultaneously providing a window into the advancement of deep learning, notably the convolution neural network, in the context of the 4.0 Industrial Revolution. Additionally, this project will demonstrate to potential employers that I have the capacity and want to master new skills.

The objectives for this project are to train a neural network model to give the number of people within crowd using the PyTorch framework. The neural network that will be used is called MCNN. The dataset that will be used to train the said model is the ShanghaiTech Dataset, a dataset that has the coordinates of the heads of the people in the images or video and are marked and save in a. mat file. The image dataset is then applied a Gaussian kernel filter to make a ground truth density map where it will be give along with its corresponding images to the neural network model for training. After training is finished, the model would then create the image's density map and estimate the number of crowds from there. The result of the model would be calculated using the Mean Absolute Error and Mean Squared Error as it is the standard for most of the crowd counting algorithms out there.

Planning

First make the neural network and train it. The development will be done using the PyTorch framework and Python3. Also, the practice of object-oriented coding will also be used throughout this project. The neural network design will be stored in PythonModel.py, the program that will insert the image data will be saved in dataloader.py, the code to create the ground truth density map will be saved ingaussian.py and the file to train the neural network will be called train.py. Once the training of the neural network is complete and tested successfully in Google Colab, the development of the front-end can be initiated. The overall design of the GUI will be done using HTML, CSS, JavaScript to help expedite the development time. After that, the GUI has to be connected to methods that will connect with the neural network to form a working model that could imagine the crowd count and also, test the neural network for its accuracy. To show the graphs and also the density map, the matplotlib Python library is used to visualize the data.



Fig.4 GUI

The general code construction of this venture will be parted into a few Python scripts for effortlessness purpose. Also, the Pytorch structure is object-oriented based which makes developing the model a lot more straightforward.

- 4.1 gaussian.py responsible for creating the ground truth density map
- 4.2dataloader.py contains the Dataset class needed.
- 4.3PythonModel.py contains the Network model class needed.
- 4.4train.py responsible for implementing the Dataset and Network class together and train said model.
- 4.5App.py launches our localhost port 5000.
- 4.6Inferences- Initializes our model and makes the predictions.
- 4.7Commons References our model file.
- 4.8Model.pt- Our trained model.
- 4.9Templates- Contains index and result templates.



IV. ARCHITECTURAL DESIGN

Fig. 5 Process design

ECDE architectural design is based on MCNN deep learning. In that from input to output one structure is designed that means architectural design which is show in fig. 5.

In that structure, data is collected from the dataset for segmentation. After that process, data goes to the trained GP model through the feature extraction. The Gaussian processes model is a chances supervised machine-learning framework that has been widely used for regression and classification tasks. A Gaussian process regression (GPR) model can create assumptions corporation prior knowledge (kernels) and give uncertainty measures over assumptions. The gaussian processes model is an observed method developed by computer science and statistics institutes. Researchers with engineering backgrounds often search it difficult to gain a clear understanding of it. To understand GPR, even only the basics need to have data of multivariate normal distribution, kernels, non-parametric model, and joint and conditional situations.

Tools needed

A working computer equipped with a CUDA-supported discrete graphics processing unit (GPU). In that preferably an NVIDIA card with RT cores, at least 16GB RAM and a CPU with at least 4 cores and 8 threads. Pytorch machine learning library, python 3, pycharm, package installer for python, serval external python libraries.

MCNN

MCNN, which stands for Multi-Column Convolutional Neural Network, is a deep learning model specifically designed for crowd counting tasks. It was proposed by Zhang et al. in their paper titled Single- Image Crowd Counting via Multi-Column Convolutional Neural Network in 2016. The MCNN model aims to estimate the number of individuals in a crowd by analyzing a single image. It consists of three columns, each representing a separate pathway in the network. The three columns process the input image at different scales to capture features of different sizes, allowing the model to detect individuals at various scales within the crowd. The MCNN model's efficiency for crowd counting stems from its ability to capture multi-scale features, shared weight strategy, density map estimation, and appropriate loss function. These design choices enable MCNN to achieve competitive crowd counting accuracy while maintaining computational efficiency.



Fig. 6 Structure of the proposed Multi-Column Convolutional Neural Network

The improvements of the neural network model will be measured in mean absolute error as well as mean squared error. Mean Absolute Error is an evaluation tool used for regression models where the mean absolute error of a model in terms of its test set is the mean of the absolute values of the individual prediction errors on every step in the test set. So, the difference between the true value and the redacted value for the instance is the prediction error. The targeted MAE f or this model is to be as low as possible shown below.

$$mae = rac{\sum_{i=1}^n abs \left(y_i - \lambda(x_i)
ight)}{n}$$

Mean Squared Error on the other hand is also an evaluation tool used for regression models. The mean square error of the model in terms of its test set is the mean of the squared prediction error on all the instances of the test set. The difference between the true value and the predicted value of the instance is the prediction error. The targeted MSE for this models to be as low as possible [13] shown below.

$$mse = rac{\sum_{i=1}^n (y_i - \lambda(x_i))^2}{n}$$

VGG-16 structure

VGG-16 structure is a deep convolutional neural network architecture that was proposed by Simonyan and Zisserman in 2014. It is called VGG-16 because it consists of 16 layers, including 13 convolutional layers and 3 fully connected layers. VGG-16

has become a popular model due to its simplicity and strong performance on various computer vision tasks, including image classification.

VGG-16 has gained popularity and is considered better than some other architecture is its deep and uniform structure. It has a straightforward design, with small 3x3 filters and max pooling layers applied consistently throughout the network. This uniformity makes it easier to understand and implement compared to architectures with complex skip connections or varying filter sizes.



Fig. 7 VGG-16 structure

Software Development

In software development given steps of whole development process. Before the system can be developed, a neural network needs be trained. Steps contains are following,

Train dataset using Google colab

A dataset has to be prepared. To ready the dataset for the neural network, images from the dataset must be insert into a Gaussian filter to make a density map in the form of NumPy array. For this, gaussian.py is used where the converter coordinates of the people's head in the image are applied a Gaussian filter and the resulting map is saved as. npy file. Google Colabs like Jupyter notebooks. They are convenient because Google Colab hosts them, so we don't use any of our own computer sources to run the notebook. We can also share these notebooks so other people can quickly run our program, all in a standard environment since it's not dependent on our own machines. However, we need to install some libraries in our environment during initialization. Google Colab steps are ready for the Google Colab notebook, Loading the necessary libraries, Constant for image dimension and batch size, giving the model architecture, accessing data from Google Drive, Define the directions for ShanghaiTech dataset, process the data and make data generators, Build the MCNN model, compile the model, Define call back and Training that.



Fig. 8 Final trained model saved in model.pt

Development of Web application using Flask

Flask is a light and flexible web framework for Python. It is contract to make it reliable to build web features and APIs. Flask follows the WSGI (Web Server Gateway Interface) specification and can be used with any WSGI-compliant web server. Here

are many key features and characteristics of the Flask Framework Minimalistic, quickly to get started, increasable, flexible routing, tempting engine, RESTful API development, testing- debugging and ecosystem, applicable, interfaces, commons, model.pt



Fig. 9 UI index.HTML



Fig. 10 UI Result.HTML

V. TESTING

Testing is must to all developments. Through testing find errors for more flexible and reliable development. In testing whole process and run that unit wise for find issues. After testing and correcting errors development is completed.

VI. DEVELOPMENT

Final step is final development. In that check process from select dataset and it send to GP model for trained and after that through UI index.HTML estimate the crowd density.

VII. CONCLUSIONS

In this work, an end-to-end recurrent adversarial learning network is proposed for efficient crowd density estimation with trail of Multi-Column Convolutional Neural Network block. Multi-column Convolutional Neural Network block extracts multi column features with the help of Multi-Column Convolutional Neural Network block. Refinement block is employed to refine the learned feature maps of the preceding layers. It takes the learned feature maps from multi-column dense residual block and respective target encoder block through skip-connections and output a refined set of feature maps. The extracted features from different layers and output for previous frame are shared through recurrent technique to provide enough temporal information. Further, the learning of network is concentrated on different ways like global and cross-dataset training-testing for ECDE. Extensive experimental analysis of the proposed network outperforms the existing state-of-the-art methods on two benchmark datasets for ECDE.

Conflicts of Interest Authors must identify and confirm any personal situations or interests that may be understood as inappropriately influencing the representation or definition of reported research results. If there is no dispute of interest, please state "The authors declare no dispute of interest."

REFERENCES

- 1. S. M. Metev and V. P. Veiko, Laser Assisted Microtechnology, 2nd ed., R. M. Osgood, Jr., Ed. Berlin, Germany: Springer-Verlag, 1998.
- 2. Emada A. Felemban, Faizan UR Rehman, "Digital Revolution for Hajj Crowd Management: A Technology Survey," Creative Commons Attribution 4.0 License. IEEE, 2021
- 3. Daw-Tung Lin, Wei-Teng Weng, "Crowd Density Estimation Based on a Modified Multicolumn Convolutional Neural Network," IEEE, 2018.
- 4. Renhe Jiang, Zekun Cai, "Deep Crowd: A Deep Model for Large-Scale Citywide Crowd Density and Flow Prediction," IEEE, 2021.
- 5. Elizabeth B. Varghese, Sabu M. Thampi, "Towards the cognitive and psychological perspectives of crowd behaviour: a vision-based analysis," Taylor & Franc, 2020.
- 6. Elizabeth B Varghese, Sabu M Thampi, "Application of Cognitive Computing for Smart Crowd Management," IEEE Computer Society, 2020.
- 7. Rodolfo W. L. Coutinho, Azzedine Boukerche, "Crowd Management: The Overlooked Component of Smart Transportation Systems," IEEE Communications Magazine. April 2019.
- 8. Vu-Hoang Tran, Ching-Chun Huang, Van-Su Huynh, "IUML: Inception U-NET based multitask learning for density level classification and crowd density estimation," IEEE, 2019.
- 9. Ujwala Bhangale, Suchitra Patil, "Near real time crowd counting using deep learning approach," Procedia Computer Science, 2020.
- 10. Jung-jae Kim, Ali Zonoozi, "Periodic-CRN: A Convolutional Recurrent Model for Crowd Density Prediction with Recurring Periodic Patterns," Joint Conference on Artificial Intelligence (IJCAI-18), 2018.
- 11. Neeraj N Sajjan, Deepak Babu Sam, "Almost Unsupervised Learning for Dense Crowd Counting," Artificial Intelligence(AAAI-19), 2019.
- 12. Sarkar Snigdha Sarathi Das, "CCCNet: An Attention Based Deep Learning Framework for Categorized Counting of Crowd in Different Body States," Auckland University of Technology, 2020.
- 13. Miao, Y., Han, J., Gao, Y. & Zhang, B., 2019. ST-CNN: Spatial-Temporal Convolutional Neural Network for crowd counting in videos. Pattern Recognition Letters, Volume 125, pp. 113-118. Sammut, C. & Webb,
- 14. G. I., 2011. Enyclopedia of Machine Learning. 2010 ed. Boston, MA: Spring

Model for Retinal Disease Diabetic Retinopathy

¹Isharani, ²Vimala Kulkarni, ³Prerana, ⁴Amul Guru Nanak Dev Engineering College Bidar ,Karnataka.India

Abstract—Diabetic retinopathy is a prevalent problem worldwide, causing many people to lose their vision. If left untreated, the disease can become severe, leading to damage in the retinal blood vessels and blocking the passage of light through the optical nerves. This can result in blindness for patients with Diabetic Retinopathy. Our research aimed to find a solution to this problem, and we were able to use a Convolutional Neural Network (ConvNet) to detect multiple stages of severity for Diabetic Retinopathy. While manual screening is another method of detecting this condition, it requires a skilled ophthalmologist and takes up a considerable amount of time. Our automatic diabetic retinopathy detection technique can replace such manual processes, allowing ophthalmologists to spend more time providing proper care to patients and potentially reducing the severity of this disease.

Keywords: retinopathy, convolutional neural network, the ophthalmologist.

I.INTRODUCTION

Diabetic retinopathy (DR) is a serious complication of diabetes that can potentially lead to vision loss. It affects the retina, which is the light-sensitive tissue located at the back of the eye. The condition can develop without noticeable symptoms in the early stages, making early detection crucial for effective intervention and prevention of vision loss. As diabetes becomes more prevalent worldwide, there is an urgent need for accurate and efficient methods of diagnosing diabetic retinopathy.

To address this challenge, our machine learning project aims to develop a reliable model for detecting diabetic retinopathy. Leveraging artificial intelligence and image processing, we aim to create a system that can analyze retinal images and identify signs of diabetic retinopathy with high accuracy.

By using machine learning algorithms, we seek to automate the process of diagnosing diabetic retinopathy, providing a faster and more scalable solution compared to traditional manual screening methods. The integration of machine learning into healthcare has the potential to revolutionize the early detection and management of diabetic retinopathy, ultimately improving patient outcomes and reducing the burden on healthcare systems.

This project is not just a technological advancement but also a step towards making early diabetic retinopathy screening accessible to people in regions with limited healthcare resources. Developing an accurate and efficient machine learning model could significantly contribute to global efforts aimed at reducing the impact of diabetic retinopathy on individuals' vision and quality of life.

Our interdisciplinary team of data scientists, healthcare professionals, and experts in the field of ophthalmology collaborates to bring together the best of technology and medical expertise. Through the fusion of cutting-edge machine learning techniques and domain-specific knowledge, we aim to create a powerful tool that can make a tangible difference in the lives of individuals at risk of diabetic retinopathy.

II. METHODOLOGY

Methodology for Diabetic Retinopathy Detection using ConvNets

Step 1: Data Collection and Preprocessing

Gather a dataset of retinal images, labeled for diabetic retinopathy severity. Preprocess images: resize, normalize pixel values, and perform data augmentation to increase dataset size if needed.

Step 2: Data Splitting

Split the dataset into training, validation, and testing sets (e.g., 70-15-15 ratio) to train, tune, and evaluate the model, respectively.

Step 3: Building the ConvNet Model

Design Convolutional Neural Network architecture suitable for image classification tasks. Typical layers might include convolutional layers, pooling layers, batch normalization, and fully connected layers at the end. Experiment with various architectures to find the best-performing one.

Step 4: Training the Model

Train the ConvNet using the training dataset. Use a loss function (e.g., categorical cross-entropy) and an optimizer (e.g., Adam) to minimize the loss. Monitor the performance on the validation set to avoid overfitting.

Step 5: Evaluation and Validation

Evaluate the trained model on the test dataset to assess its performance. Measure accuracy, precision, recall, F1-score, and confusion matrix to understand the model's performance in detecting diabetic retinopathy. Fine-tune the model parameters if needed based on evaluation metrics.

III. WHAT WAS OUR APPROACH?

Why we have selected ConvNet to solve this problem? / Objective

Diabetic retinopathy is a major cause of blindness among working-age individuals in developed countries, affecting more than 93 million people. Efforts have been made to develop an automated and comprehensive screening method for diabetic retinopathy, using image classification, pattern recognition, and machine learning. The aim of this project is to create a new model that can realistically be used in clinical settings, using photos of eyes as input.

The project has two main motivations: firstly, image classification has been a personal interest for a long time, particularly on large-scale datasets. Secondly, there is a significant amount of time lost between patients getting their eyes scanned, their images being analyzed by doctors, and scheduling a follow-up appointment. EyeNet, the proposed model, aims to process images in real time, enabling patients to seek and schedule treatment on the same day.



Fig 1. FLOWCHART

The image is a flowchart that shows the steps involved in a machine-learning algorithm for the early identification of diabetic retinopathy. The algorithm takes a fund-us image as input and outputs a classification of the image as either normal, non-proliferative diabetic retinopathy (NPDR), proliferative diabetic retinopathy (PDR), or diabetic macular oedema (DME).

The first step in the algorithm is per-processing, which involves cleaning the image and removing any artefacts. The next step is optic disc detection, which involves identifying the optic disc in the image. The optic disc is the central part of the retina and is important for detecting diabetic retinopathy.

The next step is blood vessel detection, which involves identifying the blood vessels in the image. Blood vessels are also important for detecting diabetic retinopathy. The next step is microaneurysms detection, which involves identifying microaneurysms in the image. Microaneurysms are small bulges in the blood vessels that are an early sign of diabetic retinopathy.

The final step in the algorithm is disease classification, which involves classifying the image as either normal, NPDR, PDR, or DME. The algorithm uses the information from the previous steps to make this classification.

The flowchart shows that the algorithm is a sequential process, with each step building on the previous step. The algorithm is designed to be efficient and accurate, and it is effective in identifying diabetic retinopathy in the early stages.

IV. HOW WE CAN MAKE IT ACCESSIBLE TO DOCTORS?

: In our research, to tackle the aforementioned challenges, we built a predictive model for Computer-Aided Diagnosis (CAD), leveraging eye fundus images that are widely used in present-day hospitals, given that these images can be acquired at a relatively low cost. Additionally, based on our CAD model, we developed a novel tool for diabetic retinopathy diagnosis that takes the form of a prototype web application. The main contribution of this research stems from the novelty of our predictive model and its integration into a prototype web application.

V. ACKNOWLEDGEMENTS

"We express our sincere gratitude for the utilization of the advanced model for retinal disease, specifically in diabetic retinopathy detection. Our acknowledgement extends to the developers and contributors whose efforts have significantly contributed to the success and accuracy of the model, enhancing our understanding and diagnosis of diabetic retinopathy".

VI. CONCLUSION:

A successful deep-learning system for detecting diabetic retinopathy requires a multi-disciplinary approach involving collaboration between data scientists healthcare professionals and other stakeholders. Regular updates and continuous monitoring ensure the liability and effectiveness of the system in the real world.

REFERENCES:

- 1. Liew, G., Michaelides, M. & Bunce, C. A comparison of the causes of blindness certifications in England and Wales in working-age adults (16–64 years), 1999–2000 with 2009–2010. BMJ Open 4, e004015 (2014).
- 2. Kristinsson, J. K., Hauksdottir, H., Stefansson, E., Jónasson, F. & Gíslason, I. Active prevention in diabetic eye disease: a 4-year follow-up. Acta Ophthalmol. Scand. 75, 249–254 (1997).
- 3. Wang, L. Z. et al. Availability and variability in guidelines on diabetic retinopathy screening in Asian countries. Br. J. Ophthalmol. 101, 1352–1360 (2017)1.
- 4. Poplin, R. et al. Prediction of cardiovascular risk factors from retinal fundus photographs via deep learning. Nat. Biomed. Eng. 2, 158–164 (2018).
- 5. Arcadu, F. et al. Deep learning predicts OCT measures of diabetic macular thickening from colour fundus photographs. Invest. Ophthalmol. Vis. Sci. 60,852–857 (2019).
- 6. Samala, R. K. et al. Mass detection in digital breast tomosynthesis: deep convolutional neural network with transfer learning from mammography. Med. Phys.43, 6654–6666 (2016).
- 7. Lakhani, P. & Sundaram, B. Deep learning at chest radiography: automated classification of pulmonary tuberculosis by using convolutional neural networks. Radiology 284, 574–582 (2017).
- 8. Esteva, A. et al. Dermatologist-level classification of skin cancer with deep neural networks. Nature 542, 115–118 (2017).
- 9. Litjens, G. et al. A survey on deep learning in medical image analysis. Med. ImageAnal. 42, 60–88 (2017).

- 10. [12]Gulshan, V. et al. Development and validation of a deep learning algorithm for the detection of diabetic retinopathy in retinal fundus photographs. JAMA 316,2402–2410 (2016).
- 11. D. S. W. et al. Development and validation of a deep learning system for diabetic retinopathy and related eye diseases using retinal images from multiethnic populations with diabetes. JAMA 318, 2211–2223 (2017).
- 12. Murchison, A. P. et al. Nonadherence to eye care in people with diabetes. BMJ Open Diab. Res. Care 5, e000333 (2017).
- 13. Brar, M. & Ho, A. C. Diabetic eye disease: a multidisciplinary discussion on managing diabetic eye disease. EyeNet Magazine Suppl. 1–4 (2016).
- 14. American Diabetes Association. 11. Microvascular complications and foot care: standards of medical care in diabetes-2020. Diabetes Care 43, S135–S151 (2020).
- 15. International Diabetes Federation. IDF Diabetes Atlas 7th edn (International Diabetes Federation, Brussels, Belgium, 2015).

Design and Analysis of Microstrip Array Antenna for IoT Applications

¹Dr.Gurudev Zalki1, ²Dr.Mohammed Bakhar

¹Department of Electronics and Communication Engineering Sharnbasva University Kalaburagi-585103, Karnataka, India. ²Department of Electronics and Communication Engineering Guru Nanak Dev Engineering College, Bidar-585403, Karnataka, India.

Abstract— This paper presents the design of a microstrip array antenna (MAA) for IoT applications at a frequency of 2.4 GHz. Our objective is to design and analyse single-element and 1x4 rectangular MAA for IoT smart homes, medical fields, and industrial applications. For designing, we use FR4 substrate material with a dielectric constant of 4.4 with a 1.6mm thickness, and microstrip line feeding technique is used in designing of MAA. The Various antenna parameters namely Return Loss (RL), Impedance Bandwidth, Gain and Radiation Patterns are calculated. The 3DEM of Mentorgraphics software is used for design and simulation purposes.

Keywords: Microstrip antenna arrays, FR-4 Substrate, microstrip feed technique, 2.4 GHz and IoT.

I INTRODUCTION

The importance of microstrip antennas in the field of wireless communication applications is very advantageous because of their small size, low cost and easy for fabrication. Every antenna in known because of their design implementations, characteristics and different applications. The present days are difficult to be imagined without the use of technology and microstrip antennas are strength and backbone for different wireless communication and their applications. In the today's era of wireless communication technology application world the microstrip patch antennas perform a very important role [1].

The rectangular MAA are frequently considered as compared to hexagon, circular, triangular etc. The MAA are used in simple as well as in most challenging applications. The broad bandwidth, feed line flexibility, beam scanning and circular polarizations can be achieved using MAA. We use the FR4 (flame retardant) dielectric material with dielectric Constant ϵ r=4.4. FR4 is chosen because reasonably it is low-cost material and it is easily available in the market.

II. ANALYSIS AND DESIGN

The width and length of the single element is calculated using the formula [2] The width 'W' of designed patch is calculated by using the equation

$$W = \frac{c}{2f_r}$$
(1)

The Length 'Lp' of designed patch is calculated by using the equation

$$L_p = L_{eff} - 2\Delta L \tag{2}$$

The Extension Length ' Δ L' of designed patch is calculated by using the equation

$$\frac{\Delta L}{h} = 0.412 \frac{(\epsilon_{reff} + 0.3)}{(\epsilon_{reff} - 0.258)} \left(\frac{\frac{W}{h} + 0.264}{\frac{W}{h} + 0.8}\right)$$
(3)

The relationship between actual length 'L' and effective length 'Leff' of designed patch is calculated by using the equation

$$L_{eff} = L + 2\Delta L \tag{4}$$

The effective length of patch is calculated by using the equation
$$L_{eff} = \frac{\lambda_g}{2} \tag{5}$$

While designing the MAA we considered spacing to be 62.85 mm between each element [3].

III. DESIGN OF SINGLE PATCH AND 1x4 ANTENNA ARRAY

3.1 Design of Single Rectangular Patch Antenna



Fig.1. Design of Single Patch Antenna

The designed single microstrip patch antenna of size 38.03 mm width and 29.44 mm length is designed for 100Ω feedline whose width is 0.7mm and length is 5mm is shown in Fig.1.

3.2 Design and Analysis of 1x4 Antenna Array [4]



Fig.2. Design of 1x4 MAA

The Fig.2 shows design of 1x4 MAA having 100 Ω feed line is used to connect all the patches. The quarter wave transformer impedance is calculated as $Z = = 70.7\Omega$ is positioned among in between 50 Ω equivalent point and 100 Ω line. The calculated patch and feedline dimensions considered are tabulated in below Table 1.

Table 1 Patch & Feedline Dimensions of 1x4 Array

Size of each patch	38.03 mm x 29.44 mm
Spacing between patches	62.85 mm
100Ω feedline width	0.7 mm
70.7 Ω feedline width	1.6 mm
50Ω feedline width	3 mm

IV. SIMULATION RESULTS OF SINGLE PATCH AND 1x4 ANTENNA ARRAY

The proposed antenna is designed [5-10] and simulated in 3DEM of Mentorgraphics software. We calculated Return loss, Impedance bandwidth, Gain and Radiation patterns.

4.1 Simulation Results of Impedance Bandwidth & Return Loss



4.2 Simulation Results of Radiation pattern

The radiation patterns of single Patch and 1x4 antenna array at 2.4 for IoT smart homes, medical field and industrial applications are shown in Fig. 4 (a) and (b) respectively. The Fig. 4(a) shows radiation pattern of single patch, where main lobe is widely spread and side lobes are almost null in nature. Whereas Fig. 4(b) shows radiation pattern of 1x4 array, where main lobe is well directed in uniform direction with two minor side lobes.

The measured maximum Half Power Beam Width (HPBW) values of the single Patch and 1x4 antenna array are 101.620 and 50.140 respectively.



(a)



Fig.4. Radiation Patterns obtained for the (a) single patch (b) 1x4 array

4.3 Simulation Results of Gain

The Fig. 5 (a) and (b) shows maximum total field gain values of the single and 1x4 MAA. The total field gain value obtained for single patch antenna is 1.87 dB and total field gain value obtained for 1x4 antenna array is 6.28 dB respectively at frequency of 2.4 GHz.



(a)



Fig.5. Gain (a) single patch (b) 1x4 MAA

The obtained simulated results are summarised for a single patch, and a 1x4 MAA is represented in Table 2. The return loss value for both antennas is less than -10 dB, which is desirable. There is a slight improvement in bandwidth value and a significant improvement in gain, that is, for the array configuration, we obtained three times more gain as compared to the gain for a single antenna and overall good radiation patterns.

Table 2 Combined Results Obtained for Single Patch and 1x4 Antenna Array

	Return loss (dB)	Impedance Bandwidth (MHz)	Gain (dB)	HPBW
Single Patch	-40.5 dB	64.58 MHz	1.87 dB	101.62°
1x4	-27.7 dB	110 MHz	6.28 dB	50.14°

V. CONCLUSION

The rectangular single patch and 1x4 antenna arrays were designed and analysed successfully using the FR4 dielectric material. It was observed that the antenna resonated at a frequency of 2.4 GHz, which can be used for IoT smart homes, medical fields, and industrial applications and results were obtained by using 3DEM of Mentor graphics software. From the proposed design for a single patch, the gain obtained is 1.87 dB and for a 1x4 antenna array the gain obtained is 6.28 dB.

REFERENCES

- 1. Balanis, C. A. (2005). Antenna Theory. John Wiley & Sons.
- 2. Qurratul Ayn , Dr. P.A. Nageswar Rao , Prof. P. Mallikarjuna Rao. (2017). Design and Analysis of High Gain 2x1 and 4x1 Circular Patch Antenna Arrays for 2.4 GHz Applications.
- 3. Norfishah Ab Wahab, Zulkifli Bin Maslan, Wan Norsyafizan W. Muhamad, Norhayati Hamzah. (2010). Microstrip Rectangular 4x1 Patch Array Antenna at 2.5GHz for WiMax Application.
- 4. Chandrahasa R. Salian, Santhosh B., Sandeep Vedagarbham. (2017). Design and Development of High Gain Patch Antenna Array for ISM Applications.
- 5. K. Shambavi, Zachariah c alex, T. Naveen phani krishna. (2009). Design and analysis of high gain millimeter wave microstrip antenna array for wireless applications.
- 6. Pallavi Singhal and Kuldeep Jaimini. (2014). Rectangular Microstrip Patch Antenna Design using IE3D Simulator. 3DEM of Mentorgraphics User's Manual, Release 9, Zeland Software.
- 7. Wen-Shan Chen and Yung-Chi Lin. (2018). Design of 2×2 microstrip patch array antenna for 5GC-band access point applications.
- 8. A.R.G.Chandra Mouli.K, Dr M.Satya Anuradha, Darshana Kuppili , Dattatreya Gopi. (2018). Analysis of Rectangular and Triangular Microstrip Antenna Arrays.
- 9. Jagtar Singh Sivia, Sunita Rani. (2015). Design of Rectangular Microstrip Patch Antenna Array.

Leaf Based Plant Disease Detection Using Convolution Neural Network

¹Rajshekar Gaithond Arpita, ²Amisha ^{1,2} Prof ,Deptartment of Computer Science and Engineering, Guru Nanak Dev Engineering College Bidar

Abstract—Conventional has the problems of slow training speed, single characteristic scale and low recognition accuracy. To solve these problems, a convolution neural network identification model based on Inception module and dilated convolution is proposed in this study. The inception module combined with dilated convolution, could extract disease characteristics at different scales and increase the receptive field. By setting different parameters, six improved models were obtained. They were trained to identify 15 diseases of 10 different crops; then the authors selected optimal recognition model. On this basis, the segmented dataset and the grey-scaled dataset were trained as comparative experiments to explore the influence of background and color features on the recognition results. After only two training epochs, the improved optimal model could achieve an accuracy of over 85%. Moreover, the final average identification accuracy reached 95.37%. Contrast experiments indicate that color and background features may influence the recognition effect. The improved model can extract disease information from different scales in the feature map to identify diverse diseases of different crops. The proposed model has faster training speed and higher recognition accuracy than the traditional model, and thus it can provide a reference for crop disease identification in actual production.

Keyword- CNN, Texture, classification, Caffe framework, AlexNet

I. INTRODUCTION

These Crop diseases [[1]] are the main factors affecting agricultural development. The yield and quality of agricultural production are closely related to the occurrence area and harm the degree of crop diseases. Although modern chemical pesticides can effectively control crop diseases, different crops have diverse diseases. Farmers identify crop diseases relying on visual observation and experience, which is prone to misdiagnosis and easily leads to the misuse of pesticides. Therefore, it is of great significance to help farmers diagnose crop disease types timely and accurately [[2]-[4]]. To solve the problem, experts and scholars have been working to develop a method [[5], [6]] that can automatically, quickly and accurately identify a variety of crop diseases.

Provided good support to extend the related research and application of image processing [[7], [8]] in the field of agriculture. In recent years, many types of research based on computer vision technology have been used to identify crop diseases. Tellaeche et al. [[9]] studied the issue of weed disease identification among the fields. According to the principle of the orderly arrangement of weeds among the fields, they used the Hough transform method and regional marking method to identify weeds between crops in different observation angles and spaces. Liu et al. [[10]] firstly applied the fast independent component method to removing random noise of the disease images, then extracted the colour and texture features of the disease persimmon surface image. Finally, the support vector machine (SVM) was used to identify and classified plant diseases according to edge detection and histogram matching methods. Ma et al. [[12]] firstly segmented the lesion image, then extracted 25 features including colour, texture and shape features, and constructed an SVM classifier based on radial basis kernel function, which accurately identified downy mildew of cucumber, with an accuracy of 90%.

Most researches in the above literature are based on image feature extraction and traditional instructed algorithms to identify crop diseases. Although these methods have a good recognition effect for specific crop diseases, the characteristics such as colour and texture sometimes do not reflect the disease information well and the recognition effect is weakened. In addition, the processes of image pretreatment and feature extraction require artificial selection and calibration, which consume a lot of time and energy. It is not conducive to the promotion and application of disease identification methods. Another problem is that the identification of these research methods focuses on limited kinds of crops and diseases. In the actual production, farmers who grow different crops usually need to diagnose various diseases, so the above methods are perhaps not effective for different demands.

In recent years, deep learning has become a hot research pot in machine learning [[13], [14]] and has been successfully applied in many fields. Convolutional neural network (CNN) with strong ability to automatically extract features has been

widely used in the field of image recognition [[15]-[17]], such as street scene recognition [[18]], face recognition [[19], [20]], object detection [[21]] and image segmentation [[22], [23]]. Meanwhile, CNN has been applied in the agricultural field to detect and identify crop diseases. Dyrmann et al. [[24]] trained a typical CNN model and successfully identified 22 species with a classification accuracy of 86.2%. Lu et al. [[25]] identified ten common rice diseases using a novel method based on the CNN technology, with an average recognition accuracy of 95.48%. Cheng et al. [[26]] constructed a deep residual learning model to identify ten crop pest images in the complex background and achieved an accuracy of 98.67%. Too et al. [[27]] studied crop disease identification using a method of fine-tuning existing deep learning models.

The above researches are sufficient to show that the recognition methods based on deep learning, especially CNN, are effective in the field of agriculture. However, there exist some defects and difficulties in the above researches, involving the recognition rate are higher for specific samples under certain circumstances. In the diagnosis model, some parameters are not optimal and the convergence speed of the training algorithm is slow. The purpose of this study is to construct an improved CNN-based model to achieve fast and accurate automated recognition of crop leaf disease images.

In this paper, we refer to some new developments [[28]] in the field of deep learning, and propose an improved disease recognition model combined with dilated convolution and multi-scale feature fusion structure (Inception module). The improved model is trained to identify 38 different classes, including diseased and healthy crop leaf images. The recognition effect of improved model is compared with that of the traditional AlexNet [[29]] model. Based on the above works, the improved model is also trained via the grey-scaled dataset as well as segmented dataset corresponding to the raw images. Therefore, the influences of colour features and background noise on disease identification methods can be explored. The core motivation of this paper is to provide a reference for developing a crop disease identification system which is suitable for farmers by using a common digital camera.

Our main contributions can be summarised as follows: (i) We propose a CNN framework based on AlexNet for the identification of various crop diseases in the agricultural field.

(ii) An inception module combined with dilated convolution is proposed to obtain a larger receptive field and extract multiple features.

(iii) The influence of background and colour features on the recognition results is explored.

III. MATERIALS

Plant Village Project (www.plantvillage.org) is an open-source database for diagnosing crop diseases. It aims to enrich images of the database and use the database to solve problems related to plant diseases. The database contains diseased and healthy leaf images of various crops. This study uses 54,105 leaf images collected by the Plant Village Project as an experimental dataset, with 26 diseases for 14 crops and healthy leaves of some crops. Several typical images are shown in Fig. 1.



Fig. 1

Raw crop leaf image. Note: Label 1 is apple scab leaf. Label 2 is blueberry healthy leaf. Label 3 is cherry powdery mildew leaf. Label 4 is corn grey spot leaf. Label 5 is grape black rot disease leaf. Label 6 is orange yellow dragon disease leaf. Label 7 is peach bacterial spotted disease leaf. Label 8 is pepper bacterial spotted disease leaf.

We counted the total number of images and different samples, as shown in Table 1. It was found that the number of samples ranged from 152 to 5527, and the distribution was seriously unbalanced. This problem may lead to deviations in the identification and classification of various diseased leaves, which will affect the final recognition accuracy of the model. In order to balance the distribution of different types of images, the data augmentation procedures were conducted:

- (i) Flipping the image from left to right;
- (ii) Flipping the image from top to bottom;
- (iii) Rotating the image by random angle;

(iv) Enlarging the original image randomly. In addition, images of corresponding classes were downloaded from the Internet to supplement the dataset. As for the classes with a large number of images, some original images were retained by eliminating method.

Table 1. Information of the database images

Class	Crop name	Disease name	Images number
0	Apple	apple scab	630
1	Apple	black rot	712
2	Apple	cedar rust	276
3	Apple	—	1835
4	blueberry	—	1735
5	cherry	—	854
6	cherry	powdery mildew	1052
7	Corn	grey leaf spot	1457
8	Corn	common rust	1610
9	Corn	—	2450

The number of images of the training set and test set in the original dataset is 43,424 and 10,681, respectively. After data augmentation, the number of images of each class is about 2000, and the total number of expanded images is 76,000. The expanded dataset is further divided into a training set and test set according to the ratio of 4:1. The training set and test set contain 60,800 and 15,200 images, respectively. Finally, the dataset is labelled as 0-37 by category and the images are resized to 256×256 pixels. Comparison experiment is carried out based on different versions of the dataset provided by PlantVillage project to explore the influence of colour feature and background information on identification results. Fig. 2 shows the different versions of the same leaf for a randomly selected set of leaves.



Fig. 2 Sample images from the two different versions of the Plant Village dataset

I. ARCHITECTURE OF CROP DISEASE IDENTIFICATION MODEL

The traditional CNN architecture mainly includes convolution layers, pooling layers and fully-connected layers. The convolution layer is used to extract features from the images, the pooling layer is used to reduce the amount of data, and the fully-connected layer is used to connect the neurons between the layers. As a classic CNN architecture, the AlexNet model has made a major breakthrough in image recognition tasks. However, the convolution kernel size (11×11) of the first layer is large and the fully-connected layer is used in the network. This leads to problems such as slow iterative training and single feature scale, which make the model difficult to play a role in practical applications. For these limitations, this paper draws on the latest research progress and the original AlexNet made three improvements:

using the global pooling layer instead of the fully-connected layer to improve the iterative training speed;

- (i) adding the inception module to the network to extract features of various scales; and
- (ii) combining different scales of dilated convolution in the inception module to increase the receptive field.

A. INCEPTION MODULE

The 'Inception module' first appeared in the GoogLeNet proposed by Szegedy et al. [[34], [35]] and then there were multiple versions of Inception vN (where N is from 1 to 4 for the number of versions). The CNN-based architecture in this paper is combined with Inception v2. Before the advent of GoogLeNet, experts and scholars had been working to increase the depth and width of the network. Although the network architecture seemed to be more abstract, it caused problems such as too many parameters, slow iteration training and over-fitting. For the sake of reducing the parameters and speeding up the model training while increasing the depth and width of the network, the inception module came into being.

In a nutshell, the inception structure introduces modular

function into the traditional CNN architecture, which is mainly composed of convolutional layers and pooling layers. The sizes of convolution kernels are usually 1×1 , 3×3 and 5×5 . Diverse sizes of convolution kernels are used in the inception module to capture features of different scales, which is quite different from the traditional network architectures. Combining these features together will result in better classification results than using a single-size convolution kernel. Another significant advantage of inception module is the use of 1×1 convolutional layer before and after the convolutional and pooling layers, which effectively reduces computational requirements.

B. ARCHITECTURE OF IMPROVED MODEL

The proposed model architecture is presented in Fig. 4. The structure removes the fully-connected layer in the original AlexNet and replaces it with a global pooling layer. Also, an inception module is connected after the

pooling layer 4 to extract high-dimensional features of different scales. Meanwhile, the dilated convolution combines with Inception module to increase the receptive field of high-level feature maps. The improved model architecture consists of eight convolutional layers (Conv1—Conv5, the inception module uses a parallel combination of 1×1 , 3×3 , and 5×5 convolutional layers), one Concat layer and five pooling layers (pooling 1–global pooling 5, the inception module includes one pooling layer). The output layer uses Softmax classifier to get 38 labels (not shown in Fig. 4).



Fig. 3 Architecture of improved model in this paper

The main relevant parameters of the improved model architecture are summarized in Table 2. The size of the input image is set to $256 \times 256 \times 3$ according to the experience, which can be divided by 2, and computing capability of the computer. The convolution kernel size of Conv1 layer directly affects the extraction of low-level features. The commonly used convolution kernel sizes are 11×11 , 9×9 and 7×7 , and the Conv1 layer convolution kernel size is set to 7×7 in our architecture. The other convolution kernel sizes are shown in Table 2. The sizes of the first layer of convolution kernels in the Inception module are all 1×1 , and the size of the second convolution kernel is 1×1 , 3×3 , 5×5 , respectively. The number of convolution kernels of Conv1, Conv2,. ..., Conv5 are 96, 128, 192, 192, 128, respectively; the number of convolution kernels of the first layer and the second layer in the inception module is 96, 16, 128, 128, 128, 64, respectively.

IV. MODEL TRAINING

Model training is the most important step in the experiment because training the appropriate model can improve the classification accuracy. This paper uses different datasets to train the improved CNN to obtain different models, for the purpose of comparing the classification effects of each model. Although the training sets of each model are different, the experiment mode and hyper-parameter configuration of this paper are standardised to ensure the validity of the experiment. The experimental setup is shown in Table 3, using Caffe as a deep learning framework, and GPU acceleration module is used to improve the training speed.

Proper parameter setting is very important during CNN training. We trained the networks with the Batch size of 64, 128 and 256 in different experiments. Each of the experiment ran for a total of 50 epochs where the epoch was the number of training iterations. The training mechanism used in the model is stochastic gradient descent (SGD), which [[36]] is the same training mechanism used in the original AlexNet. It ensures that the improved CNN framework can be compared with the original AlexNet objectively. This optimisation algorithm is currently widely used in deep networks; it runs faster and converges easily [[37]]. The learning rate was set to 0.01 for all networks. To prevent over-fitting, the weight decay was set to 0.0005, and the learning rate was gradually reduced to 0.1 (gamma) times in stages. Due to the GPU memory constraints, the step size was set to 9000. The loss function used in the model is the softmax-loss function. The loss function used in Caffe is determined by the last layer of the classifier. Since the last layer of the network is softmax classifier (mentioned in Section 3.4), the use of softmax-loss makes it more stable in computing. Other parameters used the default parameters of the Caffe framework.

V. RESULT AND DISCUSSION

Across all our experiments, we used the augmented dataset and three different versions of the whole PlantVillage dataset to train the network. Considering that different parameter settings will affect the performance of the model, this paper also compared the impact of the global pooling type, batch size and other parameters on the accuracy of disease identification, and then optimised the model. Finally, we got an optimal model which was expected to achieve automatic, fast and accurate identification of crop diseases.

As shown in Table 4, the inception module and dilated convolution improve the performance remarkably. Compared with the baseline AlexNet, employing inception module yields a result of 97.93% in AA, which brings 1.52% improvement. It indicates that inception module can extract the disease characteristics at different scales on the surface of diseased leaves. Due to the different sizes of the lesions on the surface of the diseased leaves, the inception module brings benefit to disease identification. Meanwhile, employing dilated convolution individually outperforms the baseline by 1.16%. This shows that the increase of receptive field contributes to disease identification. When we integrate the inception module and dilated convolution together, the performance further improves to 98.44%. Results show that the proposed architecture, combined with inception module and dilated convolution brings benefit to crop disease identification. Based on the above experiments, the subsequent experiments are conducted to obtain a suitable recognition model.

Effects of global pooling type on model performance

Due to different settings, six improved models were obtained in our experiments. Classification accuracies of diverse models are summarised in Table 5. As described in Section 3.1, global pooling includes global average pooling and global maximum pooling. In our experiments, models 1, 3, 5 use global maximum pooling, and models 2, 4, 6 use global average pooling. It can be observed from Table 5 that using global average pooling gets better results than using global maximum pooling. For example, the accuracy of Model 2 is 0.5% higher than that of Model 1, training on the original dataset. We suppose that global average pooling averages the pixel values of the entire feature map. It could make full use of the information and effectively reduce the estimation error caused by the domain constraints. In contrast, the global maximum pooling can only retain the low-level texture feature. Therefore, the deep feature information is lost, which results in lower recognition accuracy. However, we are not sure that the global average pooling is always the best option, since different classification tasks may have various suitable configurations.

Model	Model	Model parameter		Accuracy, %	
number		Global pooling type	Batc h size	Origi nal image datas et	Augment ed image dataset
0	AlexNet model	-	64	96.41	97.62
1	improved model	maximum pooling	64	97.13	98.56
2		average pooling	64	97.63	98.93
3		maximum pooling	128	98.26	99.25
4		average pooling	128	98.45	99.37
5		maximum pooling	256	96.25	97.34
6		average pooling	256	96.47	97.78

Table 5. Classification accuracies with different parameters during diverse experiment

Effects of batch size on model performance

It can be seen from Table 5 that the global average pooling contributes to the performance of models. In order to further optimise the model, the batch size is taken into consideration during the experiments. The batch training method is used to train the network model in experiments, and the training set is divided into several batches. In Table 5, the best accuracy 99.37% is got by the improved model combining with batch size 128 and global average pooling (Model 4). Compared with other models, employing batch size 128 yields better results in AA. Thus, it is supposed that the batch size 128 is quite effective in this work.

In the case of deep learning, if the dataset is enough, the gradient calculated from half of the batch data (even more less) is almost the same as that from all the dataset. Thus, increasing the batch size within a reasonable range might effectively improve the computational efficiency of matrix multiplication and reduce the number of iterations. However, compared with Model 3, employing batch size 256 (Model 5) yields a result of 96.25% in AA, which causes 2.01% loss. Thus, improving batch size blindly may be getting the opposite results. The memory utilisation is increased, yet the memory capacity may not meet the calculation requirements. In addition, as the batch size increases, the number of iterations required to train a full dataset is reduced. To achieve the same accuracy, the time it takes is greatly increased, and the correction of the parameters becomes slow. As a result, the accuracy stops rising if the batch size is increased to a certain extent.



Fig. 4 Result

The later experiment is compared to the experiments with the raw dataset and augmented dataset. The recognition model trained on augmented dataset performs the best, with the highest recognition accuracy of 99.37% and the lowest loss of 0.023. As expected, the performance of the recognition model obtained from training grey-scaled dataset is significantly lower than that of the experiments on the other version of the datasets, but still achieves the accuracy of 96.73%. It indicates that the improved CNN-based model has learnt to extract colour features and the lack of colour features (especially green) in the image would affect the final recognition accuracy. Compared with the model trained on the raw dataset, training on the segmented dataset yields a result of 98.83% in accuracy, which brings 0.38% improvement. We suppose that the model which trained on the segmented dataset can investigate the role of the background of the images in overall performance. Thus, the redundant information and noise introduced during model training might be reduced to increase the final accuracy.

To further verify the convergence effect of the four recognition models, the relationship between accuracy and epochs is illustrated in Fig. 7. It is obvious that after only ten training epochs, all models start to converge. Meanwhile, after 25 training epochs, all models have converged and the test accuracy of the four models can reach over 95%. Besides, it can be found from Fig. 7 that the recognition model based on the segmented dataset and the raw dataset has a convergence speed close to each other, and the final recognition accuracy has little difference. On the one hand, it can be concluded that a small amount of redundant information introduced from the simple background has little effect on the recognition effect of the model. On the other hand, if the background is complex, a better classification model is likely to be obtained by training the images which segment the background.



Fig. 5 Relationship between test accuracy and epoch

Results compared with other studies

Table 7 shows the best results obtained in this study, compared with other studies that previously classified diseases using computer vision and deep learning. The last column shows the classification accuracies when the methods are trained and tested on our dataset with 38 classes. The classification accuracy of this study is higher than that of other studies. Overall, deep learning methods perform better than traditional computer vision methods.

The methods used by Ma et al. [[12]] and Girish and Priti [[38]] both rely on feature extraction, which consumes a lot of time and energy. As our dataset contains various disease images, their methods are therefore not applicable to this dataset. Instead, the methods described by Lu et al. [[25]] and Jayme [[39]] have been implemented and tested against our method using the same dataset. Lu et al. [[25]] designed a typical CNN architecture to identify ten common rice diseases. As shown in Table 7, training their models resulted in a classification accuracy of 96.86%. This accuracy is higher than the original classification accuracy since our dataset is richer than their original dataset. It indicates that rich dataset benefits to identification effect. Jayme [[39]] used transfer learning technology in a pretrained GoogLeNet CNN and achieved identification of plant diseases. By using this approach, an accuracy of 98.24% was achieved on our dataset. These accuracies could be compared to the classification accuracy of 99.37% achieved using our method. In general, the proposed method outperforms these approaches that classified diseases previously.

VI. CONCLUSION

In this paper, CNN is applied to the identification of many crop diseases. Traditional AlexNet model is improved by combining the inception module with dilated convolution. Then the proposed model is trained to identify many crop diseases and is able to effectively classify 26 crop diseases and healthy crops through images recognition. The improvedel could extract disease features of different scales and ease the receptive field of the network compared with the inal AlexNet model, and therefore the final average ognition accuracy reached 99.37%. In addition, the higher convergence rate of the model is obtained.

After only two epochs of training, the accuracy of the improved model is over 95% and the loss value is only 0.023. The contrast experiments based on the segmented dataset and grey-scaled dataset also verify that noise in background and colour features maybe have an impact on the recognition effect, which provides valuable support for further improvement of the recognition method.

The results show that the improved CNN-based model trained on augmented dataset has good adaptability and robustness to image spatial position changes. Our optimal model can identify different diseases of multiple crops. We hope and believe that the research and observations of this paper will provide help for disease identification of crops in actual production. In future work, we are supposed to apply and improve other deep CNN-based architectures and train the model with images of more complex background.

VII. ACKNOWLEDGMENTS

This work was partially supported by the National Natural Science Funds projects (grant no. 31471413), a project funded by the Priority Academic Program Development of Jiangsu Higher Education Institutions (PAPD), Six Talent Peaks Project in Jiangsu Province (ZBZZ-019), Practical Innovation Training Program for College of Jiangsu University.

REFERENCES

- 1. Zhang, Y., Gao, Y., Wang, X.: 'Problems and countermeasures of crop pest control', Agric. Technol., 2017, 37, (1), pp. 117–118 Google Scholar
- 2. Liu, T., Zhong, X., Sun, C. et al.: 'Computer vision-based identification of rice leaf diseases', Agric. Sci. China, 2014, 47, (4), pp. 664–674Google Scholar
- 3. Tan, W., Zhao, C., Wu, H. et al.: 'Hierarchical pathology image recognition based on elastic momentum deep learning neural network', Trans. Chi. Soc. Agric. Mach., 2015, 46, (1), pp. 20–25 Google Scholar
- 4. Sun, J., Tan, W., Mao, H. et al.: 'Recognition of multiple plant leaf diseases based on improved convolutional neural network', Trans. Chin. Soc. Agric. Eng., 2017, 33, (19), pp. 209–215Google Scholar
- 5. Srdjan, S., Marko, A., Andras, A. et al.: 'Deep neural networks based recognition of plant diseases by leaf image classification', Comput. Intel. Neurosci., 2016, 2016, pp. 1–11 Web of Science®Google Scholar
- 6. Mohanty, S.P., Hughes, D.P., Salathé, M.: 'Using deep learning for image-based plant disease detection', Front. Plant. Sci., 2016, 7, p.1419 View Web of Science®Google Scholar
- 7. Johannes, A., Picon, A., Alvarez-Gila, A. et al.: 'Automatic plant disease diagnosis using mobile capture devices, applied on a wheat use case', Comput. Electron. Agric., 2017, 138, pp. 200–209.
- 8. Liu, P., Tu, K., Xu, H. et al.: 'Persimmon surface disease recognition based on support vector machine', Modern Food Sci. Technol., 2011, 27, (3), pp. 349–353 Google Scholar
- 9. Bankar, S., Dube, A., Kadam, P. et al.: 'Plant disease detection techniques using canny edge detection & color
- 10. Ma, Y., Wen, Y., Li, X. et al.: 'Diagnostic system of greenhouse cucumber downy mildew based on image processing', Trans. Chin. Soc. Agric. Mach., 2017, 48, (2), pp. 195–202
- 11. Fukushima, K.: 'Neocognitron: a self-organizing neural network model for a mechanism of pattern recognition unaffected by shift in position', Biol. Cybern., 1980, 36, (4), pp. 193–202
- 12. Lecun, Y., Boser, B., Denker, J.S. et al.: 'Backpropagation applied to handwritten zip code recognition', Neural. Comput., 1989, 1, (4), pp. 541–551
- 13. Gong, D., Cao, C.: 'Plant leaf classification based on convolutional neural network', Comput. Modernization, 2014, 2014, (4), pp. 12–15
- Zhang, S., Huai, Y.: 'Research on plant leaf recognition based on layered convolution depth learning system', J. Beijing Forestry Univ., 2016, 38, (9), pp. 108–115
- 15. Zeiler, M.D., Fergus, R.: 'Visualizing and understanding convolutional networks', Dept. Comput. Sci., New York Univ., 2013, 8689, pp. 818-833
- 16. Goodfellow, I.J., Bulatov, Y., Ibarz, J. et al.: 'Multi-digit number recognition from street view imagery using deep convolutional neural networks' arXiv: 1312. 6082, 2013
- 17. Parkhi, O.M., Vedaldi, A., Zisserman, A.: 'Deep face recognition', British Mach. Vision Conf., 2015, 41, pp. 1–12
- 18. Sun, Y., Liang, D., Wang, X. et al.: 'DeepID3: face recognition with very deep neural networks', Comput. Sci., arXiv: 1502.00873, 2015
- 19. Dai, J., Li, Y., He, K. et al.: 'R-FCN: object detection via region-based fully convolutional networks'. 30th Conf. on Neural Information Processing Systems (NIPS), Barcelona, Spain, 2016
- 20. Chen, L.C., Papandreou, G., Kokkinos, I. et al.: 'Semantic image segmentation with deep convolutional nets and fully connected CRFs', Comput. Sci., 2014, 4, (4), pp. 357–361
- Chen, L.C., Papandreou, G., Kokkinos, I. et al.: 'Deeplab: semantic image segmentation with deep convolutional nets, atrous convolution, and fully connected CRFs', IEEE Trans. Pattern Anal. Mach. Intell., 2018, 40, (4), pp. 834– 848
- 22. [22]Dyrmann, M., Karstoft, H., Midtiby, H.S.: 'Plant species classification using deep convolutional neural network', Biosyst. Eng., 2016, 151, pp. 72–80
- 23. Lu, Y., Yi, S., Zeng, N. et al.: 'Identification of rice diseases using deep convolutional neural networks', Neurocomputing, 2017, 267, pp. 378–384
- 24. Cheng, X., Zhang, Y., Chen, Y. et al.: 'Pest identification via deep residual learning in complex background', Comput.Electron. Agric., 2017, 141, pp. 351–356
- 25. Too, E.C., Yujian, L., Njuki, S. et al.: 'A comparative study of fine-tuning deep learning models for plant disease identification', Comput. Electron. Agric., 2019, 161, pp. 272–279

- 26. Guan, T., Zhu, H.: 'Atrous faster R-CNN for small scale object detection'. Int. Conf. on Multimedia and Image Processing, 2017, pp. 16–21
- Krizhevsky, A., Sutskever, I., Hinton, G.E.: 'Imagenet classification with deep convolutional neural networks'. Int. Conf. on Neural Information Processing Systems, Curran Associates Inc., Lake Tahoe, Nevada, USA., December 2012, pp. 1097–1105
- Wan, L., Zeiler, M.D., Zhang, S. et al.: 'Regularization of neural networks using DropConnect'. Int. Conf. on Machine Learning, Atlanta, GA, USA., June 2013, pp. 1058–1066
- 29. Srivastava, N., Hinton, G., Krizhevsky, A. et al.: 'Dropout: a simple way to prevent neural networks from overfitting', J. Mach. Learn. Res., 2014, 15, (1), pp. 1929–1958
- 30. Lin, M., Chen, Q., Yan, S.: 'Network in network', Comput. Sci., arXiv: 1312.4400, 2013
- 31. Kudo, Y., Aoki, Y.: 'Dilated convolutions for image classification and object localization'. Fifteenth Iapr Int. Conf. on Machine Vision Applications, Nagoya, Japan, May 2017, pp. 452–455
- 32. Szegedy, C., Vanhoucke, V., Ioffe, S. et al.: 'Rethinking the inception architecture for computer vision'. Computer Vision and Pattern Recognition, Las Vegas, Nevada, USA., June 2016, pp. 2818–2826
- 33. Szegedy, C., Ioffe, S., Vanhoucke, V. et al.: ' Inception- v4, inception-ResNet and the impact of residual connections on learning', 2016
- 34. Ketkar, N.: ' Stochastic gradient descent', in Deep Learning with Python (Apress, Berkeley, CA, 2017), pp. 111-130
- 35. He, K., Zhang, X., Ren, S. et al.: ' Identity mappings in deep residual networks', 2016, pp. 1–15
- Girish, A., Priti, B.: 'Potato leaf diseases detection and classification system', Int. J. Comput. Sci. Mob. Comput., 2016, 2, pp. 76–88
- Jayme, G.A.B.: 'Plant disease identification from individual lesions and spots using deep learning', Biosystems Eng., 2019, 180, pp. 97–107

Suspicious Activity Tracking & Detecting Ai Camera

¹Aishwarya.H, ²Namratha.S, ³Md.Bakhar ¹UG Student, Guru Nanak Dev Engineering College, Bidar, Karnataka ^{2,3} Asst. Prof. Guru Nanak Dev Engineering College, Bidar, Karnataka

Abstract— Recognition of suspicious humanoid activity from surveillance video is an energetic research area of image processing and computer visualization. In order to avert violence, theft, accidents, chain snatching, crime, and other suspicious activities, human activity can be supervised visually in sensitive and public spaces like bus stations, railway stations, airports, banks, shopping malls, schools, and colleges, parking lots, roads, etc. Because it is very hard to keep an eye on public spaces all the time, intellectual audiovisual surveillance that can follow human activity in real-time, classify it as normal or exceptional, and send out notifications is necessary. In the previous ten years, a lot of research in the area of anomalous activity recognition in visual surveillance has been published. Furthermore, a number of studies are accessible in the literature to identify different abnormal activities; but none of them dealt with it differently abnormal activity in the review. In this paper, the state-of-art demonstrating the overall improvement in recognizing suspicious activities from surveillance videos in last decade is presented. In broad, we discussed all steps taken to recognize human activity from surveillance videos; such as forefront object abstraction, tracking-based object detection or non-tracing approaches, feature extraction, classification, movement investigation and recognition.

Keywords: Suspicious Activity, AI Camera, Machine Learning, Google Teachable, Recognition

I. INTRODUCTION

IoT platforms have grown into a worldwide behemoth over the past ten years, grabbing hold of every aspect of our everyday lives and improving human life with its inexplicable smart services [1]. IoT is currently confronting greater security concerns than it has in the past because to its easy accessibility and the rapidly growing demand for smart devices and networks. IoT can be secured with readily available security techniques. Conventional methods, however, are less operative when dealing with progression booms and various attack kinds and their severances. Therefore, a next-generation Internet-of-things system requires a robust, continuously enhanced, and modern security system. The suggested work intends to decrease time and effort wasted on video surveillance camera monitoring through the usage of Deep- Learning and Image- Processing [3].

Video Surveillance illustrates a vital role in present world [2]. It is an unique budding areas of Computer visualization and artificial intelligence [6]. An instantaneous, proactive surveillance scheme is desirable, which could instantly sense dubious personnel, recognize suspicious activities, or raise momentous alerts [8]. Insecure security systems provide criminals an advantage during armed robberies in settings like banks and airports. Out of dread, people won't be able to call the police for assistance. One vital task in video investigation is identifying unusual events such as traffic accidents, crimes or illegal activities Such systems require frequent rule-base updates and signature updates, and are not capable of detecting unknown attacks[11]. However, similar crimes can be avoided if a smart camera can capture any suspicious movement and automatically raise an alarm to notify the authorities.

II. RELATED WORK

In [1], the authors proposed a paper on Machine- Learning Based Solutions for Security of Internet of Things (IoT). IoT platforms have developed enormously dominating every aspect of our daily existence by improving human life through its unique suite of intelligent services.

Because of informal ease and reckless-rising demand for smart devices and network, IoT is now facing additional safety challenges than ever beforehand. There are on hand security measures that can be applied to protect IoT. However, conventional techniques are not as competent with the progression booms as well as diverse attack types and their severances. Thus, a robust-vigorously superior and up to date security system is necessary for nextGen IoT system. Machine-Learning (ML) is a prevailing technology that has opened up many promising exploration windows to address current and future challenges in the Internet of Things (IoT). In this paper [1], the design of IoT is discussed; after a widespread literature assessment on ML, the implication of IoT security is approached in terms of numerous types of potential attacks. Additionally, ML-based emerging solutions for IoT security have been presented, and forthcoming challenges are discussed. In [2], the authors has described that, in present world, Video Scrutiny plays a fundamental role. When artificial intelligence, machine-learning and deep-learning inclined into the system, the technologies are advanced too much. Utilizing above blends,

various frameworks are situated which assists to distinguish diverse doubtful performances from the live tracing of footages.

The utmost impulsive one is human behavior and it is very complicated to detect whether it is abnormal or normal. To notice suspicious or normal activity, deep-learning methodology is used in various environments, which guides a vigilant information to the analogous authority, in case of foreseeing a suspicious-activity. Supervision is often performed through successive frames which are extracted from the video. The complete context is separated into dualistic phases. In the main phase, the features are computed from video frames and in next phase, based on the attained features classifier predict the class as suspicious or normal.

In [3], the authors has described that nowadays, video surveillance is used almost everywhere for security. The traditional method of monitoring cameras requires constant human intervention. Using Deep-Learning and Image Processing, the anticipated work aims to eliminate time and effort wasted on monitoring video surveillance cameras. Predicting humanoid behavior is almost impossible. Deep Learning is used to detect suspicious and non-suspicious activity and to warn the user if any suspicious activity is detected. The proposed system strives for the detection of real-world suspicious activities such as burglaries, assaults etc. in surveillance videos.

III. METHODOLOGY

We Make an ML model first using datasets representing common activities like chatting, reading, sitting, etc. Based on the supervised learning model, after being trained on a collection of data, these models are given an algorithm to analyze data, find patterns in feed data, and gain knowledge from the data. These models can be used to forecast the unfamiliar dataset once they have been trained. Then feed datasets of questionable activity—such as fighting, boxing, pointing weapons, or any other suspicious violent movement—into the machine learning algorithm. Additionally, carry out these actions in front of the smart camera to catch different gestures. This would be helpful for building a smart AI camera using Raspberry Camera by training the ML model and installing it on the Raspberry Pi. There are number of adaptable choices available for creating and training ML models, including TensorFlow, Google Teachable, Edge Impulse, Lobe, etc.

IV. HARDWARE & SOFTWARE DESCRIPTION

The proposed block diagram is shown in Fig.1. RASPBERRY PI: Raspberry Pi is a tiny single board computer. By linking peripherals like Keyboard, mouse, display to the Raspberry Pi, it will act as a mini personal computer. Raspberry Pi is extensively used for actual time Image/Video Processing; IoT based applications and Robotics applications. Raspberry Pi is slower than laptop or desktop but is still a computer which can offer all the likely features or abilities, at a low power utilization.

BUZZER:

There are 2 types of buzzers: passive and active. Active buzzers are easier to use and allow us to use them independently, even if we're just using constant DC power.So we will use an active buzzer. With constant DC voltage, it will hum at a predefined frequency of about 2300 Hz. Ideally, the buzzer works at 5V. Since the output voltage of our Raspberry Pi's GPIO pins is only 3.3V, this seems a bit too low for our 5V buzzer. But the buzzer works even at 3.3V. However, at 3.3V, the volume of the sound produced is less powerful. So if we have an NPN transistor we will be able to supply 5V to the buzzer and it is better to use it.

Two different RASPBERRY PI camera boards are available: an 8MP camera and a 12MP High Quality (HQ) camera. The 8MP gadget is also offered without an IR filter in No IR form. The Raspberry Pi no longer sells the original 5MP camera. All Raspberry Pi cameras can capture full HD 1080p video and high-resolution images, and they can all be completely programmed.



Fig.1Proposed Block diagram

V. MACHINE LEARNING MODELLING USING GOOGLE TEACHABLE

A.Data Preparation

Google Teachable Machine: Teachable Machine is an online device that makes building AI models quick, simple, and open to everybody. Workable Machine is helpful - use documents or catch models live. It's a conscious method for taking care of your business. We could decide to utilize it completely on-gadget with next to no webcam or amplifier information leaving your PC. The models we make with Teachable Machine are genuine TensorFlow.js models that work anyplace JavaScript runs, so they play well with instruments like Glitch, P5.js, Node.js, and then some. Besides, we can commodity to various configurations and utilize your models somewhere else like Coral, Arduino, and more The enchanted behind Teachable Machine depends on a famous profound learning strategy called move learning. The majority of the brain network design of the completely prepared model is saved, while a more modest piece of it is supplanted in view of the information. This approach requires less figuring power along with more modest dataset for preparing. Google utilizes probably the best profound learning and brain network models for the Teachable Machine. Preparing Google Teachable is a stage for building models.

B. Training

What can we use to teach it?

Images: We teach the model to classify images using files or your webcam. Sounds: Teach the model to classify sound by recording short sound samples. Poses: Teach the model to classify body poses using files or striking poses in your webcam.

Sample positions in Google Teachable can be formed and many as we requested creating many sample positions and adding them to the data file can increase the accuracy of reading and tracking activity.

How to teach a model using Google Teachable: It Includes gather data, Train the model and product the information base Collect and gathering our models into classes or classifications that we believe the PC should learn. Train the model and afterward quickly test it to check whether it can accurately characterize new models export the data set export our model for our tasks: sites, applications and that's just the beginning.

C.Exporting the Database

Creating & Exporting Pose Model database: New position Project Place a person through the webcam so that the camera can see the person to craft a new image project on Teachable Machine. Train the first class: Fill your first class with first class photos captured from your webcam. Training dataset using normal actions without suspicious activity. Train the second class next, populate your second class with pictures of the second class as well captured from your webcam. Next, press "model train" and that's it! Trained model. Try to test how well your model works for normal activity and suspicious activity. The model will try to classify the miscellaneous actions at the bottom right of the site. If your model doesn't recognize very accurately, try adding more images to the classes that seem weak. Then click "export model" and then "upload my model" to upload to the cloud.

VI. RESULT

Working set up of Suspicious Activity Tracking and Detecting is shown in fig.2.We see that the accuracy of detecting and tracking was seen best i.e. 95% in the Google Teachable platform, the accuracy is also depend on the clarity of the camera and the database of the machine learning pose module that is created On google Teaching model. If there is difference in the class one models and class two Pose models then the Value cross to 1, if the value of class 1 pose models is Zero then it is considered a normal activity. If the value of class 2 is one then it is considered as Suspicious Activity and there is alert Sound.



Fig 2: Working of Suspicious Activity Tracking and Detecting

Identification of Normal Activity:

CASE1: Identification of Normal Activities.

Fig.3 shows identifying Non-suspicious activity as sitting.

- Class-one of the Google Teachable have the Data set of Non-Suspicious Activity Images.
- These Non Suspicious activity include sitting, walking, standing etc..
- Comparing to the Data set of class1images, Here Non-Suspicious activity is deducted.



Fig.3.Identifying Non-suspicious activity as sitting with accuracy

CASE2: Identification of Suspicious activity.



Fig.4. shows identifying the Action of Fighting

- Class- two of the Google Teachable have the Data set of Suspicious Activity Images.
- These Suspicious activity includes Fighting, beating &Weapons like Gun.
- Comparing to the Data set of class 2 images ,Here Suspicious activity is deducted with Accuracy []

VII. CONCLUSION & FUTURE SCOPE

The numerous methods for detecting abandoned objects, theft, falling objects, accidents, unlawful activity, violence, and other events have been covered in this article. In previous decades, a number of researchers suggested cutting-edge methods for detecting objects to lessen erroneous object detection. Numerous researchers have also labored to construct real-time intelligent surveillance systems, but the processing speed of the video frames is not as good as needed, and no scheme has been assembled that has 100% detection accuracy and 0% false detection rate for movies with complicated backgrounds. This AI camera may be utilized for surveillance at numerous public places using the newest technology, such as artificial intelligence and machine learning, to detect and track suspicious actions in real time.

REFERENCES

- 1. Syeda Manjia Tahsiena , Hadis Karimipoura, Petros Spachosa "Machine Learning Based Solutions for Security of Internet of Things (IoT)" computer science arxiv 2020
- 2. Amrutha C.V, C. Jyotsna, Amudha J "Deep Learning Approach for Suspicious Activity Detection from Surveillance Video" 2020 2nd International Conference on Innovative Mechanisms for Industry Applications (ICIMIA)
- 3. Phalguna Kadam, Shweta Gawande, AkshitaThorat, Rohini Mul "Suspicious Activity Detection using Image Processing" Journal of Science and Technology, Vol. 06, Special Issue 01, August 2021, pp114-119
- 4. Teachable Machine Train a computer to recognize your images, sounds, & poseshttps://teachablemachine.withgoogle.com
- 5. Ashwini Kumar www.Electronicforu.com "Suspicious Activity Tracking AI Camera"
- 6. P.BhagyaDivya, S.Shalini, R.Deepa, Baddeli Sravya Reddy, "Inspection of suspicious human activity in the crowd sourced areas captured in surveillance cameras", International Research Journal of Engineering and Technology (IRJET), December 2017.
- 7. Jitendra Musale, Akshata Gavhane, Liyakat Shaikh, Pournima Hagwane, Snehalata Tadge, "Suspicious Movement Detection and Tracking of Human Behavior and Object with Fire Detection using A Closed Circuit TV (CCTV)

cameras ", International Journal for Research in Applied Science & Engineering Technology (IJRASET) Volume 5 Issue XII December 2017.

- 8. U.M.Kamthe,C.G.Patil "Suspicious Activity Recognition in Video Surveillance System", Fourth International Conference on Computing Communication Control and Automation (ICCUBEA), 2018.
- 9. Zahraa Kain, AbirYouness, Ismail El Sayad, Samih Abdul-Nabi, Hussein Kassem, "Detecting Abnormal Events in University Areas", International conference on Computer and Application, 2018.
- 10. Tian Wanga, Meina Qia, Yingjun Deng, Yi Zhouc, Huan Wangd, Qi Lyua, HichemSnoussie, "Abnormal event detection based on analysis of movement information of video sequence", Article-Optik, vol 152, January-2018.
- 11. Tejashri Subhash Bora, Monika Dhananjay Rokade "Methodology For Human Suspicious Activity Detection" IJASRET,vol.6,May2021.
- 12. Dinesh Jackson Samuel R, Fenil E, Gunasekaran Manogaran, Vivekananda G.N, Thanjaivadivel T, Jeeva S, Ahilan A, "Real time violence detection framework for football stadium comprising of big data analysis and deep learning through bidirectional LSTM", The International Journal of Computer and Telecommunications Networking, 2019.
- 13. Kwang-Eun Ko, Kwee-Bo Sim"Deep convolutional framework for abnormal behaviour detection in a smart surveillance system."Engineering Applications of Artificial Intelligence ,67 (2018).
- 14. Yuke Li "A Deep Spatiotemporal Perspective for Understanding Crowd Behavior", IEEE Transactions on multimedia, Vol. 20, NO. 12, December 2018.
- 15. Javier Abellan-Abenza, Alberto Garcia-Garcia, Sergiu Oprea, David Ivorra-Piqueres, Jose Garcia-Rodriguez "Classifying Behaviours in Videos with Recurrent Neural Networks", International Journal of Computer Vision and Image Processing, December 2017.

Visualizing and Forecasting Stocks

¹Miss.Smita Gawade, ²Sunil Kadam, ³Yash Bagal, ⁴Adesh Kulkarni ¹Assistant Professor SVERI's College of Engineering, Pandharpur ²UG Student, SVERI's College of Engineering, Pandharpur

Abstract— The "Review and Forecast" project aims to create a software application to analyze historical data, identify trends and predict future market prices. This project includes data collection, forecasting, visualization and forecasting, providing users with the right tools to understand stock market dynamics. Reliable historical data originally from platforms like Yahoo Finance and Alpha Vantage. This data is then cleaned and prioritized to resolve important and negative situations. The cleaned data is converted into a suitable format for analysis. The main function of the application is interactive visualization created using libraries such as Matplotlib, Seaborn and Plotly. These views include historical price charts, indicators such as moving averages and Bollinger Bands, and market forecasts. Users can interact with the view by viewing zoom, pan, and instructions.

Keywords: Review and forecast, software application, historical data analysis, trend identification, future market price predication, data collection, visualization, stock market dynamics.

I. INTRODUCTION

The change in product demand is huge. But as technology advances, so does the way to make a steady profit from desired products and it also helps experts find the best tips to inject better reaction. Determining the asking price is important to help you get the most out of your product selection while keeping risk low. Interval Neural Networks (RNN) have proven to be one of the most important models for continuous data processing. Short-term memory is one of the most efficient RNN infrastructures. LSTM refers to the memory of the brain, a computer that replaces traditional artificial intelligence in the remaining process of the network. With memory cells, the network can well integrate memory and ideas in the remote area over time, thus being suitable for processing the structure of data over time and high preservation capacity. The third paper presents the model and predicts NIFTY 50 stock returns using LSTM. We collected data of 5x NIFTY 50 and used it for training and implementation purposes of the model. The next part of this article will be the process where we will explain each process in detail. We will then graphically represent the analysis we used and also consider the possible outcomes.

II. METHODOLOGY

Data collection and preparation

The first step in the process is data collection and preparation. Historical transaction data, including price and trading volume data, can be collected from trusted sources such as financial data providers or APIs. Data collection will be performed subsequently, including data cleaning, design, and handling of missing values or results. This step ensures that the data is in the appropriate format for analysis and prediction.



Fig 1 Data Preprocessing Feature

Construction and selection

After preliminary information, the next step is construction and selection. Relevant features can be extracted or extracted from the raw data to increase the predictive power of the predictive model. Indicators such as moving averages, relative strength index (RSI) and Bollinger Bands can be calculated based on price and volume data. A special selection process will be applied. Specifies the most important and useful data for forecasting models. Model Selection and Training Model will include the selection and training of a suitable model. Various models can be considered, such as autoregressive integrated moving average (ARIMA), exponential smoothing (ES), or machine learning algorithms such as random forests, support vector machines (SVM), or neural networks. The choice of model will depend on its suitability for capturing key patterns and relationships in stock data. The selected model is trained using historical data and its parameters are optimized using methods such as grid search or cross-validation. Visualization TechniquesThe visualization method involves creating data reports and visualizing charts, graphs, and dashboards to represent data and analysis. Visualizations can include bar charts, graphs, or area charts to show price movements over time. Additionally, other visual points such as volume, moving averages, and trend lines can be combined to provide a comprehensive view of stock data. Interactive features or guides may also be used to allow users to explore information and gain a deeper understanding. Charts are a simple but effective visual tool for showing the price history of a stock. It links the closing price all at once, allowing users to track the total price. Charts are especially important for analyzing long-term trends and evaluating stock performance over time. The combination of these visualization techniques enables understanding and interpretation of market data. This technology allows users to identify patterns, trends, disparities, and relationships, allowing them to make investment decisions. Visualizations can include bar charts, graphs, or area charts to show price movements over time. Additionally, other visual elements such as volume, moving average and trend lines can be combined to provide a comprehensive view of stock data. Interactive features or guides may also be used to allow users to explore information and gain a deeper understanding. Charts are a simple but effective visual tool that shows historical price trends of a stock.

Machine Learning Based Predictive Models

The solution strategy includes machine learning based predictive models that are unique in business analytics. By leveraging the power of machine learning algorithms, solutions can learn from historical data, identify complex patterns, and make accurate predictions about future products. This new approach goes beyond traditional forecasting methods to achieve more accurate and reliable results. Learning models continually adjust and improve their performance over time, providing users with new and predictable changes to make better decisions.

Real-Time Data Integration

Another new way to solve problems is the instant integration of data. The solution allows users to access and analyze data in real time, enabling timely decision making and effective investment strategies. Real-time data integration and APIs give users access to new data, allowing them to quickly respond to changes in business. This unique feature provides users with a competitive advantage in an environment where timely information is critical to investment decisions.

Interactive Trading Simulation

The solution strategy has interactive trading simulation as a special feature. Users can simulate trading scenarios based on historical data and evaluate the performance of different investment strategies. This innovation allows users to better understand the consequences of their investment decisions by allowing them to test and adjust their business strategies in a risk-free environment. Interactive trading simulation provides a practical, hands-on experience for learning and testing, allowing users to improve their trading skills and confidence.

Customization and Personalization

Solutions mean customization and personalization based on specific features. Users can customize views, metrics, and forecast models to meet their specific needs and preferences. The solution provides flexible options for selecting and configuring indicators, allowing users to create analytical and decision-making tools. This update enables users to customize the solution according to their investment strategy and business, improving their experience and efficiency.

User-friendly interface and intuitive user experience

User-friendly interface and intuitive user experience are important innovative aspects of the solution. This solution, which is important in terms of simplicity and ease of use, enables it to reach a wide range of users, including those without expertise. Intuitive design and navigation allow users to easily search and interact with views and tools. This feature allows users to use the solution effectively, eliminating the need for extensive training or expertise. In summary, solutions for visualizing and estimating products stand out due to the integration of many new and unique features. The integration of advanced visual technologies provides users with an interactive and intuitive interface to search and analyze business data. Using advanced

technology, the solution provides informative charts, graphs and dashboards that help identify patterns, trends and anomalies. Additionally, the integration of predictive models based on machine learning differentiates this solution from traditional approaches. These models continue to learn from historical data to make more accurate decisions. With its innovative features, the solution allows users to better understand the stock market and make informed investment decisions. Additionally, although LSTM models are promising in stock price predictions, their complexity and lack of interpretation pose significant limitations. The black box nature of LSTM models hinders our ability to understand the logic behind their predictions. To close this gap, future research should focus on improving the interpretation and interpretation of LSTM models in the context of cost estimation. By developing strategies to present visual content and interpret learning patterns, we can close the gap between LSTM prediction and investment decision making. Additionally, the integration of other factors into the LSTM model should also be investigated. By integrating a variety of data such as economic indicators, news, geopolitical events and specific company news, we can increase the accuracy and efficiency of our estimated costs. Investigating how LSTM models can be used by these outliers will pave the way for more powerful predictive models. The visionary concept and solution represents the integration of many new and unique features. The integration of advanced visual technologies provides users with an interactive and intuitive interface to search and analyze business data. Using advanced technology, the solution provides informative charts, graphs and dashboards that help identify patterns, trends and anomalies. Additionally, the integration of learning-based prediction models makes this solution different from traditional methods. These models continue to learn from historical data to make more accurate decisions.Deep learning is becoming a trend as financial predictions continue. Among these technologies, the long distance model (LSTM) is a variant of the recurrent neural network (RNN) that has become popular in the last few years. LSTM overcomes the gradient vanishing problem found in traditional RNNs and is good at capturing long-term dependencies in data series. This makes LSTM particularly suitable for stock price prediction. LSTM models can capture patterns in connected data by combining storage and gateways that store or discard data over time. The ability to recall key information from early stages is critical to successful product cost estimation. In the next section, we will delve deeper into LSTM concepts and applications in the context of business analysis. In recent years, the use of deep learning techniques for financial forecasting has gained significant attention. One such technique is the Long Short- Term Memory (LSTM) model, a type of recurrent neural network (RNN) known for its ability to capture longterm dependencies in time series data.

Understanding LSTM

LSTM is a type of RNN that solves the vanishing problem commonly encountered in traditional RNN. The LSTM model is particularly useful for capturing patterns in continuous data by combining cells and gates that can selectively store or discard data over time. These storage units allow the model to remember important information from previous steps, which is important for predicting the correct stock price. Data preparation: The first step in developing an LSTM model for stock price prediction is data preparation. Collect historical price data, which typically includes features such as opening price, closing price, trading volume, and other metrics. This data is preprocessed by normalizing or standardizing features to ensure scale in stock.

III.MODEL ARCHITECTURE

LSTM models consist of multiple LSTM layers and one or more thickness layers for prediction. The input of the model is the historical data of the stock price, and the output is the price prediction for the next step. The LSTM layer learns to extract relevant patterns and dependencies in the components, while the density layer provides the final prediction. The LSTM (long-range long) model is a recurrent neural network (RNN) architecture designed to overcome the limitations of traditional RNNs in capturing long-lived populations in the network. Natural language processing, speech recognition, time series analysis, etc. It is widely used in many tasks such as.



Fig. 2 Graph Visualizing stock prices with respect to time.

Forecasting and evaluation:

The learned forecasting model will be used to predict future market prices or trends. The model will be applied to invisible data or test data to evaluate its effectiveness and accuracy. Metrics such as mean error (MAE), root mean square error (RMSE), or precision measurement can be used to compare and evaluate the performance of forecast models. This method allows iterating and improving models and visualizations based on evaluations. If the prediction model or visualization does not meet the desired accuracy or efficiency, it can be modified or improved. This iterative process may involve fine- tuning the model parameters, exploring alternative models, or enhancing the visualizations to provide clearer insights.

Algorithms and tools used:

The success of the project in visualizing and forecasting stocks relies on the effective utilization of r elevant algorithms, techniques, and tools.

Forecasting algorithm:



Fig 3 Forecasting Algorithm

LSTM (Long Short Term Memory) is a deep learning algorithm belonging to the Recurrent Neural Network (RNN) category. It is designed to be effective and model long-term expectations or trends in connected products. The main advantage of LSTM is its ability to solve the problem of incompleteness, which is a common problem in training deep neural networks. The vanishing problem means that when gradients are repeated across layers, their effects weaken, making it difficult for the network to learn its dependency on the system over long periods of time. LSTM solves this problem by introducing a storage and control algorithm. Memory is the main component of the LSTM unit. It collects data over time and selectively decides what to keep and what to discard. The unit has an integrated recycling feature that allows information from previous steps to be preserved. It can be thought of as a conveyor belt where information flows in and out and is modified or controlled based on the information it carries. LSTM has three types of gates to control the flow of information in the unit: input gates, forget gates, and exit gates. Each gate is a layered neural network that uses a sigmoid and convolutional algorithm to process data. Using this gating technique, LSTM networks can selectively retain or discard long-term data, allowing them to capture and display relevant information over time. This makes LSTM especially useful for tasks with complex components, such as natural language processing, speech recognition, and real-time processing.

Training and evaluation

Consolidate your data by cleaning, tokenizing, and coding it into a model for your version of LSTM. This will include removing uppercase and lowercase letters, splitting the row, and drawing parentheses for index numbers. Divide the data set into study, validation and analysis units. The training method is used to train the LSTM model, the validation method is used to tune hyperparameters and demonstrate the overall performance, and the testing method is used to evaluate the final model. Create a version of LSTM using a deep learning framework such as TensorFlow or PyTorch. You can specify the number of LSTM layers, the number of hidden objects in the layer, and the input form. Feed the training data into the LSTM version and repeat the training set for different time periods. These forms make predictions at any given time, calculate losses and adjust weights used in recovery. Tweak as necessary to understand the value and other hyperparameters. Check the performance of the version of the application installed at exit at any time. Collect metrics that include precision, loss, accuracy, recovery, or F1 metrics to measure how well the model is learning.

Hyperparameter Tuning

The overall performance of the LSTM version depends on many hyperparameters, including the number of LSTM layers, the number of units in a layer, the training cost, and the batch size. These hyperparameters can affect the accuracy and reliability of the model. Therefore, it is very important to perform hyperparameter tuning to find the most reliable parameter that produces good prediction results.

IV.RESULTS AND DISCUSSION

Once the LSTM model is well known and refined, it can be used to estimate the inventory value of unobserved data. Prediction accuracy can be measured using a variety of metrics, including mean error (MAE), standard deviation of error (RMSE), and standard deviation percent error (MAPE). Also, look at the estimated and actual stock prices on the chart to better understand the effectiveness of the release. The results obtained in the project "Visualizing and Forecasting Stocks Using LSTM" demonstrate the effectiveness of the proposed method in capturing stock dynamics and making accurate forecasts. A combination of visual and LSTM versions has proven useful in analyzing trades, trends and potential trading opportunities.

Many types of neural networks can be created by combining various factors such as network topology and learning methods. In this experiment we consider neural connections and long-term memory. At this stage, we can talk about the technology of our device.

Our site has different levels as follows:- Level 1: Raw Data:

In this level, historical data is collected from https://www.quandl.com/information/ NSE, this historical fact Future stock Used to predict prices.

Level 2: Fact Preprocessing:

The preprocessing stage should include: a) Data discretization: some of the facts are reduced, but are especially important for statistics.

b) Data transformation: standardization.

c) Actual Cleanup: List missing values.

d) Real integration: Integrate data files.

After the data set is converted into simple data, the data set will be divided into teaching and evaluation for evaluation. Here the current rate on the course fee will be higher. Try to keep statistics on 5-10% of the total data set.

Stage 3: Feature Extraction:

In this stage, select only the features that can be fed into the neural network. We will select Day, Open, High, Low, Close and More.

Stage 4: Neural Network Learning:

In this stage, data is fed into a neural network and trained to predict the distribution of random numbers and heavy numbers. Our LSTM model consists of a communication layer (displayed by 2 LSTM layers), a dense layer with ReLU activation, and a density output layer with a linear activation function. The Neural ensemble rule used in Keras is as follows: Fifth stage: output generation:

In this stage, the output value produced from the output layer of the RNN is compared with the target value. The error or difference between the target and the generated value is minimized using the return exposure method, which adjusts the weight and bias of the ensemble.

Evaluation:

We use root mean square error (RMSE) for machine readout. By using RMSE estimates, the error or difference between the target and the results can be reduced. RMSE is the square root of the implied/mean value of each error squared. The use of RMSE is not uncommon and provides the best possible margin of error for numerical estimation. RMSE reinforces and severely penalizes the large compared to the equivalent implicit error.





Experimental Work:

• Dataset description: We acquired the records from https://www.quandl.com. we have amassed the historical stock facts of NIFTY 50 from the national stock alternate. we have collected a day-by-day dataset and saved a window length of 22 days. Facts degrees from 01.01.2011 to 31.12.2016.

• Collection records: We were given 1312 sequences from 01.01.2011 to 31.12.2016. From these facts set we used 1180 samples for schooling cause and 132 samples for validation motive.

• Education detail: For schooling the version we used RMSprop as the optimizer and normalized every vector of the sequence. We use Google Cloud Engine as the training platform [Machine type: n1-standard-2 (2 vCPU, 7.5 GB memory), CPU platform: Intel Ivy Bridge] and Ubuntu Sixteen.04, Keras (front-end) and TensorFlow (backend) we use end)) because of the learning environment. For experimental use, we use different methods with different time intervals to measure the RMSE of training and testing data.

Experimental Result:

Parameters	No.Of Epochs	Education RMSE	Education RMSE
Open/close	250	0.01491	0.01358
Open/close	500	0.01027	0.00918
High/low/close	250	0.01511	0.014
High/low/close	500	0.01133	0.01059
High/low/close/o pen	250	0.0133	0.01236

Once the LSTM model is trained and updated, it can be used to estimate the value of unknowns. The accuracy of the forecast can be estimated using a variety of methods, including mean absolute error (MAE), root mean square error (RMSE), and mean absolute error (MAP). Additionally, presenting estimated costs and actual market costs on a graph provides insight into the model's performance. The results obtained from the design of "Visualization and Inoculation of Inventory Using LSTM" show the effectiveness of the proposed method in extracting the required inventory and making the correct forecast. The combination of visualization and LSTM modeling has proven to be very useful in demand interaction, conversion and open market intervention.



Fig 5 Prediction of Reliance stock using LSTM

V. CONCLUSION:

In conclusion, the "Forecasting and Weather" project provides a good insight into the field of business analysis and forecasting. In one of these papers, we explored a variety of techniques and tools to visualize inventory data, learn historical characteristics, and create predictive models to help people make purchasing decisions. One of the main findings of the visit is the importance of visual analysis in sharing information. By seeing past price movements, volumes and different indicators, we can clearly understand the market and trends. This insight not only improves our language, but also allows us to convey statistics accurately. Additionally, the company is trying to develop a forecasting model to predict future stock prices. We explore extraordinary techniques such as real- time analysis, machine learning algorithms and deep learning models. These methods have proven effective in capturing underlying patterns and making accurate predictions

REFERENCES

- Roondiwala, M., Patel, H., & Varma, S. (2015). Predicting Stock Prices Using LSTM. *International Journal of Science and Research, 10*(1), 234-245.
- 2. Hefny, A., Ali, A., Fikry, H., & Khedr, A. (2016). Stock Price Forecasting Using Artificial Neural Networks. In *Proceedings of the International Conference on Computer Engineering & Systems (ICCES)*, 123-136.
- Zhang, Z., Li, Q., & Yang, J. (2018). Stock Market Forecasting Using Recurrent Neural Networks. In *Proceedings of the International Conference on Cloud Computing and Big Data Analysis (ICCCBDA)*, 45-58. Zhou, P., Liu, S., Lu, Y., & Ma, Z. (2016).
- 4. Visual Analytics for Stock Market Prediction: A Survey. *IEEE Transactions on Visualization and Computer Graphics, 22*(1), 112-125.
- 5. Jiang, W., Han, Y., Liu, W., & Sun, R. (2017). Stock Price Prediction Using Deep Learn ing. In*Proceedings of the International Conference on Machine Learning and Cybernetics (ICMLC)*, 78-92.
- Sharma, A., Jadhav, P. D., & Borade, A. P. (2017). A Comparative Study of Stock Market Prediction Using Machine Learning Techniques. *International Journal of Advanced Research in Computer Science and Software Engineering (IJARCSSE), 7*(4), 321-334.
- Tang, M., Jiao, B., Huang, Y., & Huai, J.(2017). Visual Analysis of Financial Time Series with Deep Learning Models. In*Proceedings of the International Conference on Computer Supported Cooperative Work in Design (CSCWD)*, 213-227.
- 8. Agarwal, K. K., Lal, K., & Rattan, M. (2017). Stock Market Prediction using Hidden Markov Models. In *Proceedings of the International Conference on Computer Communication and Informatics (ICCCI)*, 56-69.
- Nallamothu, V. K., & Yadavalli, S. (2019). Deep Learning for Stock Market Prediction Using Multiple Financial Indicators. In *Proceedings of the International Conference on Computing, Communication and Automation (ICCCA)*, 102-115.
- 10. Gondhalekar, A., & Ramachandra, R. (2018). Predicting Stock Market Trends Using Fusion of News and Social Media Sentiment. In *Proceedings of the International Conference on BigData Analytics (BDA)*, 78-91.
- 11. Nguyen, H., & Ramachandra, R. (2019). Ensemble Methods for Stock Price Prediction: A Survey. In *Proceedings of the International Conference on Machine Learning and Data Science (ICMLDS)*, 143-156.
- 12. Huai, J., Jiao, B., Tang, M., & Chen, Z. (2016). Visual Analytics for Financial Time Series Data: A Survey. *IEEE Transactions on Visualization and Computer Graphics, 22*(1), 45-58.

Directivity Enhancement Of Multilayer Direct Feed Hexagonal Micro Strip Patch Antenna Array For Wireless Applications

¹Anil K. Rathod, ²Dr. Md. Bakhar, ³Dr. S. R. Chougule, ⁴Amol C. Bhosale
¹Research Scholar, Department of ECE, GNDEC, Bidar and VTU, Belagavi, India
²Professor, Department of ECE, GNDEC, Bidar and VTU, Belagavi, India
³Profrssor, Department of E&TC, KITs COE, Kolhapur, Maharashtra, India
⁴Research scholar, ECE Department, BKIT Bhalki VTU, Karnataka, India

Abstract— In this research paper, Directivity enhancement is investigated in Multilayer direct feed Hexagonal Micro strip patch antennas array. Performance analysis of Multilayer Direct feed Hexagonal Micro Strip Patch Antenna Array is done to improve its parameters like Bandwidth, and Directivity. Comparing the proposed Multilayer direct feed Hexagonal Micro strip patch antenna array to conventional alternatives, it demonstrates enhanced bandwidth of 180 MHz and directivity of 7.737 dB. The proposed antenna design models have a functional center frequency ranging from 2.400 GHz to 2.58 GHz, suitable for wireless applications. The Return Loss and VSWR have been examined within an acceptable range. The electric field and current flow of the antenna are contained inside the intended radiating area. The utilization of the probe feed technique in the design model of a multilayer direct feed hexagonal microstrip patch antenna array has been found to improve the bandwidth, directivity, and other parameters of the antenna compared to traditional antennas.

Keywords: Hexagonal, Bandwidth, Gain, Directivity, Multilayer, Direct feed technique.

I. INTRODUCTION

In the present era, antennas play a crucial role in every wireless communication system. Antennas are of utmost importance in communication due to the significant role they play, especially in light of technological improvements [1-2]. Microstrip antennas (MSAs) possess various advantages, including their lightweight and small-volume nature, as well as their ability to be made conformal to any sort of host surface. In addition, PCB Technology can be utilized for the physical fabrication of Microstrip Patch Antennas, enabling cost-effective mass production [1]. The use of Micro Strip Patch Antenna in defense and commercial applications is being considered as a viable alternative to conventional antennas. Nevertheless, it has been stated that the MSA is experiencing a limitation in its bandwidth capacity. Expanding the bandwidth of the MSA has emerged as a fundamental necessity in the realm of research. This is evident in a substantial volume of scholarly articles on the topic that have been published in academic journals and conference proceedings [1]. Various broadband MSA design models have been examined in recent decades. Both the coaxial and probe feed approaches are applicable in various design methodologies. In order to achieve impedance matching, it is necessary to solder the patch and center conductor of the coaxial connector. The probe feeding approach offers the advantage of being able to place the feed point at any desired location to achieve input impedance matching [1, 10]. A novel microstrip antenna with a Ψ -shaped configuration is described in this paper. The antenna's bandwidth is enhanced by transforming it from a rectangular shape to an E-shape and then to a Ψ -shape [3]. The investigation focused on achieving wide bandwidth by utilizing the stacking of two substrates and slotted patches [4]. The usage of Defected Ground Structure (DGS) as a strategy to enhance the gain and bandwidth of the ground plane has been described [5]. The study presents the first-ever use of multilayer stretchable conductors (SCs) as a cost-effective solution for creating conformal and reconfigurable antennas on flexible substrates [6]. Researchers have utilized micromachining processes to design patch antennas on single, double, and multilayer dielectric substrates [7]. A small patch antenna is suggested for use in satellite applications operating in the C-band, X-band, and Ku-band frequencies. The multiband antenna that has been built is suitable for satellite applications [8-9]. The design of a fractal antenna is optimized for the 2.4 GHz frequency band, making it well-suited for Wireless Local Area Network (WLAN) applications. The antenna is constructed using FR-4 dielectric substrate material [10]. To improve the bandwidth and Directivity of Micro Strip patch antenna deferent methodologies investigated and anyone technique can be adopted.

II.PROPOSED ANTENNA DESIGN MODEL

Figure 1 shows the proposed Multilayer Direct feed Hexagonal MSA Array with probe feeding. The design model comprises a copper ground plane, a radiating patch, an air gap, and a FR-4 dielectric substrate. The probe feed technique has been employed to achieve impedance matching with the radiating element.



Fig. 1 Multilayer Direct feed Hexagonal MSA Array

The air gap functions as a dielectric substrate with a dielectric constant of one. The proposed model now comprises a Multilayer structure, including a copper ground plane, an air gap with a height of 6 mm, a dielectric substrate with a readily available height of 1.6 mm, and a Hexagonal radiating patch [11-12]. The permittivity can be selected near to 1 to obtain a wide bandwidth.

III.SIMULATION RESULTS OF THE PROPOSED DESIGN MODEL



3.1 Bandwidth And Return Loss Of Multilayer

Fig. 2 Bandwidth and Return loss of Multilayer Direct feed Hexagonal MSA Array

3.2 VSWR of Multilayer Direct feed Hexagonal MSA Array



Fig 3 VSWR of Multilayer Direct feed Hexagonal MSA Array

3.3 Directivity of Multilayer Direct feed Hexagonal MSA Array



Fig 4 Directivity of Multilayer Direct feed Hexagonal MSA Array

3.4 GAIN OF MULTILAYER DIRECT FEED HEXAGONAL MSA ARRA



Fig 5 Gain of Multilayer Direct feed Hexagonal MSA Array

3.5 Radiation Pattern of Multilayer Direct feed Hexagonal MSA Array



Fig 6. Radiation Pattern of Multilayer Direct feed Hexagonal MSA Array

3.6 E-Field of Multilayer Direct feed Hexagonal MSA Array



Fig 7 E-Field of Multilayer Direct feed Hexagonal MSA Array

IV. RESULT ANALYSIS TABLE OF ULTILAYER DIRECT FEED HEXAGONAL MSA ARRAY

Table No.1 consists of result analysis table for Multilayer Direct feed Hexagonal MSA Array with performance Parameters like Operating Frequency in GHz, Bandwidth in MHz, Return Loss in dB, VSWR, Directivity in dB and Total Gain in dB as shown below:

Table No.1 Result analysis table for Multilayer Direct feed Hexagonal MSA Array.

Parameters	Multilayer Direct feed Hexagonal MSA Array
Operating Frequency (GHz)	2.400-2.580
Bandwidth (MHz)	180
Return Loss(dB)	-19.073
VSWR	1.255
Directivity (dB)	7.737
Total Gain (dB)	4.383

The Bandwidth can be defined as a reappearance extends over which VSWR is always under two (which relates to an arrival loss of 9.5 dB or 11% reflected force). The VSWR or impedance BW of the MSA is characterized as the recurrence extend over which it is coordinated with that of the feed line inside determined cut off points.

The Bandwidth of proposed Multilayer Direct feed Hexagonal MSA Array is 180 MHz, which is within the specified limits as shown in figure 2.

Return Loss has been observed in an acceptable range. VSWR can be defined in terms of the input reflection coefficient (Γ) as:

$$VSWR = \frac{1 + |\Gamma|}{1 - |\Gamma|} - \dots$$
 (1)

The VSWR (Voltage Standing Wave Ratio) of the Multilayer Direct feed Hexagonal MSA (Microstrip Antenna) Array is below 2, as depicted in figure 3. Figure 4 illustrates the Directivity of the Multilayer Direct feed Hexagonal MSA Array, while Figure 5 depicts its Gain. Figure 6 displays the radiation pattern of the antenna that has been suggested. The electric field of the multilayer direct feed hexagonal MSA array is depicted in Figure 7. All of the suggested antenna parameters have been observed within the acceptable range.

V.CONCLUSION

In this paper, Bandwidth and Directivity improvement is investigated in Multilayer Direct feed Hexagonal Micro Strip Antenna Array by using suspended and probe feed techniques. The design model of Multilayer Direct feed Hexagonal MSA Array is operating at 2.486 GHz centre frequency. The Multilayer Direct feed Hexagonal Micro Strip Antenna Array, as compared to a regular antenna, exhibits an enhanced bandwidth of 180 MHz and a directivity of 7.737 dB. The proposed developed model has a return loss of -19 dB, which is below the threshold of -10 dB. The Voltage Standing Wave Ratio (VSWR) has been measured to be 1.255, which falls within the allowed limit. The design is modeled using FR-4 dielectric substrate material, which has a relative permittivity of 4.4. It is more cost-effective compared to other materials. The Multilayer Direct feed Hexagonal Micro Strip Antenna Array is a superior choice for wireless applications.

REFERENCES

- 1. Girish Kumar, K. P. Ray "Broadband Micro Strip Antennas", Artech House Boston, London, 2003
- 2. Moh'd Ghaith Al-Halawani, Mohammed Al-Najjar and Mohamed K. Abdelazeez, "Microstrip Antenna Design For UWB Applications", 978-1-5090-2886-3/16/\$31.00 ©2016 IEEE
- 3. S Amit A. Deshmukh and K. P. Ray, "Analysis of Broadband Psi (Ψ)-Shaped Micro strip Antennas", IEEE Antennas and Propagation Magazine, Vol. 55, No. 2, April 2013.
- 4. Khushboo Tiwari and Dhaval Pujara, "Multilayer Slotted Microstrip Antenna for Wi-Fi Application", 978-1-4799-7815-1/15/\$31.00 ©2015 IEEE.
- Amar B.Kachare, Mahesh S.Mathpati and Anil K.Rathod, "Effect of Size and Location of the DGS on Characteristics of Rectangular Micro Strip Patch Antenna", International Journal of Computer Applications (0975 – 8887), Conference on Emerging Applications of Electronics System, Signal Processing and Computing Technologies (NCESC 2015).
- Riaz Ahmed Liyakath, Arash Takshi, and Gokhan Mumcu, "Multilayer Stretchable Conductors on Polymer Substrates for Conformal and Reconfigurable Antennas", IEEE Antennas and Wireless Propagation Letters, Vol.12, 2013
- Rajesh Saha, Santanu Maity and Nripendra Trigunayat, "Enhancement of Gain, Bandwidth and Directivity of a Patch Antenna by Increasing Dielectric Layers of the Substrate through Micromachining Technique for RFID Application", 2015 International Conference on Advances in Computer Engineering and Applications (ICACEA)IMS Engineering College, Ghaziabad, India, 978-1-4673-6911-4/15/\$31.00 ©2015 IEEE
- 8. L. Nageswara Rao and V. Santhosh Kumar, "A Novel Multiband Micro strip Patch Antenna for Satellite Applications", International Journal of Advanced Science and Technology, Vol. 29, No. 3s, (2020), pp. 177-183, ISSN: 2005-4238 IJAST
- M.S. Mathapti and A.K. Rathod, "Design, Simulation and Fabrication of Multilayer Structured PsiShape Microstrip A M.S. Mathapti and A.K. Rathod, "Design, Simulation and Fabrication of Multilayer Structured PsiShape Microstrip Antenna for Wireless Applications", International Journal on Emerging Technologies (Special Issue on NCRIET-2015) 6(2): 243-248(2015), ISSN No. (Print): 0975-8364.
- Kotaibah Mohammed Ali, Chandrasekharan Nataraj et.all, "Bandwidth Analysis of Microstrip Antenna With Fractal Design for 2.4 GHZ Wireless Applications", International Journal of Advanced Science and Technology, Vol. 29, No. 1, (2020), pp. 1344 – 1358, ISSN: 2005-4238 IJAST.
- 11. Y. S. Faouri, N. M. Awad, and M. K. Abdelazeez, "Hexagonal patch antenna with triple band rejections," 2019 IEEE Jordan Int. Jt. Conf. Electr. Eng. Inf. Technol. JEEIT 2019 Proc., pp. 446–448, 2019.

12. S. Hong, Y. Kim, and C. Won Jung, "Transparent Microstrip Patch Antennas with Multilayer and Metal-Mesh Films," IEEE Antennas Wirel. Propag. Letter vol. 16, no. c, pp. 772–775, 2017.

Author information



Anil K. Rathod received B.E (E&TC) and M.E (E&TC) in 2012 and 2015, respectively. He is currently pursuing a Ph.D. at Research Scholar, Department of ECE,Guru Nanak Dev Engineering College, Bidar and Visvesveraya Technological University,Belagavi,India.e-mail: rathod.anil102@gmail.com



Dr. Mohammed Bakhar is a Professor in the E&CE Dept. of Guru Nanak Dev Engg. College, Bidar Karnataka, India. He completed M.E(Communication Systems) in 1998 & Ph.D. in Microwave Electronics 2013 from Gulbarga University, Kalaburagi, Karnataka, India. He started his teaching profession in 1995and has 25 years of teaching experience in the field.e-mail: mohammed.bakhar@gmail.com



Dr. S. R. Chougule is a Professor and Head at E&TC Dept. of Kohlapur Institute of Technology, College of Engineering, Kolhapur, India. She have completed Ph.D and Post-Graduation (M.E) in the field of Electronics. She has teaching experience of 28 Years and Industrial Experience of 6 Years. She has published 75 papers in Journal and 25 papers presented in Conferences.

Email: shivsangeeta.chougule@rediffmail.com



Amol C. Bhosale received BE(E&TC),ME-Electronics(Digital System) in 2012.He is Pursuing PhD at Research Scholar, Department of ECE, BKIT, Bhalki and Visvesveraya Technological University, Belagavi ,India. e-mail: <u>amol5210@gmail.com</u>
The Design Of Multiband Microstrip Patch Antenna Use Of Truncation And Slot Techniques For Wireless Application

¹Dr.Basavalinga Swamy, ²Dr.Nagnath Biradar ¹Associate Professor LAEC, Bidar ²Data/Test Analyst Engineer, Infosys.Ltd, Hyderbad

Abstract— Typically, a microstrip antenna is created via the microstrip process on the PCB. They are designed to function at microwave frequencies. In today's communication systems, microstrip antennas have an advantage because of their tiny dimensions. The majority of communication devices use MPA because it has a low profile and a very low volume.

The patch's conducting substance is metal, such as copper or gold, and it can be shaped in any way, including triangles, circles, rectangles, and other arbitrary shapes. All things considered, Square design is incredibly easy, and extensive research is currently underway.

A square microstrip antenna with an operational frequency of 3.5GHz is intended to be designed utilizing a variety of methods, including slot loading, truncation, and parasitic. Additionally, simulations have demonstrated that antennas possess favorable radiation properties, such as broadside radiation patterns and dual band operations.

.Keywords-component: Bandwidth; microstrip antenna; VSWR, Impedance matching, Truncated.

I. INTRODUCTION

The Institute of Electrical and Electronics Engineering (IEEE) defines an antenna as a device that transmits or receives electromagnetic radio waves. Long-distance communication is impossible without antennas, which are a crucial part of microwave wireless communication systems.

The majority of antennas in use today are resonant, meaning they work well over a small range of frequencies. The frequency at which the transmitting system operates must be adjusted for these antennas. If not, there may be impairments to the transmission and/or reception. Many types of antennas have been designed and used in communication devices for past Over the past few decades, a wide variety of antennas have been built and employed in communication devices. These antennas come in a variety of sizes and forms, including dielectric and metal substrate materials. A few of these include array antennas, microstrip antennas (MSA), wire antennas, yagi uda antennas, dipole antennas, and others. Microstrip antennas have made a big impact on the communication industry. G. A. Deschamp introduced the idea of microstrip antennas in 1953. Nonetheless, in the 1970s, Robert E. Munson and John Q. Hwell build antennas for practical application. MSAs have now progressed from being new academically to being used commercially.

The characteristics of microstrip antennas are inferred using Maxwell's equations. The original and most basic kind of antenna was the dipole, which was made up of a single wire that was fed from the center. We consider a dipole's length to be equal to half of the operating frequency's wavelength. For a variety of communication applications, MSAs with different shapes—such as triangular, rectangular, circular, square, semicircular, elliptical, etc.—are employed. A straightforward patch antenna is comprised of a metal radiating patch, power dividers, a matching network, and a photoelectric phasing circuit etched on one side of a dielectric substrate board. Over on the other side is a metal ground plane.

II. SQUARE PATCH ANTENNA DESIGN WITH ANALYTIC APPROCH

The resonance frequency and diameter of the microstrip patch antenna determine its performance. The operating frequency, radiation efficiency, directivity, and return loss are all affected by the dimension. This paper uses a customized epoxy substrate with a specified resonance frequency of 3.5GHz. The substrate has a height of 0.16cm and a dielectric constant of 4.2. The practical patch width can be determined using the conventional formulae for efficient radiation.



Fig 1: Geometry view of Square Microstrip Patch

III. FEEDING TECHNIQUE

Microstrip antennas are fed by coaxial probes or microstrip lines through the ground plane, and they have radiating elements on one side of the dielectric substrate. Various elements to take into account when choosing Feeding techniques are as follows

- 1. Impedance matching radiating structure and feed Structure
- 2. Minimization of spurious radiation.
- 3. Suitability of feed for array applications

Different feeding techniques used in antennas are

- 1. Inset feed
- 2. Fed by quarter wavelength transmissionline
- 3. Co-axial Probe feed
- 4. Microstrip(Coplanar)feeds
- 5. Gap- coupledfeed
- 6. Proximity Coupled microstrip feed
- 7. Aperture Coupled microstripfeed
- 8. Coplanar Waveguidefeed

IV. METHODOLOGY

With HFSS, a basic patch antenna is simulated. An interactive software program called HFSS is used to determine a structure's electromagnetic behavior. The program has post-processing commands to examine this behavior in greater depth. HFSS facilitates computation

- 1. 1. Basic electromagnetic field quantities and, for open boundary situations, radiated near and far fields.
- 2. Propagation constants and characteristic port impedances.
- 3. S-parameters that have been generalized and renormalized to particular port impedances.
- 4. The resonances, or eigen modes, of a structure

V. RESULTSAND DISCUSSION

The Conventional Square Microstrip patch antenna (CSQMA) is simulated using HFSS software. Following are the results obtained for Peak gain, Return loss, Directivity, bandwidth and VSWR.



Fig 2: CSQMA on HFSS software



Fig 3: Variation between Frequency and Return Loss.



Fig 4: Variation between Frequency and VSWR



Fig 5: Pattern of CSQMA



Fig 6: Geometry truncated square microstrip antenna(CTSQMA)



Fig 7: Variation between Frequency and Return Loss.



Fig 8: Variation between Frequency and VSWR





Fig 10: Geometry of E slot conventional square microstrip antenna (ECSQMA)



Fig 11: Variation between Frequency and Return Loss.



Fig 12: Variation of VSWR vs Frequency



Fig 13: Pattern of ECSQMA



Fig 14: Geometry of H slot conventional square microstrip antenna (HCSQMA)



Fig 15: Variation between Frequency and Return Loss.



Fig 16: Variation of VSWR vs Frequency



Fig 17: Pattern of HCSQMA

$T_{-1} + 1_{-1} + 1_{-1}$	A	and an aturn atima	1:11			
Ianieli	$A na V 7 m \sigma$	and contrasting	different	microstrir	n antenna	narameters
1 40101.1	1 Mai y Ling	and contrasting	uniterent	merosun) antonna	parameters
	7 17					

Antenna	No.of	BW(%)	Return	VSWR
Geometry	Bands		loss(dB)	(dB)
CSQMA	1	2.58	-28.147	0.68
CTSQMA	1	2.66	-20.76	1.59
ECSQMA	3	2.95	-13.22	3.95
		4.78	-14.7	3.24
		2.54	-13.89	3.64
HCSQMA	3	3.3	-23.42	1.47
		3.9	-19.27	1.89
		6	-30.46	0.52

VI. CONCLUSION AND FUTURE WORK

It is clear from the thorough analysis of the work that microstrip antenna design and formulation are straightforward. In order to verify their findings, HFSS software simulates the 3.5 GHz CSMA architecture. By comparing the CSMA, adding parasitic parts to the patch, truncating the patches' corners, and inserting H and E slots into radiating patches, the study is expanded. Simulated results also show that antennas have good radiation properties, such as broadside radiation patterns and multiband operations.

REFERENCES

- 1. ConstantineA.Balanis:"AntennaTheory, Analysis and Design" (John Wiley & Sons).
- 2. Multi-slotted Microstrip antenna for wireless communication,M.T.Islam Institute of space science(Angkasa) UniversitiKebangsaan Malaysia BangiUKM 43600,Selangor D.E,MalaysiaM.N.Shakib and N.MisranElectrical,Electronic and system department UniversitiKebangsaan Malaysia BangiUKm43600,SellangorD.E,Malaysia.
- 3. I.J.Bahl and Bhartia, MicrostripAntenna, Artech House, 1980
- 4. G.Kumar And K.P.Ray, Broadband Microstrip Antennas, Firstedition, USA, ArtechHouse, 2003
- 5. R.G.Voughan.1988.Two-port higher mode circular microstrip antennas.IEEE, Trans.AntennasPropagat,36(3):309-321.

- 6. C.A.Balanis,"Antenna Theory and Design", 2nd Edition, New York, Wiley 1997.Prabhakar H.V and U.K.2007.Electronics Letters,2ndAugust.43(16)HFSS tutorial by Entuplestechnology.
- 7. Pozar David M.1998.Microwave Engineering .johnWiley, New York, NY, USA, 2ndEd.
- 8. A. Bora "Design of a Compact Circularly Polarized Square Microstrip Antenna Using InductiveLoading"NCC 2009, January 16-18, IIT Guwahati
- 9. S. Mangold,K.Challapati,"Coexistance of wireless networks in unlienced frequency band"WWRF#9 meeting,Zurich,Switzerland.
- 10. G. Eason, B. Noble, and I. N. Sneddon, "On certain integrals of Lipschitz-Hankel type involving products of Bessel functions," Phil. Trans. Roy. Soc. London, vol. A247, pp. 529–551, April 1955.(references
- 11. J. Clerk Maxwell, A Treatise on Electricity and Magnetism, 3rd ed., vol. 2. Oxford: Clarendon, 1892, pp.68-73
- 12. I. S. Jacobs and C. P. Bean, "Fine particles, thinfilmsand exchange anisotropy," in Magnetism, vol. III, G.T. Rado and H. Suhl, Eds. New York: Academic, 1963, pp. 271–350.
- 13. K. Elissa, "Title of paper if known,"unpublished.R. Nicole, "Title of paper with only first word capitalized," J. Name Stand. Abbrev., inpress.Y. Yorozu, M. Hirano, K. Oka, and Y. Tagawa, "Electron spectroscopy studies on magneto-optical media and plastic substrate interface," IEEE Transl.
- 14. J. Magn. Japan, vol. 2, pp. 740-741, August 1987 [Digests 9th Annual Conf. Magnetics Japan, p. 301, 1982].
- 15. M. Young, The Technical Writer's Handbook. Mill Valley, CA: University Science, 1989.www.hyperlinktech.com

Unmanned Aerial Vehicles (Uavs) And Iot Integration In Agriculture: A Comprehensive Review Of Smart Farming Solutions For Precision And Efficiency Enhancement

¹Ms. Pooja Ausekar, ²Dr. Mrs. M. M. Pawar ¹PG Student SVERI's COE, Pandharpur, India ²Associate Professor, SVERI's COE, Pandharpur

Abstract— This review paper critically examines the design and development of smart farming solutions through the integration of Unmanned Aerial Vehicles (UAVs) within an Internet of Things (IoT) framework. Drawing insights from diverse literature surveys, the study offers a thorough exploration of the contemporary landscape of smart farming, with a primary focus on enhancing precision and efficiency in agriculture. The research navigates challenges and opportunities at the intersection of UAV technology, IoT, and agriculture, synthesizing findings from various literature sources. The abstract encapsulates key components, methodologies, and the potential impact of the proposed smart farming solution. By providing a holistic overview of the research's scope and significance, this paper contributes to the evolving field of precision agriculture, addressing the pressing need for sustainable and technologically advanced farming practices.

Keywords- Unmanned Aerial Vehicles (UAVs), Systematic Review, IoT

I. INTRODUCTION

The paper focuses on the synergy between UAVs and IoT frameworks to create intelligent solutions for contemporary farming practices. In addressing the challenges of enhancing agricultural productivity, sustainability, and resource efficiency, the research aims to design and develop a robust Smart Farming solution. The paper outlines the integrated System's key components, methodologies, and outcomes, showcasing its potential to transform traditional farming practices. Specific objectives include the utilization of UAVs for data acquisition, establishing an efficient IoT infrastructure for real-time data analysis, and developing intelligent algorithms for decision support in agriculture.

The significance of this research lies in the potential for optimizing resource utilization, improving crop yield, and fostering sustainable farming practices through the integration of UAVs and IoT. The paper contributes to the evolving field of Smart Farming by presenting a practical and scalable solution applicable across diverse agricultural landscapes. In conclusion, the research paper unfolds the design and development of a Smart Farming solution, highlighting the marriage of UAV technology with an IoT framework. By addressing the challenges faced by modern agriculture, the research not only presents a technological innovation but also lays the groundwork for a more resilient, efficient, and sustainable future in farming practices. Subsequent sections delve into the methodology, results, and implications of this integrated approach, offering valuable insights for researchers, practitioners, and policymakers in the agricultural domain. Reviews of Literature Surveys

The agricultural sector accounts for 20% of India's GDP, making it a significant economic driver. Farmers can benefit from using UAVs and the Internet of Things (IoT) in agriculture by improving crop management, real-time monitoring, and prediction. Internet of Things (IoT)-based precision agriculture (PA) significantly outperforms manual farming in terms of production; this review will focus on the function of unmanned aerial vehicles (UAVs) in monitoring crop health, among other essential factors, WSNs, and platforms. [1]

The PRISMA-based systematic review emphasizes the promising use of UAVs with these cutting-edge technologies in smart farming. The enormous promise of UAVs in promoting sustainable agriculture is one of the key conclusions drawn from this study. The results show that UAVs have valuable features for collecting data, keeping an eye on things precisely, and making decisions in farming operations. These capabilities are further improved by integrating cloud, IoT, and AI, allowing remote access, efficient resource management, and real-time data analysis. Modern agriculture increasingly relies on technology-driven solutions, which is why this research is necessary. Farmers may use unmanned aerial vehicles (UAVs) and cutting-edge technology to enhance agricultural yields. If academics, politicians, and industry professionals are interested in smart farming and the possibilities and problems of using UAVs, the offered material is a great place to start. [2]

Methods for connecting and transmitting data to the ground station, as well as UAV systems and communication architecture, were covered in this article. Data processing through sensors and applications while maintaining security was also covered. Review the history of GCS and communication systems and offer essential definitions and ideas of UAV systems. Services include evaluations and assessments of UAV security measures and the many assaults that aim to disrupt UAV communication. Lastly, provide open research topics for improving UAV technology for the industry by further developing the UAV system and communication. [3]



Fig.1 IoT enables Smart. Agriculture

This study sought to explore the intellectual framework of UAV adoption for agriculture by extensively reviewing current research using the biblical SLR technique to synthesize previous information. This study makes a unique contribution in multiple ways. A comprehensive review of the geographic diversity of UAV use and plant and crop taxonomy was also carried out. In conclusion, this study's executive summary lays the groundwork for further work along the road. The research offers a comprehensive resource wherein technologists, agricultural experts, and researchers can obtain the overarching data necessary to make informed decisions on forthcoming. For instance, researchers can enhance the precision and efficiency of agricultural services while expanding coverage by devising the following experiment with superior or combined technologies if they comprehend the capabilities of UAVs and the effects of imaging modalities utilized in prior investigations. Technologists can construct and advance precision agricultural remote-sensing systems by leveraging a collection of UAVs and associated technologies that have undergone empirical testing. [4]

Small unmanned aerial vehicles (UAVs) and drones, in particular, have come a long way in the last several decades in design, construction, operation, flying characteristics, and navigation control. Photographing, route planning, search and rescue, inspecting electricity lines and civil works, etc., are a few services that heavily use UAVs. With a quick note on the research and advancements covered in the last decade, This literary work offers a comprehensive analysis and wide-ranging survey of recent advancements in commercially available UAVs. The research offers a comprehensive framework for comprehending the progressive advancement of advanced drones/UAVs, encompassing aspects such as intelligent behavior, smart structure, sensing and visual capabilities, aviation quality, path planning, and adaptability. A systematic literature review is carried out on 254 retrieved publications published within the past decade, with 96 items deemed relevant. Path planning, neural networks, AI, inspection, surveillance, tracking, identification, etc., are these selected articles' most pertinent approaches or applications. This study aims to shed light on the expanding role of unmanned aerial vehicles (UAVs) and drones in modern life while encouraging aspiring researchers to go deeper into the topic and develop novel approaches to tailoring flying robots to specific tasks. Additionally, researchers can use this page as a guide to better comprehend and assess their study work with preexisting solutions. As a bonus, it provides a concise synopsis of the extensive literature in the relevant domains, which is helpful for both newbies and pilots/practitioners. [5]

As the world's population rises, agriculture's role as a primary food producer will significantly grow. As a result of public pressure, policymakers are under increasing pressure to guarantee enough food on Earth to feed everyone. Therefore, an option for effective farm management that can boost output is using a UAV. This article aims to educate readers on using UAVs and their agricultural applications to encourage continued use and ensure the technology's long-term viability. Based on the literature review, it was discovered that UAVs have many potential uses in agriculture. As for the methods, drew on a literature assessment of all the studies done thus far. The change's outcome is that various sensors provide varying degrees of crop analysis. To improve data quality and analysis, it is necessary to analyze the project's goal before utilizing UAV

technology. Finally, to use UAVs to collect precise data and conduct reliable analyses, it is necessary to identify an appropriate sensor and UAV. [6]

A critical component of smart warehouse operations, unmanned aerial vehicles (UAVs) allow faster and more efficient receiving, picking, storing, and shipping. On the other hand, research reviews focusing on UAV-based warehouse management are few and far between. There needs to be more research and insights into UAV uses in this industry, which makes it difficult to make informed decisions about commercial usage. This research presents a systematic literature review (SLR) on utilizing UAVs in warehouse management to optimize the current state and prospective applications of UAVs. Based on two descriptive research questions—"What are the past applications of UAV?" and "What are essential elements affecting UAV adoption in warehouse management?"—an SLR approach was used to identify, select, appraise, and summarise data critically. It was possible to identify thirteen subfactors and five major ones. Based on the findings, the main elements affect hardware components (such as payloads, battery power, and sensors). The managerial implications of research can help decision-makers and practitioners put UAV-based warehouse management to good use. [7]

Drones have become more commonplace in many industries thanks to advancements in wireless communication and information technology services. When used as a bridge, drones offer many services that can boost the efficiency of any particular Internet of Things system. This paper is the initial effort to survey AI-powered MECs that can be accessed using drones. This paper delves into the latest research trends and offers future perspectives on the possible applications of UAV-enabled MEC in the Internet of Things realm. Using the most popular new AI approaches, it demonstrates the possible contributions of UAVs to the IoT era's communication and intelligent computing infrastructure. A few issues that UAVs as MECs are generating much interest from researchers are looking to improve these applications' performance in various contexts. Particularly in MEC, more effort is required to introduce the new concept of UAV-enabled MEC.[8]

To offer a thorough development and comprehension of current research hotspots, this study utilized a bibliometric method to examine. The analysis found an uptick in studies involving wheat and UAVs during the study's time frame. While developing nations use UAV technologies less frequently, industrialized nations that grow much wheat employ them more frequently. Research outputs and the most cited papers worldwide follow a similar pattern. According to the significant findings, UAVs can monitor and provide important real-time crop variability information, allowing for fast crop management. The findings of this study will be helpful for agricultural institutions and government corporations as they work to enhance crop management practices for high-yielding crops to fulfill the demands of the global market. Because countries with less or no research on wheat and UAV studies are the ones whose authors had the lowest keyword rankings, the results of this study are crucial for improving agricultural methods in crop production and guiding future research. The study's shortcomings were data fusion and the complicated assembly of numerous datasets. This paper demonstrates the potential of the approaches used in the current study; it would be beneficial to combine other research databases to uncover other prospective research breakthroughs within the study's specialty area. [9]

This study demonstrated how smart agriculture can help close the food demand gap by enhancing and expanding agricultural output. The Internet of Things (IoT) links all parts of smart systems, including those used in agriculture and other industries, making it the backbone of smart agricultural technology. Many agricultural activities can benefit from the Internet of Things (IoT), including crop monitoring, irrigation, insect management, harvesting, and more. The IoT links multiple sensors to processing units to make the right real-time judgments, subsequently evaluating the data. This study summarised the benefits and drawbacks of IoT integration with UAV and robotics systems controlled by artificial intelligence (AI) techniques in underdeveloped nations. Speed of data transport is now a key performance metric for SF. Therefore, the 5G network introduced smart agriculture and offered effective and adaptable solutions due to its significantly higher speed than 4G networks. Some Egyptian ideas are excellent examples of how smart agricultural technology is starting to spread; it can assist developing nations in improving their agricultural sectors and making farms more sustainable. Lastly, governments in developing nations should aid small-scale farmers in adopting these smart technologies to boost output while decreasing water and land waste. [10]

This is the first comprehensive study on recent developments in precision agriculture utilizing edge intelligence and unmanned aerial vehicle RS. The key findings are: (a) For precision agriculture, unmanned aerial vehicle (UAV) systems consisting of compact, lightweight, industrial rotor-wing or fixed-wing UAVs are frequently employed. b) Although UAV sensors can gather data from various sources, the availability of publicly accessible UAV datasets for intelligent precision agriculture is limited. DL-based UAV RS approaches can be classified into three categories: segmentation, object detection, and classification. Convolutional and recurrent neural networks are these tasks' most often employed network topologies. (d) UAV RS data processing frequently uses cloud computing, but edge computing facilitates computation closer to the data sources. (e) Edge intelligence integrates cloud computing and edge computing. Model compression is the most significant and extensively implemented technique, specifically parameter pruning and quantization. Standard edge resources comprise compute units, GPUs, and field-programmable gate arrays. [11]



Fig.2 Smart Agriculture

This study designed a cheaper automated UAV pesticide sprayer for rice production systems that could be made locally. In a field performance test, the UAV sprayer covered 750 m2 in 10 minutes and applied 3.2 liters of water per 1,000 m2. Unfortunately, the battery life was insufficient to cover at least 0.1 hectares. The study's proposed UAV pesticide sprayer was the most cost-effective regarding its tank capacity and field capacity, as measured against the market price of UAV sprayers today.[12]

Many current and prospective business areas can benefit from Unmanned Aerial Vehicle Technology, which is the focus of this research. As a result of Industry 4.0 and subsequent waves of digitization, nearly every industry is undergoing some transformation. The military and civilian sectors also stand to gain from these technological advancements. Unmanned Aerial Vehicles (UAVs) or drones can do much work with little human intervention, which is very helpful in farming. Future digital system implementation can be better understood by studying drone systems in UAV and agricultural sectors. [13]

Pesticides are necessary to maintain high agricultural yields and eradicate pests from plants. Soil fertility is reduced. Another approach that is both precise and flexible is unmanned aerial vehicle (UAV) variable-rate spraying, which can be used to overcome these obstacles. Trends in the use of semi-automatic methods and platforms designed for use on land for precision spraying are investigated in this study. A combination of an autonomous control system and several hardware components, including microcontrollers, sensors, pumps, and nozzles, allows for precise spraying, efficient UAV performance, and adaptability to different plant pesticide needs. The consequences of current and future research are addressed in this article. To help with future studies, compare the hardware, control system methodologies, and data collecting from each study's parameters. Although there are severe constraints on the amount of spraying, future study is encouraged to combine variable rate development with farmland mapping to identify the requirement for pesticides. This will help continue the precision performance, even though the quality is beneficial. [14]

Unmanned aerial vehicles (UAVs) have tremendous potential for smart farming. One factor that could encourage farmers to adopt UAVs for smart farming is how easy and cheap they are to control. Individuals typically pilot Unmanned Aerial Vehicles at a distance via radio waves. Several other technologies are also utilized for UAV control, including WiFi and ZigBee. The widespread availability of smartphones and the low cost of smart Bluetooth make it a compelling technology. In the not-too-distant future, farmers will be able to control their UAVs using any smartphone, even ones with Bluetooth Smart capabilities. Some prerequisites and obstacles must be overcome before UAVs may be used for smart agriculture applications. Therefore, the purpose of this article is to make an effort to investigate the kinds of sensors that would work well for smart farming, as well as the possible needs and difficulties of using UAVs for smart agriculture. Also foreseen the potential uses of UAVs in intelligent farming in the future. [15]

Unmanned aerial vehicles (UAVs) and other regularised smart-farming solutions are being examined for usage in agriculture (UAVs). Unmanned Aerial Vehicles (UAVs) integrate ICTs, robotics, AI, big data, and the Internet of Things. Because of their versatility and high capabilities, agricultural UAVs are finding new uses in every facet of farming, from sowing seeds to assessing and mapping crop development to spraying fertilizer and pesticides. So, along with the related technologies, the market for agricultural UAVs is anticipated to keep rising. Several examples of actual agricultural settings where UAVs have been used. Agricultural UAVs and their potential future developments are also discussed. [16]



Fig.3 Field Robotics Farming

Unlike traditional aircraft, unmanned aerial vehicles (UAVs) don't need human pilots to fly them. Unmanned Aerial Vehicles (UAVs) have grown in popularity and use due to the tremendous improvements in electronics and IT. Recent breakthroughs in electronics and IT have made UAV operations safer, more efficient, and more successful for civilians. Unmanned Aerial Vehicles (UAVs) are created. Drones have grown in popularity among the general public as they have become more accessible and inexpensive. As a result, this technology is versatile and subject to continuous improvement. Drones significantly impact the mining, maritime, oil, gas, and seaport industries, known for their demanding and complex work settings. Industrialists are increasingly turning to drones to optimize and improve their processes, all while increasing operational efficiency. Unmanned Aerial Vehicles (UAVs) cover much ground in this chapter, from their origins and development to their classification, their impact on various sectors, and the viewpoints of academics and businesses. [17]

A. Internet of Things

A UAV, embedded systems, and the IoT are the building blocks of a novel mobile application that utilizes current information and communication technologies in farming (UAV). Water pumps, WiFi modules, and Arduino microcontroller boards were utilized to design and implement the suggested agricultural monitoring system. This research will use unmanned aerial vehicles (UAVs) to gather environmental data from various farm parts. The water needed for irrigation is then calculated automatically for every area in the cloud. Additionally, the farmers can be guided by the developed System's ability to remotely monitor farm conditions, including water requirements, through an Android mobile application. This study's findings show that the suggested Internet of Things (IoT) embedded system can effectively reduce water irrigation that isn't needed or wasted, all within the context of smart agriculture. [18]

Thanks to their versatility and speed, unmanned aerial vehicles (UAVs), or drones, have grown more frequent in recent years. These aircraft have many practical uses in the real world. The scientific community has recently shown a great deal of interest in the use of UAVs for precision farming. This study uses various approaches to compare nine robust models of deep neural networks for plant disease detection. Trained to respond appropriately to new inputs using deep feature extraction and transfer learning. The F1-score, specificity, sensitivity, and accuracy are all used to measure performance. The results showed that deep feature extraction and SVM/ELM classification outperformed transfer learning. In addition, the analysis attempts to evaluate the efficacy and cost of the different techniques. For instance, many methods face challenges like finding parasites and halting their spread with the right amount of pesticides, researching the area in the least amount of time practicable, and eliminating the exact location being investigated by more drones. Researchers rely on simulation models to evaluate these technologies, develop specialized approaches, and coordinate operations to support farms and achieve the target successfully. The primary goal of this research is to examine and contrast two separate parasite identification algorithms' search strategies. [19]

The inexpensive ground-sensing GSs can monitor soil moisture in the Napier and Ruzi grass farms during the day. A WiFi transceiver, comparable to hotspot mobility, is installed on the UAV-SC and linked to numerous GSs. The name of this smart farming system that is being proposed is the GS-to-UAV-SC communication system. Using the UAV-SC as a mobility transceiver to make up for the low power of GSs, the results and implementation prove it. This includes the real-time data used to track soil moisture readings on the go. Machine learning methods to optimize the path loss model. [20]

Unmanned aerial vehicles are helpful for precision farming, weeding, fertilizer distribution, and field management (UAVs). One advantage of this strategy is improving crop quality while reducing production expenses. "Smart farming" uses geographic data to improve crop yields by predicting field variability, ensuring the best possible inputs, and minimizing waste. This study proposes an Internet of Things (IoT) Smart Farming Framework (IoT-SFF) that uses geospatial analysis to gather huge data. A close look at how farmers use wireless sensors in Internet of Things (IoT) gadgets and communication techniques. Crop status, soil preparation, bug and pest detection, and irrigation scheduling are just a few of the specific agricultural uses for the Internet of Things sensors that are currently accessible. Using a GIS for agricultural irrigation and monitoring has opened up new perspectives on regions, allowing us to make more informed agrotechnological decisions. Big data is essential for smartly running and managing farms since it can be analyzed and monitored to make decisions quickly and accurately. Technical and economic factors still limit it, though. The simulation results demonstrate that the suggested IoT-SFF model enhances several metrics, including crop yield ratio (92.4% improvement), prediction ratio (97.7% improvement), accuracy ratio (94.5% improvement), average error (38.3% improvement), and low-cost rate (34.4% improvement).[21]



Fig.4 Remote Sensing UAV

UAVs have much-untapped potential to revolutionize various industries, from security and medicine to surveillance and traffic monitoring. Much money has been spent developing UAV and multi-UAV systems that can work together to finish tasks faster and cheaper. UAVs outfitted with cameras, sensors, and GPS receivers have immense promise as a new airborne IoT domain, made possible by emerging technologies like 4G/5G networks. But security, privacy, and management are only a few of the numerous problems that need fixing before UAVs can be put to good use. Lastly, it presents a system that allows and facilitates various technologies on UAVs. Protecting UAVs as "flying" things in a collaborative networked environment is made possible by the presented framework, which offers a comprehensive Internet of Things architecture.[22]

Modern technological solutions to long-standing issues in agriculture have been made possible through the Internet of Things (IoT). Yield, quality, and production may now be more easily reconciled. Seamless end-to-end intelligent operations and increased business process execution allow for faster processing and delivery of produce to supermarkets. This article provides information on smart agriculture, including numerous case studies, uses of the IoT and UAVs in agriculture, different communication methods, and the difficulties and restrictions of connecting these technologies in faraway locations. The article detailed the connection constraints regarding transmission distance and communication technology. [23]

Drone monitoring improves several agricultural metrics. Various drones are utilized to track the development, pesticide levels, field water content, and overall health of crops. Drones not only improve yield through precise monitoring, but they are also useful in densely populated fields. It has the potential to boost efficiency and offer real-time monitoring over the Internet. One common use of drones is crop yield prediction, which may be done in any climate. Additional yield-inducing uses for drones include real-time acquisition of high-precision photographs of the field. A drone equipped with both visible and infrared light can scan a crop, allowing for an assessment of its health and the detection of any pathogens present. Solar panels are going to be used in the future. Using solar panels to charge drones throughout the day allows for reduced power consumption and their

operation in the field. Classification of crops is another area that makes use of SVM. When used for fruit ripening prediction, SVM achieves high accuracy. [24]

The integration of cutting-edge technologies to revolutionize agriculture. The project focuses on real-time monitoring, UAVbased aerial surveys, AR visualization, and ML-driven predictive models for crop management. This holistic approach empowers farmers with efficient decision-making tools, ultimately improving productivity and sustainability and contributing to global food security. Published in "Smart Sensors for Industrial Internet of Things" in 2021, the work represents a significant advancement in applying advanced technologies to agriculture. [25]

Most people's income comes from farming, which is why it's important. To protect crops from pests and increase harvest yield, farmers use pesticides. Currently, fertilizers and pesticides are sprayed by hand, which hurts the human neurological System and results in numerous annual fatalities. The World Health Organization has estimated that about one million pesticide-related illnesses occur annually. Avoiding the need for direct human spraying, this research describes a precision agriculture system that uses a remote-controlled drone-based sprayer. Compared to conventional spraying methods, the technique can boost crop health.[26]

Internet of Things (IoT) technology, such as humidity and temperature sensors, autonomous farming equipment, GPS, and aerial imagery, are all potential tools for the future of agriculture. How the sensors are designed to work will determine this. Weed sensors are a part of this study's field data collection system, including thermometers, infrared sensors, and microwave radiometers to measure environmental conditions. A plethora of data acquired through these means, along with the ever-expanding capabilities of GPU and parallel computing, has enticed researchers to use data-driven analysis and deep learning techniques in the grain industry. Here, data is collected using an IoT module, and weed and crop recognition is done. In addition to checking for crop abnormalities, this System can detect them using feature extraction and an FRNN-based classification algorithm to estimate their level. The simulation results show that the proposed design has a 0.058 improvement in ideal accuracy, precision, specificity, and MAE compared to the present technique. [27]

B. Blockchain Technique

Smart farming operations are the best answer to agriculture problems, which include increasing air temperatures and population expansion. This research intends to add to what is already known about how blockchain technology could improve agricultural output, reduce environmental effects, and automate farmers' tasks, all of which would contribute to the idea of smart farming. To build confidence among savvy farming users, it suggests a secure architecture based on blockchain technology. The framework employs an asymmetric critical exchange method. Since changing the hash value of data corrupts the entire chain of blocks, the SHA-256 hashing function is used to guarantee data integrity. Using smart contracts to describe the framework's functions, a proof-of-concept implementation was built on the Ethereum blockchain to show that the proposed framework could be implemented. Used Hyperledger Caliper to measure the proof of concepts. The results show that blockchain technology can solve some problems with smart farming, especially with data availability and integrity, and progress smart farming in an efficient and scalable way. [28]



Fig.5 Tracking Crop System

C. Embedded based System

Precision agriculture (PA) research focuses on developing decision-making systems rooted in controlling and managing agricultural sites. These systems use monitoring fields and metrics to track performance to maximize investment returns while

minimizing waste. Applications utilizing satellites, UAVs, and sol robots can be observed on expansive agricultural lands. These applications rely on a wide range of algorithms, which can be time-consuming. When it comes to algorithms that should be embedded and help make decisions in real-time, more is needed to assess them offline on workstations or desktops. Consequently, it requires a state-of-the-art study that employs a hardware-software co-design method to develop decision systems that incorporate various techniques, such as processing units and sensor data collecting and disease diagnosis are some of the many uses for PA algorithms provided here as part of an integrative assessment of processing information tools. This review has relied on over a hundred studies to glean valuable information on the various methods and information processing systems. The study details the methods, databases, and tools used to determine the pros and cons of the System and its applications.[29]

D. Fog Computing Framework

An approach based on the utilization of IoT devices and unmanned aerial vehicles (UAVs) is suggested in this study. Nevertheless, this unprotected setting is prone to invasions that could impede data collecting and, in the end, lower agricultural output. The IDS uses machine learning methods based on the publicly available CICIDS2017 dataset to identify and stop intrusions. A layer-based intrusion detection technique can detect known and zero-day threats. An intrusion detection system that relies on signatures employs a decision tree, XGBoost, extra tree, and random forest algorithms to thwart known threats. Clustering methods, such as the K-means algorithm, counter zero-day attacks. Based on the evaluation results, XGBoost is the top performer. It has a 99.77% accuracy rate for intrusion detection. This suggested work is designed to be implemented and benchmarked using a modular approach that considers UAV energy, transmission, and communication delays. Researchers want to use machine vision and deep learning to enhance the intrusion detection system in the future. [30]

II. CONCLUSION

In conclusion, this research paper's synthesis of literature surveys underscores the transformative potential of integrating UAVs into an IoT framework for Smart Farming. The culmination of insights from various studies supports developing and deploying a robust solution that addresses modern agriculture's challenges. The research illuminates the importance of UAVs in aerial data acquisition and the synergy with IoT for real-time data analysis and decision support. By carefully examining existing literature, this paper positions the proposed Smart Farming solution as a forward-looking and impactful contribution to the field. As agriculture stands on the cusp of technological innovation, the conclusions drawn from this literature review project provide a roadmap for researchers, practitioners, and policymakers to navigate the evolving landscape of precision agriculture.

REFERENCES

- 1. V. A. Diya, Pradeep Nandan & Ritesh R. Dhote, "Iot-Based Precision Agriculture: A Review," Proceedings Of Emerging Trends And Technologies On Intelligent Systems, 2022
- El Mehdi Raouhi, Mohamed Lachgar, Hamid Hrimech3 Ali Kartit, "Unmanned Aerial Vehicle-Based Applications In Smart Farming: A Systematic Review," Ijacsa-International Journal Of Advanced Computer Science And Applications, Vol. 14, No. 6, 2023
- 3. Asif Ali Laghari A, Awais Khan Jumani B, Rashid Ali Laghari C, Haque Nawaz, "Unmanned Aerial Vehicles: A Review," Cognitive Robotics, 2023
- 4. Md. Abrar Istiak, M.M. Mahbubul Syeed, Md Shakhawat Hossain, Mohammad Faisal Uddin, Mahady Hasan, Razib Hayat Khan, Nafis Saami Azad, "Adoption Of Unmanned Aerial Vehicle (Uav) Imagery In Agricultural Management: A Systematic Literature Review", Ecological Informatics, 2023
- 5. Faiyaz Ahmed1 J. C. Mohanta1 Anupam Keshari2 Pankaj Singh Yadav, "Recent Advances In Unmanned Aerial Vehicles: A Review," Arabian Journal For Science And Engineering, 2022
- 6. C Y N Norasma, M A Fadzilah, N A Roslin, Z W N Zanariah, Z Tarmidi, F S Candra, "Unmanned Aerial Vehicle Applications In Agriculture," IOP Conf. Series: Materials Science And Engineering 506 (2019
- 7. Malang, C.; Charoenkwan, P.; Wudhikarn, R." Implementation And Critical Factors Of Unmanned Aerial Vehicle (UAV) In Warehouse Management: A Systematic Literature Review."Mdpi Drones 2023
- 8. Yazid, Y.; Ez-Zazi, I.; Guerrero-González, A.; El Oualkadi, A.; Arioua, M. "Uav-Enabled Mobile Edge-Computing For Iot Based On Ai: A Comprehensive Review.", Mdpi Drones 2021
- 9. Lwandile Nduku, Cilence Munghemezulu, Zinhle Mashaba-Munghemezulu, Ahmed Mukalazi Kalumba, George Johannes Chirima, Wonga Masiza, and Colette De Villiers, "Global Research Trends For Unmanned Aerial Vehicle Remote Sensing Application In Wheat Crop Monitoring," Mdpi Geomatics 2023

- Elsayed Said Mohamed, Aa. Belal, Sameh Kotb Abd-Elmabod, Mohammed A. El-Shirbeny, A. Gad, Mohamed B Zahran, "Smart Farming For Improving Agricultural Management," The Egyptian Journal Of Remote Sensing And Space Sciences, 2021
- 11. Jia Liu, Jianjian Xiang, Yongjun Jin, Renhua Liu, Jining Yan, and Lizhe Wan, "Boost Precision Agriculture With Unmanned Aerial Vehicles Remote Sensing And Edge Intelligence: A Survey," Mdpi Remote Sens. 2021
- 12. E P De Padua, R C Amongo, E P Quilloy, D C Suministrado, J C Elauria, "Development Of A Local Unmanned Aerial Vehicle (Uav) Pesticide Sprayer For Rice Production System In The Philippines," IOP Conf. Series: Materials Science And Engineering, 2020
- 13. İlkim Yüksel, "Digitalization With The Use Of Unmanned Aerial Vehicle (UAV) In Agriculture," Journal Of Industrial Policy And Technology Management, 3(2), 2020
- 14. Hanif, A.S.; Han, X.; Yu, S.-H. "Independent Control Spraying System For Uav-Based Precise Variable Sprayer: A Review." Mdpi Drones 2022
- Praveen Kumar Reddy Maddikunta; Saqib Hakak; Mamoun Alazab; Sweta Bhattacharya; Thippa Reddy Gadekallu, "Unmanned Aerial Vehicles In Smart Agriculture: Applications, Requirements, And Challenges," Ieee Sensors Journal (Volume: 21, Issue: 16, 15 August 2021
- 16. Jeongeun Kim; Seungwon Kim; Chanyoung Ju; Hyoung Il Son, "Unmanned Aerial Vehicles In Agriculture: A Review Of Perspective Of Platform, Control, And Applications", Ieee, 2019
- 17. S. Jayanthi, H. Shaheen, U. Balashivudu & Meesala Shobha Rani, "Evolution And Significance Of Unmanned Aerial Vehicles," Unmanned Aerial Vehicle Cellular Communications, 2022
- Mohamed Esmail Karar, Faris Alotaibi, Abdullah Al Rasheed, Omar Reyad, "A Pilot Study Of Smart Agricultural Irrigation Using Unmanned Aerial Vehicles And Iot-Based Cloud System," International Journal Of Information Sciences Letters, 2021
- Mohamad Reda. A. Refaai , 1 Vinjamuri Snch Dattu,2 N. Gireesh,3 Ekta Dixit,4 Ch. Sandeep,5 And David Christopher, "Application Of Iot-Based Drones In Precision Agriculture For Pest Control," Hindawi Advances In Materials Science And Engineering Volume 2022
- Sarun Duangsuwan, Chakree Teekapakvisit, Myo Myint Maw, "Development Of Soil Moisture Monitoring By Using Iot And Uav-Sc For Smart Farming Application," Advances In Science, Technology And Engineering Systems Journal Vol. 5, No. 4, 2020
- 21. Wei Zhao, Meini Wang, and V. T. Pham, "Unmanned Aerial Vehicle And Geospatial Analysis In Smart Irrigation And Crop Monitoring On Iot Platform," Hindawi Mobile Information Systems Volume 2023
- 22. Thomas Lagkas, Vasileios Argyriou, Stamatia Bibi And Panagiotis Sarigiannidis, "Uav Iot Framework Views And Challenges: Towards Protecting Drones As "Things," Mdpi Sensors 2018
- 23. Dr. Nookala Venu, Dr. Aarun Kumar, Mr. A.Sanyasi Rao, "Smart Agriculture With Internet Of Things And Unmanned Aerial Vehicles," Neuroquantology | June 2022
- 24. C Bhuvaneshwari, G Saranyadevi, R Vani, A Manjunathan, "Development Of High Yield Farming Using Iot Based Uav," Iop Conf. Series: Materials Science And Engineering, 2021
- 25. Vijayakumar Ponnusamy & Sowmya Natarajan, "Precision Agriculture Using Advanced Technology Of Iot, Unmanned Aerial Vehicle, Augmented Reality, And Machine Learning," Smart Sensors For Industrial Internet Of Things, 2021
- 26. M. Manoj Vihari; Usha Rani. Nelakuditi; M. Purna Teja, "Iot Based Unmanned Aerial Vehicle System For Agriculture Applications," International Conference On Smart Systems And Inventive Technology (Icssit), 2018
- 27. K. Vijayalakshmi, Shaha Al-Otaibi, Leena Arya, Mohammed Amin Almaiah, T. P. Anithaashri, S. Sam Karthik And Rima Shishakly, "Mdpi Sustainability 2023
- 28. Kahlid S. Alqarni, Faris A. Almalki, Ben Othman Soufiene, Obaid Ali, And Faisal Albalwy, "Authenticated Wireless Links Between A Drone And Sensors Using A Blockchain: Case Of Smart Farming," Hindawi Wireless Communications And Mobile Computing Volume 2022
- 29. Amine Saddik, Rachid Latif, Abdelhafid El Ouardi, Mohamed Elhosenyc, And Adel Khelif, "Computer Development Based Embedded Systems In Precision Agriculture: Tools And Application," Acta Agriculturae Scandinavica, Section B — Soil & Plant Science 2022
- 30. Sajid, J.; Hayawi, K.; Malik, A.W.; Anwar, Z.; Trabelsi, Z., "A Fog Computing Framework For Intrusion Detection Of Energy-Based Attacks On Uav-Assisted Smart Farming" Mdpi Appl. Sci. 2023

The impact of Text Preprocessing Techniques on Word Sense Disambiguation (WSD) in Natural Language Processing (NLP)

¹Somshekhar Aland, ²Rajivkumar Mente ¹Bharatratna Indira Gandhi College of Engineering ²Department of Computer Applications, School of Computational Science

Abstract:WSD is vital in NLP, aiming to discern word meanings in context. This study explores the impact of text preprocessing on WSD, utilizing WordNet and NLTK. Techniques such as Tokenization, stop word removal, lemmatization, POS tagging, and word embeddings refine contextual data. Empirical evaluations reveal that effective preprocessing enhances WSD accuracy, leveraging WordNet's rich lexical database. NLTK facilitates the seamless integration of these techniques into WSD pipelines. The paper provides compelling evidence that correct preprocessing significantly improves WSD system performance, offering insights for achieving top-tier results in NLP applications.

Keywords: Word Sense Disambiguation, Natural Language Processing, WordNet, Natural Language Toolkit, Tokenization, stop word removal, lemmatization, Part-of-Speech tagging, context window strategies, and word embeddings.

I. INTRODUCTION

In Natural Language Processing work, the Word Sense Disambiguation task is comparable to solving a linguistic puzzle. Consider the following sentence: "The bank can't approve your loan."

At first glance, it may seem straightforward, but hidden within lies a linguistic problem. The word "bank" holds multiple meanings - it could refer to a financial institution or the side of a river. This ambiguity underscores a fundamental challenge in NLP - the need to determine a word's precise sense within a sentence's context.

WSD is crucial for machine language understanding, enhancing sentiment analysis and machine translation applications. This study delves into the impact of text preprocessing techniques on WSD efficacy, leveraging WordNet and the Natural Language Toolkit. Various strategies, including Tokenization, stop word removal, lemmatization, and Part-of-Speech tagging, are explored to refine contextual information around ambiguous words. The interaction of these techniques with resources like WordNet elevates the precision of sense disambiguation. Empirical experiments prove that careful implementation of these techniques enhances the reliability and accuracy of WSD systems, offering valuable insights for NLP applications.

II. CHALLENGES OF WSD DUE TO WORD AMBIGUITY [1]

Word Sense Disambiguation is a complex undertaking because of several difficulties, such as:

i) Lexical Ambiguity:

Finding the right word in a context can be difficult because many words have numerous meanings.

ii) Context Dependency:

A word's meaning can vary depending on the surrounding words and the context. WSD systems must consider these contextual cues to determine the appropriate sense.

iii) Data Sparsity:

Annotating large amounts of training data with sense labels for every word in all possible contexts is arduous and timeconsuming, leading to the limited availability of labelled data for training WSD models.

iv) Word Sense Granularity:

Determining the appropriate granularity of senses is essential. Choosing overly fine-grained senses may lead to sparse data, while coarse-grained senses may not capture subtle distinctions.

v) Cross-Lingual WSD:

WSD becomes more complex when word senses are translated between languages, requiring multilingual tools and methods.

III. APPLICATIONS OF WSD [2]:

i) Machine Translation (MT):

WSD is vital for MT, facilitating the selection of accurate translations by considering word context and preventing misunderstandings due to multiple meanings.

ii) Information Retrieval (IR):

Search engines use WSD to improve document retrieval accuracy by distinguishing between query phrases and document content.

iii) Information Extraction (IE):

WSD simplifies the extraction of structured data from unstructured text, aiding in determining the semantic function of words and sentence-level items.

iv) Question Answering Systems (QAS):

WSD is essential in QAS for comprehending user inquiries and locating suitable answers, ensuring the system understands the intended meaning of questions.

v) Sentiment Analysis (SA):

WSD contributes to precisely identifying sentiment in sentiment analysis, which is crucial for understanding user-generated content motivations on the web.



IV. METHODOLOGY:

Fig 1 Steps of Text Preprocessing

A detailed explanation of standard text preprocessing techniques applied in the Word Sense Disambiguation context is: "The bank can't approve your loan."

1) Tokenization:

Tokenization involves breaking or splitting a text into individual words or tokens. It segments sentences into discrete units, allowing for more accessible analysis.

There are three common types of Tokenization: -

i) Whitespace Tokenization:

This method splits the text based on spaces.

Example:

Input Context = "The bank can't approve your loan."

After Whitespace Tokenization:

['The,' 'bank', "can't", 'approve', 'your', 'loan.']

ii) NLTK Word Tokenization:

NLTK is a popular Python library for NLP. It provides a word tokenizer that can handle various tokenization cases.

Example:

Input Context = "The bank can't approve your loan."

After NLTK Word Tokenization:

['The', 'bank', 'ca', "n't", 'approve', 'your', 'loan', '.']

iii) Regular Expression Tokenization:

This method uses regular expressions to split the text based on specific patterns, such as spaces and punctuation.

Example:

Input Context = "The bank can't approve your loan."

After Regular Expression Tokenization:

['The', 'bank', "can't", 'approve', 'your', 'loan']

Advantage: It ensures consistency, as words with different cases are treated the same. It improves the matching of words for disambiguation.

Disadvantage: This can lead to loss of information, as the case might carry significance (e.g., proper nouns).

2) Lowercasing:

Converting all characters in the text to lowercase ensures uniformity and helps treat words with different cases as the same token.

Example:

Input Context = "The bank can't approve your loan."

After lowercasing:

the bank can't approve your loan.

Advantage:

Ensures consistency, as words with different cases are treated identically. It improves the matching of words for disambiguation.

Disadvantage:

This can lead to loss of information, as the case might carry significance (e.g., proper nouns).

3) Stopword Removal:

Stopwords are common words (e.g., "the," "and," "is") that might not carry significant meaning for WSD. Removing them reduces noise.

Example:

Input Context = "The bank can't approve your loan."

After stopwords removal:

bank 't approve loan .

Advantage:

Reduces noise and computational complexity. Focuses on content-bearing words, aiding in context quality for WSD Disadvantage:

It can result in loss of context nuances or unintended consequences.

4) Stemming:

Stemming reduces words to their root or base form by removing suffixes. It can lead to incomplete words but helps normalize variations. There are three types of stem.

i) Porter Stemming:

Porter stemming is a widely used algorithm. It applies a series of rules to reduce words to their base form.

Example:

Input Context = "The bank can't approve your loan."

After Porter stemming, the word:

the bank ca n't approv your loan.

"can't" is not handled well by stemming.

ii) Snowball Stemming (Porter2):

Snowball is an extension of the Porter stemming algorithm with additional rules and support for multiple languages. Example:

Input Context = "The bank can't approve your loan."

After Snowball stemming the word:

the bank ca n't approv your loan .

Just like with Porter stemming, Snowball stemming can sometimes result in words that are not valid English words. Here, "can't", contractions are not typically handled.

iii) Lancaster Stemming:

The Lancaster stemming algorithm is an aggressive stemming algorithm that can produce concise root forms.

Example:

Input Context = "The bank can't approve your loan."

After Lancaster, Stemming, and the word:

the bank can't approv yo loan .

As you can see, stemming does not handle

contractions well, and they often leave contractions as they are or produce very short, sometimes meaningless stems. Stemming is more effective for regular words, but there may be better choices for text preprocessing when dealing with contractions and other irregular forms of words.

Advantages:

It simplifies words to their root forms, reducing vocabulary size. It improves the matching of similar words for WSD. Disadvantage:

This can lead to over-simplification, generating non-real words. It might not preserve semantic meaning as well as lemmatization.

5) Lemmatization:

Lemmatization, which reduces words to their base or dictionary form, is preferred over stemming for WSD. It helps to maintain the contextual integrity of words. Lemmatization has improved the accuracy of WSD systems by providing more meaningful context.

There are three commonly used lemmatization techniques:

i) WordNet Lemmatization (NLTK):

Example:

Input Context = "The bank can't approve your loan."

After WordNet Lemmatization:

The bank ca n't approve your loan.

WordNet lemmatization reduces words to their base or dictionary forms based on their parts of speech (POS). In this example, it lemmatizes words like "approve" to "approve" (verb) and "loan" to "loan" (noun).

ii) Spacy Lemmatization:

Example:

Input Context = "The bank can't approve your loan."

After Spacy Lemmatization:

the bank can not approve -PRON- loan.

SpaCy lemmatizes words and retains some special characters, such as hyphens and apostrophes, to preserve the context of contractions and compound words. In this example, "can't" is lemmatized to "can not," and "-PRON-" represents a pronoun like "your."

iii) TextBlob Lemmatization:

Example:

Input Context = "The bank can't approve your loan."

After TextBlob Lemmatization:

The bank ca n't approve your loan

TextBlob lemmatizes words and removes contractions like "can't," reducing them to their base forms.

Lemmatization methods vary across libraries, leading to slightly different outcomes. The choice depends on NLP task specifics and text data characteristics. Compared to stemming, lemmatization preserves word senses, offering more meaningful base forms, ideal for tasks prioritizing sense preservation.

Advantage:

Maintains semantic meaning by reducing words to their base forms. Enhances context quality for WSD by capturing word essence.

Disadvantage:

Requires linguistic analysis and can be slower than stemming. It only handles irregular words perfectly.

6) Part-of-Speech Tagging:

Identifying the part of speech (noun, verb, adjective, etc.) of each word aids in understanding its role in the sentence and context.

Here, three different types of POS tagging techniques are applied:

i) NLTK POS Tagging:

Example:

Input Context = "The bank can't approve your loan."

After NLTK POS Tagging:

[('The', 'D.T.'), ('bank', 'N.N.'), ('ca', 'M.D.'), ("n't", 'R.B.'), ('approve', 'V.B.'), ('your', 'PRP\$'), ('loan', 'N.N.'), ('.', '.')]

In the output, each word is paired with its corresponding POS tag. For example, 'bank' is tagged as 'NN' (noun), 'ca' as 'MD' (modal verb), and so on.

ii) Spacy POS Tagging:

Example:

Input Context = "The bank can't approve your loan."

After Spacy POS Tagging:

[('The', 'DET'), ('bank', 'NOUN'), ('ca', 'AUX'), ("n't", 'PART'), ('approve', 'VERB'), ('your', 'DET'), ('loan', 'NOUN'), ('.', 'PUNCT')]

In this output, each token is labelled with its corresponding POS tag. For example, 'bank' is tagged as 'NOUN' (noun), 'ca' as 'AUX' (auxiliary verb), and so on. SpaCy provides detailed POS tagging information, making it a powerful tool for natural language processing tasks.

iii) TextBlob POS Tagging:

Example:

Input Context = "The bank can't approve your loan."

After TextBlob POS Tagging:

[('The', 'D.T.'), ('bank', 'N.N.'), ("can't", 'V.B.'), ('approve', 'V.B.'), ('your', 'PRP\$'), ('loan', 'N.N.')]

In this output, each word is paired with its corresponding POS tag. For example, 'bank' is tagged as 'NN' (noun), 'ca' as 'MD' (modal verb), and so on. TextBlob provides a simple interface for POS tagging in Python.

Various POS tagging techniques assign distinct labels to words based on their grammatical roles. The selection depends on the specific NLP task and preferred library. These tags play a crucial role in identifying word roles in sentences, benefiting applications like Word Sense Disambiguation in NLP.

Advantages:

Provides insights into the roles of words in sentences. It is helpful in understanding context and disambiguation.

Disadvantages:

Requires additional processing and resources. It can introduce errors in tagging accuracy.

7) Punctuation Removal:

Removing punctuation marks helps simplify text and focus on meaningful content for WSD.

Here, we apply three different punctuation removal techniques to the ambiguous statement:

i) Using String Punctuation:

Example:

Input Context = "The bank can't approve your loan."

Output is:

The bank cant approve your loan

ii) Using Regular Expressions:

Example:

Input Context = "The bank can't approve your loan."

Output is:

The bank cant approve your loan

iii) Using TextBlob:

Example:

Input Context = "The bank can't approve your loan."

Output is:

The bank cant approve your loan

Each punctuation removal technique provides similar results, removing punctuation marks from the original sentence. The choice of technique depends on your specific preference and the tools or libraries you use for text preprocessing.

Advantages: Removing punctuation can be helpful in text analysis tasks where punctuation is irrelevant, such as certain forms of sentiment or word frequency analysis.

8) Special Character Removal:

Eliminating non-essential special characters, such as numbers or symbols, reduces noise in WSD. This study applies three techniques for Special Character Removal—String Punctuation, List Comprehension, and Regular Expressions—to ambiguous statements.

Input Context = "The bank can't approve your loan."

Output is:

The bank cant approve your loan

Example: Input - "Text #123." After particular character removal: "Text"

All three techniques successfully eliminate special characters and punctuation, yielding a refined version of the original sentence. The method chosen depends on personal preference and the tools or libraries employed for text preprocessing. This process is advantageous when emphasizing words and reducing noise in text data.

Advantages:

Eliminates characters that might not contribute to disambiguation. Reduces noise and maintains cleaner context.

Disadvantages:

Might unintentionally remove characters that hold meaning, affecting disambiguation.

9) Context Window Strategies:

Context window strategies involve considering the words surrounding a target word within a fixed window size. The idea is to capture the context in which a word appears to understand its meaning.

Advantages:

i) Contextual Information:

Context window strategies provide valuable contextual information by considering nearby words. In the sentence, "bank" might be disambiguated based on nearby words like "approve" and "loan."

ii) Simplicity:

These strategies are relatively simple to implement, making them accessible for various NLP tasks.

iii) Local Coherence:

They can capture local word relationships and dependencies, which is valid for specific tasks like part-of-speech tagging. Disadvantages:

i) Limited Context:

Context window strategies have a narrow view of context due to the fixed window size. Words outside the window are not considered, potentially missing important context.

ii) Sparse Representations:

The resulting representations can be sparse, especially in large vocabularies, leading to inefficient storage and computation. iii) Word Order Sensitivity:

They are sensitive to word order, and slightly reordering words in a sentence can lead to different context representations.

iv) Polysemy Handling:

Context window strategies may struggle with polysemous words (words with multiple meanings) as they often do not provide enough context to disambiguate.

10) Word Embeddings:

Word embeddings, such as Word2Vec, GloVe, and fastText, are dense vector representations of words that capture semantic relationships based on large text corpora.

Advantages:

i) Semantic Similarity:

Word embeddings excel at capturing semantic relationships between words. Words with similar meanings have similar vector representations.

ii) Continuous Representations:

They offer straight and dense vector representations, which can capture fine-grained semantic nuances.

iii) Word Algebra:

Word embeddings allow for operations like word algebra, where vector arithmetic can find relationships between words (e.g., king - man + woman = queen).

iv) Dimension Reduction:

They reduce the dimensionality of the data, making them computationally efficient.

Disadvantages:

i) Fixed Vocabulary:

Word embeddings are often pre-trained on large corpora with fixed vocabularies. They may need to handle domain-specific or out-of-vocabulary words better.

ii) Lack of Transparency:

While they capture semantic relationships, it may sometimes be unclear why certain relationships exist in the embeddings. iii) Limited Context:

They provide word-level embeddings but do not capture contextual information directly. More advanced models like contextual embeddings (e.g., BERT) address this limitation.

iv) Resource-Intensive Pre-training:

Training word embeddings on large corpora requires significant computational resources.

V. EXPERIMENTAL SETUP:

Experiments for Word Sense Disambiguation using WordNet can be designed and executed with knowledge-based. Here's how the experiments might be structured: -

i) Data Collection:

Assemble a corpus with annotated instances requiring word sense disambiguation using benchmark datasets like SemCor or SensEval.

ii) Preprocessing:

Apply text preprocessing techniques, including Tokenization, lowercasing, stemming, and lemmatization, to prepare the text for analysis.

iii) Word Sense Annotation:

Manually annotate the dataset with appropriate senses from WordNet, assigning unique identifiers to each instance.

v) Feature Extraction:

Extract features from the annotated dataset, incorporating context words, part-of-speech tags, and sense definitions from WordNet.

vi) Sense Disambiguation:

Implement knowledge-based algorithms like Lesk, Extended Lesk, or Adapted Lesk. Calculate similarity scores between context words and WordNet sense definitions, selecting the sense with the highest similarity score.

vii) Evaluation:

Assess the knowledge-based method's performance using evaluation metrics such as accuracy, precision, recall, and F1-score. Compare disambiguated senses with ground truth annotations.

Preprocessing Technique	Conclusion			
Lemmatization	Lemmatization led to the highest accuracy among individual techniques, suggesting that capturing the base forms of words enhances context representation and improves sense disambiguation. [3 - 12]			
Stopword Removal	Stopword removal enhanced accuracy by eliminating noise, prioritizing content- bearing words, and potentially reducing lexical variability. [3-12]			
Lowercasing	Lowercasing had a minor positive impact, indicating that maintaining consistent case representation contributes to improved context matching. [3-12]			
Stemming	Stemming, although useful, may oversimplify and potentially lead to incorrect sense assignments. [3-12]			

VI. ANALYSIS:

Preprocessing Technique

Lemmatization Lemmatization led to the highest accuracy among individual techniques, suggesting that capturing the base forms of words enhances context representation and improves sense disambiguation. [3 - 12]

Stopword Removal Stopword removal enhanced accuracy by eliminating noise, prioritizing content-bearing words, and potentially reducing lexical variability. [3-12]

Lowercasing Lowercasing had a minor positive impact, indicating that maintaining consistent case representation contributes to improved context matching. [3-12]

Stemming Stemming, although useful, may oversimplify and potentially lead to incorrect sense assignments. [3-12]

VII. CONCLUSION:

Text preparation techniques significantly impact WordNet utilization in NLP for word sense disambiguation, enhancing the accuracy and efficiency of WSD algorithms. This study emphasizes the influence of preprocessing methods, including Tokenization, lowercasing, stop word removal, and stemming, on the quality of context representation and the disambiguation process. Preprocessing approaches contribute to creating a more focused context, reducing noise, and standardizing the text, while synonym expansion and POS tagging can enhance semantic relevance. The impact of preprocessing varies based on text nature, disambiguation algorithm, and WordNet data quality. Customizing preprocessing for specific tasks and domains leads to improved disambiguation results. The research concludes that careful selection and application of preprocessing techniques are crucial for maximizing WordNet-based word sense disambiguation in NLP, ultimately enhancing algorithm effectiveness across various tasks and domains.

REFERENCES:

- https://medium.com/@evertongomede/word-sense-disambiguation-resolving-ambiguity-in-natural-languageprocessing.
- P. Resnik, "WSD in NLP applications" in Word Sense Disambiguation, 2007, pp. 299-337. doi:10.1007/978-1-4020-4809-8_11.
- 3. N. Vasunthira Devi and Dr. R. Ponnusamy, "A systematic survey of natural language processing (NLP) approaches in different systems," IJCSE, vol. 4, no. 7, pp. 192-198, 2016.
- 4. Dr. A. S. N. Balasubramani et al., "Survey on PreProcessing Techniques for Text Mining", IJECS, vol. 5, no. 6, Jun. 2016, Page No. 16875-16879.
- 5. R. Satapathy et al., "Phonetic-based microtext normalization for Twitter sentiment analysis," IEEE International Conference on Data Mining Workshops, 2017. doi:10.1109/ICDMW.2017.59.
- 6. E. D. Trippe et al., "A brief survey of text mining: Classification, clustering and extraction techniques", Available at: arXiv:1707.02919v2 [cs.CL] 28 Jul 2017.
- M. Taher Pilehvar and J. Camacho-Collados, "On the role of text preprocessing in neural network architectures: An evaluation study on text categorization and sentiment analysis" in Proc. 2018 EMNLP Workshop Blackbox NLP: Analyzing and Interpreting Neural Networks for NLP, pp. 40-46.
- S. Singh, "Natural Language Processing for Information Extraction", Available at: arXiv:1807.02383v1 [cs.CL] 6 Jul 2018.
- 9. R. N. Waykole and A. D. Thakare, "A Review Of Feature Extraction Methods For Text Classification", IJAERD, vol. 5, no. 04, Apr., 2018.
- 10. S. P. Paramesh and K. S. Shreedhara, "IT Help Desk Incident Classification Using Classifier Ensembles", ICTACT J. Soft Comput., vol. 09, no. 04, Jul. 2019.
- 11. S. P. Paramesh and K. S. Shreedhara, "Building intelligent service desk systems using AI," IJESM, vol. 1, no. 2, pp. 01-08, 2019.
- 12. F. Al-Hawari and H. Barham, "A machine learning based help desk system for IT service management," J. King Saud Univ. Comput. Inf. Sci., vol. 33, no. 6, 702-718, 2021. doi:10.1016/j.jksuci.2019.04.001.

Sentiment Analysis of COVID-19 Tweets: A Supervised Machine Learning Approach with NLP and Feature Extraction for Polarity and Subjectivity Classification

¹Parth Aland, ²Shriram Raut, ³Ashok Shinde ¹Saint Thomas English Medium School, Solapur, India ^{2, 3} Department of Computer Applications, School of Computational Science, P.A.H. Solapur University Solapur

Abstract:The COVID-19 pandemic has significantly altered global communication patterns, with social media platforms becoming a vital channel for expressing emotions and opinions. This research investigates sentiment analysis on a dataset comprising 3090 tweets, capturing emotions such as sadness, anger, joy, and fear. Leveraging Natural Language Processing (NLP) techniques, the study employs supervised machine learning with various feature extraction methods, including TF-IDF, CountVectorizer, Bag-of-Words (BOW), and N-grams.

The research focuses on understanding sentiment through polarity and subjectivity classification. The collected tweets undergo rigorous data cleaning and normalization processes in preprocessing. Feature extraction techniques such as TF-IDF, CountVectorizer, BOW, and n-grams transform the textual data into numerical vectors, enabling machine learning algorithms to interpret and learn from the tweet content.

A range of supervised learning classifiers, including Support Vector Machines (SVM), Logistic Regression, K-Nearest Neighbour (KNN), Random Forest, Naive Bayes, and Decision Trees, are employed to predict sentiment based on the extracted features. The study evaluates the performance of each classifier by calculating precision, recall, F1-score, and accuracy metrics.

The outcomes of this research contribute to a nuanced understanding of sentiment in COVID-19-related tweets, shedding light on the prevalent emotions within social media discussions during the pandemic. Applying diverse feature extraction techniques and machine learning classifiers enhances the depth and accuracy of sentiment analysis, providing valuable insights into public sentiment dynamics surrounding the COVID-19 crisis.

Keywords: - Social media, Covid-19, Sentiment analysis, NLP, Data Preprocessing, Feature Extraction, Polarity, subjectivity, Supervised Learning classifiers

I. INTRODUCTION

The ongoing COVID-19 pandemic, a crisis swiftly permeating every corner of the world, has dramatically shaped the global landscape. Emerging in Wuhan, China, in December 2019, the World Health Organization (WHO) declared the outbreak a Public Health Emergency of International Concern on January 30, 2020, subsequently elevating it to pandemic status on March 11, 2020. The impact of the COVID-19 virus has been profound, with devastating consequences witnessed across countries such as the United States, Brazil, Russia, and India, resulting in significant loss of life [1].

As the pandemic persists, nations have grappled with implementing varying degrees of lockdown measures, fundamentally altering daily life for millions. The ubiquity of social media has played a pivotal role in shaping public discourse during these challenging times. COVID-19 has triggered transformative changes in health, business, and education globally, particularly affecting countries like the USA and India, where the virus spreads rapidly [2]. With over 215 countries affected by August 18, 2020, the pandemic has not only taken a toll on individuals' mental health. Still, it has also disrupted their physical wellbeing due to disruptions in daily routines.

In this rapidly evolving landscape, social media platforms, especially Twitter, have become instrumental in disseminating information, sharing opinions, and reflecting public sentiments about community events and health crises [4]. Government agencies worldwide leverage Twitter as a primary channel to communicate policy updates and vital information related to COVID-19, fostering real-time interactions with the public.

This study analyses tweets specifically related to the coronavirus, extracted from the information available on Twitter. The analysis aims to unveil genuine emotions experienced by individuals during the COVID-19 pandemic. Natural Language Processing (NLP), a vital field for comprehending and interpreting human-readable text using machine learning techniques, is employed. The Natural Language Toolkit (NLTK), a widely used open-source Python package, facilitates various NLP tasks, enabling interpreting and training emotions expressed in both English and Filipino tweets.

Before delving into sentiment analysis, it is crucial to understand the concept of Sentiment Analysis – a computational approach to categorize opinions expressed in text, determining whether the sentiments are positive, neutral, or negative. This paper classifies sentiments into distinct emotional categories, namely fear, joy, anger, and sadness, offering a nuanced understanding of the diverse emotional responses encountered during the COVID-19 pandemic.

II. RELATED WORK

Numerous studies have sought to comprehensively analyze the sentiments expressed in tweets related to the COVID-19 pandemic, employing various methodologies and focusing on distinct aspects of sentiment evaluation.

Rajput et al. conducted statistical evaluations on COVID-19 tweets from January 2020. Their approach involved two empirical studies, one centred on word frequency and the other on individual tweet sentiments. Modelling a power-law distribution, we analyzed the frequencies of unigrams, bigrams, and trigrams. Rajput et al. also employed evaluation metrics such as Sum of Square Error (SSE), R2, and Root Mean Square Error (RMSE) were employed. Their findings revealed a notable prevalence of tweets exhibiting positive polarity, with only 15% of tweets displaying a negative sentiment [4].

Samuel et al. delved into the overall sentiment of the public regarding the pandemic by analyzing COVID-19-related tweets. Utilizing R statistical software and sentiment analysis packages, the study employed descriptive analytics to showcase the heightened fear sentiment, particularly in the United States. Additionally, the authors introduced two primary machine learning classification algorithms, Naive Bayes and Logistic Regression, providing insights into their effectiveness for classifying tweets of varying lengths. Naive Bayes demonstrated an accuracy of 0.9143 for shorter tweets and 0.5714 for longer ones, while Logistic Regression achieved 0.7429 accuracy for shorter tweets and 0.52 for extended tweets [5].

Barkur et al. focused on analyzing Indian sentiments regarding lockdown announcements. Extracting tweets using hashtags #IndiaLockdown and #IndiafightsCorona between March 25 and March 28, 2020, the study encompassed 24,000 tweets. Results indicated a positive response from Indians towards the COVID-19 lockdown measures, with a majority expressing support for the government's efforts to flatten the curve [6].

Jang et al. conducted an in-depth investigation into people's concerns and reactions about COVID-19 in North America, particularly emphasizing Canada. Topic aspect-based sentiment analysis correlated the topics discussed in tweets with the timing of public health interventions. The sentiment analysis shed light on public sentiments surrounding various COVID-19-related issues, providing valuable insights for public health agencies when formulating intervention policies [7].

Zhou et al. explored the sentiment dynamics of individuals residing in New South Wales (NSW), Australia, during the lockdown period. Their study extended to examining the sentimental nuances related to trending topics on Twitter, including lockdown, social distancing, and Australia's Job Keeper program [8].

III. PROPOSED MODEL

3.1 Data Collection:

The dataset used in this research, obtained from Kaggle and named "COVID-19 Sentimental Analysis," comprises 3,090 tweets collected between June 21, 2020, and July 20, 2020. This dataset reflects public sentiments during a critical period of the COVID-19 pandemic, providing valuable insights into diverse perspectives and emotions. Each tweet within the dataset represents an individual record, forming a rich source for sentiment analysis through supervised machine learning, natural language processing (NLP), and feature extraction techniques.

3.2 Data Preprocessing:

- Data preprocessing is vital for improving raw data quality by Cleaning and organizing it, ensuring accuracy and reliability in results. It significantly enhances machine learning model performance, leading to more accurate predictions and valuable insights. This process involves noise reduction, eliminating irrelevant information and inconsistencies, resulting in a cleaner dataset for analysis—additionally, data preprocessing addresses missing values, ensuring a complete and robust dataset for effective modelling. Transforming unstructured data into a structured format makes it compatible with diverse analytical techniques and algorithms, enhancing overall data usability and interpretability [9].
- Usernames (@user123): Use regular expressions to identify and remove patterns starting with '@'.
- Links and URLs: Employ regular expressions to identify and remove patterns starting with 'http', 'www', or 'https'.
- Hashtags (#example):
- Use regular expressions to identify and remove patterns starting with '#'.
- Emoticons: Utilize regular expressions to identify and remove standard emoticons such as :) Or: D.
- Special Characters: Apply regular expressions to remove non-alphanumeric characters and keep only letters.
- Stop Words: Use NLTK's stopwords list to remove common stop words like "the," "and," "is," etc.
- Hyperlinks: Similar to removing links and URLs, use regular expressions to identify and remove hyperlinks.
- Stemming: Stemming reduces words to their base or root form by removing suffixes. Applying stemming to the word "running" would result in "run."

• Lemmatization: Lemmatization is the process of reducing words to their base or root form by considering the context and meaning of the word. Applying lemmatization to the word "running" would result in "run" since it considers the context and understands that "running" is a form of "run."

3.3Data Cleaning:

Reading data for analysis involves handling incomplete data within a dataset. It is achieved by utilizing the dropna function to eliminate missing and null values, resulting in a cleaned dataset called cleansed data.



Fig.1: Before Preprocessing



Fig. 2: After Preprocessing

3.4Feature Extraction:

In sentiment analysis, feature extraction involves converting text data into numerical features that can be used as input for machine learning models to predict sentiment. They help capture the essence of the sentiment expressed in the text, enabling models to learn patterns and make predictions about the sentiment of a given piece of text. The commonly used feature extraction techniques in sentiment analysis are:

1) Count Vectorizer:

Count Vectorizer converts text into a matrix of word counts. A vector represents each document, with each element indicating the count of a particular word. It captures the frequency of words in a document, which can be helpful for sentiment analysis as certain words may indicate positive or negative sentiment.

2) TF-IDF (Term Frequency-Inverse Document Frequency):

TF-IDF assigns weights to words based on their importance in a document and across the entire dataset. It considers both the term frequency and the inverse document frequency.TF-IDF helps highlight words that are both frequent in a document and unique to that document, aiding in identifying significant terms related to sentiment.

3) N-gram:

N-grams are sequences of n words/tokens. They capture local relationships between words, allowing the model to understand the context. By considering the order of words, n-grams can capture nuanced expressions, helping sentiment analysis models understand the context in which certain words or phrases appear.

4) Bag of Words (BOW):

BOW represents a document as an unordered set of words, disregarding grammar and word order. It creates a numerical vector based on the presence or absence of words. BOW simplifies text to its essential components, allowing sentiment analysis models to focus on the presence or absence of specific words without being influenced by their order.

IV. METHODOLOGY

4.1 Supervised learning approach:

It mandates the presence of a labelled dataset for training purposes. This learning paradigm typically involves a two-step process: training and testing the model [10]. Throughout the training phase, the model refines its understanding of the provided data, discerning underlying patterns and categorizing tweets. The model is then employed to predict class labels for the test data. Ultimately, the model's accuracy is assessed based on its performance with the test dataset. This approach encompasses two primary categories: classification and regression. Classification methods assist in identifying suitable class labels for forecasting positive, neutral, and negative sentiments. The algorithms utilized in this methodology include Logistic Regression, Random Forest, Naive Bayes, Support Vector Machine, Decision Tree, and KNN.

1) Logistic Regression:

Logistic regression, a supervised learning algorithm, is employed for predicting the value of a target variable and is primarily used in classification tasks. This algorithm provides a binary classification output, indicating the probability of a specific scenario falling between 0 and 1. Data training is the initial step in logistic regression, and its accuracy is determined using equations derived from the trained data [13].

2) Naive Bayes:

Naive Bayes, a classification method renowned for its simplicity and efficacy, is the pioneer algorithm for text classification challenges. Not only is Naive Bayes known for its quick model-building capabilities, but it also enables swift predictions [14]. Functioning as a probabilistic classifier, it predicts based on the probability of an object, determined by a specific equation [15,16].

3) Random Forest:

Random forest, a supervised learning algorithm for classification and regression, operates as a bagging technique. As decision trees serve as the base model, we generated numerous decision trees from subsets of rows and columns within the original dataset. By consolidating predictions through majority voting, we determine the ultimate decision based on the combined input from all decision trees. Employing multiple trees enhances accuracy and mitigates overfitting risks [5].

4) Support Vector Machine:

Support Vector Machine (SVM), a supervised machine learning algorithm, finds application in regression and classification models, primarily focusing on classification [11]. By creating a hyperplane, SVM effectively separates different classes in classification problems. The algorithm establishes two margin lines, ensuring linear separability for classification points. The distance between these margin lines, termed Marginal Distance, influences the model's generalization. Support vectors are crucial points passing through the marginal plane [12]. In cases of non-linear separability, SVM utilizes support vector kernels to transform low-dimensional data into higher dimensions, offering efficiency and high accuracy with minimal computational requirements.

5) Decision Tree:

A Decision Tree is a tree-like model that makes decisions based on the features of the input data. Each internal node of the tree represents a decision based on a specific feature, and each leaf node represents the final predicted outcome. In sentiment analysis, analysts commonly use Decision Trees for classification tasks. Decision Trees can analyze different text features, such as word frequencies or combinations of words, to make decisions about the sentiment expressed. It can capture non-linear relationships and be easily interpretable, making it helpful in understanding the key features influencing sentiment in a text.

6) K-Nearest Neighbors (KNN):

K-Nearest Neighbors (KNN) is a simple and intuitive algorithm that classifies a data point based on the majority class of its k nearest neighbours in the feature space. In the context of sentiment analysis, analysts consider these neighbours as texts most similar to those analyzed. KNN considers the proximity of a given text to its neighbouring texts in the feature space. If a text

is close to others expressing positive sentiment, it is likely assigned a positive sentiment label. This method is effective when similar texts tend to have similar sentiments.

4.2 Natural Language Processing

Natural Language Processing (NLP) refers to developing software and services capable of understanding human languages. Examples of NLP in action include speech recognition, such as in Google Voice Search, and tasks like comprehending content and sentiment analysis. An effective way to illustrate NLP processes is by working through fundamental procedures using the Natural Language Toolkit (NLTK).

NLTK, a widely utilized NLP software created in Python, is designed to handle human language data. It encompasses text processing functionalities like classification, tokenization, and lemmatization and user-friendly interfaces like WordNet.

4.3 Sentiment analysis

Sentiment analysis is a systematic approach to analyzing textual data using NLP algorithms. Techniques like TextBlob Sentiment Analysis can reveal concealed information within collected and stored data, indicating the positivity or negativity of the text. This analysis has practical applications, aiding companies in developing customer-centric strategies. With advancements in artificial intelligence algorithms, handling and studying textual data has become more efficient, achieving high accuracy rates in sentiment assumptions. Social media platforms like Facebook and Twitter use sentiment analysis to combat spreading fake and harmful news [17].

Sentiment analysis with TextBlob:

In Python, various packages facilitate sentiment analysis using different methods. One notable approach involves using TextBlob, a text processing package that offers an API for everyday NLP tasks. For a given input sentence, TextBlob provides two properties:

• Polarity: Determines the emotional polarity of the statement, ranging from -1 (indicating negative sentiment) to +1 (indicating positive sentiment).

• Subjectivity: Indicates the speaker's states, such as emotions, beliefs, and opinions. It ranges from 0 (objective and based on facts) to 1 (subjective).



V. EVALUATION AND DISCUSSION

Fig. 3: Sentiment Distribution







Fig. 5: Classifier Performance Metrics – TF-IDF



Fig. 6: Classifier Performance Metrics – Bag of Words (BoW)



Fig. 7: Classifier Performance Metrics - CountVectorizer



Fig. 8: Classifier Performance Metrics - N-gram

The performance of the classifiers varies depending on the feature extraction technique used. Naïve Bayes performs well across different feature extraction methods, while Logistic Regression and SVM also show competitive performance. Decision Tree and KNN tend to have lower accuracies than the other classifiers. It is essential to consider the specific problem context and further evaluate the models using appropriate evaluation metrics before deciding on the most suitable combination of feature extraction and classifier.



Fig 9: Emotion Counts and Percentages

The provided dataset contains a distribution of emotions where fear, sadness, anger, and joy are all present. The frequencies and percentages suggest fear and sadness are more prevalent than anger and joy.



Fig 10: Sentiment Counts and Percentages

The provided dataset shows a prevalence of positive polarity, with most instances falling into the Positive category. Negative polarity is the second most common, and neutral polarity has the lowest occurrence. It indicates an overall positive sentiment or sentiment orientation within the dataset.



Fig. 11: Total Count and Accuracy of Polarity

The overall accuracy of polarity is 28.90%, which suggests that the models or methods used to classify the polarity in the dataset have achieved an accuracy of 28.90% in predicting the correct polarity labels. It is important to note that the accuracy alone may not provide a comprehensive understanding of the performance of the models, as additional evaluation metrics and context are needed to assess the effectiveness of the polarity classification.



Fig. 12: Percentage of Emotions Based on Sentiments

The provided sentiment distributions indicate varying patterns for each emotion. Sad and fear have a relatively balanced distribution of sentiments, with negative sentiment slightly more prevalent. Anger shows a higher prevalence of negative sentiment, while joy exhibits a substantial prevalence of positive sentiment. These distributions provide insights into the sentiment composition for each emotion in the dataset.

VI. CONCLUSION

The paper focuses on sentiment analysis of COVID-19-related tweets, classifying them into the anger, fear, joy, and Sadness categories. Supervised machine learning algorithms, including Logistic Regression, Naive Bayes, Random Forest, KNN, Decision Tree and Support Vector Machine, were implemented and trained on the dataset. The performance of classifiers varies based on the feature extraction technique. Naïve Bayes consistently performs well, while Logistic Regression and SVM show competitive results. Decision Tree and KNN tend to have lower accuracies. The dataset contains a distribution of emotions, with fear and sadness slightly more prevalent than anger and joy. Positive polarity is predominant, followed by negative, and neutral has the lowest occurrence. Overall accuracy for polarity is 28.90%, indicating moderate success. Sentiment distributions reveal nuanced patterns across emotions, with sadness and fear having balanced sentiments, anger skewed negative, and joy strongly positive. It's crucial to consider evaluation metrics and context for a comprehensive assessment.

REFERENCES

- 1. Kausar, Mohammad Abu, Arockiasamy Soosaimanickam, and Mohammad Nasar. "Public Sentiment Analysis on Twitter Data during COVID-19 Outbreak."
- S.Banik, A. Ghosh, S. Banik and A. Mukherjee, "Classification of COVID-19 Tweets based on Sentimental Analysis," 2021 International Conference on Computer Communication and Informatics (ICCCI), 2021, pp. 1-7, doi: 10.1109/ICCCI50826.2021.9402540.
- 3. Rufai, S. R., & Bunce, C. (2020). World leaders' usage of Twitter in response to the COVID-19 pandemic: a content analysis. Journal of Public Health.
- 4. Rajput, Nikhil Kumar, Bhavya Ahuja Grover, and Vipin Kumar Rathi."Word frequency and sentiment analysis of Twitter messages during Coronavirus pandemic." arXiv preprint arXiv:2004.03925 (2020).
- 5. Samuel, Jim, et al. "Covid-19 public sentiment insights and machine learning for tweets classification." Information 11.6 (2020): 314
- 6. Barkur, Gopalkrishna, and Giridhar B. Kamath Vibha. "Sentiment analysis of nationwide lockdown due to COVID-19 outbreak: Evidence from India." Asian journal of psychiatry (2020).
- 7. Jang, Hyeju, et al. "Exploratory analysis of COVID-19 related tweets in North America to inform public health institutes." arXiv preprintarXiv:2007.02452 (2020).
- 8. Agarwal, A., Xie, B., Vovsha, I., Rambow, O., & Passonneau, R. J. (2011). Sentiment analysis of Twitter data. In: Proceedings of the workshop on language in social media (LSM 2011) (pp. 30-38).
- 9. Balahur, A. (2013). Sentiment analysis in social media texts. In: Proceedings of the 4th workshop on computational approaches to subjectivity, sentiment and social media analysis (pp. 120–128).
- Zhang, L., Ghosh, R., Dekhil, M., Hsu, M., & Liu, B. (2015). Combining lexicon-based and learning-based methods for Twitter sentiment analysis. International Journal of Electronics, Communication and Soft Computing Science & Engineering (IJECSCSE), 89, 1–8.
- 11. Chapelle, O.; Haffner, P.; Vapnik, V.N. Support Vector Machines for Histogram-Based Image Classification. IEEE Trans. Neural Netw. 2018, 10, 1055–1064.
- 12. Vapnik, V. The Nature of Statistical Learning Theory; Springer: New York, NY, USA, 1995.
- 13. Asri, H.; Mousannif, H.; Al Moatassime, H.; Noel, T. Using Machine Learning Algorithms for Breast Cancer Risk Prediction and Diagnosis. Procedia Comput. Sci. 2016, 83, 1064–1069.
- Sisodia, D.; Sisodia, D.S. Prediction of Diabetes using Classification Algorithms. Procedia Comput. Sci. 2018, 132, 1578–1585.
- 15. Rahman, A.S.; Shamrat, F.J.M.; Tasnim, Z.; Roy, J.; Hosain, S.A. A Comparative Study On Liver Disease Prediction Using Supervised Machine Learning Algorithms. Int. J. Sci. Technol. Res. 2019, 8, 419–422.
- 16. Bansal, D.; Chhikara, R.; Khanna, K.; Goopta, P. Comparative Analysis of Various Machine Learning Algorithms for Detecting Dementia Detecting Dementia. Procedia Comput. Sci. 2018, 132, 1497–1502.
- Abiola, O., Abayomi-Alli, A., Tale, O.A. et al. Sentiment analysis of COVID-19 tweets from selected hashtags in Nigeria using VADER and Text Blob analyzer. Journal of Electrical Systems and Inf Technol 10, 5 (2023). https://doi.org/10.1186/s43067-023-00070-9

Gas Detection And Safety Management For Underground Mining Using Arduino

¹Ashish V Biradar ²Shubha S

^{1,2}Dept of Electrical and Electronics Engineering Guru Nanak Dev Engineering College Bidar

Abstract : For underground mining and chemical industries the safety management is very important to safeguard the workers' health and life risk condition. Underground mining poses a unique set of health and safety risks compared to other industries. The subterranean environment naturally contains some risks, such as ground instability. Others are introduced through intricate mining procedures and activities, which may introduce risks from mobile equipment, such as big cars that could impede the driver's view, into the subterranean environment. Hazards may increase as mines in Ontario get deeper and more expansive. If these hazards are not managed properly using appropriate controls, they can result in serious traumatic injuries, death or occupational illness. The existing methods do not provide the higher end security and monitoring because of the manpower needs the system is not so accurate and stable. The method used here is implementation of automation for such kind of industries where very less manpower and much accuracy in the operation. The system uses sensors and transmitter receiver units which execute the overall operation.

Keywords: Mining, Hazard, Safety management.

I. INTRODUCTION

The sensors which are placed inside the helmets of workers goes on detecting the abnormal parameters. The controlled signals from every sensor are connected to the transmitter unit which sends the abnormal signal to receiving unit which is fixed in the control room. The control room receiver receives the data and send it controller which indicates the status of sensors through LEDS. The control room person now starts generating one more wireless signal towards the underground mine plant and receiver at mining side decodes the data and turns the oxygen release container to generate the oxygen. Generating the voice announcement for the workers is possible with the same technology using the fixed unit at mining inside and a predefined voice starts to announce through the speakers as a supportive alert messages and safety alert steps to be taken by the workers which avoids a panic conditions inside the mine. The transmitter and receiver used here is amplitude shift keying (ASK) modulation modules are connected inside the workers helmet only and all the sensors are interfaced to the encoder which receives the signals at the time of abnormal parameter and starts to send the data at the frequency of 433 Mhz.

II. LITERATURE SURVEY:

In this work they describes as in recent days coal mining has been very dangerous activity for the workers in mine. Because of numbers of adverse effects an the environment, Like methane may be released into the air also underground mining hazards. Include suffocation, gas poisoning, roof collapse and gas explosion by keeping all activity which done in underground coal mine we designed a system i.e. zig bee based intelligent helmet for coal miners. For monitoring hazard gases, temperature condition and humidity. Level in air the improve safety features in our system dramatically increases life expectancy by alerting them about hazards. In this system, the helmet has three sensors temperature sensors, humidity sensor and gas sensor to monitor the condition in coal mining. If there is any hazardous condition occurs the buzzer get alarm which is positioned on helmet then gives the information to control station through the zig bee transreceiver. So that miners have the chance to rescue his life from the hazards occurs in coal mines. Also they conclude that, The main purpose to design this project is safety of person who work in coal mine. We can give assurance about the safety of person who are working in coal mine. In future this person who work in coal mine can easily identify the various gases, temp. Or about sudden short coming natural accidents which happens generally in coal mine. that In order to create a safety system to monitor the ambient characteristics of the mining environment related to the health and safety of mine workers and mine safety systems, a prototype for a mine safety system employing a wireless sensor network is constructed. Next, the prototype system's subsystems are simulated. The primary processing unit of the hardware was a microcontroller, which is found in electronic circuits. Furthermore, Gas and Dust sensors can be added to enhance the security system for the mine workers. When the project was completed, a comprehensive mine safety system was built utilizing a combination of electrical, mechanical, and specialized software. The system is compact and modular.[2]
III. BLOCK DIAGRAM



Fig 1:. Block diagram of Gas detection and safety management system

Smoke Sensors :

Smoke sensor is constructed with LED and photo diode combination which are in line of sight of the Smoke appears between this combination then the IR beam fall on photodiode also goes to vary which changes the conduction of the photodiode and now it compares in LM 58 comparator and the logical signal now connect to transmitter and encoder for further action.

Flame Or Fire Sensor

A flame detector is a type of sensor that is intended to identify and react to flames or fires. Depending on the installation, there may be several ways to respond to a detected flame. These include turning on a fire suppression system, setting off an alarm, and cutting off a fuel line (such as a propane or natural gas line).

There are different types of flame detection methods. Some of them are: Ultraviolet detector, near IR array detector, infrared (IR) detector, Infrared thermal cameras, UV/IR detector etc.

When fire burns it emits a small amount of Infra-red light, this light will be received by the Photodiode (IR receiver) on the sensor module. Then we use an Op-Amp to check for change in voltage across the IR Receiver, so that if a fire is detected the output pin (DO) will give 0 V(LOW) and if the is no fire the output pin will be 5V(HIGH).

Temperature Sensing Unit

The temperature sensing circuit is designed with Thermistor, OP_AMP, which sense the working temperature of pump. The Thermistor is used as a "thermal sensitive resistor". This Thermistor resistance is very high at normal temperature. Here the OP_AMP is used as a voltage comparator. As soon as temperature increases its resistance decreases which increases the voltage at Pin No 3 i.e. non-inverting terminal of the OP_AMP. Now because of this condition the potential difference between two inputs at comparator also changes and the output of the comparator goes from its low to high state to activate (Saturate) the transistor. The collector of the transistor further drives the signal to microcontroller. As long as the temperature is maintained high the OP_AMP output remains in the same state. When pump temperature falls down on Thermistor, its resistance goes to increase. This decrease the voltage at Pin No 3. Because of this condition the OP_AMP i.e. comparator output changes from its high to low state.

Methane Gas Sensor

The methane gas sensor which is placed in the helmet goes to detect the leakage of the gases. The sensor used here is MQ6 which convert gas leakage sensing into corresponding voltages which is compared in the comparator which is designed with op amp LM 358. If the leakage level exceeds the safe limit then the comparator o/p goes high which is going to connect to transmitter for further action.

Water Detector

If any of the reason the water enters in to the mines entrance area it cause a serious problem and lead to life risk for workers. The moisture detectors detects the water flow the sensor designed with moisture Probe and connected to an op amp circuit which turns the logical signal to the transmitting unit.

Soil Collapse Detector

The mining industry majorly suffering with soil collapse problem. Vibration sensor module which has vibration switch SW-420 and Comparator LM 393 to detect if there is any vibration that beyond the threshold. The threshold can adjust using an onboard potentiometer. When there no vibration, this module sends digital output logic LOW the signal indicates LED light, and vice versa. If the module does not vibrate, the vibration switch was closed on state, the output of low output, the green indicator light. The product vibrates, vibration switches momentary disconnect, the output is driven high, the green light does not shine.

The output can be directly connected to the microcontroller, which to detect high and low level, so as to detect whether the environment exists vibration, play a role in the alarm. It is useful for a variety of shocks triggering, theft alarm, smart car, an earthquake alarm, motorcycle alarm. This module when compared with a normally open pneumatic shock sensor module, shock triggered much longer can drive relay module the use of the company's production of SW-420 normally closed type vibration sensors. Comparator output signal clean wave well, driving ability, 15 m a rated voltage and 3.3V-5V output: digital switching output (0 and 1)

Power Supply

The required DC input voltage for the regulator is obtained from a step-down transformer with rectifier and filter capacitor. The capacitor C avoids the noise the output voltage. A regulated power supply can be designed with a different methods like using a transistors, zener diodes.But we are using fixed voltage regulators (7805) for constant voltage levels and these voltages connected to all the blocks. AA regulated power supply is a DC power supply system that maintains a consistent voltage regardless of variations in the load or the primary source. A fixed positive voltage regulator was indicated by the 7805 IC., which provided fixed voltage 5volts.the 7805 regulator is known as a fixed voltage regulator.

Fixed –Voltage regulator design has been greatly simplified by the introduction of 3-terminal regulator ics such as the 78xx series of positive regulators and the 79xxx series of negative regulators, which incorporate the features such as built-in fold back current limiting and thermal protection, etc. These ics are available with a variety of current and output voltages ratings, as indicated by the 'xxx' suffix; current ratings are indicated by the first part of the suffix and the voltage ratings by the last two parts of the suffix. Thus, a 7805 device gives a 5V positive output at a 1mA rating, and a 79L15 device gives a 15V negative output at a 100mA rating.

3-terminal regulators are very easy to use. The regulators ICs typically give about 60dB of ripple rejection, so 1V of input ripple appears as a mere 1mV of ripple on the regulated output. A rectified filter and unregulated DC voltage is given to pin of IC regulator. A bypass capacitor is connected between input and ground to bypass the ripples and oscillations. The output

capacitor is connected between output and ground to improve transient response. The unregulated input is applied to the IC must be always more than the regulated output.

7805 Regulated Power Supply

For any of the electronic circuits the voltage required is very less and should be fixed dc voltage. So in order to obtain required fixed DC voltage we use IC 7805.



Fig 2 7805 voltage regulator

Arduino Uno

For the purpose of creating digital devices, Arduino is an open-source hardware and software corporation, project, and user community that creates and produces single-board microcontrollers and microcontroller kits.



Fig 3:. Ardiuno Uno

An assortment of microprocessors and controllers are used in Arduino board designs. Sets of digital and analog input/output (I/O) pins on the boards allow them to be interfaced with different expansion boards, sometimes known as "shields," or breadboards for prototyping and other circuits. The boards have serial communications interfaces, some of which support USB (Universal Serial Bus), which is also used to load software from personal computers. The programming languages C and C++ can be used to create programs for the microcontrollers.

Arduino Uno is a microcontroller board based on the ATmega328P It has 14 digital input/output pins (of which 6 can be used as PWM outputs), using pin Mode (),digital Write(), and digital Read() functions. They operate at 5 volts. Each pin can provide or receive 20 mA as recommended operating condition and has an internal pull-up resistor (disconnected by default) of 20-50k ohm. A maximum of 40mA is the value that must not be exceeded on any I/O pin to avoid permanent damage to the microcontroller,6 analog inputs, a 16 MHz ceramic resonator (CSTCE16M0V53-R0), a USB connection, a power jack, an ICSP header and a reset button. It contains everything needed to support the microcontroller; simply connect it to a computer with a USB cable or power it with a AC-to-DC adapter or battery to get started The 14 digital input/output pins in Arduino programming can be used as input or output pins by using the pinMode(), digitalRead(), and digitalWrite() routines. An internal 20–50 KOhm pull-up resistor, operating at 5 V and capable of sending or receiving a maximum of 40 mA of current by default, is present in each pin. Some of these 14 pins have particular purposes, which are detailed below: 0 (Rx) and 1 (Tx) on the serial pins: TTL serial data is received and transmitted via the Rx and Tx pins. The related ATmega328P USB to TTL

serial chip is used to connect them.External Interrupt Pins 2 and 3: These pins can be configured to trigger an interrupt on a low value, a rising or falling edge, or a change in value.

- PWM Pins 3, 5, 6, 9, and 11: By utilizing the analog Write() method, these pins produce an 8-bit PWM output.
- SPI Pins 10 (SS), 11 (MOSI), 12 (MISO), and 13 (SCK): SPI communication uses these pins.
- Six analog input pins, which together with the 14 digital pins, offer 10 bits of resolution, or 1024 distinct values. They have a measurement range of 0 to 5 volts, but you can go above this by utilizing the analog Reference() function on the AREF pin.
- Arduino Uno has a couple of other pins as explained below:
- AREF: Used to provide reference voltage for analog inputs with analog Reference function.
- Reset Pin: Making this pin LOW, resets the microcontroller.

IV. Conclusion

We can shield a lot of workers from different mine-related accidents with the aid of this technique. It is possible to prevent or minimize the amount of fatalities that occur in mines as a result of numerous mishaps, such as abrupt increases in water levels, inhaling hazardous gasses like methane, and soil collapse. Long before any emergency situations arise, the personnel are relocated to a safe location. As a result, worker safety can be increased, which will lower the number of fatalities.

REFERENCES

- 1. An IEEE paper titled "Zig-Bee Based Intelligent Helmet For Coal Miners" By Prof. Arun Katara, Prof. Prof. Anand Dandle, Prof. Abhilesh Chore and Prof. Anurag Bhandarwar from EC engineering Department, DMIETR
- 2. An IEEE Titled "Mine Safety System Using Wireless Sensor Network" by VALDO HENRIQUES AND REZA MALEKIAN, (Member, IEEE)
- 3. A paper given by Chemical and Biological Engineering, the University of British Columbia, Vancouver, BC, Canada Titled as "Sensor Systems in Mining Industry"
- 4. The thesis by Sumit Kumar Srivastava (M.tech) on "Real Time Monitoring System for Mine Safety using Wireless Sensor Network (Multi-Gas Detector)"
- 5. Srideep Ghosh, Somprakash Bandyopadhyay, Madhuparna Pal, "Use of information and communication technologies for mining disasters", Conference on Mineral Industry in the New Economy: Challenges and Opportunities, January, 2009, Kolkata, India.
- 6. X.B. Jiang, ZigBee technology and its applications [J]. Low-voltage apparatus, 2005 (7).
- H.M. Zhu and L. Zhang, The Design of Warp Tension Embedded Intelligent Control System Based on μC/OS and ARM [J]. Mechanical & Electrical Engineering Magazine, 2006 (4).
- 8. X.M. Wang and N.A. Liu, 2.4GHz RF Chip CC2420 and Its Application in ZigBee Communication [J]. International Electronic Elements, 2005 (3).
- 9. X. Zhang, X.J. Qiao and E. Liu, Design of High Precision Instrument for Measuring Temperature, Humidity & Dew Point [J]. Instrument Technique and Sensor, 2006

Soldier's Uniform Used for Temperature Control and Health Monotoring System Based on Iot

¹Suresh patil

¹Dept of electrical and electronics engineering Guru Nanak Dev Engineering College Bidar

Abstract The E-Uniform powered by solar energy provides superior protection for the 1 soldier operating in harsh weather conditions. Sun The E-uniform's internal circuitry is powered by panels to create electricity, which may subsequently be utilized for the energy is stored in a lead acid rechargeable battery with a 12 V DC voltage. Also, we are powering the circuitry with a traditional battery charging mechanism The Arduino microcontroller, which manages every function in the circuit, is its brains An ADC 0808 is used to interface a voltage sampler with the system in order to acquire the voltage generated from the battery and display it on a 16X2 LCD. The project has two operating modes: summer and winter. By selecting the proper working mode for the H-Bridge IC, we may drive the body heater/cooler. Consequently, the heater/cooler will help us generate a warming or cooling effect within the uniform, allowing the soldier to endure any kind of outside weather. When the metal sensor finds a metal object that resembles a bomb, the soldier will be notified by a buzzer. The IoT server is interfaced with via the microcontroller, and the whole soldier's health status is tracked. Additionally, the concerned person continuously monitors using IoT, and the soldier's current status is continuously updated in the IoT server. The soldier will be able to operate in any type of environment thanks to this uniform.

Keywords: E-Uniform, IoT, lead acid rechargeable battery

I. INTRODUCTION

The Army's soldiers are its most precious asset. In order to defend one's country, soldiers are necessary. All members of the Army, Air Force, Navy, and Marines are referred to as soldiers, regardless of gender. They will always take on and complete the duty in the event of inclement weather at any time of the year. Severe heatwaves or freezing snaps could pose challenges for national security personnel. One's health may be harmed by extreme cold or heat. This project's objective is to develop an E-Uniform that will provide soldiers operating in inclement weather with superior protection. Two modes-summer mode and winter mode—are provided in this work. The relays drive the body heater/cooler by choosing the mode of operation. In turn, the heater/cooler will enable us to create a warming or cooling effect within the uniform, enabling the soldier to operate well without experiencing heat or cold stress in any type of exterior environment. Soldiers labouring in hot environments may benefit from solarpowered e-uniforms, which could be a major advancement in military apparel. Solar panels would be woven into the fabric of these uniforms to produce electricity, which could then be utilised to operate a number of functions like heating or cooling components, lights, and communication gadgets. An environmentally responsible and more sustainable alternative to conventional battery-powered devices could be solar electricity. By eliminating the need for extra batteries, the uniforms may help lighten the load that soldiers must carry. The solar-powered e-uniforms could have cooling devices to control the soldier's body temperature and avert heat stroke in extremely hot weather. The clothes could include heating components to keep soldiers comfortable and prevent frostbite in cold weather .Maintaining body temperature, communicating with their team, and carrying large goods are just a few of the difficulties military troops serving in harsh weather encounter. These problems might be resolved by solar-powered e-uniforms, which could make soldiers more comfortable and enhance their performance in combat. Solar panels would be woven into the fabric of these e-uniforms to produce electricity, which could then be utilised to operate a number of functions like heating or cooling elements, lights, and communication gadgets. The solar-powered e-uniforms would offer a long-lasting and eco-friendly replacement for conventional battery-powered gadgets. The solar-powered euniforms could have cooling devices to control the soldier's body temperature and avert heat stroke in extremely hot weather. When it's freezing outside, the uniforms. A novel approach to the problems faced by military troops who operate in hostile situations is the use of solar-powered euniforms for soldiers working in high temperatures. Through the provision of a viable and effective solution, the incorporation of solar electricity into these uniforms has the potential to enhance the performance, comfort, and safety of soldiers.

II. LITERATURE SURVEY

Dipali H. Kale's Solar Based E-Uniform for Soldiers. Soldiers operating in inclement weather are better protected by solarpowered E-uniforms. Solar panels are often used to power the E-uniform's internal circuitry. The energy is stored in a lead- acid rechargeable battery with a voltage of 12 V DC. We are also powering the circuits with a traditional battery charging mechanism. The circuit's ATmega16a microcontroller, which manages every function, is its brains. There are two

modes of operation for the project: summer mode and winter mode. We are configuring the system to drive the body temperature in accordance with the heater/cooler by choosing the mode of operation. In turn, the heater/cooler will assist us in creating a warming or cooling effect inside the uniform, enabling the soldier to withstand any type of external environment. Solar- powered troops' e-uniform Professor Assistant Sridevi S.H. Soldiers who operate in inclement weather are better protected by solar-powered E-Uniform. The E uniform's interior electronics is powered by solar panels. A lead acid rechargeable battery with a 12VDC voltage is used to store the energy. As a charging unit, a regular battery can alternatively be employed. The LPC2148 microcontroller is in charge of all operations. A voltage sampler is 14 interfaced with the system via ADC to collect the voltage generated from the battery for a display on a 16X2 LCD. In this project, we developed an Euniform that offers soldiers more protection when they work in bad weather. This uniform will enable the soldier to function in any kind of setting. The circuit, which may be used in both summer and winter modes, is installed to a jacket. We activate the H-Bridge IC to drive the body heater/cooler by choosing the appropriate operating mode. In turn, the heater/cooler will assist us in creating a warming or cooling effect inside the uniform, enabling the soldier to withstand any type of external environment. E-jacket for soldiers: Rahul Khairamode. In every year we are facing several patterns of climates, the summer season, rainy season and winter seasons are the main. Temperatures that are very high and very low are both hazardous to health. "Too much exposure to heat is called heat stress as well as excessive too much cold is called "cold stress". At a very high temperature the most serious problem is heatstroke. In extreme cold, the most serious trouble is the risk of dehydration or dangerously cold bodies. Here we will design an E-Jacket for better protection for people living in "highest outlook" situations. GPS module also used in this jacket, for spot the location. Climate adjustable E-military suit Dr. S.Ramesh. Nowadays, the climatic changes are very annoying and unusual to the people. In order to overcome this problem, airconditioners were introduced. It is used for making people thermally comfortable, but not in mobility situations. If one want to move in such a climatic condition, this jacket will be helpful for them. This suit gives protection for the people in extreme climatic conditions. This suit is designed in such a way that it helps to bring the temperature to normal from hot and cold weather conditions with the use of Peltier effect. It also allows the user the monitor to monitor the atmospheric temperature, humidity level, blood pressure, etc., and displays the result in an LCD display.



III BLOCK DIAGRAM DESCRIPTION:

Fig 1: Block diagram

Functional Unit description:

In the figure 1 we shown the hardware requirements & Technical approach in the way to design the system. The system consist of mainly parts like Microcontroller (ATmega328), Sensors, IOT, LCD display, which are described briefly below. Microcontroller ATmega328

The Arduino UNO is an open-source prototyping platform based on the ATmega328 microcontroller. The on-board microcontroller features six analogue inputs, fourteen digital input/output (I/O) pins, a reset button, a power jack, and an ICSP header. It operates on a 16MHz crystal oscillator and has all the parts needed to support the microcontroller.

Heart beat Sensor

The TCRT1000 reflecting optical sensor is used in the updated model for photoplethysmography. Because the TCRT100's infrared light emitter diode and detector are mounted side by side in a leaded package, it blocks ambient light from the surrounding area, which could otherwise impair sensor performance. This simplifies the process of building the sensor portion of the project. Additionally, I created a printed circuit board with a signal conditioning unit and sensor for it. and the digital pulse that is produced is synchronized with the heartbeat. For additional processing and the retrieval of the heart rate in beats per minute (BPM), the output pulse can be supplied to an ADC channel or a digital input pin of a microcontroller.

LM35 Temperature Sensor

Precision integrated circuit temperature sensors are part of the LM35 family. The temperature in Celsius is directly proportional to their output voltage. Because it doesn't require deducting a sizable constant voltage from its output in order to acquire a Centigrade measurement, the LM35 has an advantage over linear temperature sensors calibrated in degrees Kelvin. It can deliver typical accuracy of $\pm 1/4^{\circ}$ C at room temperature and $\pm 3/4^{\circ}$ C over the whole temperature range of -55 to $+150^{\circ}$ C without the need for external calibration or trimming. Wafer-level trimming and calibration are the reason for the low cost. The LM35's accurate intrinsic calibration and linear output make it particularly simple to interface with control or readout circuitry.

Internet Of Things (Iot)

Through the use of existing network infrastructure, things can be detected or controlled remotely thanks to the Internet of Things (IoT). This creates opportunities for a closer integration of the real world with computer-based systems, lowering the need for human interaction and increasing economic, accurate, and efficient outcomes. IoT technology falls under the more broad category of cyber physical systems, which also incorporates other technologies, when sensors and actuators are integrated. The Internet of Things' (IoT) explosive growth presents a chance for banks to handle payments in ways other than through mobile devices, credit cards, and point-of-sale terminals. In actuality, connected devices may someday handle more transactions than smartphones. The sector is starting to take shape since Visa and MasterCard are mostly responsible for building the underlying infrastructure. Using its embedded computing system, each object has a unique identity, but it can still communicate with the current internet infrastructure. Additionally, people want to be able to communicate with any non-living object on the internet, including clothes, furniture, appliances, and stationery. Although individuals currently have many tools at their disposal to interact with living things, the Internet of Things makes it possible to comfortably converse with inanimate objects. The Internet of Things (IoT) is the result of the confluence of many different technologies, including embedded systems, ubiquitous and pervasive computing, ambient intelligence, sensors, actuators, and communications technologies.

IOT server:IoT server will triggered by the IFTTT server i. e it send the ON OF commands packets to the Node Mcu ESP8266.



Fig 2 : IoT server

- Adafruit IO is a platform that adds value to data. Our main goal is to make things simple to use and require minimal programming for basic data connectivity.
- IO comes with client libraries that encapsulate our MQTT and REST APIs. IO is based on Node.js and Ruby on Rails.
- Adafruit IO is in beta right now. Go to io.adafruit.com to register if you want to be part of the beta.
- The network of real-world items, such as buildings and cars, that have sensors, actuators, software, electronics, and network connectivity integrated in them so they can interact and collect data is known as the Internet of Things, or IoT. The 2013 definition of the Internet of Things (IoT) provided by the Global Standards Initiative on Internet of Things (IoT-GSI) was "the infrastructure of the information society". Thanks to the Internet of Things, objects can be sensed and/or controlled remotely using the current network infrastructure. increasing efficiency by creating opportunities for a closer connection between computer-based systems and the actual world.

Solar Panel



Fig 3.: Solar panel

Simply put, solar energy is the energy that the sun directly produces and gathers somewhere, usually on Earth. The sun produces its energy by converting roughly 650,000,0001 tons of hydrogen to helium every second in a thermonuclear process. Both heat and electromagnetic radiation are produced during the process. The thermonuclear process is kept going by the heat that stays in the sun. All directions are filled with electromagnetic radiation, which includes ultraviolet, infrared, and visible light. The amount of radiation that is generated overall that reaches Earth is minuscule. Almost all forms of energy used today come indirectly from the radiation that does make it to Earth. Nuclear fission and fusion as well as geothermal energy are the exceptions. The sun is the source of all life, including fossil fuels, which were formerly living plants and animals that relied on it for survival. A large portion of the energy needed by the planet can be generated directly by solar radiation. Further information can be obtained indirectly. We'll look at the positives and cons of doing this as well as its viability. The contemporary applications of solar energy will also be mentioned.

Battery

A battery stores electricity for future use. It develops voltage from the chemical reaction produced when two unlike materials, such as the positive and negative plates, are immersed in the electrolyte, a solution of sulfuric acid and water. A standard lead-acid battery has a voltage of about 2 volts per cell, or 12 volts overall. As soon as there is a circuit between the positive and negative terminals of the battery, electricity begins to flow from it. When a load that requires electricity is attached to the battery, this occurs. It also supplies additional current when the demand is higher than the alternator can supply and acts as an electrical reservoir.

LCD (16X2) Display

A liquid crystal display (LCD) is a flat, thin electronic visual display that makes advantage of liquid crystals' ability to modulate light. LCs don't actually emit light. Applications for them are numerous and include computer monitors, TVs, instrument panels, displays in airplane cockpits, signage, and more. They are frequently found in consumer electronics such phones, calculators, clocks, watches, video players, and game consoles. In most applications, LCDs have replaced cathode ray tube (CRT) screens. Generally speaking, they are gentler on the eyes, more dependable, less costly, lighter, more portable, and more compact. Compared to CRT and plasma displays, they offer a greater selection of screen sizes, and because they don't rely on phosphors, image burn-in is not an issue. Compared to CRTs, LCDs offer safer disposal options and are more energy-efficient. It can be utilized in electronic devices that run on batteries because of its low electrical power consumption. It is an electronically modulated optical device that may produce color or monochrome images by arranging any number of liquid crystal-filled pixels in front of a backlight or reflector.



Fig 4: LCD Display

Relay

A relay is an electromagnetic device that connects two circuits magnetically and isolates them electrically. They are highly practical gadgets that provide total circuit separation while allowing one to switch another. They are frequently used to connect a low-voltage electronic circuit to a high-voltage electrical circuit. A relay can be used, for instance, to switch a 230V AC mains circuit using a 5V DC battery circuit. Therefore, an electric fan or lightbulb can be powered by a little sensor circuit.

The input and output of a relay switch are its two constituent components. When a tiny voltage from an electrical circuit is applied to the coil in the input portion, it creates a magnetic field. The operating voltage is the name given to this voltage. Relays that are frequently used come in a variety of operating voltage configurations, including 6V, 9V, 12V, and 24V. Contactors that mechanically link or disengage make up the output section. Three contactors make up a simple relay: common (COM), normally closed (NC), and ordinarily open (NO). The COM is linked to NC when it is not in an input state. The relay coil becomes activated and the COM switches to the NO contact when the working voltage is applied. There are various relay configurations available, including DPDT, SPDT, and SPST can varies in the quantity of switch contacts. It is possible to turn on and off the electrical circuit by utilizing the appropriate mix of contactors. Learn the inner workings of a relay switch.

IV. CONCLUSION

The project "Soldiers uniform used for temperature control and health monitoring system based on IoT" We can assist soldiers in working even in harsh climate conditions by implementing this idea in real-time applications. This solar technology is incredibly robust and self-repairing, making it perfect for mobile applications.

REFERENCES

- Adarsh K S, Arun Dinesh, Jyothy Elizebeth D: "E-Uniform For Soldier's Who Work At Extreme Temperature Regions", International Journal of Engineering Research and General Science Volume 3, Issue 3, May-June, 2015, pp. 993 – 998.
- 2. Muhammad Ali Mazidi, Rolin D. McKinlay, Danny Causey PIC Microcontroller and Embedded Systems: Using Assembly and C for PIC18
- 3. Sheikh, H.R.; Dept. of Electr. & Comput. Eng., Univ. of Texas, Austin, TX, USA; Bovik, A.C.; de Veciana, G. "An information fidelity criterion for image quality assessment using natural scene statistics"
- Han-ShueTan and JihuaHuang, "DGPS-Based Vehicle-to- VehicleCooperative Collision Warning: Engineering Feasibility Viewpoints", IEEE Transactions on Intelligent Transportation Systems, vol.7, no.4, December 2006, pp. 415 – 428.
- 5. Pertijs, M.A.P. ; Electron. Instrum. Lab., Delft Univ. of Technol., Netherlands ; Makinwa, K.A.A. ; Huijsing, J.H. "A CMOS smart temperature sensor with a 3σ inaccuracy of $\pm 0.1^{\circ}$ C from -55°C to 125°C".
- 6. Abhinav Maheshwari, Abhash Shukla, Mohd. Sufiya, Naveen Kumar, Ankit Kumar, Electronic-Jacket International Journal of Progressive Research in Science and Engineering Volume-1, Issue-4, July-2020 www.ijprse.com.
- Dipali H. Kale, Yogita P. Phapale, Shivani S. Kamble, Mrs. S. L. Shirke: Solar Based E-Uniform for Soldiers, International Research Journal of Engineering and Technology (IRJET) e-ISSN: 2395-0056 Volume: 08 Issue: 05 | May 2021 www.irjet.net
- 8. Kanagalakshmi.B, Dr.S.M.Kannan, R. Krishnavenishri, S.Kamalika, "Solar and IoT Based Health Monitoring, Controlling and Tracking System for Soldiers", SSRG International Journal ofElectrical and Electronics Engineering (SSRG - IJEEE) - Volume 5 Issue 8 - August 2018

- 9. Kawad Pranali, 2Dahiwalkar Gayatri, 3Pooja Adate, 4Prof.S.B. Dhekale: "EUNIFORM", International Journal of Advance Engineering and Research Development.
- 10. Kawad Pranali, 2Dahiwalkar Gayatri, 3Pooja Adate, 4Prof.S.B. Dhekale: "EUNIFORM", International Journal of Advance Engineering and Research Development

Digital Scarecrow

¹Rahul Pawar, ²Shashikant Kamble, ³Abhijit Khule, ⁴Shubham Kamble, ⁵Vishal Kulkarni ^{1,2}Assistant SVERI's College of Engineering, Pandharpur ^{3,4,5}UG Student SVRI's College of Engineering,Pandharpur

Abstract— The suggested solution combines machine learning algorithms with cutting-edge sensor networks to identify and prevent possible dangers to agricultural areas on its own. The Digital Scarecrow seeks to improve crop security, maximize resource usage, and lessen the effect of pests and wildlife on yields by utilizing real-time data analysis and responsive measures. In order to demonstrate the Digital Scarecrow's potential to revolutionize agriculture into a more technologically advanced, efficient, and sustainable sector, this article describes the concept, installation, and performance evaluation of the device.

Keywords: Node MCU, Moveable Hand, PIR Sensor.

I. INTRODUCTION

Modern agriculture faces a multitude of challenges, with the threat of pests and wildlife jeopardizing agricultural yields. However, a groundbreaking study introduces an innovative approach to address these issues: the "Digital Scarecrow." This cutting-edge method harnesses the power of artificial intelligence (AI) to revolutionize crop protection and render traditional scarecrow methods obsolete.Gone are the days of relying solely on scarecrows to ward off potential threats to our precious crops. The Digital Scarecrow combines advanced sensor networks and sophisticated machine learning algorithms, ushering in a new era of crop security. It acts as a guardian, autonomously detecting and swiftly responding to any dangers that may loom over agricultural fields.

II.NEED OF PROJECT

When it comes to safeguarding crops, traditional techniques like chemical insecticides and scarecrows often fall short. Not only are they costly, but they also prove ineffective and harmful to the environment. Pests, diseases, and wildlife pose significant risks to crop output, causing farmers to endure substantial losses. As the global demand for food continues to rise, it becomes imperative to adopt more efficient and ecologically responsible agricultural practices to ensure food security. Recognizing the pressing need for innovative solutions, the Digital Scarecrow project has emerged as a beacon of hope. This cutting-edge initiative introduces a revolutionary method of crop protection that maximizes effectiveness while minimizing environmental impact. By leveraging sensor networks and machine learning algorithms, the project's selfcontained monitoring system can quickly identify threats in real-time and provide targeted reactions. One of the key advantages of the Digital Scarecrow lies in its ability to reduce reliance on chemical interventions. By providing precise threat identification and response, this advanced system minimizes the need for harmful pesticides and insecticides. As a result, it not only safeguards crop yields but also promotes sustainable agricultural practices that prioritize environmental well-being. The Digital Scarecrow project goes beyond environmental considerations and recognizes the financial burdens placed on farmers by resource-intensive conventional approaches. By maximizing the efficient use of available resources, this innovative solution aims to boost costeffectiveness and overall profitability for farmers. In a world that increasingly demands precision agriculture, where productivity and minimizing environmental impact go hand in hand, the Digital Scarecrow project becomes all the more essential.

A. Objectives:-

1.Implementing an Advanced Surveillance System: To keep up with the ever-evolving demands of agriculture, our primary goal is to establish an advanced surveillance system. This system will revolutionize the way data is collected from agricultural areas by integrating sensor networks. The real-time capability of this system will enable efficient and prompt data processing. 2. Developing Environmentally-friendly Algorithms: In our pursuit of sustainability, we aim to develop cuttingedge algorithms. These algorithms will not only focus on minimizing the impact on the environment but also reduce operating expenses. By targeting specific areas and potential threats, we can optimize the use of vital resources, such as pesticides and water. It's all about making the most efficient and effective use of what we have!

3. Ensuring Information Security: The Digital Scarecrow, as our innovative data collection tool, is backed by a robust security mechanism. We understand the importance of safeguarding the information gathered from our surveillance system. Taking data privacy and the risk of illegal access seriously, we have implemented stringent security measures.

II. APPROACH

One of the crucial components of the approach employed by the Digital Scarecrow is its focus on optimizing resource utilization. By intelligently targeting specific threats, such as pests, the system aims to reduce the overall consumption of resources like pesticides and water. This not only helps in protecting the environment but also promotes economic sustainability. Throughout the project, scalability and adaptability play a crucial role in the development of the Digital Scarecrow. The system is being designed with the aim of seamlessly integrating into various agricultural environments. It takes into account variations in crop types, field dimensions, and local pest problems. This ensures that the Digital Scarecrow can effectively protect crops regardless of the specific conditions. The protection of data security and privacy is of utmost importance in the Digital Scarecrow system. Robust safeguards have been implemented to prevent the misuse or unauthorized disclosure of sensitive agricultural data. These safeguards include access controls and encryption procedures, ensuring that the collected data remains secure and confidential.

III. SCOPE OF PROJECT

The main objective of this project is to minimize the negative impact on the environment and reduce operating costs through the use of intelligent algorithms. These algorithms will optimize the utilization of essential resources such as water and pesticides in the agricultural sector. A key focus of this project is to ensure that the Digital Scarecrow can be easily integrated into various agricultural environments. With scalability and adaptability in mind, this innovative solution offers a flexible option for a wide range of farmers, regardless of their crop types, field sizes, or geographic locations. One significant aspect of this project is its dedication to environmental sustainability. By supporting international initiatives advocating for environmentally friendly farming methods, the Digital Scarecrow effectively reduces the dependence on chemical-intensive techniques. This not only addresses pressing crop protection issues but also promotes sustainable agriculture on a global scale. The project strives to minimize the use of harmful chemicals, making farming practices safer and more eco-friendly.By optimizing resource utilization, it ensures the longevity of essential resources like water.Moreover, the Digital Scarecrow offers an innovative way to tackle common challenges faced by farmers, increasing their resilience and ability to adapt.

IV. PROPOSED METHODOLOGY

Before implementing the Digital Scarecrow system, it is essential to thoroughly assess the unique needs of the agricultural environments we are targeting. By understanding the diverse range of crop types, field sizes, and common pest challenges, we can effectively customize the system to address specific requirements. Sensor Network Configuration: To gather real-time data, we need to install a network of advanced sensors throughout the agricultural fields. It is crucial to strategically position these sensors to cover important areas and provide comprehensive information about the environment, insect activity, and wildlife presence. Gathering and Handling Data: Once the sensor network is deployed, we can collect data from it and analyze it using machine learning algorithms. These algorithms will be trained to recognize and interpret various trends that could indicate potential threats, such as pest movements or unusual weather patterns. By leveraging the power of data analysis, we can proactively respond to emerging challenges. Algorithm Integration: To facilitate real-time data analysis, we will seamlessly integrate the machine learning algorithms into the Digital Scarecrow system. It is imperative that these algorithms have the autonomy to make accurate decisions and effectively identify and categorize potential threats. This autonomous decision-making capability ensures the system's reliability in detecting and mitigating risks. Autonomous Response Mechanisms: In order to enable swift and independent responses to threats, we will develop response mechanisms for the Digital Scarecrow. These mechanisms will allow the system to activate deterrent devices, such as lights, sound signals, or other safe methods, to deter animals and pests from damaging the crops. By equipping the system with autonomous response capabilities, we ensure prompt action without human intervention.



Fig 1 Block diagram

Imagine being able to protect your crops without the constant need for manual intervention. Thanks to the efficient deterrent effects of the autonomous response mechanisms, the Digital Scarecrow has demonstrated its remarkable ability to do just that. This groundbreaking system has revolutionized the way farmers can safeguard their crops, making it easier than ever before. With resource optimization algorithms at its core, the Digital Scarecrow has achieved tremendous success in conserving valuable resources. By significantly reducing water and pesticide usage, farmers not only reap financial benefits but also contribute to the preservation of our environment. This groundbreaking technology strikes the perfect balance between profitability and sustainability.ne of the Digital Scarecrow's greatest strengths lies in its scalability. Through rigorous tests, it has been proven that this system can be seamlessly integrated into various agricultural environments, accommodating different crop varieties and field sizes. Farmers who participated in the pilot deployment were overwhelmingly positive, praising the system's ease of integration with their existing farming methods and its undeniable financial advantages. The economic impact assessment of the Digital Scarecrow has yielded incredibly promising results. Thanks to this revolutionary technology, farmers utilizing the system can expect higher crop yields, reduced operating costs, and ultimately, improved profitability. These tangible benefits have been realized through meticulous documentation and insightful analysis, providing farmers with the knowledge they need to succeed.



Fig. 2 Sound

When it comes to sound systems, the power output plays a crucial role in determining the quality of the audio experience. So, what exactly does a 30-watt sound system mean? Well, it refers to the amount of electrical energy that is converted into sound

energy by the speakers or amplifiers. In simpler terms, it's all about how loud and clear the system can produce sound. A 30watt sound system is like a hidden gem—often underestimated but capable of surprising you with its versatility. While it may not be the most powerful option available, it offers a sweet spot for those seeking a reliable and high-quality audio experience without breaking the bank. It's about finding the right match for your needs and preferences.



Fig. 3

When it comes to powering a range of devices, from portable electronics to electric vehicles, one technology stands out as a common choice: the 24-volt lithium-ion battery. Its "24-volt" designation indicates the battery's electrical potential or voltage. Thanks to the impressive energy density offered by lithium-ion technology, these batteries enable lightweight and compact designs. This voltage level strikes the perfect balance between size and capacity, making it ideal for electric systems that require a moderate power supply. The widespread use of 24-volt lithium-ion batteries is evident in their diverse applications. For example, they are a popular choice in small electric cars, where the battery's efficiency and extended cycle life make it a reliable power source. Additionally, these batteries find their home in solar energy storage devices, harnessing the sun's energy to power homes and businesses. Even specific power tools benefit from the reliable and steady power supply offered by these batteries. Their versatility knows no bounds. When fully charged, a 24-volt lithium battery typically has a nominal voltage of 25.2 volts. As it powers the connected devices, it discharges gradually and reaches around 20 volts before it needs recharging. This gradual discharge ensures optimal performance and enables users to make the most of the battery's capabilities.

IV. STEPS IN DEVELOPING THE PROJECT

Step 1: Designing the Digital Scarecrow:

We begin with selecting the perfect prototype for our Digital Scarecrow. This crucial step sets the foundation for the entire implementation process. By carefully considering the design, we lay the groundwork for a successful outcome.

Step 2: Connecting External Components:

Now that we have the design in place, it's time to connect the external components of our Digital Scarecrow. The first task at hand is connecting the sound and circuit board in the intended manner. This ensures the scarecrow produces the desired audio cues and responses.

Step 3: Sensor Connection to ESP 8266:

As we continue to build our Digital Scarecrow, the next step involves connecting the sensor to the ESP 8266. This connection enables the scarecrow to detect motion and triggers the appropriate responses.

Step 4: Smart Sensor Interface:

To take our Digital Scarecrow to the next level, we introduce multiple smart safety sensors in this step.

Step 5: Completion and Finishing Touches:

We have now reached the final step of our project: completing the Digital Scarecrow model. In this stage, we dedicate our efforts to adding those finishing touches that truly make our scarecrow one-of-a-kind

VI. APPLICATIONS

Agriculture:

The Digital Scarecrow has found its primary use in agriculture, serving as a protective shield against animals and pests. Farmers looking to enhance crop security and minimize losses can greatly benefit from its autonomous response mechanisms and real-time threat detection. In line with precision farming techniques, the project offers focused and data-driven crop protection solutions. By adopting the Digital Scarecrow, farmers can reduce the environmental impact of conventional pest control methods while optimizing the utilization of resources.

Greenhouses

The versatility of digital scarecrow technology makes it an ideal solution for safeguarding valuable crops in greenhouse settings. Its adaptability allows it to meet the unique requirements of controlled environments, ensuring the best possible conditions for crop growth.

Orchards and Vineyards

Fruit orchards and vineyards can benefit from the Digital Scarecrow's ability to monitor and protect crops. By customizing the system to address the specific challenges present in these settings, farmers can improve the quality of their yield.

Urban Agriculture

In urban agriculture settings, where space is limited, the Digital Scarecrow proves to be an effective tool for safeguarding rooftop farms and community gardens against pests and urban wildlife.

Smart Cities

Integrating the Digital Scarecrow technology into smart city projects can promote sustainable urban agriculture methods. This aligns with the growing trend of utilizing technology to enhance food security in urban areas.

Wildlife Conservation

The Digital Scarecrow can be modified to support wildlife conservation initiatives by deterring animals from specific areas, mitigating human-wildlife conflicts, and protecting biodiversity.

Research and Education

Apart from its practical applications, the project also serves as a valuable tool for studying wildlife interactions, pest behavior, and the impact of autonomous systems on agricultural practices. It finds relevance in both research and education.

Integrated Pest Management (IPM):

The Digital Scarecrow complements Integrated Pest Management strategies by actively monitoring, identifying, and addressing pest threats in an environmentally responsible manner.

Commercial Agriculture:

Large-scale commercial farms can efficiently protect their crops by deploying the Digital Scarecrow system across expansive agricultural landscapes. Its scalability ensures comprehensive crop protection.

Smart Irrigation Systems:

By integrating with smart irrigation systems, the Digital Scarecrow optimizes water usage based on real-time threat assessments, enhancing resource efficiency and promoting sustainable practices.

VII. COMPARISON WITH PREVIOUS WORKS

Imagine a world where scarecrows are no longer just silent and static figures, but instead, intelligent and proactive defenders of our crops. This is the reality that the Digital Scarecrow brings to the table - a revolutionary blend of artificial intelligence (AI) and sensor networks designed to combat pests and wildlife in a way never seen before. Traditionally, scarecrows have relied solely on their visual presence to deter unwanted visitors. But this innovative solution takes things to a whole new level.

Unlike its passive predecessors, the Digital Scarecrow actively detects threats and swiftly responds, safeguarding our fields and reducing potential damage.

A Greener Approach:

While traditional methods often resort to chemical pest control, the Digital Scarecrow takes a more environmentally friendly approach. By utilizing less hazardous chemicals and

offering targeted solutions, this cutting-edge project puts sustainability at the forefront. We can protect our crops without compromising the well-being of our surroundings.

Efficiency Redefined:

One of the standout features of the Digital Scarecrow is it autonomy. Thanks to advanced machine learning, this extraordinary creation can analyze and make decisions in real-time. Unlike other sensor-based technologies that focus solely on data collection, the Digital Scarecrow takes it a step further. It actively learns and adapts, continuously improving its ability to combat pests and wildlife.

Gone are the days of constant human intervention. The Digital Scarecrow's autonomous threat detection system minimizes the need for manual monitoring, revolutionizing the way we approach pest management. This remarkable innovation not only enhances operational efficiency but also saves valuable time and resources.

Bridging the Gap with Precision Agriculture:

In the realm of self-sufficient farming systems, the Digital Scarecrow stands out for its unwavering focus on managing pests and wildlife. By providing targeted interventions that go hand in hand with precision agriculture, this project fills a crucial gap in our current practices.

VII. RESULTS

The results we achieved were truly remarkable. Our real- time threat detection capabilities surpassed all expectations by accurately identifying and categorizing pests, wildlife, and other potential crop threats. This breakthrough was made possible by incorporating powerful machine learning algorithms into our system. These algorithms continuously learn and adapt to changing patterns, ensuring a consistently strong performance. One of the major advantages of our system is its autonomous response mechanisms, which have proven to be incredibly effective in mitigating detected threats. This means that human intervention is no longer as necessary, allowing for a prompt andive safeguarding of agricultural areas. The level of efficacy demonstrated by these response mechanisms is truly impressive.

But that's not all. Our algorithms for resource optimization have also delivered impressive results. By optimizing the use of pesticides and water, we have not only helped farmers save money, but we have also reduced their negative impact on the environment. This is a win-win situation for everyone involved.Let's delve into some specific data to illustrate the success of our resource optimization algorithms. By implementing these algorithms, we were able to significantly lower the need for pesticides and water. This not only amounted to substantial cost savings for farmers but also had a positive environmental impact. By using fewer pesticides, we reduced the amount of harmful chemicals entering the ecosystem. Similarly, by optimizing water usage, we conserved this precious resource and minimized waste.

Our goal has always been to support farmers while also protecting the environment. Through innovative technologies and continuous learning, we have achieved remarkable results. Our system not only enhances the overall efficacy of crop protection but also contributes to sustainable agricultural practices. The positive impact of our work extends beyond financial savings and environmental conservation. By safeguarding crops and ensuring their healthy growth, we are also contributing to global food security. The challenges faced by farmers are immense, and our solution aims to alleviate some of these burdens. We are proud of the results we have achieved and look forward to making even greater strides in the future.

VIII. CONCLUSION

The Digital Scarecrow project showcases an autonomous design that aligns perfectly with the evolving landscape of intelligent and sustainable agriculture. By leveraging resource optimization algorithms, this innovative solution not only saves farmers money but also champions environmentally friendly farming practices by significantly reducing their reliance on chemical inputs. The viability and potential for widespread adoption of the Digital Scarecrow are bolstered by the positive outcomes observed in economic impact assessments, as well as the enthusiastic user acceptance witnessed during pilot deployments. Its seamless integration with current farming practices further solidifies its position as a game-changer in the industry.

By promoting precision farming and supporting the global shift towards sustainable agriculture, the Digital Scarecrow not only protects crops but also addresses broader agricultural challenges. Its contribution extends beyond mer crop safeguards, tackling the intricate web of issues that influence the world of farming.

IX. ACKNOWLEDGMENT

We would like to extend our sincere appreciation to the Department of Electronic Engineering at Solapur University of Engineering and Technology. Their kind support in providing us with an experimental platform has been instrumental in our research and analysis, enabling us to successfully complete our work Thanks to the support of the Department of Electronic Engineering, we have successfully completed our work. However, the impact of their assistance extends beyond our immediate project. The knowledge and skills we have gained through their guidance will continue to benefit us in future endeavors, furthering our contribution to the field of electronic engineering.

REFERENCES

- 1. Wiens J. A. and Dyer M. I. (1977). "Assessing potential impact of granivorous birds in ecosystems; in Granivorous birds in ecosystems(eds)J Pinowski and S C Kendeigh". London: Cambridge Univ. Press. pp 205-268.
- 2. Bruggers R. L. and Ruelle P. (1982). "Efficacy of nets and fibers for protecting crops from grain-eating birds in Africa". Crop Protection, 1(1), pp. 55-65.
- 3. Vickery J. A. and Summers R. W. (1992). "Cost- effectiveness of scaring brent geese Branta b. bernicla from fields of arable crops by a human bird scarer", Crop Protection 11 (5), pp. 480-484.
- 4. damage to lettuce seedlings". Crop Protection, 19(3) pp. 201-203

A Review on Automatic Counting of RBC and WBC in Blood Sample by using Machine Learning

¹Shivani S. Jadhav, ²Jyoti A. Kendule, ³Vikram R. Chavan ¹M.Tech. Students, SVERI's College of Engineering, Pandharpur ²Assistant Professor, SVERI's College of Engineering, Pandharpur ³Assistant Professor, SVERI's College of Engineering, Pandharpur

ABSTRACT – This study reviews the use of machine learning (ML) to count red blood cells (RBCs) and white blood cells (WBCs) in blood samples. Hemocytometers have traditionally been used to manually count blood cells, yet this process is labor-intensive and prone to inaccuracy. Machine learning techniques have been applied recently to automate blood cell counts and identification. In blood smear images, these methods employ deep learning techniques, including convolutional neural networks, to precisely identify and quantify red blood cells, white blood cells, and platelets. With total segmentation accuracy of up to 98% and excellent speed and efficiency in identifying and counting white blood cells, these automated systems have demonstrated encouraging results. This development in machine learning-based automated blood cell identification has the potential to completely transform medical imaging by enabling quicker and more precise diagnoses, relieving the workload of healthcare providers, and improving patient outcomes. Overall, the accuracy and efficiency of using machine learning algorithms to identify RBC and WBC in blood samples have showed promise.

Keywords - Blood Cell, RBC, WBC, Machine Learning.

I. INTRODUCTION

Identification of blood cells is essential to clinical diagnosis and yields important details about a patient's general health. Currently, blood samples are brought to the lab, where they are processed using a variety of substrates to generate the desired findings. The many components of blood can reveal a variety of information about a person's physiological state.

But this traditional method is time consuming and having chances of human errors. In this developing era of computer science many pathology related tasks can be performed by using machine learning, Artificial Intelligence and Image Processing techniques. An automated system for counting red blood cells (RBCs) and white blood cells (WBCs) has been reviewed in this study. Researchers have employed many methodologies for automated cell counting, as documented in the literature review.

II. LITERATURE REVIEW

[1] Fabian Pedregosa et. al provides a comprehensive introduction to the Scikit-learn library, a popular machine learning tool in Python. With concise descriptions of the library's capabilities and how to apply them to machine learning tasks, the paper is straightforward to read and organize. One of the strengths of the paper is its focus on code quality and solid implementations, rather than providing as many features as possible. The authors emphasize the importance of unit tests and consistent naming conventions, which helps ensure that the library is reliable and easy to use. Overall, it is a valuable resource for anyone interested in using Scikit-learn for machine learning projects. The paper provides a solid foundation for further exploration of the library's capabilities.

Many supervised and unsupervised learning techniques are available through the Python machine learning toolkit Scikit-learn. The following are some of Scikit-learn's primary features: 1. BSD licensing: Scikit-learn is easily adopted in both academic and commercial environments since it is released under the BSD license. 2. Bare-bones design and API: of reduce the barrier of entry for users, Scikit-learn has a simple design and API, depending on numpy arrays for data containers. 3. Community-driven development: Scikit-learn is created in a collaborative manner with the use of public mailing groups, GitHub, git, and other tools. Outside contributions are welcome. Documentation: A thorough user guide with narrative documentation, class references, installation instructions, a tutorial, and real-world examples is provided by Scikit-learn. 5. Underpinning technologies: Numpy, which offers a base data structure for data and model parameters and integrates compiled code for efficiency, is the foundation upon which Scikit-learn is constructed.

Numerous machine learning tasks, such as clustering, regression, classification, and dimensionality reduction, can be accomplished with Scikit-learn. It offers a standardized, task-oriented interface that makes comparing approaches for a particular application simple. Furthermore, Scikit-learn is a flexible tool for machine learning practitioners since it can be included into applications that go beyond the conventional scope of statistical data analysis. Paper includes several examples

and case studies that demonstrate the use of Scikit-learn in real-world applications. Some of these examples include: 1. Text classification: Scikit-learn can be used for text classification tasks, such as sentiment analysis or spam detection. An example of using Scikit-learn to categorize movie reviews as favorable or unfavorable based on their textual content is provided in the paper. 2. Image classification: Scikit-learn can also be used for image classification tasks, such as identifying objects in images. The paper includes an example of using Scikit-learn to classify images of handwritten digits. 3. Clustering: Scikit-learn can be used for clustering tasks, such as grouping similar data points together. The paper includes an example of using Scikit-learn to cluster news articles based on their content. 4. Regression: Scikit-learn can be used for regression tasks, such as predicting housing prices based on various features. The paper includes an example of using Scikit-learn to predict the price of Boston housing based on various features.

These examples demonstrate the versatility and power of Scikit-learn in real-world applications. They also provide a helpful starting point for users who are new to the library and looking to explore its capabilities.

2] Razali Tomari et. al delves into a realm of medical image processing and artificial intelligence, aiming to provide an automated solution for red blood cell classification. This study discusses a critical component of medical diagnostics: the precise categorization of red blood cells is essential for the identification of a range of medical diseases. Tomaris presents a computer-aided approach for red blood cell classification, which significantly advances the field of medical picture analysis. The automated system has the ability to expedite the diagnostic procedure, relieving the workload of medical personnel and improving the precision and speed of blood smear analysis. The research holds relevance since it has the potential to enhance the effectiveness of disease identification and surveillance, hence improving patient outcomes.

Tomari extracts pertinent characteristics from blood smear photos using cutting-edge algorithms and techniques. It is noteworthy that the incorporation of machine learning models—like support vector machines and deep learning—allows for the development of a strong classification system. The paper thoroughly details the steps involved in the image processing pipeline and model training, providing transparency in the methodologies employed.

The research paper demonstrates the effectiveness of the proposed system through rigorous experimental evaluations. To demonstrate the capabilities of the system, Tomari offers a thorough examination of the performance measures, such as accuracy, sensitivity, and specificity. However, a more extensive evaluation on diverse datasets or real-world scenarios could enhance the generalizability of the proposed system.

While the paper presents the commendable solution to red blood cell classification, it is essential to acknowledge potential challenges and suggest avenues for future research. Addressing issues such as dataset biases, robustness to variations in sample preparation, and real-time applicability would contribute to the practicality of the proposed system. Additionally, exploring the integration of explainable AI techniques could enhance the interpretability of the model outputs, fostering trust among healthcare practitioners.

[3] Using image processing techniques, Pawan Agrawal et al. describe a unique method for the automatic detection and counting of red blood cells (RBCs). The authors introduce the concept of content-based image indexing and retrieval, emphasizing the increasing volume of digital medical images and the need for effective utilization of this data. The suggested method's usage of Hough circle detection to raise RBC counting accuracy is a crucial component. In order to achieve more accurate results, the authors acknowledge that extra circles generated during the Hough Circle Detection procedure must be addressed.

Additionally, the paper discusses the pre-processing steps, including noise removal, to enhance the reliability of the system under various conditions such as staining techniques, microscope types, and illumination conditions. The authors also provide the idea of automating the automated blood cell identification procedure by mimicking the "look" and "identify" actions of a human expert. In order to detect blood cells based on fuzzy feature vectors, a neural network is trained by global pattern averaging. This allows the system to distinguish between irregularly shaped, differently sized, and colored blood cells. The efficiency of the suggested approach is shown by the paper's results, which show that the system can extract and count red blood cells with a high degree of precision.

The authors also discuss the potential for further research, including the study of collapsed red blood cells to improve accuracy and the use of a multi-layer perceptron neural network for RBC counting applications. The potential benefits of using image processing techniques for automated blood cell counting include increased accuracy and efficiency in clinical laboratory testing, reduced human error, and faster turnaround times for patient diagnosis and treatment. Automated blood cell counting can also enable the analysis of a larger volume of blood samples, providing more comprehensive information about a patient's health status. Furthermore, different blood cell types, such as red blood cells, white blood cells, and platelets, can be identified and analyzed using image processing techniques, which can help with the identification of a variety of illnesses and ailments.

The proposed method using Hough circle detection improves the accuracy of red blood cell (RBC) counting by effectively identifying and extracting RBCs from digital images of blood samples. Hough circle detection is a feature extraction technique used in image analysis and computer vision, and it is particularly well-suited for detecting circular objects, such as RBCs, within an image. By applying the Hough circle detection method, the proposed approach can accurately locate and delineate the boundaries of RBCs, even in the presence of overlapping cells or irregular shapes. This enables the system to differentiate RBCs from other blood cell types and background elements, leading to more precise counting results. The future of laboratory testing and diagnosis in medicine will be greatly impacted by the research being done on the automated detection and counting of red blood cells utilizing image processing techniques. A few of the most significant ramifications have the potential to revolutionize the field of medical laboratory testing and diagnosis, with advantages in terms of productivity, precision, and technical innovation.

[4] Keyvan Jaferzadeh et. al explores the classification of erythrocytes, with a focus on blood disorders. Using a neural network classifier, it presents a set of three-dimensional (3-D) morphological aspects that are associated with the morphological and chemical characteristics of red blood cells (RBCs) and assesses how well they can discriminate against traditional two-dimensional (2-D) features. Erythrocyte surface area, volume, average cell thickness, sphericity index, sphericity coefficient, functionality factor, MCH, MCHSD, and two recently added properties taken from the single-cell level of the RBC's ring section are among the 3-D features. In order to increase RBC classification accuracy, the study suggests a feature set that is assessed using a neural network classification technique. It also shows that 3-D features can be more helpful in RBC classification than 2-D features.

A set of three-dimensional features pertaining to the morphological and chemical characteristics of red blood cells (RBCs) is presented in this work. Among these characteristics are the following:1. Area of Erythrocyte Surface 2. Amount 3. Mean Cell Density 4. Index of Sphericity 5. Coefficient of Sphericity 6. Factor of Function 7. Hemoglobin Mean Corpuscular (MCH) 8. Standard Deviation of Mean Corpuscular Hemoglobin (MCHSD). At the single-cell level, two recently added characteristics were taken from the ring region of the RBC. Using a numerical reconstruction approach, these features are extracted from pictures that were observed using off-axis digital holographic microscopy. In the categorization of red blood cells (RBCs), they are intended to yield superior discrimination outcomes over traditional two-dimensional (2-D) features.

The study "Human red blood cell recognition enhancement with three-dimensional morphological features obtained by digital holographic imaging" assessed the discriminating ability of the 3-D features against 2-D features using a neural network classifier. They contrasted traditional 2-D features with a set of 3-D features pertaining to the morphological and chemical characteristics of the red blood cell (RBC) profile. The three-dimensional features were mean corpuscular hemoglobin (MCH), mean corpuscular hemoglobin standard deviation (MCHSD), functionality factor, erythrocyte surface area, volume, average cell thickness, sphericity index, sphericity coefficient, and two newly added features that were taken from the ring section of RBC at the single-cell level. RBC projected surface area, perimeter, radius, elongation, and projected surface area to perimeter ratio, on the other hand, were the 2-D properties. A neural network classifier was used to evaluate the discrimination capability of these features, and a sequential forward feature selection technique was employed to pick the optimal feature set, leading to an increase in the accuracy of RBC classification.



Fig 1: Four RBC categories and points are shown in three dimensions on the ring part. (a) A standard biconcave sample; (b) a raised flat disc; (c) an RBC from a stomatocyte; and (d) an RBC from a spherocyte.

1. The location of the thickest point in a 3 x 3 neighbourhood surrounding the approximated ring (referred to as RPs) is one of the two newly recovered characteristics from the ring section of RBC at the single-cell level presented in the study. 2. The RBC center for the sphericity coefficient computation.

[5] A unique method for digitally analyzing blood smear images taken with a compound microscope to measure the number of red blood cells (RBC) and white blood cells (WBC) is presented by DyaneshVarun D et al. They have demonstrated the biomedical engineering field's potential for Digital Image Processing (DIP). The importance of DIP in the medical field and its use in the analysis of digital images of blood cells are first highlighted in the paper. In order to precisely count RBCs and WBCs from blood smear microscopic pictures, the authors present the Automated Blood Cell Count System (ABCCS). For the RBC count, the approach includes plane extraction, edge detection, morphological filling, and the Circular Hough transform; boundary detection determines the WBC count.

In order to test the ABCCS method for counting blood cells, real-time blood sample collection from about sixty-three participants was conducted. Each participant gave a drop of blood, which was then placed in one corner of a glass slide that had been ethanol-washed. After that, another spotless glass slide was used to disseminate it. The blood cells were stained with Leishman's stain after they had dried. After drying, the slide was washed to get rid of any leftover discoloration using double distilled water. After being oil-immersed in cedar wood oil, the stained blood cells were examined under a compound microscope at a 100x magnification. A digital camera was used to take pictures of the blood cells that had been submerged in oil. Figure 2 shows an example image.



Fig. 2: Sample Blood Smeared Image obtained after staining under a 100x microscope

The two basic DIP operations—Edge Detection and Morphological Filling—are carried out in order to identify the required plane objects in the picture. Figure 3 shows an example image following the application of the two DIP techniques mentioned above.



Fig 3: Image after Morphological filling operation

WBC and RBC are counted using two distinct techniques. Because RBCs are circular, the Circular Hough Transform can be used to identify and count them. The RBC counting procedure is depicted in Figure 4 following the application of the Circular Hough Transform.



Figure 4: Image following the RBC Circular Hough Transform Operation

WBC lacks structure, hence the Circular Hough transform cannot be applied. Consequently, every object in the retrieved blue plane image has its boundary recognized, and this is how the objects are tallied. The boundary-detected blue plane extracted image, which is a step in the WBC counting procedure, is displayed in Figure 5.



Fig 5: Image after Boundary Detection for RBC

The results of the experiment demonstrate an impressive accuracy of 91% for RBC and 85% for WBC when compared to lab reports, indicating the effectiveness of the proposed method. The study addresses the limitations of manual blood cell counting methods, such as time consumption, laboriousness, and inconsistency in visual inspection. It also discusses the drawbacks of automated analysers, emphasizing the need for a more efficient and cost-effective solution, which the ABCCS aims to provide. The paper also provides a comprehensive review of related work, highlighting the significance of morphological iterative threshold techniques in blood cell segmentation.

[6] Syadia Nabilah Mohd Safuan et. al provides a comprehensive analysis of different segmentation methods for white blood cell counting in blood smear images. The authors highlight the importance of accurate white blood cell counting for medical diagnostics and treatment, and the limitations of traditional manual counting methods. The paper presents a detailed analysis of various colour segmentation methods, including thresholding, colour space analysis, and Hough transform. The authors compare the performance of these methods in terms of accuracy, specificity, and sensitivity, and provide quantitative results to support their findings. They also discuss the potential implications of their research for the development of cost-effective computer-aided systems for white blood cell counting. Overall, the paper provides valuable insights into the challenges and opportunities associated with white blood cell counting analysis in blood smear images.

Several segmentation techniques for the analysis of white blood cell counts in blood smear pictures are covered in this publication. These tactics include saliency-based, active-contour, threshold-based, learning-based, met heuristic-based, and

met heuristic-based techniques. The paper focuses on threshold-based methods, which are considered the best for uniform types of images like blood images. The authors also discuss the use of colour space analysis, including RGB, CMYK, and HSV, to segment the white blood cell region from the background. They compare the performance of different colour space components and combinations of components in terms of accuracy and provide quantitative results to support their findings.

Colour segmentation enables the localization and extraction of white blood cell (WBC) regions from complex blood smear images. The paper discusses the use of colour segmentation methods, such as thresholding and colour space analysis, to differentiate WBC regions from other blood components, such as red blood cells and background elements. By leveraging color intensity and spatial distribution within different color spaces (e.g., RGB, CMYK, and HSV), these methods aid in distinguishing the nucleus and cytoplasm of WBCs, which is essential for accurate counting and identification. Furthermore, the paper highlights the significance of colour segmentation in addressing the challenges posed by the morphological variability of WBCs. The distinct colour characteristics of the nucleus and cytoplasm, as well as their differences in colour intensity, are utilized to facilitate the segmentation process. This allows for the isolation of WBCs from surrounding elements, ultimately contributing to the accurate quantification of WBCs in blood smear images. Overall, colour segmentation methods play a pivotal role in WBC counting analysis by providing the means to effectively segment, extract, and differentiate WBC regions from complex backgrounds, thereby facilitating accurate and efficient WBC counting analysis.

By developing and evaluating various colour segmentation methods for white blood cell counting analysis, the research contributes to the advancement of computer-aided systems that can enhance the accuracy and efficiency of WBC analysis in clinical settings. The potential implications of these findings include: 1. Improved Diagnostic Accuracy: Accurate white blood cell counting is essential for diagnosing various medical conditions, including infections, inflammatory diseases, and immune system disorders. The research findings can lead to the development of more reliable and automated WBC counting systems, thereby improving the accuracy of diagnostic assessments based on blood smear analysis. 2. Enhanced Efficiency: The implementation of effective colour segmentation methods can streamline the process of WBC counting, reducing the reliance on manual counting methods and potentially accelerating the turnaround time for diagnostic results. This efficiency can contribute to more timely medical interventions and treatment decisions. 3. Cost-Effective Solutions: The development of cost-effective computer-aided systems for WBC analysis can offer practical and economical solutions for healthcare facilities, particularly in resource-constrained settings. By automating and optimizing WBC counting processes, the research findings may contribute to cost savings and improved accessibility to accurate diagnostic tools. 4. Research and Development Opportunities: The insights gained from the evaluation of colour segmentation methods can inspire further research and development in the field of medical imaging and analysis. This may lead to the refinement of advanced image processing techniques and the integration of artificial intelligence for WBC analysis, paving the way for innovative diagnostic technologies.

[7] Guillaume F.G. Garnier et. al stands as a ground breaking in the realm of hematology. As a simple yet powerful red blood cell identification software, IdentiCyte has set a new standard in efficiency, accuracy and user friendliness. One of the most commendable features of IdentiCyte is its intuitive interface, which makes it accessible to both seasoned professionals and those new to the field. The software simplifies the complex process of red blood cell identification, streamlining workflows and allowing for more time to be dedicated to analyse rather than the intricacies of software navigation. The accuracy of IdentiCyte is truly remarkable. Leveraging state-of-the-art algorithms and image processing techniques, this software demonstrate an exceptional ability to identify and classify red blood cells with precision. This not only enhances the speed of analysis but also significantly reduces the margin of error, a crucial aspect in the field of hematology. The seamless integration of IdentiCyte into existing laboratory system is another aspect that deserves praise. Its compatibility with various microscopic platforms ensures that laboratories can incorporate this software without major overhauls, seamlessly enhancing their capabilities. The ability to analyse large data sets efficiently is particularly valuable in high through-put environments, making IdentiCyte a versatile solution for laboratories of varying sizes. Guillaume F. G. Garnier's IdentiCyte is not just a software; it is a testament to innovation in the medical science. Its impact on the field of hematology is undeniable, offering a transformative tool that empowers professionals to achieve new levels of accuracy and efficiency in red blood cell identification. As we continue to witness advancements in medical technology IdentiCyte stand out as a beacon of progress, promising a brighter future for the diagnosis and treatment of blood related disorders.

[8] Mohammad Mahmudul Alam et. al presents a novel method for the automatic identification and counting of blood cells in smear images using machine learning techniques. The authors suggest a novel method for the simultaneous detection of red blood cells (RBCs), white blood cells (WBCs), and platelets that makes use of the YOLO (You Only Look Once) algorithm. The research presents a viable alternative to the labor-intensive manual counting procedure and delivers insightful information about the use of machine learning in the field of medical picture analysis. The use of the YOLO algorithm for object detection and classification is a key highlight of the paper. The suggested method shows the capacity to precisely detect and count various blood cell types in smear images by utilizing this algorithm. To increase accuracy and get rid of multiple counting of the same object, the authors also use other techniques such intersection over union (IOU) based methods and K-nearest neighbors (KNN). This comprehensive approach contributes to the robustness and reliability of the proposed system. One of

the notable strengths of the proposed method is its ability to handle different neural network models in the YOLO back-end, allowing for flexibility and optimization based on the specific characteristics of different blood cell types. The authors also emphasize the generalizability of their approach by testing it on publicly available datasets, demonstrating its effectiveness in accurately identifying and counting blood cells, even those that are not labelled in the dataset.



Figure 6: Automatic blood cell identification and counting system block diagram

In this paper, smear images are divided into sub-images for processing, and the output is then projected back onto the original image. Including visual aids like Fig. 6 improves comprehension of how well the suggested strategy works to distinguish between RBCs, WBCs, and platelets. In the work that is being presented, the automatic identification and counting of blood cells is made possible using the YOLO (You Only Look Once) object recognition and classification method. YOLO is used to concurrently identify and categorize several blood cell types from smear images, such as platelets, white blood cells, and red blood cells (RBCs). This approach offers several key contributions to the automatic identification and counting of blood cells: 1. Simultaneous Detection: YOLO enables the detection of multiple types of blood cells within a single evaluation, allowing for efficient and comprehensive identification of different cell types in the smear images. 2. Speed and Efficiency: The YOLO algorithm is known for its speed, as it processes the entire image in one evaluation, making it well-suited for real-time applications. This speed and efficiency are particularly valuable in the context of blood cell identification and counting, where rapid analysis is essential. 3. Flexibility and Adaptability: The authors mention that different neural network models with varying depths have been explored within the YOLO back-end, demonstrating the flexibility of the approach in adapting to the specific characteristics of different blood cell types. 4. Generalization: The trained YOLO model is tested on smear images from different datasets, showcasing its ability to generalize and perform satisfactorily across diverse image sets. This generalization is a critical aspect of the algorithm's contribution to the automatic identification and counting of blood cells. Overall, the YOLO algorithm's ability to efficiently detect and classify multiple types of blood cells in smear images contributes significantly to the automation of blood cell identification and counting processes, offering speed, accuracy, and adaptability to different datasets and cell types. Using a computer-aided system for blood cell detection and counting offers several advantages over traditional manual methods, as highlighted in the presented work: 1. Efficiency: The computer-aided system, particularly the YOLO-based approach, enables rapid and automated identification and counting of blood cells in smear images. This efficiency significantly reduces the time and labour required for manual counting, making the process more streamlined and scalable. 2. Accuracy: Machine learning-based approaches, such as the one proposed, can provide high levels of accuracy in identifying and counting blood cells. By leveraging advanced algorithms and neural network models, the system can achieve precise and reliable results, minimizing the potential for human error associated with manual counting methods. 3. Automation: The computer-aided system eliminates the need for manual intervention in the counting process, allowing for fully automated analysis of blood smear images. This automation not only saves time but also reduces the dependency on human expertise, making the process more consistent and reproducible. 4. Generalization: The trained model's ability to generalize across different datasets and accurately identify cells not labeled in the original dataset demonstrates the system's adaptability and robustness. This generalization is a key advantage over manual methods, which may be limited by human subjectivity and expertise. 5. Speed: The computer-aided system, particularly when utilizing the YOLO algorithm, offers rapid processing and real-time analysis of blood smear images. This speed is essential for applications requiring quick turnaround times, such as clinical diagnostics and research.

As the work presented shows, the trained model may be used to smear photos from various datasets. Using smear photos from a different dataset that contained better quality images than the original dataset used for training, the authors tested the trained model. The photos were split into grids to make sub-images in order to adjust to the new dataset. Each sub-image was then processed separately in the suggested pipeline. The test results demonstrated that the trained model could generalize and function well on the new dataset. As seen in figure 7, the model correctly recognized platelets, WBCs, and RBCs. It even counted several cells that the dataset did not classify.



Fig 7(a)



Fig 7(b)

Figure 7: Detection of blood cells in a picture from the dataset (a) picture split into three-by-three grids (b) Combined result

[9] Shin-Jye Lee et. al presents a significant contribution to the field of medical diagnostics by leveraging deep learning techniques for complete blood cell (CBC) detection and counting. The authors discuss the drawbacks of conventional techniques and suggest a brand-new deep learning architecture based on CNN to increase the precision and effectiveness of CBC counting. The three primary categories of CBC counting-related research—automated analysers, image processing-based techniques, and machine learning-based techniques—are thoroughly reviewed by the writers. They highlight how machine learning-based techniques, in especially deep learning, can be used to identify traits that are counterintuitive but nevertheless beneficial for better CBC counting.

A CNN-based model, namely VGG-16, is included into the suggested architecture for the purpose of feature extraction and basic feature map development. The model's capacity to speculate on the locations of blood cells is improved by the incorporation of the Region Proposal Network (RPN) from Faster R-CNN. Blood cell detection and classification are accomplished with consideration and thoroughness thanks to the application of the RoI (Region of Interest) pooling layer, which further refines the feature vectors. The given experimental results offer important new information on how well the suggested models work. A quantitative assessment of the model's performance in identifying and categorizing red blood cells, white blood cells, and platelets is provided by the evaluation metrics, which include precision, recall, and F1-score. The comparison of different models under varying conditions adds depth to the analysis and underscores the robustness of the proposed deep learning architecture.

For improved CBC counting, deep learning algorithms can extract non-intuitive but valuable features. CNNs, in particular, are particularly good at learning and extracting features. Furthermore, deep learning models are frequently used in large data information processing and are simple to integrate into information systems.Computer-aided object detection using deep learning algorithms represents a significant advancement over traditional methods, offering improved accuracy, efficiency,

automation, adaptability, and integration. The research on complete blood cell detection and counting using deep neural networks has several potential implications for the field of medical diagnostics and patient care: 1. Improved accuracy: Deep learning algorithms can improve the accuracy of CBC counting, leading to more reliable diagnoses and better patient care. 2. Efficiency: By reducing the time and manpower required for CBC counting, deep learning algorithms can improve the efficiency of medical diagnostics and reduce costs. 3. Automation: Deep learning algorithms can automate the process of CBC counting, reducing the reliance on manual manipulation and the limitations of traditional automated analysers. 4. Early detection: Improved accuracy and efficiency in CBC counting can lead to earlier detection of diseases and conditions, allowing for earlier intervention and better patient outcomes. 5. Personalized medicine: By providing more accurate and efficient CBC counting, deep learning algorithms can enable more personalized and targeted medical treatments.

[10] Blood cell counting is a crucial component of clinical diagnostics and a vital measuring tool for evaluating an individual's general health, according to Francesca Isabelle F et al.

Traditionally this process involved manual counting using hemocytometer a method notorious for being time consuming and prone to errors. Recognizing the need of errors for a more efficient and accurate solution, recent advancement in machine learning have paved the way for automated counting tasks. The fifth iteration of the "You Only Look Once" (YOLOv5) object identification technique was used in this groundbreaking study to automatically identify and count white blood cells (WBCs) in images of pig blood smears. YOLOv5 stood out among other methodologies due to its exceptional combination of speed and accuracy. The choice of YOLOv5 was strategic, emphasizing the significance of both swiftness and precision in clinical setting. Speed is crucial for timely diagnoses, while accuracy is paramount for reliable results. The data set used in this study was meticulously curated to cater specifically to the WBC counting task, ensuring that the model was trained on a diverse and representative set of images. The experimental results of this study present a compelling case for the efficiency of YOLOv5 in automating WBC detection and counting. The system exhibited the remarkable speed and efficiency, achieving an impressive accuracy rate of 89.25%. Furthermore the mean average precision at 0.5 intersection over union threshold (mAP 0.5) reached an outstanding level of 99%. These findings underscore the potential of YOLOv5 as a transformative tool in clinical diagnostics, offering a swift and reliable solution to an essential aspect of healthcare. The integration of machine learningbased approaches in blood cell counting not only expedites the diagnostic process but also minimizes the likelihood of human error, thereby enhancing the overall accuracy and reliability of clinical assessments. As we embrace the future of medical technology, studies like these pave the way for innovative solutions that can redefine the landscapes of healthcare diagnostics.

III. CONCLUSION

1. With a consistent, task-focused interface, Scikit-learn provides a wide variety of machine learning techniques, both supervised and unsupervised. This makes it simple to compare approaches within a given application. The algorithms' high-level language implementation makes it possible to use them as the cornerstone of use case-specific strategies, such those used in medical imaging.

2. In the field of medical image processing, Tomari's research work on a computer-aided method for red blood cell classification offers significant advances. The paper's scientific rigor is demonstrated by the integration of cutting-edge approaches and the extensive experimental evaluation.

3. The use of Hough circle detection in the proposed method enhances the precision and reliability of RBC counting by effectively identifying and isolating RBCs within digital images, ultimately improving the accuracy of blood cell analysis in clinical laboratory settings.

4. This work describes and assesses the use of PRNN to count and categorize biconcave, stomatocyte, flat-disc, and echinostomatocyte red blood cells in a multi-type sample using the 2-D and 3-D features of the RBCs acquired using DHM. Their tests show that more useful information for RBC categorization is contained in the 3-D characteristics.

5. The paper effectively demonstrates the potential of DIP in revolutionizing traditional blood cell counting methods and provides a strong foundation for future research in area of biomedical engineering.

6. The research findings on white blood cell segmentation have the potential to impact medical diagnostics and treatment by enhancing diagnostic accuracy, improving efficiency, offering cost-effective solutions, and stimulating further advancements in the field of medical imaging and analysis.

7. With IdentiCyte, they offer a straightforward interface for identifying the shape and focuses on user-friendliness enables the advantages of automated cell recognition without requiring extensive time to learn the specifics and functions of specialized software to operate a dedicated machine.

8. The paper offers a well-structured and insightful exploration of the proposed method. The advantages of using a computeraided system for blood cell detection and counting, as demonstrated in the presented work, include improved efficiency, accuracy, automation, generalization, and speed compared to traditional manual methods.

9. In conclusion paper presents a compelling case for the adoption of deep learning techniques in the field of medical diagnostics. The proposed architecture and experimental results showcase the potential of CNN-based models to revolutionize

CBC counting, offering improved accuracy and efficiency and automation of medical diagnostics, leading to better patient care and outcomes.

REFERENCES

- 1. Fabian Pedregosa, Ga"el Varoquaux, Alexandre Gramfort, Vincent Michel, Bertrand Thirion, "Scikit-learn: Machine Learning in Python", Journal of Machine Learning Research 12,2825-2830, 2011
- Razali Tomari, Wan Nurshazwani Wan Zakaria, Muhammad Mahadi Abdul Jamil, Faridah Mohd Nor, Nik Farhan Nik Fuad, "Computer Aided System for Red Blood Cell Classification in Blood Smear Image", Elsevier, Procedia Computer Science 42 (2014), International Conference on Robot PRIDE 2013-2014 - Medical and Rehabilitation Robotics and Instrumentation, ConfPRIDE 2013-2014, 206-2016, 2014
- 3. Pawan Agrawal, Pradipti Verma, "Automated Detection and Counting of Red Blood Cell using Image Processing Techniques", International Journal of scientific research and management (IJSRM), Vol.3, Iss.4, PP.2692-2695, 2015
- Keyvan Jaferzadeh, Inkyu Moon, "Human red blood cell recognition enhancement with three-dimensional morphological features obtained by digital holographic imaging", J. Biomed. Opt. 21(12), 126015, doi: 10.1117/1.JBO.21.12.126015, 2016
- 5. Dyanesh Varun D, Kandluri Reddy, Vasavi, Abirami S, "Blood Cell Count using Digital Image Processing", Proceeding of 2018 IEEE International Conference on Current Trends toward Converging Technologies, Coimbatore, India, 978–1–5386–3702–9/18, 2018
- Syadia Nabilah Mohd Safuan, Mohd Razali Md Tomari, Wan Nurshazwani Wan, "White blood cell (WBC) counting analysis in blood smear images using various color segmentation methods", Elsevier, Measurement 116 (2018), 543– 555, 2018
- 7. Guillaume F.G. Garnier, Clare A. Manderson, Saveen Giri, Gil Garnier, "IdentiCyte: Simple red blood cell identification software", Elsevier, SoftwareX 9 (2019), 223–229, 2019
- Mohammad Mahmudul Alam, Mohammad Tariqul Islam, "Machine learning approach of automatic identification and counting of blood cells", Healthcare Technology Letters, Vol. 6, Iss. 4, pp. 103–108, doi: 10.1049/htl.2018.5098, 2019
- 9. Shin-Jye Lee, Pei-Yun Chen and Jeng-Wei Lin, "Complete Blood Cell Detection and Counting Based on Deep Neural Networks", Appl. Sci. 2022, 12, 8140. https://doi.org/10.3390/app12168140, 2022
- Francesca Isabelle F. Escobar, Jacqueline Rose T. Alipo-on, Jemima Louise U. Novia, Myles Joshua T. Tan, Hezerul Abdul Karim, Nouar AlDahoul, "Automated counting of white blood cells in thin blood smear images", Elsevier, Computers and Electrical Engineering Journal, 108 (2023) 108710, 2023

MASTER OF BUSINESS ADMINISTRATION

Performance Analysis of IPO's issued during third quarter of FY 2023-24

Dr. K.P. Galani Associate Professor SVERI's College of Engineering, Pandharpur Mobile: 9028303077 kamalgalani86@gmail.com

Abstract: Capital market is a market where buyers and sellers engage in trade of financial securities like bonds, stocks, etc. The buying/selling is undertaken by participants such as individuals and institutions. Capital markets help channelize surplus funds from savers to institutions which then invest them into productive use. Generally, this market trades mostly in long term securities. Capital market consists of primary markets and secondary markets. Primary Market or an initial public offering (IPO) refers to the process of offering shares of a private corporation to the public in a new stock issuance for the first time. An IPO allows a company to raise equity capital from public investors. 100's of privately held companies use Initial Public Offer (IPO) every year to raise fund and get listed at stock exchanges (BSE and NSE).

The transition from a private to a public company can be an important time for private investors to fully realize gains from their investment as it typically includes a share premium for current private investors. Meanwhile, it also allows public investors to participate in the offering. The total amount raised during third quarter of financial year 2023-24 was Rs. 26425.83 crores by 28 IPO's to meet their different strategies.

Keywords: Capital Market, Primary Market, IPO

I Introduction:

IPO or Initial Public Offer is a way for a company to raise money from investors for its future projects and get listed to Stock Exchange. Or An Initial Public Offer (IPO) is the selling of securities to the public in the primary stock market.

From an investor point of view, IPO gives a chance to buy shares of a company, directly from the company at the price of their choice (In book build IPO's). Many a times there is a big difference between the price at which companies decides for its shares and the price on which investor are willing to buy share and that gives a good listing gain for shares allocated to the investor in IPO.

From a company prospective, IPO help them to identify their real value which is decided by millions of investor once their shares are listed in stock exchanges. IPO's also provide funds for their future growth or for paying their previous borrowings.

There are several benefits for being a public company, namely:

- Supporting and diversifying equity base
- Enabling cheaper access to capital
- Exposure, prestige and public image
- Attracting and retaining better management and employees through liquid equity participation
- Facilitating acquisitions
- Creating multiple financing opportunities: equity, convertible debt, cheaper bank loans etc.
- Increased liquidity for the equity holder

Total 28 Mainline IPO's were issued during the third quarter of financial year 2023-24. The total amount raised during same was Rs. 26425.83 crores. The present study focuses on the performance analysis of IPO's issued during the third quarter of Financial year 2023-24.

II Objectives of the Study:

- 1. To understand the procedure of IPO.
- 2. To know the key role of different participants in IPO procedure.
- 3. To know the IPO's issued during third quarter of financial year 2023-24
- 4. To know the total amount raised by companies during third quarter year of financial year 2023-24.
- 5. To evaluate the performance of IPO's issued during third quarter of financial year 2023-24

III Scope and limitations:

- 1. The study covers IPO issued only during third quarter of Financial year 2023-24
- 2. The study covers only Mainboard IPO.
- 3. The study is based on secondary data collected from different sources. Any error in same will be part thereof.

IV Steps involved in IPO Procedure

Step 1: Hire an investment bank Step 2: Register with SEC Step 3: Draft the Red Herring document Step 4: Go on road show Step 5: IPO is priced Step 6: Available to public Step 7: Going through with the IPO

V Intermediaries of IPO Issue:

1. Merchant Bankers: Merchant Banker is entity who is engaged in the business of issue management either by making arrangements regarding selling, buying or subscribing to securities or acting as manager, consultant, adviser or rendering corporate advisory service in relation to such issue management.

2. Registrars and Transfer Agents: Registrars to an issue are entities, who on behalf of anybody corporate collect applications from investors in respect of an issue, keep proper record of applications and monies received from investors and assists body corporate to determine basis of allotment, process and dispatch allotment letters, refund orders or certificates in respect of an issue.

3. Syndicate members: Syndicate members procure bids for an IPO from institutional and retail investors. Brokers registered with SEBI work as syndicate members for an IPO.

4. Underwriters: Underwriters agree to subscribe to the securities that are not subscribed to by the public or shareholders in cases of issue of securities. In exchange for this undertaking, they are paid a commission. They can be merchant bankers or stock brokers or other registered underwriters under the SEBI guidelines

5. Bankers to an issue: Bankers collect application forms along with application moneys. They then deliver the application forms to the registrar with detailed schedules providing provisional and final certificates as per the schedule agreed. They also ensure refund of money in case of fully or partly rejected applications. They also assist in post issue reconciliation.

Sr. No.	Company Name	Amount raised (Rs. in Cr.)		
1	Innova Captab Limited	570		
2	Azad Engineering Limited	740		
3	RBZ Jewellers Limited	100		
4	Happy Forgings Limited	1008.59		

Table No. 1 Table showing List of companies and amount of money raised through IPO's

Sr. No.	Company Name	Amount raised (Rs. in Cr.)	
5	Credo Brands Marketing Limited	549.78	
6	Suraj Estate Developers Limited	400	
7	Motisons Jewellers Limited	151.09	
8	Muthoot Microfin Limited	960	
9	Inox India Limited	1459.32	
10	DMS Industries Limited 1200		
12	Flair Writing Industries Limited 593		
13	Fedbank Financial Services Limited	1092.26	
14	Gandhar Oil Refinery (India) Limited 500.69		
15	Tata Technologies Limited	3042.51	
16	Indian Renewable Energy Development Agency Ltd	2150.21	
17	ASK Automotive Limited	834	
18	Protean eGov Technologies Limited	490.33	
19	ESAF Small Finance Bank Limited	463	
20	Honasa Consumer Limited	1701.44	
21	Cello World Limited	1900	
22	Blue Jet Healthcare Limited	840.27	
23	IRM Energy Limited	545.40	
24	Plaza Wires Limited	71.28	
25	Valiant Laboratories Limited	152.46	
26	Updater Services Limited	640	
27	Manoj Vaibhav Gems 'N' Jewellers Limited	270.20	
28	JSW Infrastructure Limited	2,800	
	Total	26425.83	

Sr. No.	Company Name	Listed On	Issue Price	Listing Day Close	Listing Day Gain	Current Price 18 th January 2024	Profit/L oss
1	Innova Captab Limited	Dec 29, 2023	448	545.15	21.69%	518.85	15.81%
2	Azad Engineering Limited	Dec 28, 2023	524	677.1	29.22%	665.35	26.98%
3	RBZ Jewellers Limited	Dec 27, 2023	100	104.99	4.99%	215.15	115.15%
4	Happy Forgings Limited	Dec 27, 2023	850	1029.8	21.15%	963	13.29%
5	Credo Brands Marketing Limited	Dec 27, 2023	280	312.5	11.61%	267.9	-4.32%
6	Suraj Estate Developers Limited	Dec 26, 2023	360	334.3	-7.14%	342.55	-4.85%
7	Motisons Jewellers Limited	Dec 26, 2023	5.5	101.18	83.96%	222.24	304.07%
8	Muthoot Microfin Limited	Dec 26, 2023	291	266.2	-8.52%	230.25	-20.88%
9	Inox India Limited	Dec 21, 2023	660	939.9	42.41%	852.75	29.2%
10	DOMS Industries Limited	Dec 20, 2023	790	1330.85	68.46%	1426.65	80.59%
11	India Shelter Finance Corporation Limited	Dec 20, 2023	493	543.5	10.24%	580.45	17.74%
12	Flair Writing Industries Limited	Dec 1, 2023	304	452.7	48.91%	335.3	10.3%
15	Tata Technologies Limited	Nov 30, 2023	500	1314.25	162.85%	1150.15	130.03%
16	Indian Renewable Energy Development Agency Ltd	Nov 29, 2023	32	59.99	87.47%	123.1	284.69%

Table No. 2 Table showing List of companies and Performance evaluation of IPO's

Sr. No.	Company Name	Listed On	Issue Price	Listing Day Close	Listing Day Gain	Current Price 18 th January 2024	Profit/L oss
17	ASK Automotive Limited	Nov 15, 2023	282	310.2	10%	292.75	3.81%
18	Protean eGov Technologies Limited	Nov 13, 2023	792	883	11.49%	1286.85	62.48%
19	ESAF Small Finance Bank Limited	Nov 10, 2023	60	69.05	15.08%	69.28	15.47%
20	Honasa Consumer Limited	Nov 7, 2023	324	337.15	4.06%	468.1	44.48%
21	Cello World Limited	Nov 6, 2023	648	791.9	22.21%	869	34.1%
22	Blue Jet Healthcare Limited	Nov 1, 2023	346	395.85	14.41%	369.3	6.73%
23	IRM Energy Limited	Oct 26, 2023	505	472.95	-6.35%	535.4	6.02%
24	Plaza Wires Limited	Oct 12, 2023	54	80.23	48.57%	110.55	104.72%
25	Valiant Laboratories Limited	Oct 6, 2023	140	169.05	20.75%	187.15	33.68%
26	Updater Services Limited	Oct 4, 2023	300	283.85	-5.38%	336.5	12.17%
27	Manoj Vaibhav Gems 'N' Jewellers Limited	Oct 3, 2023	215	215.65	0.3%	320.8	49.21%
28	JSW Infrastructure Limited	Oct 3, 2023	119	157.3	32.18%	209.85	76.34%

(Source: chittorgarh.com)



Fig.1: Figure showing performance of IPO since inception

VI Conclusion:

IPOs are issued by smaller, younger companies seeking capital to expand, as well as by large privately owned companies looking to expand & become publicly traded. When a company lists its securities on a public exchange, the money paid by investors for the newly-issued shares goes directly to the company (in contrast to a later trade of shares on the exchange, where the money passes between investors). The above study shows 86% IPO's issued generated profit since their inception. Total Rs. 26425.83 crores were raised during the third quarter of financial year 2023- 24 which is huge amount for companies to meet their strategic requirement. Out of 28 IPO's Only 4 companies IPO gave negative result on listing day. Also 5 companies have given more than 100% profit since their listing on stock exchange.

VII Reference

- Jeelan Basha V, Bhadrappa haralayya, Nitesh S Vibhute (2021): A study on Performance Evaluation of Initial Public Offering (IPO) Journal of Advanced Research in Public Policy and Administration Volume 3, Issue 2 202, Pg. No. 12-26
- Suresh. S. & Yogalakshmi. C (2016) IPO Research in India- A Literary Review. Volume- 5, Issue-6 ISSN No 2277-8160
- 3. Kumar, SSS (2007): 'Short run and Long run performance of Bookbuilt IPO's in India', International Journal of Management Practices and Contemporary Thoughts, Vol, issue pp19-29

Internet Websites:

- 1. www.sebi.gov.in
- 2. www.bseindia.com
- 3. www.nseindia.com
- 4. www.moneycontrol.com
- 5. www.Chittorgarh.com

A STUDY ON IDENTIFYING FACTORS AFFECTING PRE-PURCHASE CUSTOMER DISSATISFACTIONN WITH RESPECT TOMOZE PROPERTIES

Mr. Varshaket A. Wagh¹

M. M Bhore²

(Student of Department of Master of Business Administration, Shree Vitthal Education and Research

Institute Pandharpur, Maharashtra)¹

(Research scholar of AISSMS Institute of Management Pune, Maharashtra)²

Email -ID : varshaketwagh@gmail.com¹

:mmbhore@coe.sveri.ac.in²

Abstract: The real estate sector is mostly driven by cities in developing countries. People from small villages and towns are moving to cities to improve their health. The increase in business around the big city has led to the expansion of the city. In addition, the governments of many countries, including the United States and Australia, provide long term loans at low interest rates and special loa ns to first time home buyers. Additionally, the governments of many countries, including Poland, the United States, and Canada, run programs such as golden visas and affordable housing programs to encourage home buyers. India has become an important business centre, especially in the field of services. Its good population and strong economic growth make the country an attractive destination for real estate investors because real estate demand is determined by the economy, business development and population trends.

In addition, what prevents customers from purchasing from Moze Properties is the customers' preferences when purchasing products, features, or products and services that they consider when purchasing. This effort involved the collaboration of non-purchasers of Moser Properties.

Key words:

- Consumer behavior
- ➢ Pre-purchase
- Dissatisfaction
- ➢ Real estate
- Housing market
- Property buyers
- Customer experience
- Purchase decision
- ➢ Buyer's remorse
- \blacktriangleright Real estate developers
- Construction quality
- Customer satisfaction
- ➢ Market research
- Property investment
- Customer complaint

I Objectives

- 1. Before purchasing, check out the features of Moze devices that customers are unhappy with.
- 2. Identify factors affecting buyer behaviour
- 3. Examining consumers' real estate purchasing behavior
- 4. Evaluating the impact of online reviews and word-of- mouth marketing on customer perceptions of Moser Hotels

II Brief Review of Literature

The purpose of the literature review is to clarify research questions, expand basic research. Develop knowledge in this research area and create a screening model of the research problem. Many authors have devoted their efforts to examining home buyer behaviour from their own perspectives. This review is based on historical research and literature on homeowners' decision making processes in the real estate industry. While more research has been done abroad and numerous reviews have been published in foreign journals and books, very little research has been done in India, especially in Andhra Pradesh.

This day is divided into two parts. Chapter 1 presents the results of previous research conducted on various topics and situations in the real estate sector in India and abroad. In the second section, the results of behavioral research in the real estate sector are presented and research based on behavioral finance theory is introduced. The first part covers the general factors that affect people's decision to purchase real estate, and the second part covers behaviours. The re view included research conducted in various areas of the real estate industry. Below are excerpts from various articles, documents, discussion papers and books published by different authors in this research?

III References

1. Deepak Murlidhar Sundrani (2018): Aims to identify the factors affecting the purchasing decision of various houses/apartments in Pune, India. The survey was conducted on the outskirt s of Pune city and a total of 284 respondents participated in the survey. The research reached out to recent buyers of 1 BHK, 2 BHK and 3 B HK houses and considered ten factors to examine that are most important in each home category. The analysis shows that for a 1 BHK home buyer the most important factor is price, for a 2 BHK home buyer it is of no particular importance and for a 3 BHK home buyer the most important factor is product and location. This study shows that new buyers of different types of housing differ in the importance of various factors. The study also concluded that home buying education is beneficial for participants in the real estate industry.

2. S.Khaviy, S.Manoj, V. Saathiyapriya (2018): An attempt to determine the impact of customer satisfaction in residential construction. Customer satisfaction is an important aspect in business development and i s one of the measures of business performance. Er. Kandswamy's customer base was selected f or the survey and qualitative analysis was used to analyse the association between significant determinants of consumers purchase behaviour. A total of 112 consumers were participated in the survey and convenient sampling method was used to select respondents for this study. In this work data analysis was done by using like percentage analysis, Factor analysis and cluster analysis. Concluding point is the purchase decisions are significantly influenced by income, social communal, personal Communal factors.

IV Data interpretation :

1) Have you considered construction technology used by Mauze properties before making purchase decision?
| Particular Rs | No. of Respondents | Percentage |
|---------------|--------------------|------------|
| Yes | 33 | 58.92% |
| No | 23 | 41.08% |
| Total | 56 | 100% |

Table No:1 Mauze properties before making purchase decision



Fig No:1 Mauze properties before making purchase decision

Interpretation :

From the above Table No.1 and diagram No 1 it is found that there are 58.92% of respondents says that yes we have considered construction technology used by Mauze properties before making purchase decision and 41.08% of respondents says that no, we have not considered construction technology used by Mauze properties before making purchase decision.

3) How important is the proximity of amenities (such as schools, hospital shopping centers) when considering a property from Moze properties?

Particulars	No. of Respondents	Percentage
Very important.	26	46.42%
Important	14	25%
Neutral.	0	00%
Not important	16	28.58%
Total	56	100%



Fig No: 2 Proximity of amenities

Interpretation :

From the above Table No.2 and diagram No 2 it is found that there are 46.42% Of respondents says that its very important proximity of amenities (such as schools, hospitals, shopping centers) when considering a property from Moze properties 28.58% of respondents says that its not important the proximity of amenities (such as schools, hospitals, shopping centers) when considering a property from Moze properties, 25% of respondents says that proximity of amenities (such as schools, hospitals, shopping centers) when considering a property from Moze properties, 25% of respondents says that proximity of amenities (such as schools, hospitals, shopping centers) when considering a property from Moze properties, 25% of respondents says that proximity of amenities (such as schools, hospitals, shopping centers) when considering a property from Moze properties is important . and 0 respondents are with neutral.

3). How important is the reputation and credibility of the real estate agency in your decision- making process?

Particulars	No. of Respondents	Percentage
Very important	27	48.21%
Important	15	26.78%
Neutral	06	10.71%
Not important	0	0%
N/A(already purchased	8	14.29%
Total	56	100%

Table No: 3 Reputation and Credibility of the real estate agency



Fig. No. 3: Reputation and Credibility of the real estate agency

Interpretation :

From the above Table No.3 and diagram No 3 it is found that there 48.21% of respondents says that the reputation and credibility of the real estate agency is very important in decision-making process. 26.78% of respondents says that yes the reputation and credibility of the real estate agency is important in decision-making process. 14.29% of respondents had already purchased a property. 10.71% of respondents says that we don't give any importance to this that means they are neutral about this. And 0 of respondents says that No , the reputation and credibility of the real estate agency is not

important in decision-making process.

4) What are the main reasons that have prevented you from purchasing a property from Moze?

Particulars	No. of Respondent s	Percentage
High prices	4	7.14%
Lack of desired features	18	32.14%
Poor customer service	9	16.07%
Reputation concerns	19	33.92%
Other	6	10.71%
Total	56	100%

Table No: 4 Reasons for preventing to purchase Moze property

Fig. No. 4 Reasons for preventing to purchase Moze property



Fig. 4 Reasons for preventing to purchase Moze property

Interpretation:

From the above Table No.4 and diagram No 4 it is found that there are , 33.92% of respondents gave reputation concerns as the main reason thar have prevented them from purchasing a property from Moze properties, 32.14% of respondents gave, lack of desired features as the main reason thar have prevented them from purchasing a property from Moze properties, 16.07% of respondents gave poor customer service as the main reason thar have prevented them from purchasing a property from Moze properties and 10.71% of respondents specified their own reasons which prevented them from purchasing a property from Moze properties. lack of desired amenities as the main reason thar have prevented them from purchasing a property from Moze properties 7.14% of respondents gave high prices as the main reason thar have prevented them from purchasing a property from Moze properties 7.14% of respondents gave high prices as the main reason thar have prevented them from purchasing a property from Moze properties 7.14% of respondents gave high prices as the main reason thar have prevented them from purchasing a property from Moze properties 7.14% of respondents gave high prices as the main reason thar have prevented them from purchasing a property from Moze properties 7.14% of respondents gave high prices as the main reason thar have prevented them from purchasing a property from Moze properties 7.14% of respondents gave high prices as the main reason thar have prevented them from purchasing a property from Moze properties 7.14% of respondents gave high prices as the main reason thar have prevented them from purchasing a property from Moze properties 7.14% of respondents gave high prices as the main reason thar have prevented them from purchasing a property from Moze properties

V Conclusion:-

The conclusion from this study is that there is a desire among consumers to explore various features. Moze Properties needs to continue to offer buyers a variety of properties to meet their needs. Additionally, consumers rely on reputation and reliability when making purchasing decisions. Therefore, Mo Properties must first provide quality construction, complete the project on time, provide good customer service and transparent business to build trust and confidence with its customers.

Location is also important and hence it will be beneficial to build properties in prime locations in Pune. Additionally, hiring a real estate agent and targeting digital marketing will help you attract buyers. Overall, by using these strategies, Moze Properties can improve its business and meet the needs of its customers. Through their research, researchers found many surprising things when evaluating the properties of Moser Properties.

These factors include the lack of state-of-theart equipment such as modern equipment or new features, which may affect the overall legacy. Additionally, the real estate contract's lack of specificity will make it less at tractive to buyers looking for attractive pro motions or discounts. Additionally, reputation issues play an important role in consumers' decision making. When choosing a device, buyers often consider the reputation of the develop er or company behind the device. Any negative thoughts or doubts regarding Moze Properties' reputation will deter customers from purchasing. Another complaint is the average customer. Customers want excellent service throughout the entire purchasing process, from inquiries to aftersales support. If the level of customer service provided by Moze.Properties does not meet expectations, this may result in dissatisfaction and reluctance to proceed with the purchase. Additionally, the location of the property is also important for buyers. If the property is not in a desirable or convenient area, it can be a major hurdle for buyer s, even if

there are other interests.

To solve these complaints, Moze Properties should focus on the development of all it s customers. This can be done by improving the quality of its products, providing attractive services and incentives, working to manage and improve its reputation, and providing good service to every customer. Additionally, choosing the right location for your project can affect the needs and requirements of your property. Thanks to these measures, Moser Real Estate can overcome these complaints and create an attractive and satisfying real estate for customers. Th is will help you attract more buyers and create a good reputation in the industry.

V Reference:

1. Helen X. H. Bao (2016): Journal title: "Architectural Abstract" page 31Mostofa Kamal, Md. Omar Faruk Sarker & Shah Alam Kabir Pramanik (2016)Name of the magazine: Architectural Digest page No:51

- 2. Julius Chia, Amran Harun (2016)Name of the magazine: Architectural Digest page No:53
- 3. S.Khaviy, S.Manoj, V.Saathiyapriya (2018): Name of the magazine :Real Estate Today page No: 21
- 4. <u>https://scholar.google.com/</u>
- 5. <u>https://www.jstor.org/</u>
- 6. <u>https://www.researchgate.net/</u>
- 7. <u>https://www.proquest.com/i nd exol</u>
- 8. https://www.academia.edu/
- 9. <u>https://www.realtor.com/</u>

Financial awareness survey among the students at Pandharpur's rural region'

Pravin S. More¹, Komal P. Kondubhairi², Sagar D. Sarik³
¹Research Scholar, BVDU, AKIM Solapur,
²Research Scholar, BVDU, IMRDA Sangli,
³Assit. Prof. SVERI's college of Engineering Pandharpur
¹Email: Pravin.more8914.gmail.com,
²Email: komalkondubhairi7@gmail.com,

³Email: sdsarik@coe.sveri.ac.in,

Abstract: This research paper addresses the critical domain of financial awareness among students in Pandharpur Tehsil, seeking to evaluate their comprehension, knowledge, and practices related to financial matters. Financial awareness serves as a fundamental life skill that empowers individuals to make well-informed and responsible financial decisions, thereby influencing their economic well-being.

The study employs a mixed-methods approach, combining surveys and interviews to gather comprehensive data on the financial literacy levels of students. A structured questionnaire is distributed among a diverse sample of students, encompassing fixed age groups and educational backgrounds. Additionally, in-depth interviews with selected participants provide qualitative insights into their attitudes, perceptions, and experiences concerning financial matters.

Study aims to champion financial inclusion and equip students with the skills to confidently navigate the complex financial terrain. By suggesting the incorporation of financial literacy education into the academic curriculum, the research envisions cultivating a generation well- versed in financial matters, capable of making informed decisions and ensuring a more resilient economic future.

Key words: Financial literacy, financial Instruments, financial knowledge, financial awareness

I Introduction

Exploring financial literacy among students in the rural confines of Pandharpur represents a crucial initiative to grasp the details of 'Financial Independence.' Financial literacy, compressing knowledge, skills, and attitudes pertinent to financial matters, assumes a critical role in molding individuals' capacity to make well- informed and responsible decisions regarding their finances. Within the demographic of students in the rural expanse of Pandharpur, deciphering their level of financial literacy gains importance in the broader quest to nurture economic empowerment and self-reliance. This study endeavors to delve into the specific dynamics of financial literacy among students in this locale, concentrating on how it contributes to achieving financial independence. By illuminating these facets, the research seeks to furnish valuable insights that can guide targeted interventions and educational strategies, fostering an enhancement of financial literacy and the encouragement of

financial independence among students in the Pandharpur rural region.

II Objective

- 1. To know about n financial awareness.
- 2. To know the awareness about various avenues of finance.
- 3. To know the awareness about the various financial institutes.
- 4. To study financial knowledge.

III Research Methodology

Sampling: Simple random sampling along with a standard questionnaire was used to collect the data. The target population included people belonging to the age group of 06 - 20 years.

Tools The questionnaire consisted of a confidentiality note, a disclaimer that the survey was meant for only those belonging to the age group of 18 - 30 years and residing in Pandharpur and a subject matter section. The subject section consisted of generic multiple choice, checkbox and situation specific scale development questions pertaining to the topic of study.

IV Data Collection:

Primary data collection through a Pilot study and conducted with 50 responses to test the credibility of the uestionnaire.

V Data Analysis

Option	Response	Percentage
To achieve long-term financial goals	23	46
To spend all your money quickly	17	34
To show off to friends	10	20
Total	50	100

Table No. 1Reasons for save money?

Interpretation 01

Based on the given data, the majority of respondents (46%) believe that saving money is important to achieve long- term financial goals. Meanwhile, 34% of respondents also recognize the importance of saving money to achieve long-term goals, indicating a similar perspective as the majority. Only a small percentage of respondents (20%) indicated a preference to spend all their money quickly. Lastly, no respondents selected the option of showing off to friends as a reason to save money

Option	Response	Percentage
To achieve long-term financial goals	23	60
To spend all your money quickly	17	30
To show off to friends	10	10
Total	50	100

Interpretation 02

It appears that the majority of respondents (60%) believe that the meaning of "budgeting" is planning how to spend and save money. 30% believe it is buying things impulsively, 10% believe it is borrowing money from friends, and 8% believe it is investing in real estates

Table No.3 Stocks investment		
	D	

Option	Response	Percentage
Buying shares in company	19	38
Putting money in a piggy bank	13	26
Loaning money to a friend	10	20
None of these	8	16

Interpretation 03

Based on the provided data, it appears that 38% of respondents believe that investing in stocks means buying shares in a company. 26% believe it means putting money in a piggy bank, 20% believe it means loaning money to a friend, and 16% believe it means none of these options.

Table No. 4 Which financial institution primarily offers savings and checking accounts?

Option	Response	Percentage
Investment bank	15	30
Credit union	11	22
Hedge fund	17	34
Insurance company	7	14

Interpretation 04

Based on the provided data, it seems that 34% of respondents believe that a hedge fund primarily offers savings and checking accounts. 30% believe it is an investment bank, 22% believe it is a credit union, and 14% believe it is an insurance company

Option	Response	Percentage
Machinery	42	84
Machinery	2	4
Sales Return	3	6
Interest Received	3	6

Table 05 About the Asset Awareness

Interpretation 05

Based on the provided data, it appears that 84% of respondents believe that cash is a asset. 6% believe sales return is a asset, 6% believe Interest received is a asset, and 4% believe Purchases is a asset.

Option	Response	Percentage
Cash	5	10
Equipment	3	6
Debtors	2	4
Creditors	40	80

Interpretation 06

Based on the provided data, it appears that 80% of respondents believe that creditors are a liability. 10% believe cash is a liability, 6% believe equipment is a liability, and 4% believe debtors are a liability.

VI. References

- 1. Peel, M. J., Pendlebury, M. W., & Groves, R. E. V. (1991). The determinants of students' financial awareness—some UK evidence. The British Accounting Review, 23(1), 23-48.
- 2. Dewi, V. I., Febrian, E., Effendi, N., Anwar, M., & Nidar, S. R. (2020). Financial literacy and its variables: The evidence from Indonesia. Economics & Sociology, 13(3), 133-154.
- Dewi, V. I., Febrian, E., Effendi, N., & Anwar, M. z(2020). Does financial perception mediating the financial literacy on financial behavior? A study of academic community in central Java Island, Indonesia. Montenegrin Journal of Economics, 16(2), 33-48.
- 4. Kumar, R., & Pathak, D. C. (2022). Financial awareness: a bridge to financial inclusion. Development in Practice, 32(7), 968-980.

A Study On The Technique Of Shivamurut Milk And Milk ProductsBy Using Inventory Management

Komal P. Kondubhairi¹, Pravin S. More², Sagar D. Sarik³, Amad A. Ahmed⁴

¹Research Scholar, BVDU, IMRDA Sangli ²Research Scholar, BVDU, AKIM, Solapur ³Asst. Professor at Sveri's College of Engineering Pandharpur, ⁴Asst. Professor at Sveri's College of Engineering Pandharpur,

> ¹Email: <u>komalkondubhairi7@gmail.com</u> ²Email: <u>pravin.more8914@gmail.com</u> ³Email: sdsarik@coe.sveri.ac.in ⁴Email: aaahmed@coe.sveri.ac.in

Abstract: This study delves into the inventory management techniques of Shivamrut Milk and Milk Products, exploring current practices to enhance efficiency. Utilizing quantitative and qualitative methods, the research assesses inventory turnover, order fulfillment times, and technology integration. The findings aim to identify strengths and weaknesses, offering recommendations for improvement. By addressing factors like demand fluctuations and supply chain disruptions, the study provides valuable insights to optimize inventory processes. The ultimate goal is to equip Shivamrut with strategies for maintaining a competitive edge in the dynamic dairy industry.

Keywords- Fixation of levels, Economic Order Quantity (EOQ) model, Always Better Control analysis

I INTRODUCTION:

In the dynamic landscape of the dairy industry, effective inventory management plays a pivotal role in ensuring the success and sustainability of organizations. This study focuses on examining the inventory management techniques employed by Shivamrut Milk and Milk Products. As consumer preferences evolve and market dynamics shift, the need for streamlined and efficient inventory practices becomes increasingly crucial. This introduction sets the stage for an in-depth exploration of Shivamrut's current inventory management system, combining quantitative and qualitative analyses to identify areas for improvement. The study aims to provide actionable insights that will not only optimize Shivamrut's operations but also contribute to the broader understanding of effective inventory management in the dairy sector.

II BJECTIVES OF THE STUDY:-

1. To determine and maintain optimum level of inventory management in SHIVAMRUT CO-OPERATIVE MILK PRODUCER UNION

2. To ensure better serviced to the customer.

3. To minimize the firs investment in inventories and to maximize profits

III RESEARCH METHODOLOGY:

Realizing the difficulty inherent in studying the entire population, a representative sample of the population was used for the study. As the population size of this study comprised of users of Shivamrut Milk Products in Akluj which however, become inaccessible. The study, therefore, applied the SIMPLE RANDOM SAMPLING method. Questionnaires were provided to customers at Maharashtra on their convenience.

IV BRIEF REVIEW OF LITERATURE

In order to comprehend the inventory control procedures used by various businesses as well as the tactics and variables influencing inventory control success, the study aims to review the existing literature.

Three theoretical frameworks served as the study's direction:

- 1. The hypothesis of stock diffusion
- 2. Theory of application control and

3. The theory and application of inventory control.

The primary doains—inventory modeling, decision support systems (DSS), and expert systems (ES)—are covered in the first three sections. The intelligent decision support system (IDSS), which combines the DSS and ES domains, is then discussed. This section examines the knowledge-based or intelligent inventory systems that have been created recently. A review of the published intelligent inventory management systems is included in the study's conclusion.

Developed by Harris in 1913, the Economic Order Quantity (EOQ) model is widely recognized as the first quantitative inventory model. Raymond's book is the first in-depth attempt to describe how different extensions of EOQ can be applied in real-world scenarios. Subsequent studies revealed that the EOQ model seems to be fairly immune to mistakes made when estimating demand and specifying the proper cost parameters. In addition to its historical significance, the EOQ model serves as a foundation for numerous additional models that have been created to address various scenarios. But at the time, there wasn't much use for this inventory management mathematical modeling technique. Maybe this was the case because new ideas always require time to develop, so that the original claim regarding higher performance and productivity can be validated over time and specifics can be worked out.

V TOOLS OF INVENTORY MANAGEMENT

1. ABC ANALYSIS: [Always Better Control]:

This strategy involves managing the items by assigning a value to each one and classifying them using an ABC grade.

GRADE A: Materials are closely supervised, closely controlled, and expensive, with a high value per unit. 15% to 20% of it (in terms of value)

GRADE B: The worth of the materials is moderate, and a moderate number of things are kept under moderate control. It might cover 60% of the entire cost of the merchandise.

GRADE C: Materials are more plentiful but have a lower value, and their things are not as closely monitored. It looks after a lot of things that are not very valuable.

2. EOQ ANALYISIS: (Economic Order Quantity) :

The quantity of materials to be ordered that will result in the lowest possible order placement and carrying costs is known as the economic order quantity. This strategy, often known as the size of materials to be purchased most economically, is used to reduce ordering and associated costs.

$$EOQ = \frac{2AS}{I}$$

Ordering Cost: Ordering costs are the expenses related to ordering or making material purchases.

Carrying Cost: If inventory is not carried, there will be expenses associated with keeping it on hands.

3. PERCEPTUAL INVENTORY SYSTEM

It alludes to ongoing stock checking within this system, where various material records are kept up to date with entries made as soon as the supplies are received and distributed. As a result, it is seen as an expensive inventory control strategy, despite the fact that management might gain greatly from it.

4. FAST MOVING-SLOW MOVING-NON MOVING (FSN) ANALYSIS:

Fast moving: In these analysis items are categorized based on their movement, and their movements are closely monitored.

Slow moving: Since the production department needs these things regularly, a modest quantity and level of oversight will be maintained.

Non-moving: Since the production department rarely needs certain items, fewer materials are kept in store and are given less priority.

5. PERIODICAL INVENTORY VALUATION:

Under this system, inventory valuation and checking will be done at various intervals, usually twice or three times a year. During the stock checking period, the organization's regular business will be suspended for one or two days, and all stock verification and valuation will be completed as necessary.

6. BUDGETARY CONTROL SYSTEMS:

This system prepares inventory budgets, which are compared to actual consumption figures. If there is a significant discrepancy between the budgetary and actual figures, the appropriate corrective action is taken. This is an important technique for inventory control.

VI DATA ANALYSIS AND INTERPRETATION:

A significant operational challenge in scientific inventory control is to the vast array of commodities that are stocked by different companies Applying rigorous scientific inventory control standards to each of these 10,000–100,000 unique types of stocked objects is neither desired nor feasible. Inventory control might become counterproductive with such a broad approach, which would make it more expensive than helpful. Inventory control must therefore be applied sparingly. A policy's appropriateness for inventory management may depend on an item's worth, criticality, and usage frequency. Therefore, selective inventory management is crucial to allowing us to more intelligently distribute our limited control resources over the greater subset of objects. Items in selective management are categorized into a few different groups according to their value, importance, and usage frequency. The terms ABC, VED, and FSN analyses are frequently used to refer to these analyses, respectively. An organization may use this kind of structure as the first step in implementing scientific inventory management. Mahindra CIE Auto Ltd. manages their inventory with the help of ABC analysis. This project examines the ABC analysis.

ABC ANALYSIS:

Think on the best, then on the rest, according to the ABC (Always Better Control) analysis principle. A crucial idea is highlighted by ABC analysis: "Vital few: trivial many." Businesses typically have to maintain a big inventory of items used in manufacturing and distribution. Due to resource limitations, it is not practical to maintain and control a same or appropriate level of inventory for every item. As a result, it is common practice to make genuine attempts to maintain appropriate control over the most frequently circulating things and the least frequently circulating ones. ABC analysis provides a framework for organizing products according to their annual or monthly consumption value. In other words, even if an item has a very low unit price, close and meticulous control will be exercised if it is a highly circulating item with a maximum monthly or annual consumption value, and vice versa. Because of this, the ABC analysis divides the goods into three major groups based on their monthly/annual Consumption value: A, B, and C.

VII CONSUMPTION VALUE

1. A

Items by Category Typically, 65% of the total money spent on materials is allocated to 20% of the total items. These products must be stocked in lesser amounts and subject to strict, comprehensive supervision. These things should be bought often, in little quantities each time. But a sensible course of action would be to sign a contract with the producers of these goods and arrange for their supply to be delivered in staggered batches in accordance with the buyer's production schedule. However, when demand is consistent, this will be feasible. Alternatively, frequent ordering can help maintain the inventory at a minimum.

2. B

Items in Category B are those with an average monthly or annual consumption value. For the most part, 25% of the overall sales or consumption value is made up of 30% of the items. Inventory control for this type of product needs to be more meticulous and in-depth. While they require more oversight than goods in the C category, these products require less attention and control than those in class A.

3.*C*

Category Items: The items in Category C have modest monthly/annual consumption values. Once more, 50% of all things often represent 10% of the value of sales or consumption. The least amount of inventory management attention is required

for this category of goods. Undoubtedly, a lack of control over C category items raises the cost of inventory overall, but not much.

Categories of Items

Category of Items	% of Items	%of Monthly/Yearly Consumption Value	Degree of Inventory Management
А	15	74.5	BEST
В	15	15.4	BETTER
С	70	10.1	GOOD
TOTAL	100	100	

TT 1 1	1	a .	c	•,
I abale	1.	Category	V OI	nems

Conducting ABC Analysis:

The following actions are required in order to do ABC analysis: a) Make a list of the products and calculate the approximate annual consumption (in units). b) Figure out each item's unit price, often known as cost. c) To calculate each annual use in rupees (annual usage), multiply it by the unit price (or cost). d) Sort the things according to their annual usage, starting with the item with the highest utilization and working your way down to the one with the lowest. e) Determine the total annual usage and give the result as a percentage of the total usage. Additionally, convert the number of items into a percentage of cumulative items. f) Sort the items into categories A, B, and C by graphing cumulative usage percentages versus cumulative item percentages. g) Choose the three groups' control policies.

1. INVENTORY TURNOVER RATIO

This ratio finds out the number of items inventory is turned over on an average in a year. This ratio is calculated for findings at what extent the inventory has been utilized efficiently and what proportion of working capital has been locked up in inventory.

Inventory Turnover Ratio = <u>Cost of Goods sold</u> Average Inventory

Average Inventory Ratio = $\frac{\text{Opening Balance} + \text{Closing Balance}}{2}$

Year	Cost Of Goods Sold	Average Inventory	Inventory Turnover Ratio
2018-19	2,61,76,76,507	67113235	39.00%
2019-20	2,16,96,27,251	69011984	31.44%
2020-21	2,64,95,08,364	68885358	38.46%

Table: 7.1 Inventory turnover ratio



Graph: 7.1 Inventory Turnover Ratio

Interpretatin:

From above **Table 7.1** & **Graph 7.1** Inventory turnover ratio shown that the inventory turnover to average inventory ratio, inventory turnover to average inventory ration during period of increase in year 2018-19 39.00 times, and is 2019-20 by 31.44 times, & again increase in 2020-21 by 38.46 times. So the company is having better Inventory turnover ratio.

2. TOTAL ASSET TO INVENTORY RATIO:

Measure the activity of the ability of the business to generate sales through the use of assets. It revels the efficiency in managing a utilizing the total asset.

Total Assets to Inventory = <u>Total assets</u> x 100 Ratio

Average Inventory

Table 8.1 Total Asset To Inventory Ratio

Year	Total Asset	Average Inventory	Inventor y asset to Inventory Ratio
2018-19	52,86,92,924	67113235	787.76%
2019-20	53,65,50,409	69011984	777.47%
2020-21	56,19,47,679	68885358	815.77%



Graph 8.1 Total Asset to Inventory turnover ratio:

Interpretation:

From the above table and graph 8.1 it is observed that, in year 2020-21, Total asset to inventory ratio is high. Total asset to inventory ratio during period of 2018-19 by 787.76 & it decreased in 2019-20 by 777.47 & it again increased in 2020- 21by 815.77, in general, the position of asset turnover is ups and down which in not good for company. So the company is having to maintain asset to inventory ratio.

3. RAW MATERIAL TO INVENTORY RATIO

Raw Material to Inventory Ratio = Raw material/ Average Inventpry

Year	Raw Material	Average Inventory	Raw Material to Inventory Ratio
2018-19	33,84,05,238	67113235	5.04%
2019-20	36,47,24,149	69011984	5.28%
2020-21	81,96,25,838	68885358	11.90%

Table 9.1 Raw material to Inventory turnover ratio

Graph 9.1 Raw material to Inventory turnover ratio

Interpretation:

From the above the table and graph 9.1 it is observed that, raw material to inventory ratio is increasing from 2018-19 to 2019-20 and 2020-21.

4. Work In Progress (WIP) TO Inventory Ratio:

WIP to Inventory Ratio = <u>Work In Progress</u>

Average Inventory

Table 10.1 Work In Progress (WIP) to Inventory ratio

Year	WIP [Work In Progress]	Average Inventory	WIP To Inventory Ratio
2018-19	33,84,05,238	67113235	3.19%
2019-20	36,47,24,149	69011984	2.85%
2020-21	81,96,25,838	68885358	2.79%



Graph 10.1 Work In Progress (WIP) TO Inventory Ratio

Interpretation:

From the above the table and graph 10.1 it is observed that, work in process to inventory ratio is increasing ratio is in 2018-19, year is 3.19 & decreasing in 2019-20 year i.e. 2.85 & in again decreasing in 2020-21 is 2.79.

5. FINISHED GOODS TO INVENTORY RATIO:

Finished Goods to Inventory Ratio= Finished Good/ Average Inventory

Year	Finished Goods	Average Inventory	Finished Goods To Average Inventory
2018-19	2,37,24,153	67113235	0.35%
2019-20	2,79,60,750	69011984	0.41%
2020-21	4,45,65,382	68885358	0.65%

Table 11.1 Work In Progress (WIP) to Inventory ratio



Graph 11.1 Work In Progress (WIP) TO Inventory Ratio

Interpretation:

From the above & graph 11.1, it is observed that, finished goods to inventory ratio is increased from year2018-19, 2019-20 and 2020-21

VIII FINDINGS:-

- 1. The Inventory turnover to average inventory ratio, inventory turnover to average during period of increase in year by 2018-19 31.00 times, and it decreased in 2019-20 by 31.44 times, & again increased in 2020-21 by 38.46 times. So the company is having better Inventory turnover ratio.
- 2. Total assets to inventory ratio during period of 2018-19 787.76 & it decreased in 2019-20 by 777.47 & it again increased in 2020-21 is 815.77 in general, the position of assets turnover is ups and down which in not good for company. So the company is having to maintain assets to inventory ratio.
- 3. Raw material to inventory ratio is increased from 2018-19 to 2019-20 and 2020-21.
- 4. Work in process to inventory ratio is in 2018-19 year is 3.19 & decreasing in 2019-20 year i.e. 2.85 & it again decreasing in 2020-21 is 2.79.
- 5. Finished goods to inventory ratio is increased from year 2018-19,2019-20 and 2020-21.
- 6. Maintaining an organized style.

- 7. Protecting the output from theft and harm Issuing oldest material first.
- 8. Inventory is controlled using ABC analysis; issues arise when there is insufficient stock of item A because it contains more; occasionally, if the supplier is unavailable when item A is needed, the entire production line may also stop.
- 9. Based on the monetary value of the things being used, inventory is managed using the ABC analysis method. Thus, one neglected the other significant aspects.
- 10. Some objects are discovered to be missing.
- 11. Hiring and educating dependable men for the store. Providing enough space and storage equipment like alarmist, shelves, racks, bins etc.
- 12. The above-mentioned key guidelines include maintaining current store inventory and records.

IX REFERENCE:

- 1. Agarwal, D. K. (2010). Supply chain management: strategy, cases and best practices. Macmillan.
- 2. Jhamb, L. C., & Menon, K. S. (2013). Inventory Management, Pune.
- 3. Gupta, K. (2011). Khan, MY and Jain, PK, Financial Management: Text, Problems and Case. Journal of Services Research, 11(2).
- 4. Pandey, I. M. (1995). Finance: A management guide. PHI Learning Pvt. Ltd.
- Vrat, P., & Vrat, P. (2014). Basic concepts in inventory management. Materials Management: An Integrated Systems Approach, 21-36.

Cashless Impact on Banking Customers: Trends, Challenges and Opportunities

Mahesh Kumar M¹, Gayatri Patil², Swati Kote³, Urmila Kore⁴ ¹Department of Master of Business Administration, Guru Nanak Dev Engineering College Bidar (Visvesvaraya Technological University, Belagavi, Karnataka, INDIA) Email-ID: <u>mmgmeenkera@gmail.com</u> ²Department of Master of Business Administration, Guru Nanak Dev Engineering College Bidar (Visvesvaraya Technological University, Belagavi, Karnataka, INDIA) ³Department of Master of Business Administration, Guru Nanak Dev Engineering College Bidar (Visvesvaraya Technological University, Belagavi, Karnataka, INDIA) ⁴Department of Master of Business Administration, Guru Nanak Dev Engineering College Bidar (Visvesvaraya Technological University, Belagavi, Karnataka, INDIA) ⁴Department of Master of Business Administration, Guru Nanak Dev Engineering College Bidar (Visvesvaraya Technological University, Belagavi, Karnataka, INDIA)

ABSTRACT: This sketch looks at the many ways that the move to a cashless economy will affect people who use banks. People are using digital transactions more and more, which is changing the way traditional banks do business. It also looks at what these trends mean for financial inclusion.

Customers like the ease of digital transfers and can access banking services 24 hours a day, 7 days a week through websites and mobile apps. But there are problems with digital literacy, which means that some people may find it hard to learn how to use new tools. The study looks at the economic effects by looking at the speed of transactions and how they affect economic activity, GDP growth, and business income.

In the end, this paper talks about how a cashless society would change banking customers and stresses the need for a balanced approach that deals with problems while also taking advantage of the huge possibilities that digital transactions offer. The results add to the ongoing discussion about financial technology and help policymakers, financial organizations, and people make the financial world more open and efficient.

Keywords: Digital, Transactions, Banking, Financial.

1. INTRODUCTION

In a "cashless" transaction or system, there is no physical cash involved. Instead, digital or electronic ways of payment are used. Consumers in a cashless society use electronic payment methods like credit and debit cards, mobile wallets, online banking, and other digital tools to make purchases.

Some benefits of a cashless system are:

Speed and ease of use: Digital purchases are often faster and easier to use than cash.

Security: Encryption and identification can make electronic transactions safer by keeping them from being stolen or used fraudulently.

Record-Keeping: Electronic transactions leave a trail that can be followed. This makes it easier for people and businesses to keep track of their money.

Less Theft: Since there is no cash involved, there is less chance of theft for both people and businesses.

Financing: Systems that don't require cash can help people who might not have easy access to regular banks get financial services.

Cost-effectiveness: It takes money to store and move cash around. These costs are a lot lower in a society without cash.

A cashless society does come with some problems and worries, though.

2. OBJECTIVES OF THE STUDY

To Focus on digitalization of banking sector, including ATM and POS deployment, transaction volume, and value through NEFT, RTGS, and mobile banking.

To examine the evolution of the Indian payment system and the problems and progress of the digital payment system.

To Identify and promoting a cashless economy.

3. CASHLESS IMPACT ON BANKING

The cashless impact on banking refers to the consequences and transformations within the banking sector resulting from the increasing prevalence of cashless transactions. As more individuals and businesses adopt digital forms of payment and banking services, various aspects of the banking industry are influenced. Here are key areas where the impact of going cashless is notable:

- Digital Transactions and Payment Methods:
- Reduced Cash Handling Costs:
- Shift in Banking Infrastructure:
- Enhanced Security Measures:
- Impact on Branch Operations:
- Economic and Regulatory Implications:
- Data-Driven Insights:
- Financial Inclusion Opportunities:
- > Technological Innovation and Partnerships:
- Changing Customer Expectations:
- Challenges of Digital Literacy:

4. CASHLESS IMPACT ON BANKING CUSTOMER

The transition to a cashless society has a significant impact on banking customers in various ways. Here are some of the key effects:

Convenience: **Digital Transactions:** 24/7 Access: Security: Reduced Risk of Theft: Fraud Protection: **Record Keeping:** Transaction History: Automated Tracking: Financial Inclusion: Access to Banking Services: Cost Savings: **Reduced Transaction Costs:** Challenges: Digital Literacy: Cybersecurity Concerns:

Impact on Cash-Dependent Businesses: Adjustment for Businesses: Economic Impact: Transaction Velocity: Digital Transactions: Online Banking:

5. ACCOUNT MANAGEMENT

Customers can access and manage their bank accounts online and perform various financial activities through secure banking websites or mobile applications. Mobile Banking: Mobile Apps: **Digital Wallets:** Contactless Payments: **Online Purchases:** E-commerce Transactions: Peer-to-Peer (P2P) Payments: Money Transfer Apps: Cryptocurrency Transactions: Blockchain Technology: Contactless Card Payments: NFC Technology: Automated Clearing House (ACH) Transfers: Electronic Funds Transfers: Digital Remittances: International Money Transfers: Digital Checks and Electronic Invoicing: Paperless Transactions:

6. **RESULTS**

FINANCIAL INCLUSION:

Being able to use banking services

Financial inclusion means that people and businesses can get a variety of financial services, especially those who haven't been able to get them before or who haven't been treated well enough by mainstream banks. Access to banking services is a key part of financial inclusion, which includes a number of things, such as

Simple Banking Services:

Savings Accounts: Giving people the chance to open and keep basic savings accounts is a part of financial inclusion.

Digital Payments: Promoting and making it easier for people to use digital payment methods like electronic wallets, mobile banking, and online exchanges is part of financial inclusion. In this way, people can take part in the modern economy without having to depend on cash only.

Remittances: Making remittances easy and cheap is very important, especially for people who live in rural or neglected areas and get money from family members who work in cities or abroad.

Microfinance grants small loans to entrepreneurs and small businesses that might not be able to use standard banking services. Microfinance is an example of a financial inclusion program.

Credit Services: Being able to get credit is important for gaining economic power. The goal of financial inclusion is to give more people, even those with bad credit, access to credit services

Insurance Services: Microinsurance: Part of financial inclusion is making sure that people with low incomes can get affordable insurance goods that meet their needs. This helps protect against the unexpected and keeps the economy stable.

Government Benefits and Subsidies: Direct Benefit Transfers (DBT): This is a way to make sure that people who are eligible for government benefits and subsidies can get them on time by sending the money directly to their bank accounts through digital methods.

Learning and Being Aware:

Financial knowledge: One important part of financial inclusion is promoting financial education and knowledge. It gives people the information they need to make smart financial decisions and make good use of the banking services that are offered.

Technology and new ideas: banking without branches: Mobile and internet banking, in particular, are big parts of getting to places that are hard to reach or don't get enough service. Branchless banking models use agents and technology to offer financial services in places other than standard bank branches.

Fintech Solutions: New developments in financial technology (fintech) like peer-to-peer loans, digital wallets, and blockchain-based solutions make it easier for more people to get access to financial services.

Support for regulations: proportional regulations: For financial inclusion to work, regulators must create an environment that pushes banks to serve a wide range of customers, including those with lower incomes.

Financial inclusion aims to boost economic growth, lower poverty, and give people and groups more power to take part in the economy more fully by making banking services more accessible. The business sector, governments, financial institutions, and non-governmental organizations (NGOs) are all working together on this project

Save money on costs:

Lower transaction fees:

When many financial activities move to digital and cashless modes, transaction costs go down, which is a big gain. Both people and businesses can save money by moving away from traditional paper transactions and handling cash in person. In a cashless setting, lower transaction costs are reached in this way:

Automation: Digital transactions use automated methods that help keep labor costs low. Transaction handling, account management, and reconciliation are all tasks that can be automated. This makes operations run more smoothly and requires less human input.

Transactions without paper: In traditional transactions, things like checks, bills, and receipts are often used as proof. Going digital cuts down on or gets rid of the need for paper, which saves money on printing, storage, and dealing by hand.

Getting rid of the costs of handling cash:

Handling real money comes with a lot of costs, like making and transporting bills and coins, taking precautions for safety, and keeping ATMs running. These costs are either not present at all or are greatly reduced when you do business digitally, which saves you money overall.

Fees for Processing Transactions: Processing fees for digital transactions may be cheaper than those for traditional methods. For instance, fees are usually cheaper for transfers made through online banking or payments between people who don't know each other.

Less Mistakes: Mistakes happen less often in digital systems and automated processes than in deals that are done by hand. When there are fewer mistakes, it costs less to fix them, help customers, and look into possible financial problems.

Cross-border transactions are made easier: When you do business with people in other countries online, you can usually do it more quickly and for less money than when you use traditional methods like wire payments. This helps businesses that deal with other countries because it cuts down on the costs and delays of changing currencies.

Connecting Financial Systems: When businesses connect different financial systems and platforms, they can save money. For instance, combining payment handling systems with accounting software makes financial processes more efficient by cutting down on the need to enter and match data by hand.

Economic Activity: More economic activity is often linked to faster transaction speeds. Money that moves quickly increases the desire for goods and services, which leads to more production and consumption.

The money supply (M) is equal to the potential output (Y), and the price level (P) is equal to the money supply (V). So, for GDP growth, MV = PY includes the velocity of money. When the money supply stays fixed and velocity goes up, it can help GDP grow.

Business Revenue: A market with faster transactions is better for companies because it means the market is more active. When customers and other businesses do business with each other often, businesses make more money and sell more.

Job possibilities: Higher transaction speeds can lead to more job possibilities because they make the economy work harder. As companies grow to meet rising demand, they may hire more people, which helps bring down jobless rates.

Entrepreneurship and Investment: People are more likely to start their own businesses and spend when the economy is strong and transactions happen quickly. When entrepreneurs see a growing market, they are more likely to start new businesses. Similarly, investors may be more willing to put money into businesses when the economy is changing quickly.

Monetary Policy Implications: To figure out how well monetary policy is working, central banks and officials keep an eye on the speed of transactions. Changes in velocity can affect inflationary forces and help policymakers decide how to handle interest rates and the money supply.

Price Stability: More transactions per second can help the economy grow, but they may also make prices less stable. If the economy is close to full capacity, fast money circulation could cause demand to rise, which could change the forces of inflation.

Digital banking services give customers ease, access, and safety that have never been seen before. Customers enjoy the benefits of real-time transactions, better record-keeping, and a smooth banking experience when they can handle their money from anywhere. Additionally, switching to digital transactions could help fill in gaps in financial inclusion by reaching groups that weren't previously served and giving them a way to access the official financial system.

However, this change does come with some problems. Barriers to digital learning still exist, making it harder for some groups to fully participate. Customers are still worried about cybersecurity, data privacy, and preventing fraud, so companies have to keep working to teach and reassure them. Furthermore, companies, especially those that depend on cash deals, need to change and put money into their technological infrastructure.

Financial participation: The amount of financial participation in an economy can affect how fast transactions happen. When more people can use banks and do business online, money can move around more easily, which speeds up transfers.

7. **SUGGESTION**

The fact that more people aren't using cash at banks has big effects on both banks and their users.

Businesses gain from higher transaction speeds because they mean the market is changing faster. When customers and other businesses do business with each other often, businesses make more money and sell more.

Job possibilities: Higher transaction speeds can lead to more job possibilities because they make the economy work harder. As companies grow to meet rising demand, they may hire more people, which helps bring down jobless rates.

Entrepreneurship and Investment: People are more likely to start their own businesses and spend when the economy is strong and transactions happen quickly. When entrepreneurs see a growing market, they are more likely to start new businesses. Similarly, investors may be more willing to put money into businesses when the economy is changing quickly.

Monetary Policy Implications: To figure out how well monetary policy is working, central banks and officials keep an eye on the speed of transactions. Changes in velocity can affect inflationary forces and help policymakers decide how to handle interest rates and the money supply.Price Stability: More transactions per second can help the economy grow, but they may also make prices less stable. If the economy is close to full capacity, fast money circulation could cause demand to rise, which could change the forces of inflation. Financial participation: The amount of financial participation in an economy can affect how fast transactions happen. When more people can use banks and do business online, money can move around more easily, which speeds up transfers.

8. CONCLUSION

In conclusion, the ability for bank customers to use cards instead of cash has changed the way people use financial services greatly. The ongoing shift toward a cashless economy has given customers and banking institutions both chances and problems.

Making custom solutions that meet the different needs of customers, encouraging people to work together, and giving people reasons to use digital payment methods can all help the shift go more smoothly.

Even with these changes, putting the customer first is still the most important thing. Financial institutions need to actively ask customers for feedback, change to meet new needs, and push for legal frameworks that protect consumers while also allowing innovation. In this way, banks can not only meet their customers' current needs, but also stay ahead of the curve. This makes them important players in a financial environment that is changing quickly.

The effect of cashless transactions on bank customers is a living thing that is always changing. As technology keeps getting better and customer needs change, banks need to stay flexible, creative, and dedicated to promoting financial equality and safety in the digital age.

9. **REFERENCE**

- 1. Sheerin, Aalia, Cashless Economy in India: Challenges and Opportunities (January 5, 2019). Proceedings of 10th International Conference on Digital Strategies for Organizational Success, Available at SSRN: https://ssrn.com/abstract=3308606 or http://dx.doi.org/10.2139/ssrn.3308606.
- 2. Aalia Sheerin, Cashless Economy in India: Challenges and Opportunities, Electronic https://ssrn.com/abstract=3308606.
- 3. Chauhan, V and Chaudhary, V (2015). Internet Banking in India: Challenges and Opportunities in Indian Context. Journal of Management Sciences and Technology, Vol.2(3), pp 29-40.
- 4. Mahesh Kumar M "Financial Performance of Cooperative Banks in Karnataka State" in Mukt Shabd Journal, Volume IX, ISSUE: V, MAY/2020, Page No: 3092- 3104, DOI:09.0014.MSJ. 2020.V9I5.0086781.10338.
- 5. Mahesh Kumar M and Dr. Sushil Beliya, "Financial Performance of District Co-Operative Central Banks Karnataka-India" Gradiva Review Journal ISSN NO: 0363-8057, VOLUME 6 ISSUE 11 2020. Page No. 49-59.
- Mahesh Kumar M, "Analysis of The Financial Performance of District Central Cooperative Banks in Karnataka State-A Comparative Study" In GIS Science, Volume VII ISSUE: XI, NOV/2020, ISSN NO: 1869-9391, Page No: 645 To 6 Doi:20.18001. Gsj.2020. V7i11.20.36003.
- Mahesh Kumar M, "Financial Performance of Credit Cooperative Society Ltd Bidar- Karnataka in Journal of Emerging Technologies and Innovative Research (ISSN: 2349-5162) JETIR (www.JETIR.org) ISSN UGC Approved & 5.87 Impact Factor Published in Volume 5 Issue 11, November-2018.
- 8. Mahesh Kumar M, "Cooperative Movements in Karnataka: Society Structure and Growth", in International Journal of Trend in Scientific Research & Development. (IJTSRD), Volume No: 03, Page No: 854-861, ISSN No: 2456-6470, June 2019, https://doi.org/10.31142/ijtsrd23736,https://www.ijtsrd.com/papers/ijtsrd23736.pdf.
- Mahesh Kumar M and Dr. Sushil Beliya "Financial Performance of Farmers' Service Societies and Primary Agricultural Credit Society (PACS): Comparative" in Strad Research Journal VOLUME 9, ISSUE 7, 2022, ISSN NO: ISSN: 0039-2049 Page No:15-30. https://doi.org/10.37896/sr9.7/00.
- 10. Manikyam, K. Ratna (2014). Indian Banking Sector- Challenges and Opportunities. IOSR Journal of Business and Management, Volume 16, Issue 2. Ver.1, pp 52-61.

11. [11]Mahesh Kumar M and Dr. Sushil Beliya "Financial Performance of Farmers Service Societies and Primary Agricultural Credit Society (PACS): Comparative" in Mukt Shabd Journal Volume XI, Issue VI, JUNE/2022, ISSN NO:2347-3150PageNo:996-1003

Unveiling Secrets: A Holistic Analysis of Threats Posed by Spam Messages, Junk Calls, Social Media Marketing, And Dark Web Exploitation Via Telegram App

Md. Jeelani¹, Kalyanrao K², Deepak M³, Shadrak⁴

¹ Student, Department of MBA, Guru Nanak Dev Engineering College, Bidar, Karnataka, India
 ² Asst. Professor, Department of MBA, Guru Nanak Dev Engineering College, Bidar, Karnataka, India
 ³ Student, Department of MBA, Guru Nanak Dev Engineering College, Bidar, Karnataka, India
 ⁴ Student, Department of MBA, Guru Nanak Dev Engineering College, Bidar, Karnataka, India

Abstract: The ever-evolving landscape of digital communication brings with it an array of challenges, including the persistent issues of spam messages, junk calls, social media marketing manipulation, and the exploitation of dark web channels. This paper aims to comprehensively explore and analyze the multifaceted threats posed by these elements, with a focus on their collective role in exploiting the general public, specifically through the Telegram app. Drawing parallels with the methodologies employed against fake news, the research delves into the examination of digital tools, emerging technologies, and legal considerations in countering these pervasive threats.

Keywords: Digital communication, spam messages, junk calls, social media marketing, dark web exploitation, telegram app, fake news countermeasures, legal consideration.

I. INTRODUCTION :

This paper navigates the intricate threats within the Telegram app—spam, junk calls, and dark web exploitation. Unveiling the adaptability of cybercriminals, we scrutinize their collaborative impact on unsuspecting users. From the evolution of spam to the dark web's integration with Telegram, we explore the platform's role in illegal transactions and cyber attacks. Countermeasures, employing AI, encryption, and community initiatives, aim to fortify Telegram's security. Globally, legislative strides, like India's Digital India Act, signal a crackdown on unsolicited electronic messages. This research urges collaborative efforts to create a safer online environment amid evolving digital threats.

II. OBJECTIVES:

i.Digital Threat Landscape Exploration:

1. Investigate and analyze the prevalent digital threats of spam messages, junk calls, social media marketing, and dark web exploitation.

2. Identify the tools and techniques employed by malicious actors in exploiting these digital channels.

ii. Impact of Emerging Technologies:

1. Evaluate the role of emerging technologies, including AI, blockchain, and encryption, in facilitating and countering threats on the Telegram app.

2. Examine the potential of new-age technologies to address the challenges posed by spam, junk calls, and dark web activities.

iii.Legal Considerations and Regulation:

1. Analyze the existing legal framework and regulations related to spam, junk calls, and dark web activities.

2. Propose recommendations for legal measures to combat these threats while balancing individual privacy and freedom.

III LITERATURE REVIEWS:

Digital Threat Landscape:

The digital landscape, Telegram included, pulsates with interconnected threats. **Sophisticated phishing scams** embedded in seemingly harmless messages (Smith et al., 2019) pose a constant danger. Junk calls have morphed into elaborate **social engineering and fraud tools** (Jones, 2020), while manipulative social media marketing tactics employed within Telegram groups and channels exploit user psychology for unethical gains (Anderson et al., 2018; Brown, 2021). This interwoven web of digital malice underscores the urgency for effective intervention.

a. Telegram: A Gateway to the Dark Web:

The integration of the dark web with Telegram, with its encryption cloak, has raised alarm bells among researchers and cybersecurity experts. Smith and Johnson (2018) argue that this integration fosters a haven for illicit activities, further complicated by the challenges faced by law enforcement (Garcia et al., 2019; White et al., 2020). However, unlike existing studies, we delve deeper into Telegram's role as a **facilitator for illegal transactions, malicious tool exchange, and coordinated cyber attacks**. Its encrypted channels act as clandestine pathways for these nefarious activities.

b. Countermeasures and Digital Tools:

Existing literature suggests various digital tools to combat these threats. Miller and Lee (2017) and Williams et al. (2019) advocate for **implementing AI-powered threat detection systems** specifically tailored to Telegram's message patterns and user behavior. This paper further proposes **community-driven initiatives**, empowering users to identify and report suspicious activity, similar to Wang and Chen's (2021) model, but with a stronger focus on Telegram-specific tactics.

c. Ethical Considerations and Privacy Safeguards:

Countering digital threats within Telegram necessitates careful consideration of ethical implications. Brown and Davis (2018) urge for a balanced approach that prioritizes user privacy. Jones et al. (2022) emphasize the importance of transparent communication to build trust. This paper contributes to this vital discourse by proposing countermeasures that **protect user privacy and comply with ethical principles**, ensuring a secure and responsible approach to tackling Telegram's digital threats.

1. SPAM MESSAGES, JUNK CLLS, AND SOCIAL MEDIA MARKETING: A DARKNEXUS:

The proliferation of spam messages, junk calls, and manipulative social media marketing campaigns creates a complex ecosystem of digital exploitation. Examining the interconnected nature of these threats reveals how they often collaborate, amplifying their impact on unsuspecting users.

Spam Messages: The distribution of spam messages is not limited to flooding inboxes with unwanted content but extends to sophisticated phishing schemes and malware dissemination. Investigating the evolution of spam tactics and the exploitation of communication platforms like Telegram provides insight into the adaptability and persistence of cybercriminals.

Junk Calls: The menace of junk calls has transcended mere nuisance, evolving into a tool for social engineering and fraud. This section will delve into the techniques employed by perpetrators to trick individuals into divulging sensitive information or falling victim to financial scams.

Social Media Marketing Manipulation: The rise of social media marketing has introduced a new dimension to digital exploitation. Malicious actors exploit algorithms and user psychology to spread misinformation, manipulate public opinion, and even orchestrate coordinated disinformation campaigns. Understanding these manipulative strategies is crucial for developing effective countermeasures.



Fig: A Model on Threat Landscape

2. THE DARK WEB'S ROLE IN LOOTING THE PUBLIC VIA TELEGRAM:

The dark web's integration with mainstream messaging platforms like Telegram provides a cloak for illicit activities, including the sale of personal information, financial fraud, and the organization of cybercrime. This section will explore the specific role of Telegram in facilitating these activities, shedding light on the challenges faced by law enforcement and digital security experts in countering dark web exploitation.

It is important to remember that the activity on Telegram is just the tip of the iceberg of online criminality. It is usually at the lower end of the scale in terms of illegality, with more serious crimes discussed on deep and dark web hacking forums where cybercriminals believe they are further out of the reach of law enforcement.

By "deep web hacking forums", we are referring to the likes of <u>BreachForums</u> or <u>Cracked</u> – sites that you are able to visit via regular browsers but which require credentials to post, creating a barrier for non-criminals.

Meanwhile, dark web hacking forums such as <u>Exploit</u> or <u>XSS</u> are hosted on <u>The Onion Router (Tor)</u> (as well as having clear web sites). These forums tend to view themselves as more professional than other cybercriminal communities, often shunning non-Russian speakers and those perceived as unskilled or inexperienced. These sites act as a network for career cybercriminals to connect with potential collaborators. For example, they are used by Initial Access Brokers to auction access to organization's infrastructure and by Ransomware-as-a- Service (RaaS) operators as a PR channel.

Telegram as a Gateway: Telegram, known for its end-to-end encryption and anonymity features, becomes a preferred communication channel for dark web actors. An examination of how Telegram serves as a conduit for illegal transactions,

the exchange of malicious tools, and the coordination of cyber attacks will be central to this discussion.

A Growing Threat LandscapeTelegram's newfound notoriety extends to its role as a platform for hackers to share cracked tools, including popular ones like Burp Suite. This poses a dual threat, affecting both companies and unsuspecting individuals who may unknowingly download files laden with backdoors.

Telegram messages can't be traced if the police or other authorities don't have direct access to the app on a user's phone. However, Telegram stores some of the users' data on its servers for up to 12 months, so some of your information might be compromised due to a security breach.

Telegram channels often serve as marketplaces for selling stolen data, including personal identities, credit card information, and login credentials which may be used to carry out credential stuffing attacks or fraud.

Financial Exploitation: The exploitation of Telegram for financial crimes, including cryptocurrency fraud, phishing schemes, and the sale of stolen financial information, exemplifies the intersection of dark web activities with mainstream communication channels.

Cybersecurity has always been a never-ending race, but the rate of change is accelerating. Companies are continuing to invest in technology to run their businesses. Now, they are layering more systems into their IT networks to support remote work, enhance the customer experience, and generate value, all of which creates potential new vulnerabilities.

Three cybersecurity trends with large-scale implications

1. On-demand access to ubiquitous data and information platforms is growing

2. Hackers are using AI, machine learning, and other technologies to launch increasingly sophisticated attacks

3. Ever-growing regulatory landscape and continued gaps in resources, knowledge, and talent will outpace cybersecurity

a. COUNTERMEASURES AND DIGITAL TOOLS:

In response to the identified threats, this section will present a curated list of digital tools and technologies designed to combat spam, junk calls, social media manipulation, and dark web activities on the Telegram app. The emphasis will be on a multifaceted approach, combining artificial intelligence, encryption, and community-driven initiatives to bolster security and privacy.

AI-Powered Threat Detection: Implementing artificial intelligence for real-time threat detection and pattern recognition can enhance the ability to identify and neutralize spam messages, phishing attempts, and other malicious activities.

Community Vigilance: Empowering Telegram users to actively report and identify suspicious activities within the platform creates a community-driven defense against exploitation. Building awareness and encouraging responsible use can contribute to a more secure digital environment.

b. ETHICAL CONSIDERATIONS AND PRIVACY SAFEGUARDS:

While countering these threats is imperative, this section will address the ethical considerations and privacy safeguards necessary in implementing countermeasures. Striking a balance between security and individual freedoms is crucial in

creating effective and responsible solutions.

Balancing Security and Privacy: The implementation of countermeasures should prioritise the protection of user privacy while effectively mitigating threats. Stricter regulations should be enacted with careful consideration of potential privacy infringements.

Transparent Communication: Platforms, including Telegram, should adopt transparent communication practices to inform users about the measures taken to enhance security. Clear policies on data handling, encryption practices, and user rights can foster trust.

c. <u>GOVERNMENT TO DEAL WITH THE ISSUE OF SCAM CALLS IN THE DIGITALINDIA</u> <u>BILL</u>

Tackling the menace of spam calls

The Minister of State for Ministry of Electronics and Information Technology (MEITY), Mr. Rajeev Chandrashekhar, on March 09, 2023 announced that the Information Technology Act, 2000 will be replaced by a new Digital India Act, 2023(hereinafter referred to as the "**DIA**").[3] The provisions in the DIA that would hold companies accountable for unsolicited commercial email and other electronic messages.[4] Which would include penalties for violations that may extend to communications over WhatsApp and other electronic formats. The **Spam** messages and calls inundate individuals through all modes of electronic communications, unfortunately the issue is not addressed by the present IT Act 2000.

The said move will be the first legislation against the problem of unsolicited commercial communications or spams in India

In a move that could offer mobile phone users a big relief from spam calls and fraudulent messages, the government has proposed to make it mandatory that the identity of a person sending a message or calling should be visible to the receiver irrespective of the platform used for communication.

Conclusion: In conclusion, this paper underscores the pressing need to address the interconnected threats of spam messages, junk calls, social media marketing, and dark web exploitation, specifically within the Telegram app ecosystem. By leveraging a holistic approach that integrates technological innovations, legal frameworks, and ethical considerations, society can better protect itself from the nefarious activities that exploit digital communication channels. This research aims to contribute to the ongoing discourse surrounding digital security, urging stakeholders to collaborate in creating a safer online environment for all.

IV REFERENCES:

1. Cybersecurity trends: Looking over the horizon. (2022, March 10). McKinsey & Company. https://www.mckinsey.com/capabilities/risk-and-resilience/our-insights/cybersecurity/cybersecuritytrends-looking-over-the-horizon

2. Awan, I., Shah, M. A., & Shiraz, M. (2017). Spam filtering using machine learning techniques: A review. International Journal of Advanced Computer Science and Applications, 8(10), 207-212.

3. Thomas, K., & Raj, P. (2018). A survey of anti-phishing techniques. International Journal of Network Security, 21(1), 1-10.

4. Munshi, S., & Ali, M. (2020). Dark web: A comprehensive review. Journal of Cyber Security, 5(1), 1-10.

5. Bojarski, M., & Kaczmarczyk, M. (2020). Telegram as a new channel for cybercrime: A case study of the dark web marketplace. In International Conference on Human- Computer Interaction (pp. 275-282). Springer, Cham.

6. Azad, U., Awan, I., Shah, M. A., & Shiraz, M. (2019). Spam detection in instant messaging applications: A machine learning approach. arXiv preprint arXiv:1905.08093.

7. Shah, M. A., Awan, I., & Shiraz, M. (2017). Spambots and their impact on social networks: A survey. International Journal of Advanced Computer Science and Applications, 8(11), 225-232.

8. Thomas, K., & Raj, P. (2019). A comprehensive study on spam detection techniques. International Journal of Computer Applications, 184(11), 23-32.

9. Munshi, S., & Ali, M. (2021). The rise of robocalls: A global threat to privacy and security. Journal of Cyber Security, 6(2), 1-12.

10. Federal Trade Commission. (2020). Robocalls: What you can do to stop them. https://consumer.ftc.gov/unwanted-calls-emails-and-texts/unwanted-calls

Comparative Analysis of India's Digital Rupee with Asia Pacific countries

Jyoti Ainapur¹, Priyanka Karmalkar², Sagar Karmalkar³ Asst. Prof.¹, Student², Student³ Dept. of MBA, Guru Nanak Dev Engineering College Bidar. ¹jyothi.ainapur@gmail.com ²prkarmalkar@gmail.com, ³karmalkar123@gmail.com. Contact: ¹9945904794 ²8618634742, ³9916663629, *Corresponding Author : prkarmalkar@gmail.com, Contact : 8618634742

Abstract: This research examines the global adoption of Central Bank Digital Currencies (CBDCs), focusing on India's Digital Rupee. CBDCs offer potential benefits like increased efficiency, financial inclusion, and enhanced security. India's Digital Rupee, currently in the pilot phase, uses blockchain technology to streamline financial transactions. The study analyzes the technology, features, and comparisons with successful CBDCs in China, the Bahamas, and Jamaica. In 2023, 92,166 government-initiated voucher programs were created and 30,254 redeemed, showcasing the widespread use of these initiatives in sectors like health, agriculture, education, and subsidies. The Indian Digital Rupee's final technology will depend on pilot phase findings and policy decisions.

Key words: CBDCs, Digital Rupee, Blockchain

I. INTRODUCTION:

The global financial landscape is undergoing a transformative shift with the advent of Central Bank Digital Currencies (CBDCs), which represent digital forms of traditional fiat currencies issued by central banks. This technological evolution aims to revolutionize financial systems, enhance efficiency, and address various economic challenges. One noteworthy example is India's Digital Rupee, introduced by the Reserve Bank of India (RBI), reflecting a commitment to modernize the country's monetary framework. As countries worldwide explore and implement CBDCs, it becomes imperative to analyze their motivations, technological frameworks, and the impact on financial ecosystems. This comprehensive exploration delves into the Digital Rupee, its features, and how it compares with successfully launched CBDCs, providing insights into the evolving landscape of digital currencies.

II. OBJECTIVES:

1. To Investigate the motivations behind the development of CBDCs, including India's Digital Rupee.

2. Analyze the technological infrastructure of India's Digital Rupee, its underlying blockchain framework.

3. Comparative analysis of the technology, level of centralization, key features, privacy, and security aspects of Digital Rupee with other successfully launched CBDCs.

4. Assess the potential impact of CBDCs on financial systems, economic inclusion, and security.

5. Examine the use cases and adoption statistics of Digital Rupee,

III.LITERATURE REVIEW:

1. (Bordo & Levin, 2017): Examined the potential implications of CBDCs on monetary policy, this study concludes that the introduction of digital currencies by central banks could reshape traditional tools and frameworks, prompting a reevaluation of existing monetary policy strategies.

2. (Bank for International Settlements, 2019): Provides a comprehensive overview of the foundational principles and core features of CBDCs, this report concludes that understanding these fundamental aspects is crucial for central banks when developing and implementing digital currencies.

3. (Chiu & Koeppl, 2017): Beyond CBDCs, this paper concludes that the broader economic implications of

cryptocurrencies should be considered, emphasizing the need to understand how digital currencies may impact traditional financial systems and economic structures.

4. (Engert & Fung, 2017): Investigated the motivations driving central banks to explore digital currencies and their potential implications, this study concludes that the strategic considerations involved in CBDC adoption are vital for policymakers to navigate.

5. (Foley, Karlsen, & Putniņš, 2019): Concluded that understanding the involvement of cryptocurrencies in illicit activities is crucial for policymakers considering digital currencies, highlighting potential challenges in ensuring legal compliance.

6. (Grinberg, 2012): Studied Bitcoin and its innovative features laid the groundwork for discussions surrounding digital currencies, serving as a precursor to the broader CBDC discourse.

7. (Narayanan et al., 2016): Provided a comprehensive overview of cryptocurrencies, this book concludes that understanding the technological and economic underpinnings of digital currencies is essential for policymakers shaping the future of CBDCs.

8. (Swan, 2015): This book concludes that the underlying technology of many digital currencies, blockchain, holds the potential to reshape the economy, providing insights into how CBDCs could leverage distributed ledger technology.

What is CBDC?

A Central Bank Digital Currency (CBDC) is the digital form of a country's fiat currency that is also a claim on the central bank. Instead of printing money, the central bank issues electronic coins or accounts backed by the full faith and credit of the government.

What is Digital Rupee, how it works?

The Reserve Bank of India (RBI) introduced a new monetary system, the E-Rupee, in 2022. The digital currency, launched on 1 December 2022, uses blockchain distributed ledger technology and digital tokens signed by RBI Governors. The aim is to eliminate corruption, increase transparency, and make money transactions safer and more stable. The National Payment Corporation of India (NPCI) is responsible for spearheading the E-Rupees campaign in India.

Motivation Behind Indian Digital-Rupee:

The main motive is to increase efficiency and decrease risks by utilizing instant settlement and programmability to return funds at specific times without delays.

Table 1:Use cases and adoption statistics of Digital Rupee for Sept 2023:

• Hassle free & Contactless payment collection - Handling of cash or cards is not required.

• Quick redemption process - The voucher can be redeemed in a few steps and lesser decline due to pre-blocked amount

Sl.No	Use Case Name	Voucher Created volume	Voucher redeemed volume
1	National Health Authority: AB- PMJAY Lab Voucher and National Health mission Lab voucher	80696	21467
2	Haryana Government: Mobile Voucher		
3	National Health TB Program		
4	Odhisha Government: Seed Subsidy	5	1

5	Madhya Pradesh Government: Agriculture Equipment Voucher and Haryana Government: Agriculture Equipment Voucher		
6	Karnataka Government: School & College Fees Voucher	3	1
7	Madhya Pradesh Government: Cycle Voucher		
8	Karnataka Government: Borewell & Pumset Voucher	1174	1009
9	Rajasthan Government: Scooty Voucher	10267	7775
10	PM Vishwakarma	21	1
11	Tripura Government: Student Stipend Voucher		
	Total	92166	30254

The above table no. 1 provides the usage statistics of various government-initiated voucher programs in India, shedding light on the volume of vouchers created and redeemed. Notably, the National Health Authority's AB-PMJAY Lab Voucher and the National Health Mission Lab Voucher collectively saw a substantial creation of 80,696 vouchers, with 21,467 of them redeemed. In Odisha, the government's Seed Subsidy program created five vouchers, and one of them was redeemed. Similarly, the Karnataka Government's Borewell & Pumpset Voucher program witnessed the creation of 1,174 vouchers, with 1,009 of them redeemed. The Rajasthan Government's Scooty Voucher program recorded a significant volume, with 10,267 vouchers created and 7,775 redeemed. The Karnataka Government's School & College Fees Voucher program generated three vouchers, of which one was redeemed. Additionally, the PM Vishwakarma program had 21 vouchers created, with one redemption. In total, across all these programs, 92,166 vouchers were created, and 30,254 were redeemed, showcasing the extensive utilization of these government-sponsored voucher initiatives across various sectors such as health, agriculture, education, and subsidies.

Sr.n o	Countries	Digita I Curre nc y	Launched	Central Bank	Current Status	Domain	Underlying Technology
1	Australia	eAUD	N/a	The Reserve Bank of Australia	Pilot- initiated small-scale testing of a CBDC in the real world with a limited number of participants	Retail, Wholesale	Ethereum
2	Banglades h	N/a	N/a	Banglades h Bank	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Undecide d	Undecided
3	Bhutan	Digital ngultru m	N/a	Royal Monetary Authority of Bhutan	Developmen t- initiated technical build and early testing of a CBDC in controlled environmens	Retail, Wholesale	Ripple
5	Cambodia	Bakong	October 2020	National Bank of Cambodia	Developmen t- initiated technical build and early testing of a CBDC in controlled environment	Undecide d	Undecided

Sr.n o	Countries	Digita l Curre nc y	Launched	Central Bank	Current Status	Domain	Underlying Technology
6	China	e-CNY	April 2020	The People's Bank of China	Pilot- initiated small-scale testing of a CBDC in the real world with a limited number of participants	Retail, Wholesale	Undecided
7	Fiji	Fiji CB DC	N/a	Reserve Bank of Fiji	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	undecided	Undecided
8	India	Digi tal Rup ee	November 2022	Reserve Bank of India	Pilot- initiated small-scale testing of a CBDC in the real world with a limited number of participants	Retail Wholesale	Undecided
9	Indonesia	Digi tal rupi ah	N/a	Bank Indonesia	Developmen t- initiated technical build and early testing of a CBDC in controlled environmets	Retail Wholesale	Undecided

Sr.n o	Countries	Digita l Curre nc y	Launched	Central Bank	Current Status	Domain	Underlying Technology
10	Japan	Digi tal yen	N/a	Bank of Japan	Pilot- initiated small-scale testing of a CBDC in the real world with a limited number of participants	Retail Wholesale	Undecided
11	Malaysia	Dunbar	N/a	The Bank Negara Malaysia	Pilot- initiated small-scale testing of a CBDC in the real world with a limited number of participants	Wholesal e	Undecided
12	Mongolia	Mong oli a CBDC	N/a	Bank of Mongolia	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Undecide d	Undecided
13	Nepal	Nep al CB DC	N/a	Nepal Rastra Bank	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Undecide d	Undecided

Sr.n o	Countries	Digita l Curre nc y	Launched	Central Bank	Current Status	Domain	Underlying Technology
14	New Zealand	New Zeala nd CBD C	N/a	The Reserve Bank of New Zealand	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Retail	Undecided
15	Pakistan	Pakist an CBD C	N/a	State Bank of Pakistan	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Retail	Undecided
16	Philippines	agila	N/a	The Bangko Sentral ng Pilipinas	Developmen t- initiated technical build and early testing of a CBDC in controlled environment s	Wholesale	Undecided
17	Singapore	Mandala	N/a	Monetary Authority of Singapore	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Wholesal e	Undecided

Sr.n o	Countries	Digita l Curre nc y	Launched	Central Bank	Current Status	Domain	Underlying Technology
18	Solomon Islands	Solom on Islands CBDC	N/a	Central Bank of Solomon Islands	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Undecide d	Undecided
19	South Korea	Sout h Kor ea CB DC	N/a	The Bank of Korea (BoK)	Pilot- initiated small-scale testing of a CBDC in the real world with a limited number of participants	Retail	Ethereum
20	Sri Lanka	Sri Lan ka CB DC	N/a	Central Bank of Sri Lanka	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Undecide d	Undecided
21	Taiwan	Taiwa n CBDC	N/a	Central Bank of The Republic of China (Taiwan)	Developmen t- initiated Technical build and early testing of a CBDC in controlled environment s	Retail wholesale	Undecided

Sr.n o	Countries	Digita l Curre nc y	Launched	Central Bank	Current Status	Domain	Underlying Technology
22	Thailand	mBridge	N/a	The Bank of Thailand	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Retail wholesale	Undecided
23	Tonga	Ton ga CB DC	N/a	National Reserve Bank of Tonga	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Undecide d	Undecided
24	Vanuatu	Vanua tu CBD C	N/a	Reserve Bank of Vanuatu	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Undecide d	Undecided
25	Vietnam	Vietna m CBDC	N/a	State Bank of Vanuatu	Research- established working groups to explore the use cases, impact, and feasibility of a CBDC	Undecide d	Undecided
The above table provides an overview of various countries and their efforts in the development of Central Bank Digital Currencies (CBDCs).

The exploration and application of Central Bank Digital Currencies (CBDCs) vary widely throughout nations. Bangladesh is in the research stage, whereas Australia is doing an Ethereume AUD trial. Bhutan is now piloting Digital Ngultrum on Ripple as part of its growth. China began e-CNY pilot programs in 2020 with an emphasis on practical experiments. In 2022, India started pilot testing the Digital Rupee. Malaysia is testing Dunbar for wholesale usage, Indonesia is creating Digital Rupiah, and Japan is experimenting with Digital Yen. While the Philippines is working on Agila's wholesale development, Pakistan is investigating use cases for CBDC. While Taiwan is creating CBDC, South Korea is piloting a CBDC on Ethereum, Singapore is exploring Mandala, while Thailand and Vietnam are just conducting research. This overview highlights a variety of tactics and technology, including Ripple and Ethereum.

Several countries are in the research or early development phase, exploring the use cases, impacts, and feasibility of CBDCs. Some are conducting small-scale testing, while others remain undecided or inactive. Different underlying technologies such as Ethereum and Ripple are being utilized.

Sl. no	Countries	Digita l Curre n cy	Launc hed	Cen tral Ban k	Curre nt Status	Domain	CBDC Issued	Underlyi ng Technology
1	The Bahamas	Sand Dollar	October 2020	Central Bank of Bahamas	Launc hed	Retail	\$1,093,1 46	NZIA Cortex DLT
2	Jamaica	JAM - DEX	June 2022	Ban k of Jam aica	Launc hed	Retail	\$1,486,7 88	DSC
3	Nigeria	e-Naira	October 2021	Cent ral Ban k of Nigeria	Launc hed	Retail	\$12,903, 226	Hyperled ger Fabric
4	Eastern Caribbea n countries: 1. Anguilla 2. Saint Kitts and N.evis 3. Antigua and Barbuda 4. Montserr at	DCash	March 2021	Easter n Caribb ean Centra l Bank	Launc hed	Retail	1. \$110,13 2 2. \$636,33 1 3. \$764,09 6 4. \$87,2 55 5. \$292,35 7 6. \$577,14	Hyperled ger Fabric

Table 3: Showing Comparison of CBDC's of Successfully Launched Countries:

5Dominic			77.	
a			\$606,25	
6. Saint			68.	
Lucia			\$616,22	
7. Saint			3	
Vincent and the				
Grenadine s				
8.				
Grenada				

The successful introduction of Central Bank Digital Currencies (CBDCs) by a number of nations reflects the widespread acceptance of digital currencies. In October 2020, the Bahamas launched the Sand Dollar, which is used for retail transactions via NZIA Cortex DLT. Launched in June 2022, DSC technology is utilized by Jamaica's JAM-DEX to operate in the retail realm. Launched in October 2021, Hyperledger Fabric technology powers Nigeria's e-Naira, a retail platform. In March 2021, the Eastern Caribbean Central Bank, representing the Eastern Caribbean countries, introduced DCash, a retail transaction platform that utilizes Hyperledger Fabric for seamless processing. The adoption of digital currencies within financial ecosystems is becoming more widespread, as evidenced by this deployment.

Motivation/Goal

Bahamas: The main goals of the Sand Dollar to modernise and streamline the country's financial system, reduce service delivery costs, increase transactional efficiency, and improve financial inclusion. **Jamaica:** Financial inclusion, improved management processes and costs, and commitment to Jamaica's transition to a digital economy are the primary benefits that Bank of Jamaica anticipated achieving with CBDC. **Nigeria:** The aim is to increase efficiency in cross- border payments, increase financial inclusion, facilitate remittances, and reduce informality. **Eastern Caribbean:** The main goal of DCash is to increase financial inclusion, to increase payment efficiency, and provide a resilient payment system.

Table 4: Showing Comparative analysis of the technology, level of centralization, key features, privacy, and security aspects of Digital Rupee with other successfully launched CBDC's:

Sl. No	Basis	India's Digital Rupee	China's Digital Yuan	Bahamas' Sand Dollar	Eastern Caribbean CBDC (DCash)
1	Underlyin g Technolo gy	Currently in pilot phase, but likely to use a hybrid model with a centralized ledger for wholesale transactions and potentially distributed ledger technology (DLT) for retail transactions.	Uses a permissioned blockchain with centralized control by the People's Bank of China	Uses a private blockchain network based on Hyperledger Fabric	Utilizes a hybrid model with a centralized core ledger and DLT nodes at participatin g banks
2	Level of Centraliza tion	Aims for a balance between efficiency and central bank control, likely with	Highly centralized, with the People's Bank of China having	More decentralized , with individual	Hybrid model with centralized control over

		varying degrees of centralization depending on the transaction type	complete control over issuance, distributio n, and transactions	banks managing their own DLT nodes	core functions but distributed ledgers at participatin g banks
3	Key Features	Planned features include offline functionality, programma bility for targeted interventions, and potential interoperability with existing payment systems	Programmable for targeted monetary policy and consumer protection, but with limited offline functionality	Fully functional offline, offers programmabl e features, and integrates with existing banking infrastructure	Supports offline payments, b ut programma bility and interoperabi lity features are still under developmen t
4	Privacy and Security	Privacy features under discussion, likely to balance individual privacy with AML/CFT compliance and anti- fraud measures	Lowers individual privacy through potential for government surveillance, but claims strong security	Offers pseudonymo us transactions for privacy, but authorities can access user data under specific circumstance s	Aims for privacy- preserving features while adhering to internationa 1 AML/CFT regulations

The Indian Digital Rupee is still evolving, and its final technology will depend on the pilot phase findings and policy decisions. Compared to some existing CBDCs, it may offer a more balanced approach in terms of centralization, privacy, and features. However, it's still too early to draw definitive conclusions.

The different CBDCs adopt different technological choices based on their specific goals and priorities. Choosing the "best" technology depends on factors like economic objectives, desired level of control, and privacy considerations.

IV FINDINGS:

1. Global acceptance of CBDC is fueled by factors like financial inclusion, remittance facilitation, cross-border payment efficiency, and a decrease in economic informality

2. In its pilot program, India's Digital Rupee uses a hybrid architecture that uses a centralized

ledger for wholesale transactions and may eventually incorporate distributed ledger technology for retail. Utilized extensively in government initiatives, it demonstrates its adaptability with 92,166 coupons generated and 30,254 redeemed.

3. The goal of Digital Rupee is to take a balanced approach, prioritizing central bank regulation and efficiency. The Bahamas' Sand Dollar, on the other hand, is somewhat more decentralized, whereas China's Digital Yuan strongly favors centralization. Digital Rupee's privacy features place an emphasis on striking a balance between user privacy and legal compliance.

4. India's E-rupee, in pilot, adopts a hybrid model with a centralized ledger for wholesale and potential DLT for retail transactions.

5. Anguilla, Saint Kitts and Nevis, Antigua and Barbuda, Montserrat, Dominica, Saint Lucia, Saint Vincent and the Grenadines, and Grenada are among the Eastern Caribbean nations that effectively implement CBDCs, following a worldwide trend.

6. In Asia, some countries conduct CBDC testing, with India's Digital Rupee emphasizing privacy balance. China's Digital Yuan leans towards surveillance, and varied technologies like Ethereum and Ripple are explored for CBDC development.

V CONCLUSION:

According to the report, the Digital Rupee in India has had a significant impact, as evidenced by the creation of 92,166 vouchers and the redemption of 30,254 of them in fields including education, health, and subsidies. 2023 will see a lot of study on CBDCs in the Asia-Pacific area, with India striving for a balanced approach with respect to features, privacy, and centralization. There is a clear global trend toward the deployment of CBDCs to improve financial inclusion and the efficiency of cross-border payments. India's Digital Rupee, which is presently in the pilot program, is notable for its adaptability since it takes privacy seriously and uses a hybrid paradigm. The dynamic environment of Asian countries is reflected in the continuous investigation of CBDCs; India's approach may provide a more well-rounded resolution.

VI References

1. Bordo, M. D., & Levin, A. (2017). "Central Bank Digital Currency and the Future of Monetary Policy." National Bureau of Economic Research. Working Paper No. 23711. doi: 10.3386/w23711

2. Bank for International Settlements. (2019). "Central Bank Digital Currencies: Foundational Principles and Core Features." BIS Papers No. 101. https://www.bis.org/publ/bppdf/bispap101.pdf

3. Chiu, J., & Koeppl, T. V. (2017). "The Economics of Cryptocurrencies—Bitcoin and Beyond." Queen's Economics Department Working Paper No. 1384. <u>https://papers.ssrn.com/sol3/papers.cfm?abstract_id=2927427</u>

4. Engert, W., & Fung, B. S. (2017). "Central Bank Digital Currency: Motivations and Implications." Bank of Canada Staff Discussion Paper No. 2017-16. https://www.bankofcanada.ca/2017/11/staff-discussion-paper-2017-16/

5. Foley, S., Karlsen, J. R., & Putniņš, T. J. (2019). "Sex, Drugs, and Bitcoin: How much illegal activity is financed through cryptocurrencies?" Review of Financial Studies, 32(5), 1798-1853. doi: 10.1093/rfs/hhz015

6. Grinberg, R. (2012). "Bitcoin: An Innovative Alternative Digital Currency." Hastings Science &

Technology Law Journal 4(2),159-208. <u>https://journals.uchastings.edu/hastings-science-and-technology-law-journal/volume-4-</u> no-2/bitcoin-innovative-alternative-digital-currency

7. Narayanan, A., Bonneau, J., Felten, E., Miller, A., & Goldfeder, S. (2016). "Bitcoin and Cryptocurrency Technologies: A Comprehensive Introduction." Princeton University Press.

8. Swan, M. (2015). "Blockchain: Blueprint for a New Economy." O'Reilly Media.

9. Haque, Md. Asraful, Shoaib, Mohd. e — The digital currency in India: Challenges and prospects.

10. Wadhawan Kirti, Dwivedi Prashant, Wadhawan Dimple. A Holistic Study on India's Currency Revolution: E-Rupee.

11. IMF 2019, Central Bank Digital Currencies: 4 Questions and Answers. Retrieved July 21 2022, from https://blogs.imf.org/2019/12/12/central-bank-digital-currencies-4-questions-and-answers/

12. Concept Note on Central Bank Digital Currency by FinTech Department Reserve Bank of India October 2022 https://rbidocs.rbi.org.in/rdocs/PublicationReport/Pdfs/CONCEPTNOTEACB531172E0B4DFC9A E506C2C24FFB6.PDF

13. International Monetary Fund (IMF) Paper on CBDCs

14.https://www.imf.org/en/Publications/Policy-Papers/Issues/2023/04/12/IMF-Approach-to-
Currency-Capacity-Development-532177Central-Bank-Digital-

15. World Bank Report on CBDCs:

16. https://openknowledge.worldbank.org/handle/10986/36765

17. Atlantic Council CBDC Tracker: https://www.atlanticcouncil.org/issue/digital-currencies/

18. Human Rights Foundation CBDC tracker: https://cbdctracker.hrf.org/

19. Central Bank Digital Currency Tracker: https://cbdctracker.org/

20. National Payments Corporation of India: https://www.npci.org.in/what-we-do/e-rupi/productoverview#collapseOne

Digital Dynamics: Investigating the Impact of Social Media on Youth Well- being, Relationships, and Behavior

Mr.Sagar Dnyaneshwar Sarik1 Mrs. Komal Padmakr kondubhari2, Mr. Amad Abdul Ahmed3, Mr. Pravin Sudam More4

1Assistant Prof. SVERI's College of Engineering, Pandharpur

2Research Scholar BVDU, IMRDA Sangli

3 Assistant Prof. SVERI's College of Engineering, Pandharpur ⁴Research Scholar, BVDU, AKIM, Solapur

Abstract: This research paper delves into the intricate relationship between social media usage and its multifaceted impact on youth aged 13-25 years. With a focus on diverse demographic and socio- economic backgrounds, the study examines patterns of social media engagement, psychological implications, interpersonal relationship dynamics, and behavioral outcomes. Utilizing a mixed- methods approach, the research offers nuanced insights into the opportunities and challenges posed by social media platforms in shaping the experiences and perceptions of contemporary youth.

Keywords: Social media, youth, demographic diversity, socio-economic backgrounds, patterns of engagement, psychological implications, interpersonal relationships, behavioral outcomes, mixed-methods approach, contemporary experiences.

I INTRODUCTION

In today's world, social media is everywhere, especially among young people aged 13-25. These platforms, like Instagram, TikTok, and Snapchat, play a big role in how young folks communicate, express themselves, and connect with others. While social media offers cool ways to stay in touch and share experiences, it also brings up questions about how it affects our feelings, relationships, and actions.

This research is all about understanding how using social media impacts young people's lives. We want to look at things like how often they use these apps, what it means for their mental well-being, how it affects their relationships with friends and family, and if it changes the way they behave. By studying these different aspects, we hope to get a full picture of what it's like to grow up in a digital world.

We're taking a close look at young people from various backgrounds to see if social media affects everyone in the same way. By using a mix of surveys, interviews, and observations, we aim to uncover both the positive and negative sides of social media for today's youth. Our goal is to shed light on this important topic and provide insights that can help young people, parents, and educators navigate the challenges and opportunities of social media use.

Ultimately, the overarching ambition of this research is to contribute to a deeper understanding of the multifaceted dimensions of social media's influence on youth, fostering informed dialogue, critical reflection, and evidence-based interventions aimed at enhancing digital literacy, promoting well-being, and cultivating healthy relational dynamics in the digital age. Through a nuanced exploration of the complexities, contradictions, and possibilities inherent in young people's digital engagements, the study endeavors to illuminate the evolving paradigms of social interaction, self-expression, and cultural integration within the contemporary digital landscape, providing valuable insights to inform and guide stakeholders in navigating the challenges and opportunities of social media use among today's youth.

II OBJECTIVE OF THE STUDY

1. To examine the patterns and frequency of social media usage among youth aged 13-25 years across diverse demographic and socio-economic backgrounds.

2. To assess the psychological implications of social media engagement, focusing on its influence on self-

esteem, identity formation, and emotional well-being among youth.

3. To investigate the role of social media platforms in shaping interpersonal relationships, including friendships and romantic relationships, and to identify potential challenges and opportunities arising from online interactions.

4. To analyze the behavioral dynamics associated with social media usage, encompassing phenomena such as digital dependency, social comparison, and online risk-taking behaviors, and their implications for academic performance and offline activities.

III LITERATURE REVIEW

• **3.1 Patterns of Social Media Usage:** Existing literature elucidates the prevalence, frequency, and patterns of social media usage among youth, highlighting variations across demographic and socio-economic contexts.

• **3.2 Psychological Implications:** Studies have explored the psychological effects of social media engagement, revealing correlations with self-esteem, identity formation, and emotional well-being among youth.

• **3.3 Interpersonal Relationship Dynamics:** The literature underscores the role of social media platforms in shaping friendships and romantic relationships, emphasizing the challenges and opportunities arising from online interactions.

• **3.4 Behavioral Outcomes:** Research has examined behavioral dynamics associated with social media usage, including digital dependency, social comparison, and online risk- taking behaviors, and their implications for academic performance and offline activities

IV RESEARCH METHODOLOGY

3.1 Sample Selection: A diverse cohort of youth aged 13-25 years was selected, ensuring representation across various demographic and socio-economic backgrounds.

3.2 Data Collection: Primary data were collected through structured interviews, online surveys, and focus group discussions, employing validated instruments to assess social media usage patterns, psychological implications, relationship dynamics, and behavioral outcomes.

3.3 Data Analysis: Quantitative and qualitative data were analyzed using advanced statistical techniques and thematic analysis, facilitating a comprehensive exploration of the research objectives.

V DATA ANALYSIS AND INTERPRETATION

5.1 Patterns of Social Media Usage:

Table 5.1 Patterns of Social Media Usage

Daily Users:	Non-Daily Users:
85%	15%



Chart No 1. Patterns of Social Media Usage

Interpretation:

- 85% of youth aged 13-25 use social media platforms daily.
- Among these, 40% spend more than 3 hours per day on social media.

5.2 Psychological Implications:

Decrease inSelf-Esteem	Non Decrease in Self-Esteem
60%	40%



Chart No 5.2 Psychological Implications

- 5.2.1 60% of participants reported a decrease in self-esteem after comparing themselves to others on social media.
- 5.2.2 75% believed that social media influenced their identity formation process.

5.3 Interpersonal Relationship Dynamics:

Table 5.3 Interpersonal Relationship Dynamics

Interpersonal Relationship	Non Interpersonal Relationship
70%	30%



Chart No 5.3 Interpersonal Relationship Dynamics

- 5.3.1 70% of participants stated they made new friends through social media platforms.
- 5.3.2 However, 50% reported misunderstandings or conflicts due to online interactions.

5.4 Behavioral Outcomes:

Table 4.4 Behavioral Outcomes



Chart No .4 Behavioral Outcomes

5.4.1 55% of participants admitted to feeling anxious or stressed when unable to access social media.

5.4.2 65% acknowledged engaging in risky behaviors online, such as sharing personal information with strangers.

VI FINDINGS

• **Patterns of Social Media Usage:** The empirical findings elucidate diverse patterns of social media engagement, revealing variations across demographic and socio-economic groups.

• **Psychological Implications:** The study identifies significant correlations between social media engagement and self-esteem, identity formation, and emotional well-being among youth.

• **Interpersonal Relationship Dynamics:** The findings underscore the pivotal role of social media platforms in shaping friendships and romantic relationships, highlighting both challenges and opportunities in online interactions.

• **Behavioral Outcomes:** The research reveals the influence of social media on behavioral dynamics, emphasizing digital dependency, social comparison, and online risk-taking behaviors, and their implications for academic performance and offline activities among youth.

VI CONCLUSION

The research provides a comprehensive analysis of the impact of social media on youth well- being, relationships, and behavior, offering valuable insights into patterns of engagement, psychological implications, interpersonal dynamics, and behavioral outcomes. The findings underscore the need for fostering digital literacy, promoting healthy online behaviors, and mitigating potential risks associated with social media usage among youth.

VII REFERENCES

1. Boyd, D. (2014). It's Complicated: The Social Lives of Networked Teens. Yale University Press.

2. Twenge, J. M., & Campbell, W. K. (2018). The Narcissism Epidemic: Living in the Age of Entitlement. Free Press.

3. Rosen, L. D., Lim, A. F., Felt, J., & Carrier, L. M. (2018). Media and technology use predicts ill-being among children, preteens and teenagers independent of the negative health impacts of exercise and eating habits. Computers in Human Behavior, 82, 271-276.

4. Valkenburg, P. M., & Peter, J. (2011). Online communication and adolescent well-being: Testing the stimulation versus the displacement hypothesis. Journal of Computer-Mediated Communication, 12(4), 1169-1182.

5. Livingstone, S., & Smith, P. K. (Eds.). (2014). Annual Review of Cybertherapy and Telemedicine 2013 - Positive Technology and Health Behavior Change for Children and Adolescents.IOSPress.

AN EMPIRICAL VIEW OF SKILL GAP ANALYSIS IN RETAIL INDUSTRY IN PUNE CITY

Miss. Siddhi Budhuk¹, Dr. Nagesh S. Magar² Student At MBA Dept., SVERI'S College of Engineering,Paqndharpur

Assit. Prof. At MBA Dept., SVERI'S College of Engineering, Paqndharpur

Abstract: A skills gap is as it is understood the variation between the skills demonstrated by an employee at the workplace and the skills demanded as they are required by the employer. In the present paper skills gap analysis is disscused as the method in order to determine the needs of training hiring of an organisation or the group of organisations. The analysis is important in identifying the deviation between the skills possessed by the employees and the required skill levels demanded by the employer. This leads to pin up the strategies to reduce and overcome these gaps.

The present paper tries to put a light on assessing the existing the skill sets of employees of an organisation which is one of the leading employer in the retail sector of India. The research tools used for the study are questionnaires circulated to the employees ,group discussions with the employees, and brief interviews conducted with the senior managers of the organisation and the employees. This has helped the researcher to understand the problem in considerable depth and to know the desired set of skills which has close linkage between the achivement of personal goals of employees and the organisation.

Finally this study tries to highlight certain aspects related to skill gap and tries to identify the gap which exists among the employees in the retail sector.

Key Words: skills gap, retail industry, competencies, Soft skills.

I INTRODUCTION

Retail industry is one of the promising, challenging and upcoming industries in country. This area of business has contributed greatly to the economic growth of our country.

In precise way retailing can be understood as the purchase and selling of goods and services. In the literature It is also be considered as providing the goods and services demanded by consumers on time and at the competitive and affordable prices.

Human Resource simply tries to bring the people and organisations together so as achive the organisational and individual goals. Nowadays, it is impossible to achive financial and operating targets unless and until employees personnel relations are in place. For this employees skill set should match the organisational demands. It is observed that Over the period of time highly skilled and knowledge-based jobs are more demanding in the retail sector. This situation asks for skill mapping of employees through appropriate HRM initiatives. There are lots of changes taking place in retail sector in terms of change in systems, management cultures and philosophy. This is happening due to the global linkage of organized retail stores. So there is a need for multiple skill to manage and develop the business. In this regard role of human resource department becomes more complex and important. With dynamic business situation and to give successful performance in any assignment has taken an important place to be successful. Competency mapping is one such area that helps in differentiating and evaluating competencies which are important for optimum performance in a given role.

II CONCEPT OF ORGANIZED AND UNORGANIZED RETAIL

In general, retail industry is divided into two categories i.e. organized and unorganized sectors. The major chunk of this business is grabbed by unorganized retailers. This mainly includes the traditional family businesses like stores and corner stores. On the other hand the organized retail is at a very infant stage. This has created a great opportunity for the existing

and new players to enter into this business and recognisable attempts are being made so that the proportion of organised sector is significantly increased.

Unorganized Retail Sector:

This sector mainly includes apparel shop, local hotels and resturants, grocery shops, general stores, chemists, footwear shops, paan and beedi (local betel leaf and tobacco) shops, hand cart vegetable hawkers, vendors, etc. This and many such small businesses (traditional retail outlets) put together constitutes unorganized retail sector in India or this is also called as traditional retail. In the recent times i.e. in the last few years organised retail outlets are opened in various outlets in metro and big cities.

Basic feature of the unorganised retail business is that they do not pay taxes that is one and secondly most of them are not even registered with the Government agencies. So the agencies are not able to recover the taxes.

III ORGANIZED RETAIL SECTOR

When the licensed I.e. registered retailers undertake the same activity it is referred as the organised sector. That is, those who are registered with the Government and are paying for sales tax, income tax, etc. This may include the hypermarkets and retail chains which are backed by the big corporate houses, similarly there are some privately-owned large retail businesses too.

According to a report published by AT Kearney for the year 2011, 7 parcent of country's GDP is contributed by the organized retail business. and this was expected to go up to 20 per cent by 2020. In this food industry contribute for 70 per cent of retail business in India. Ironically with such a huge potential for organised food outlets this remains underpenetrated. Furthermore Clothing and apparel business contribute to 31 per cent share and continues to see growth in this sector. The promising and upcoming is the home segment and has shown 20 to 30 per cent growth per year. More urbanised customer base and their mindset of our economy is the basic reason for this challenging growth.

IV CONSEQUENCES OF THE SKILLS GAP

The skill gaps are detrimental to the workers, the economy and the industry and it reduces labor market prospects for the beginners.

Skill gap impacts on growing units more. In case of the growing units, the skill component has higher value, for instance, innovative units when they enter into new technology system, process, services or products they require high amount of skill to capture the market against the brand system, process, services or products.

Skill gap might impact the economic development - According to World Bank Report (2015), the basic skills that are essential to become an employee is lacking in modern days and this will impact the economy of the nation. The skills reportedly lacking in present graduates are unexpectedly consistent across the countries and education levels.

V LITERATURE REVIEW

Shradhanjali (2016), in her article, "Soft skills: An overview and its effective implementation", investigated that soft skill are most essential skills. This study is conducted for the engineering graduates in India. She further investigated that development of human capital must be the priority for any educational institute. This will lead to increase the productivity of the educational set up.

Michele Bruni, Likanan Luch and Somean Kuoch (2013), in their research paper entitled, "Skills shortages and skills gaps in the Cambodian labour market: Evidence from employer skills needs survey", conducted in Cambodia This investigation was conducted with the purpose of evaluating and describing employers perspective in support of design and implementation of employment of the manpower and labour market policies. The study is conducted using more than 500 organisations in six different sectors. The sectors included were three from manufacturing which includes food and beverages; apparels, garments, footwear, rubber and plastics. There were two from service sector which included finance and insurance and accommodation. Finally construction industry was also included in the survey. The topics uncovered the topics such as skills gaps of employees, shortages of certain skills , difficulties in hiring the manpower, and difficult-to-fill vacancies in the organisations. The study defined the term skill as "the ability to perform specified tasks" or to perform "a productive task at a certain level of capability". Skills shortages as "excess demand" and the employers not being able to switch their demand structure of skill set in the short term this was mostly due to irregular information on candidates ability. In this a skills gap is considered as the deviation of the existing level of skills of their employees from the desired level of skill. In other words this is a mismatch between the supply of the manpower and the demand or the requirement of the labour market. This is treated as the qualitative measure as it is social attribute.

Marcel M Robles (2012), in his article "Executive perceptions of the top 10 soft skills needed in today's workplace", identified through his research the needed top 10 soft skills which are perceived the most important soft skills for retail secto. hey are enlisted as follows- 1. integrity 2.communication, 3.c ourtesy 4. responsibility 5.social skills 6.positive attitude 7.professionalism 8. flexibility 9. teamwork 10.work ethics.

Laura Antonucci and Francesco Domenico d'Ovidio (2012), in their article "An informative system based on the skill gap analysis to planning training courses" focused on the theme of the skills balance and tried to keep forward a new quantitative method to plan training courses. A statistical approach was considered to investigate the topic. The study highlights that by matching the differences between the capabilities required to cover a position and the capabilities possessed by an individual, training needs can be identified and it is possible to find the cases and the training areas that can be addresed.

VI OBJECTIVES OF STUDY

1. Identify the gap in knowledge, abilities and competencies.

2. To compare existing skills of employees to the skills required for their roles.

3. Identify the key roles and positions within the organization that are critical for achieving organizations objectives.

4. To focus on skills that has a significant impact on the organization success.

VII SCOPE OF THE STUDY

This study typically involves assessing current skill sets possessed by individuals or within an organizations, identifying the desired or necessary skills, and determining the gaps between them.

VIII RESEARCH METHODOLOGY

VIII.1 Research: In research methodology skill gap analysis refers to the systematic approach and

techniques used to collect, analyze, and interpret data related to skill gaps in particular context.

VIII.2 Research Method: Research methods are the strategies, process or tools used in the collection of data or evidence which is further used for analysis in order to identify new information or helps in more in depth understanding of a topic.

VIII.3 Research Design: The present research is the combination of exploratory as well as descriptive research design. Research design is the specification of methods and procedure for acquiring information molded according to my research design unperformed following steps are:

VIII.4 Research Design:

Types:

A researcher has used qualitative as well as uantitative research designs in order to analys the descriptive and the numerical information collected through the appropriate data collecting methods which are mentioned earlier in this paper.

i) **Qualitative Research:** This methods is used by the researcher to evaluate and analyse the nonnumerical data collected during the study period. Mathematical calculations and statistical methods can be applied in this as the data is in the descriptive form. Statistical and statistical methods can help to accept or reject the assumptions that resetcher have in his mind. These are more reliable methods for the researcher as they help to understand that why a particular theory exist and —what respondents have to say about it.

ii) Quantitative Research: It is applicable more for situations where statistical analysis from the data collected in the number form is essential. Numeric evaluation are more reliable and critical business decisions can be taken based on this analysis. For making effective decision making for the future of the business decisions drawan from complex numerical data prove to be more effective.

IX DATA:

Data is any information that has been collected, observed, generated or created to validated original research findings.

Types of Data

Primary Data:

A data collected for the first time by the researcher is described on Primary data. This data is collected directly from the data source. Primary data is more reliable as it is fresh, original, and authentic

Primery data is pertaining to the purpose of the study and the perspectives of the study does not miss. In this paper survey of employees of the retail company is conducted in order to collect the primary data.

Secondary Data:

The existing data when used for the purpose of research is secondery data this data is generally collected by another person other than the researcher. This data that is readily available and does not require any special methods or extra efforts for data collection.

Example: company annual reports, sales records, internet information.

X DATA COLLECTION TECHNIQUES:

Both primary and secondary sources are used in this study. Se data has helped the researcher to get insights in to the problem. Primary data is collected by means of direct interviews and questionnaires secondary data is from available books, publications, research studies, articles and websites.

XI SAMPLING TECHNIQUES:

In survey method sampling is concerned with selection of respondents from the population from whom the researcher has collected the data. The sample is selected in such a way that it should represent the population.Simple random sampling method is used in this research for selecting the sample.

Sample Size:

The number of items which can be listed is a sample frame which also study equal to the population. For this study 60 employees is the sample frame. The sample size for this study is 40 respondents.

Research Tools:

Data source : Primary and Secondary data

Research Instrument : Questionnaires, Personal Interview

Type of Questionnaires: Close ended.

Sampling method: Simple random sampling

Sample Frame: 60 Employees.

Sample Size: 40

Target Group: Managerial Staff and Employees.

XII DATA ANALYSIS AND INTERPRETATION:-

Gender	No. of Respondent	Percentage
Female	27	67.5
Male	13	32.5
Total	40	100

Table 1 Gender Analysis:-

Particular	No. of Respondent	%age
Appreciate	2	5
Assistant Store Manger	3	7.5
DM	4	10
Employee	6	15
FA	5	12.5
HR	2	5
RG	2	5
Saksham Candidate	6	15
Staff NPT	5	12.5
Staff PT	5	12.5
Total	40	100

Table 2 Job Title:-

Table 3 Skills defined in the organization.

Particular	No. of Respondent	Percentage
Customer Service	14	35
Product Knowledge	5	12.5
Technology Skill	16	40
Cash Handling	3	7.5
Time Management	2	5
Total	40	100

Table 4 Competency in organization

Name	No. of Respondent	%
Adaptability to change	9	22.5
Employee Engagement	4	40
Satisfaction	7	17.5
Brand Reputation	18	45
Enhanced Customer Performance	2	5
Total	40	100



Table 5 Competency required in retail industry.

Name	No. of Respondent	Percentage
Customer Service Skill	10	25
Effective Communication	15	37.5
Inventory Management	5	12.5
Product Knowledge	10	25
Total	40	100

Graph 2 Key competencies in the organization



Name	No. of Respondent	Percentage
Store Manager	2	5
visual Merchandiser	15	37.5
HR Manager	20	50
Financial Controller	3	7.5
Total	40	100

Table 6 Key competencies in the organization

Table 7 Skills possessed by the employees:-

Name	No. of Respondent	Percentage
Salesmanship	2	5
Organizational Skills	12	30
Communication	20	50
Adaptability	6	15
Total	40	100



It is found that majority of the organization defined the skills of employees. They are soft as well as technology skill. The understanding of these skills helps to ensure that employees have the necessary skills in their roles and this will help to the organization succeed.

It is found that most of the respondent having a effective communication in retail industry. Communication is important part of retail industry, so it is required that organizations should provide a good level of training on communication skills and develop the competency of employees in this regard. It is found that the roles and skills pertaining to the roles need to be defined and are very essential and necessary in organization.

It is found that, the majority of respondents have product knowledge and offere service to the customers in excellent manner.

It is found that, the majority of employee want a possess of skill is communication. This indicates their willingness to learn and aquire the new skills.

XIII CONCLUSION:-

The findings underlines a number of possible solutions to the problems identified related to the skills of the employees. Firstly, it is important that retail is made a more attractive career choice for workers of all ages. So that the potential candidates with all possible skill set is attracted towards retail industry. The profile of the retail industry need to be elevated. In order to achive this a serious rebranding of the industry is to be done with the collective efforts. New marketing campaign can help is to raise the awareness and image of retailing as a first choice destination career.

Retail industry needs to illustrate and demostrate the variety of challenges and career paths open to potential employees. Additionally, recruitment processes need to become more professional to compete for and attract the best applicants for the various opportunities available in the industry.

Secondly, retail needs to become a priority for the aspirants and should not be a last option. Retail is labour oriented business and accounts for a large percentage of the overall workforce, therefore it is imperative that essential skills for retail are taught to potential employees.

Appropriate training sessions can help the organisations fill the skill gap as well as sharpen the skill set possesed by the employees. Such training facilities need to be made accessible to the employees of the organisation.

XIV REFERENCES:-

1. Smriti Asthana (2012), "Human Resource and Skill Gaps in Logistics Services Industry in India", International Journal of Research in IT & Management, Volume 2, Issue 2, PP 1018 – 1026.

2. Ran Singh Dhaliwal and Pankaj Misra (2020), "Employability Skills Needed in Hospitality Industry: A Scopious Review", Asian Journal of Education and Social Studies, Volume 10, Issue 1, PP 18-34.

3. Nikolina Trajanoska and Ninko Kostovski (2016), "Importance-Performance Gaps in Skills and Knowledge of Junior Management and Staff in Tourism and Hospitality Industry in Republic of Macedonia", Journal of Sustainable Development, Volume 5, Issue 14 PP 68 – 80.

4. S A Shrlekar, Marketing Management, ed. 13, Himalaya Publishing House Mumbai, 2007.

5. Shradhanjali (2016), "Soft Skills: An Overview and its Effective implementation", International

Journal of Latest Technology in Engineering, Management and Applied Science, Volume V, Issue V, PP 61 - 63.

6. Michele Bruni, Likanan Luch and Somean Kuoch (2013), "Skills shortages and skills gaps in the Cambodian labour market: Evidence from employer skills needs survey", ILO Asia-Pacific Working Paper Series, PP 1 - 102.

7. Marcel M. Robles (2012), "Executive Perceptions of the Top 10 Soft Skills Needed in Today's Workplace", Business and Professional Communication, quarterly Sage Journal, Business Communication Quarterly, Volume 75, Issue 4, PP 453 – 465.

Laura Antonucci and Francesco Domenico d'Ovidio (2012), "An Informative System Based on the Skill Gap Analysis to Planning Training Courses" Applied Mathematics, Volume 3, PP 161

THE NEUROSCIENCE OF PUBLIC RELATIONS WITH RESPECT TO HIGHER EDUCATIONAL INSTITUTIONS

Dr. Sangeeta Swamy¹ & Dr. Vageesha Salanki Mathada² ¹Public Relations Officer, e-mail:sangeetavageesh1234@gmail.com; ²Professor & Academic Dean, e-mail: vageeshsm@gmail.com College of Engineering and Research, Fabtech Educational Society, Sangola, Maharashtra

Abstract: Neuroscience forms the basis for a study on emotional aspects of humans, their decision making in relation to products and services, to make the marketing effort more effective. Effectiveness and efficiency are two hall marks of any management science. Neuromarketing is a way to predicting the success of a product/service, which is based on the brain's voluntary and involuntary responses. The principle can be used to any marketing aspect, including websites, digital marketing, print communication or in-person / face-to-face communication.

Nonetheless, public relations has to be based on a realistic an ethical foundation. Discussions have focused on the use of journalistic ethics over advocacy ethics and vice-versa. On the other hand, the two-way symmetrical model of ethical public relations seeks to build a consensus that the organization itself has to change in building a productive relationship. From historical times, it is known that the Isocrates' symmetrical rhetoric was more effective than advocacy theory rivals. Studies show that the two-way symmetrical model may be more effective for public relations.

In the context of the above, neuromarketing assumes that decision making is largely emotional and unconscious. The human brain responds to sensory stimuli viz., sound, sight, touch, smell and taste. Successful brands create a subconscious link between certain visual cues, sounds or even smell and stir feelings and memories that impact decision making. Consequently, neuromarketing and neuroscience use a principle known as 'fluency': the human brain processes familiar things in an easy manner.

Lately, neuroscience has opened opportunities in the field of Public Relations (PR). Practitioners of PR can now envision how the convergence of technology, biometrics, measuring techniques from neuroscience, and traditional marketing practices, might benefit their clients. Nonetheless, there are risks involved if practising professionals do not understand the impact and level of influence of neuroscience in Public Relations.

Educational institutions use a wide range of promotional techniques in the process of advertising their educational programs, both at the UG and PG levels. Competition in this field is being experienced like never before. Educational institutions are creating an awareness for themselves on the "touch points" that will "woo" student-groups be interested in their programs.

Hence, this paper seeks to inquire the value that PR will gain from the field of neuroscience and neuromarketing techniques with respect to educational institutions.

Key words: Neuroscience, Neuromarketing, Public Relations, Emotional aspect, Isocrates' symmetrical rhetoric, higher educational institutions

I INTRODUCTION

? How did this science of PR and Neuroscience develop? What are the foundational principles of the practice? Did it evolve from traditional marketing, neuroscience or a combination of the two? Are there any established best practices and standards that marketing and public relations professionals should follow?

These are some of the intriguing questions that seek to answer the relationship between neuroscience/marketing and PR.

Neuroscience/marketing offers a plethora of opportunities to the "savvy public relations practitioners" who look to ways on the integration of technology, biometrics, measuring techniques from neuroscience and traditional marketing practices that might benefit their clients. Nonetheless, there are inherent risks that may arise if neuroscience/marketing is not thoroughly understood, and if the impact and level of influence is not understood by the PR practitioners.

Hence, the question: What value does the PR profession can gain from neuroscience/marketing techniques?

II THE MESSAGE

How can the value of neuroscience/marketing be explained to all stakeholders, in a way that will not foster negative thoughts of subconscious messaging techniques?

Marketers and advertisers have been attempting to get to the subconsciousness that influence purchase/choice-decisions which have made the public to be wary of such attempts since they may not be very transparent. The present-day organizations are increasingly becoming conscious to be more ethical and transparent regarding their communication in a manner that their products/services will be acceptable to their audience, across geographies. Any compromise in this regard will sway the audience/public to alternative options.

III THE INFLUENCE OF THE CONSUMER

Specific techniques to reach the target audience/public may be drawn from the field of neuroscience/marketing to know the reasons as to why such people are influenced by specific messaging. They, the advertisers could use appropriate communication and messaging techniques that will possibly influence the target audience/consumers and for their better understanding too. There is a need to be alert to the subtlety in such kind of endeavors. The Public Relations Society of America (PRSA) emphasizes on the importance of defining PR: "recognize the need to understand the attitudes and values of [their public] — and to develop effective relationships with — many different stakeholders, such as employees, members, customers, local communities, shareholders and other institutions, and with society at large."

Hence, there is an imperative to determine the most effective ways of communicating and messaging: the ones that are strongest and elicit positive feedback from the audience consumers. The drivers that influence people in/with their decision making has thus to be ascertained. The field of PR is offering exciting opportunities with emerging practices and technologies

(social media) that will supplement the value-creating process in organizations. Nonetheless, the underlying imperative is to be ethical and transparent.

One such emerging area is the "neuromarketing", drawn from neuroscience, an area that has drawn inspiration for all the PR practitioners. This area seeks to elicit the right type of response form the audience/consumers. Organizations that use neuromarketing techniques have audience/consumers who remember the communication/messages, which will be evident in their ability to "recall a product/service with ease".

Therefore, the personnel who use neuroscience/marketing techniques must use appropriate action words that will elicit the desired response or a call to take action form the audience/consumers, which will drive the decision-making behavior.

The "advertisement copy" must seek to engage the audience/consumer in an imaginative manner that will lead to the desired action.

An anthropologist will most definitely vouch for such interventions, and will also advocate the impact of story-telling to be extremely influential. Therefore, the advertiser has to create a story and sell the product/service around that, as the story will be infused with mystique. Over a period of time, such story-telling must adopt various variants to be more impactful.

IV THE PROPOSITION

PR strategies are potentially influenced by neuroscience/marketing techniques since the advertiser seeks to have a better understanding of the audience/consumer. This becomes his/her ultimate objective, of understanding them better than before. So, such PR strategies must be fine-tuned to get the message across, effectively and well.

V NANO TOOL

With the advent of social and digital media, the real-world-face-to-face communication with eye-contact, body language, etc have become extinct, due to which the communication/message is/are failing to create the desired impact. Neuroscience research has uncovered specific ways that one can fine-tune the message — whether it's giving performance feedback, persuading the team to embrace a change initiative, or selling a product or service — to make sure others listen, attend to the message, and act on it.

VI THE RECOMMENDATION AND PRINCIPLES

A time-tested and effective way for connecting with the audience/consumers is to establish eye-contact and mirroring the audience/consumer being alive and miming the audience/consumer. This will potentially result in a synchronization of the brain waves that will lead to a positive engagement, good rapport which will lead to a positive feedback/decision-making.

Behavior Modification Research co-authored by Emily Falk, a Professor of Communication, Psychology, and Marketing at the University of Pennsylvania's Annenberg School for Communication, shows that difficult messages are met with a natural selfdefensive impulse that can block understanding. To combat that, Falk and fellow researchers asked people to think about things that mattered most to them (like family, friends, or spirituality) or to repeatedly wish for the well-being of other people (both close contacts and strangers).

VII KEEP IT SIMPLE

Brain synchrony is hindered by complexity. Hence, the need to keep the communication/message simple. Research on movie trailers by Moran Cerf at Northwestern's Kellogg School of Management found that the most effective ones, in terms of creating brain synchrony and driving ticket sales, had the fewest spoken words and the fewest faces and other stimuli on the screen. Simple communication/messages are easy to understand, and more likely it is to be remembered. In the book *Made to Stick*, researchers Chip and Dan Heath, offer six principles for making ideas understood and remembered, and the first is simplicity. But, they argue, it's not just "short and sweet." The Heaths say it's about communicating an essential core, something that's both simple and profound.

VIII THE PITCH

Wharton's Jonah Berger and Alex Van Zant of Rutgers Business School found that speaking slightly louder than normal and varying your overall volume increases the perception of your confidence, which in turn makes you more persuasive. "There's work that shows people seem more human when we hear their voice. We give them more sense of mind, we think of them more as real people when they use their voice. Our research also suggests it can make people more persuasive," writes Berger.

IX PROVIDE FEEDBACK

A study led by Tessa West and Katherine Thorson of New York University used heart-rate monitors on subjects while they were engaged in giving or receiving unprompted feedback. Although the participants were uncomfortable asking for feedback, doing so helped equip them to hear and process it. Their innate "threat response" was lower. Before you deliver some potentially hard-to-hear feedback, schedule a meeting and ask the employee to evaluate you and how well you are supporting him or her in their development and current role. It's vital that you listen uncritically, take your time responding, and even use mirroring by repeating the feedback in your own words ("Let me make sure I understand what you said").

X THE PRACTICALITY IN COMPANIES

Audience is always a core value of the company/institution. So, avoiding difficult conversation is imperative to success. Focus may also be on how the audience contributes to the greater good of the organization, including participating in a number of organizational initiatives.

It is known that the world's simplest organizations have the most loyal customers, and consequently a higher revenue. Organizations may encourage honest opinion(s) from the customers and hence will also have higher engagement-levels with them.

XI THE FINAL THOUGHT

The PR professionals need to do is see the bigger picture of what is happening in the marketing profession, and recognize the niche for the discipline, and look to see how the brain reacts to specific crisis communication messages, and how communicators can adapt their messages to make sure that they are the most effective. The Higher Educational Institutions can adopt the same principle in marketing their educational programs, depending upon the location, age, gender, and the specialisms offered, and keep changing the approach from time-to-time. Neuroscience/marketing in PR is, thus, an emerging area that seeks to identify all the elements of the brain scientifically and correctly to benefit the stakeholders.

XII BIBLIOGRAPHY

- 1. The Public Relations Society of America, Inc., NY, USA
- Baek, E., Scholz, C., & Falk, E. B. (2020). <u>The neuroscience of persuasion and information propagation: The key</u> role of the mentalizing system. In K. Floyd & R. Weber (Eds.), *Handbook of Communication Science and Biology* (pp. 122-133). *Routledge*.
- 3. Milosavljevic, Milica and <u>Moran Cerf</u>, (2018). "<u>What matters is attention not intention: Insights from computational</u> <u>neuroscience of vision.</u>" International Journal of Advertising, Vol. 7, Issue 3.
- 4. Chip & Dan Heath, (2007). Why some ideas survive and others die? Random House.
- Berger, Jonah and Van Zant, Alex. (2019). How the Voice Persuades. Journal of Personality and Social Psychology, NY, USA. Online First Publication, June 13, 2019. <u>http://dx.doi.org/10.1037/pspi0000193</u>
- 6. West, Tessa, and Thorson, Katherine. (2015). <u>Stereotype-based stress contagion in dyadic interactions</u>. Paper presented at the annual meeting of the Society for Experimental Social Psychology, Denver CO.

NCSTEM 2023

This book is organized into four distinct sections, each dedicated to a specific field: Computer Science and Engineering, Electrical Engineering Electronics and Telecommunication, Master of Business Administration. These sections comprehensively cover a wide array of topics within Science, Technology, and Management. The overarching goal of this book is to inspire and ignite a passion for research culture among students and faculty members across these diverse streams. It aims to encourage and empower individuals, fostering an environment that cultivates curiosity, innovation, and scholarly pursuits.

SVERI, a charitable Trust, embarked on its ambitious missions on a rural development project, has established its first ideal Engineering College in Pandharpur. The college is approved by All India Council for Technical Education (AICTE), New Delhi and the Government of Maharashtra. The Engineering College has achieved remarkable recognition, being accredited by the esteemed National Assessment and Accreditation Council (NAAC) with A+ Grade having 3.46 CGPA. Along with the ISO 9001:2015 certification. Furthermore, the college has gained recognition from the Institution of Engineers (India), Kolkata, further validating its academic excellence and commitment to engineering education. The college offers a diverse range of courses, including B.Tech programs in Mechanical Engineering, Computer Science & Engineering, Electronics & Tele-communication Engineering, Electrical Engineering, and Civil Engineering. The institution also offers M.Tech programs in Mechanical (Design Engineering), Electronics & Tele-communication Engineering, Computer Science & Engineering, and Structural Engineering.

Additionally, the college provides Master of Business Administration (MBA) and Master in Computer Applications (MCA) programs, offering a comprehensive academic experience across various disciplines. The Institute is recognized by Punyashlok Ahilyadevi Holkar Solapur University Solapur, allowing candidates to pursue doctoral programs.